

PROCEEDINGS 2016

27 - 29 OCTOBER 2016 STONEHENGE IN AFRICA NORTH WEST



THROUGH THE LOOKING GLASS





Proceedings of the 27h Annual Southern African Institute for Industrial Engineering Conference

27 - 29 October 2016

Stonehenge in Africa Parys, North-West Province, South Africa

> ISBN: 978-1-86822-671-9 ECSA CPD Nr: SAIIE / CPD / I / 06-16

Editor: Prof L Van Dyk, North-West University, South Africa Contents available on the conference website:www.saiie.co.za/SAIIE27 Published by: Southern African Institute for Industrial Engineering (SAIIE) - see www.saiie.co.za



SAIIE Copyright © 2016



PREFACE



The vision of the Southern African Institute for Industrial Engineering (SAIIE) states: "We are a vibrant, learned society, representing and promoting Industrial Engineering in Southern Africa". SAIIE's annual conference epitomizes this vision by bringing together academics and professionals from all over the country and abroad to share best practices and new knowledge concerning Industrial Engineering.

The first call for papers was sent out almost a year ago. About 10% of the initial 163 responding authors opted from the onset for the *Abstract and Presentation Only* track, which does not require a full paper. The remaining 90% of submissions were made to the *Abstract and Full Peer-reviewed Paper* track.

These papers were provisionally approved on the basis of an abstract, followed by a doubleblind peer review process.

Of the 163 initial abstracts, 68 full papers successfully passed the double-blind peer review process. Of these 68 papers, 52 papers appear in these proceedings. The remaining 16 papers were diverted to a special edition of the *South African Journal for Industrial Engineering* (SAJIE).

All review comments and editorial decisions taken during the process have been recorded and can be traced by means of the online conference management system used for this conference. In an effort to avoid any possible conflict of interest, the reviewers were usually not from the same institution as the author to whose papers they were allocated. When the reviewers recommended minor improvements, the final checks were performed by only one of the two reviewers and the programme director. When major improvements were required, both reviewers were involved in a second (and in some cases, a third) review round. All papers published in these proceedings passed this double peer-reviewed process.

This conference has culminated in three outputs:

- 1. The **conference proceedings** (this document) is an electronic document that will be distributed on USB Flash drives to all delegates. It contains the full-length papers that were submitted, reviewed and approved for the "Full Peer Reviewed Track", excluding those diverted to the special edition of SAJIE. The proceedings are also available online to ensure that it remains accessible and indexed by scholarly search engines.
- 2. An abstract of each conference presentation is printed in the **conference programme and abstract book**. This includes papers printed in these proceedings, those diverted to the special edition of SAJIE, as well as all the other non-peer-reviewed submissions (presentations, workshops and keynote speakers).
- 3. Sixteen papers from this conference will be published in a special edition of SAJIE.

We trust that you will enjoy interacting with the authors of the papers that appear in these proceedings and other delegates of the 27th Annual SAIIE Conference, and, in doing so, being part of this "vibrant and learned society, presenting and promoting Industrial Engineering in South Africa".

Prof Liezl van Dyk Faculty of Engineering, North-West University Proceedings Editor October 2016





ORGANISING TEAM

ORGANISING COMMITTEE

Liezl van Dyk Hannelie Nel

Jacques Fauré Lynette Pieterse Thereza Botha

EDITORIAL COMMITTEE

Prof Liezl Van Dyk Prof Hannelie Nel Faculty of Engineering North-West University, Potchefstroom University of Johannesburg, Johannesburg/ North-West University, Potchefstroom Fortna, Johannesburg SAIIE and SAJIE Administrator, Pretoria TechnoScene (Pty) Ltd (conference organiser), Pretoria

Faculty of Engineering North-West University, South Africa University of Johannesburg, Johannesburg/North-West University, Potchefstroom SAJIE Editor (SAIIE27 Shortlist Selection) Stellenbosch University, South Africa

Prof Corné Schutte

PEER REVIEW PANEL

The Editorial Committee would like to acknowledge the following reviewers who all contributed to the reviewing process:

Themba Amukelani Baloyi	University of Johannesburg, South Africa
Wouter Bam	Stellenbosch University, South Africa
Louzanne Bam	Stellenbosch University, South Africa
James Bekker	Stellenbosch University, South Africa
Prof Thomas Bobga	Vaal University of Technology, South Africa
Prof Alan Brent	Stellenbosch University, South Africa
Rojanette Coetzee	North-West University, South Africa
Pieter Conradie	Fourier Approach. South Africa
Hasan Darwish	North-West University, South Africa
Edward Davies	Nelson Mandela Metropolitan University, South Africa
Prof Deon Johan de Beer	North-West University, South Africa a
Dr Marné de Vries	University of Pretoria, South Africa
Joubert de Wet	North-West University, South Africa
Mendon Dewa	Durban University of Technology, South Africa
Dr Anton du Plessis	Stellenbosch University, South Africa
Partson Dube	University of Johannesburg, South Africa
Lilian Ganduri	Stellenbosch University, South Africa
Talon Garikayi	Stellenbosch University, South Africa
Charles Harebottle	EOH, South Africa
Dieter Hartmann	University of the Witwatersrand South Africa
Teresa Hattingh	University of the Witwatersrand, Johannesburg
Prof Alwyn Hoffman	North-West University, South Africa
Prof Johann Holm	North-West University, South Africa
Dr Christianah Olakitan Ijagbemi	Tshwane University of Technology, South Africa
Wyhan Jooste	Stellenbosch University, South Africa
Dr Grace Mukondeleli Kanakana	Tshwane University of Technology, South Africa
Denzil Kennon	Stellenbosch University, South Africa
Gerrit Kotze	Sasol Mining, South Africa
Willie Krause	Stellenbosch University, South Africa
Prof David Kruger	University of South Africa, South Africa
Mr Louis Krüger	University of Pretoria, South Africa



- Dr Ann S Lourens Whisper Maisiri Dr Stephen Matope Kenneth Moodlev Mphegolle Ephraim Moshidi Miss Zanele Promise Mpanza Mr Tshepo Mpshe Dr Michael Mutingi Andrew Kisten Naicker Prof Hannelie Nel Lungile Nyanga Dr Gert Adriaan Oosthuizen **Prof Tinus Pretorius** Prof Leon Pretorius Dr Kemlall Ramdass Prof Corné Schutte
- Kgomotso Simango Liezl Smith Henrietta (Rita) Steenkamp Bernadette Patricia Sunjka Prof Fanie Terblanche Carin Tredoux Prof Herman van der Merwe Prof Andre Francois van der Merwe Dr Karl Robert van der Merwe Prof Liezl van Dvk Joubert van Eeden Quintin Van Heerden Dr Johann van Rensburg Dr Chris van Schoor Prof Jan H van Vuuren Maria Van Zyl Prof Jacobus Krige Visser Prof PJ Vlok

Konrad von Leipzig

Nelson Mandela Metropolitan University, South Africa North-West University, South Africa Stellenbosch University, South Africa South Africa North-West University, South Africa Council for Scientific and Industrial Research, South Africa South Africa Namibia University of Science & Technology, Namibia Durban University of Technology, South Africa South Africa Stellenbosch University, South Africa Stellenbosch University, South Africa University of Pretoria, South Africa University of Pretoria, South Africa University of South Africa, South Africa SAJIE Editor (SAIIE27 Shortlist Selection), Stellenbosch University, South Africa North-West University, South Africa University of Pretoria, South Africa University of Johannesburg, South Africa University of the Witwatersrand, Johannesburg North-West University Potchefstroom, South Africa South Africa North-West University, South Africa Stellenbosch University, South Africa Nelson Mandela Metropolitan University, South Africa North-West University, South Africa Stellenbosch University, South Africa South Africa Council for Scientific and Industrial Research, South Africa North-West University, South Africa 14BE, South Africa University, South AfricA Stellenbosch University, South Africa North-West University, South Africa University of Pretoria, South Africa Stellenbosch University/Gaussian/Asset Care Research Group, South Africa Stellenbosch University, South Africa



SAIIE27 would like to thank the following partners and exhibitors for their generous support







TABLE OF CONTENTS

Ref ¹	Paper Title with Authors	Page
2448	RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PROSTHETIC HANDS USING OPEN SOURCE DEVICES	1
	D Hagedorn-Hansen, L P Steenkamp, G A Oosthuizen, M M Jansen van Rhensburg	
2509	LEAN MANAGEMENT IN THE HEALTHCARE INDUSTRY	11
	A P Muthivhi, GM Kanakana, K Mpofu	
2544	VALUE STREAM MAPPING APPLIED TO A TYPICAL THREE YEAR HIGHER EDUCATION QUALIFICATION	23
	K R Van der Merwe	
2548	ENGINEERING MANAGEMENT AND BUSINESS INTELLIGENCE: THE IMPORTANCE OF PLANT DESIGN BASE	31
	H F Swanepoel, J H Wichers	
2550	SELECTION, TRANSFER AND ADOPTION OF NEW TECHNOLOGY IN THE MINING INDUSTRY	51
	F Tuta, M W Pretorius	
2562	CONCEPTUALISING GLOBAL MINERAL VALUE CHAINS: A TOOL FOR MINERAL POLICY DESIGN	61
	W G Bam	
2563	EVALUATION OF ELECTRICAL ENERGY CONSUMPTION BY A HOT-DIP GALVANISING PLANT	75
	M Dewa, B Dzwairo, B Nleya	
2564	A PROPOSED MODEL FOR SOUTH AFRICA TO EFFECTIVELY RECYCLE AND DISPOSE WASTE ELECTRICAL AND ELECTRONIC EQUIPMENT	87
	A J J Mouton, J H Wichers	
2565	BUSINESS INTELLIGENCE PERFORMANCE MEASURES IN LOCAL GOVERNMENT	103
	J J Fourie, J Stimie, PJ Vlok	
2570	IMPLEMENTATION OF FMECA APPROACH FOR RELIABILITY IMPROVEMENT ON DRUM PRODUCTION LINE	115
	M Dewa, N Luvuno	
2571	UNDERSTANDING THE REAL COST OF POOR QUALITY: CASE STUDY AT BOSCH SOUTH AFRICA (AUTOMOTIVE INDUSTRY)	127
	T Mpshe, G M Kanakana, J Trimble	
2576	EVALUATION OF WASTE-TO-ENERGY GRATE INCINERATION POWER PLANT DRIVERS AND BARRIERS FOR A SMALL SOUTH AFRICAN CITY: A SWOT ANALYSIS APPROACH	139
	W Maisiri, L Van Dyk	
2577	RESPONSIVE STRATEGIES AND SUPPLY CHAIN DECISION-MAKING FOR THE SOUTH AFRICAN WINE INDUSTRY	151
	L Knoblauch, J Van Eeden, R Edwards	
2578	DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL OF INJECTION MOULDS	163
	M T Dewa, A F Van der Merwe, S Matope, L Nyanga	

¹The Reference Number is a unique paper reference that is used throughout the conference to identify the paper. This is also used as a page number prefix in the proceedings, and papers are sorted according to this number in the proceedings.



REF ¹	Paper Title with Authors	Page
2581	DEVELOPMENT OF AN MHEALTH REHABILITATION ACTIVITY MONITORING SYSTEM FOR TRANSTIBIAL AMPUTEES	177
	T Garikayi, D Van den Heever, S Matope	
2596	IMPACT OF MANUFACTURING STRATEGY ON OPERATIONAL PERFORMANCE: A CASE STUDY OF ZIMBABWEAN MANUFACTURING INDUSTRY	191
	B Chindondon, S Chindondondo	
2601	A SYSTEMATIC COMPARISON OF DONOR FUNDED SUPPLY CHAIN AND COMMERCIAL SUPPLY CHAIN CHARACTERISTICS	201
	D Lingervelder, L Bam, W G Bam	
2604	AN EVALUATION OF WASTE AND COSTS AT A TEXTILE FACILITY: A CASE STUDY K Ramdass	213
2608	EMPLOYEE RETENTION STRATEGIES: FACTORS FOR GENERATION Y BURSAR GRADUATES B Tladi	223
2611	DESIGNING A MOBILE LABORATORY SOLUTION FOR RAPID TUBERCULOSIS TESTING BY MEANS OF THE CEPHEID® GENEXPERT® DEVICE	239
	M Hattingh, L Bam	
2613	REDUCING THE COMMUNICATION COSTS OF A REMOTE MONITORING AND MAINTENANCE SYSTEM FOR AN ENERGY SERVICES COMPANY (ESCo)	253
	J N Du Plessis, J Prinsloo, J Vosloo	
2619	AN EVALUATION OF PROJECT TEAM WORK: TEAM LEADERS' PERSPECTIVES	263
2620	DEVELOPMENT AND IMPLEMENTATION OF AN ADVANCED MOBILE DATA COLLECTION SYSTEM	271
	I M Prinsloo, J N Du Plessis, J C Vosloo	
2622	ENGINEERING THE FACE OF THE FUTURE FACTORY	283
	J Durbach, D Hartmann, T Hattingh	
2630	DEVELOPMENT OF A MODEL FOR ROAD TRANSPORT FUEL MANAGEMENT M Van der Westhuizen, A Hoffman	293
2639	DEVELOPMENT OF A DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY OF CAPITAL EQUIPMENT FOR A SOUTH AFRICAN RETAILER	307
	H S Swart, J L Jooste, D L van Blommestein	
2643	EVALUATING LEAN IMPLEMENTATION IN SOUTH AFRICAN CASTING FOUNDRIES Y N Mawane, G Muyengwa	321
2647	TECHNOLOGY TRANSFER TO UPCOMING COMMERCIAL - COTTON FARMERS IN THE MAKHATHINI REGION	335
	J J Mashala	
2657	PREVENTATIVE MAINTENANCE IN PUBLIC HOSPITALS H Steenkamp	345
2659	AN EVALUATION AND COMPARISON OF BUSINESS SIMULATION SOFTWARE J K Visser	353
2661	IDENTIFYING BARRIERS FACED BY KEY ROLE PLAYERS IN THE SOUTH AFRICAN MANGANESE INDUSTRY	365
	H J Van Zyl, W G Bam, J Steenkamp	
2666	SUPPLY CHAIN MANAGEMENT: PERFORMANCE BENCHMARKS FOR SOUTH AFRICAN COMPANIES	377
	L Smith, E Van der Lingen	
2000	AFRICAN COMPANIES L Smith, E Van der Lingen	37



REF ¹	Paper Title with Authors	Page
2671	DEVELOPMENT OF A WAREHOUSE MANAGEMENT MATURITY MODEL FRAMEWORK FOR THE SOUTH AFRICAN WAREHOUSING ENVIRONMENT	393
	J Pillay, S Grobbelaar, K Von Leipzig	
2673	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	405
	J F Oberholzer, G A Oosthuizen, P De Wet, M D Burger, C I Ras	
2684	FAILURE ANALYSIS OF LOCOMOTIVE ALTERNATORS USING THE SIX SIGMA METHODOLOGY	417
	A Mekwa, M G Kanakana, K Mpofu	
2686	STRUCTURING DATA FOR A RSA SECTION 12L ENERGY EFFICIENCY TAX INCENTIVE APPLICATION	435
	H M Janse van Rensburg, W Booysen, S W Van Heerden	
2688	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	449
	L Nyanga, A F Van der Merwe, N Masanganise, S Matope, M T Dewa	
2690	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	461
	L Nyanga, A F Van der Merwe, T Musiyarira, S Matope, M T Dewa	
2692	DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE	477
	T Munetsiwa, L Masiyazi, M T Dewa, D Museka	
2693	THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY	489
	T D Mallane, T B Tengen	
2694	PHYSICAL ASSET MANAGEMENT MATURITY IN MINING: A CASE STUDY	493
	B J Mona, B P Sunjka	
2702	THE EFFECTIVENESS OF THE INTERNAL QUALITY AUDITING AT A CABLE MANUFACTURING COMPANY IN SOUTH AFRICA	505
	F Chiromo, F Ngobeni	
2734	PRACTICAL IMPLEMENTATION OF THE ISO 50 001 STANDARD FOR LARGE INDUSTRIES	515
	M Van Heerden, M J Mathews, J H Marais	
2737	THE IMPACT OF MAJOR ENVIROMENTAL, SOCIAL AND ECONOMIC FORCES ON THE FIELD OF INDUSTRIAL ENIGINEERING	525
	H Darwish	
2738	BENCHMARK INDICATORS FOR COMPETITIVENESS ANALYSIS OF TIER 2 AUTOMOTIVE INDUSTRY	537
	O T Laseinde, G M Kanakana	
2752	A DYNAMIC OPTIMAL CONTROL SYSTEM FOR COMPLEX COMPRESSED AIR NETWORKS	545
	S W van Heerden, J N Du Plessis, J H Marais	
2867	AN INVESTIGATION INTO THE CONTRIBUTION OF VARIOUS HUMAN FACTORS TOWARDS THE SUCCESSFUL OUTCOME OF A COMPLEX WEAPON SYSTEM ACQUISITION PROCESS	557
	G D P Pretorius, N D Du Preez, L Louw	
2876	APPLICATION OF BERNOULLI PRINCIPLE IN ADDRESSING BOTTLENECKS AT A LOCAL PRODUCTION COMPANY IN THE VAAL	573
	T B Tengen	



REF ¹	Paper Title with Authors	Page
2878	CRITICAL SUCCESS FACTORS FOR IMPLEMENTING A LABOUR PRODUCTIVITY IMPROVEMENT INITIATIVE IN A COMPETITIVE SOUTH AFRICAN MANUFACTURING PLANT FOR GREATER INTERNATIONAL COMPETITIVENESS	579
	R N Govender, B Emwanu	
2880	FACTORS AFFECTING THE QUALITY OF SERVICE DELIVERY IN A GOVERNMENT DEPARTMENT OVERSEEING EFFICIENT MAINTENANCE OF PHYSICAL ASSETS IN HEALTHCARE SERVICES IN A SELECTED PROVINCE IN SOUTH AFRICA	591
	T Taruvinga, B Emwanu	
2886	ENGINEERING WORK INTEGRATED LEARNING: A CASE STUDY IN PROBLEM-BASED RESEARCH AND DEVELOPMENT PROJECTS	599
	M Della Tamin, J Meyer, H Nel	
2921	ASSESSMENT OF CONTINUOUS IMPROVEMENT INITIATIVES' IMPACT WITHIN THE MANUFACTURING SECTOR IN SOUTH AFRICA	609
	S Theko, G M Kanakana, K Mpofu	



ALPHABETIC AUTHOR INDEX

Author	Paper Title	Ref ²
A P Muthivhi, A P	LEAN MANAGEMENT IN THE HEALTHCARE INDUSTRY	2509
Bam, L	DESIGNING A MOBILE LABORATORY SOLUTION FOR RAPID TUBERCULOSIS TESTING BY MEANS OF THE CEPHEID® GENEXPERT® DEVICE	2611
Bam, L	A SYSTEMATIC COMPARISON OF DONOR FUNDED SUPPLY CHAIN AND COMMERCIAL SUPPLY CHAIN CHARACTERISTICS	2601
Bam, W G	CONCEPTUALISING GLOBAL MINERAL VALUE CHAINS: A TOOL FOR MINERAL POLICY DESIGN	2562
Bam, W G	A SYSTEMATIC COMPARISON OF DONOR FUNDED SUPPLY CHAIN AND COMMERCIAL SUPPLY CHAIN CHARACTERISTICS	2601
Bam, W G	IDENTIFYING BARRIERS FACED BY KEY ROLE PLAYERS IN THE SOUTH AFRICAN MANGANESE INDUSTRY	2661
Booysen, W	STRUCTURING DATA FOR A RSA SECTION 12L ENERGY EFFICIENCY TAX INCENTIVE APPLICATION	2686
Burger, M D	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	2673
Chindondo, B	IMPACT OF MANUFACTURING STRATEGY ON OPERATIONAL PERFORMANCE: A CASE STUDY OF ZIMBABWEAN MANUFACTURING INDUSTRY	2596
Chindondo, S	IMPACT OF MANUFACTURING STRATEGY ON OPERATIONAL PERFORMANCE: A CASE STUDY OF ZIMBABWEAN MANUFACTURING INDUSTRY	2596
Chiromo, F	THE EFFECTIVENESS OF THE INTERNAL QUALITY AUDITING AT A CABLE MANUFACTURING COMPANY IN SOUTH AFRICA	2702
Darwish, H	THE IMPACT OF MAJOR ENVIROMENTAL, SOCIAL AND ECONOMIC FORCES ON THE FIELD OF INDUSTRIAL ENIGINEERING	2737
De Wet, P	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	2673
Della Tamin, M	ENGINEERING WORK INTEGRATED LEARNING: A CASE STUDY IN PROBLEM-BASED RESEARCH AND DEVELOPMENT PROJECTS	2886
Dewa, M	EVALUATION OF ELECTRICAL ENERGY CONSUMPTION BY A HOT-DIP GALVANISING PLANT	2563
Dewa, M	IMPLEMENTATION OF FMECA APPROACH FOR RELIABILITY IMPROVEMENT ON DRUM PRODUCTION LINE	2570
Dewa, M T	DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL OF INJECTION MOULDS	2578
Dewa, M T	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	2688
Dewa, M T	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	2690

²The Reference Number is a unique paper reference that is used throughout the conference to identify the paper. This is also used as a page number prefix in the proceedings, and papers are sorted according to this number in the proceedings.



Author	Paper Title	Ref ²
Dewa, M T	DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE	2692
Du Plessis, J N	REDUCING THE COMMUNICATION COSTS OF A REMOTE MONITORING AND MAINTENANCE SYSTEM FOR AN ENERGY SERVICES COMPANY (ESCo)	2613
Du Plessis, J N	DEVELOPMENT AND IMPLEMENTATION OF AN ADVANCED MOBILE DATA COLLECTION SYSTEM	2620
Du Plessis, J N	A DYNAMIC OPTIMAL CONTROL SYSTEM FOR COMPLEX COMPRESSED AIR NETWORKS	2752
Du Preez, N D	AN INVESTIGATION INTO THE CONTRIBUTION OF VARIOUS HUMAN FACTORS TOWARDS THE SUCCESSFUL OUTCOME OF A COMPLEX WEAPON SYSTEM ACQUISITION PROCESS	2867
Durbach, J	ENGINEERING THE FACE OF THE FUTURE FACTORY	2622
Dzwairo, B	EVALUATION OF ELECTRICAL ENERGY CONSUMPTION BY A HOT-DIP GALVANISING PLANT	2563
Edwards, R	RESPONSIVE STRATEGIES AND SUPPLY CHAIN DECISION- MAKING FOR THE SOUTH AFRICAN WINE INDUSTRY	2577
Emwanu, B	FACTORS AFFECTING THE QUALITY OF SERVICE DELIVERY IN A GOVERNMENT DEPARTMENT OVERSEEING EFFICIENT MAINTENANCE OF PHYSICAL ASSETS IN HEALTHCARE SERVICES IN A SELECTED PROVINCE IN SOUTH AFRICA	2880
Emwanu, B	CRITICAL SUCCESS FACTORS FOR IMPLEMENTING A LABOUR PRODUCTIVITY IMPROVEMENT INITIATIVE IN A COMPETITIVE SOUTH AFRICAN MANUFACTURING PLANT FOR GREATER INTERNATIONAL COMPETITIVENESS	2878
Fourie, J J	BUSINESS INTELLIGENCE PERFORMANCE MEASURES IN LOCAL GOVERNMENT	2565
Garikayi , T	DEVELOPMENT OF AN MHEALTH REHABILITATION ACTIVITY MONITORING SYSTEM FOR TRANSTIBIAL AMPUTEES	2581
Govender, R N	CRITICAL SUCCESS FACTORS FOR IMPLEMENTING A LABOUR PRODUCTIVITY IMPROVEMENT INITIATIVE IN A COMPETITIVE SOUTH AFRICAN MANUFACTURING PLANT FOR GREATER INTERNATIONAL COMPETITIVENESS	2878
Grobbelaar, S S	DEVELOPMENT OF A WAREHOUSE MANAGEMENT MATURITY MODEL FRAMEWORK FOR THE SOUTH AFRICAN WAREHOUSING ENVIRONMENT	2671
Hagedorn-Hansen, D	RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PROSTHETIC HANDS USING OPEN SOURCE DEVICES	2448
Hartmann, D	ENGINEERING THE FACE OF THE FUTURE FACTORY	2622
Hattingh, M	DESIGNING A MOBILE LABORATORY SOLUTION FOR RAPID TUBERCULOSIS TESTING BY MEANS OF THE CEPHEID® GENEXPERT® DEVICE	2611
Hattingh, T	ENGINEERING THE FACE OF THE FUTURE FACTORY	2622
Hoffman, A	DEVELOPMENT OF A MODEL FOR ROAD TRANSPORT FUEL MANAGEMENT	2630
Janse van Rensburg, H M	STRUCTURING DATA FOR A RSA SECTION 12L ENERGY EFFICIENCY TAX INCENTIVE APPLICATION	2686
Jansen van Rhensburg, M M	RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PROSTHETIC HANDS USING OPEN SOURCE DEVICES	2448



Author	Paper Title	Ref ²
Jooste, J L	DEVELOPMENT OF A DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY OF CAPITAL EQUIPMENT FOR A SOUTH AFRICAN RETAILER	2639
Kanakana, G M	LEAN MANAGEMENT IN THE HEALTHCARE INDUSTRY	2509
Kanakana, G M	UNDERSTANDING THE REAL COST OF POOR QUALITY: CASE STUDY AT BOSCH SOUTH AFRICA (AUTOMOTIVE INDUSTRY)	2571
Kanakana, G M	FAILURE ANALYSIS OF LOCOMOTIVE ALTERNATORS USING THE SIX SIGMA METHODOLOGY	2684
Kanakana, G M	BENCHMARK INDICATORS FOR COMPETITIVENESS ANALYSIS OF TIER 2 AUTOMOTIVE INDUSTRY	2738
Kanakana, G M	ASSESSMENT OF CONTINUOUS IMPROVEMENT INITIATIVES' IMPACT WITHIN THE MANUFACTURING SECTOR IN SOUTH AFRICA	2921
Knoblauch, L	RESPONSIVE STRATEGIES AND SUPPLY CHAIN DECISION- MAKING FOR THE SOUTH AFRICAN WINE INDUSTRY	2577
Laseinde, O T	BENCHMARK INDICATORS FOR COMPETITIVENESS ANALYSIS OF TIER 2 AUTOMOTIVE INDUSTRY	2738
Lingervelder, D	A SYSTEMATIC COMPARISON OF DONOR FUNDED SUPPLY CHAIN AND COMMERCIAL SUPPLY CHAIN CHARACTERISTICS	2601
Lourens, A	AN EVALUATION OF PROJECT TEAM WORK: TEAM LEADERS' PERSPECTIVES	2619
Louw, L	AN INVESTIGATION INTO THE CONTRIBUTION OF VARIOUS HUMAN FACTORS TOWARDS THE SUCCESSFUL OUTCOME OF A COMPLEX WEAPON SYSTEM ACQUISITION PROCESS	2867
Luvuno, N	IMPLEMENTATION OF FMECA APPROACH FOR RELIABILITY IMPROVEMENT ON DRUM PRODUCTION LINE	2570
Maisiri, W	EVALUATION OF WASTE-TO-ENERGY GRATE INCINERATION POWER PLANT DRIVERS AND BARRIERS FOR A SMALL SOUTH AFRICAN CITY: A SWOT ANALYSIS APPROACH	2576
Mallane, T D	THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY	2693
Mallane, T D	THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY	2693
Marais, J H	PRACTICAL IMPLEMENTATION OF THE ISO 50 001 STANDARD FOR LARGE INDUSTRIES	2734
Marais, J H	A DYNAMIC OPTIMAL CONTROL SYSTEM FOR COMPLEX COMPRESSED AIR NETWORKS	2752
Masanganise, N	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	2688
Mashala, J J	TECHNOLOGY TRANSFER TO UPCOMING COMMERCIAL - COTTON FARMERS IN THE MAKHATHINI REGION	2647
Masiyazi, L	DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE	2692
Mathews, M J	PRACTICAL IMPLEMENTATION OF THE ISO 50 001 STANDARD FOR LARGE INDUSTRIES	2734
Matope, S	DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL OF INJECTION MOULDS	2578



Author	Paper Title	Ref ²
Matope, S	DEVELOPMENT OF AN MHEALTH REHABILITATION ACTIVITY MONITORING SYSTEM FOR TRANSTIBIAL AMPUTEES	2581
Matope, S	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	2688
Matope, S	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	2690
Mawane, Y N	EVALUATING LEAN IMPLEMENTATION IN SOUTH AFRICAN CASTING FOUNDRIES	2643
Mekwa, A	FAILURE ANALYSIS OF LOCOMOTIVE ALTERNATORS USING THE SIX SIGMA METHODOLOGY	2684
Meyer, J	ENGINEERING WORK INTEGRATED LEARNING: A CASE STUDY IN PROBLEM-BASED RESEARCH AND DEVELOPMENT PROJECTS	2886
Mona, B J	PHYSICAL ASSET MANAGEMENT MATURITY IN MINING: A CASE STUDY	2694
Mouton, A J J	A PROPOSED MODEL FOR SOUTH AFRICA TO EFFECTIVELY RECYCLE AND DISPOSE WASTE ELECTRICAL AND ELECTRONIC EQUIPMENT	2564
Mpofu, K	LEAN MANAGEMENT IN THE HEALTHCARE INDUSTRY	2509
Mpofu, K	FAILURE ANALYSIS OF LOCOMOTIVE ALTERNATORS USING THE SIX SIGMA METHODOLOGY	2684
Mpofu, K	ASSESSMENT OF CONTINUOUS IMPROVEMENT INITIATIVES' IMPACT WITHIN THE MANUFACTURING SECTOR IN SOUTH AFRICA	2921
Mpshe, T	UNDERSTANDING THE REAL COST OF POOR QUALITY: CASE STUDY AT BOSCH SOUTH AFRICA (AUTOMOTIVE INDUSTRY)	2571
Munetsiwa, T	DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE	2692
Murray, A	AN EVALUATION OF PROJECT TEAM WORK: TEAM LEADERS' PERSPECTIVES	2619
Museka, D	DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE	2692
Musiyarira, T	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	2690
Muyengwa, G	EVALUATING LEAN IMPLEMENTATION IN SOUTH AFRICAN CASTING FOUNDRIES	2643
Nel, H	ENGINEERING WORK INTEGRATED LEARNING: A CASE STUDY IN PROBLEM-BASED RESEARCH AND DEVELOPMENT PROJECTS	2886
Ngobeni, F	THE EFFECTIVENESS OF THE INTERNAL QUALITY AUDITING AT A CABLE MANUFACTURING COMPANY IN SOUTH AFRICA	2702
Nleya, B	EVALUATION OF ELECTRICAL ENERGY CONSUMPTION BY A HOT-DIP GALVANISING PLANT	2563
Nyanga, L	DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL OF INJECTION MOULDS	2578
Nyanga, L	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	2688
Nyanga, L	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	2690



Author	Paper Title	Ref ²
Oberholzer, J F	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	2673
Oosthuizen, G A	RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PROSTHETIC HANDS USING OPEN SOURCE DEVICES	2448
Oosthuizen, G A	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	2673
Peters, B D B	MANAGING CONTRACTOR CASH FLOW CONSTRAINTS DURING PROJECTS	2262
Pillay, J	DEVELOPMENT OF A WAREHOUSE MANAGEMENT MATURITY MODEL FRAMEWORK FOR THE SOUTH AFRICAN WAREHOUSING ENVIRONMENT	2671
Pretorius, G D P	AN INVESTIGATION INTO THE CONTRIBUTION OF VARIOUS HUMAN FACTORS TOWARDS THE SUCCESSFUL OUTCOME OF A COMPLEX WEAPON SYSTEM ACQUISITION PROCESS	2867
Pretorius, M W	SELECTION, TRANSFER AND ADOPTION OF NEW TECHNOLOGY IN THE MINING INDUSTRY	2550
Prinsloo, I M	DEVELOPMENT AND IMPLEMENTATION OF AN ADVANCED MOBILE DATA COLLECTION SYSTEM	2620
Prinsloo, J	REDUCING THE COMMUNICATION COSTS OF A REMOTE MONITORING AND MAINTENANCE SYSTEM FOR AN ENERGY SERVICES COMPANY (ESCo)	2613
Ramdass, K	AN EVALUATION OF WASTE AND COSTS AT A TEXTILE FACILITY: A CASE STUDY	2604
Ras, C I	SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA	2673
Schoeman, M	MANAGING CONTRACTOR CASH FLOW CONSTRAINTS DURING PROJECTS	2262
Smith, L	SUPPLY CHAIN MANAGEMENT: PERFORMANCE BENCHMARKS FOR SOUTH AFRICAN COMPANIES	2666
Steenkamp, H	PREVENTATIVE MAINTENANCE IN PUBLIC HOSPITALS	2657
Steenkamp, J	IDENTIFYING BARRIERS FACED BY KEY ROLE PLAYERS IN THE SOUTH AFRICAN MANGANESE INDUSTRY	2661
Steenkamp, L P	RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PROSTHETIC HANDS USING OPEN SOURCE DEVICES	2448
Stimie, J	BUSINESS INTELLIGENCE PERFORMANCE MEASURES IN LOCAL GOVERNMENT	2565
Sunjka, B P	PHYSICAL ASSET MANAGEMENT MATURITY IN MINING: A CASE STUDY	2694
Swanepoel, H F	ENGINEERING MANAGEMENT AND BUSINESS INTELLIGENCE: THE IMPORTANCE OF PLANT DESIGN BASE	2548
Swart, H S	DEVELOPMENT OF A DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY OF CAPITAL EQUIPMENT FOR A SOUTH AFRICAN RETAILER	2639
Taruvinga, T	FACTORS AFFECTING THE QUALITY OF SERVICE DELIVERY IN A GOVERNMENT DEPARTMENT OVERSEEING EFFICIENT MAINTENANCE OF PHYSICAL ASSETS IN HEALTHCARE SERVICES IN A SELECTED PROVINCE IN SOUTH AFRICA	2880
Tengen, T B	THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY	2693
Tengen, T B	APPLICATION OF BERNOULLI PRINCIPLE IN ADDRESSING BOTTLENECKS AT A LOCAL PRODUCTION COMPANY IN THE VAAL	2876



Author	Paper Title	Ref ²
Tengen, T B	THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY	2693
Theko, S	ASSESSMENT OF CONTINUOUS IMPROVEMENT INITIATIVES' IMPACT WITHIN THE MANUFACTURING SECTOR IN SOUTH AFRICA	2921
Tladi, B	EMPLOYEE RETENTION STRATEGIES: FACTORS FOR GENERATION Y BURSAR GRADUATES	2608
Trimble, J	UNDERSTANDING THE REAL COST OF POOR QUALITY: CASE STUDY AT BOSCH SOUTH AFRICA (AUTOMOTIVE INDUSTRY)	2571
Tuta, F	SELECTION, TRANSFER AND ADOPTION OF NEW TECHNOLOGY IN THE MINING INDUSTRY	2550
Van Blommestein, D L	DEVELOPMENT OF A DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY OF CAPITAL EQUIPMENT FOR A SOUTH AFRICAN RETAILER	2639
Van den Heever, D	DEVELOPMENT OF AN MHEALTH REHABILITATION ACTIVITY MONITORING SYSTEM FOR TRANSTIBIAL AMPUTEES	2581
Van der Lingen, E	SUPPLY CHAIN MANAGEMENT: PERFORMANCE BENCHMARKS FOR SOUTH AFRICAN COMPANIES	2666
Van der Merwe, A F	DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL OF INJECTION MOULDS	2578
Van der Merwe, A F	DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL	2688
Van der Merwe, A F	DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES	2690
Van der Merwe, K R	VALUE STREAM MAPPING APPLIED TO A TYPICAL THREE YEAR HIGHER EDUCATION QUALIFICATION	2544
Van der Westhuizen, M	DEVELOPMENT OF A MODEL FOR ROAD TRANSPORT FUEL MANAGEMENT	2630
Van Dyk. L	EVALUATION OF WASTE-TO-ENERGY GRATE INCINERATION POWER PLANT DRIVERS AND BARRIERS FOR A SMALL SOUTH AFRICAN CITY: A SWOT ANALYSIS APPROACH	2576
Van Eeden, J	RESPONSIVE STRATEGIES AND SUPPLY CHAIN DECISION- MAKING FOR THE SOUTH AFRICAN WINE INDUSTRY	2577
Van Heerden, M	PRACTICAL IMPLEMENTATION OF THE ISO 50 001 STANDARD FOR LARGE INDUSTRIES	2734
Van Heerden, S W	A DYNAMIC OPTIMAL CONTROL SYSTEM FOR COMPLEX COMPRESSED AIR NETWORKS	2752
Van Heerden. S W	STRUCTURING DATA FOR A RSA SECTION 12L ENERGY EFFICIENCY TAX INCENTIVE APPLICATION	2686
Van Zyl, H J	IDENTIFYING BARRIERS FACED BY KEY ROLE PLAYERS IN THE SOUTH AFRICAN MANGANESE INDUSTRY	2661
Visser, J K	AN EVALUATION AND COMPARISON OF BUSINESS SIMULATION SOFTWARE	2659
Vlok, P J	BUSINESS INTELLIGENCE PERFORMANCE MEASURES IN LOCAL GOVERNMENT	2565
Von Leipzig, K	DEVELOPMENT OF A WAREHOUSE MANAGEMENT MATURITY MODEL FRAMEWORK FOR THE SOUTH AFRICAN WAREHOUSING ENVIRONMENT	2671



Author	Paper Title	Ref ²
Vosloo, J	REDUCING THE COMMUNICATION COSTS OF A REMOTE MONITORING AND MAINTENANCE SYSTEM FOR AN ENERGY SERVICES COMPANY (ESCo)	2613
Vosloo, J C	DEVELOPMENT AND IMPLEMENTATION OF AN ADVANCED MOBILE DATA COLLECTION SYSTEM	2620
Wichers, J H	A PROPOSED MODEL FOR SOUTH AFRICA TO EFFECTIVELY RECYCLE AND DISPOSE WASTE ELECTRICAL AND ELECTRONIC EQUIPMENT	2564
Wichers, J H	ENGINEERING MANAGEMENT AND BUSINESS INTELLIGENCE: THE IMPORTANCE OF PLANT DESIGN BASE	2548





RESOURCE EFFICIENT PROCESS CHAINS TO MANUFACTURE PATIENT SPECIFIC PROSTHETIC HANDS USING OPEN SOURCE DEVICES

D. Hagedorn-Hansen¹, L.P. Steenkamp¹, M.M. Jansen van Rhensburg¹ & G.A. Oosthuizen¹

¹ Stellenbosch Technology Centre STC-LAM, Department of Industrial Engineering University of Stellenbosch, South Africa devonh@sun.ac.za; 16051505@sun.ac.za; 16069641@sun.ac.za; tiaan@sun.ac.za

ABSTRACT

The success of 3D printed prosthetics over the last few years has made way for further research and development within the field to improve the functionality and aesthetics of low-cost prosthetics. An opportunity now exists to develop a product, similar to that of Robohand and e-Nable, but which uses myoelectric control rather than mechanical control. Online open source projects such as the Open Hand Project, e-Nable, Robohand and various other projects have, to date, successfully supplied low-cost prosthetics to thousands of users and enabled hundreds of people to make their own prosthetics. The next step in the 3D printing revolution is to create the same type of platform, but to improve the designs and their functionality by incorporating myoelectric technology and more lifelike aesthetics into the process chain. This study suggests a process chain that can be used to produce patient specific prosthetic hands with myoelectric control, a 3D printed skeleton structure, and silicone fingers. A time and cost analysis was also performed.



1. INTRODUCTION

Affordable healthcare is one of the needs in South Africa and there are an increasing number of prosthetics required, most of which are not very affordable for the masses [1]. In 2013, the International Society of Prosthetics and Orthotics estimated that, of the 32 million amputees in the world, 80 percent live in developing countries, and only 5 percent have been fitted with a prosthesis [2]. Medical prosthetics are devices that are located either inside or outside of the patient's body to perform a function either aesthetically or practically or both [3]. After the face, the hands are reported to be the principal representation of the self-image which is received by other individuals [4], [5]. Research [1] indicates that an individual's tendency to conceal an affected hand, due to embarrassment, makes them as functionally disabled as a scapula-thoracic amputee. Further research [4] indicated that improved aesthetic prostheses advocate greater psychological well-being. Due to limited funds and available skill in developing regions, people do not have access to quality and aesthetically appealing prostheses. If one had access to affordable prostheses, a life-like passive prosthesis is preferred in order to promote better psychological well-being [6].

Three dimensional printing has become a popular area of manufacturing and medical science. Institutes are now looking into the applications in the medical fields for 3D printers and scanners [7]. Currently the focus is on organ printing where a robotic bio-printer applies tissue, layer upon layer to a print bed where the organ constructs using self-assembling tissue [8]. There is also focus on the 3D printing of prosthetics and whether or not it is feasible to create low cost prostheses that can be donated to underprivileged amputees [9]-[12]. The success of 3D printed prosthetics over the last few years has made way for further research and development within the field to improve the functionality and aesthetics of low-cost prosthetics [9], [10]. The current traditional method of prosthesis fabrication is very costly and not very time efficient [11]. In order to be a worthwhile venture the benefits of the new technology should outweigh that of the traditional methods. The benefits of 3D printing is that the machines can print a prototype or mould in just a few hours and the machines can run nonstop if they have to [13]. This means less waiting time for the patient and less skills required to fabricate the prosthesis.

Open source technology is freely available to download and does not require one to buy the product or expensive licences. The mentality behind this technology is to create a platform for rapid development and to do so using minimal or shared resources [14]. Multiple online databases, or 3D warehouses as they are called, could be accessed to download designs, upload new models or collaborate. The idea of open design platforms is to change the way we construct knowledge around manufacturing, to be able to generate new designs and then be able to share it with others for improvements [15]. This new mentality is providing a new competitive edge to the manufacturing paradigm shift and has also resulted in the uprising of collaborative open source projects. Open source projects are generally crowd funded, collaborative projects where communities of people contribute to a specific cause in the form of finances, knowledge, design, ingenuity, feedback, or experience in order to create awareness and to make the project a success. Several open source projects for the development of prosthesis exist, such as e-Nable, Robohand, and the open hand project.

The 3D-printed prosthetic hands offered by open source projects e-Nable and Robohand are designed for users who still have a section of palm and a movable wrist or elbow. This type of disability can either be a birth defect similar to the user as shown in Figure 1(a), or a partial amputation due to an accident as shown in Figure 1(b). The movement or control of a mechanical prosthesis relies on the upward movement of the palm, section of palm by the wrist, or the elbow in order to bring about movement of the fingers.



Figure 1: (a) Infant hand showing birth defect where fingers are not developed [16] and (b) user with amputated fingers after accident [17]



The objective of this research was to develop a resource-efficient process chain for the manufacturing of aesthetically appealing patient specific prosthetic hands using additive manufacturing technologies with open-source myoelectric control.

2. LITERATURE STUDY

2.1 Open Community Manufacturing and Online open source prosthetic projects

Open community manufacturing (OCM) is a concept described by Oosthuizen et al. [18]. OCM uses open design platforms to create value in developing communities or the base of the pyramid (BoP). The BoP represents a socio-economic group of four billion people who live on less than US\$ 2.50 per day [10]. In South Africa more than half of the working population falls into the BoP category [18]. The OCM model could be used to manufacture prostheses in developing communities in order to grow the formal economy and mobilize entrepreneurs within the communities.

Organisations such as Open Hand Project, Enabling the Future, and Robohand operate open source projects in order to develop low cost prostheses for people in developing countries or for people who cannot afford the traditionally manufactured prostheses. These organisations use a similar model to the OCM model but they do not assist with the incubation of entrepreneurs. A design challenge is presented on a public forum by one of the organisations. The community of designers then collaborate in order to achieve the design goal. The intellectual property (IP) does not belong to anyone as it was a collaborative effort over a public forum. The designs and models are then uploaded onto the Internet of Things (IoT) where anyone can freely download and manufacture the final product. The public and each of these organisations makes use of the FDM technology in order to manufacture the major components of the prostheses [19]-[21].

In order to simplify the prostheses the three organisations designed prosthetics that make use of mechanical control. Mechanical control requires the user to have either a wrist or elbow that is functional. This requirement limits the application of these designs to only a specific group of amputees and the mechanism used for mechanical movement limits the functionality of the prosthesis.

The Robohand prosthetic shown in Figure 2(b) brings about movement using a similar mechanism as the Cyborg Beast in Figure 2(a), but Robohand makes use of only one set of cords. The cords are tightened just enough to keep the fingers straightened when the wrist is in the normal resting position. For users who do not have a palm or section of palm, but who has a functional elbow, some projects provide an adapted design to use the extending motion of the elbow to facilitate the opening and closing of the fingers.



Figure 2: (a) Movement mechanism on the Cyborg Beast by e-Nable [20]. (b) Robohand prosthetic showing the movement mechanism [21]

2.2 Myoelectric control and Arduino controls

Myoelectric control uses electromyography (EMG), electronic components and motors to bring about movement of the fingers. EMG is a diagnostic procedure used to assess the health of muscles and the nerve cells that control them. These nerve cells are called motor neurons and they transmit electrical signals from the brain that cause muscles to contract. Surface electrodes placed on the skin over a muscle sense the myoelectric activity within the muscle. For the application of myoelectric prosthetic hands, the sensors are placed on a muscle of either the lower or upper arm, depending on the severity of the amputation. The surface electrodes transmit the sensed myoelectric activity to a processor. The processor is configured to interpret the myoelectric signals and then give relevant commands to the motors to close either one, multiple, or all of the fingers [22], [23].

Myoelectric control is more complicated than mechanical control for various reasons. Firstly, the specific myoelectric activity within the muscles differs from user to user. This issue is partially overcome by complex



programming and signal processing techniques that enable the processor to correctly interpret a whole range of myoelectric activity as the command for a specific movement of the fingers. Each myoelectric prosthesis thus requires some extent of user specific calibration.

The second issue with myoelectric control is the limited processing capabilities of the electronics used in the prosthetics. The human body has an extremely complex and sophisticated network of nerves that respond to commands sent by the human brain, which is a very complex and powerful processor. Muscles and limbs are created to respond to the commands that travel through the nerves in real time and with extreme precision. The processors, motors and actuators used in the application of myoelectric prosthetics are unable to read, interpret and respond to each and every command that travels through the nerves. It is thus necessary to develop a series of recognisable commands that the components can recognise and interpret successfully. The added functionality of myoelectric control does however strongly outweigh the complexity of its application [22]. In order to keep the costs of the prosthesis low, open source electronics such as Arduino hardware was used for the myoelectric control system.

2.3 Additive manufacturing technologies

Gibson, Rosen, and Stucker [24] define additive manufacturing as the process of joining material layer by layer to make objects from 3D model data, as opposed to subtractive cutting processes (e.g. milling, turning). In order to lower the production costs of prostheses, a fused deposition modelling (FDM) machine was used in the process chain for this experiment. AM is a simpler process to produce 3D objects where very little skill is required compared to other manufacturing processes. With other manufacturing processes a detailed analysis of the object's geometry is required in order to determine the different features that can be fabricated and the different machines and tools needed to do so [24]. An entire process chain needs to be developed in order to produce just one part. With AM the process chain is considerably shorter and less complex than other traditional manufacturing methods. There are many different types of AM technologies and each can be classed into four categories which are separated by the phase in which the component material is used. These four categories are liquid phase material, filament or paste materials, powdered materials, or solid sheet materials [25]. Some of the different AM technologies are classed into their categories and displayed with the years of significant development in Table 1 below.

Name	Acronym	Category	Development years
Stereolithography	SLA	Liquid Phase	1986 - 1988
Solid ground curing	SGC	Liquid Phase	1986 - 1988
Laminated Object Manufacturing	LOM	Solid Sheet	1985 - 1991
Fused Deposition Modelling	FDM	Filament or Paste	1988 - 1991
Selective Laser Sintering	SLS	Powder	1987 - 1992
3D Printing (Drop on Bed)	3DP	Powder	1985 - 1997

Table 1: The types and development years	of additive manufacturing technologies [26]
--	---

Each AM technology has its advantages and disadvantages. The most cost effective and popular AM technology is fused deposition modelling. The FDM process builds parts layer by layer, where each layer is constructed by depositing a filament of material in a point-wise fashion. The thermoplastic filament is fed into a heating chamber and melted. A set of rollers pushes the material into the heating chamber, and it is this flow of material into the constant volume heating chamber that then produces the required pressure for extrusion. Once a layer has been completed, the build platform or the extrusion head will shift, either one layer down or up, depending on the machine setup, and the next cross-sectional layer is then deposited according to Gibson, Rosen, and Stucker [24].

3. RESEARCH METHODOLOGY

Firstly, a process chain that could be used to produce a fully functional, patient specific prosthetic hand with myoelectric control had to be developed. Once the process chain was developed and refined the prosthesis had to be manufactured following the process steps. The times and costs were recorded throughout each step in the process chain. The research methodology can be observed graphically in Figure 3 below.



Prosthetics Open-source	Evaluation Affordable - Open	Time and Cost Study
hardware Open Community Manufacturing (OCM) AM technologies	source technologies FDM evaluation Process chain development	Assembly of myoelectric control unit Printing of skeleton structure
		Printing of moulds and silicon finger production

Figure 3: Research Methodology

4. RESULTS AND DISCUSSION

The results from this study are presented and discussed in this section. The process chain that was developed can be observed as well as the finished prototypes and the costs to manufacture the prosthesis at each step of the process chain.

4.1 Open source process chain

The process chain that was developed in order to produce an aesthetically appealing patient specific myoelectric prosthetic hand can be observed in Figure 4 below.





Each step of the process chain is elaborated on in the following sections.

4.2 Hand digitising and myoelectric control system

Firstly the patients hand needed to be scanned in order to replicate their intact functional hand. In this study a Kreon KLS51 laser line scanner attached to a Zeiss coordinate measurement machine (CMM) was used to scan the hand. The integrated system combines the advantages of a CMM and a laser scanner, for a faster and non-contact method of surface digitisation that allows for very high scan resolutions [10]. Once the hand was scanned and digitised into CAD software format the myoelectric system could be manufactured.

In order for the prosthetic hand to function, similarly to the E-Nable and Robohand, without wrist movement, an open source electronic myoelectric control system needed to be used. One of the best suited open source hardware systems is the Arduino system. The hardware component configuration that was used to create the myoelectric control system is displayed in Figure 5 below.





Figure 5: Pin connections and interfaces of electronic subsystem.

Once fully assembled the components were tested and a trial run was performed in order to determine whether the system was functioning properly before moving on with the next step in the process chain.

4.3 Hand skeleton structure

Firstly the dimensions of the designed skeleton structure had to be edited in order to match the dimensions of the patient's intact hand. Once the design was edited the different parts were converted to an STL file format and uploaded onto the machines build software. The skeleton structure was printed on an UP Mini FDM printer. The printing parameters were as follows: the layer thickness was 0.25 mm with an 80% fill density. The material that was used was ABS. The final printed hand skeleton structure can be observed in Figure 6 below.



Figure 6: Hand skeleton structure to house all of the myoelectric components

To increase the patient's confidence and psychological well-being the prosthesis needs to look lifelike. So the next step in the process chain is to replicate the patient's intact fingers and mirror them in CAD software and mould them out of medical grade silicon.

4.4 Silicon prosthetic fingers

The aesthetic look of the prosthesis needs to be lifelike and in order to achieve this; the patient's intact hand was scanned in the first step of the process chain. The 3D digitisation of the patient's hand can then be



mirrored with the CAD software and edited on an open source sculpting software like Sculptris Alpha or Blender. Once the digital hand is to the patient's satisfaction, each finger can be separated from the model. Once this is performed a Boolean operation needs to be done on each finger in order to create a mould for each finger. Once the moulds are in STL file format they can be loaded onto the 3D printers build software. The mould is then printed and post processing is performed as displayed in Figure 7 below. The fingers from the skeleton structure are then set into position in the moulds. The silicon is mixed, along with the correct colour dye in order to match the patients existing skin colour. The silicon is then poured around the skeleton structure until the mould is filled. The silicon is left to cure for 23 hours and then the fingers are removed and attached to the skeleton structure. The final silicon fingers can be observed in Figure 8 below.



Figure 7: Mould produced using an UP Mini FDM printer [10]



Figure 8: Top (left) and bottom (right) view of the final silicon fingers [10]

In order to not interfere with the motor and other features on the skeleton structure, a latex glove is used to cover the rest of the hand. The glove is dyed the same colour as the patients skin and then cut to allow the silicon fingers to show.

4.5 Time and Cost Study

Tables 2, 3, and 4 display the times and costs for each process chain.

Table 2: Time and costs of 3D digitising and the myoelectric control system

ITEM	COST	TIME
3D Scanning of Hand	R550.00	01H00
Arduino Pro Mini Micro Processor 5V	R144.95	00H05
FTDI Basic Breakout Module 5V	R194.95	00H10
Advancer Technologies Muscle Sensor Kit	R749.95	00H12
Sparkfun Stepper Motors x2	R296.00	00H13
Servo Motor	R148.00	00H09
Easy Drive Stepper Motor Driver	R403.00	00H06
Total	R2486.85	01H55

Table 3: Time and	costs to print the	hand skeleton structure
-------------------	--------------------	-------------------------

PART	COST	TIME
Palm of the hand	R24.00	01H37
4 Fingers with Joints	R24.00	01H36
Thumb with Joints	R5.00	00H20
Total	R53.00	03H33



PART	COST	TIME
Printing of Mould	R50.00	22H36
Post Processing	R10.00	00H30
Silicon Work	R60.00	23H00
Total	R120.00	46H06
Total for 5 fingers	R600.00	115H30

Table 4: Time and costs to produce the silicon fingers

The total time take to produce an aesthetically appealing, patient specific myoelectric controlled prosthetic hand was 120 hours and 58 minutes. The majority of the time is spent on the last step of the process chain with the printing of the silicon moulds and the curing of the silicon in the moulds. The total cost of the hand was roughly R3139.85 which is very affordable in comparison to a professionally manufactured myoelectric controlled prosthesis.

CONCLUSION 5.

A process chain to manufacture a patient specific prosthetic hand with open source myoelectric control was successfully developed. The myoelectric control system operated correctly and the hand could open and close as it was designed to do. The skeleton structure would need to be redesigned in order to allow the silicon fingers to bond better to the finger structure. The aesthetic appearance of the silicon fingers were up to the standards of a prosthetic sculptor.

6. REFERENCES

- P. Bartolo, J. P. Kruth, J. Silva, G. Levy, A. Malshe, K. Rajurkar, M. Mitsuishi, J. Ciurana, and M. Leu, [1] "Biomedical production of implants by additive electro-chemical and physical processes," CIRP Ann. Manuf. Technol., vol. 61, no. 2, pp. 635-655, 2012.
- [2] J. Falconer, "Inexpensive home-brewed prostheses created using 3D printers," Gizmag, 2013. [Online]. Available: http://www.gizmag.com/inexpensive-prostheses/26009. [Accessed: 20-Sep-2015].
- L. J. Vorvick, "Medline Plus," A service of the U.S. National Library of Medicine, 2013. [Online]. [3] Available: http://www.nlm.nih.gov/medlineplus/ency/article/ 002286.htm. [Accessed: 20-Feb-2015].
- H. Burger and C. Marincek, "Return to work after lower limb amputation," Disabil. Rehabil., vol. 29, [4] no. 17, pp. 1323-1329, 2007.
- D. Desmond and M. MacLachlan, "Psychological issues in prosthetic and orthotic practice: a 25 year [5] review of psychology in Prosthetics and Orthotics International.," Prosthet. Orthot. Int., vol. 26, no. 3, pp. 182-188, 2002.
- A. M. Carroll and N. Fyfe, "A Comparison of the Effect of the Aesthetics of Digital Cosmetic Prostheses [6]
- on Body Image and Well-Being," JPO J. Prosthetics Orthot., vol. 16, no. 2, pp. 66-68, 2004. K. Subburaj, C. Nair, S. Rajesh, S. M. Meshram, and B. Ravi, "Rapid Development of Auricular [7] Prosthesis using CAD and Rapid Prototyping Technologies," Int. J. Oral Maxillofac. Surg., vol. 36, no. 1, pp. 938-943, 2007.
- [8] V. Mironov, V. Kasyanov, and R. R. Markwald, "Organ Printing: from bioprinter to organ biofabrication line," Curr. Opin. Biotechnol., vol. 22, pp. 667-673, 2011.
- G. A. Oosthuizen, D. Hagedorn-Hansen, and T. Gerhold, "Evaluation of Rapid Product Development [9] Technologies for Production of Prosthesis in Developing Communities," in SAIIE25 Proceedings, 2013, no. July, pp. 590.1-590.14.
- D. Hagedorn-Hansen, G. A. Oosthuizen, and T. Gerhold, "Resource-Efficient Process Chains To Manufacture Patient-Specific Prosthetic Fingers," South African J. Ind. Eng., vol. 27, no. May, pp. 75-[10] 87, 2016.
- P. Liacouras, J. Garnes, N. Roman, A. Petrich, and G. T. Grant, "Designing and manufacturing an [11] auricular prosthesis using computed tomography, 3-dimensional photographic imaging, and additive manufacturing: A clinical report," J. Prosthet. Dent., vol. 105, no. 2, pp. 78-82, 2011.
- K. Kokalitcheva, "How Google is Helping Amputees get 3D-printed Prosthetics," Time, 2015. . [12]
- [13] B. Berman, "3-D printing: The new industrial revolution," Bus. Horiz., vol. 55, no. 2, pp. 155-162, 2012.
- C. I. Ras, J. Botes, M. Vermeulen, G. A. Oosthuizen, and J. W. Uys, "Social manufacturing business [14] model elements to support local suppliers," in Rapid Product Development Association of South Africa, 2015.
- [15] J. Day and H. Zimmerman, "OSI Reference model-The ISO Model of Architecture for Open Systems Interconnection," IEEE Trans. Commun., vol. 28, no. 4, 1980.
- C. Goldfarb, "Congenital Hand and Arm Differences," 2015. . [16]
- M. A. R. Inc., "Prosthetic Restoration of the finger/toe," 2016. [Online]. Available: http://www.medicalartresources.com/services-directory/finger-toe-2/. [Accessed: 14-Jun-2016]. [17]
- G. A. Oosthuizen, P. Butala, S. Bohm, A. Rebensdorf, and A. Gergert, "Open Community [18]



Manufacturing," in SAIIE 26 Proceedings, 2014, pp. 1-14.

- [19] J. Gibbard, "Open Hand Project," 2013. [Online]. Available: http://www.openhandproject.org/. [Accessed: 13-Jun-2016].
- [20] Gravida, "Enabling The Future," 2015. [Online]. Available: http://enablingthefuture.org/. [Accessed: 13-Jun-2016].
- [21] "Robohand," 2016. [Online]. Available: http://www.robohand.net/. [Accessed: 13-Jun-2016].
- [22] C. M. Light, P. H. Chappell, B. Hudgins, K. Engelhart, C. M. Lighty, B. Hudginsz, K. Engelhartz, P. H. Chappell, B. Hudginsz, and K. Engelhartz, "Intelligent multifunction myoelectric control of hand prostheses," J. Med. Eng. Technol., vol. 26, no. 4, pp. 139-146, 2002.
- [23] T. a Kuiken, G. a Dumanian, R. D. Lipshutz, L. a Miller, and K. a Stubblefield, "The use of targeted muscle reinnervation for improved myoelectric prosthesis control in a bilateral shoulder disarticulation amputee," *Prosthet. Orthot. Int.*, vol. 28, pp. 245-253, 2004.
- [24] I. Gibson, D. W. Rosen, and B. Stucker, Additive Manufacturing Technologies: Rapid Prototyping to Direct Digital Manufacturing. New York: Springer, 2010.
- [25] N. Guo and M. C. Leu, "Additive manufacturing: Technology, applications and research needs," Front. Mech. Eng., vol. 8, no. 3, pp. 215-243, 2013.
- [26] J. J. Beaman, J. W. Barlow, D. L. Bourell, R. H. Crawford, H. L. Marcus, and K. P. McAlea, Solid Freeform Fabrication: A new direction in manufacturing. New York: Springer US, 1997.





LEAN MANAGEMENT IN THE HEALTHCARE INDUSTRY

A.P. Muthivhi

M-Tech I.E. Student Tshwane University of Technology South Africa <u>Aluwani.muthivhi@gmail.com</u>

Dr. M.G Kanakana

Department of Industrial Engineering

Tshwane University of Technology

South Africa

kanakanamg@tut.ac.za

Dr K. Mpofu

Department of Industrial Engineering

Tshwane University of Technology

South Africa

mpofuk@tut.ac.za

ABSTRACT

Lean management began in the manufacturing industry as a strategy for increasing efficiency and reducing costs of manufacturing. This strategy proved very successful for Toyota Motor Company and other manufacturing companies. Consequently, based on the success recorded, the service industry embraced it. This study seeks to explore the effects of implementing Lean methodology in the healthcare industry and explore its strategy. A qualitative approach was utilised to evaluate what sets apart lean from other popular strategies. The study focused on the prominence that Lean healthcare management is gaining momentum as a preferred approach to healthcare service delivery in the 21st century. It identified the differences between lean and other popular strategies utilised in healthcare. Management's responsibilities and steps to Lean Management were deciphered and five basic steps developed to better determine the lean journey in simple terms.



1. INTRODUCTION

[12] "Heavy investment over the past 30 years has made the health care sector the largest expenditure category of the health system in most developed and developing countries". Harding and Preker [16]. This is because health care sector is responsible for servicing, repairing and assisting with most human beings health problems. Besides the importance of this sector "There are still significant inefficiencies in the health system stemming from poor quality of care". David Harrison [9]

"The relatively high healthcare expenditure coupled with poor outcomes speaks to weak management" Kleinert and Horton [19]

meaning management is the key needed to be enhanced to get better service delivery in the healthcare sector. "Striving to achieve a process without non-value-added time, monetary waste, and product waste has been a long-standing tradition in manufacturing, but the healthcare industry has been slow to adopt these principles". (Benfield et. al, [6]concurs "clearly it shows that the evolution of the healthcare sector has been slow as compared to the manufacturing sector".

This is because the healthcare sector was focused on developing new technology and better processes to improving healthcare neglecting the basis of its day to operations.

There are many approaches to improve healthcare that have been explored but one in particular has picked up in recent trends in developed countries which is lean in health care. Waldhausen, John HT, et al. [37] concluded that Lean methodology can be used to improve clinic efficiency as well as patient and staff's experience.

From the manufacturing industry, Toyota motor company developed the Toyota Production System known as lean manufacturing which has now migrated to the service industry Arlbjorn [4] and Kanakana [18] concur. Healthcare organizations have been one of the latest services settings adopting Lean principles, tools and techniques feeding a crescent stream of literature. Guimaraes & Carvahlo [14]

Why lean in health care one might ask?

"The core competence of Toyota Motor Corporation is its ability to produce automobiles of great quality at best prices, thereby providing a value for money to the customers" Nkomo 2014, [34]. Nkomo [34] further states:" Toyota's distinctive competence is its production system known as the "Toyota Production System" or TPS.

TPS is based on the Lean Manufacturing concept furthermore Nkomo [34] states that "Overall, Toyota has outperformed the industry over the past five years and its 8 times more profitable than the industries average" which shows that lean manufacturing is one of the best if not the best approach to adopt if you want to be competitive in the market.

Furthermore, Nkomo [12] states that "Toyota has outperformed the industry over the past five years". Critics note that alternatives to lean production have not gained much acceptance Dankbaar, [7] and stipulate that ''lean production will be the standard manufacturing mode of the 21st century'' Rinehart et al. [12].

With the increasing pressure on healthcare providers to reduce costs and improve quality, an increasing number of organizations are looking to Lean tools and techniques as a breakthrough solution for performance improvement" White Paper on Lean Healthcare, Philips, [27]

Lean identifies gaps, waste and opportunities for improvement and eliminates or reduces those through continuous improvement. Furthermore, Philips white paper [27] expresses that "Over the last four decades, Lean has emerged as one of the most impactful approaches to help increase an organization's competitiveness through improvements in process efficiency and a reduction in operational waste".

Today, Lean is used in most global industries and virtually all organizational sectors including healthcare. (Kanakana [18] concurs.

Lean begins with driving out waste so that all work adds value and serves the customer's needs. *Institute of health Improvement* 2005, p.2, Innovation Series [17]

To eliminate the 7 waste only a human can influence change so which brings us to the 8th waste which is a lack of creativity or waste of none improvement in lean terms by employees.

Lean aims to tap into this 8th waste in every employee in an organisation to create a culture of continuous improvement using lean tools for solving problems (in lean terms opportunity for improvement) by eliminating waste and coming up with better ways to provide the service or produce a product.



Lean production in healthcare is mostly used as a process improvement approach and focuses on three main areas: Poksinska [28]

- Defining value from the patient point of view,
- Mapping value streams and
- Eliminating waste in an attempt to create a continuous flow.

Lean healthcare is still in an early stage of development if compared to the same process in the automotive industry, Souza [33]. Kanakana[18] quoted Burgess and Radnor[2013] in their study on evaluating Lean in healthcare, claiming that Lean in English National Health Service is being successfully implemented in many cases and continuous to gain momentum in English hospital trusts and also continues to spread throughout the world.

This study aims to show how lean management is one of the best strategies to adopt in order to increase quality in the health care sector by comparing it to PDSA and Six Sigma.

2. LITERATURE REVIEW

The primary target for lean principles in Healthcare is to improve quality healthcare where quality healthcare is defined by Agency for Healthcare Research as:

- "Doing the right thing at the right time in the right way for the right person, and having the best possible results" Varkey and Kollengode[36]
- "Lean is a systematic approach to identifying and eliminating waste (non-value-added activities) through continuous improvement by flowing the **product at the pull of the customer in pursuit of perfection." Lockwood [24].

Most of the definitions focus on eliminating waste but lean focuses on increasing the quality of the product or service to the customer by having the following conditions in mind when producing products and delivering a service according to the Levantar organisation. <u>www.levantar.co.uk</u> [44]

- 1) The customer must be willing to pay for the activity.
- 2) The activity must change the product making it closer to the end product or service that the customer requires and is willing to pay for.
- 3) The activity must be done right the first time.

To achieve these three conditions lean looks at increasing the efficiency of its resources to attain the most efficient process to develop a quality product or deliver a service with the aim of being competitive, making the profit and staying in business.

Instead of the old traditional 7 lean wastes lean has identified the 8 waste and a lot of literature has been written to validate the 8th waste under lean Six Sigma by Dr. Pat Hammett, 2008 and The Lean Enterprise Institute and more. This is Waste of none creativity (e.g. Nurses with ideas to improve procurement but withholding it since it's not their department) because between all the resources required to produce a product or service the only resource that can influence change and solve problems is the people (management and employees).

According to Zameer [43] "Many authors regard the respect for people as fundamental to Lean implementation and acknowledge the value of staff input in creating improvements". Lean aims to develop employee's creative thinking so that they come up with ways to improve the processes, products, eliminate waste; solve problems and come up with improvements in order to produce a quality product or service. Liker[23] concurs by stating that "the more I have studied TPS and the Toyota Way, the more I understand that it is a system designed to provide the tools for people to continually improve their work" this is made possible by employees trained and equipped with tools to take initiative through the leadership of management.

Kollberg et al. [20] cited (*Miller*, *D.*, *et. al.*, 2005).""Adoption of Lean management strategies while not a simple task can help healthcare organizations improve processes and outcomes, reduce cost, and increase satisfaction among patients, providers, and staff". Balle [5] states that: "the adoption of lean in healthcare is very young it started in the late 90s but started to gain a little bit of momentum in the last decade and now in this decade many health organisations are looking at lean principles since it has been proven to be very successful in many cases around the world".

However Vinodh et al. [37] states that "a Lean method is a good option for optimizing clinical workflow, because it focuses on detailed process components, such as workflow and problems, and then redesigns the processes by removing waste". Andersen on the above statement basically wants to show that each organisation will have different lean solutions although they will be driven by the same lean principles. According to Wu [41]moving from theory to practice may be challenging as lean thinking requires a deep change of operating mentality, which frequently collides with embedded mind-set".



3. METHODOLOGY

The methods of evaluating change and improvement strategies are not well described, **Eccles** et al. [12] A systematic review of the literature was done in order to establish quality management strategies that are been adopted from the manufacturing sector and applied in healthcare in the 21st century with the aim of

comparing them with lean Management in Healthcare; and PDSA and Six Sigma were chosen.

A qualitative method was used in order to determine the effects of lean implementation in the health care industry and the differences it has as compared to other strategies reviewed in order to identify the critical element that sets apart lean from the two strategies. "A primary goal of qualitative research is to produce knowledge that is transferable from one context to another" Goodridge et al. [13]From the available literature relevant literature was reviewed in order to investigate lean health care strategies mainly to establish what sets it apart from other management strategies adopted from manufacturing implemented in healthcare recently.

4. RESULTS

4.1 PDSA

Table 1: Positive PDSA implementation in Healthcare industry

Reference	Year	Country	Study based on	Results
Zack	2008	USA	Zero tolerance for central line-associated bacteraemia.	Positive
Torkki et al. 2	2006	Finland	Managing urgent surgery as a process: case study of a trauma centre.	Positive

Torkki et al. used the PDSA cycle to improve Hospital efficiency and reduce waiting times and the results showed a positive outcome:

- ✤ Waiting times decreased by 20.5 percent (p<.05)</p>
- None operative times in the operating room were reduced by 23.1 percent (p<.001)
- efficiency was increased by 9.7 percent (p<.001)
- Overtime hours decreased by 30.9 percent after re-engineering of the care process.

PDSA cycle was introduced by Edward Deming as a strategy for implementing change management. This is an often used process to help management and technical teams improve the quality of care in healthcare. Basically, the strategy is mostly used for addressing long-term strategic change but can also be used for short-term objectives.

PDSA is a tool for "developing, testing and implementing changes leading to improvement" Langley et al. [22] which has four stages. Langley et al. [22] furthermore state that "they developed a quality improvement program to reduce patient waiting times in the Hospital Management System using Plan-Do-Study-Act (PDSA) methodology" which proves that this methodology is being adopted in healthcare.

Luther, Hammersley and Chekairi, A., [25] established the key feature of the PDSA cycle is that it recognizes that the objectives of a project are rarely accomplished first time round, and encourages learning from the errors of the first attempt.




4.1.1 Plan

Objectives are placed based on e.g. (in a health care institution service and patient needs) key important questions asked at this phase.

- Mainly what is the Aim or objective of what u trying to achieve? Donnelly and Kirk [45] this is called the *aimed statement*.
- Underpinning this is the question ... What is the problem? You then have to formulate an answer to the question.
- How do you know it is a problem? To answer this, baseline measurements must be acquired.

Donnelly and Kirk [45] "These inform the understanding of the problem and its scale. It is also important to acquire data which informs the likely or actual cause of the problem". Donnelly and Kirk [45] furthermore they state that "The points above naturally lead toward producing a description of the solution"

4.1.2 The second step is DO.

Donnelly and Kirk [45] "This is where you carry out the change, test or intervention and record what has happened. It is important to do this from a particular point in time and to take measurements over a period of time to record the pattern of data"

4.1.3 The third step is STUDY.

Langley GL et al. [22] "In essence, this is about studying or analysing your data and the process itself". Some Important questions in this step are:

- What was the outcome based on what we planned?
- Did things go according to the implemented plan?
- How the experience was and what was gained from the experience (lessons learned)?

4.1.4 The next step is ACT.

Donnelly and Kirk [45] "Act is an important element in that you have to consider what measures and procedures are in place to ensure that whatever solution or solutions you have realised remain effective", Key questions such as:

• What changes are needed to improve the process?



Justification of improvement changes?

Such questions lead to a plan which in turn generates the next PDSA cycle.

According to the Authors PDSA is a strategy that seeks perfection by continuously improving the processes (continuous Improvements serves as the key to ingredient that drives the PDSA cycle).

4.2 Six Sigma

Table 2: Positive Six Sigma implementation in Healthcare industry

Reference	Year	Country	Study based on	Results
Frankel et al.	2005	USA	Performance-improvement strategies to reduce incidence of catheter-related bloodstream infections in a surgical ICU.	Positive
Does et al.	2009	Netherlands	Reducing start time delays in operating rooms.	Positive

Ian et al. (2010) stated that "in business the term **Six Sigma** refers to a set of tools and methodologies developed by Motorola to improve the business by eliminating defects". Pandi, Sethupathi and Rajesh [46] concur by stating that "the core concept of **Six Sigma** is to reduce the defects in the process".

"Six Sigma has been proved as an effective approach for quality improvement in service sectors, especially at healthcare and financial services" Schroeder et al. [31] and also in the table above the researcher shows two success stories in healthcare.

This means that it is an approach that is suitable to be used in eliminating defects in healthcare. Nicolay et al. [47] states that quality in six sigma is "measured in terms of defect (error) rates and the aim is to reduce this to six standard deviations (sigma or σ) from the process mean" furthermore Nicolay et al. [47] expresses that Six Sigma translates as 99-99966 percent defect free or error-free or 3.4 defects per million opportunities (DPMO).

Six Sigma uses the DMAIC methodology basically divided into 5 steps.

4.2.1 Step 1: Define

Define is where the opportunity for improvement or problem is defined in detail to understand its details and constraints. (Hikmet Erbiyik and Muhsine Saru [15] express key questions to ask at this stage

- What is the source of problems causing to failure?
- ✤ What are the relations?
- Between cause and effect of the problem?

Furthermore Hikmet Erbiyik & Muhsine Saru [15] state that "By searching the answers for above questions the relevant problem is defined".

4.2.2 Step 2: Measure

Is when the current situation is measured to evaluate its impact and contribution.

4.2.3 Step 3: "Third, the capability

The capability of the process needs to be analysed in order to determine if it is delivering what is required (accurate stability predictions or estimates), and if not, improve". De Vore [49]

4.2.4 Step 4: Improve

This is divided into 3 steps according to (Alexandra, Carneiro & Pinto [3]

- Step 1: Solution Identification
- Step 2: Solutions Prioritization
- Step 3: Solution Implementation

4.2.5 Step 5: Karl De Vore(2008) state step 5: control as "the stability testing process by ensuring that the improvements that have been implemented are maintained through time".



Six Sigma mainly deals with eliminating problems in a process by eliminating its short comings in simple terms the authors express that (six sigma seeks design perfection in the process)

4.3 Lean.

Reference	Year	Country	Study based on	Results
Waldhausen et al.	2010	USA	Application of Lean methods improves surgical clinic experience.	Positive
Muder et al.	2008	USA	Implementation of an industrial systems engineering approach to reducing the incidence of methicillin.	Positive

Table3. Positive Lean implementation in Healthcare industry

Lean management in healthcare has been discussed and evaluated in many ways and many frameworks developed mostly discussing the tools. From the literature reviewed by the researcher thus far and personal experience, the Researcher developed five basic steps to Lean management.

4.3.1 Control your territory

- Basically, this means to control your working area and know its limits
- Identifying your resources and know what you have and what they can achieve and their costs to run, cost to maintain, what is useless and etc. (Preventative Maintenance concept)
- Create a clean working Area which will able you to spot any anomalies immediately
- Also, this stage, is mainly concerned with Housekeeping (A place for everything and everything in its place defined). This is done by using a lean tool called 5s Method.
- In this initial stage the manager must do regular field visits spotting anomalies and encourage employees to iron them out.
- Lastly to reduce or eliminate the risks of mistakes from happening this is known as Error Proofing outlined by Womack, jones, and Ross, [39]

4.3.2 Establish meaning to actions

- In this step, you Identify your all your customers and what they expect from you.
- Then design performance indicators based on your customers' expectations that you can measure on a daily basis. It's then easy to study the processes and clearly indicate what is required to process and what is not necessary for the processes.
- In this step tools like value stream mapping, Kanakana [18] stated that it provides a graphical representation of the process activities and provides a scope of the project by defining the current and future states of the system.
- Evaluating your suppliers because without the supply of raw materials nothing can be produced. This is done by creating continuous flow in the process.
- Daily activities should be aligned according to customer needs and Tools for monitoring established (Visual Management) which assists in exposing anomalies immediately to the day to day activities.

4.3.3 Training your employees in the lean culture and giving them a platform to express challenges, problems and solutions.

- This will be done by assembling your team around a central location where you will have a visual management board where the performance will be evaluated.
- It must be done in the morning for 15 minutes (sort of a ritual)at least weekly and employees must indicate what they have done, their challenges, discoveries, problems to be solved, lessons learned, opportunities for improvement, actions in progress, update the visual management board and etc.
- This should become a daily ritual and you must acknowledge achievements by your team and always motivate during this time.

4.3.4 Delegating problem-solving

Problems or opportunities for improvement discussed in the daily or weekly ritual can't all be solved immediately. Delegating a team (including Management if possible) to analyse the problem and stamp out the root cause of the problem discuss the solution and give responsibility employees to implement the solution.



In this stage method known as Kaizen is used; basically if the solution is simple it will be implemented immediately but if a study needs to be done a team will assess the problem develop a solution and implement.

4.3.5 Managing the progress plan...

- Instead of Kaizen day to day improvements here is when the organisation is running smoothly now breakthrough solutions to meet the strategic challenges are needed.
- This is done by many ways but the cycle is the same evaluate the state of the company's current situation and where you want it to be and set up steps to reach those goals and implement them.

Criteria	PDSA	SIX SIGMA	LEAN
Methodology main focus	Reengineering Processes	Eliminating Defects	Eliminating waste
Tools used	Changing the process to a better more efficient one	Reduce variations in the process	Reducing none value adding processes according to the customer
level of employee involvement	Exploring options to improve the system	Exploring the causes of defects in the process and eliminating them	Making sure the activities of the process add value to the end service or product
Implementation approach	Normally by Management and technical stuff	Mostly Quality experts, Management and technical employees	Every employee can influence change
Feasibility	Meduim	Expensive	Low in most instances

Table 4: Key differences of PDSA, Six Sigma and Lean.

Discusion

There are Hybrids of improvement strategies in the healthcare sector being applied to complement for the weakness of the other strategy e.g. Lean Six Sigma but for the sake of these study original strategies where evaluated.

The PDSA cycle is driven by the nature repetition of the cycle to seek perfection from the lessons learned in the previous cycle and it's echoed in the DMAIC process of Six Sigma and apparent in the Kaizen continuous improvement methodology of lean. Which one can argue that it can fall under Lean as a tool for continuous improvement? Although unlike Kaizen which uses baby steps (small improvements) to arrive at its goal PDSA completes a goal and re-evaluates the new strategy and sets a different course of improvement.

Six Sigma, however, uses statistical analyses which are very accurate in spotting and eliminating the root cause of defects and process variations all through the DMAIC process defines and improve its application to business re-engineering is very farfetched. Instead of statistical analyses Lean uses value stream mapping and GEMBA (which stands for go observe for yourself) and kaizen which are similar to six sigma but are less complex and anyone can adopt them easily.

Lean seems to embody the two processes in its culture although it's in a different form. To be able to fully integrate Lean into healthcare isn't an easy task because it incorporates a lot of tools from the production industry which might not necessarily be applicable to the service industry e.g. In manufacturing when a defect occurs in any process a lean factory stops and solves that problem but for a hospital it's impossible if a defect occurs in admin we can't expect the doctors to stop operating a patient in theatre.

Nicolay et al. [47] express clearly that methodologies such as Six Sigma require trained employees in Six Sigma (ideally people with statistical background) because it requires a lot of data collection and statistical analysis. This leaves the sustainability of the program in the hands of the Six Sigma specialists.

PDSA is a re-engineering model this process although doesn't require much training the researchers express that it needs at least technical staff and management to execute this process on the other hand although advance lean integration needs specialist the basic form of lean which is termed lean thinking developed by Womack and Jones refers to the spotting and elimination of waste is easy enough to be integrated into the whole organisation easily at a low cost.

From the analyses, Lean is mostly set apart from the other strategies because its continuous improvement methodology known as lean thinking can be easily trained to every employee at a low cost which makes Lean the most viable strategy to adopt in healthcare in the 21 century.



5. CONCLUSION

Dobrzykowski, McFadden, and Vonderembse [11] "To improve patient safety and financial performance, many healthcare organizations have implemented quality initiatives such as lean process improvement". Mainly this as a result of other studies in other industries provides evidence on the effectiveness of lean in reducing lead times, improving space utilization, increasing throughput, improving quality, and increasing financial performance Shah and Ward, [48].

"It should also have become clear for the readers, that quality improvements and waste reduction require a systematic and well-planned process, where all employees are involved in identifying the problems, route causes behind the problem and the needed action for eliminating the problem". (Dahlgaard, Pettersena and Park[21]

And from the strategies that are common in the 21st century only Lean can achieve that (if we exclude hybrid strategies like Lean Six Sigma).

The researcher expresses that Lean is mostly set apart from the other strategies because it equips all employees in an organisation with tools, principles, and philosophies to influence change and also provides them with a platform to make those changes. Womack and Jones [40] concur by expressing that Lean seeks to reconfigure organisational processes to eliminate waste and enhance productivity based upon the application of specialist analytical tools and techniques coupled with creating a culture of continuous improvement.

Based on the above the Lean management in healthcare although it was adapted from the manufacturing sector seems to work for the service industries including health care.

Lean's uniqueness of being able to engage management to promote growth in the organisation through making sure that every employee is equipped with a continuous Improvement ideology, from the shop floor operator to the CEO or Director of the organisation sets lean apart from other methods and the researcher expresses lean management in healthcare as the most viable approach to adopt in healthcare.

6. REFERENCES

- 1. A. Pal Pandi; P.V.Rajendra Sethupathi and R. Rajesh A Tittle: Conceptual Model for Achieving Global Quality in Engineering Educational Institutions in Indi. International conference on modelling optimization computing. (2012) Procedia Engineering 38
- 2. Albeanu, Mircea, Radford, Jo, Hunter, Ian Six Sigma in HR Transformation : Achieving Excellence in Service Delivery 2010
- 3. Alexandra Teneraa and Luis Carneiro Pinto^a Tittle: A Lean Six Sigma (LSS) project management improvement model (27th IPMA World Congress) Procedia Social and Behavioral Sciences 119 (2014) 912 920 by
- Arlbjørn, J.S., P.V. Freytag and H.D. Haas, Service supply chain management: A survey of lean application in the municipal sector. International Journal of Physical Distribution & Logistics Management, 2011. 41(3): p. 277-295.
- 5. Balle, M. 1991. Lean attitude. Manufacturing Engineer, 84(2), 14-19, 2005
- 6. Benfield, J.A., Rainbolt, G.N., Bell, P.A. and Donovan, G.H., 2015. Classrooms With Nature Views Evidence of Differing Student Perceptions and Behaviors. *Environment and Behavior*, 47(2), pp.140-157.
- 7. Dankbaar, B.,. Lean production: denial, confirmation or extension of sociotechnical systems design? 1997 Human Relations 50 (3), 653-670.
- 8. David D. Dobrzykowski a, *, Kathleen L. McFadden b, Mark A. Vonderembse Examining pathways to safety and financial performance in hospitals: A study of lean in professional service operations. 2012 Journal of Operations Management 1-13
- 9. David Harrison: An Overview of Health and Health care in South Africa 1994 2010: Priorities, Progress and Prospects for New Gains 2009,
- 10. Does RJMM, Vermaat TMB, Verver JPS, Bisgaard S, Van den Heuvel J. Reducing start time delays in operating rooms. J Qual Technol 2009; 41: 95-109.
- Douglas, T. J., & Judge, W. Q. Title: Total quality management implementation and competitive advantage: The role of structural control and exploration. Academyof Management Journal, (2001). 44(1), 158-169.
- 12. Eccles, M., Grimshaw, J., Campbell, M. and Ramsay, C., 2003. Research designs for studies evaluating the effectiveness of change and improvement strategies. *Quality and Safety in Health Care*, 12(1), pp.47-52.
- 13. Goodridge, D., Westhorp, G., Rotter, T., Dobson, R. and Bath, B., 2015. Lean and leadership practices: development of an initial realist program theory. *BMC health services research*, 15(1), p.1.



- 14. Guimaraes & Carvahlo 2012, "Lean Healthcare across Cultures: State Of the Art" Vol. 2. European Journal of Cross-Cultural Competence and Management,
- 15. Hikmet Erbiyik and Muhsine Saru Six Sigma Implementations in Supply Chain: An Application for an Automotive Subsidiary Industry in Bursa in Turkey. Procedia - Social and Behavioral Sciences 195 (2015) 2556 - 2565
- 16. Harding A, and AS Preker, AS.. "A Conceptual Framework for Organizational Reforms in the Hospital Sector.":V.I Corporatization of Public Hospitals. 2002
- 17. Institute of health improvement. White Paper, 2005 "Going Lean in Health Care" innovation series.
- 18. Kanakana M.G. Lean In Service Industry Journal Saiie Stellenbosch S.A. 2013. P. 574-1 To 10.
- 19. Kleinert S., Horton R., South Africa's Health: departing for a better future?, Lancet 2009; 374: 759-760
- <u>Kollberg</u> Beata, Jens J. Dahlgaard, Per-Olaf Brehmer, (2007) "Measuring lean initiatives in health care services: issues and findings", International Journal of Productivity and Performance Management, Vol. 56 Iss: 1, pp.7 - 24Liker J.K., The Toyota Way: 14 Management Principles from the World's GreatestManufacturer, New York: McGraw-Hill 2004,
- 21. Jens J. Dahlgaarda, Jostein Pettersena and Su Mi Dahlgaard-Park Tittle: Quality and lean health care: a system for assessing and improving the health of healthcare organisations 2011 Total Quality Management Journal of Taylor & Francis Vol. 22, No. 6, June 2011, 673-689
- 22. Langley GL, Nolan KM, Nolan TW, Norman CL, Provost LP: The Improvement Guide: APractical Approach to Enhancing Organizational Performance. 2nd edition. San Francisco: Jossey Bass. (2015)<u>http://www.institute.nhs.uk/quality_and_service_improvement_tools/</u>
- Liker J Tittle: the Toyota way: 14 Management principles from the world's greatest manufacturer. Mcgraw-Hill, New York 350 pp.
- 24. Lockwood Steve "Introduction to Lean Healthcare" Citec Business Advisor 2013
- 25. Luther, V., Hammersley, D. and Chekairi, A., 2014. Improving patient handover between teams using a business improvement model: PDSA cycle. *British Journal of Hospital Medicine (17508460)*, 75(1).
- Muder RR, Cunningham C, McCray E, Squier C, Perreiah P, Jain R et al. Implementation of an industrial systems-engineering approach to reduce the incidence of methicillin-resistant Staphylococcus aureus infection. Infect Cont Hosp Ep 2008; 29: 702-738
- 27. Philips., white paper 2011. "Successfully deploying Lean in healthcare"
- 28. Poksinska Bozena The current state of Lean implementation in health care: literature review, 2010, Quality Management in Health Care, (19), 4, 319-329.
- 29. Rachna Shah and Peter T. Ward. Tittle: *Defining and developing measures of lean production(2007)* Journal of Operations Management 25 785-805
- 30. Rinehart, J., Huxley, C., Robertson, D., 1997. Just Another Car Factory? Cornell University Press, Ithaca, NY.
- 31. Schroeder, R. G., Linderman, K., Liedtke, C., & Choo, A. (2008). Tittle Six-Sigma: Definition and Underlying Theory. Journal of Operations Management, Vol. 26, No. 4, pp. 36-54.
- Stewart Robinson, Zoe J. Radnor, Nicola Burgess and Claire Worthington Tittle: SimLean: Utilising simulation in the implementation of lean in healthcare 2012 European Journal of Operational Research 219. Pp 188-197
- 33. Souza Brandao de , L., 2009. Trends and approaches in lean healthcare. *Leadership in Health Services*, 22(2), pp.121-139.
- 34. Torkki PM, Alho AI, Peltokorpi AV, Torkki MI, Kallio PE. Managing urgent surgery as a process: case study of a trauma center. Int J Technol Assess Health Care 2006; 22: 255-260
- 35. Thembani Nkomo 2014, Analysis of Toyota Motor Corporation, Harvard journal
- 36. Varkey P, Kollengode A. A framework for healthcare quality improvement in India: The time is here and now! J Postgrad Med 2011;Vol 57:237-41
- Vinodh, S., Arvind, K.R. and Somanaathan, M., 2010. Application of value stream mapping in an Indian camshaft manufacturing organisation. *Journal of Manufacturing Technology Management*, 21(7), pp.888-900.
- 38. Waldhausen JH, Avansino JR, Libby A, Sawin RS. Application of Lean methods improves surgical clinic experience. J Pediatr Surg 2010; 45: 1420-1425.
- 39. Womack, J.J., D.T. Jones, and D. Ross, *The machine that change the world, Thestory of Lean Production.* 1991, New York: Harpert-Collins.
- 40. Womack, J. P., & Jones, D. T. Beyond Toyota: how to root out waste and pursue perfection. Harvard Business Review, (1996). 74(5), 140e158.
- 41. Y. C. Wu, "Lean Manufacturing: A Perspective of Lean Suppliers," International Journal of Operations & Production Management, Vol. 23, No. 11, 2003, pp. 1349-1376.
- Zack J. Zeroing in on zero tolerance for central line-associated bacteremia. Am J Infect Control 2008; 36: S176.e171-e172.
- 43. Zameer Brey, Towards Building a theory of Lean Implementation in Healthcare: Thesis graduate school of business UCT, 2011.
- 44. <u>http://www.levantar.co.uk/images/uploads/What%20is%20Lean%20Manufacturing%20pdf.pdf</u> visited (24-May-2016)



- 45. Donnelly, P. and Kirk, P., 2015. Use the PDSA model for effective change management. *Education for Primary Care*, 26(4), pp.279-281.
- 46. Pandi, A.P., Sethupathi, P.R. and Rajesh, R., 2012. A Conceptual Model for Achieving Global Quality in Engineering Educational Institutions in India. *Procedia Engineering*, 38, pp.3628-3634.
- 47. Nicolay, C.R., Purkayastha, S., Greenhalgh, A., Benn, J., Chaturvedi, S., Phillips, N. and Darzi, A., 2012. Systematic review of the application of quality improvement methodologies from the manufacturing industry to surgical healthcare. *British Journal of Surgery*, 99(3), pp.324-335.
- 48. Shah, R. and Ward, P.T., 2007. Defining and developing measures of lean production. *Journal of operations management*, 25(4), pp.785-805
- 49. De Vore, K., 2008. A six-sigma approach to stability testing. *Journal of pharmaceutical and biomedical analysis*, 47(2), pp.413-421.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



VALUE STREAM MAPPING APPLIED TO A TYPICAL THREE YEAR HIGHER EDUCATION QUALIFICATION

K.R. van der Merwe

Department of Industrial Engineering Nelson Mandela Metropolitan University karl.vandermerwe@nmmu.ac.za

ABSTRACT

Given the challenges that higher education institutions are facing in South Africa due consideration needs to be given to efficiencies associated with the current processes. This article focused on the use of value stream mapping as a means to identify waste in a typical three-year national diploma course in its current format. Statistics and timelines associated with students enrolled for an industrial engineering national diploma at a comprehensive university were analysed and transferred to a value stream map. A comparison of total lead time and value added time revealed ratios that would be totally unacceptable in any other environment. Taking note of these issues could guide academics in the process of re-engineering the entire academic value stream.



1. INTRODUCTION

The current higher education (HE) landscape in South Africa is characterised by unusually high levels of change, a significant component of which, is emanating from forces linked to affordability and efficiency. In the midst of calls for change academics at many universities are in the process of re-designing curricula in line with the the new HEQSF framework and a unique opportunity therefore exists to radically change the way in which qualifcations are structured.

Existing methodologies for offering three-year diploma type engineering courses have been largely unchanged for several decades and have become the de facto blueprint for the new qualifications. In effect, the same content is being re-packaged under new module names and outcomes with no fundamental change to learning process and the associated duration. This article endeavours to bring proven industrial engineering and lean techniques to the diploma value stream with the aim of exposing waste and highlighting current inefficiency. Exposing the levels of waste and the constraints that are crippling flow throughout the process has revealed that there is an obvious and urgent need for a major re-think on the way in which qualifications are offered.

The sections that follow include a brief review of the situation prevailing in HE and an overview of the lean tools used in the analysis of the current three-year engineering diploma. An analysis of the diploma at a selected comprehensive university is conducted and the results discussed.

2. HIGHER EDUCATION IN SOUTH AFRICA

As is the case globally, higher education institutions in South Africa are experiencing a bewildering array of forces aimed at bringing about significant change. The main sources of these forces include reductions in state funding, competition from private learning organisations and a growing resistance to increasing fees [1]. The manner in which they respond to these forces will in many ways determine their future success and in some cases ability to survive. Current responses, to a large extent, depend upon the way the university in question views itself. These views range from the traditional monastery world of careful reflection to the modern marketplace mind-set anchored in quick response to changing requirements. Kirp [2] provides interesting arguments supporting a policy (for U.S. institutions) that includes both viewpoints, the aim being to retain academic integrity with a commitment to need and ability rather than the ability to pay whilst simultaneously recognising the forces of commercialisation. Whether this lofty goal can be achieved or not remains to be seen but it is also evident that South African universities face challenges that are more urgent emanating as they do from a polarised population desperate to gain access to higher education as a means of realising economic freedom. Adding to the complexity of the problem in South Africa is the relatively small tax base [3] where less than two percent of the entire population pay more than fifty percent of the total personal income tax collected. Growing tax revenue to further subsidise higher education is clearly going to be a challenge. Ironically, evidence suggests that national economic growth is dependent upon increased investment in education and training [5].

Although it could be argued that financial challenges are the major force contributing to change in the global higher education landscape there are other challenges that include [4]:

- higher public expectations
- increased parental concern about quality of education
- greater emphasis on university ratings, and
- demographic changes to the student population.

The development of new technologies has also stimulated thought around new ways to organise and deliver programmes. Competition from online learning and the awarding of credits for prior learning has become common practice in private learning institutions [1]. As is the case with most service industries, customers are demanding better quality at a lower cost and those universities that do not acknowledge this trend may do so at their own peril. Social media is believed to be an increasingly important factor in the development of higher academic service quality transparency [6]. In addition, increased student mobility gives students the freedom to choose a university based on the factors discussed in this section

Addressing all the forces that are currently exerting an influence on the higher education sector is beyond the scope of this paper as the final solution will undoubtedly be multi-faceted. Given that funding is unlikely to increase, this paper seeks to address the factor that is arguably the most important and urgent namely that of cost reduction. A number of options for reducing costs exist but for reasons that will be discussed in the following section the lean option of waste reduction and increased flow has been selected.



3. LEAN

Reducing manufacturing costs per unit can be achieved in two ways, either cut costs and maintain output or keep costs static whilst increasing output. In reality most organisations try to realise cost per unit reductions using both methods. Lean is a well-known philosophy associated with waste reduction (and by association, cost reduction) and increased output which is generally referred to as flow [7]. Although developed in the manufacturing sector, lean has since proven effective in a number of other fields such as the service sector and health care. Of particular relevance to this paper are the recent successes achieved by lean in the higher education sector [4].

Lean can be described [4] as a philosophy "intended to reduce cost and cycle time significantly throughout the entire value chain while continuing to improve product performance". As such it clearly focuses on cost reduction and faster throughput, both factors that are relevant and sought after in the context of higher education. A number of scholarly articles have been published that describe successes achieved through lean applications related to administration functions in higher education institutions [1,4,5,8]. Examples of lean applications related to the design and delivery of actual academic programmes are less common. Reasons for this phenomenon are not clear but may reside in the logic that administrative functions are not the core function of a university and are not therefore afforded the almost untouchable monastery status of the academic programmes. Whatever the reason, it is clear that the building pressure to reduce costs and increase throughput will erode this status.

One of the notable exceptions to the trend of neglecting programme design is the work done by Emiliani [9,10, 11] where he outlines the process of redesigning courses using lean principles. This paper seeks to address the topic of course design in a similar fashion but using different lean tools. Kaizen, which has been utilised for more than three decades now, was the tool of choice for Emiliani whereas this paper describes the use of the value stream mapping (VSM) tool.

4. VALUE STREAM MAPPING

Value stream maps are diagrams, usually drawn on A3-size paper, that provide an overview of how a group of processes produce a product or service [12]. According to Emiliani [13] the technique was developed by the Operations Management Consulting Division of Toyota Motor Corporation in the late 1980's with the aim of improving information and material flow. Used almost exclusively in manufacturing for many years the technique has since gained popularity in service industries where it is used to map processes such as order entry, procurement and design [14]. The purpose of doing a value stream mapping exercise is to identify and eliminate waste or non-value adding activities in a series of processes. In the case of conventional value stream maps the waste is normally expressed in terms of time although hybrid versions have been developed that focus on other resources such as energy and water [15].

The procedure for carrying out a value stream mapping exercise can be divided into three stages that include:

- drawing the current state depicting the value stream as it is,
- an analysis phase where waste is identified, and,
- developing the future state which reflects an improved value stream.

The diagrams are completed using a standardised set of icons where possible augmenting these with personalised icons where one does not already exist. Any set of processes can be mapped using this technique with the key being to ensure that the current state truly reflects reality and not someone's perception of reality.

5. MAPPING A TYPICAL 3-YR ENGINEERING DIPLOMA

The mapping exercise that constitutes the main body of this paper was conducted at a comprehensive university in South Africa that offers a range of engineering diplomas that are designed to take three years to complete. Consisting of a planned two-year period where theory is taught in a conventional classroom setting and one year of experiential learning in industry the process appears fairly simple and logical yet almost everyone involved in the system including academics, students and sponsors (parents or students themselves) are aware that it is inefficient. The raison d'etre for this study is to highlight inefficiencies in the diploma value stream using lean tools, or more specifically value stream mapping, whilst the importance of the study is highlighted by the discussion surrounding the need for meaningful change in the way higher education institutions design engineering programmes.

The mapping process was divided into three distinct phases including the drawing of a map depicting the desired current state, amendments to the map to depict reality and an analysis of areas where problems occur.



5.1 Desired current state

Although it goes against accepted procedure, the point of departure for the value stream mapping exercise was to map the process as it was *meant* or *planned* to transpire. The reason for the deviation (the first map is usually an actual current state) was that this step would show the interested parties just how removed from reality the desired state had become.

The desired current state map is shown below in figure 1. The map shows 55 students entering the system annually and the same amount graduating three years later. A two-year period is set aside for students to complete 24 main modules with the goal of passing six every semester (six months). In the original course design each of the six modules was afforded the same amount of time per week in a timetable that had six keys. This has been mapped by numbering each module (1 to 24) and adding a prefix letter (A to F). The industrial engineering department was responsible for the programme, offering the bulk (14) of the modules, while service departments offered the balance.

For reasons of simplification, the timeline focuses on the two-year period containing the theory component of the programme. A total of 474 days were deemed available over this period of which 280 days were value-adding. The balance of the days was either non-value adding (134) or set aside for assessment in the form of exams (60).



Figure 1: Desired state map (own construction using MS Visio)

This scenario represents the original planned programme and even so results in a fairly dismal picture where non value adding activities (or inactivity) account for almost thirty percent of planned time. To anyone intimately familiar with the entire process it will also be evident that these timelines err on the conservative side. Available academic time (planned) has been eroded by administrative functions over the last two decades.

5.2 Actual current state

Experts in the field of value stream mapping advocate that a mapping team should walk the entire process collecting data as it is observed [12]. This type of approach creates a "snapshot" of the value stream at a specific point in time which is fairly meaningless when a two-year process is being mapped. Developing a map to cover this type of event requires that the mapper bases the current state on a typical (or average) student flow. Doing so results in a map that is clearly very different from the desired current state. The complete actual current state map is shown in figure 3.

Although there is considerable variation, the average student takes approximately four years to complete the required 24 modules of theory and 35 percent of each intake drop out altogether. The experiential learning



component of the programme takes just over the planned year - revealing that this is not a major factor in the lengthy throughput time. It may, however, be an important factor contributing to the high drop-out rate. Another significant difference between the two current state maps is that the average student fails six modules during the entire process.

It is also worth noting the addition of a seventh module key was introduced since the original plan in order to help students overcome scheduling issues resulting from timetable clashes



Again using a conservative approach, the value adding time remaining the same, it becomes evident that the non-value adding time is more than four times larger than originally planned.

5.3 Value stream map analysis

The actual current state map shows that students are taking significantly longer to complete the two-year theory component of the engineering diploma in question. There are, of course, repeats where students failed a module and analysis of the map reveals that there are a number of occurrences of "no studies" where the student did not register for a module on a particular key. Eight students were questioned to find out why they do not use all the available keys. In nearly all the cases (only one indicated finances as an issue) they indicated that they were not able to register because of clashes and precedence rules. Verifying this response required that a network diagram be constructed to show the flow of modules and the associated precedence rules. The resulting diagram is shown in figure 3.

Two obvious types of constraints were found to reduce the options available to a student when selecting modules every semester. The first of these factors is linked to the timetable; each module is offered on a certain key (depicted by colour) every semester and therefore the choice is limited by clashes in the module key allocations when two required modules are held in the same time slot. The second factor is as a result of the precedence relationships between certain modules; Work Study 1, for example has to be passed before the student can register for Work Study 2. In total, the programme in question contains sixteen of these precedence rules.





Figure 3: Programme network diagram (own construction using MS Visio)

Timetable and precedence factors, particularly combined, make it very difficult for a student to fill the available keys on his or her timetable. Making this situation even worse is the fact that module keys occasionally have to be changed by the department in order to accommodate lecturer availability changes.

The abovementioned analysis prompted further thinking on the current methodology associated with the design of the three-year diploma (or any diploma for that matter). Leaving aside the reluctance of most academics to consider academic programmes in industrial terms the following anecdotal comparison was made by the mapping team.

"Raw material in the form of students enter the system. Variability among the raw material is certainly high but is also largely unmeasured (except for a system we know is flawed). Virtually every unit is re-worked and approximately 35 percent of the work-in-process is scrapped. When exactly, and for what reason we do not know. The majority of the work-in-process takes roughly twice as long to process as the production plan originally intended. We spend somewhere between 10 and 20 percent of our time inspecting roughly half of the attributes each unit should have. Approximately 60 percent of our time is allocated to either re-working the work-in-process of leaving it in storage. We are also completely unaware of the cost per unit."

Whilst the narrative above is obviously anecdotal it nature it is also cause for concern.

6. CONCLUSION

Global reports [16] would indicate that the non-continuation rates of students studying engineering diplomas in South Africa is not unusually high. Contact programme design is fairly similar around the world and it is reasonable to assume that institutions see no cause for alarm because of this commonality. Cost pressures are certainly being experienced in the higher education sector in other nations but in light of South Africa's unique history and uneven society it is also evident that innovation in programme design is required. The amount of constraints and variability inherent in the system would suggest that there must be a better way, not only for the students but also for academics (who are constantly switching between teaching and research throughout the day). In closing, the system in use is a classic example of an outdated system that was designed to suit the education service providers of yesteryear.



REFERENCES

- [1] Waterbury, T. 2015. Learning from pioneers, International Journal of quality & Reliability Management, 32 (9), pp. 934-950.
- [2] Kirp, D.L. 2004. Shakespeare, einstein, and the Bottom Line, Harvard University Press.
- [3] SARS. 2014. Tax Statistics, National Treasury and South African Revenue Services.
- [4] Comm, C.L. & Mathaisel, D. 2003. Less is more: a framework for a sustainable university, International Journal of Sustainablity in Higher Education, 4(4), pp. 314-323.
- [5] Vauterin, J.J., Linnamen, L. & Martilla, E. 2011. Issues of delivering quality customer service in a higher education environment, *International Journal of Quality and Services Sciences*, 3(2), pp. 181-198.
- [6] Tsai Tan, A.H., Muskat, B & Zehrer, A. 2016. A Systematic Review of Quality of Student Experience in Higher Education, International Journal of Quality and Service Sciences, 8(2), awaiting publication.
- [7] Womack, J. & Roos, aD. 1996. Lean Thinking, Simon & Schuster.
- [8] Comm, C.L. & Mathaisel, D. 2005. An exploratory study of best practices in higher education, *Quality* assurance in Higher Education, 13(3), pp. 227-240.
- [9] Emiliani, M.L. 2004. Improving business school courses by applying lean principles and practices, *Quality Assurance in Education*, 12(4), pp. 175-187.
- [10] Emiliani, M.L. 2005. Using kaizen to improve graduate business school degree programs, *Quality Assurance in Education*, 13(1), pp. 37-52.
- [11] Emiliani, M.L. 2015. Engaging faculty in Lean Teaching, International Journal of Lean Six Sigma, 6(1), viewpoint article.
- [12] Rother, M. & Shook, J. 1999. Learning to See, Lean Enterprise Institute.
- [13] Emiliani, M.L. & Stec, D.J. 2004. Using value-stream maps to improve leadership, *The Leadership & Organisation Development Journal*, 25(8), pp. 622-645.
- [14] Swank, C. 2003. The lean service machine, Harvard Business Review, 81(10), pp. 123-129.
- [15] Davies, E. & van der Merwe, K.R. 2015. Development of a Framework for a Lean based Water and Energy Efficiency Assessment Tool, *Journal of Engineering, Project and Production Management*, 5(2), pp. 98 106.
- [16] Higher Education Statistics Agency. 2014. UK Non-continuation rates, viewed online at https://www.hesa.ac.uk/stats on the 25th May 2016.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



ENGINEERING MANAGEMENT AND BUSINESS INTELLIGENCE: THE IMPORTANCE OF PLANT DESIGN BASE

H.F. Swanepoel^{1*} & Prof. J.H. Wichers²

¹Department of Mechanical Engineering North West University, South Africa Chief Technologist: Engineering, Eskom Holdings (Pty) Ltd <u>swanephf@eskom.co.za</u>

²Department of Mechanical Engineering North West University, South Africa Director: School for Mechanical and Nuclear Engineering <u>Harry.Wichers@nwu.ac.za</u>

ABSTRACT

Most popular business improvement models center their framework and approaches around *business <u>process</u>* improvement - thus on people, process and technology aspects of the business. They tend to drive business process improvement cycle elements, and seldom evaluate the impact of not using high quality critical plant design and control data effectively. As a result, these business models generally struggle to quantify their value proposition as they lack the plant and process data needed to demonstrate/prove value. This paper highlights the importance of understanding and using the plant design base as a primary input to process plant business improvement models and initiatives.

¹ The author is enrolled for a PhD (Mechanical) degree in the Department of Engineering Management, North West University, South Africa.



1. INTRODUCTION: THE IMPORTANCE OF THE PLANT DESIGN BASE

Most of the popular business improvement models centers their framework and approaches around *business* <u>process</u> improvement. Very few of these models, if any, evaluate the impact of not using critical plant design and control data effectively. Limited focus is therefore placed on the ideal set of plant data and information required for business improvement and analytics. As a result, these business models generally find it difficult to quantify the value proposition sold to the business as it lacks the supporting plant and process data that can be used to demonstrate and prove the value proposition.

According to Biehn^[2], data scientists believe that as little as 5% of the "big data" gathered results in 95% of the value contribution of the data. And herein lies one of the biggest problems with data in business today - *effectively identifying, modelling and analysing the 5% critical data to improve business operations*. Many companies gather vast amounts of data, but rarely take the effort to analyse the data or even asking the basic question of *WHY* they are gathering the data. Although data storage costs have significantly reduced, the impact of analysing critical business and plant data when it is buried in 95% of "low value data" has a significant impact on productivity, situational analysis capability, incident response and decision times.

The research study has developed an integrated plant information (IPI) framework and Business Improvement Model (IPI-BIM) depicted below in Figure 1. At the core of this model is the Plant Design Base Data contained in an IPI system environment, with the design base content supported by the required engineering processes and advanced analytics capability to manage it.



Figure 1: The Integrated Plant Information Business Improvement Model (IPI-BIM)

The research study scope included the identification, sourcing, validation and information management requirements of the most critical and core design base content needed for advanced analytics and decision making. Although the research study was focused on the Power Utility environment, any process plant owner would be able to identify with the core design base content listed in Table 3.

Most of the current Power Utility Generation fleet are either mid-design life or post this midpoint in terms of the asset lifecycle (Figure 2). The research scope is thus of great interest to the Utility that is currently undertaking a process of Validating and Verifying (V&V) their plant design base and implementing more extensive configuration control capability for the management of the plant design base. V&V of the entire potential design base can come at significant cost to the Power Utility, so if a reduced critical and core design base can be identified, there is significant cost saving potential for the Power Utility in a very funding-constrained environment.





Figure 2: Eskom Generation Fleet Age [4]

The operational research and prototyping further evaluated whether the identified core design base proposed for the IPI-BIM would support predictive analysis of plant behaviour in order to increase reliability of plant. Proposed system integration, core design base identified, the required data interaction, validation and visualisation capability, all formed part of the operational research effort to prove the IPI-BIM elements and implementation approaches suggested.

The research included an evaluation of information delivery options and technology within the Power Utility to define the most cost efficient, intuitive and effective/productive way to make the information available. More integrated design base data management would also support business improvement by supplying consistent, validated and verified design base data in one central information repository. This paper focuses on a sub-set of the overall research undertaken to build and prove the IPI-BIM Model and Approach. It specifically covers the core design base definition and operational prototyping/research undertaken to prove that it is sufficient to support business improvement initiatives in process plant environments.

2. PLANT DESIGN BASE AS THE FOUNDATION FOR ADVANCED ANALYTICS

The IPI-BIM Framework works on the premise that a systematic and integrated approach to configuring and building the required Integrated Plant Information (IPI) infrastructure and architecture can set up the organisation for success in driving more informed decision making and improved business operations (Figure 3).

Research into the building of Artificial Intelligence (AI) capability and algorithms clearly indicated that it is a research study field in its own right, and the aim of this study was not to build or establish this AI capability and was thus excluded from the research scope. The research aimed to establish the framework capability stack to enable predictive analytics capability level, with the premise that AI would be a natural outflow from the initial predictive analytics frameworks and fault models built during the research prototypes.

Figure 3 indicates how the Plant Design Base *is the most fundamental requirement for advanced plant analytics*. The challenge is to define the core design base data set, specifically on plants that were built before the electronic information era, and confirming that what has been sourced is sufficient to support the advanced analytics capability required for business improvement using plant information.

Searches of the Electrical Power Research Institute (EPRI) research database (www.epri.com) returned very limited information on exactly what is deemed prescribed design base information and scope, and rather focused on actual design examples - confirming the need to define the core design base data set needed for advanced analytics in the proposed Business Improvement Model.





Figure 3: Advanced Analytics Capability Building Blocks ^[15]

The PAS-55 ^[10] framework makes it clear that good decisions should be balanced between deterministic *and* probabilistic analysis to provide the required insights, perspective, comprehension and balance in the decision making process. The PAS-55 framework also emphases the importance of a well-managed design base information and data-set to assist in decision making.

The Nuclear industry study on advanced AI on control systems ^[18] sees the development of advanced algorithms and fault models as exciting new future trends that can have a major impact on operator efficiency in dealing with plant upsets. This industry study confirms the crucial importance of a well-managed, integrated plant database where all critical plant design base information is contained. The study also supports the view that it is key to apply rule sets in the management of information and when dealing with the ordering of large datasets for analytical analysis.

3. RESEARCH SCOPE AND OUTCOMES

Part of the overall operational research performed aimed to define the scope, effort, establishment methodology and extent of design base information in an integrated plant information system (IPIS) hub for a Brownfields plant. As indicated, it is considered a core capability needed to deploy and use more advanced predictive plant information analytical capability to enhance and improve business efficiency and decision making. It is also a key input to achieve the Power Utility SmartUtility Strategy ^[8] desired outcomes.

Part of the research process was to understand and characterise the core elements that make up the critical plant design base for a Brownfields power utility plant and investigating whether there is any difference in content and methodology of establishing the critical plant design base between Greenfield and Brownfields Plants.

Although there is obvious benefit with a reduced core design base (reduced validation and verification [V&V] effort and cost if only the most critical information has to be subjected to this process), it remains important to ensure that the scaled down design base information set is still sufficient to support asset management, advanced analytics and Business Intelligence (BI).

The research effort further included an investigation into the viability of reverse engineering data and plant process design base content. This was required in cases where original design base data-sets could not be sourced with conventional data mining/sourcing processes. Reverse engineering results were subjected to a three-way process of V&V using Plant Operating Simulator, Engineering Simulator/Flow Simulation and Actual Plant performance data evaluation to ensure its correctness.



Subsequent prototyping was then conducted to confirm that the core design base elements identified in the Plant Information System hub is extensive enough in nature to support efficient asset management and predictive diagnostics.

The research also aimed to identify the most efficient methods of information delivery based on user needs, type of user and specific business requirements. The engineering design base is very data-centric in nature and options for delivering it in a more user-friendly and intuitive manner was prototyped.

3.1 Software System Implementation Methodology

The research developed a templatised, standardised, balanced implementation methodology and system configuration that contains the required elements of data content, workflow, engineering business processes and information integration needed to enable and support the engineering organisation and core engineering processes required across the asset lifecycle (Figure 4).



Figure 4: IPI Engineering Business Process Integration [16]

This implemented configuration and capability forms part of the core system framework in the IPI-BIM and acts as the integrated repository for engineering processes (with their associated BI meta-data for analytical purposes), design base data and the 3D plant visualisation platform.

Four Engineering Information Management System Project implementation methodologies were evaluated during the research study (Table 1). The four projects all had the same intent - the implementation of an integrated engineering information management system for the Power Utility that can manage the plant design base and associated workflows.

Although these projects had the same intent and intended outcomes, they followed very different approaches and execution methods. The biggest impact on system implementation timelines are the engineering business process configuration scope and implementation methods. Delays on Projects 2, 3 and 4 were primarily caused by challenges on engineering process workflow configuration. These configurations generally ended up being the critical path items defining/ affecting the project timeline. Enabling a more optimal method to define and configure workflows would therefore add significant value in system implementation scopes of work.



	PROJECT 1 METHOD	PROJECT 2 METHOD	PROJECT 3 METHOD	PROJECT 4 METHOD
Implementation Timeframe (Planned) ^{9}	6 Months	24 Months	9 Months	6 Months ^{2}
Implementation Timeframe (Actual)	6 Months	36 Months	24 Months ^{1}	42 Months (Still Ongoing) ^{2}
Deployment Timeframe	4 Weeks	2 Years	8 Weeks (Prototype)	52 Weeks (Still Ongoing)
Number of Eng. Processes Enabled	6	4	22	8
Extent of 2D Intelligence ^{7}	30%{4}	0%	60%	40%{5}
Extent of 3D Intelligence ^{8}	25% (Point Cloud, Limited 3D Model) ^{4}	10% (Only 3D Review models, no intelligence)	60% (Limited 3D use due to reference data availability) 3D-Pact Ops Simulator Implemented	30% (Reference 3D Models only - reference data still an issue to enable full intelligence 3D Models)
Operational Use	Full operational use of all functionalities created - Used at Power Station for 7 years	Partial - 1 New Build Project, no use in Brownfields Generation plants. No 2D/3D deployed in operational use.	Prototype implementation only - SPO OOTB implementation favored and GEDI Project closed out at feasibility study and concept delivery phase.	Limited use as it is still in deployment phase. Partial Implementation at 2 sites. A number of functionality additions still required for full operational use.
Implementation Cost	R9M ^{3}	R60M	R15M ^{3}	R120M+ ^{6}

Table 1: Research Findings - IPI System Implementation Methodologies

SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Notes:

- {1} Increased timeframe caused by scope creep in number of business processes required. (7 core processes were defined at the start of the project to prove the concept a total of 22 processes were eventually implemented under the full project scope).
- {2} "Out of the Box" (OOTB) Timeframe of 6 months was claimed by vendor. The OOTB implemented solution was not accepted or signed off by business as it did not meet business requirements.
- [3] Implementation cost included the proof of concept implementation of 2D and 3D design base deliverables for Project 3. The other 3 projects only focused on *enabling* the capability to create intelligent 2D/3D deliverables.
- {4} The initial project scope excluded 2D & 3D Deliverables but was added subsequent to initial project scoping.
- The 2D/3D design-base related work undertaken in Project 3 was re-used in Project 4's Implementation.
- [6] Implementation cost for this project *excludes* extensive design base data migration. The project's scope only covers legacy EDMS system replacement efforts.
- {7} Two-dimensional (2D) intelligence implies the conversion of conventional drawings into data-centric "intelligent drawings" where design base content is related to objects on the drawing and available on "query" of the object on the drawing.
- {8} Three-dimensional (3D) intelligence implies the use of visual/three-dimensional information datasets to display the plan design base in a visual format. As with 2D intelligence drawings, 3D model "objects" can be queried for design base information like design criteria, specification values, etc.
- {9} The engineering process workflow configuration would typically dictate the anticipated project timeframe and on all four projects was the "critical path" scope.

From evaluation, the Project 3 Implementation Method was found to be preferred baseline to establish the IPI System implementation approach for the IPI-BIM. Enhancement of this method and the development of a templatised 6-step implementation methodology in the research study resulted in a well-documented rapid application development (RAD) approach to integrated engineering system implementation.



3.2 Engineering Business Process Implementation Methodology

To prove the methodology developed, the enhanced implementation methodology was used on Project 3 to configure and implement the 15 additional engineering processes identified in addition to the original 6 core processes. Table 2 below demonstrates how the structured and templatised approach and methodology developed by the research significantly reduced business process implementation cycle time and configuration error rates.

	COMPLEX ENGINEERING PROCESS	AVERAGE COMPLEXITY ENGINEERING PROCESS	SIMPLISTIC ENGINEERING PROCESS	
Target Processes	1 st Pass: Engineering & Project Change Management (Design & Field Changes) 2 nd Pass: Technical Risk Analysis 3 rd Pass: Technical Documentation Management	1 st Pass: Occurrence & Incident Management 2 nd Pass: Non- Conformance Management 3 rd Pass: Authorisation Management	1 st Pass: Action Management 2 nd Pass: Interface Management 3 rd Pass: Spares Strategy Management	
No of Workflow Steps	>20	10-20	<10	
First Pass Process Configuration & Implementation Cycle Time	42 Days	14 Days	6 Days	
2 nd Pass Process Confi- guration & Implemen- tation Cycle Time	29 Days	11 Days	4 Days	
3 rd Pass Process Confi- guration & Implemen- tation Cycle Time	21 Days	7 Days	2 Days	
Error Rate (UAT/FAT NCR - % rework)	1 st Pass: 22% 2 nd Pass: 19.8% 3 rd Pass: 13%	1 st Pass: 15% 2 nd Pass: 7.8% 3 rd Pass: 4.5%	1 st Pass: 11% 2 nd Pass: 6.3% 3 rd Pass: 2.8%	

Table 2: Engineering Business Process Implementation Method Findings

The projected business process system configuration time for the additional 15 processes was reduced from 9 months to 6 months (with the resulting benefits of reduced project cost and earlier project scope delivery).

3.3 Identifying the Core Design Base

An Intergraph study ^[3] states the goals of having core design base information available as being:

- Reduced Time-to-Market (TTM)
- Maximised Time-in-Market (TIM)
- Optimised Operating Parameters (OOP)

The study very aptly shows how certain information creates a cascading negative business effect later in the asset lifecycle when it is not available, as well as the inter-relatedness of the TTM, TIM and OOP elements.

Numerous examples were sourced from industry players in the EPC space ^[3, 6, 11, 13, 14, 17] to establish the typical Design Base Handover scope that Owner/Operators ask for and measure at handover of plant into commercial operation.

The challenges with obtaining a full design base were demonstrated on the Greenfields Power Plant Project as well as the Brownfields Power Station that formed part of the research study. The research study concluded that the ability to source a full-on design base data set is problematic if a proper and extensively detailed handover information specification was not issued to the Design Authorities/Contractors.

Given the analysis of information sourced/available against the full design base, the research study theorised that it will be possible to identify the set of core design base information most needed by the Owner/Operator



to manage, operate and maintain the asset for the design asset lifecycle timeframe. And to demonstrate that supply/sourcing and validation of this core design base data-set is both possible and feasible.

The research study also theorised that it may be more difficult to source such design base content on an old Brownfields plant, and that there would be a need to do reverse engineering of missing design base content. To prove this hypothesis, design base sourcing was therefore conducted in both a Greenfields and Brownfields scenario to compare the availability of information and determine the need for reverse engineering.

The research study concluded that the design base elements and content needed for a full C&I control system refurbishment (Figure 5) would be the closest requirements definition of what plant and technical information should be deemed *CORE* design base content for a power plant asset (in this case the plant Operating Design Baseline). Further detailed analysis of the typical design artefacts that would define the Operating Design Baseline, is provided in Table 3.

This Operating Design Baseline is typically augmented by a Maintenance Design Baseline for the plant asset, defining the maintenance strategy and tasks for plant equipment based on its criticality classification, operating duty and operating environment conditions.



Figure 5: Core Design Base Content needed for C&I Upgrades

The research study focused further data sourcing and evaluation of design base artefacts listed in Table 3. Although the research study was focused in the Power Utility environment, the commonality of these design base artefacts to most process industries should be evident. This view is confirmed in the numerous examples sourced from industry players in the EPC space ^[3, 6, 11, 13, 14, 17].

Operational research was used to determine completeness and availability of this information in a Greenfields Power Plant Project, as well as an operational Brownfields Power Plant, to confirm that this is a viable hypothesis and design base scope.

Table 3 indicates the actual volumes of design base artefacts sourced for each core design base element. It demonstrates that the information is indeed available, and can be sourced successfully for the Greenfields as well as Brownfields plant scenarios.

Interpretative notes at the end of Table 3 explain anomalies or large differences found with the comparative analysis done in the research study.



CORE DESIGN BASE CONTENTGRAMINES OF DESIGN BASE ARTER INTERS PLANTProcess Flow Diagrams (PFD's)144122Process & Instrumentation Diagrams (P&ID's)4,4673,198Material, Mass & Energy Balances24Plant Process Set-point Lists951 ^[8] Operating Envelopes13513 ^[11] Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control & Operating Philosophies17Control & Derating Philosophies17Control & Derating Specifications27232System Functional Specifications793 ^[9] 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] Call Setpoint List16555Sil Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)01Plant Control & Protection Logic (Including Protections)01Interlocking244 ^[3] 373 ^[3] Protection Systems Connectivity to SCADA Systems01Interlocking244 ^[3] 373 ^[3] Protections)244 ^[3] 373 ^[3] Protections)244 ^[3] 45Switchgear Schedules2,4044,696Equipment Datashets/Data-lists2,404	Table 3: Core Design Base Content Va	ILLE - Brownfields vs Greenfields Plant		
Coccess Flow Diagrams (PED's)144122Process Flow Diagrams (PED's)144673,198Material, Mass & Energy Balances24Plant Process Set-point Lists951181Operating Envelopes13513 ⁽¹⁾ Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ Cdi Setpoint List16555SI Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)023'd Party Control & Protection Logic (Including Interlocking)01Interface Bus Requirements (IT Level)01Interface Bus Requirements (Including Protections)244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule2441,696Equipment Datasheets/Data-lists2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 8Maintenance Technical Specifications0 ⁽¹³⁾ 4,978Test & Inspection Plans723399 ⁽¹⁵⁾ Operating Chenkal Spec	CORE DESIGN BASE CONTENT	GREENEIELDS DI ANT BROWNEIELDS DI ANT		
Process & Instrumentation Diagrams (P&ID's)4.4673,198Material, Mass & Energy Balances24Plant Process Set-point Lists951(8)Operating Envelopes13513 ^[11] Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)0 ^[2] 0 ^[2] Automation Concepts/Strategy023'd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule244 ^[3] 373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[9] Operating Technical Specifications0 ^[13] 33 <td< td=""><td>Process Flow Diagrams (PFD's)</td><td>144</td><td>122</td></td<>	Process Flow Diagrams (PFD's)	144	122	
Atterial, Mass & Energy Balances24Plant Process Set-point Lists951 ^[8] Operating Envelopes13513 ^[13] Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ^[9] 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] Cáll Setpoint List16555Sil Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)01Interlock B Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy02373 ^[3] Drive & Actuator Schedule38373373 ^[3] Drive & Actuator Schedule38373373 ^[3] Drive & Actuator Schedule38373373 ^[3] Drive & Actuator Schedule244 ^[1] 497 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Centrical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[3] Operating Technical Specifications0 ^[13] 33	Process & Instrumentation Diagrams (P&ID's)	4.467	3.198	
Interfact matrixImage of the resultPlant Process Set-point Lists951181Operating Envelopes13513 ⁽¹⁾ Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control & Operating Philosophies17Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ Câl Setpoint List16555Sil Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)01Interlocking)023'd Party Control & Protection Logic (Including Interlocking)01Interface Bus Requirements (IT Level)01Interface Bus Requirements (Including Protections)244 ⁽¹³⁾ 373 ⁽¹³⁾ Drive & Actuator Schedule38373Switchgear Schedules2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 8Maintenance Technical Specifications0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)723399 ⁽³⁾ Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications </td <td>Material Mass & Energy Balances</td> <td>2</td> <td>4</td>	Material Mass & Energy Balances	2	4	
Name Hole: Solid point Lists1351Operating Envelopes13513 ^[11] Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ^[9] 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)01Plant Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023 rd Protections)244 ^[3] 373 ^[3] 373Protections)244 ^[3] 373 ^[3] 145Lectrical Settings Documents (Including Protections)0 ⁽¹³⁾ 145Lectrical Settings Connectivity to SCADA Systems0 ⁽¹³⁾ 8Maintenance Philosophies/Strategy0 ⁽¹³⁾ 8Maintenance Philosophies/Strategy0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[9] Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifi	Plant Process Set-point Lists	95	1{8}	
Operating Liveopes1.531.53Operating Curves (Start-up, Shutdown, Specific Conditions)016Single Line Diagrams8691,321Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)023rd Party Control & Protection Logic (Including Interlocking)01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule38373Switchgear Schedules2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 4Drive & Actuator Schedule38373Switchgear Schedules2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 4Switchgear Schedules0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Control Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical	Operating Envelopes	135	13{1}	
Single Line Diagrams8691,321Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)02Jard Party Control Systems Connectivity to SCADA Systems01Switchgar Schedule244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule244 ⁽³⁾ 373 ⁽³⁾ Switchgar Schedules2441,696Equipment Datasheets/Data-lists2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 8Maintenance Technical Specifications0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Sci23399 ⁽⁵⁾ Operating Technical Specifications0 ⁽¹¹⁾ 33Operating Technical Specifications0 ⁽¹¹⁾ 33Operating Precision Plans723399 ⁽⁵⁾ Operating Technical Specifications0 ⁽¹¹⁾ 33Operating Technical Specifications0 ⁽¹¹⁾ 33Operating Technical Specifications0 ⁽¹¹⁾ 33Operating Technical Specifications0 ⁽¹¹⁾ 33	Operating Curves (Start-up, Shutdown, Specific Conditions)	0	16	
Control & Operating Philosophies17Control Functional Specifications27232System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)023rd Party Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including 	Single Line Diagrams	869	1,321	
Control Functional Specifications27232System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)0(2)0 ⁽²⁾ Qard Party Control & Protection Logic (Including Interlocking)023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 8Maintenance Technical Specifications0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)723399 ⁽⁵⁾ Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Check sheets591 ⁽¹¹⁾ 202	Control & Operating Philosophies	1	7	
System Functional Specifications793 ⁽⁹⁾ 42Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ⁽²⁾ C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ⁽²⁾ 0 ⁽²⁾ Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (Including Protections)244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-Lists2,404970 ⁽⁴⁾ Maintenance Pchlosophies/Strategy0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ⁽⁵⁾ Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33	Control Functional Specifications	272	32	
Operating Manuals0135Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023 rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Drive & Actuator Schedule38373Switchgear Schedules244(³)373 ^[3] Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 4,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	System Functional Specifications	793 ^{9}	42	
Process Alarms List - Alarm Response Procedures21,578Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023 rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Drive & Actuator Schedule38373Switchgear Schedules244(³¹)8Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 4Auton Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[113] 33Operating Check sheets591 ^[11] 202	Operating Manuals	0	135	
Plant Trip & Interlock Schedule2,81712,695 ^[2] C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023 rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[113] 33Operating Technical Specifications0 ^[113] 202	Process Alarms List - Alarm Response Procedures	2	1,578	
C&I Setpoint List16555SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ^[2] 0 ^[2] Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33	Plant Trip & Interlock Schedule	2,817	12,695 ^{2}	
SIL Report (Safety Instrumentation List)171Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0 ⁽²⁾ 0 ⁽²⁾ Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ⁽³⁾ 373 ⁽³⁾ Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ⁽⁴⁾ Maintenance Philosophies/Strategy0 ⁽¹³⁾ 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Operating Technical Specifications0 ⁽¹³⁾ 33Operating Technical Specifications0 ⁽¹³⁾ 33Operating Check sheets591 ⁽¹¹⁾ 202	C&I Setpoint List	165	55	
Instrument Schedule (Preferably including Locations)101304Plant Control & Protection Logic (Including Interlocking)0[2]0[2]Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244[3]373[3]Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970[4]Maintenance Technical Specifications0[13]145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Operating Technical Specifications0[13]33Operating Technical Specifications0[13]33Operating Technical Specifications0[13]33Operating Technical Specifications0[13]33Operating Technical Specifications0[13]33	SIL Report (Safety Instrumentation List)	17	1	
Plant Control & Protection Logic (Including Interlocking) $0^{[2]}$ $0^{[2]}$ Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 202	Instrument Schedule (Preferably including Locations)	101	304	
Automation Concepts/Strategy023rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Plant Control & Protection Logic (Including Interlocking)	0 ^{2}	0{2}	
3rd Party Control Systems Connectivity to SCADA Systems01Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33	Automation Concepts/Strategy	0	2	
Interface Bus Requirements (IT Level)01Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Operating Technical Specifications0 ^[13] 33Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	3 rd Party Control Systems Connectivity to SCADA Systems	0	1	
Electrical Settings Documents (Including Protections)244 ^[3] 373 ^[3] Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Interface Bus Requirements (IT Level)	0	1	
Drive & Actuator Schedule38373Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Electrical Settings Documents (Including Protections)	244 ^{3}	373{3}	
Switchgear Schedules2441,696Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Drive & Actuator Schedule	38	373	
Equipment Datasheets/Data-lists2,404970 ^[4] Maintenance Philosophies/Strategy0 ^[13] 8Maintenance Technical Specifications0 ^[13] 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Switchgear Schedules	244	1,696	
Maintenance Philosophies/Strategy $0^{\{13\}}$ 8Maintenance Technical Specifications $0^{\{13\}}$ 145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ⁽⁵⁾ Operating Technical Specifications $0^{\{13\}}$ 33Operating Check sheets591 ^{{\{11\}} }202	Equipment Datasheets/Data-lists	2,404	970 ^{4}	
Maintenance Technical Specifications0[13]145Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Maintenance Philosophies/Strategy	0{13}	8	
Layout Drawings (Plant Control/Switchgear Rooms)15,2134,978Test & Inspection Plans723399 ⁽⁵⁾ Operating Technical Specifications0 ⁽¹³⁾ 33Operating Check sheets591 ^{11} 202	Maintenance Technical Specifications	0{13}	145	
Test & Inspection Plans723399 ^[5] Operating Technical Specifications0 ^[13] 33Operating Check sheets591 ^[11] 202	Layout Drawings (Plant Control/Switchgear Rooms)	15,213	4,978	
Operating Technical Specifications0[13]33Operating Check sheets591[11]202	Test & Inspection Plans	723	399 ^{5}	
Operating Check sheets 591 ^{11} 202	Operating Technical Specifications	0{13}	33	
	Operating Check sheets	591 ^{11}	202	



Commissioning Check sheets (Running, Start-up & Shutdown)	1,750	833
Panel Standby Check sheets	O ^{6}	O{6}
Control Panel Releases	51 ^{6}	O ^{6}
Plant Simulation / Simulator Strategy	0	1
Simulator User Requirement Specification	0	1
Technical Procedures - Operating	584	893
Technical Procedures - Maintenance	563	199
Plant Occurrences History	0	6,944
Plant FMECA Analysis (Occurrence Statistics)	90 ^{12}	0{7}
Plant Modifications History & Detail	355 ^{10}	1,726

Notes:

- {1} Operating Envelopes are typically described in the OTS and not found as a stand-alone/separate document. Some Operational/Performance Requirements documents were identified that to some extent provided the information needed to define the operating envelope.
- 2} Does not exist as unique entities, but part of the set of Logic Diagrams of the Control System.
- 33 Some of the information was contained in the electrical load list, and some of it in the Control Logic Diagrams.
- {4} Number reflects the unique datasheet entities. This *excludes* equipment data contained in MSExcel format spreadsheets extracted from manuals and other sources.
- [5] Excludes 177 actual test specifications for the Brownfields plant (content defines test specifics and not just test frequency and high level criteria) and 383 for the Greenfields Project.
- [6] Panel Standby checks forms part of the Commissioning Check sheets and not separate entities.
- [7] Statistics derived from Occurrence History captured in the Occurrence Management Process. Data mined and analysed on "as/when required basis"
- {8} Information exists as "Process Design Criteria".
- [9] Delivered as System Descriptions on Greenfields Project
- [10] Greenfields Project still under construction, changes handled under Project Engineering Change Notice (ECN) or Field Change Notice (FCN) process. Number reflects the quantities of these registered under the project.
- {11} Issued as Operating Instructions.
- {12} Safety Study artifacts, feeding into FMECA process.
- 13} Normally a document compiled by the System Engineer for the Plant system once all commissioning, maintenance guidelines and operating information is provided and signed off at hand-over.

A low volume of information & data was found on operating envelopes and expected plant process response behaviours on the Brownfields Plant. This design base content was therefore earmarked as the ideal candidates for the reverse engineering scope of the research study.

3.4 Structuring Design Base content

The research study theorised that it is crucial to implement a structured classification system to identify and manage the plant design base and its associated artefacts more effectively. The Power Utility implemented the IEC 61355 Document Classification standard^[9] in 2008 and this standard was used to define and group the design base artefacts listed in Table 13. Prior to this standardisation, very little consistent structuring or classification methods were employed in the Power Utility to order engineering design base artefacts and content, making assessment of design base artefact maturity and completeness problematic.

The first pass assignment of design base artefacts to the IEC classification standard was done manually for the research study Greenfields Project. It was found to be very time consuming, and the research theorised that it should be possible to define and utilise data analytics rule-sets to assign IEC classes to the documents automatically. The data-mining would focus on the titles of documents and drawings and sometimes internal content (if Optical Character Recognition [OCR] technology was executed on the documents).

The first data-mining algorithmic rule-set was defined and executed as "Iteration 1" on the Greenfields Project data-set. This was then compared against the manual IEC allocation done previously to determine the accuracy of the automated rule-set. A success rate of 53.7% accuracy on IEC level 4 assignments was achieved.



Table 4 summarises how the data-mining and analytical success rate was improved using further iterations and enhancements to the rule-set (and using both Brownfields and Greenfields plant documentation).

Iteration 2 significantly enhanced the success rate, mostly due to the fact that a mature Brownfields Power Station data set was used to enhance the set developed using the Greenfields Project. Many design base artefacts were not available on the Greenfields Project during Iteration 1, reducing the number of artefacts automated (57 artefact types). With Iteration 2, the IEC artefact types were expanded to 107 instances.

Iteration 3 used data from a second Brownfields plant (with additional new data mining keywords and search terms), which resulted in a further notable improvement.

ITERATION 1	ITERATION 2	ITERATION 3*
Greenfields Rule-set	Greenfields enhanced with Brownfields design base data and naming conventions used on older plants	Greenfields enhanced with design base data from 2 Brownfields plants
53.7%	83%	90.3%

Table 4: Automated Data Analytics Mining Tool Success Rate

*Note: For the final iteration, the remainder of the document types was manually assigned by evaluating the document titles in batches and allocating the correct IEC document classes

The productivity benefit and thus usability of this automated classification tool in an operational environment was confirmed by the research - a previously labour intensive, manual IEC class assignment process of nearly 9 months was reduced to 2 months by applying the automated data-mining rule-sets on another power plant of the Power Utility.

3.5 Reverse Engineering Core Design Base content

The research study scope included an element of reverse engineering using the Brownfields Plant Operating Simulator content to re-build and validate the missing design base content identified. This allowed the recreation of the operating envelopes and the expected plant process response behaviours of the Brownfields Plant. The research process and methodology is indicated below, and Figure 6 is an example of the typical process followed by the research study (this process may differ depending on the scope of the reverse engineering exercise).





Figure 6: Reverse Engineering Design Base using Plant Operating Simulator

Notes on Figure 6:

- 1. WW = Invensys WonderWare technology deployed as Plant Data Historian platform.
- 2. etaPRO = engineering simulation software deployed by the Power Utility at the Brownfields Power Station of the research study.
- 3. ODB = Operating Design Base.
- 4. WTP = Water Treatment Plant.

Research data analysis also revealed that there was a significant number of missing or mismatched data-points between the 3 systems used in the reverse engineering process (plant control system, the Operating Simulator and Engineering Simulator). Resolving this data quality issue resulted in a notable improvement in the outcome of the advanced analytics (as certain values in the engineering simulator previously used "interpreted" or estimated values instead of real-time data values from the plant).

Research efforts included alignment of the Engineering and Operator simulator systems in terms of design base content, and then comparison to actual plant performance monitoring data-sets to confirm re-engineered datasets and expected behaviour observed in the two simulators.

The research proceeded to evaluate the load curves (that only existed in unverified paper format) and automated these in the Operating Simulator. Using actual plant data-sets and start-up scenarios, the operating curve was evaluated and confirmed against control system data (confirming that the simulated configuration and outputs mimic real life conditions and behavior). This reverse engineered and developed plant operating curve and analytical model can therefore be used with confidence for further analytics and measurement of plant reliability.

The advanced analytics model built of the Brownfields Plant Rankine cycle (and the outcomes achieved running analytical scenarios on this cycle model), created a useful plant process design baseline.

Introduction of transient process analytics using the FlownexTM software platform further enhanced the simulation capability in the research study. A drawback of the etaPRO Virtual Plant systemTM is that it provides a static analysis of plant process conditions and does not cover transient conditions and the subsequent process condition normalisation usually experienced when plant process conditions change.

3.6 Reducing Information Delivery Complexity

Gentile ^[7] states from his research that there are notable benefits from using data visualisation in an effort to reduce perceived and actual information/data complexities.



Typical benefits listed are:

- Understanding and absorbing complex technical engineering concepts when visually described.
- Better visualisation and understanding of relationships and behavior patterns for operational and business activities.
- Emerging trends and patterns can be recognised and acted on faster.
- The ability to directly interact and manipulate data.
- Complex business concepts can be better implemented in a visual format to enable a new business langue or bring about a change in business paradigms.

Part of the IPI-BIM operational research entailed the evaluation of the various information delivery technologies available within the Utility and defining the most appropriate infrastructure to use. Due to funding constraints, the research brief was to contain the research and data sources prototyped within the realm of the Power Utility's approved software technology stack, and as such research was not undertaken into alternative systems or software solutions.

The research study prototyping scope proposed and proved portal technology as a preferred medium of information delivery in a diverse organisation with different needs and uses for the information contained in business systems. The diversity of information covered is depicted in Figure 7.

The research and prototype scope considered both static and dynamic data sources within the Utility organisation. The frequency of data storage in the source systems varied from continuous on-line data capturing systems to systems where data is captured and trended on a monthly (or infrequent/"as and when required") basis.



Figure 7: Implemented Portal Technology and Scope of Business Information Visible

Further research undertaken demonstrated the ease with which modern 3D CADD Models could be used to enable 3D visualisation capabilities in a portal environment. Being an integral part of the plant design base, the 3D Model of the plant could be enabled as a prototype research to demonstrate the ease of creating an interactive, virtual plant 3D environment on the Greenfields Project (Figure 8) from where design base information can be interrogated.





Figure 8: Visualisation of Design Base Content in 3D interactive plant model

As part of the information delivery research portion, a Proof of Concept (PoC) study demonstrating the integration of the 2D advanced analytics FlownexTM capability into the 3D Plant Operating & Engineering Simulator was undertaken. The PoC focused on the Flue Gas Desulpherisation portion of the plant on the Greenfields Project that formed part of the research study. The reason for the choice was two-fold:

- A more mature 3D model of the Plant Design Base existed.
- A business need for Operating Staff to be trained on a very complex and intricate plant process technology that has not been implemented on a power generation plant in the Utility before.

Figure 9 below shows the outcome of this PoC - controllable parameters linked to a dynamic process simulator (FlownexTM) was enabled (on the right hand side of the User Interface) allowing the end user to make changes to these parameters and evaluate the impact on plant performance and outputs. In the same visual interface it also exposed the end user to the integrated design base information system (on the left hand side of the User Interface) where more design base content can be accessed to evaluate the impact of plant control and parameter changes.

The plant breakdown structure of the Plant Design Base becomes the integration lever to expose more design base information and data when required by the operator/user, creating an "information on demand" capability and reducing potential information overload by exposing too much information in one user interface.





Figure 9: Integrating Advanced Analytics with the 3D Information Delivery and Design Base

It also evaluated of methods to simplify complex plant process models by focusing on core operating design base content known as "Controllable Parameters" in the process plant control environment. Work done in the reverse engineering phase of the research project added significantly to this evaluation.

The research study scope included an evaluation of the viability of creating a similar virtual, interactive and 3D visualisation capability enabled environment for the Brownfields Plant using 3D Laser scanning technology and data outputs.

The research study scope was extended to include a small prototype to prove that the same 3D virtual environment capability can be enabled using 3D laser point cloud scanning technology on top of which design base content can be enabled for a Brownfields plant that does not have access to a modern CADD 3D Model. The prototype outcome confirmed that similar results were achievable and feasible.

4. BENEFITS OF A WELL MANAGED DESIGN BASE

4.1 Better identification of "Plant Hot Spots" and addressing performance issues

The research prototyping showed how effective management, ordering of design base data and analysing plant historical information allowed the Brownfields Plant in the research study to focus on problem plant areas.

The Brownfields Plant could identify the top 10 problem plant areas using improved plant design base data and information classification and management approaches, and it also assisted the research study in the identification of candidate plant areas to use for advanced analytics prototyping.





Figure 10: Top 10 Problem Plant Areas (Brownfields Research Plant) [19]

4.2 Improving Plant Abnormal Condition Analytics

The Design Base information contained in the Integrated Plant Information System platform, will enable the organisation to effectively build, test and refine analytical fault models.

An Emerson study ^[5] confirms that some of the biggest challenges in process plant are overwhelmed operators and complex plant operations that demand a new approach to managing plant more efficiently and predictively. The study confirms that abnormal situation *prevention* is one of the biggest potential productivity gains in the Process Plant operating space.

The research postulated that the IPI Design Base content can be used to efficiently and correctly define, build and prototype advanced analytical models that will make it possible to:

- Measure efficiency of current operating practices against operating design base and make recommendations on remedial action where there is room for improvement.
- Demonstrate that improved early warning failure detection is possible using big data and advanced analytics capabilities to analyse plant operating and control data.
- Measure efficiency of operations during abnormal and/or test conditions of process plant.
- Improve operator training outcomes if training can be integrated with design base simulation capability and predictive capability it will create a highly efficient operator workforce that has the tools and means to timeously act on plant deviations to prevent trips and load losses.

The research outcome confirmed the theory postulated that a combination of analytical methods will most likely be required to provide a holistic plant improvement framework. The analytical methods employed in the overall research study was found to cover all 3 types of analytical techniques - descriptive, predictive and prescriptive^{[12].}

From the research study literature survey, it was suggested that Angeli^[1] most aptly describes the basic research methodology and elements that made up the advanced analytics models of this research study.

The researcher could find practical application of this method already employed on the Brownfields Plant that formed part of the research study, so it made sense to leverage the method further when it came to defining more advanced analytical models on the plant.

Several derivations of the plant ideal state with regards to the Rankine cycle were developed and evaluated in the research. This allows an analysis of the impact when plant operations and maintenance do not align with the requirements of the design base.

Identification of controllable parameters in the Plant Operating Design Base further allowed online continuous trending of plant operations to evaluate how well it is operated within this baseline, and identify areas of



continuous improvement where further technical training can enhance the Operator's ability to better control and manage plant within the required design base parameters.

Fig. 12 shows an example of how the performance comparison was done and trended in the Brownfield Plant's boiler system area.

CONTROLLABLE	PARAMET	ERS - DES	SIGN		GROSS LOAD NET LOAD	399.6 MW 375.6 MW
	ACTUAL	DESIGN	-40	-30 -20	-10 0 10	20 30 40
THROTTLE PRESSURE	16.2 MPag	16.0 MPag	-0.90 R/h			
SH STEAM TEMP	533 °C	535 °C	2.17 R/h-			
RH STEAM TEMP	533.5 °C	535 °C	1.26 R/h			
CONDENSER PRESSURE	3.86 kPaa	4.267 kPaa	-10.57 R/h-			
CONDENSER PRESSURE	5.42 kPaa	5.791 kPaa	-6.65 R/h			
COOL TWR OUT TEMP	22.6 °C	21.9 °C	9.50 R/h-			
SH SPRAY FLOW	43.3 kg/s	0.0 kg/s	8.88 R/h-			
RH SPRAY FLOW	12.6 kg/s	0.0 kg/s	20.41 R/h—			
FEEDWATER TEMP	220.7 °C	225.5 °C	4.81 R/h—			
HOTWELL TEMP	35.2 °C	33.7 °C	0.00 R/h-			
LH PAH GAS OUT TEMP	111.6 °C	131.0 °C	0.00 R/h-			
LH SAH GAS OUT TEMP	144.5 °C	131.0 °C	9.30 R/h-			
RH PAH GAS OUT TEMP	125.8 °C	131.0 °C	0.00 R/h-			
RH SAH GAS OUT TEMP	145.1 °C	131.0 °C	9.78 R/h—			
EXCESS OXYGEN	3.56 %	2.82 %	5.76 R/h-			
INCOMPLETE COMB	1.20 %	1.6 %	-8.68 R/h-			
AUXILIARY POWER	24.00 MW	16.0 MW	72.15 R/h-			
NOX EMISSION	0.0 ppm		LU6 Net U	nit Heat Rate Actu	ial (GPHI) & LU6 Net Uni	t Heat Rate Design
NET UNIT HEAT RATE	10 231 kJ/kWh	9 732 kJ/kV	/h 10500	LU6 CRZ.PNO.A	LUG	CRZ.PN0.D
TOTAL ANNUAL COST	1 842 624 F	Z	版 2 10000	$\sim \sim$		
			9500			
			E Tuo Apr 2016	9:00	9:3	:0

Figure 12: Plant Controllable Parameter Monitoring

The research also successfully demonstrated the impact of design base decisions during plant design, and the long term impact it can have on product output and plant efficiency.

As an example, the original Brownfields Plant design base was one where steam-driven feed pumps formed the basis for both normal as well as stand-by duty. The final implemented design configuration however, was one Steam-turbine driven feed pump with two electrically driven feed pumps as back-up in emergencies when the main pump is not available.

The research prototype evaluated the impact of using the two Electric Feed pumps (EFP) for feed water supply (normally reserved for standby/emergency purposes). This would be contrary to the ideal design base scenario of using the steam-turbine feed pump (SFP) for normal operations.

The impact of this decision on available output generation and GTCHR is significant - higher fuel consumption and heat transfer rate requirements for less MW's output to the electricity grid (Table 5) - which can be a significant impact considering a plant design life of at least 40 years.

PROCESS SCENARIO	GROSS GENERATION (MW)	HEAT RATE - GTCHR (Btu/kWh)
BFPT providing feedwater to Boiler	619.384	7,683
EFP's providing feedwater to Boiler	604.309	7,876

Table 5: SFP vs EFP feedwater Suppl	y Impact on Generation and Heat Rate
-------------------------------------	--------------------------------------



5. CONCLUSION

The research has provided a practical and workable methodology and framework within the overall IPI-BIM to identify, source, validate and implement the core design base information data-set required for plant asset management and advanced analytics to improve business efficiency. It confirmed the importance to have engineering business processes to support the management of the design base content over the plant asset lifecycle. The research study further showed that the choice of system implementation methodology can have a significant impact on cost and timelines involved to implement the IPI system and associated design base contents of the IPI-BIM.

The research demonstrated the benefit of design base data and information structuring into a well-controlled classification system. This greatly assisted in evaluation and identification of core design base candidates. Using the research to introduce an effective and very accurate automation document type classification tool sped up the process of this classification exercise significantly. This was shown to significantly improve the productivity of staff involved with any Design Base back-fit or V&V exercise.

The research managed to successfully define a core set of design base information required for advanced plant condition analytics and associated improved business intelligence. The research hypothesis was proven that C&I Control System Upgrade design base requirements make up the core Operating Design Base content. It was shown to be a feasible scope for design base V&V, regardless of whether it is a Greenfields Project or Brownfields Process Plant.

Where identified core design base content could not be sourced, sufficient alternative information was available to reverse engineer and/or re-build the missing information.

The advanced analytics and plant process modelling/simulation portion of the research study further confirmed that the identified core design base set was sufficient for the research scope of work. The data set could be successfully used to identify opportunities for business improvement using the integrated plant information.

By undertaking the design base definition and additional reverse engineering exercise in the research, and enabling it in Portal technology and 3D visual/interactive format, opened up significant value propositions for the business:

- Improved visibility of plant performance and design base data.
- Increased understanding of plant behavior in upset conditions, and the impact of this on plant output and reliability.
- Online and continuous monitoring of performance against design base values.
- Timeous management of performance deviations in cases where the plant deviates from the target or expected operating envelopes.
- Increased usage (and thus better Return-on-Investment) for the advanced analytics software investment made in the Utility.

By bringing together critical business, plant process and design base information into a single integrated plant information platform, a powerful business improvement capability is enabled using integrated plant information. It empowers business users to make plant and process decisions in the fastest and most efficient manner.

REFERENCES

- [1] Angeli, C. 2004. On-Line Fault Detection Techniques for Technical Systems: A Survey. Department of Mathematics and Computer Science. Technomathematics Research Foundation. Greece
- [2] **Biehn, N.** May 2013. The Missing V's in Big Data: Viability and Value. PROS. www.wired.com/insights.
- [3] **Botterill.** 19 January 2007. SmartPlant Technical Specification Guide: Capitalising on a "Smart Handover" Strategy for the Engineering Enterprise. PPM-05-063. Intergraph Corporation,
- [4] **Department of Minerals and Energy.** 2010. Integrated Resource Plan for Electricity 2010-2030, Revision 2, Final Report. South Africa.
- [5] Emerson. September 2003. White Paper: Reducing Operations & Maintenance Costs. .
- [6] **Fallon, K.K. and Palmer, M.E.** January 2006. EPISTLE Facilities Information Handover Guide, Part I. NISTIR7529. National Institute of Standards and Technology, US Department of Commerce.
- [7] Gentile, B. 29 September 2014. The Top 5 Business Benefits of using Data Visualisation. TIBCO Analytics Product Group, TIBCO Software Inc.
- [8] Hales, K. 2013. Eskom Smart Strategy. Eskom Holdings (Pty) Ltd.



- [9] International Electricity Commission. 2008. IEC 61355: Classification and Designation of Documents for Plants, Systems and Equipment. www.iec.com. Geneva, Switzerland.
- [10] ISO/PC251, ISO Standards Organisation. 15 January 2014. ISO 55001: Asset Management Systems: Requirements. 1st Edition. Switzerland.
- [11] Lewis, S and Thomson, A. 10 September 1998. EPISTLE Process Industries Data Handover Guide, Part II. Issue 2.2
- [12] Lustig, I, Dietrich, N. Johnson, C. and Dziekan, C. 2004. The Analytics Journey. Institute for Operations Research and the Management Sciences.
- [13] **Port, Dr. S.** 2002/2007. Plant Information Management in Offshore Projects: an e-Engineering/e-Business Implementation. Statoil and Intergraph Process, Power & Marine, Intergraph publication DDPP244EA0, Intergraph Corporation.
- [14] Smit, M.J. June 2009. Completion of Power Plant Projects, Commissioning, Take-over from Contractors and Hand-over to the Generation Business. Eskom Holdings (Pty) Ltd.
- [15] Swanepoel, H.F. October 2014. Lethabo Plant Information System Concept: Supporting the Eskom Smart Utility Drive. Eskom Holdings (Pty) Ltd.
- [16] Swanepoel, H.F. October 2012. SmartPlant GEDI Implementation: Training Guideline. Eskom Holdings.
- [17] Smit, M.J. November 2008. Generation Project Management. Eskom Holdings.
- [18] Sun, B.K.H., Kossilov, A. and Negoyan, V. July 1997. Advanced Control Systems to Improve Nuclear Power Plant Reliability and Efficiency. International Atomic Energy Association. Vienna, Austria.
- [19] Wegener, R. & Sinha, V. 2013The Value of Big Data: How Analytics Differentiates Winners. Bain & Company. Atlanta.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE


SELECTION, TRANSFER AND ADOPTION OF NEW TECHNOLOGY IN THE MINING INDUSTRY

F.U. Tuta¹ & M.W. Pretorius²

¹Department of Engineering and Technology Management, University of Pretoria Fefekazi.Tuta@implats.co.za

²Department of Engineering and Technology Management, University of Pretoria tinus.pretorius@up.ac.za

ABSTRACT

The mining industry is known for mass production with minimal technology innovations. The mining industry is known for mass production with minimum technological innovations, owing to the capital intensive operations that cannot be easily changed. Nonetheless, various reasons such as government legislation and improvements in process efficiencies prompt the need for innovation within the industry and when this occurs, the success of such endeavours is of utmost importance. One inescapable threat to this success is the risk of rejection of new technology by end users coupled with the potential losses with respect to capital injected into the venture. To encourage successful technology implementation, the development of methodologies pertaining to the selection, transfer and adoption of new technology in the mining industry is vital. The objective of this research is the evaluation of new technology implementation processes and the investigations conducted offered a framework for use within the mining industry.



1. INTRODUCTION

The identification of alternative energy sources as possible substitutes for electrical power has long been a topic of interest and has recently been made a mandate by the Department of Mineral Resources as a result of the large power constraints experienced by Eskom. The mining industry is known for consuming large quantities of power and thus a reduction on the load imposed on Eskom by the industry is both of national and strategic importance. To this effect, various alternative sources of power were considered by the mining industry in order to identify viable options. The switch from electrical power to other alternatives although mandated and some alternatives viable, is capital intensive and as such, success of any endeavor to be taken is vital. One inescapable threat to this success is the possible risk of rejection of the new technology by end users coupled with the potential losses with respect to capital that would be incurred in the event of failure. It is thus important that the selection, transfer and adoption of new technology are investigated in order to gain insight into the many considerations that need to be taken into account when new technology is being introduced.

Technology selection, being the first stage of the process, refers to the choice of technology to be invested in by an organisation; an all-important decision that can catapult an organisation into possessing a significant market share by creating a competitive advantage in its industry [13]. After the selection of technology, technology transfer, which loosely refers to the relay of technological information from one party (transferor) to the next (recipient) [7], is the next step in the technology implementation process that needs to be executed. Lastly, technology adoption, which refers to the acceptance of transferred technology and its continued use in an organization, is the third and final step of the process [4]. The three stages described above are important to the success of any technology implementation programme and thus warrant further investigation across the various industries. Each of the three stages involves a number of steps to be completed to aid success in technology implementation projects.

1.1 Problem Statement

Viable technology alternatives as sources of energy is a relatively new concept in the mining industry aimed at reducing the electricity generation load placed on Eskom. Despite the well documented advantages of various energy options and taking into consideration the fact that the measurable success of any technology lies with its application, successful selection, transfer and adoption of the new technology is not guaranteed. Failure to successfully select, transfer and adopt new technology places into jeopardy efficacious integration of forthcoming technologically advanced concepts scheduled for implementation in the near future. It is thus imperative that all possible stumbling blocks to the successful integration of new technology are identified and a viable model developed.

1.2 Research Objectives

The various factors governing the selection, transfer and adoption of new technology are investigated to encourage the worthwhile implementation of new technology while also indirectly making headway for future projects. The objective of this research is the evaluation of new technology with regards to the three stages namely selection, transfer and adoption. Ultimately, the goal is the development of a framework for the mining industry that rallies the success rate of technology implementation projects by taking into consideration the factors that affect effective selection, transfer and adoption of new technology. In addition, highlighting the pitfalls that retard the successful implementation of technology projects is also valuable. The research topic will attempt to answer the questions:

- What factors need to be considered for effective technology selection?
- What factors need to be considered for effective technology transfer?
- What factors need to be considered for effective technology adoption?
- What pitfalls risks need to be avoided?
- ٠

2. A CONCEPTUAL THREE-STAGE FRAMEWORK BASED ON LITERATURE REVIEW

Based on an analysis of relevant literature, the following steps in the execution of technology selection, - transfer and -adoption activities were structured to form the basis for a proposed three-stage framework:

2.1 Technology Selection steps

Identification of relevant stakeholders: The identification of relevant stakeholders is the initial step before any other action can be taken [12]. Stakeholders describe all parties who have vested interests in the organisation. Having all stakeholders present during the technology selection phase ensures that due diligence is conducted by all affected parties who will remain accountable for the final decisions taken.



Definition of systems requirements: From a systems engineering point of view, the transferor and the transferee of the technology must, where possible, work together [2] as early as the research phase of the technology's life cycle. This will ensure that all requirements are filtered through to the design, development, production, maintenance and the usage phase of the technology. Ideally, the end user must be present at all stage gates to give approval to proceed with the technology development process so that all dissatisfactions are addressed as and when they occur rather than waiting until towards the end of the life cycle where the costs of the change are astonishing.

Analysis of technology alternatives: All of the possible technology alternatives are selected from a variety of technologies available in the market. It is imperative that the list of possible alternatives is exhaustive and also explores foreign technology that would be beneficial to the organisation [11]. This mandates the organisation to conduct extensive research into current trends and to also obtain free lessons learnt from other organisations that have attempted similar endeavours.

Evaluation of core competencies: Core competency evaluation is pivotal to the selection process [10]; the technology to be selected needs to build on these competencies i.e. enhance the current capabilities. Any of the alternatives identified in the preceding step that complement current competencies must be considered first based on what the organisation perceives to be its core capabilities that provide a competitive advantage against its competitors.

Development of selection criteria: Development of selection criteria will provide a system upon which the various alternatives can be weighed for evaluation against each other [13], thereby ensuring that the same rigorous and fair set of principles are applied at all times. The criteria to be used must include cost implications, opportunities and threats; market trends as well as the existing skills level of the workforce.

Appraisal of alternatives: Appraisal of viable alternatives is the last step in the technology selection process. The employment of the requirements and adoption filters [11] which ensure that issues pertaining to the risk, strategy alignment, government regulations and payback periods are addressed is ideal for this purpose.

2.2 Technology Transfer steps

Core competency zone partnerships: Stemming from the selection of technology that complements existing core competencies, zone 1 category relationships are the most ideal. Zone 1 relationships refer to the employment of cooperative R&D agreements that allow for mutual beneficiation between parties [1]. This type of collaboration is more productive stemming from the initial involvement of the technology end user at the research phase of the technology life cycle. Collaborations also make way for long term relationships that do not just end when the technology is transferred but rather extend to other R&D projects that the end user organisation may be involved in.

Technology transfer environment: Involvement of the transferee at the initial stages of technology development awards the end user the opportunity to describe the type of environment that the technology will be used in [5]. This in conjunction with the intrinsic knowledge of their workforce allows for adaptation of the technology for the receiver's environment. It is thus important that existing conditions are taken into consideration and the technology designed to fit into this environment.

Selection of transfer mechanism: Once again, due to the knowledge of the workforce and challenges faced by the organisation, the technology transfer mechanism must be chosen by the recipient of the technology who is privy to the best possible transfer environment and processes [5]. The chosen mechanism must adequately address the expected reactions, some of which will have been communicated by the various stakeholders identified during the selection process.

Iterative training: Training of the technology end user should not be a single event but rather designed as a recurring process. Employee training by the technology inventor must be planned carefully and aimed specifically at bridging the gap in the capabilities of the organisation's employees with respect to the use of the new technology being introduced. The continuous training will allow end users to learn all aspects of the technology with the aim of reaching a stage where, with thorough extensive knowledge, the end users can identify further opportunities for improvement [9].

2.3 Technology Adoption

Communication of selected technology: Personnel in the organisation must be made aware of the decision to introduce new technology and must be informed of the technology selected as soon as a decision has been made [3]. Reasons behind the choice and decision (inclusive of government endorsed/legal compliance) to



implement the technology should be communicated as early as possible. This step goes a long way in addressing minor issues which, if left unattended, may result in major resistance to the technology adoption process.

Targeted personnel training: Literature revealed that age plays a major role in the adoption of technology [8] and to this effect; the age distribution of employees in the organisation should be taken into consideration. Training facilitators must be made aware and assigned to particular age groups for targeted training sessions. Highly skilled facilitators are thus required for this change management exercise.

Selection of technology ambassadors: Stemming from the targeted training interventions that are aimed at curbing the technology knowledge gap for all employee age groups, ambassadors for each age group [8] must be selected. It is advised that the selected ambassadors be trained as "super users", a process which entails rigorous training of the super users prior to the introduction of the technology to the remainder of the workforce. The super users can then be positioned as change agents during the targeted training workshops where they will endorse the technology and serve to encourage sceptics on the benefits of the technology.

Selection of transfer mechanism: The best decision pertaining to the transfer mechanism can only be made by the end user [5] who is privy to the most knowledge about its employees i.e. the transferee. The transferee can choose from transfer mechanism options which include training of end users at the developers site, training at the final destination/ end user' site or a combination of the two options.

Continuous technology exposure: This goes hand in hand with the iterative training process discussed during the technology transfer phase. Constant exposure to the new technology will help break down all defence barriers [6] as the employees interact with the technology thus dissolving the fear factor. With time, all fears pertaining to the change will be alleviated and the technology gradually adopted by the workforce.

2.4 Proposed Hypotheses

Based on the development of the three-stage framework for the selection, transfer and adoption of new technologies, the following sets of hypotheses are stated for each of the three stages:

2.4.1 Technology Selection

H1: Government regulations and improvement of existing core competencies drive new technology implementation in an organisation.

H2: An exhaustive list of stakeholders and a variety of technology alternatives aid informed technology selection decisions. (PM)

H3: Existing core competencies are the primary consideration during technology selection.

2.4.2 Technology Transfer

H1: Recipient involvement at the design phase of new technology allows for retrofitting of new technology for the recipient's environment which favours technology transfer. (SE)

H2: Collaborative contractual agreements increase the success rate of technology transfer.

H3: Practical training sessions favour successful technology transfer.

H4: Iterative training sessions ensure maintenance of the skills and capabilities required to utilise the new technology.

2.4.3 Technology Adoption

H1: Early communication of selected technology to the workforce assists technology adoption.

H2: Targeted training interventions at the recipient's site promote successful technology adoption.

H3: Technology ambassadors accelerate the process of technology adoption.

H4: The duration of exposure to new technology favours technology adoption.

3. RESEARCH METHODOLOGY

As a first step, a literature study relating to the selection, transfer and adoption of technology was conducted. The review covered relevant definitions, existing models and methods employed in previous studies. This was done for all three stages i.e. selection, transfer and adoption of technology in order to provide the background for the conceptual model to be tested and analysed. Stemming from the literature study conducted and the conceptual model developed, hypotheses for technology selection, transfer and adoption to be tested and either proved or negated by the data collected were formulated.

The data was collected by means of survey questionnaires sent to various respondents within the mining industry. For the purposes of external validity, the framework was tested in other industries as well (see Figure



1). The results of the survey in the other industries are not reported here since the focus of this paper is on a framework for the mining industry. To this effect, two electronic survey questionnaires were developed, one for the mining industry (platinum refinery) and the other for external organisations. The questions included in the surveys were extracted from the proposed hypotheses such that each hypothesis could either be distinctly proven or negated. The survey developed for distribution to multiple industries was identical to the one distributed within the refinery with an additional question that required respondents to indicate the industry within which they operated. Responses from the mining industry were utilised to developed a framework for each of the three stages i.e. selection, transfer and adoption of technology to be tested for validity. Descriptive statistics were applied to the data gathered.



Figure 1: Distribution of respondents

4. RESULTS

Descriptive analyses were conducted on the data collected from a total of 18 responses received within the refinery and the 58 responses obtained for the survey distributed to various industries. Of the 58 responses, 96% of the respondents stemmed from the mining industry and thus the results of the survey were dominated by the mining industry owing to the significant percentage of respondents from this sector.

4.1 Descriptive Analysis

The following section summarises the conclusions from the descriptive analysis for each step of the three stages of selection, transfer and adoption.

4.1.1 Technology Selection

Years of experience: The years of experience of respondents ranged between 15 months and 36 years within the refinery and between 5 months and 37 years in the various industries, thus vast experience from the respondents of the survey.

Motives for new technology implementation: The most common reason for the implementation of technology in the mining industry was found to be the improvement of process efficiencies as this translated to lower running costs and higher metal recoveries thus the generation of income for the business.

Identification of relevant stakeholders: Stakeholders involved were found to be based on internal structures, the importance of the project and the disciplines (knowledge field) involved. The finance department was found to be involved in all decisions made and what stuck out however was the absence of labour union representatives in the list generated by the respondents.

Definition of systems requirements: Mining organisations were found to be involved as early as the design phase of the technology life cycle owing to the value of seeing the technology right through its entire life cycle (cradle to grave) in order to prevent failures which are as a result of avoidable matters.

Analysis of technology alternatives: The respondents explained that the number of alternatives considered was dictated to by the purpose of the project; geographical source of the technology; patented/ novel technologies, availability of suppliers and the intended application of the technology.



Evaluation of core competencies: Existing core competencies were found to be a secondary consideration for technology selection. The cost of the new technology was the primary consideration above core competencies in decision making processes.

Development of selection criteria: Decision making criteria reported included strategic intent; SHEQ aspects; financial aspects; risks appetite; current vs. future needs; alignment of the new technology with existing infrastructure and maturity of the technology.

Appraisal of alternatives: Appraisal methods in use included financial benefits; SWOT analysis; supplier reputation; technology life cycle and ease of use of the new technology.

4.1.2 Technology Transfer

Core competency zone partnerships: Zone 1 partnerships i.e. collaborations and partnership contractual agreements were entered into during technology transfer due to the new era of information sharing and dissemination that in turn gives birth to powerful long term relationships.

Technology transfer environment: It was found that new technology was retrofitted to suite the organisation. The results were considered to be sound when taking into consideration that all operations differ from one section to the next. In line with getting involved at the inception/design phase of the technology life cycle, retrofitting was considered to be the most effective transfer tool because the technology could be tailor made for a specific use within the organisation.

Selection of transfer mechanism: Practical and modular training of employees were the most favoured technology transfer mechanisms. This was found to be in line with the retrofitting of technology to an organisation as discussed above. In addition, it was offered that the type of training intervention employed depended mostly on the type of technology to be implemented.

Iterative training: In line with the modular training discussed above which incorporated changes and improvements in the new technology recently implemented, continuous/iterative training processes were found to be most favoured.

4.1.3 Technology Adoption

Communication of selected technology: Communication of technology implementation was found to be mostly done when the selection process was concluded however, communication upon delivery was also found to occur quite often as well. It was considered that communication of technology implementation upon delivery accompanied by the absence of labour representatives in the technology selection stage was bound to generate feelings of ambush and uncertainty amongst employees.

Targeted personnel training: It was found that combined training sessions were the most common training intervention employed. Age targeted training intervention were rarely used thus all employees were subjected to the same training methods regardless of age. It is believed that this could be the result of the costs associated with age specific training since combined training sessions were found to be the most common, cheaper and more convenient.

Selection of technology ambassadors: Data gathered showed that the use of technology ambassadors was the second most common method for stimulation of technology adoption. The most common method employed was the promotion of the technology's benefits to the intended recipients as well as financial incentives. From the results above, it was found that faith was placed on acceptance of the new technology by placing emphasis on the technology's benefits with regards to improved efficiencies and improved productivity offered by the new technology.

Selection of transfer mechanism: Training interventions were reported to occur at a combination of both the developer's and recipient's sites. Rarely was technology transferred at the developer's site. It was considered that this was because this translated to major costs owing to the large number of employees who would have to be transported to the developers' site for possibly long durations. An alternative offered entailed training of an organisation's employees at another organisation's (third party) site/premises. The third party organisation would ideally be one that had perfected the use of the technology and had gathered enough experience and knowledge to share with other organisations.

Continuous technology exposure: Technology exposure was reported to be based on:

• The rationale that for personnel (operators) who were required to operate the new technology on a frequent (daily) basis, exposure accounted for more than 75%. For employees trained to understand the mechanisms for troubleshooting purposes, exposure accounted for more than 50%. For employees who required background and basic understanding of the technology, exposure accounted for less than 50%.



• The level of interest and commitment conveyed by the employee; complexity of the technology and the employee's level of responsibility.

Impact of duration on technology exposure: Almost unanimously, it was found that the duration of exposure to the new technology had a positive effect on technology adoption. This was considered to be a logical result because familiarity with the new concept eventually resulted in comfort and ultimately reception and acceptance of the new technology. It was explained that in light of the fact that change was intimidating, constant exposure eliminated the unknown, eradicated resistance and replaced all of the fears with excitement and passion to learn more about the new technology.

4.2 Hypotheses Testing

4.2.1 Technology Selection

Hypothesis 1 offered that government regulations and improvements to existing core competencies drove new technology implementation in an organisation. From the data gathered, it was found that this was not the case and that improved process efficiencies were the main driver for technology implementation within the mining industry. The second and third reasons were found to be cost saving initiatives and government regulations respectively.

Hypothesis 2 offered that a holistic list of stakeholders ensured that all avenues were explored and that all possible scenarios were discussed and resolved. Data gathered generated a comprehensive list of stakeholders involved in the selection process. It was also found that the purpose of the technology (strategic vs. incremental innovation) informed the list of shareholders involved. Irrespective of the scale of the technology innovation project, the finance department was however always involved. It was thus concluded that a holistic list of stakeholders varied from one project to the next and that it also differed from one organisation to the next.

Hypothesis 3 offered that existing core competencies were the primary consideration during technology selection. From the data gathered, it was found that in the mining industry, costs of new technology were the main driving factor in technology selection. Existing core competencies were found to be a secondary consideration. The conclusion was made that the proposed hypothesis does not hold true for the mining industry.

4.2.2 Technology Transfer

Hypothesis 1 offered that recipient involvement during the technology development phase allowed for retrofitting of new technology for the recipient's environment which favoured technology transfer. The data gathered validated the offering within the mining industry and it was found that involvement of the recipient organisation during the design phase ensured all requirements were taken into consideration. This also ensured that retrofitting of the technology for the recipient's environment was also incorporated into the design phase.

Hypothesis 2 offered that collaborative contractual agreements increased the success rate of technology transfer processes. From the data gathered, this was indeed found to be the case. Collaborations between the developer and recipient organisations were the most common type of partnerships as these symbiotic relations are maintained throughout the years.

Hypothesis 3 offered that practical training sessions favoured successful technology transfer by bridging the gap between current capabilities and those required to operate the new technology. The data collected substantiated the proposition in that practical training interventions were the most common and effective form of training employed in mining organisations.

Hypothesis 4 offered that iterative training sessions maintained the skills and capabilities required to utilise the new technology. From the data gathered it was found that a combination of iterative training, as per need training and once-off training sessions was employed and that the type of training intervention chosen was informed by the type of technology (retrofit vs. plug and play) that was being implemented. Iterative training albeit most common, was thus not the only type of effective training employed to promote technology transfer.

4.2.3 Technology Adoption

Hypothesis 1 offered that early communication of selected technology to the workforce assisted technology adoption by allowing employees sufficient time to adjust to imminent changes. Data obtained revealed that communication of the selected technology in the mining industry was mostly done after the selection as proposed; however it was also found that the remainder of the time, communication was conducted during the



commissioning stages of the technology life cycle. It could thus not be conclusively stated that the proposition was completely proven and thus remains an area open for further investigation.

Hypothesis 2 offered that targeted training interventions allowed for interaction between the transferor and transferee organisations thereby developing a trust relationship which in turn facilitated the adoption process. Data obtained to test the hypothesis was found to be inconclusive as most mining organisations seldom used targeted training sessions but rather opted for the conventional combined training sessions. The hypothesis was thus neither substantiated nor negated due to inconclusive data obtained.

Hypothesis 3 offered that technology ambassadors accelerated the process of technology adoption. Data collected was found to be inconclusive due to the fact that most mining organisations rarely used technology ambassadors. Instead, the promotion of the technology's benefits was the main stimulant used to encourage technology adoption.

Hypothesis 4 offered that the duration of exposure to new technology favoured technology adoption. Data collected attested to this proposition serving as proof that prolonged exposure greatly encouraged technology adoption.

5 CONCLUSIONS AND RECOMMENDATIONS

An amendment of the conceptual model offered above as informed by the findings of the research is set below. The model is considered to be valid within the confines of the mining industry as informed by the respondents of the surveys who stemmed mainly from mining organisations.

5.1 Technology Selection in the Mining Industry



Figure 2: Revised Technology Selection in Mining Industry

Stakeholders involved in the process will vary from one organisation to the next but it is imperative that labour unions are involved at the early stages in order to prevent labour unrest due to feelings of exclusion. Stakeholders with varying years of experience should be selected to draw from past experiences (seasoned employees) while at the same time taking advantage of current industry trends brought forth by younger employees who keep abreast of technology trends. Improvements in process efficiencies and productivity are the primary motives for technology implementation followed by government regulations. Cost considerations are the main criterion used for selection purposes followed by existing core competencies of the organisation. The number of alternatives for consideration is informed by the purpose, origin and the availability of suppliers of the technology respectively. Appraisal methods employed include financial, technical and SHEQ consideration that are prevalent in the organisation.

5.2 Technology Transfer in the Mining Industry

Figure 3 shows the proposed steps found to be effective in the mining industry.





Figure 3: Revised Technology Transfer in Mining Industry

Involvement of the recipient organisation at the technology design phase is most beneficial and this in turn paves the way for collaborations between the transferor and transferee organisations. Involvement at the initial stages allows for technology retrofitting and ensuring that the new technology is designed to align with the existing systems within the recipient's organisation. Practical training as the most effective (followed by modular training) enables employees to possess the skills required for operating the new technology. Practical training is complemented by continuous training interventions aimed at retraining employees on an agreed upon basis. As per need training is also an effective training mechanism that is conducted as and when required.

5.3 Technology Adoption in the Mining Industry

The technology adoption steps in the mining industry are as described in figure 4.





The communication of impending changes must be conducted immediately after the technology selection step; change management processes must also be kick-started at the same time in order to address employees



concerns and fears. As was the case with technology transfer, targeted training is seldom used in favour of combined training sessions that are less costly than age specific training sessions. The benefits of the technology must be highlighted and where deemed necessary, ambassadors must be selected to endorse the new technology. Training of employees at both the transferor and the transferee's sites allows for effective training sessions and where possible the use of third party training facilities can be utilised. The duration of technology exposure is informed by the purpose and frequency of use of the technology, which for direct end users warrants prolonged exposure in order to obtain the skills required and to also enable troubleshooting abilities which favour technology adoption.

5.4 Recommendations

5.4.1 External Validity

As a result of a significant number or respondents to the surveys being from the mining industry, the findings of the current research cannot be generalised to other industries and thus could not be externally validated. The findings are therefore limited to the mining industry and it is proposed that the modified model be tested for external validity outside the confines of the mining industry.

5.4.2 Technology Ambassadors

The use of technology ambassadors to encourage technology adoption was not completely verified during the current study and as such, the benefits (if any) of technology ambassadors need to be investigated further.

5.4.3 Age Targeted Training

Owing to the fact that targeted training sessions were seldom used in the mining industry, it was not established whether or not the employment of various training techniques per age group provided any benefits. Further studies into this aspect of technology adoption need to be conducted in order to obtain conclusive data on the subject.

REFERENCES

- [1] Amesse, F. & Cohendet, P. 2001, "Technology transfer revisited from the perspective of the knowledgebased economy", Research Policy, vol. 30, no. 9, pp. 1459-1478.
- [2] Gibson, D.V. 1994, R & D collaboration on trial: the Microelectronics and Computer Technology Corporation, Harvard Business Press.
- [3] Hall, B.H. & Khan, B. 2003, Adoption of new technology.
- [4] Karahanna, E., Straub, D.W. & Chervany, N.L. 1999, "Information Technology Adoption Across Time: A Cross-Sectional Comparison of Pre-Adoption and Post-Adoption Beliefs", MIS Quarterly, vol. 23, no. 2, pp. 183-213.
- [5] Khabiri, N., Rast, S. & Senin, A.A. 2012, "Identifying Main Influential Elements in Technology Transfer Process: A Conceptual Model", Procedia - Social and Behavioral Sciences, vol. 40, no. 0, pp. 417-423.
- [6] Luthra, S., Kumar, S., Garg, D. & Haleem, A. 2015, "Barriers to renewable/sustainable energy technologies adoption: Indian perspective", Renewable and Sustainable Energy Reviews, vol. 41, no. 0, pp. 762-776.
- [7] Madu, C.N., Chinho, L. & Kuei, C.-. 1998, "A goal compatibility model for technology transfers", Mathematical and Computer Modelling, vol. 28, no. 9, pp. 91-103.
- [8] Morris, M.G. & Venkatesh, V. 2000, "Age differences in technology adoption decisions: Implications for a changing work force", Personnel Psychology, vol. 53, no. 2, pp. 375-403.
- [9] Parente, S.L. & Prescott, E.C. 1994, "Barriers to Technology Adoption and Development", Journal of Political Economy, vol. 102, no. 2, pp. 298-321. Phaal, R., Farrukh, C.J.P.
- [10] Ruder, K. A., Pretorius, M. W. & Maharaj, B. T. 2008. A technology selection framework for the telecommunications industry in developing countries. 2008 Ieee International Conference on Communications, Proceedings, Vols 1-13. Ch, pp 5490-5493.
- [11] Shehabuddeen, N., Probert, D. & Phaal, R. 2006, "From theory to practice: challenges in operationalising a technology selection framework", Technovation, vol. 26, no. 3, pp. 324-335.
- [12] Stacey, G.S. & Ashton, W.B. 1990, "A structured approach to corporate technology strategy", International Journal of Technology Management, vol. 5, no. 4, pp. 389-407.
- [13] Torkkeli, M. & Tuominen, M. 2002, "The contribution of technology selection to core competencies", International Journal of Production Economics, vol. 77, no. 3, pp. 271-284.



CONCEPTUALISING GLOBAL MINERAL VALUE CHAINS: A TOOL FOR MINERAL POLICY DESIGN

Wouter Bam

Department of Industrial Engineering, Stellenbosch University, South Africa <u>wouterb@sun.ac.za</u>

ABSTRACT

This paper presents a model by which policy makers can conceptualise the shape and impact of mineral value chains (MVCs) and how their decisions influence the local footprint and value capture of mineral related activities. The model combines a high-level value chain perspective with the triple bottom line dimensions of sustainable development. The proposed model aims to stimulate increased discourse surrounding a more holistic view of value capture from mineral value chains. This is done by investigating the role of productive actors, government and civil society and how each of these players' roles interlink.



1. INTRODUCTION

The last fifty years have seen the metals producing and consuming nations moving into different camps, with the major consumers no longer being the major producers. This has led to consumer countries increasingly moving to adopt policies that ensure the supply security of critical minerals. On the other hand, producing countries have increasingly become prone to resource nationalism and the adoption of policies that aim to maximize the benefit achieved from the extraction of their resources [1]. However, this is often a difficult task where there may be trade-offs between short to medium term value for the nation and the long term sustainability of their resource activities [2].

In some cases, this resource nationalism has found expression in policies that are aimed at localising the downstream processing of minerals to the mineral producing countries. The reasoning behind this usually being that the local processing of minerals will result in greater economic value add, an improved balance of trade and the creation of more jobs [1-5].

Academia has also been investigating several aspects regarding the increased sustainable value capture from mineral resources in support of the drive to ensure more benefit from mineral endowments for mineral producing countries and to combat the so called "resource curse". This has included research on aspects ranging from the role of corporate social responsibility (CSR) in development [6, 7], managing small scale and artisanal mining (ASM) for improved regional outcomes [8], trust between different stakeholders in promoting sustainable development [9] and incentivising the recycling of minerals [10].

However, there appears to have been limited recent attempts to collate the research into a conceptual framework to enable policy makers to consider the wide ranging impacts of their resource policies in a holistic manner. This is particularly important due to the increased emphasis on localising mineral processing activities where multiple factors need to be considered. The nearest to such a framework that the author is aware of is the conceptualisations suggested by Franks et al. [11] for representing cumulative impacts in resource regions, presented in their introduction of the special issue of the journal, *Resources Policy*, on 'Understanding and Managing Cumulative Impact in Resource Regions'.

This paper aims to address this gap by proposing a model conceptualising mineral value chains (MVCs) with a focus on the downstream processing of extracted minerals, the value capture from the local MVC footprints and the factors that influence both the location of and value capture from mineral processing activities. The framework combines a high-level value chain perspective similar to that used by authors such as Gereffi and Lee [12] and Barrientos et al. [13] with the triple bottom line (TBL) dimensions of sustainable development (economic, social and environmental) that has become increasingly important in various areas of research [14].

From this dual perspective, the conceptual model was inductively constructed through reviewing 304 articles published in the journal *Resources Policy*. These articles were selected by reviewing all the articles published in this journal since the start of 2010 up until the 12th of July 2015. The identification of these articles was based on a Scopus® search on the latter date for all articles from this journal, filtered by the requirement of being published in or after 2010. Instead of delivering the final word, the proposed model aims to stimulate increased discourse surrounding a more holistic view of value capture from mineral value chains, particularly for mineral rich countries and to provide policy makers with a starting point from which to consider their mineral related policies.

The first element of the conceptual model, presented in Section 2, focuses on the elements of a mineral value chain. The second element of the conceptual model, presented in Section 3, is the national value capture from mineral value chains. The third element of the conceptual model, presented in Section 4, focuses on the factors that influence the shape of the mineral value chain.

The final element of the model focuses on the factors that influence the value capture from mineral value chains, presented in Section 5. In Section 6, these four elements are collated into a single conceptual model. Finally, Section 7 provides concluding remarks to the article.

2. CONCEPTUALISATION OF THE MINERAL VALUE CHAIN

There are various ways in which to conceptualise a mineral value chain. The specific focus of a study, the mineral considered and the types of policies being investigated will dictate which of these conceptualisations may be more applicable. This section presents a typical conceptualisation of the MVC for the investigation of the location mineral processing activities. Thereafter, other considerations that might be important in the current mineral related policy environment are discussed.

In order to analyse the location of downstream mineral processing activities, the proposed conceptualisation integrates the theoretical work of Humphreys [1] focussing on the tension between producing and consuming



countries for control over the processing steps of the mineral value chain and the GVC conceptualisation work of authors such as Gereffi and Lee [12] and Barrientos et al. [13]. This proposed conceptualisation is shown in Figure 1.



Figure 1: Proposed conceptualisation of the local footprint of the mineral value chain within a global context.

This conceptualisation allows for the evaluation of the structure of MVC by distinguishing between the extraction source of the mineral and the market where the products using this mineral are eventually used or consumed. Between these two poles, the model allows for the evaluation of where manufacturing takes place - in the focal country or not. It could be expected that the refining and processing takes place close to the extraction activity in order to minimize the costly transport of unprocessed ore. It could also be expected that the final assembly might take place close to large final markets in order to minimize the transport cost and lead times to the final consumers. The manufacturing, forming and assembling might take place somewhere in between, though this need not be the case, and different factors play a role in determining where these activities are located.

By conceptualising the MVC in this way, it is possible to glean new insights regarding the location of MVC activities and the drivers of this location. This high level conceptualisation will have to be adapted for different minerals depending on the different and possibly parallel processing steps, the importance of recycling and other mineral specific properties.

Other considerations that might be important in the current mineral policy environment include the distinction between large scale and artisanal mining (ASM), the clustering of activities, linkages and the mineral life cycle. These are briefly discussed here in turn.

Recent literature emphasises the distinction between and consideration of both large scale conventional mining and artisanal and small scale mining (ASM). The failure to distinguish and accommodate both can lead to sub-optimal outcomes, particularly on the local social level [8, 15-27].

Another important consideration is the clustering of activities. Activity clusters can generate critical mass and unlock positive feedback loops that generate additional growth and activities. They can also assist in overcoming constraints through the pooling of resources. However, due to the often observed mining revenue flow to cities, abroad to shareholders and to fly-in fly-out employees and the flow of royalties to national governments, these clusters are often not allocated sufficient resources and underexploited. This can lead to the suboptimal development of mining regions [24, 28-36].

Related to clusters is the consideration of linkages. Linkages (both upstream and downstream) are also important in terms of the shape of MVCs. Increasing them allows governments to increase the local footprint of MVCs and the value capture from them [37]. Fessehaie [30], Morris et al. [37] and Hanlin and Hanlin [38] emphasise the importance of "deeper" upstream linkages that stretch beyond the first tier to ensure that the maximum local activity footprint is achieved.

Toward better environmental management, Fleury and Davies [39] conceptualises the MVC as part of a wider life cycle. This life cycle starts with the societal demand for minerals and includes the disposal, reuse and recycling of minerals. By conceptualising minerals in this way, policy makers can consider the effects of minerals after extraction to the point of disposal.

This section presents a generic conceptualisation of the MVC to analyse the location of mineral processing activities. Thereafter, other considerations that were identified during the literature review that may be important to incorporate into a MVC conceptualisation for other policy related studies were identified. The



following section extends the proposed view of the MVC by incorporating the sustainable value capture dimensions of the triple bottom line.

3. NATIONAL VALUE CAPTURE FROM MINERAL VALUE CHAINS

The value that is captured from the local footprint of MVC activities can be conceptualised according to the three sustainability dimensions of the *triple bottom line* (TBL) concept, popularised by Elkington [40]. Each of these dimensions can also be conceptualised as entailing gain and cost, as the complex impacts of mineral related activities can have both positive and negative effects on the host nations. This is illustrated in Figure 2.

The triple bottom line, consisting of an economic, social and environmental dimensions, has been increasingly used in supply chain literature and provides a useful conceptualisation of the aspects related to the sustainability of a variety of activities and how these activities translate to value [41-44]. The distinction between gain and cost is somewhat problematic, as choosing in which category certain impacts fall might be a matter of perspective. However, the goal of distinguishing between the two categories is to support decision-making by ensuring the consideration of both positive and negative aspects. A number of aspects can be identified to form part of each of these dimensions of value capture. These are each considered in Section 3.1 through 3.3.



Mineral value chain

Figure 2: The national value capture from the local resource value chain footprint.

3.1 Economic value capture

Some direct economic benefits from mineral related activities cited in literature include capital investment [32, 45, 46], employment sustained [1, 29, 32, 46-48], economic growth [32], improved income levels [1, 32, 46, 48, 49], local development, lower inequality [48, 50], shareholder returns [32], tax revenue [1, 21, 32, 51, 52] and local value added [47]. Indirect economic benefits include the appreciation of the national currency [32], a diversification of exports [53], the stimulation of downstream activities, employment in other sectors,



particularly in transport, rental and accommodation [1, 29, 36, 54], infrastructure development [37, 45] and other linkages [4, 5, 32, 37, 38, 45, 46, 55-62].

There may also be several economic costs related to the exploitation of minerals that should be considered. These have been identified to include the opportunity cost, which is particularly important due to the relatively capital intensive nature of mining and thus might translate to fewer jobs for the equivalent investment [32, 54], the exhaustion of minerals [32], the outflow of revenue to foreign investors [28], low job spill-overs into manufacturing and agriculture [54], poorer financial development [63], regional economic problems such as low entrepreneurial activity, pressure on local services and infrastructure, specialisation on minerals and unaffordable housing [32, 35, 64], resource dependence of the economy which may lead to the crowding out of other sectors, instability and insecurity due to the dependence on the fluctuating external demand [32] and possible slower national economic growth [65].

3.2 Social value capture

Some of the potential factors related to the social gain from resource related activities include better education [49], a reduction in poverty [66], improved communication access [49], an increase in disease prevention initiatives [67], improved income levels [1, 22, 32, 49], lower unemployment [48] and a more equitable distribution of income [1]. Some of the social problems that have been associated with resources related activities include conflict [21, 65, 68], corruption [65, 69], the displacement and relocation of communities [21, 70, 71], the disruption of traditional cultures [72], risks to human health such as deaths through contamination, mine injuries, the increased spread of sexually transmitted diseases, mental health issues, addiction, family stress, increased violence towards women and the intake of toxins through contaminated water [73-75], the aggravation of societal issues through the white washing effect of corporate social responsibility [72, 76], poor working conditions [35], poorer education [32], inequality [65], rentier states [65], stranded regions [32] and human rights suppression [75].

3.3 Environmental value capture

As far as minerals and the environment is concerned, the focus in literature and elsewhere is generally on minimising the negative impacts on the environment. However, positive impacts such as man-made habitats for protected birds have been reported [77]. Some of the reported negative effects include deforestation [35], degraded recreational resources [78], general environmental degradation [35, 69, 75], erosion [35, 78], the loss of fauna [72], the loss of habitats for fauna [78], the introduction of non-native species [78], the pollution and contamination of the environment leading to air quality degradation [11, 73], ecosystem degradation [73, 78] and heavy metal contamination [75], toxic spills [75], risks to clean water supplies [11, 35, 78] and visual degradation [11]. There has also been an increasing trend towards the cumulative assessment of impacts as to not view impacts in isolation [11, 75, 79, 80].

4. FACTORS INFLUENCING THE SHAPE OF MINERAL VALUE CHAINS

The factors that influence the location of mineral processing activities and thus the shape of mineral value chains can be conceptualised as enabling and constraining factors that influence the context for stakeholder decisions that ultimately determine the shape of mineral value chains. These decisions again impact on the enabling and constraining factors. This is illustrated in Figure 3.



Figure 3: Factors influencing the shape of mineral value chains.

The enabling factors identified include access to capital [30, 53], expectation of sustained demand [81, 82], clusters and corridors [11, 29], sizeable domestic markets [57], improved technology and processing techniques [30, 53, 83, 84], infrastructure [30, 81], local suppliers [29, 38], competitiveness supported by labour productivity, management practices and quality ore [85], supportive policies [29, 30, 38, 70], continued research



and development [33, 60, 86], the regulation and formalisation of industries [21], local skills [30], subsidies [10], tax breaks [1, 21] and trust [38].

The constraining factors identified in the literature review include demand and price volatility [32, 87], competition [32, 81], environmental concerns [83], exhaustion of resources [32], increased cost of production [32], land acquisition difficulty [53], a restrictive and uncertain legislative environment [82, 88, 89], logistical costs [47], physical infrastructure [29, 57], a decrease in investor confidence brought on by resource nationalism [1], substance restrictions [39], supplier readiness [38] and the availability of water [83].

In the environment of the enabling and constraining factors at a given time, different stakeholders make decisions that impact the shape of the mineral value chain. These include end-users making product choices determined by a variety of factors. These may also include their perceptions on the sustainability of the products that they are buying [39].

Governments also influence the shape of mineral value chains through policy stances, which may vary from neoliberal to resource nationalistic [90] and may directly interfere through, for example, the nationalisation of mines or the protection of specific industries [3, 91, 92]. Company decisions also influence the shape of mineral value chains through supplier decisions and procurement policies [38, 39]. Other influencing stakeholders may include employees, NGOs and local communities [21].

5. FACTORS INFLUENCING THE VALUE CAPTURE FROM MINERAL VALUE CHAINS

Similar to the impact drivers suggested by Franks et al. [11], this section presents a conceptualisation of the value capture from mineral value chains as being determined primarily by the actions of productive actors within the context created by government and other stakeholders as illustrated in Figure 4.



National value capture

Figure 4: An illustration of the factors influencing the value capture from mineral value chains.

The actions of companies and other individuals and entities involved in productive activities play a predominant role in the value capture outcomes of mineral related activities [7, 11, 72, 90, 93]. These actions may be influenced by the boundaries set by government [11, 28, 36, 70, 73, 89, 90, 94- 101], the business model and values of the companies or individuals involved in productive activities [93], NGO's [21] or civil society and local communities [11, 90, 99].

These non-productive actors fulfil a crucial role by influencing the actions of government and those involved in productive activities. This goes some way in explaining the increasing emphasis in literature on the so called "social licenses to operate" [96, 102-115] and multi-stakeholder forums and collaboration to achieve better outcomes for all the stakeholders involved [72, 80, 103, 116].

6. COLLATED CONCEPTUAL MODEL

The four conceptual aspects presented in Section 2 through 5 can be collated into a single conceptual model, as shown in Figure 5.





Mineral value chain

Figure 5: Collated mineral value chain value capture conceptual model.

The proposed model is a major simplification of all the dynamics at play. In particular, there may be factors that influence the shape of MVCs and the value capture from them. For example, a policy of higher mineral taxation may increase the short to medium term income from minerals for the country, but may lead to less investment in the host nations over the medium to longer term.

The model also does not explicitly account for feedback loops that may exist in the system. For example, the level of national value capture achieved from minerals may influence the expectations and demands of civil society, which may, in turn, influence how they interact with other stakeholders to influence the national value capture.

By simplifying the dynamics at play the model aims to provide a starting point for mineral policy discussions that ensure a sustainable development agenda and a best practice perspective. It also serves to support an appreciation of the important roles of different role players in the MVCs within host countries. This is achieved by providing a clearer conceptualisation of how policies might impact the local footprint of and national value capture from mineral resource value chains.

7. CONCLUSION

This paper aimed to address the need for a conceptual model by which policy makers can conceptualise the shape and impact of mineral value chains and how their decisions influence the shape of and value capture from mineral value chains. The model combines a high level value chain perspective with the triple bottom line dimensions of sustainable development. From this dual perspective, the conceptual model was inductively constructed by reviewing 304 articles published in the journal Resources Policy since the start of 2010 up until the 12th July 2015. Instead of delivering the final word, the proposed model aims to stimulate increased discourse surrounding a more holistic view of value capture from mineral value chains, particularly for mineral rich countries and to provide policy makers with a starting point from which to consider their mineral related policies.

The first element of the conceptual model focuses on the elements of a mineral value chain. Although various possible dimensions are identified, the proposed conceptualisation focusses on the producer-consumer dimension of the value chain. The second element of the conceptual model is the national value capture from mineral value chains. It explores the positive and negative value capture aspects of the economic, social and environmental



sustainability dimensions from mineral related activities identified in the literature review. The third element of the conceptual model focuses on the factors that influence the shape of the mineral value chain. It explores the role of stakeholders decisions and how these are influenced by local and global enabling and constraining factors. The final element of the model focuses on the factors that influence the value capture from mineral value chains. It emphasises the role of productive actors in bringing about the different aspects of local value capture, while illuminating the role of government and civil society to influence the actions of these productive actors.

The proposed model captures the current thinking on mineral value chains and how they translate to value for host nations. The model also highlights the tensions between ensuring a greater local value chain footprint, while ensuring a more positive value attainment from this footprint.

REFERENCES

- [1] D. Humphreys, New mercantilism: A perspective on how politics is shaping world metal supply, Resources Policy 38 (3) (2013) 341-349, ISSN 03014207.
- [2] M. Kahn, Natural resources, nationalism, and nationalization, Journal of the Southern African Institute of Mining and Metallurgy 113 (1) (2013) 3-9, ISSN 22256253.
- [3] E. Machacek, N. Fold, Alternative value chains for rare earths: The Anglo-deposit developers, Resources Policy 42 (2014) 53-64, ISSN 03014207.
- [4] J. Wu"bbeke, Rare earth elements in China: Policies and narratives of reinventing an industry, Resources Policy 38 (3) (2013) 384-394, ISSN 03014207,
- [5] L. Zhang, Q. GUO, J. Zhang, Y. Huang, T. Xiong, Did China's rare earth export policies work? Empirical evidence from USA and Japan, Resources Policy 43 (2015) 82-90, ISSN 03014207.
- [6] A. Ackah-Baidoo, Enclave development and 'offshore corporate social responsibility': Implications for oilrich sub-Saharan Africa, Resources Policy 37 (2) (2012) 152-159, ISSN 03014207.
- [7] E. Gilberthorpe, G. Banks, Development on whose terms?: CSR discourse and social realities in Papua New Guinea's extractive industries sector, Resources Policy 37 (2) (2012) 185-193, ISSN 03014207.
- [8] G. Hilson, J. McQuilken, Four decades of support for artisanal and small-scale mining in sub-Saharan Africa: A critical review, The Extractive Industries and Society 1 (1) (2014) 104-118, ISSN 2214790X.
- [9] L. Marais, Resources policy and mine closure in South Africa: The case of the Free State Goldfields, Resources Policy 38 (3) (2013) 363-372, ISSN 03014207.
- [10] N. Johansson, J. Krook, M. Eklund, Institutional conditions for Swedish metal production: A comparison of subsidies to metal mining and metal recycling, Resources Policy 41 (1) (2014) 72-82.
- [11] D. M. Franks, D. Brereton, C. J. Moran, The cumulative dimensions of impact in resource regions, Resources Policy 38 (4) (2013) 640-647.
- [12] G. Gereffi, J. Lee, Why the World Suddenly Cares About Global Supply Chains, Journal of Supply Chain Management 48 (3) (2012) 24-32.
- [13] S. Barrientos, F. Mayer, J. Pickles, A. Posthuma, G. Gereffi, Decent Work in Global Production Networks, International Labour Review 150 (3-4) (2011) 300-3016, ISSN 00207780.
- [14] G. M. Mudd, The Environmental sustainability of mining in Australia: key mega-trends and looming constraints, Resources Policy 35 (2) (2010) 98-115, ISSN 03014207.
- [15] F. Andriamasinoro, J.-M. Angel, Artisanal and small-scale gold mining in Burkina Faso: Suggestion of multiagent methodology as a complementary support in elaborating a policy, Resources Policy 37 (3) (2012) 385-396, ISSN 03014207, doi:\let\@tempa\bibinfo@X@doi10. 1016/j.resourpol.2012.04.004.
- [16] M.-R. Bashwira, J. Cuvelier, D. Hilhorst, G. van der Haar, Not only a man's world: Women's involvement in artisanal mining in eastern DRC, Resources Policy 40 (1) (2014) 109-116, ISSN 03014207.
- [17] J. Childs, A new means of governing artisanal and small-scale mining? Fairtrade gold and development in Tanzania, Resources Policy 40 (1) (2014) 128-136.



- [18] M. J. Clifford, Pork knocking in the land of many waters: Artisanal and small-scale mining (ASM) in Guyana, Resources Policy 36 (4) (2011) 354-362, ISSN 03014207.
- [19] S. Geenen, Dispossession, displacement and resistance: Artisanal miners in a gold concession in South-Kivu, Democratic Republic of Congo, Resources Policy 40 (1) (2014) 90-99, ISSN 03014207.
- [20] S. Geenen, A dangerous bet: The challenges of formalizing artisanal mining in the Democratic Republic of Congo, Resources Policy 37 (3) (2012) 322-330, ISSN 03014207.
- [21] G. Hilson, The extractive industries and development in sub-Saharan Africa: An introduction, Resources Policy 40 (1) (2014) 1-3, ISSN 03014207.
- [22] M. Hirons, Shifting sand, shifting livelihoods? Reflections on a coastal gold rush in Ghana, Resources Policy 40 (1) (2014) 83-89, ISSN 03014207.
- [23] J. T. Kelly, "This mine has become our farmland": Critical perspectives on the coevolution of artisanal mining and conflict in the Democratic Republic of the Congo, Resources Policy 40 (1) (2014) 100-108, ISSN 03014207.
- [24] B. Milanez, J. A. Puppim de Oliveira, Innovation for sustainable development in artisanal mining: Advances in a cluster of opal mining in Brazil, Resources Policy 38 (4) (2013) 427- 434, ISSN 03014207.
- [25] F. K. Nyame, J. Blocher, Influence of land tenure practices on artisanal mining activity in Ghana, Resources Policy 35 (1) (2010) 47-53, ISSN 03014207.
- [26] B. Ross, S. Dessureault, M. Rieber, The Tucson Mineral Show and the market for collector minerals: The potential for artisanal and small scale miners, Resources Policy 36 (2) (2011) 168-177, ISSN 03014207.
- [27] B. A. Teschner, Small-scale mining in Ghana: The government and the galamsey, Resources Policy 37 (3) (2012) 308-314, ISSN 03014207.
- [28] L. Ebert, T. La Menza, Chile, copper and resource revenue: A holistic approach to assessing commodity dependence, Resources Policy 43 (2015) 101-111, ISSN 03014207.
- [29] T. Ejdemo, P. S[°]oderholm, Mining investment and regional development: A scenario-based assessment for Northern Sweden, Resources Policy 36 (1) (2011) 14-21, ISSN 03014207.
- [30] J. Fessehaie, What determines the breadth and depth of Zambia's backward linkages to copper mining? The role of public policy and value chain dynamics, Resources Policy 37 (4) (2012) 443-451, ISSN 03014207.
- [31] F. M. Haslam McKenzie, A. Hoath, The socio-economic impact of mine industry commuting labour force on source communities, Resources Policy 42 (2014) 45-52, ISSN 03014207.
- [32] G. Ivanova, The mining industry in Queensland, Australia: Some regional development issues, Resources Policy 39 (1) (2014) 101-114, ISSN 03014207.
- [33] S. Kinnear, I. Ogden, Planning the innovation agenda for sustainable development in resource regions: A central Queensland case study, Resources Policy 39 (1) (2014) 42-53, ISSN 03014207.
- [34] G. Lagos, E. Blanco, Mining and development in the region of Antofagasta, Resources Policy 35 (4) (2010) 265-275, ISSN 03014207.
- [35] J. A. Puppim de Oliveira, S. H. Ali, Gemstone mining as a development cluster: A study of Brazil's emerald mines, Resources Policy 36 (2) (2011) 132-141, ISSN 03014207.
- [36] P. S"oderholm, N. Svahn, Mining, regional development and benefit-sharing in developed countries, Resources Policy 45 (2015) 78-91, ISSN 03014207.
- [37] M. Morris, R. Kaplinsky, D. Kaplan, "One thing leads to another" Commodities, linkages and industrial development, Resources Policy 37 (4) (2012) 408-416, ISSN 03014207.
- [38] R. Hanlin, C. Hanlin, The view from below: 'lock-in' and local procurement in the African gold mining sector, Resources Policy 37 (4) (2012) 468-474, ISSN 03014207.
- [39] A.-M. Fleury, B. Davies, Sustainable supply chains-minerals and sustainable development, going beyond the mine, Resources Policy 37 (2) (2012) 175-178, ISSN 03014207.



- [40] J. Elkington, Partnerships from cannibals with forks: The triple bottom line of 21st-century business, 1998.
- [41] N. Bocken, S. Short, P. Rana, S. Evans, A literature and practice review to develop sustainable business model archetypes, Journal of Cleaner Production 65 (2014) 42-56, ISSN 09596526.
- [42] C. R. Carter, D. S. Rogers, A framework of sustainable supply chain management: moving toward new theory, International Journal of Physical Distribution & Logistics Management 38 (5) (2008) 360-387, ISSN 0960-0035.
- [43] E. Hassini, C. Surti, C. Searcy, A literature review and a case study of sustainable supply chains with a focus on metrics, International Journal of Production Economics 140 (1) (2012) 69-82, ISSN 09255273.
- [44] M. Pagell, Z. Wu, Building a more complete theory of sustainable supply chain management using case studies of 10 exemplars, Journal of Supply Chain Management 45 (2) (2009) 37-56, ISSN 15232409.
- [45] L. Blanco, R. Grier, Natural resource dependence and the accumulation of physical and human capital in Latin America, Resources Policy 37 (3) (2012) 281-295, ISSN 03014207.
- [46] Y. Lei, N. Cui, D. Pan, Economic and social effects analysis of mineral development in China and policy implications, Resources Policy 38 (4) (2013) 448-457, ISSN 03014207.
- [47] T. Brown, F. McEvoy, J. Ward, Aggregates in England Economic contribution and environmental cost of indigenous supply, Resources Policy 36 (4) (2011) 295-303, ISSN 03014207.
- [48] B. Kotey, J. Rolfe, Demographic and economic impact of mining on remote communities in Australia, Resources Policy 42 (2014) 65-72, ISSN 03014207.
- [49] S. A. Hajkowicz, S. Heyenga, K. Moffat, The relationship between mining and socio-economic well being in Australia's regions, Resources Policy 36 (1) (2011) 30-38, ISSN 03014207.
- [50] P. Howie, Z. Atakhanova, Resource boom and inequality: Kazakhstan as a case study, Resources Policy 39 (1) (2014) 71-79, ISSN 03014207.
- [51] C. Mainguy, Natural resources and development: The gold sector in Mali, Resources Policy 36 (2) (2011) 123-131, ISSN 03014207.
- [52] J. Southalan, What are the implications of human rights for minerals taxation?, Resources Policy 36 (3) (2011) 214-226, ISSN 03014207.
- [53] J. T. Mensah, E. Botchway, Ghana's salt industry: A neglected sector for economic development?, Resources Policy 38 (3) (2013) 288-294, ISSN 03014207.
- [54] D. A. Fleming, T. G. Measham, Local job multipliers of mining, Resources Policy 41 (1) (2014) 9-15, ISSN 03014207.
- [55] A. O. Adewuyi, T. Ademola Oyejide, Determinants of backward linkages of oil and gas industry in the Nigerian economy, Resources Policy 37 (4) (2012) 452-460, ISSN 03014207.
- [56] R. Bloch, G. Owusu, Linkages in Ghana's gold mining industry: Challenging the enclave thesis, Resources Policy 37 (4) (2012) 434-442, ISSN 03014207.
- [57] B. Bocoum, W. C. Labys, Modelling the economic impacts of further mineral processing, Resources Policy 19 (4) (1993) 247-263, ISSN 03014207.
- [58] L. Corkin, Chinese construction companies in Angola: A local linkages perspective, Resources Policy 37 (4) (2012) 475-483, ISSN 03014207.
- [59] M. Farooki, The diversification of the global mining equipment industry Going new places?, Resources Policy 37 (4) (2012) 417-424, ISSN 03014207.
- [60] D. Kaplan, South African mining equipment and specialist services: Technological capacity, export performance and policy, Resources Policy 37 (4) (2012) 425-433, ISSN 03014207.
- [61] M. Morris, R. Kaplinsky, D. Kaplan, Commodities and Linkages: Industrialisation in SubSaharan Africa, ISBN 9781770112513, URL http://oro.open.ac.uk/30048/, 2011.



- [62] Z. Teka, Linkages to manufacturing in the resource sector: The case of the Angolan oil and gas industry, Resources Policy 37 (4) (2012) 461-467, ISSN 03014207.
- [63] K. Yuxiang, Z. Chen, Resource abundance and financial development: Evidence from China, Resources Policy 36 (1) (2011) 72-79, ISSN 03014207.
- [64] J. Rolfe, Predicting the economic and demographic impacts of long distance commuting in the resources sector: A Surat basin case study, Resources Policy 38 (4) (2013) 723-732, ISSN 03014207.
- [65] A. D. Elbra, The forgotten resource curse: South Africa's poor experience with mineral extraction, Resources Policy 38 (4) (2013) 549-557, ISSN 03014207.
- [66] J. Ge, Y. Lei, Mining development, income growth and poverty alleviation: A multiplier decomposition technique applied to China, Resources Policy 38 (3) (2013) 278-287, ISSN 03014207.
- [67] A. Calderon, J. D. Harris, P. A. Kirsch, Health interventions used by major resource companies operating in Colombia, Resources Policy ISSN 03014207.
- [68] X. S. Warnaars, Why be poor when we can be rich? Constructing responsible mining in El Pangui, Ecuador, Resources Policy 37 (2) (2012) 223-232, ISSN 03014207.
- [69] J. Ebner, The Sino-European race for Africa's minerals: When two quarrel a third rejoices, Resources Policy 43 (2015) 112-120, ISSN 03014207.
- [70] A. Adeniyi, Resource governance and the challenges of community development in the Nigerian bitumen belt, Resources Policy 40 (1) (2014) 42-47, ISSN 03014207.
- [71] J. Kidido, J. Ayitey, E. Kuusaana, E. Gavu, Who is the rightful recipient of mining compensation for land use deprivation in Ghana?, Resources Policy 43 (2015) 19-27, ISSN 03014207.
- [72] B. Y. Imbun, F. Duarte, P. Smith, "You are not our only child": Neoliberalism, food security issues and CSR discourse in the Kutubu oilfields of Papua New Guinea, Resources Policy 43 (2015) 40-49, ISSN 03014207.
- [73] O. C. Anejionu, P.-A. Ahiarammunnah, C. J. Nri-ezedi, Hydrocarbon pollution in the Niger Delta: Geographies of impacts and appraisal of lapses in extant legal framework, Resources Policy 45 (2015) 65-77, ISSN 03014207.
- [74] J. A. Shandro, M. M. Veiga, J. Shoveller, M. Scoble, M. Koehoorn, Perspectives on community health issues and the mining boom-bust cycle, Resources Policy 36 (2) (2011) 178-186, ISSN 03014207.
- [75] D. Holterman, Slow violence, extraction and human rights defence in Tanzania: Notes from the field, Resources Policy 40 (1) (2014) 59-65, ISSN 03014207.
- [76] B. Campbell, Corporate Social Responsibility and development in Africa: Redefining the roles and responsibilities of public and private actors in the mining sector, Resources Policy 37 (2) (2012) 138-143, ISSN 03014207.
- [77] P. Heneberg, Burrowing bird's decline driven by EIA over-use, Resources Policy 38 (4) (2013) 542-548, ISSN 03014207.
- [78] W. S. Breffle, D. Muralidharan, R. P. Donovan, F. Liu, A. Mukherjee, Y. Jin, Socioeconomic evaluation of the impact of natural resource stressors on human-use services in the Great Lakes environment: A Lake Michigan case study, Resources Policy 38 (2) (2013) 152-161, ISSN 03014207.
- [79] C. Moran, D. Franks, L. Sonter, Using the multiple capitals framework to connect indicators of regional cumulative impacts of mining and pastoralism in the Murray Darling Basin, Australia, Resources Policy 38 (4) (2013) 733-744, ISSN 03014207.
- [80] M. Porter, D. M. Franks, J.-A. Everingham, Cultivating collaboration: Lessons from initiatives to understand and manage cumulative impacts in Australian resource regions, Resources Policy 38 (4) (2013) 657-669, ISSN 03014207.
- [81] Y. Huang, D. Todd, L. Zhang, Capitalizing on energy supply: Western China's opportunity for development, Resources Policy 36 (3) (2011) 227-237, ISSN 03014207.



- [82] A. Marvasti, The role of price expectations and legal uncertainties in ocean mineral, exploration activities, Resources Policy 38 (1) (2013) 68-74, ISSN 03014207.
- [83] T. R. M. D. Camargo, P. R. D. C. Merschmann, E. V. Arroyo, A. Szklo, Major challenges for developing unconventional gas in Brazil - Will water resources impede the development of the Country's industry?, Resources Policy 41 (1) (2014) 60-71, ISSN 03014207.
- [84] E. H"aggquist, P. S"oderholm, The economic value of geological information: Synthesis and directions for future research, Resources Policy 43 (2015) 91-100, ISSN 03014207.
- [85] J. Joaqu´ın Jara, P. P´erez, P. Villalobos, Good deposits are not enough: Mining labor productivity analysis in the copper industry in Chile and Peru 1992-2009, Resources Policy 35 (4) (2010) 247-256, ISSN 03014207.
- [86] K.-S. Huh, Steel consumption and economic growth in Korea: Long-term and short-term evidence, Resources Policy 36 (2) (2011) 107-113, ISSN 03014207.
- [87] M. A. Haque, E. Topal, E. Lilford, A numerical study for a mining project using real options valuation under commodity price uncertainty, Resources Policy 39 (1) (2014) 115-123, ISSN 03014207.
- [88] I. G´omez-M´arquez, L. R. Alejano, F. Garc´ıa Bastante, Mining compatibility with other projects in Spain: Solutions and benefits, Resources Policy 36 (1) (2011) 22-29, ISSN 03014207.
- [89] K. S"oderholm, P. S"oderholm, H. Helenius, M. Pettersson, R. Viklund, V. Masloboev, T. Mingaleva, V. Petrov, Environmental regulation and competitiveness in the mining industry: Permitting processes with special focus on Finland, Sweden and Russia, Resources Policy 43 (2015) 130-142, ISSN 03014207.
- [90] J. Ayelazuno, Oil wealth and the well-being of the subaltern classes in Sub-Saharan Africa: A critical analysis of the resource curse in Ghana, Resources Policy 40 (1) (2014) 66-73, ISSN 03014207.
- [91] WTO, China Measures Related to the Exportation of Rare Earths, Tungsten and Molybdenum, URL https://www.wto.org/english/tratop_e/dispu_e/cases_e/ds431_e.htm, 2015.
- [92] J. D. Wilson, Chinese resource security policies and the restructuring of the Asia-Pacific iron ore market, Resources Policy 37 (3) (2012) 331-339, ISSN 03014207.
- [93] K. Slack, Mission impossible?: Adopting a CSR-based business model for extractive industries in developing countries, Resources Policy 37 (2) (2012) 179-184, ISSN 03014207.
- [94] P. Aroca, M. Atienza, Economic implications of long distance commuting in the Chilean mining industry, Resources Policy 36 (3) (2011) 196-203, ISSN 03014207.
- [95] G. Banks, Little by little, inch by inch: Project expansion assessments in the Papua New Guinea mining industry, Resources Policy 38 (4) (2013) 688-695, ISSN 03014207.
- [96] B. Y. Imbun, Maintaining land use agreements in Papua New Guinea Mining: 'Business as usual'?, Resources Policy 38 (3) (2013) 310-319, ISSN 03014207.
- [97] M. Marinescu, A. Kriz, G. Tiess, The necessity to elaborate minerals policies exemplified by Romania, Resources Policy 38 (4) (2013) 416-426, ISSN 03014207.
- [98] F. A. S. Postali, M. Nishijima, Oil windfalls in Brazil and their long-run social impacts, Resources Policy 38 (1) (2013) 94-101, ISSN 03014207.
- [99] S. M. Smith, P. T. Dorward, Nationalised large-scale mining, trade unions and community representation: Perspectives from Northern Madagascar, Resources Policy 40 (1) (2014) 31- 41, ISSN 03014207.
- [100] A. Standing, Ghana's extractive industries and community benefit sharing: The case for cash transfers, Resources Policy 40 (1) (2014) 74-82, ISSN 03014207.
- [101] E. Tang, F. Liu, J. Zhang, J. Yu, A model to analyze the environmental policy of resource reallocation and pollution control based on firms' heterogeneity, Resources Policy 39 (1) (2014) 88-91, ISSN 03014207.
- [102] M. Barber, S. Jackson, Indigenous engagement in Australian mine water management: The alignment of corporate strategies with national water reform objectives, Resources Policy 37 (1) (2012) 48-58, ISSN 03014207.



- [103] R. G. Boutilier, L. Black, Legitimizing industry and multi-sectoral regulation of cumulative impacts: A comparison of mining and energy development in Athabasca, Canada and the Hunter Valley, Australia, Resources Policy 38 (4) (2013) 696-703, ISSN 03014207.
- [104] G. Campbell, M. Roberts, Permitting a new mine: Insights from the community debate, Resources Policy 35 (3) (2010) 210-217, ISSN 03014207.
- [105] P.-Y. Le Meur, L. S. Horowitz, T. Mennesson, "Horizontal" and "vertical" diffusion: The cumulative influence of Impact and Benefit Agreements (IBAs) on mining policy-production in New Caledonia, Resources Policy 38 (4) (2013) 648-656, ISSN 03014207.
- [106] K. Moffat, A. Zhang, The paths to social licence to operate: An integrative model explaining community acceptance of mining, Resources Policy 39 (1) (2014) 61-70, ISSN 03014207.
- [107] S. Nysten-Haarala, E. Klyuchnikova, H. Helenius, Law and self-regulation Substitutes or complements in gaining social acceptance?, Resources Policy 45 (2015) 52-64, ISSN 03014207.
- [108] J. R. Owen, D. Kemp, 'Free prior and informed consent', social complexity and the mining industry: Establishing a knowledge base, Resources Policy 41 (1) (2014) 91-100, ISSN 03014207.
- [109] J. R. Owen, D. Kemp, Social licence and mining: A critical perspective, Resources Policy 38 (1) (2013) 29-35, ISSN 03014207.
- [110] R. Parsons, J. Lacey, K. Moffat, Maintaining legitimacy of a contested practice: How the minerals industry understands its 'social licence to operate', Resources Policy 41 (1) (2014) 83-90, ISSN 03014207.
- [111] J. Prno, An analysis of factors leading to the establishment of a social licence to operate in the mining industry, Resources Policy 38 (4) (2013) 577-590, ISSN 03014207.
- [112] J. Prno, D. Scott Slocombe, Exploring the origins of 'social license to operate' in the mining sector: Perspectives from governance and sustainability theories, Resources Policy 37 (3) (2012) 346-357, ISSN 03014207.
- [113] C. Richert, A. Rogers, M. Burton, Measuring the extent of a Social License to Operate: The influence of marine biodiversity offsets in the oil and gas sector in Western Australia, Resources Policy 43 (2015) 121-129, ISSN 03014207.
- [114] H. Tiainen, R. Sairinen, V. Novikov, Mining in the Chatkal Valley in Kyrgyzstan Challenge of social sustainability, Resources Policy 39 (1) (2014) 80-87, ISSN 03014207.
- [115] A. Zhang, K. Moffat, A balancing act: The role of benefits, impacts and confidence in governance in predicting acceptance of mining in Australia, Resources Policy 44 (2015) 25-34, ISSN 03014207.
- [116] J. R. Craynon, E. A. Sarver, D. P. Robertson, Could a public ecology approach help resolve the mountaintop mining controversy?, Resources Policy 38 (1) (2013) 44-49, ISSN 03014207.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



EVALUATION OF ELECTRICAL ENERGY CONSUMPTION BY A HOT-DIP GALVANISING PLANT

M. Dewa¹, B. Dzwairo² and B. Nleya³

¹Department of Industrial Engineering Durban University of Technology, South Africa <u>mendond@dut.ac.za</u>

²Research and Postgraduate Support Durban University of Technology, South Africa Ig445578@gmail.com

³Department of Electronic Engineering Durban University of Technology, South Africa <u>bakhen@dut.ac.za</u>

ABSTRACT

The galvanising industry is generally characterised by significant consumption of energy. Over 75% of South African plants in the galvanising sector use electric furnaces, an approach was originally prompted by the previous low price of electricity. The biggest challenge now is that from year 2008, electricity prices have risen considerably. It is against this background that an energy assessment was conducted on a batch hot-dip galvanising plant in order to identify potential opportunities for the reduction and more efficient use of electrical energy. The methodology which was adopted commenced with an evaluation of the organisation in terms of its awareness to energy management and commitment to improving on energy efficiency. The methodology also entailed compiling a detailed electrical energy balance and identifying the most significant energy users of electricity. An assessment of the results revealed that the galvanising kettle was the most significant energy users after which recommendations were made for reducing the energy consumption by the galvaniser.

¹ Corresponding Author



1. INTRODUCTION

The operational environment of manufacturing plants is characterised by increasing pressure to reduce their carbon footprint, motivated by trepidations associated with high energy costs and climate change, with energy as one of the most vital resources for manufacturing. Increasing the Energy Efficiency (EE) of production processes and management approaches have a greater potential of reducing energy costs and avoiding high carbon footprint [1] [2]. It has been generally acknowledged that hot dip galvanising (HDG) is a cost-effective and adequate protection system for components and fabrications required to operate with minimum or zero maintenance for extended periods of time in any environment [3]. The subject of energy optimisation is vital in any industry, especially in hot dip galvanising processes which are used to fabricate a wide variety of steel products such as large pre-fabricated items, structural steel, small washers, threaded parts and so forth. About half of the zinc produced world-wide is used to protect steel from corrosion by a through galvanising of steel and iron [4]. The dearth of energy sustainability is continuously posing major challenges to the South African government and globally. Galvanising furnaces are the predominant consumers of energy in the galvanising industry. The key factors that contribute to large energy consumption by the galvanising furnace is heat losses through convection, conduction, and radiation through pot components as well as the galvanised products. This paper evaluates the electrical energy consumption by a hot-dip galvanising (HDG) plant, against the backdrop of rising energy costs. This evaluation assisted the galvaniser to quantify its significant energy users and identify potential opportunities for the reduction and more efficient use of energy within the plant.

2. LITERATURE REVIEW

2.1 Background for Galvanising

Rahrig (2002) cited batch hot-dip galvanising and inline-continuous galvanising as two most common methods of applying zinc metal to steel. Batch hot-dip galvanising is characterised by immersing the parts or jobs as a discrete "batch" into the zinc bath [5]. Pertaining continuous galvanising, molten zinc is applied on a continuous ribbon of steel sheet as it passes through a bath of molten zinc at high speeds and the process may operate for days without interruption. The duration for the steel in the zinc bath is about two to four seconds and in spite of the favourably faster galvanization rate, continuous galvanising is limited to very thin, flexible sheets of steel [6].

Galvanising plants may also be further sub-categorised according to the type of fuel used to heat the surface preparation and galvanising tanks. Gas-fired furnaces can be categorised in terms of whether they heat the zinc directly or indirectly. Furnaces that heat the zinc directly use immersion burners while those that heat the zinc indirectly have a combustion gallery between the furnace and its exterior to facilitate the transfer of heat from the combustion gases to the zinc. Flat-flame, forced-circulation and high-velocity furnaces are the major types of indirect gas-fired furnaces prevailing within the galvanising industry [7].

Hot dip galvanising is a complex metallurgical process in which steel material is rapidly immersed into a zinc alloy bath whose temperature is normally between 450° C and 480° C [8] [9]. The hot dip galvanising process is characterised by massive heat losses and with regard to heat transfer. Cook (2005) presented basis data for calculations for heat lost at zinc-air interface for "open" and "enclosed" furnace [10]. Thereafter, Blake and Beck (2004) modelled the effect of combined radiation and convection on hot dip galvanising furnace wear [11] and Depree *et al.* (2010) used a three dimensional model to investigate temperature distribution due to radiative heat transfer in three dimensions [12].

Electrical heating is preferred for a galvanising furnace since the radiant heat produced by resistance elements is very uniform, though the high cost of electricity usually precludes the use of electrical heating [7]. There are three steps to the galvanising process which include preparation, galvanising, and post-treatment, with the preparation step being divided into degreasing, pickling, and fluxing [6].

2.2 Surface Preparation

The initial process tank in the galvanising plant is the degreasing tank which usually contains a caustic soda solution used to get rid of organic contaminants such as oils and dirt from the surface of steel. The tanks is generally heated to a temperature of about 80° C and can be agitated to hasten the cleaning process [7]. These surface contaminants need to be eliminated prior to pickling so that the surface can be "wetted" by the pickling solution.

The second surface preparation step is pickling where the degreased steel is immersed into a tank containing acid solution. This tanks either contain sulphuric or hydrochloric acid solution which is used to remove any mill scale or other oxides that may have developed on the surface of the steel [13]. Hydrochloric tanks may be used



at ambient temperatures without heating since the cleaning action of the hydrochloric solution is sufficient at room temperature. Conversely, in order to increase the cleaning action, sulphuric acid must be heated to a temperature of about 60° C.

The third surface preparation step is fluxing which involves the application of a fluxing chemical coating, usually zinc ammonium chloride, onto the surface of the steel part [6]. The fluxing chemical would chemically remove the last

vestiges of oxides prior to steel immersion into the molten zinc, and allows the steel to be wetted by the molten zinc. There are two types of fluxing which include "dry" and "wet" fluxing. Dry fluxing is characterised by immersing the

steel part into an aqueous solution of the zinc ammonium chloride flux, of which upon removal, the flux solution is dried prior to immersion into the zinc bath. The amount of flux deposited on the components is a function of the flux concentration, and the cleaning performance is a function of the drying time and temperature conditions [3]. On the other hand, in wet fluxing, a blanket of liquid zinc ammonium chloride is floated on top of the molten zinc bath so that the steel part to be zinc coated first passes through the molten flux as it is being immersed into the zinc bath. Zinc ammonium chloride is less dense and has a lower melting point than molten zinc and thus floats on the bath surface. Wet fluxing is preferred for workloads that consists of small fasteners, mixes of shapes and sizes, and when centrifuging is executed [3].

2.3 Galvanising

Hot dip galvanising is the application of the zinc coating whereby cleaned steel is immersed in molten zinc usually at a temperature of between 445° C and 450° C [14]. A metallurgical (chemical) reaction results when perfectly cleaned steel is immersed into molten zinc, forming a thick coating comprising of a series of zinc or zinc iron alloy layers. Generally, the duration for the steel to reach bath temperature and to react with the zinc is usually less than ten minutes [6]. The adhesion of the resultant coating to carbon steel is therefore determined by means of metallurgical laws and forms a chemical bond to the substrate. Chemical bonding is considered to be far superior to that of a mechanical bond. The galvanised steel product is withdrawn slowly from the galvanising bath once the coating growth is complete, and the left-over zinc is removed by draining, centrifuging or vibrating.

One key by-product of hot dip galvanising is dross which is formed inside or on top of the molten zinc as floating (galvanising ashes) or bottom dross. The intermetallic compounds of the floating and bottom dross develop when the concentration of elements of iron oxides and zinc ammonium chloride exceed their solubility limits in the zinc furnace. The floating dross is characterised by a mixture of oxides and chlorides. Despite perfect control of the zinc bath and the deliberate addition of elements such as aluminium to reduce the iron dissolution, the temperature gradient between the immersed objects and the zinc bath necessitates the crystallisation of dross [15]. The temperature at which the zinc bath is maintained has a direct effect on the galvanising furnace life since at galvanising temperatures, the partially soluble iron from the furnace wall dissolve in the liquid zinc and form dross [11]. The equation for the reaction can be indicated as:

$$Fe + 13 Zn \rightarrow Fe Zn_{13}$$
(1)

The composition of the dross layer is therefore about 4% Fe and 96% Zn.

2.4 Post Treatment and Inspection

Post galvanising treatment can be characterised by air cooling or quenching into water. Generally, quenching is the last step in most HDG processes used to promote passivation of the zinc surface and to control the growth of the zinc-iron alloy layers, with chromium-free passivating technology being preferred to toxic hexavalent chromium passivation [16]. The inspection of hot-dip galvanized steel is less complex, given that zinc does not adhere to contaminated steel, and thus, a visual inspection of the galvanized steel provides adequate evaluation of the quality of the coating [17].

A variety of simple physical tests can be performed to determine thickness, uniformity, adherence, and appearance. The duration to first maintenance of HDG products is directly proportional to the thickness of the zinc coating, given any operational environment. This is a function of the coating thickness which is a critical parameter in the specification and effectiveness of hot-dip galvanising as a corrosion protection system. Thus, inspection of the coating thickness is a key feature of hot-dip galvanized products since thicker coatings yields enhanced durability and decades of maintenance-free performance [18]. Salt-fog tests are generally used for a relative comparison with respect to service condition in order to relate the extent to which various coatings provide protection against corrosion [5]. These tests produce quick results and are less costly although the correlation between the duration in salt spray test and the expected life of some HDG coatings is generally weak [19].



2.5 Benchmarking and Energy Consumption

Energy consumption benchmarking is an energy consumption indicator as a function of the raw materials, process and product output [20]. It can be used for comparison of the energy performance of a number of plants within the same industry, to compare the dynamics in energy performance of one plant at different time periods, with the view to improve energy performance [21]. Energy consumption benchmarking can also be useful for estimating the energy-efficiency improvement potentials within a particular sector whereby the results are compared against best practice technology currently in operation [22]. According to Phuong (2010), energy consumption varies with different raw steel materials and each item has its own product and process parameters which determine the extent of energy consumption [23].

One of the key performance metrics in energy consumption benchmarking is specific energy consumption (SEC) which is an indicator of the amount of primary energy consumed by a process to produce one physical unit of product [24]. Blake and Beck (2004) presented a set of equations that describe the energy efficiency of a galvanising furnace in order to emphasise the need for energy optimisation. These equations could be used for comparing furnaces of different designs

and fuel types in a completely objective fashion [4]. Cook (2005) elaborated that future furnaces will have to operate using less energy and have higher production capabilities to be competitive [10].

3. STUDY AREA

The case-in-point galvaniser provides hot dip galvanising solutions to a wide range of products using a functional jobbing production mode. The plant has a maximum capacity of 40 tonnes per day and the bath size for degreasing, pickling, fluxing and galvanising tanks is $14m \times 1.3m \times 2.5m$, with cranes having a lifting weight capacity of 4 tonnes. Figure 1 shows the flow diagram for the hot dip galvanising process at the company.



Figure 1: Flow diagram for hot dip galvanising process

The final surface preparation step in the galvanising process is fluxing, where a zinc ammonium chloride solution is used to remove any remaining oxides and deposits a protective layer on the steel to prevent any further oxides from forming on the surface prior to immersion in the molten zinc. The dimensions of the tank are 14 m x 1.3 m x 2.5 m and this 30 000-litre tank has no insulation on the side walls and no covers at the top. The galvanising kettle is powered by 24 rows of chromium heating elements sectioned into 4 zones and furnace temperature is generally set at 600° C.

4. METHODOLOGY

The methodology which was adopted commenced with an evaluation of the organisation in terms of its awareness to energy management and commitment to improving on energy efficiency.

The questionairre for assessment of company's energy management had the following questions:

- Does the top management know that significant energy cost savings can be achieved by simple low cost measures without necessitating financial assessment?
- Is the top management committed to energy cost reduction and is there an approved energy policy in place?
- Have roles, responsibility and authority been identified for all persons having an influence on significant energy use and is this documented?
- Have the significant energy uses been quantified and documented?
- Has a baseline of energy performance been established against which progress can be measured?
- Have indicator(s) or metrics been identified to use in measuring progress against your baseline?
- Have the organisation's energy objectives and targets been identified and documented?



- Have energy action plans been established?
- Is the energy management system evaluated at least once a year and are improvements made based on the results of the evaluation.

The methodology also entailed compiling a detailed electrical energy balance and identifying the most significant energy users of electricity. A heat loss model was then developed for evaluating energy saving initiatives for the significant energy users which include the galvanising and degreasing tanks after which recommendations were developed for reducing the energy consumption of the galvaniser.

5. ASSESSMENT OF ELECTRICAL ENERGY CONSUMPTION

5.1 Assessment of Galvaniser's Energy Management

An initial evaluation of the company's energy performance was conducted. Figure 2 shows the results for assessment of the organization in terms on its awareness to energy management and commitment to improving its energy efficiency. Although there was no energy policy in place, top management was committed to energy cost reduction. However management was also generally unaware that significant energy cost savings could be achieved by simple low cost measures without huge financial investment.

There was no documentation of the roles, responsibility and authority for all persons who could have an influence on significant energy use (SEUs) and the significant energy uses have not been quantified and documented. No baselines of energy performance indicators (EnPIs) had been established against which progress can be measured and there were no metrics for measuring progress against baselines. The organisation's energy objectives and targets were neither identified nor documented and there were no energy action plans.



Figure 2: Assessment Chart for awareness to energy management and commitment

5.2 Assessment of Consumption Data

An average of 33 tonnes of metal was galvanised per day from an average of 45 dips per day. Total monthly tonnage sat at around 62 7921 tonnes of steel material using about 49 757 tonnes of zinc. Average monthly electricity consumption sat at 270779 kWh and charged at an average rate of R1.49/kWh but higher figures were noted for June (R3.00/kWh), July (R2.95/kWh) and August (R2.95/kWh). The average rate of R1.49/kWh was computed as a mean of the energy standard, energy peak, and energy off-peak as derived from the municipality monthly bills.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 3: Electricity consumption profile

Active energy data for standard usage time, peak usage time and off peak time is shown in Figure 3. The galvaniser utilised a high percentages of both standard time and off-peak times and lower amounts of peak time. It was noted that plant loading during peak periods was lower than for off-peak periods, sitting at around 50 000 kWh per month. A key

recommendation would be to further schedule as much production as is feasible during off-peak times or standard times as energy usage for peak times are billed at a rate of almost double that of off-peak time usage. Peak, off-peak and standard energy usage were fairly constant over the year, albeit that a few peaks could be noted as production and factory usage fluctuated in some months. As shown in Figure 4, it was also noted that the notified maximum demand is far greater than the average monthly network demand consumption by the company.



Figure 4: Monthly network demand versus notified demand

Figure 5 shows percentage contributions of the appliances for the consumption of electrical energy. It was revealed that the galvanising tank consumes the bulk of the energy, followed by the flux and degreasing tanks.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 5: Pie chart for distribution of electrical energy consumption

Figure 6 shows the average current drawn by each of the heating zones and the spot temperature of the galvanising fluid surface. It is worth mentioning that only the side walls of the tank can be heated, not the bottom. Heat was lost from the tanks primarily through convection and evaporation from the surface of the liquid. The spot temperature at molten zinc surface was slightly less than the galvanising temperature since dross formation at the top act as a partially insulating blanket because of its lower thermal conductivity. Since the tank was not covered at the top, heat was lost over an area of 18.2 m^2 i.e. from 14 m x 1.3 m. This is equivalent to a 14 m horizontal bare pipe with a diameter of 41.3 cm. It was recommended to cover the tank during non-production hours to reduce surface losses. The rate of loss was dependent upon the differential temperature between the zinc fluid and the surface exposed to air movement.



Figure 6: Current drawn by heating zones and the spot temperature at zinc surface

5.3 Evaluation of energy saving initiatives

5.3.1 Galvanising tank

An online calculator from CheCalc was used to compute the heat loss from the galvanising tank and estimate the saving that would be derived from insulating it [25].



Process Temperature	173°C
Ambient Temperature	28ºC
Wind Speed	0.0 km/hr
Fuel Cost	R1.48
Efficiency	90%
Hours Per Year	4224
Nominal Pipe Size	400 mm diameter
Bare Metal	Steel
Bare Surface Emittance	0.8
Outer Jacket Material	All Service Jacket
Outer Surface Emittance	0.9

Table 1: System parameters used to minimise heat loss on galvanising tank

Using the previously stated calculator, the results from Table 2 show that covering with a 40 mm thick insulating material will prevent an estimated heat loss of the difference between 12640 kWh/m/yr and 998 kWh/m/yr thus saving R19143.64/m/yr. Given that the tank is 14 m long, and is idle about 50% of the time, 81494 kWh/yr i.e. from (11642 x 14 x 0.5) will be saved annual savings will be about R134 000. About 80.67 tonnes of CO_2 emissions will be avoided, i.e. from (81494 x 0.99/1000).

Table 2: Summary	v of the savings	from variable	insulation	thickness
Tuble L. Jullinu	y or the surnings	nom variable	moulation	cific cos

Variable Insulation Thickness (mm)	Cost (Rand/m/yr)	Heat Loss (kWh/m/yr)	Savings (Rand/m/yr)
Bare	20784.35	12640	
15.0	4274.99	2600	16509.36
25.0	2350.32	1429	18434.03
40.0	1640.71	998	19143.64

5.3.2 Flux Tank

The dimensions of the tank are 14 m x 1.3 m x 2.5 m and this 30 000-litre tank has no insulation on the side walls and no covers at the top. Since the tank is not covered at the top, and has no insulation on the side walls, heat is lost over an area of $94.7m^2$ i.e. from (2 x 14 x 2.5 + 2 x 1.3 x 2.5 + 14 x 1.3). This is equivalent to a 14 m horizontal bare pipe with a diameter of 200 cm. Figure 7 shows the spot temperature for the side wall of the flux tank.



Figure 7: Spot temperature for the side wall of the flux tank

Using the CheCalc calculator, Table 3 shows the system parameters that were used to minimise heat loss on flux tank. Covering with a 40 mm thick insulating material will prevent an estimated heat loss of (2925 - 269) kWh/m/yr to give 2656 kWh/m/yr.



Process Temperature	49.8°C
Ambient Temperature	30°C
Wind Speed	0.0 km/hr
Electricity Cost	R1.48
Efficiency	80%
Hours Per Year	4224
Nominal Pipe Size	2000 mm diameter
Bare Metal	Steel
Bare Surface Emittance	0.8
Outer Jacket Material	All Service Jacket
Outer Surface Emittance	0.9

Table 3: System parameters used to minimise heat loss on flux tank

Given that the tank is 14 m long, and that part of the top will be open, of which its total surface area contributes about 25% of the total surface area, 27888 kWh/yr i.e. from (2656 x 14 x 0.75) will be saved. Annual savings will be about R68 787 and about 27.6 tonnes of CO_2 emissions will be avoided, i.e. from (27888 x 0.99/1000). Table 4 shows a summary of the savings from insulation layer which is 40 mm thick.

Table 4: Summary of savings from insulation of 40mm thickness

Variable Insulation Thickness	Cost (Rand/m/yr)	Heat Los (kWh/m/yr)	s Savings (Rand/m/yr)
Bare	5410.93	2925	
Layer 1 (40.0)	497.57	269	4913.36

5.3.3 Degreasing Tank

Figure 8 shows the spot temperature for the side wall of the degreasing tank. The dimensions of the tank are 14 m x 1.3 m x 2.5 m and this is equivalent to a 14 m horizontal bare pipe with a diameter of 200 cm.



Figure 8: Spot temperature for the side wall of the degreasing tank

Using the previously stated calculator, covering with a 40 mm thick insulating material will prevent an estimated heat loss of (1698-166) kWh/m/yr thus saving R1750.87/m/yr. Given that the tank is 14 m long, and that part of the top will be open, of which its total surface area contributes about 25% of the total surface area, 16086 kWh/yr i.e. from (1532 x 14 x 0.75) will be saved. Annual savings will be about R39 694 and about 15.92 tonnes of CO₂ emissions will be avoided, i.e. from (16086 x 0.99/1000).

5.3.4 Geyser

The 200-litre geyser with its current manufacturer's insulation has a standing energy input of 4 kWh over 24 hrs irrespective of use because the hot water is maintained at the set point temperature. In order to calculate the total amount of energy dissipated by the geyser the heat losses in W/m^2 were multiplied by the surface area of the geyser,



using the conductivity as 0.055 W/m.K and a surface heat temperature coefficient of 6.3 W/m^2 .K. The ambient temperature was considered to be 25°C and the cost of electricity at R1.48 per kWh. This loss could be reduced by adding an external insulating blanket over the geyser. The analysis revealed that the insulating blanket would reduce the heat loss from the geyser by 3.2 kWh/day, resulting in a saving of 292 kWh/yr for the geyser. This cascades to saving

R432 per year and avoiding 0.289 tonnes of CO_2 emissions. i.e. from (292 x 0.99/1000). The additional insulation would cost R2 000, and thus simple payback period would be 4.6 years.

Alternately, a better solution will be retrofitting the geyser with an instant water heater. This heater has an inline water heater utilizing electrically conductive polymer structures for electrodes and by varying the area of electrodes that confront one another would yield variable temperatures to which the water is heated. The heat is by the resistance of the water to the electrical current flowing between the electrodes [26]. Their advantages include better energy efficiency, minimised space utilisation and longer life expectancy. The efficiency of tankless hot water heaters range from 85 to 97% as opposed to conventional tank-style hot water heater systems which are characterised by efficiencies of less than 70%.

After observation for a period of month, the time taken to heat the water from using the 4 kW element was 1 hour. A day heating profile therefore consisted of eight 1 hour intervals. One can consider retrofitting the present 4 kW geyser with a 7kW instant water heater with a flow rates ranging from 1 to 6 litres per minutes and save $((4 \text{ kWh x 8h}) - (7 \text{ kWh x 3 h}), \text{ i.e. 11 kWh per day or 2640 kWh per year if we consider 20 working days per month. The resultant saving would be equivalent to R3 900. A 7kW instant water heater costs about R3 000 and installation costs about R2 000 to give R5 000, and simple payback period would be 1.2 years.$

5.3.5 Summary for estimated energy and cost savings

Table 5 shows a summary for estimated energy and cost savings from energy minimisation options. It reveals that covering galvanising kettle would yield huge energy savings followed by insulation of the flux tank. Further insulation of the geyser does not realise much energy saving. A total of 125 760 kWh of electricity at an estimated cost R242 913 would be saved by implementing the stated four energy minimisation opportunities.

Energy Minimisation Opportunities	Estimated Savings (kWh/Annum)	Estimated Savings (Rand/ Annum)	Investment (Rands)	Estimated CO ₂ emissions avoided (Tons/yr)	Simple Payback Period
Covering galvanising kettle	81 494	134 000	100 000	80.67	0.7
Insulation of flux tank	27 888	68 787	80 000	27.6	1.2
Insulation of degreasing tank	16 086	39 694	80 000	15.92	2.0
Insulation of geyser	292	432	2 000	0.289	4.6
Retrofitting with instant water heater	2640	3900	5000	2.61	1.3
Total	125 760	242913	267000	124	

Table 5: Summary for estimated energy and cost savings

5.3.6 Scheduling Production Dips

Scheduling influences the energy consumption behaviour of the whole system, and by integrating energy efficiency criteria into scheduling, a reduction of energy costs may be realised. It was noted that about 50% of the dips utilised about half the full length of the galvanising tank. Figure 9 shows a galvanised product coming out of a dip, of which the product uses less than half the size of the galvanising tank.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 9: Galvanised product coming out of a dip

It was advised to schedule dips for the long products in batches separate from shorter products to reduce to labour cost of covering and uncovering the tanks between dips. The energy saving to be derived are aligned to the covering of the galvanising, flux and degreasing tanks. The aspect of developing scheduling a scheduling algorithm with the view to save energy is a potential area for future research.

6. CONCLUSION

The aim of this paper was to evaluate the electrical energy consumption by a hot-dip galvanising plant and the evaluation assisted the galvaniser to quantify its significant energy users. The estimated savings which arise from the energy savings initiatives lie at around 125 760 kWh per annum and the payback periods range from 0.7 to 4.6 years. The results from the paper also assisted the galvaniser to identify potential opportunities for reduction and more efficient use of energy within the plant. It is recommended that management should run an appreciation course on energy efficiency for all the employee since the survey conducted revealed that there is no energy policy in the company. It is also recommended to cover the galvanising tank in between dips, insulate the fluxing and degreasing tanks as well as the geyser. Although lights were not consuming a substantial amount of electricity, retrofitting the administration block would also reduce electricity consumption at the company by a small magnitude. Future work in this area will embrace development of algorithms for optimal schedules to save energy in these non-continuous galvanising plants. It was also noted that it may be difficult to assess the reliability of an online calculator, and thus it is recommended to develop energy models using tools such as Microsoft Excel or MATLAB where the researcher can have full control of the modeling parameters.

REFERENCES

- [1] Weinert, N., Chiotellis, S. and Seliger, G. 2011. Methodology for planning and operating energy-efficient production systems. CIRP Annals *Manufacturing Technology*, 60 (1), pp 41-44.
- [2] Shrouf, F., Ordieres-Meré, J., García-Sánchez, A. and Ortega-Mier, M. 2014. Optimizing the production scheduling of a single machine to minimize total energy consumption costs. *Journal of Cleaner Production*, 67 (0), pp 197-207.
- [3] Hornsby, M. J. 1995. Hot-dip galvanising: a guide to process selection and galvanising practice. London: Intermediate Technology.
- [4] Blake, S. G. and Beck, S. B. M. 2004. Energy consumption and capacity utilization of galvanising furnaces. . Proceedings of the Institution of Mechanical Engineers. *Part E Journal of Process Mechanical Engineering*, 218 (4), pp 251-259.
- [5] Rahrig, P. G. 2002. Batch hot-dip and inline galvanising: A tale of two processes. The Tube & Pipe Journal. Available: http://www.thefabricator.com/article/tubepipefabrication/batch-hot-dip-and-inlinegalvanising (Accessed 03 July 2015).
- [6] Behrens, K. 2012. Taking Action Against Hot-Dip Galvanising Pollution. Blue Ridge Environmental Defense League.
- [7] Krzywicki, J. and Langill, T. 2003. Heat Sources & Furnaces. American Galvanizers Association, Vol. 6, pp 1.
- [8] Marder, A. R. 2000. The metallurgy of zinc-coated steel. *Progress in materials science*, 45 (3), pp 191-271.
- [9] Ilinca, F., Ajersch, F., Baril, C. and Goodwin, F. E. 2007. International Journal of Numerical Methods in Fluids. 53, pp 1629 -1646.
- [10] Cook, T. H. 2005. Burning issues impact kettle purchases: An insulated kettle cover can aid in heat recovery. Metal Finishing, 103 (5), pp 52-54.
- [11] Blake, S. G. and Beck, S. B. M. 2004. The effect of combined radiation and convection on hot dip galvanising kettle wear. *Applied Thermal Engineering*, 24, pp 1301-1319.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

- [12] Depree, N., Sneyd, J., Taylor, S., Taylor, M. P., Chen, J. J. J., Wang, S. and O'Connor, M. 2010. Development and validation of models for annealing furnace control from heat transfer fundamentals. *Computers and Chemical Engineering*, 34, pp 1849-1853.
- [13] Prasad, M., Prasad, S. and Patel, A. 2015. Thermal Analysis of the Molten Lead Kettle Failure at the Galvanising Plant and Development of Novel Design Using CFD Techniques. International Journal of Innovative Research in Science, Engineering and Technology, 4 (3), pp 1351-1360.
- [14] Wilmot, R. E. 2007. Corrosion protection of reinforcement for concrete structures. *The Journal of The Southern African Institute of Mining and Metallurgy*, 107, pp 139-146.
- [15] Vourlias, G., Pistofidis, N., Stergioudis, G. and Polychroniadis, E. K. 2005. A negative effect of the insoluble particles of dross on the quality of the galvanized coatings. *Solid State Sciences*, 7 (4), pp 465-474.
- [16] Kong, G. and White, R. 2010. Toward cleaner production of hot dip galvanising industry in China. *Journal of Cleaner Production*, 18 (10), pp 1092-1099.
- [17] GAA. 2012. The Basics of Hot Dip Galvanized Steel First and Last Line of Defence. *Galvanizers Association of Australia*, pp 6-25.
- [18] AGA. 2011. Inspection of Hot-Dip Galvanized Steel Products. American Galvanizers Association. Available from: http://www.galvanizeit.org/education-and-resources/publications/inspection-of-hot-dip-galvanized-steel-products-2011.
- [19] ISSF. 2008. The salt spray test and its use in ranking stainless steels. International Stainles Steel Forum 14-15. Available from: <u>http://www.worldstainless.org/Files/issf/non-image-files/PDF/ISSF_The_salt_spray_test_and_its_use_in_ranking_stainless_steels.pdf</u>
- [20] Worrell, E. and Price, L. 2006. An integrated benchmarking and energy savings tool for the iron and steel industry. *International Journal of Green Energy*, 3 (2), pp 117-126.
- [21] Ke, J., Price, L., McNeil, M., Khanna, N. Z. and Zhou, N. 2013. Analysis and Practices of Energy Benchmarking for Industry from the Perspective of Systems Engineering. *Energy*, 54, pp 32-44.
- [22] Saygin, D., Worrell, E., Patel, M. K. and Gielen, D. J. 2011. Benchmarking the energy use of energyintensive industries in industrialized and in developing countries. *Energy*, 36 (11), pp 6661-6673.
- [23] Phuong, Q. M. 2010. Pot heat balance analysis in continuous galvanising lines. *Master of Science West Virginia University*.
- [24] Laurijssen, J., Faaij, A. and Worrell, E. 2013. Benchmarking energy use in the paper industry: a benchmarking study on process unit level. *Energy Efficiency*, 6 (1), pp 49-63.
- [25] CheCalc. 2015. Insulation Heat Loss Calculation. Online, Available from: <u>http://checalc.com/calc/inshoriz.html</u>. Accessed on 12 February 2015.
- [26] Novotny, D. and Chaput, I. 2003. Instant water heater. U.S. Patent 6,640,048.


A PROPOSED MODEL FOR SOUTH AFRICA TO EFFECTIVELY RECYCLE AND DISPOSE WASTE ELECTRICAL AND ELECTRONIC EQUIPMENT

A.J.J. Mouton^{1*} & J.H. Wichers²

¹Department of Electrical Engineering Tshwane University of Technology, South Africa moutonajj@tut.ac.za

²School of Mechanical and Nuclear Engineering North West University, South Africa harry.wichers@nwu.ac.za

ABSTRACT

In South Africa there are daily occurrences of dumping of Waste Electrical and Electronic Equipment (WEEE), also called e-waste, at municipal refuse sites. WEEE is hazardous and comes from households and businesses and these unhealthy and unsafe practices can lead to permanent damage of the soil and water resources of areas. It can cause humans and animals to acquire different kinds of cancers and illnesses which could ultimately lead to deaths. WEEE contains valuable materials that can be recycled and there is thus loss of potential extraction of these valuable materials. South Africa has a limited WEEE infrastructure to collect, refurbish, recycle, and disposal capabilities to handle e-waste and need a management model to address WEEE effectively.

¹ The author was enrolled for a PhD (Development and Management in Engineering) degree at the Centre for Research and Continued Engineering Development, North West University

² The author was the study leader for author 1 and from the Department of Mechanical and Nuclear Engineering, North West University



1. INTRODUCTION

The retirement stage of any product, namely decommissioning and disposal, is of vital importance due to the increased awareness of environmental issues all around the world [1]. WEEE contains valuable materials and there is thus an opportunity for extraction of valuable materials. South African legislation regulating WEEE is vague and not specific causing confusion, hazardous practices, and minimal disposal and recycling. A country's infrastructure, geography, legislation, social challenges, individual knowledge, etc. are drivers that necessitate a "custom made" model to dispose of e-waste effectively. There is a need for a model that suits the unique circumstances of the South African environment and a model to address the effective recycling and disposal issues will be proposed in this paper.

According to [2], South Africa is a signatory to the Basel Convention, an international treaty designed to reduce the movements of hazardous waste between nations. It is stated in [3] that South Africa is not a signatory to the Bamako convention which implies that South Africa can import e-waste legally. In [4] it is stated that in South Africa"...only a fraction of the discarded EEE (estimated 10%) finds its way to recyclers." Authors [5] estimate that South Africa generates around 100,000 tons of e-waste annually of which only 20% is recycled by formal recyclers. Figures from [5] also indicate that that the South African average e-waste recycling is around 0.39 kg per capita per year, far below European countries. It is estimated by [6] that South African households have between 1,129,000 and 2,108,000 tons of potential e-waste which includes white goods, consumer electronics and IT. It is also estimated that around 70% of South Africa's e-waste is in storage [7].

2. IMPORTANCE OF MANAGING E-WASTE EFFECTIVELY

In a study conducted by [8] a mathematical calculation concluded that the best methods for treating e-waste are *reuse* and *recycling*. A summary by [4] indicate that e-waste is an emerging problem as well as a business opportunity due to the value of materials. E-waste components are diverse and some are classified as "hazardous" because they go beyond the threshold quantities mentioned in the Material Safety Data Sheet (MSDS) [9]. The environmental and health effects such as asthmatic bronchitis and other allergic reactions, DNA damage, abdominal cramps, vomiting, diarrhea, nausea, decreased red and white blood cell production, abnormal heart rhythm, difficulties in breathing, increased or decreased blood pressure, numbness around the face, muscle weakness, and paralysis are highlighted by [10]. Some of the substances are known human carcinogens causing cancer and possibly death. Chlorofluorocarbons (CFC's) used in refrigeration equipment also causes stratospheric ozone depletion and global warming [11].

3. METHODOLOGY

For <u>this</u> paper the following methodologies were implemented: The researcher did a literature review of ewaste practices in different countries. This included collection initiatives, recycling practices, legislation, extended producer responsibility, financing of e-waste, etc. The scenarios in South Africa were also investigated through a literature review. The researcher visited European recycling facilities and also visited South African entities such as municipal refuse sites, e-waste recyclers, government officials, environmentalists, etc. Observations were made at the different sites and informal discussions contributed to the development of a proposed e-waste management model as is discussed further in this paper. This proposed model is a first draft and will be adjusted where needed after scientific research data has been collected. The methodologies that will be implemented to collect the future data are: <u>Quantitative data</u> will be collected from South African <u>consumers</u> by using a survey questionnaire. <u>Qualitative data</u> will be collected from <u>producers</u>, <u>retailers</u>, <u>recyclers</u>, <u>municipal refuse</u> managers, <u>government</u> officials and <u>environmentalists</u>. Semi structured interviews will be conducted with knowledgeable representatives at these entities. The data will be analysed and a <u>final</u> e-waste management model will be proposed together with proposed legislation for South Africa.

4. GENERIC E-WASTE MANAGEMENT FRAMEWORK

This paper will continue the discussion under the following headings:

- Mechanisms to co-ordinate and drive the management processes
- Legislation
- Collection arrangements of e-waste
- Financing models for recycling e-waste

The framework will be expanded and discussed to highlight the important e-waste management points and to pave the way for a South African framework.



4.1 Mechanisms to co-ordinate and drive the management processes

Basic drivers-The conclusion from the literature is that different European countries have different levels of success with e-waste collection rates ranging from just above 1 kg/capita/year to 3.75 kg/capita (Switzerland) collected per year [12]. Switzerland is a world leader in effective e-waste but [13] states that although a system appears to be successful in some country the success of one system does not necessarily indicate success in another country due to different cultures, attitudes, social problems and behaviors, economic drivers, etc.

EXTENDED PRODUCER RESPONSIBILITY-Extended Producer Responsibility (EPR) is a strategy widely used around the world. EPR as a policy strategy was proposed by Thomas Lindhqvist and introduced in 1990 [14]. EPR is defined as an "environmental protection strategy to reach an environmental objective of a decreased total environmental impact from a product, by making the manufacturer of the product responsible for the entire life-cycle of the product and especially for the take-back, recycling and final disposal of the product. EPR is implemented through administrative, economic and informative instruments. The composition of these instruments determines the precise form of the Extended Producer Responsibility." [14]. According to [15] policies on EPR had been introduced in European countries to achieve waste minimization.

EPR in South Africa-The ultimate responsibility of e-waste management must lie with the producers as it is the view of the researcher that the SA government should have the role as overseer and enforcer of legal action and that the SA government do not have the capacity, capabilities, or the knowledge to ensure the effective operation of a system of this magnitude. The Waste Act, [16], entered into force on 1 July 2009 and required the Minister of environmental affairs to draft a National Waste Management Strategy. The Framework for the National Waste Management Strategy specifically mentioned e-waste, and stated that it "should be prioritized for further investigation and implementation for extended producer responsibility". In [2] it is stated "The responsibility of processing of e-waste in South Africa is not under the Extended Producer Responsibility (EPR) as what is now a practice in Europe." The *goal* of EPR in South Africa should be on waste prevention (pollution reducing and preventing).

<u>STAKEHOLDER RESPONSIBILITY</u>-According to [17] the Japanese law on WEEE clearly stipulates the roles of each participant in the e-waste recycling chain. The Environmental Protection Administration of Taiwan (EPAT) created the 4-in-1 recycling program which consisted of community residents, private collectors and recyclers, municipal collection teams, and a recycling fund [18]. In [13] it is shown that in South Korea the EPR law came in effect in 2003 and local manufacturers, distributors and importers of consumer goods such as air conditioners, TVs and PCs are required to achieve official recycling targets or face financial consequences. Stakeholders are thus identified in WEEE management programs.

<u>Stakeholder responsibility in South Africa-A</u> statement by [6] declares "a practical e-waste management solution which has the buy-in of all stakeholders" is needed.

<u>SYSTEMS COVERAGE AND ENSURING COMPLIANCE-In</u> [19] it is indicated that countries such as Belgium, the Netherlands, Sweden, Norway, and Switzerland had some form of legislation on e-waste even before the implementation of the WEEE Directive and made use of a Single National Compliance system covering all WEEE types. In a report [19] it is shown that at the time there were 76 compliance schemes in EU countries. In South Africa there is no systems coverage in place where e-waste is concerned. Compliance of effective recycling can thus not be ensured.

<u>**REGISTRY-</u>**Producers are defined on European, national and local levels due to the fact that sometimes manufacturers are or are not in the EU and producers could be wholesalers, distributors, retailers etc. depending on their status and position in the retailing chain [19]. In [2] it is suggested that "A single Registry would need to be set up and funded by producers". At present no registry exists in South Africa.</u>

4.2 Legislation

In a research survey conducted by [20] it was concluded that industrialists and academics perceived legislation to have the biggest industrial influence when designing environmentally conscious products. In a focus group study [21] point out that from the literature, legal and economic drivers have been identified as the strongest drivers.

<u>Some Asian countries</u>-It is pointed out in [4] that the Japanese Specified Home Appliances Recycling Act 1998 was published in 1998 and came into effect in April 2001. The act covers only TVs, refrigerators, washing machines and air conditioners and recycling is the responsibility of producers. A report [22] warns that India has tried to implement e-waste legislation without success. The Chinese WEEE legislation measures taken were summarized by [23] as well as some successes after implementation. In [24] mention is made that Indonesia has no regulation for managing the e-waste generated at homes. The Act on the Promotion of Saving and Recycling of Resources (APSRR) which regulates four types of home appliances in South Korea is mentioned in [25]. Personal computers were added to the program in 2003 followed by mobile phones and audio equipment (2005), and printers, copying machines and fax machines (2006). APSRR was replaced in 2008 with a new system called the Eco-Assurance System or EcoAS [25]. According to [18], Taiwan has EPAT that regulates and



ensures effective recycling of e-waste. According to Article 18 of the Waste Disposal Act, WEEE collection, storage and recycling must follow EPAT's environmental and safety standards. These standards were issued in 2002 and revised in 2007 [18].

Europe-Legally, e-waste management was introduced in 1998 by the Swiss Federal Office for the Environment (FOEN), by way of the Ordinance on "The Return, the Taking Back and the Disposal of Electrical and Electronic Equipment (ORDEE)" [26]. Directives were drafted in Europe which required an environmental approach to all designs. The first one was the RoHS Directive 2002/95/EC. The second important directive is called the WEEE Directive 2002/96/EC and this directive (European Union, 2003b) requires producers in the EU member states to take back their products and dispose them using environmentally sound methods. A new Directive (2012/19/EU) requires the member states to achieve the collection rate of 45% by weight of weight of EEE put on the market in the three preceding years by 2016 [25].

<u>America-</u>Legislative activity in the US has rapidly increased [27]. According to [25], although no national law exists in North America, states, provinces and cities have implemented various measures to address the WEEE problem. According to [28] 25 states in the USA have some form of recycling program.

Developing countries-The 1989 Basel Convention on the Control of Transboundary Movements of Hazardous Wastes and their Disposal (into force on 5 May 1992) is an international initiative to prevent the dumping of hazardous materials in developing countries, [29]. America is the only industrialized country in the world that is not a signatory to this convention [4]. To address shortcomings in the Basel Convention, African countries created the Bamako Convention for complete banning of hazardous substances such as e-waste. Referring to the Bamako convention [3] states that "these instruments share the common goal of controlling the movement of hazardous wastes across national borders". The key issues of effective management of e-waste in developing countries are a change in attitude by governments, WEEE legislation, control of WEEE dumping, implementation of EPR, and the transfer of technology on sound recycling of e-waste [30]. A comment is made by [28], "There is currently no country in Latin America with a comprehensive e-waste management system".

Legislation in South Africa-In 2007 (Updated in 2009) a report was compiled and released by Dittke, an attorney from Cape Town, to review the e-waste legislation in South Africa. The report stated that "Unlike many other countries, South Africa currently does not have any dedicated legislation dealing with e-waste. As such a whole range of environmental, as well as health and safety, laws must be examined to provide answers. Such investigation will have to cover national, provincial and local legislation. Needless to say, this is an arduous and unsatisfactory situation, and certainly does not help to clarify matters", [31]. In [16] it is stated that mandatory Industry Waste Management Plans (IndWMPs) must be prepared for the lighting industry for mercury containing lamps e.g. compact fluorescent lamps (CFL's). It also states that "Over the course of the next five years, IndWMP's will be required for different forms of e-waste and batteries, and other waste streams that are best managed through an IndWMP." Towards the end of 2013 a news release on the eWaste association of South Africa (eWasa) website stated: "As you may be aware e-waste has not been deemed a priority in terms of the waste act, therefore it was not mandatory for the sector to submit Industry Waste Management plans."

4.3 Collection arrangements of e-waste

<u>Composition of e-waste-</u>The Directive 2002/96/EC of the European Parliament and of the Council (January 2003) on Waste Electrical and Electronic Equipment (EU, 2002a) defined ten categories of WEEE.

<u>Composition of South African e-waste-</u>A claim is made by [2] that the e-waste recycling systems are not uniform and sustainable solutions for materials such as CRT tubes, brominated plastics and printed circuit boards, have not yet been found.

<u>Collection, number of collection points and efficiency of collection-</u>In table 1 some countries with known collection points and recycling rate numbers per year are analyzed. For Norway and The Netherlands the first recycling rate per year could only be obtained for 2010 and it is assumed that the collection points did not change significantly. When analyzing the figures for the 4 countries (Ireland excluded) a coefficient of correlation of 0.75 was calculated if a comparison is made between the collection points and the total weight of WEEE being collected per year, indicating a very strong relationship.

Country	Collection Points (All)	Recycle rate per year (Kg/p/y)	Population in millions	Persons/ collection point	Total weight in Kg per year
Norway (2003)	400	13.41	4.5	11250	60345
The Netherlands (2003)	600	7.5	16	26667	120000

Table 1: Collection	points and	recycling	data per	country
---------------------	------------	-----------	----------	---------



Spain (2007)	600	3.3	44.5	74167	146850
Ireland (2007)	740	6.7	4.313	5828	28897
Sweden(2006)	950	15.8	9.113	9593	143985

In calculating the coefficient of correlation of the recycling rate per person per year compared to the persons per collection point (PCP) in a country the figure is -0.70, indicating that if more collection points are available in a geographic area, the collection rate would increase. By including the four countries (Ireland excluded) in Table 1 a coefficient of correlation of 0.72 is obtained if the collection point numbers and the population of a country are analyzed. Although collection point numbers will not be only the determining factor of collection rate success, the number of collection points should play an important factor in collection efficiency.

Figure 1 shows the number of inhabitants per collection point on the Y axis and the collection in kg per inhabitant on the X axis. From the figure and the curve it can be derived that the less persons there are per collection point, meaning more collection points per persons, the better the collection rate will be. Spain has the least collection points per inhabitants and this is reflected in the collection return rate. From the key figures report released by [32] the data shows that all of the countries mentioned in Table 1 have increased the collection return rate except for Spain that shows a lower number of 2.91 kg/person collected in 2012 as compared to the rate of 3.3 kg/person being collected in 2007 [32]. The fewer collection points compared to the other mentioned countries could be a contributing factor.



Figure 1: Collection rate versus persons per collection point

<u>Collection and number of collection points in South Africa-</u>South Africa has approximately 51 million people [33]. Mention is made in [2] of some collection points but no exact figure is available. The same document lists a number of E-Waste consumer programs and drop-of points available in South Africa whereby the consumer can dispose of E-Waste responsibly. The researcher has found that some of the mentioned collection points are not collecting WEEE anymore (Pickitup) and that some do not know of collection initiatives (Builders Warehouse). Stores such as Pick and Pay and Woolworths only collect batteries, CFL's and printer cartridges.

4.4 Financing models for recycling e-waste

Mention is made by [19] that even with the EU directives which indicate that producers must finance the costs of waste management there are different interpretations and promulgated legislation on the financing of new WEEE after 13 August 2005. Five financing mechanisms are described by [19] but the authors also state "the examples provided here are not an exhaustive list of possible models used for EPR programs". In Japan the recycling fee and fund is managed by the Association for Electric Home Appliances (AEHA), [34]. It is the opinion of [17] that the majority of Japanese consumers disposes e-waste properly but is concerned that the recycling fee at disposal "may stimulate illegal dumping". The South African environment does not have any financial model to assist with e-waste recycling and the focus of recyclers is mainly on the WEEE which is profitable when recycled.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

5. WEEE DESIGN AND TREATMENT SYSTEMS

Since the realization of the hazardous nature of WEEE new measures were introduced in countries, especially in Asian and European countries to reduce the negative effect. Mentions were made of ROHS and design for environment (DfE) as measures to lessen the effects on the environment. In [18] it is explained that in Taiwan the recycling technologies for WEEE were developed using techniques from developed countries. As in many countries, WEEE recyclers in Taiwan focus on dismantling. It is the opinion of [17] that WEEE containing hazardous substances must be managed by accredited recycling plants in Japan. The results of previous research is used by [35] to indicate that vast quantities of e-waste are now being moved around the world for recycling in developing countries using manual processes in backyards of residential properties. South Africa is mentioned as one of the counties.

Where WEEE design and treatment systems for South Africa are concerned problematic practices associated with the environmental protection in South Africa are highlighted by [6]. Examples are one out of ten refurbishers and recyclers were ISO 14001 compliant, ozone-depleting practices, lack of disposal knowledge, etc. The same authors also highlight a key concern in the South African market as the potential threat of backyard electrochemical processes being set up.

6. PROPOSED COLLECTION ARRANGEMENT FOR SOUTH AFRICA

6.1 Suggested collection point numbers for South Africa

Statistical numbers suggest that the more the collection points, the better the success rate will be. The statistical website, [36], was accessed to determine the ten biggest city population figures in South Africa. From figure 1, using the formula PCP = 49286e^{-0.117R} where R indicate the collection return rate in kg/person (assume 4kg/person), it can be calculated that a collection point must be set up for every 30865 persons. For Johannesburg (4434827), Cape Town (3740026), Tshwane (2921488) and EThekwini (Durban) (3442361) the collection points are calculated as 144, 121, 95, and 112 respectively. In the Tshwane metropolitan area the landfill sites and garden refuse transfer sites are 18 [37]. Potential retailers selling EEE are: 13 Pick and Pay stores [38], 9 Builders Warehouse stores [39], 23 Checkers and Checkers Hyper stores [40], 12 Shoprite stores [41], 3 Hi-Fi Corporation stores [42], 4 Deon Wired stores [43], 8 Game stores [44], and 21 Super Spar stores [45]. The minimum stores that can be used as collection points for the Tshwane Metropolitan area are thus 111 and the suggested 95 collection points are possible and achievable.

6.2 Collection point placement

It is proposed that collection initiatives are done at:

- EEE Retailers- Collect similar WEEE of EEE type being sold in the store.
- Public collection points-Municipal refuse sites and garden refuse sites.
- Free Call to collect-Set up in cities and bigger towns to collect large WEEE.
- Containers at apartments-To collect WEEE from tenants.
- Public collection boxes-To collect small WEEE from shops selling EEE.
- Special drop-off events- Bi-annually in rural areas and small towns.
- Corporate collection points (Not for large WEEE)-Hundred or more employees.

6.3 South African collection infrastructure

It is proposed that it must be in three groups, namely: 1-CRT's, 2-all lamps, and 3-any other WEEE. Collection sites must have impermeable surfaces and must have weatherproof covering.

6.4 Voluntary or mandatory program for South Africa

A mandatory system is proposed. Producers have had many years of experience in other countries but have done little to start an effective e-waste management program.

6.5 Take-back requirements in the South African environment

All retailers of EEE should accept any similar WEEE irrespective if a customer buys a new product or not. All retailers selling EEE must accept WEEE with dimensions smaller than 30 cm³ without any obligation. Large enough shopping malls (> 50 000 m²) with more than 10% of retailers selling EEE must set up a manned WEEE collection point.

7. ECONOMIC INSTRUMENTS TO FUND AN EPR PROGRAM IN SOUTH AFRICA

The "polluter pays principle", should apply and the South African consumer of EEE should pay for the recycling of WEEE. Currently no disposal fee is levied on any South African EEE item. It is proposed that an advance disposal fees (ADF) is levied on the sale of new EEE products to cover costs that arise from the implementing and administering of an EPR program. It is also proposed that an ADF fees should be levied on imported second hand items. The ADF fee structure must be an uncomplicated system to prevent confusion.



8. THE PRODUCER RESPONSIBILITY ORGANISATIONS (PRO) IN SOUTH AFRICA

The researcher proposes three different PRO's. The *first* is for <u>household appliances</u> and electrical tools, excluding cooling appliances. The *second* is for <u>information technology</u>, telecommunications equipment and consumer equipment. The *third* one is for <u>problematic equipment</u> such as CRT, LED, LCD, cooling equipment, batteries, and different types of lighting equipment. From [46] the "Product structure" is proposed for PRO1, PRO2, and PRO3.

8.1 Detail setup of proposed PRO 1

PRO 1 for *household appliances and electrical tools*, excluding cooling appliances.



Figure 2: Proposed PRO 1 model

In figure 2 the proposed setup of PRO 1 should mainly focus to extract metals, aluminium, plastics, copper, glass, etc. as this equipment contains a minimum of electronic control parts/boards.

8.2 Detail setup of proposed PRO 2

It is proposed that PRO 2 should be for *information technology*, *telecommunications equipment and consumer equipment*.

In figure 3 the proposed setup of PRO 2 should focus on information technology (PC's, notebooks, printers, etc.), telecommunication (cell phones, telephones, answering machines, etc.), and consumer equipment (Video recorders, smoke detectors, electric toys, video games, etc.). The suggested grouping is due to the high content PC boards in the equipment.



8.3 Detail setup of proposed PRO 3

PRO 3 should be for *problematic equipment*. Problematic equipment are televisions and monitors such as CRT, LED, LCD, plasma etc., all types of cooling equipment using refrigerants as the cooling medium, all types of batteries, and all different types of lighting equipment used by consumers.





Figure 4: Proposed PRO 3 model

In figure 4 each cluster will consist of producers of the same type of EEE being produced or imported. The products in this case have different applications, but generally they have the same properties. The producer clusters have the best knowledge on recycling of their products and communication from producers to recyclers and retailers (on recycling, handling and storage) is of extreme importance and is indicated by the broken line.

8.4 PRO's and other stakeholders

In the entire proposed PRO models the following stakeholders need to be part of the group: Producers, retailers, recyclers, public representatives such as political party representatives, environmentalists, and government representatives.

9. THE PROPOSED MANAGEMENT BODY

The proposed name of the entity that will manage the WEEE process in South Africa is the South African Waste Electrical and Electronic Equipment Association (SAWEEEA). From [46] the "Functional structure" is proposed for the management body.



Figure 5: Proposed SAWEEEA management model

9.1 Composition and tasks of SAWEEEA

Figure 5 shows the proposed structure for SAWEEEA which consist of a board of members which are representing and selected by the three PRO's. To prevent corruptive actions no employee of SAWEEEA must have any links to stakeholders which can cause undue influence in the operation of managing the WEEE in South Africa. SAWEEEA should be registered as a non-profit Section 21 company or similar.

9.2 Function of the five groupings under the SAWEEEA CEO

9.2.1 Logistics

This department should be involved in contracts for transporting of WEEE, monitoring of the operation of the auction houses (to be described in section 12), collection of WEEE, ensuring safe practices are applied at collection points, compile statistics (weights collected and of fractions), SAWEEEA infrastructure maintenance and setup, setting up of non-existing recycling facilities, setting up of an administrative system at collection points, etc.



9.2.2 Information Technology

This department should be involved in the design and setting up of a website for SAWEEEA in order to have retailers to report ADF amounts received, to have producers reporting ADF amounts and sales, maintain and upgrade the website, link to auction houses, implementing links on municipal websites, implementing links on government websites, and maintaining all SAWEEEA information technology networks.

9.2.3 Communication

This department should be involved in informing the public of the hazardous nature of WEEE, making the public aware of collection points and initiatives, implementing educational material on WEEE in school programs, provide feedback to stakeholders on the EPR program, communicate the applicable ADF fees to stakeholders, etc.

9.2.4 Finances

The financial department of the SAWEEEA team must be involved in determining the ADF fee in co-operation with all three PRO's, receiving and comparing the ADF fee from producers and retailers, compensate recyclers of negative value items, compensate transport contractors, finance the setting up of collection facilities, finance the setting up of non-existing recycling facilities, payment of salaries of SAWEEEA employees, payment of operational expenses, payment for auditing of entities, etc.

9.2.5 Treatment/Accreditation

This department should issue and retract licenses to recycling facilities, assist in setting up of recycling facilities, audit recycled materials to prevent dumping, implement and upkeep the registry, ensure imported goods are RoHs compliant, etc.

9.3 EEE, ADF and ADF information flow



Figure 6: EEE, ADF and ADF information flow in proposed model

Figure 6 depicts the suggested flow of new EEE, the ADF fee, and ADF information flow. New EEE items are sold by producers to retailers. Producers can supply more than one retailer as indicated. The ADF on an EEE item is paid at the point of sale and received by the retailer. The ADF fee collected by the retailer must be paid to the producer/supplier of the item. Payment to the producer/supplier will thus become part of the transaction payment where the ADF fee will be additionally highlighted on the settling documentation. Producers should then transfer the ADF fees to the financial department of SAWEEEA. Retailers should record the ADF amounts received by consumers for specific items, and indicate from which producer the products were. Retailers should also provide this detail to SAWEEEA on a monthly basis as part of information flow. There will thus be a correlation between producers and retailers about ADF amounts received for recycling as well as detail information on the number of items and weight placed on the market. The customs department must also report figures to SAWEEEA on a regular basis. Information that must be supplied include number and dates of items that entered the customs points, the ADF fee received from producers/recyclers for second hand imported goods, and the payment of the ADF fee to SAWEEEA. In figure 6 Producer A represents an importer of second-hand EEE and an importer/manufacturer of new EEE. With all three entities reporting independently to SAWEEEA, fraud and free riding will be minimized due to the audit trail availability.

9.4 ADF and information flow for imported WEEE

South Africa is allowed to import e-waste. It is proposed that a SAWEEEA official must inspect the WEEE upon arrival. If there is uncertainty about the status (WEEE or second hand EEE) the SAWEEEA official must determine the ADF amount for EEE and payment must be done to SAWEEEA before the release of the containers at the ports or border posts. If the shipment is WEEE, a SAWEEEA official (same or another) should inspect the shipment at the recycler premises to ensure that the shipment is



WEEE. The difference between the ADF fee received and the cost to inspect the imported shipment must then be refunded to the recycler.

9.5 Financial (ADF) flow diagram

In figure 6 it has been shown that the ADF that was received from the customers, by the retailers, must be paid to producers which must again pay the ADF to SAWEEEA. In figure 7 below, the further proposed flow of the ADF fees is shown.

In Figure 7 the SAWEEEA head office is a single entity linking to an Auction house. In practice, more than one Auction house will be linked to SAWEEEA. ADF fees collected from the respective producers will be applied as shown in figure 7



Figure 7: Financial (ADF) flow diagram

9.6 ADF and information flow for second hand EEE goods

Importers of second hand goods will have to pay exactly the same ADF fee as producers of new EEE. As in the case of imported WEEE, a SAWEEEA official should also inspect the shipment and determine the ADF before the items can be released. The costs involved with this inspection must be paid by the importer on top of the ADF amount.

9.7 Setup costs for the proposes model

With the introduction of laws that demand an ADF fee from consumers for recycling it should not be difficult to fund the expenses. The ADF fee will need to be introduced from a specified date after which the setup of Auction houses, collection points, SAWEEEA main office, etc. will commence. The physical collection, transportation, auctioning and recycling will thus follow at a later stage. With the introduction of the ADF fee, private or government loans could be secured to fund the setting up of the entire system.

10. PERFORMANCE STANDARDS - SETTING TARGETS IN SOUTH AFRICA FOR RE-USE, REFURBISH AND RECYCLING OF WEEE

If possible the life of consumable items must be extended through re-use. Refilling a printer cartridge is an example of re-usable consumable items. The refurbishment of equipment is a lucrative and live sector and there is no need to set standards in this sector. For recycling, no targets will be proposed as this is a consultation process between all stakeholders.

11. THE ROLE OF THE SOUTH AFRICAN NATIONAL GOVERNMENT

Government should establish a consultative framework and start by leading the process. Government must coopt SAWEEEA to co-ordinate and run the day to day operation of the management system and set up a <u>single</u> legal policy framework which guide and support the implementation of an e-waste strategy. Government must set reasonable targets of WEEE collections and government must act as the enforcer of laws if not complied with.



11.1 Additional government/local government measures

The disposal of any WEEE onto landfills and into municipal refuse bins or bags, rivers, dams, any open area, or the sea must be banned in South Africa. The role of local government must be limited to the reception and storage of WEEE at municipal and garden refuse sites and to information communicated to the public.

12. CONSUMERS AND INCENTIVE SCHEMES

South Africa is a relatively poor country and it would be wise to incorporate incentive schemes to motivate consumers to hand in WEEE. Incentive schemes could possibly be the receiving of talk time or data bundles on cellular phones or data devices, points on reward card systems such as FNB's eBucks points, Pick a Pay's smart shopper cards points, Woolworths's WRewards points, ABSA bank's Rewards points, Standard bank's UCount Rewards etc.

13. PROPOSED AUCTION HOUSE



Figure 8: Proposed Auction house

In figure 8 the researcher is proposing an "Auction house" for WEEE recycling. An auction environment has benefits and fit into the proposed model. Klemperer (1999:227), state that "...auctions are such simple and well-defined economic environments". The auction houses should be placed in the bigger cities/towns in South Africa close to recyclers. Johannesburg and surrounding areas, Cape Town, and Durban/Ballito/Pietermaritzburg area are proposed. Due to the fairly large distances between cities in South Africa the solution would be to have collections done at remote areas ones or twice per year using larger trucks to ensure single trips and economy of scale.

From [46] the "Process structure" is proposed for the auction house. Figure 8 shows two different flows namely that of information (the solid line) and material flow (the dotted line). The *administrative section* should be the center point of information. Collections, call-to-collect, and transport arrangements are done from this section and the number/weights of received WEEE and the income generated must be reported to SAWEEEA. Auction details must be published through the administration department. The *receiving, grouping and dispatching section* must receive the collected WEEE from the transport companies, group the WEEE in different categories, separate items that can be refurbished and dispatch auctioned items. The *auction section* should deal with the auctioning of the WEEE. On auction days this section must ensure that bidders are registered, auction deposits are collected, terms of the auction are communicated, an auction list with available items are distributed and details of buyers of lots are captured.



14. PROPOSED ACCUMULATION STATIONS



Figure: Proposed Accumulation Stations

In figure 9, accumulation stations must be located in areas (Polokwane, Nelspruit, Bloemfontein, Kimberley, George, etc.) where relatively large WEEE quantities can be collected but where very few recycling facilities can be found. Although rural areas are not seen as towns, there are some areas where large communities exist and the proposed Accumulation station could also be set up in these areas. The operation of the Accumulation station will be similar to the auction houses in the sense that the public and collection points will communicate with the Accumulation Station when WEEE must be collected. When enough material is available the WEEE will be transported to the auction houses. In rural areas the municipal refuse site, schools, community halls, etc. could act as the Accumulation Station to centralize collections. If recycling facilities are established near accumulation stations the selling price of WEEE items can be negotiated by using the market prices received at the auction houses. From [46] the "Process structure" is proposed for the accumulation stations.

15. FREE-RIDERS AND MEASURES TO PENALIZE FREE-RIDING

According to [47] "Free-riders" are actors, individuals or businesses, in an EPR system that does not pay for the benefits they receive. Free riders will always be part of any system but the existence of them should be limited. In the proposed model care was taken to minimize free-riders. The proposed measures to limit free riding of stakeholders are fines (continuous if no remedies occur), criminal charges and the termination of a business entity.

16. ORPHAN AND EXISTING PRODUCTS

"Orphan products" are products of which the producers are non-existent due to bankruptcy or some other reason while "Existing (pre-existing) products" (sometimes referred as "historic" items) are products on the market at the time the EPR policy is to be introduced [47]. Orphan and existing products are a reality in South Africa and the proposed management model includes orphaned and existing products that need to be recycled.

16.1 Financing options for addressing orphan and existing products

The proposed ADF fee which should be levied at the sale of new items will be used to fund the recycling efforts of end of life (EOL) equipment, regardless if the WEEE is orphan or existing/pre-existing products. New product sales should thus fund all current WEEE.

16.2 Non-recyclability of products

If a product is essential, non-recyclable, and the only product available in South Africa, the ADF fee must be set at a level that will ensure the product is safely discarded at a hazardous site. If there are alternative recyclable products on the market the non-recyclable product must be given a time period of two years to be changed to a recyclable product. If there is more than one alternative product the ADF fee should be the same for all similar products but the difference between the normal ADF fee and the cost to dispose of the non-recyclable item must be funded by the producer.

17. COMPETITION ISSUES

The proposed EPR model ensures fair competition in the import market, the product market, the recyclable materials market, the product collection (transport) market, and the recycler market,



18. WEEE DESIGN AND TREATMENT SYSTEMS IN SOUTH AFRICA

To eliminate problematic practices associated with the environmental protection in South Africa the researcher suggests: All refurbishers and recyclers must be ISO 14001 compliant, all recyclers must be in position of a Precious Metals Refining License, a waste transportation permit must be obtained which will allow a recycler/refurbisher to transport WEEE, a second hand goods license to enable the recycler/refurbisher to be in position of second hand goods, alternatively the above three license will be replaced by a SAWEEEA recycler license with a well-defined and specified recycling focus which enable recycling of precious metals, transportation of WEEE and of being in position of second hand goods.

19. FUTURE WORK

At publication of the paper the focus is on a proposed WEEE management model for South Africa. The proposed model was developed from litereture reviews, observations, and informal discussions with stakholders in the e-waste industry. Research is still in progress and thus no data can be provided to support the proposed model. Data will be collected during the second half of 2016 to test and validate the proposed model.

20. CONCLUSION

The environmental impact induced by WEEE and the value of the materials contained in WEEE need to be addressed in South Africa. Recycling efforts are mainly conducted for the financial benefit of recyclers while environmental problems are not addressed. To address the problems and opportunities it is vital to introduce legislation and a WEEE management model. Public awareness regarding environmental protection needs urgent attention to ensure the safe disposal of WEEE.

21. ACKNOWLEDGEMENTS

The author gratefully acknowledges the financial support received from the Department of Higher Education to do a preliminary study at South African enterprises as well as in European countries. Inputs and discussions received from industry refurbishers and recyclers are also appreciated.

22. REFERENCES.

- [1] [2] International Council on Systems Engineering. 2006. Systems Engineering Handbook. 3rd ed. INCOSE.
- Advanced Tropical Environment (ATE). 2012. Identification of the magnitude of the electrical and electronic (e-waste) situation in South Africa: A strategic approach to international chemicals management (SAICM) e-waste as an emerging policy issue. November 2012.
- [3] Webster, A. Fall 2002. Keeping Africa out of the global backyard: A comparative study of the Basel and Bamako conventions.
- [4] Widmer, R., Oswald-Krapf, H., Sinha-Khetriwal, D., Schnellmann, M. & Böni, H. 2005. Global perspectives on e-waste. Environmental impact assessment review, 25(5):436-458.
- [5] Fetzer, A. 2009. E-waste management in South Africa, Kenya and Morocco: Developing a pathway to sustainable systems. Report commissioned by Hewlett-Packard 2009.
- Finlay, A. and Liechti, D. 2008. E-waste assessment South Africa. Johannesburg, South Africa: [6] Openresearch, Empa.
- [7] Finlay, A. 2005. E-Waste challenges in developing countries: South Africa case study, APC "Issue papers". Series 2005, November. 1-22.
- [8] Bereketli, I., Erol Genevois, M., Esra Albayrak, Y. & Ozyol, M. 2011. WEEE treatment strategies' evaluation using fuzzy LINMAP method. Expert systems with applications, 38(1):71-79.
- United Nations Environment Program. 2007. E-waste. Volume I: Inventory Assessment Manual. [9]
- [10] Ejiogu, A.R. 2013. E-waste economics: A Nigerian perspective. Management of Environmental Quality, 24(2), pp. 199-213.
- Tsai, W.T. 2014. Chlorofluorocarbons. In: P. Wexler, ed, Encyclopedia of Toxicology (Third Edition). [11] Oxford: Academic Press, pp. 883-884.
- 2008. Evaluating Electronic Waste Recycling Systems: The Influence of Physical [12] Fredholm, S. Architecture on System Performance. Master of Science in Technology dissertation, Massachusetts Institute of Technology.
- Kahhat, R., Kim, J., Xu, M., Allenby, B., Williams, E. & Zhang, P. 2008. Exploring e-waste [13] management systems in the United States. Resources, Conservation and Recycling, 52(7), pp. 955-964.
- [14] Lindhqvist, T. 2000. Extended Producer Responsibility in Cleaner Production: Policy principle to promote environmental improvements of product systems. Unpublished PhD, Lund University, Lund Sweden.



- [15] Gottberg, A., Morris, J., Pollard, S., Mark-Herbert, C. & Cook, M. 2006. Producer responsibility, waste minimisation and the WEEE directive: Case studies in eco-design from the European lighting sector. *Science of the total environment*, 359(1-3):38-56.
- [16] South Africa. 2008. National Environmental Management: Waste Act 59 of 2008. National Waste Management Strategy. 2012.
- [17] Aizawa, H., Yoshida, H. & Sakai, S. 2008. Current results and future perspectives for Japanese recycling of home electrical appliances. *Resources, Conservation and Recycling*, **52**(12), pp. 1399-1410.
- [18] (U.N. EPA, 2012:7) United States Environmental Protection Agency (U.S. EPA). 2012. Recycling and waste electrical and electronic equipment management in Taiwan: A case study.
- [19] Sander, K., Schilling, S., Tojo, N., Van Rossem, C., Vernon, J., & George, C. 2007. The Producer Responsibility Principle of the WEEE Directive. Final Report.
- [20] Argument, L., Lettice, F. & Bhamra, T. 1998. Environmentally conscious design: matching industry requirements with academic research. *Design Studies*, 19(1), pp. 63-80.
- [21] AndiÇ, E., Yurt, Ö. & BaltacioĞlu, T. 2012. Green supply chains: Efforts and potential applications for the Turkish market. *Resources, Conservation and Recycling*, **58**(0), pp. 50-68.
- [22] United Nations Environment Program. 2009. Sustainable innovation and technology transfer industrial sector studies. Recycling From e-waste to resources.
- [23] Zeng, X., Li, J., Stevels, A.L.N. & Liu, L. 2013. Perspective of electronic waste management in China based on a legislation comparison between China and the EU. *Journal of Cleaner Production*, **51**(0), pp. 80-87.
- [24] Panamburan-Ferse, M. & Breiter, A. 2013. Assessing the side-effects of ICT development: E-waste production and management: A case study about cell phone end-of-life in Manado, Indonesia. *Technology in Society*, 35(3), pp. 223-231.
- [25] Manomaivibool, P. & Hong, J.H. 2014. Two decades, three WEEE systems: How far did EPR evolve in Korea's resource circulation policy? *Resources, Conservation and Recycling*, **83**(0), pp. 202-212.
- [26] Khetriwal, D.S., Kraeudhi, P. & Widmer, R. 2009. Producer responsibility for e-waste management: Key issues for consideration - Learning from the Swiss experience. *Journal of environmental* management, **90**(1), pp. 153-165.
- [27] Solving the E-Waste Problem (StEP) White Paper. 2009. E-waste Take-Back System Design and Policy Approaches.
- [28] Oliveira, C.R.D., Bernardes, A.M. & Gerbase, A.E. 2012. Collection and recycling of electronic scrap: A worldwide overview and comparison with the Brazilian situation. *Waste Management*, **32**(8), pp. 1592-1610.
- [29] United Nations Environment Program. 2011. Basel Convention on the control of transboundary movements of hazardous wastes and their disposal. Protocol on liability and compensation for damage resulting from transboundary movements of hazardous wastes and their disposal.
- [30] Nnorom, I.C. & Osibanjo, O. 2008. Overview of electronic waste (e-waste) management practices and legislations, and their poor applications in the developing countries. *Resources, Conservation and Recycling*, **52**(6), pp. 843-858.
- [31] Dittke, M. 2009. A review of South African environmental and general legislation governing e-waste. Electronic Waste Association of South Africa (eWASA).
- [32] WEEE forum. <u>http://www.weee-forum.org/services/key-figures-platform [</u>Accessed 08 September 2014].
- [33] WORLD CONSUMER INCOME AND EXPENDITURE PATERNS. 2013. 13th ed. London: Euromonitor International.
- [34] Yoshida, F. & Yoshida, H. 2010. Japan, the European Union, and Waste Electronic and Electrical Equipment Recycling: Key Lessons Learned.
- [35] Kiddee, P., Naidu, R. & Wong, M.H. 2013. Electronic waste management approaches: An overview. *Waste Management*, 33(5), pp. 1237-1250.
- [36] Statistics South Africa: <u>http://beta2.statssa.gov.za/?page_id=1021&id=city-of-tshwane-municipality</u> [Accessed 22 Aug 2014].
- [37] Tshwane metro website: <u>http://www.tshwane.gov.za/Services/WasteRemoval/Pages/Dumping-Sites.aspx</u> [Accessed 22 Aug 2104].
- [38] Pick a Pay: <u>http://www.picknpay.co.za/store-search-results?store_keyword=pretoria&cat_ids=-</u> <u>1&si_5178=30&s=_5178</u> [Accessed 22 Aug 2014].
- [39] Builders Warehouse in Tshwane area: <u>https://www.builders.co.za/store-finder [</u>Accessed 01 April 2016].
- [40] Checkers stores in Tshwane area: <u>http://www.checkers.co.za/store-locator.html [</u>Accessed 01 April 2016].
- [41] Shoprite stores in Tshwane area: <u>http://web.shoprite.co.za/store-locator.html</u> [Accessed: 01/04/2016].
- [42] Hi-Fi Corporation stores in Tshwane area: <u>http://www.hificorp.co.za/storelocator/</u> [Accessed: 01/04/2016].



- [43] Dionwired stores in Tshwane area: <u>http://www.dionwired.co.za/storefinder?province=Gauteng -</u> <u>Pretoria</u> [Accessed: 01/04/2016].
- [44] Game stores in Tshwane area: <u>http://www.game.co.za/storelocator</u> [Accessed: 01/04/2016].
- [45] Spar stores in Tshwane area: <u>https://www.spar.co.za/Store-</u> Finder.aspx?province=13&searchtext=pretoria&searchmode=anyword [Accessed: 01/04/2016].
- [46] Frameworks for organizational design-Executive Brief: http://isites.harvard.edu/fs/docs/icb.topic608877.files/Class%20Nine%20Reading/CLC_Frameworks_for r_Organizational_Design.pdf [Accessed: 05/04/2016].
- [47] OECD. 2001. Extended Producer Responsibility: A Guidance Manual for Governments. OECD, Paris, pp. 84-86.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



BUSINESS INTELLIGENCE PERFORMANCE MEASURES IN LOCAL GOVERNMENT

J.J. Fourie^{1*} & J. Stimie² & P.J. Vlok³

¹Department of Industrial Engineering University of Stellenbosch, South Africa 19703392@sun.ac.za

²Department of Industrial Engineering University of Stellenbosch, South Africa johann.stimie@gmail.com

³Department of Industrial Engineering University of Stellenbosch, South Africa pjvlok@sun.ac.za

ABSTRACT

An estimated 2800 violent service delivery protests occurred from March 2004 to March 2008 in South Africa. The period between 2009 and 2013 had no less than 10 000 public gatherings. The use of violence during protests increased from under 50% of protests in 2007 to more than 80% of protests in 2014. Service delivery protests have become a way for the public to show their discontent with local government and unfulfilled promises by politicians.

Business intelligence and analysis can aid local government to make better use of available resources that are needed to support an ever growing public demand. This paper attempts to show that performance measures, as part of business intelligence and analysis, can be used to help local government address the grievances of service delivery protesters.

¹ The author was enrolled for an M Eng (Engineering Management) degree in the Department of Industrial Engineering, University of Stellenbosch

² The author graduated with a PhD (Industrial Engineering) degree in the Department of Industrial Engineering, University of Stellenbosch

³ The author held the position of Extraordinary Professor of Industrial Engineering in the Department of Industrial Engineering, University of Stellenbosch

^{*}Corresponding author



1. INTRODUCTION

The period between 2009 and 2013 had no less than 10 000 public gatherings. Poor service delivery was cited as the cause in 45% of these public gatherings, while the use of violence during protests increased from under 50% of protests in 2007 to more than 80% of protests in 2014. Metropolitan municipalities bore the brunt of protests between 2012 and 2014 with 55% of protests occurring within the borders of metropolitan municipalities. [1]

A survey done in Gauteng in 2009 showed that 57% of respondents were satisfied with national government, 50% with provincial government and only 40% were satisfied with their municipalities. Respondents stated that they were not satisfied with local government in 27% of the cases, while 13% stated that they were very dissatisfied. [2]

South Africa seems to have the required legislative framework for good governance, as will be illustrated in this paper. The apparent failure of strategy execution and a lack of skills has been cited as the biggest hurdle to the proper functioning of local government. The importance of community participation and communication between government and citizens on expenditure should, however, not be overlooked. [3]

This paper attempts to illustrate how performance measures can be used to help local government to overcome some of the barriers caused by poor strategy execution and communication, a source of frustration for service delivery protesters. The following aspects will be addressed in the paper:

- The South African local government landscape;
- Management control systems, with a specific focus on the role and function of performance measures; and
- The use of performance measures within local government in North America.

The next section focuses on the South African local government landscape.

2. THE SOUTH AFRICAN LOCAL GOVERNMENT LANDSCAPE

Local government in South Africa is faced with numerous challenges. One of these challenges is the waves of service delivery protests across South Africa. Service delivery protests mostly stem from the lack of access to basic services, including access to sanitation, water, refuse removal, electricity and basic housing. [4]

The local government landscape in South Africa will be explained in more detail in this section. The focus will fall on the levels of local government in section 2.1 and the nature and extent of service delivery in South Africa will be examined in section 2.2. The local government legislative and planning framework is presented in section 2.3.

2.1 The levels of local government in South Africa

The Republic of South Africa has a parliamentary system that operates at both national and provincial level with all bodies of government adhering to the supreme rule of law, the Constitution.

At provincial level, the burden of service delivery lies with three different kinds of municipalities:

- Category A or metropolitan municipalities. There are six metropolitan municipalities in South Africa with more than 500 000 voters in each. Metropolitan municipalities are completely distinct from local or district municipalities;
- Category B or local municipalities. There are 231 local municipalities in South Africa. Local municipalities are often directly associated to towns and cities that are not part of metropolitan municipalities; and
- Category C or district municipalities. District municipalities oversee local municipalities, between three and six local municipalities are grouped together under a district municipality. There are 47 district municipalities in South Africa that coordinate and assist development and service delivery between local municipalities. Citizens who live in low population areas, like wildlife and game parks, fall under district municipalities.

All local services and development are undertaken by metropolitan municipalities, while local and district municipalities share the burden.



Municipalities are responsible for some of the following functions:

- Electricity delivery;
- Sewage and sanitation;
- Refuse removal;
- Municipal health services;
- Municipal roads;
- Street trading;
- Parks and recreational areas;
- Local tourism;
- Water for household use;
- Storm water systems;
- Fire fighting services;
- Decisions around land use;
- Municipal public transport;
- Abattoirs and fresh food markets; and
- Libraries and other facilities. [5]

Local government, or municipalities, are responsible for the provision of services to the community through accountable and sustainable methods. Municipalities must also promote social and economic development and encourage the community to be involved with the issues that affect them. The aim of encouraging local community involvement is to ensure that services and issues can be prioritised by the community as the beneficiaries. [6] [7]

The levels of local government are clearly set out by the national government of South Africa through parliament. Service delivery has been a ground level issue for most of South Africa's young democracy.

2.2 The nature and extent of service delivery in South Africa

Service delivery protests mostly stem from the lack of access to basic services, including access to sanitation, water, refuse removal, electricity and basic housing. The Financial Crisis that occurred from 2007 to 2009, have exacerbated the problem as South Africa lost the most formal sector jobs of any major country in the world after Spain. Over one million formal jobs were lost. [8]

Besides being sparked by inadequate service delivery and a high unemployment rate, service delivery protests share a set of characteristics:

- Service delivery protests take place in informal settlements or low-income areas that form part of
 metropolitan municipalities or large cities;
- Service delivery protests bear little to no evidence of formal planning or association with institutions;
- Protesters feel marginalised; and
- Protesters feel that they are not being heard by political figures. [2]

The lack of community involvement and communication by officials and local authorities adds fuel to the fire of these protests. The perception that protestors have can be altered by better communication channels between local government and citizens. [8] [6]

Regulations have been introduced to aid local government in service delivery. In the next section reference will be made to the South African local government legislation and legal framework. Specific reference will be made to the Local Government: Municipal Systems Act, 2000 (Act No. 32 of 2000) and the National Development Plan 2030.

2.3 Regulations and guidance on performance management and measures in local government

Local government in South Africa operates within a legislative framework. Specific reference will be made to:

- The constitution of the republic of South Africa;
- The National Development Plan 2030; and
- The Local Government: Municipal Systems Act.

The constitution of South Africa states that the public must participate in policy decision making processes and that public administration should make information of an accurate nature available to the public in a timely and accessible nature.



Chapter 13 of the National Development Plan 2030 suggests ways to improve service delivery and that a new funding model for municipalities is needed. [8] [6]

The National Development Plan 2030 identifies key areas that need to be focussed on to improve local government and service delivery.

- Accountability and oversight should be strengthened Citizens need access to information to hold public servants and politicians accountable;
- Interdepartmental coordination needs to happen Communication between officials to make decision • making easier; and
- Relations and oversight between national, provincial and local government is needed An enabling framework and oversight is needed so that resource use can be planned over provinces and municipalities. [9]

The Local Government: The Municipal Systems Act dictates that municipalities must adopt a performance management system and, at the same time, ensure community involvement with setting performance measures. The Local Government: Municipal Systems Act describes a performance management system in municipalities as a framework that represents how the cycle and processes of performance planning, monitoring, measurement, review, reporting and improvement will be conducted, organised and managed, including determining the roles of the different stakeholders.

The performance measures must include input, output and outcome measures. The performance measures must be measurable, objective, relevant and precise and also be reviewed annually. [10]

The act prescribes certain performance measures. These are:

- The percentage of households with access to basic level of water, sanitation, electricity and solid waste removal;
- The percentage of households earning less than R1100 per month with access to free basic services; .
- The percentage of a municipality's capital budget actually spent on capital projects identified for a particular financial year;
- The number of jobs created through municipality's local, economic development initiatives including capital projects;
- The number of people from employment equity target groups employed in the three highest levels of management;
- The percentage of a municipality's budget spent on implementing its workplace skills plan; and
- The financial viability of a municipality as expressed by the following ratios:
 - Debt coverage = Total operating revenue received Operating grants 0
 - Debt service repayments
 - Outstanding service debtors to revenue = Annual revenue received for services Total outstanding service debtors 0
 - 0
 - Cost coverage = Available cash + Investments Monthly fixed operating expenses

The Auditor-general's report on the Public Performance Management Act for the financial year 2014-2015 covered 167 national and provincial departments and 301 public entities. The report showed a slight improvement in ungualified financial audits but these were still at a dismal rate of 28% of audits. The Auditor-general states in the report that reviewing and monitoring laws and legislation by leadership is needed. [11]

South Africa seems to have the necessary legislative framework for good governance, as shown in this section. The biggest hurdles to the proper functioning of local government has been cited as:

- Failure to execute strategies;
- A lack of skills; •
- A lack of community participation; and •
- A lack of communication between government and citizens. [3]

Strategy execution is often enhanced through the implementation of management control systems and more specifically through effective measurement and reporting on carefully selected performance measures. Over time a number of techniques and artefacts have been developed to assist managers in this process. In the next section a brief discussion on some of the most cited techniques and artefacts that are included in the management control systems discourse will be presented. The concepts management control systems and performance management are often used interchangeably in the literature. In this paper the concepts will also be used interchangeably.



3. MANAGEMENT CONTROL SYSTEMS

The definition of management control has evolved since its first recorded use by Robert Anthony (1965), where he stated that management control is a process to ensure that resources in an organisation are obtained and used efficiently and effectively towards the accomplishment of goals. Lowe (1971) expanded the definition to include accountability and feedback from stakeholders, in a system of information gathering, in an enterprise that is adapting to changes in its environment. The information is then used to track the achievement of goals and make corrections where necessary. [12] [13] [14]

The management of modern organisations requires an interdisciplinary and systems thinking approach to handle the complex interactions between departments and between strategy and execution. A number of studies around management control systems have resulted in the development of various techniques that attempt to ensure the most efficient use of resources. Performance measures are used to report on the efficiency of an organisation. Some of the most cited techniques include:

- Activity Based Costing;
- Balanced Scorecard;
- Benchmarking;
- Budgeting;
- Just-In-Time;
- Total Quality Management;
- Economic Value Added;
- Six Sigma; and
- Decision support systems. [13] [14] [15] [16] [17]

A detailed analysis of each of these techniques is beyond the scope of this paper but each of these techniques are briefly introduced in the next section. The section concludes with an introduction to decision support systems, its application and potential value.

3.1 Management Control System: Techniques and artefacts

Activity Based Costing - Activity Based Costing reveals the flow of an organisation's resources. The consumption of resources to deliver a service or create a product, as well as the generated income from these activities is revealed. [18]

Activity Based Costing consists of two phases:

- The resources consumed to create products or to deliver services are identified; and
- The resources are directly allocated to specific products or services. [13]

Managers can use Activity Based Costing to focus on areas that will have the biggest impact on expenditure and income generation. [18]

The Balanced Scorecard - The Balanced Scorecard is a management control system that makes use of financial and non-financial measures to address the shortcomings of accounting systems that rely on historic information alone. The measures used are part of four groups:

- Learning and growth;
- Business processes;
- Customer focus; and
- Fiscal focus.

The exact measures used with a Balanced Scorecard approach may differ between organisations and between business units, divisions or departments within an organisation. [13] [14]

Benchmarking - Benchmarking seeks to learn from the experiences of existing and successful organisations by establishing the standards and examining the processes used. Benchmarking attempts to correlate desired outcomes with certain inputs and reduce ambiguity. [16]

Budgeting - Budgeting is a form of planning and setting goals, but can also be used to evaluate performance in review. Resource allocation and control is the main aim of budgeting. [15]

Just-In-Time - The aim of Just-In-Time as a management control system is to improve the quality of manufacturing by improving production processes. Improving production processes are mainly achieved through the:

- Reduction of production costs;
- Elimination of waste; and
- Recognition of worker's abilities.



In essence, the right amount of goods are produced at the right time. [19]

Total Quality Management - Total Quality Management emphasises:

- Continuous improvement;
- Reducing rework;
- Constant measurement of results; and
- Long term thinking.

Total Quality Management attempts to improve the bottom line of an organisation holistically, therefore, the list above only captures the essence of Total Quality Management. [20] [16]

Economic Value Added - The creation of shareholder value in a private organisation can be measured using Economic Value Added. Economic Value Added is calculated by deducting the cost of debt and equity from operating profit. Economic Value Added is sometimes used alongside return on investment and earnings per share. [13] [15]

Six Sigma - Six Sigma uses the captured data from business processes to continually reduce defects in the delivery of goods or services. Six Sigma attempts to have less than 3.4 defects per million opportunities, therefore, a success rate of more than 99.99%. Opportunities are the number of chances that a unit, in the case of manufacturing, would have a defect. [13]

Decision support systems - Decision support systems is a term used to describe computer based systems, mostly databases and software, which aid in solving problems and making decisions. Organisations can separate the viewable data by department or management level to ensure the privacy of information, for example limiting human resource information to only be viewed by the human resources department. [13]

The technology trends that started the data era can be loosely attributed to the start of the internet in the 1970s and the ubiquitous nature of the World Wide Web that followed. [21]

Special skills and capabilities have been fostered to enable organisations to extract information from these vast amounts of data with numerous benefits, including:

- Timely access to information;
- Reduced cost;
- Increased productivity; and
- Improved employee and customer satisfaction. [13]

In the next section business intelligence and analysis is introduced. Business intelligence and analysis uses performance measures to understand and keep track of trends in an organisation, enabling easier and better decision making. The communication of ideas and abstract concepts between stakeholders, such as different levels of local government, are aided by business intelligence and analysis.

3.2 Business intelligence and analysis

Business intelligence and analysis forms part of a set of philosophies and tools that aid business leaders in decision making. Business intelligence and analysis refers to:

- The gathering and use of information to map out an organisation's position relative to its competitors and customers in a given market and economic climate An organisation's internal and external environment is mapped and put in context; and
- A system or process that an organisation uses to gather and analyse information before using the insights obtained to make decisions. [22]

Business intelligence and analysis as a decision support system has evolved over the years, starting as early as the 1970s. The two primary steps in business intelligence and analysis are:

- Collecting data; and
- Extracting the information from the data. [23]



These steps and their interaction are shown in the illustration below:



Figure 1 - A simplified business intelligence and analysis process [23]

A survey of 1 447 decision makers found that the deployment of business intelligence and analysis solutions increased by 39% since 2012. The survey also found that organisations with a longer history of using business intelligence and analysis reported a six fold increase in meeting the following objectives:

- Enhance communication and collaboration;
- Accelerate innovation of products and services; and
- Improve customer experience. [24]

In 2011, 97% of companies with revenue of more than \$100 000 000 per year made use of business intelligence and analysis. [25]

Management control systems are underpinned by performance measures and are used to determine what needs to be measured and reported on. The next chapter focusses on performance measurement and management. Focussing on performance measurement and management lays the foundation to show the value of using performance measures in local government.

3.3 Performance measurement and management

Business performance measurement and management is a combination of management and analytic processes that allows managers of an organisation to achieve pre-determined goals. [26]

Business performance measurement and management involves three main activities:

- The selection of goals;
- The consolidation of measurement information relevant to the organisation's achievement of these goals; and
- The interventions made by managers in light of this information.

Business performance measurement and management evaluates the business in a holistic way that goes beyond each division or department. The strategic and operational objectives of an organisation can be aligned. [27]

A good performance measure is difficult to determine and depends on the needs of the organisation or institution. There are similar characteristics between the different measures that organisations consider good performance measures. [28]

Good performance measures keep track of, and enable business functions that are key to the effective and efficient delivery of goods and services that are of a consistent and good quality. The data used by performance measures need to be accurate and unambiguous in order to be trustworthy and be a single version of the truth, avoiding conflict and debate that inhibits implementation. Manipulation of performance measures should be difficult, but performance measures should still be easily accessible and comprehensible. Performance measures that are easily comprehensible aids in better implementation of corrective measures when they are needed and does not create conflicting conclusions. The effectiveness of performance measures can only be of concern when the users are also responsible for decisions taken or the outcome that is being measured. Creating information from the underlying data by reducing the volume, yet still being timely enough is key. [28]



A manager may use different performance measures to keep track of unfolding events or ensure that objectives will be obtained within the allotted time given by higher management. Delegated tasks and the respective performance of employees can be tracked and measured, but what to track and measure is the key to effective performance measurement.

Performance measures can be used to monitor or alter the behaviour of systems, especially when people are a key part of the system. A caveat of performance measuring is that people will alter their usual behaviour to modify the outcome of performance measures, sometimes by circumventing safety or best practice procedures, in order to show an improvement before a management review. [29]

Manufacturers may measure certain indicators on a daily or hourly basis, sometimes even as close as possible to real-time. Measuring the performance of machines and systems on such a regular basis allows companies to keep a consistent standard without sacrificing safety or decreasing profits, allowing intervention before a major event can unfold.

[30]

The identification of applicable performance measures starts by investigating the measurable outcomes of a system or process. The outcomes are then measured against the inputs from previous parts of the process or system. A generic process for identifying performance measures is shown in the illustration below. [31]



Figure 2- A generic process for identifying performance measures [31]

The assessment of performance in the public sector is different to the assessment of performance in private organisations. The underlying difference between public sector performance measurement and private sector performance measurement is that:

- Private firms focus on profit and performance measurement and are dependent on a variety of technical procedures; and
- Public sector entities focus on a narrower set of particular social and political objectives. [28]

The legal framework and legislation discussed in section 2 provides a solid base for management control systems and can be used to eliminate at least some of the hurdles to the proper functioning of local government. The existence of a legal framework and legislation is however not enough and implementation is necessary. Evidence that performance measurement, management or reporting adequately takes place in local government in South Africa could not be identified during the research for this article.

There are different ways to identify and implement performance measures in an organisation. The discussion in section 3 shows that a properly defined decision support system provides an excellent context for identifying and implementing performance measures.

The next section discusses the results from the implementation of performance measures in North American local government entities. The discussion can be seen as a benchmark for the implementation of performance measures in local government.



4. RESULTS FROM IMPLEMENTING PERFORMANCE MEASURES IN LOCAL GOVERNMENT

The implementation and use of performance measures in local government in the United States of America are evolving from quantitative measures to more results oriented qualitative measures that also aid in decision making.

A survey of local government in the United States of America, involving 47 counties and 168 cities, showed positive results for the use of performance measures. The findings showed that using performance measures had enhanced programme management in 71% of respondent's cases and 68% agreed - while 15% strongly agreed - that the use of performance measures had resulted in an increase in efficiency. Communication between branches of local government also improved after implementing performance measures.

The most positive results yielded from the use of performance measures were:

- Improved communication with budget officers;
- Improved communication with ward councillors;
- Improved communication with citizens;
- Improved communication between departments;
- An increase in awareness and focus on results from local government;
- An increase in awareness of factors that affect results;
- An increase in service quality;
- An increase in the effectiveness of department programs; and
- Better adaptability of programs to achieve the desired results.

The research confirms that implementing performance measures in local government improved communication within and across branches of government and between local government and citizens. Discussion about the results of government activities are facilitated by performance measures and the decision making process is aided by relevant information from performance measures. [32]

Performance measures at branches of local government were mainly used as budget development tools, but some respondents have successfully used performance measures as tools to communicate satisfaction levels from citizens and as knowledge sharing to citizens. [28]

Effective communication between spheres of government and between local government and citizens may remove the perception of marginalisation. [2]

Different departments at different levels of government have different needs and uses for performance measures. Parliament may focus on policy implementation and effectiveness, while national government's treasury department is more interested in financial performance and wastage and citizens at local government level are concerned with service delivery and effective tax expenditure. [28]

The Balanced Scorecard has been cited as an invaluable framework for identifying and implementing performance measures in local government. In a survey of 184 local government entities in North America the following broad types of performance measures were created and implemented with positive results:

- Financial performance;
- Operating efficiency;
- Customer satisfaction; and
- Employee performance; [33]

The use of one or more management control systems to identify performance measures is a step in the right direction for local government in South Africa to comply with the existing legislation.

No data collection standards were present in South African municipalities in 2009 and local government information was being assembled by National Treasury. This fact makes it difficult to determine the precise performance measures that should be used, but points to the fact that data capturing processes and systems are not in place or inadequate to make the data available in a trustworthy and timely fashion to allow analysis to take place. [34]

The results from implementing performance measures are clearly positive, but a well-defined system for data collection, measurement and reporting is also necessary. Decision support systems provide an excellent context for identifying, implementing and tracking performance measures.



5. CONCLUSION

The main aim of business and government might differ but that does not mean that local government entities, like municipalities, cannot make use of tried and trusted methods to ensure that resources are used efficiently and effectively. Equating revenue in the private sector to profit and revenue in the public sector to tax, makes input performance measures comparable between the two. The same thinking can be applied to products and services provided in both sectors. [35]

Employees in local government who are included in the development and determination of performance measures are more likely to make use of and find value in performance measurement. The budget effects of performance measurement are influenced by the presence and comprehensive use of performance measures in budget processing and performance. However, the simple presence of performance measures in documentation does not mean they will be used effectively in decision making, correct performance measures at appropriate stages of the budgeting process removes the chances of information overload. [32] [33]

The results from implementing performance measures in North American local government entities are positive and enables strategy execution and better communication between stakeholders. Performance measures can, and should, be implemented into local government in South Africa. The biggest hurdle to introducing performance measures into South African local government entities seems to be the lack of data capturing standards and processes. Future research needs to determine to what extent performance measures are adequately implemented in South African local governments and how a properly defined decision support system can aid implementation.

REFERENCES

- Steyn, J. 2016. An Exploratory Study: How does Local Government Perceive Community Protest?
 Nyar, A. and Wray, C. 2012. Understanding protest action: some data collection challenges for
- [2] Nyar, A. and Wray, C. 2012. Understanding protest action: some data collection challenges for South Africa, *Transformation: Critical Perspectives on Southern Africa*, 80(1), pp. 22-43.
- [3] Mathekga, R. and Buccus, I. 2006. The challenge of local government structures in South Africa: securing community participation, *Critical Dialogue Public Participation*, 2(1), pp. 11-17.
- [4] **Booysen, S.** 2007. With the ballot and the brick: The politics of attaining service delivery, *Progress in Development Studies*, 7(February 2004), pp. 21-32.
- [5] **Tibane, E.** 2014. South Africa Yearbook.
- [6] Nembambula, M. P. 2014. Violent Service Delivery Protests in the Governance of Public Participation in a Democratic South Africa, *Mediterranean Journal of Social Sciences*, 5(9), pp. 148-151.
- [7] **Duvenhage, A.** and **Venter, J.** 2009. Local government instability in a post-2009 South African election environment: analytical and strategic perspectives, *Strategic Review for Southern Africa*, 31(2), pp. 19-39.
- [8] Bond, P. and Mottiar, S. 2013. Movements, protests and a massacre in South Africa, *Journal of Contemporary African Studies*, 31(2), pp. 283-302.
- [9] **National Planning Commission**. 2010. National Development Plan (2030), *Department: The Presedency Republic of South Africa*. p. 70.
- [10] **Mufamadi, F.** 2001. Local Government: Municipal Planning and Performance Management Reguations, *Government Gazette*, 434(7146).
- [11] Makwetu, K. 2015. Public Finance Management Act Auditor-General Report,.
- [12] Otley, D. T., Broadbent, J., and Berry, A. 1995. Research in Management Control: An Overview of its Development, *British Journal of Management*, 6(S31-S44), p. 31-.
- [13] Stimie, J. and Vlok, P. J. 2015. A Physical Asset Management Strategy Execution Enforcement Mechanism for the early detection and management of Physical Asset Management Strategy Execution Failure.
- [14] **Otley, D.** 1994. Management control in contemporary organizations: towards a wider framework, *Management Accounting Research*, 5(3-4), pp. 289-299.
- [15] Malmi, T. and Brown, D. A. 2008. Management control systems as a package-Opportunities, challenges and research directions, *Management Accounting Research*, 19(4), pp. 287-300.
- [16] Chenhall, R. and Langfield-Smith, K. 1998. The relationship between strategic priorities, management techniques and management accounting: an empirical investigation using a systems approach, *Accounting, Organizations and Society*, 23(3), pp. 243-264.
 [17] Kannan, V. R. and Tan, K. C. 2005. Just in time, total quality management, and supply chain
- [17] Kannan, V. R. and Tan, K. C. 2005. Just in time, total quality management, and supply chain management: Understanding their linkages and impact on business performance, *Omega*, 33(2), pp. 153-162.
- [18] Cooper, R. and Kaplan, R. S. 1991. Profit Priorities from Activity-Based Costing., Harvard Business Review, 69(June), pp. 130-135.
- [19] **Finch, B.** and **Cox, J.** 1986. An examination of just-in-time management for the small manufacturer: with an illustration, *International Journal of Production Research*, 24(2), pp. 329-342.
- [20] **Powell, T.** 1995. Total Quality Management as Competitive Advantage : A Review and Empirical Study,



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Strategic Management Journal, 16(1), pp. 15-37.

- [21] Chen, H. and Storey, V. C. 2012. Business intelligence and analytics : From big data to big impact, *Mis Quarterly*, 36(4), pp. 1165-1188.
- [22] **Pirttimäki, V.** and Lönnqvist, A. 2006. The Measurement of Business Intelligence, *Information Systems Management*, (Winter), pp. 32-40.
- [23] Wixom, B., Ariyachandra, T., Goul, M., Gray, P., Kulkarni, U., and Phillips-Wren, G. 2011. The current state of Business Intelligence in academia, *Communications of the Association for Information Systems*, 29(1), pp. 299-312.
- [24] **Carter, S.** and **Hupfer, S.** 2014. Raising the game: The IBM tech trends study, *The Veterinary Record*, p. 16.
- [25] **Bloomberg.** 2011. The Current State of Business Analytics : Where Do We Go From Here?, *Businessweek*. pp. 1-14.
- [26] **Frolick, M.** and **Ariyachandra, T.** 2006. Business performance management: one truth, *Information Systems Management*, 23(1), pp. 41-48.
- [27] vom Brocke, J. and Rosemann, M. 2010. Handbook on Business Process Management 2, 2nd ed. Berlin: Springer.
- [28] **Carter, N.** 1991. Learning To Measure Performance: the Use of Indicators in Organizations, *Public Administration*, 69(1), pp. 85-101.
- [29] **Parmenter, D.** *Key Performance Indicators (KPI)*, 2nd ed. Hoboken, New Jersey.: John Wiley & Sons, Inc., 2010.
- [30] Neely, A., Richards, H., Mills, J., Platts, K., and Bourne, M. 1997. Designing performance measures: a structured approach, International Journal of Operations & Production Management, 17(11), pp. 1131-1152.
- [31] **Prowse, P.** and **Prowse, J.** 2010. The dilemma of performance appraisal, *Business Performance Measurement and Management: New Contexts, Themes and Challenges*, 5(2), pp. 195-206.
- [32] Melkers, J. and Willoughby, K. 2005. Models of performance-measurement use in local governments: Understanding budgeting, communication, and lasting effects, *Public Administration Review*, 65(2), pp. 180-190.
- [33] Chan, Y. L. 2004. Performance measurement and adoption of the balanced scorecard, International Journal of Public Sector Management, 17(3), pp. 204-221.
- [34] **Collins, P. D.** 2012. Introduction To the Special Issue: The Global Anti-Corruption Discourse Towards Integrity Management?, *Public Administration and Development*, 10, pp. 1-10.
- [35] Kim, G.-H., Trimi, S., and Chung, J.-H. 2014. Big-Data Applications in the Government Sector, Association for Computing Machinery. Communications of the ACM, 57(3), p. 78.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



IMPLEMENTATION OF FMECA APPROACH FOR RELIABILITY IMPROVEMENT ON DRUM PRODUCTION LINE

M. Dewa¹ and N. Luvuno²

¹Department of Industrial Engineering Durban University of Technology, South Africa <u>mendond@dut.ac.za</u>

²Department of Industrial Engineering Durban University of Technology, South Africa <u>ntokozo.luvuno@gmail.com</u>

ABSTRACT

Manufacture of industrial packaging products requires multiple processes and machines. It is imperative to ensure that these machines are reliable and available when required to perform tasks. The case-in-point, a drum manufacturer, has experienced numerous equipment breakdowns or failures on the steel drum production line. These failures affect the reliability of the drum production line. The purpose of the study was to improve machine reliability and reduce downtime for the drum production line. Reliability analysis was conducted based on the data that was collected on the MP2 system from the records for previous three months. The analysis gave evidence that major counter-measures were pertinent to improve line reliability. A cross-functional team was instituted to conduct FMECA as risk assessment to determine failure modes and failure effects of critical equipment, and incorporate the criticality analysis. The reliability of the welder and seamer equipment improved after the implementation of FMECA and one can conclude that FMECA can be the vital tool for companies that thrive for optimal performance.

¹ Corresponding Author



1. INTRODUCTION

The manufacturing industry is machine-driven and its productivity and efficiency are key factors which influence the competitiveness of different companies in this sector. There are metrics associated with the logistics and maintenance support infrastructure, which are inherent within the overall effectiveness measures of a plant system and the degree to which the system is able to accomplish its mission. Organisational success and competitiveness is highly dependent on logistics and maintenance support infrastructure as a major element of the system and its availability when needed. Plant productivity depends on equipment maintenance which influence the plant availability and reliability. Machine reliability and availability is crucial for the sustenance of business and satisfying the customers by ensuring on time delivery of products conforming to the quality specifications. Hence a maintenance system should be instituted to reduce the frequency of equipment failure by comprehending the failure mode and effects so as to improve equipment reliability and availability. Failure Mode, Effects, and Criticality Analysis (FMECA) is the identification of potential process failures, the expected modes of failure and causes, failure effects and mechanisms, estimated rate of recurrence, level of criticality, and the procedure needed for compensation such as undertaking preventive maintenance [1]. This paper deals with the improvement of reliability of a drum production line using FMECA approach. The current and future risks are identified, assessed and examined and pertinent preventive maintenance actions are implemented with the view to strengthen the maintenance support infrastructure and reduce breakdown frequency of equipment.

2. BACKGROUND

Globalisation or the integration of economies, markets, industries, cultures and policy-making around the world, has raised the level of competition in meeting customer requirements for both local and global manufacturing companies. Failure to meet customer requirements can be suicidal for business sustainability, and thus it is imperative to improve equipment reliability and availability to enhance agility in responding to customer needs [2]. Machine breakdowns have a direct impact on reliability and availability of the plant or machines to perform its required tasks. Most companies envisage having higher reliability on their machines or plant, however they do not anticipate risks in advance in order to take necessary preventive actions or conduct preventive maintenance.

The case study company, a drum manufacturer specialises on metal packaging and supply their product to chemical and petroleum manufacturing companies in South Africa. The drum manufacturer also does corrective maintenance on most equipment and machines.

The challenge faced by the company is that it has old equipment for drum assembly line and most of these machines are not on the preventive maintenance plan. The company is also facing challenges in terms of supply support for the indispensable spare parts and the associated inventories required for the performance of unscheduled and scheduled maintenance, and hence it is difficult to get the needed spares when these machines fail. Frequent machine breakdowns have huge cost implications, since they result in downtime which may result in failure to meet customer requirements, as well as unsatisfied customers which may open the gap for competitors. Given this background, this paper focuses on improving the reliability of the drum assembly line by using FMECA approach.

Figure 1 shows the flow diagram for the drum production line. The steel drum production is a multi-step process involving the uncoiling of the steel, leveling, and shearing of the material, after which the material is edged and stacked. This is followed by body formation of the steel drum by rolling the steel sheet into a round body drum, electric resistance welding, and then flanging to create lips on the shell. The flanged body is then turned into a w-shaped body at the wedger and the rings from corrugating add rigidity and reinforces the drum shell. Material for the top and bottom lids is sent to punch press for formation of bottom and top covers. These lids are taken to the seamer to make the barrel seals. At the seamer, as the shell rotates in a spinning clamp, the rollers kill the edges of the shell and top lids. This creates an interlocking seam which is then flattened by the subsequent rollers. The same process occurs for the assembly of the bottom lid. The final assembly is then tested for possible leaks, painted, coated, and cured, after which the plug is fitted to produce a customer-ready drum.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 1: Schematic for the steel drum production line

3. OVERVIEW OF FMECA

FMECA is a methodology developed as an extension to Failure Mode and Effects Analysis (FMEA). FMEA details the distinct failures of a system through the identification of failure modes, as well as identifying the causes and effects of the respective potential failure modes on system service, after which appropriate detection procedures and corrective actions are defined. FMEA can be extended by Criticality Analysis procedure (CA) for the modes of failure to come out as Failure Mode Effects and Criticality Analysis (FMECA) [3].

FMECA is a detailed systematic procedure for analysing systems with the view to identify potential failures, their causes and effects on system performance, safety of the personnel, as well as effects on the environment by means of criticality analysis [4]. It is ideal for FMECA to be conducted in the development stage, even though conducting FMECA on an existing system or machine can also yield substantial benefits [5]. FMECA may also be used to prevent defects and improve product safety, quality and availability of machines. However, the main purpose of FMECA is to determine the product design, manufacturing process or equipment failure mode, discover the influence of failure on the equipment or system and to provide qualitative and quantifiable assessment, and then take corrective actions and preventive actions [6]. The results acquired from conducting a FMECA may be implemented to improve reliability, availability and maintainability of the equipment.

FMECA is a highly informative technical approach which is characterised by failure mode, that is the manner of failure, and causes connected at each failure mode. However, one failure mode can be connected to more than one cause. The failure mode can be examined in terms of performance measurement for each element of the equipment or system. The result of each and every failure mode is signified as effect and should be analysed and recorded in FMECA [7].

FMECA is a broadly-applied method which can be executed in diverse steps of the product life cycle such as design and development, manufacturing, distribution, services and other activities to improve quality, availability and reliability. FMECA is widely utilised in many industries and has been used in the automotive industry to identify potential failures in advance and take preventive actions [6]. FMECA is applied to determine risks of failure in the process or equipment to ensure that the company or organisation meet customer requirements such as quality products and ensure that equipment is always available to perform its function [8].

Companies use FMECA for many different reasons, but the main idea is to apply FMECA to analyse risks of failure mode and for improvement purposes. Applying FMECA provides an opportunity to investigate and identify critical elements of the equipment or system [6]. If FMECA analysis is successfully conducted, it would yield a data table that comprise highly enriched knowledge. The quality of FMECA depends on the validity and the timely nature of the processed data [9]. On overall, FMECA is an effectual tool for continual and effecting the proposed changes would yield better reliability and safety results [10]. However, in this paper, FMECA is essentially used to improve machine reliability and reduce downtime for the steel drum production line.

4. METHODOLOGY

The methodology used to improve reliability at the drum manufacturer commenced by instituting a crossfunctional team and defining the system requirements. A functional analysis of the maintenance actions that were required to restore a system to operational use were identified. The failure modes or manner in which



drum production line equipment could potentially fail to meet its requirements or fail to function were identified. The causes of failure or potential reasons behind a failure mode were then determined. Determination of the effects of failure or potential non-conformance was then undertaken after which failure means detection were identified. This was accomplished by establishing all the current controls, design features and verification procedures that would result in the detection of potential failure modes.

A qualitative assessment of the seriousness of the effect of the potential design failure mode was executed to determine the severity of a failure mode. The degree of severity was measured on a scale of 1 - 10, with very high effects being 9 to 10 and minor effects being 1. The frequency of occurrence or probability of failure of each individual failure mode was then determined using a scale of 1 - 10. Remote failure (failure is unlikely) was allocated a 1-2 while very high (failure is almost inevitable) was indicated as 8 -10.

The probability that design features, aids or verification procedures will detect potential failure modes in time to prevent a system-level failure are also crucial in implementing FMECA. Thus, it was also crucial to determine the probability that the failure will be detected and this was scaled from 1-10, with absolute certainty of detection being ranked 1-2 while absolute certainty of non-detection was ranked 9-10. The risk priority number (RPN) which is a product of the severity, occurrence, and detection rankings was then computed to facilitate the analysis of failure mode criticality. It was used as a means of prioritising potential failures and for the identification of critical areas so as to reduce the overall degree of severity, occurrence, and detection rankings.

5. SCENARIO BEFORE FMECA IMPLEMENTATION

The reliability analysis was conducted on key machines on drum assembly line to identify the reliability of each machine. The data was collected on MP2 maintenance software with the objective to analyse the reliability and focus was on the breakdowns to determine Mean Time Before Failure (MTBF) for each machine as it is a main parameter of reliability. The performance of each machine or equipment was sampled for a period of three months.

5.1 Mean Time Before Failure

Table 1 shows the MTBF results for the key machines for the drum production line. According to the results, the leak tester experienced the longest MTBF of 480 hours followed by the flanger and swedger which have 320 hours. On the other hand, the welder machine has the least MTBF (8 hours) compared to all other machine. That means the welder machine broke more frequent than other machine, and then followed by seamer with the MTBF of 80 hours.

Table	1٠	MTBF	ner	equipment
Table	••		per	equipment

Equipment	Welder	Flanger	Swedger	Corrugator	Seamer	Leak tester
MTBF (hours)	8	320	320	160	80	480

5.2 Reliability analysis

Reliability can be defined as the probability that a system or product will perform in a satisfactory manner for a given period of time, or in the accomplishment of a mission, when used under specific operating conditions. The reliability function R(t) may be expressed as

$$R(t) = 1 - F(t)$$

(1)

where F(t) is the probability that the system will fail by time t. F(t) is basically the failure distribution function or the unreliability function. If the random variable t has a density function f(t), (i.e. probability that the variable takes the value f(t)) then the expression for reliability is

$$R(t) = 1 - F(t) = \int_{t}^{\infty} f(t)dt$$
 (2)

Assuming that the time to failure is described by an exponential density function, then
$$f(t) = \frac{1}{\theta} e^{-t/\theta}$$
(3)

where θ is the mean life, t is the time period of interest, and e is the natural logarithm base (2.7183). The reliability at time t is

$$R(t) = \int_{t}^{\infty} \frac{1}{\theta} e^{-t/\theta} dt = e^{-t/\theta}$$
(4)

Mean life (θ) is the arithmetic average of the lifetimes of all items considered. An exponential probability function can be used to represent random failures and is widely used to model reliability phenomena. The assumption is that the failure rate of a system is independent of its age and other characteristics of its operating history, and given that scenario, the mean time to failure can be used to describe reliability with permissible accuracy once the exponential distribution parameter, the failure rate, is directly associated with mean time to failure [10]. The mean life (θ) for the exponential function is equivalent to Mean Time Between Failure (MTBF). Thus,







Figure 2 shows the reliability of the key equipment of the drum production line, based on reliability analysis. The plant generally operates for about 8 hours a day for 20 days a month, which cascades to 480 hours for the three month period. Figure 2 indicates that the welder and seamer have less reliability compared to other equipment from the sampling period of three months. The possible reasons that these two machines could have less reliability compared to other machines is the numerous number of breakdowns they experienced from that sampling period of three months is poor preventive maintenance system or failure to adhere to planned maintenance by the maintenance department.

6. IMPLEMENTATION OF FMECA

A cross-functional team comprising members from the maintenance department, machine operators and their supervisors as well as member from the stores department was developed. The failure modes or manner in which the welder and seamer could potentially fail to meet requirements or fail to function were identified. The causes of failure were then determined. A qualitative assessment of the seriousness of the effect of the potential design failure mode was executed to determine the severity of a failure mode using a severity rating scale shown in Table 2. The frequency of occurrence of each individual failure mode was then determined using the occurrence rating scale shown in Table 3, after which the probability that design features, aids or verification procedures will detect potential failure modes in time to prevent a system-level failure was also ascertained using the detection rating scale shown in Table 4.

Rating	Description	Definition				
10	Dangerously high	Failure could mean physical injury to the user or an employee				
9	Extremely high	Failure would result in violation of national regulations				
8	Very high	Failure makes the equipment inoperable or unserviceable				
7	High	Failure would result in significant customer dissatisfaction				
6	Moderate	Failure would cause partial breakdown				
5	Low	Failure's impact on equipment/process performance would be sufficient to generate complaints				

Table	2:	Severity	rating	scale
Tuble		Severicy	i u u i i i g	Jeale



4	Very low	Failure would result in minor performance loss				
3	Minor	Failure is a small nuisance but does not result in performance loss				
2	Very minor	Failure may have such minor consequences thus is unlikely to be apparent				
1	None	Failure would not be noticed or affect the equipment/process				

Rating	Description	Potential Failure Rate
10	Exceedingly high	Failures occurring more than once/day
9	Very high	Failures occurring every three to four days
8	High	Failures occurring once/week
7	Relatively high	Failures occurring once/month
6	Moderate	Failures occurring once every three months
5	Relatively low	Failures occurring once every six months
4	Low	Failures occurring once a year
3	Very low	Failures occurring once every one to three years
2	Relatively remote	Failures occurring once every three to five years
1	Remote	Failures occurring once in more than five years

Table 3: Occurrence rating scale

Table 4: Detection rating scale

Rating	Description	Definition				
10	Complete Uncertainly	Equipment is either not inspected or the effects of failure are not detectable				
9	Very Uncertain	Highly unlikely to detect failure				
8	Uncertain	Very unlikely to detect failure				
7	Very low	Equipment undergoes 100% manual inspection				
6	Low	Equipment undergoes 100% manual inspection usir go/no-go or other mistake proofing gauges				
5	Moderate	Fair chance of detecting failure				
4	Relatively high	Good chance of detecting failure: 95%				
3	High	High chance of detecting failure: 99%				
2	Very high	Equipment is 100% automatically inspected				
1	Almost certain	There is 100% automatic inspection with excellent maintenance of the inspection equipment				



The probability that the failure will be detected was then determined, after which the risk priority number (RPN) was then computed using the formula shown in Equation 6.

 $RPN = S \times O \times D$

(6)

In this case, S represents the severity index which relates to the assessment of the magnitude of the failure mode's effect on machine or component operation. *O* designates the occurrence index which relates to the probability that a failure mode occur within a defined period, and this generally depends on prior knowledge of the reliability of the equipment or component under study. *D* is the detection index and this is a function of the ability to identify, diagnose and eliminate or preventing the commencement of a breakdown prior to the manifestation of its effects on the system or personnel [4].

Table 5 shows the RPN rating for potential failure modes for the welder. It indicates that there is a higher risk on the lap gauge, skew roller and weld wheels potential modes since each has an RPN of 560.

Equipment component	Potential Failure Mode	Potential Failure Cause	Potential Failure Effect	Current Process Controls	Severity	Occurrence	Detection	RPN
Lap gauge	Loose lap gauge	Vibration	Misalignment of overlap	None	8	7	10	560
Skew rollers	Seized bearing	No lubrication	Misalignment of overlap	None	7	8	10	560
Weld wheel	Weld wheel poorly performing	Wear and tear	Excessive leakers	None	8	7	10	560
Insulator	Cracked or broken insulator	Loose insulation, vibration	Electrical faults	None	9	4	10	360
Control panel	Incorrect temperature readings	Electronic failure	Inconsistent temperature	Visual from screen	9	10	2	180
Cooling system	Leaking water pipes	Old pipe and excessive pressure	Welder head heat up	None	8	3	10	240
Transformer	Broken windings	Machine overheating	Equipment stops working	None	10	2	10	200

Table 5: Determination for RPN for Welder

The recommended actions were then identified for the welder components and that include instituting a Planned Maintenance (PM) once every two weeks and that responsibility was given to the maintenance planner as shown in Table 6.



				Action results				
Equipment component	Potential Failure Mode	Recommended Action	Action Taken (Yes/No)	Severity	Occurrence	Detection	RPN	
Lap gauge	Loose lap gauge	Conduct PM once every two weeks	Yes	8	4	7	224	
Skew rollers	Seized bearing	Conduct PM once every two weeks	Yes	7	5	7	245	
Weld wheel	Weld wheel poorly performing	Conduct PM once every two weeks	Yes	8	6	8	384	
Insulator	Cracked or broken insulator	Conduct PM once every two weeks	Yes	9	2	7	126	
Control panel	Incorrect temperature readings	None						
Cooling system	Leaking water pipes	Conduct PM once every two weeks	Yes	8	2	8	128	
Transformer	Broken windings	Conduct PM once every two weeks	Yes	10	2	7	140	

Table 6: Recommended actions and results for welder

Figure 3 visually depicts that there is the higher risk on the pressure accumulator potential failure mode as it has RPN of 360 and the corrective or prevention measure on the pressure accumulator needs to be priorities as it shows that there is higher criticality or risk on that potential failure mode. The recommended action that was then implemented was putting in a monthly preventive maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility may preventive maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance schedule for seaming chuck and rollers and that responsibility was given to the maintenance planner as well.



Figure 3: Seamer RPN rating


7. FINDINGS AFTER FMECA IMPLEMENTATION

The FMECA results for the key equipment were reviewed after corrective or prevention measures were taken for the high risk areas of improvement.

No.	Potential Failure Mode	RPN			
1	Weld wheels worn out	384			
2	Seized bearing	245			
3	Loose lap gauge	224			
4	Loose insulation	180			
5	Leaking water pipes	140			
6	Broken windings	128			
7	Inconsistence temperature	126			

Table 7: RPN List from welder FMECA after the improvements were implemented

Table 7 shows the RPN rating for potential failure modes of the welder after the proposed improvements have been implemented. The findings indicates that the RPN for lap gauge, skew roller and weld wheel potential failure modes have declined from 560 to 224, 245 and 384 respectively after corrective or prevention measure was taken and the average risk for equipment failure has dropped. Therefore, one can conclude that there has been a notable improvement on the reliability of the welder.





Figure 4 shows the RPN rating for potential failure modes of the seamer after the proposed improvements have been implemented. This finding rate all the RPN as based on it criticality on RPN calculations and that the RPN for pressure accumulator potential failure mode has declined from 360 to 90 after corrective or prevention measure was taken and the average risk for equipment failure has dropped. Therefore, there is an improvement on the welder and seamer as shown in Table 8.



Equipment	MTBF (hour)	Failure Rate	Reliability
Welder	120	0.0083	0.0183
Flanger	320	0.0031	0.2231
Swedger	320	0.0031	0.2231
Corrugator	160	0.0063	0.0498
Seamer	150	0.0067	0.0408
Leak tester	480	0.0021	0.3685

Table 8: MTBF per equipment after improvements were implemented

Table 8 shows the reliability per equipment after the FMECA has been applied. The findings indicates that the reliability of the welder and seamer have improved after FMECA approach and implementation of planned maintenance on the critical equipment. However, there is still more room for continuous improvement, and FMECA is a good starting point for implementation of Reliability Centered Maintenance (RCM) and company-wide Total Productive Maintenance (TPM).

5 CONCLUSION

The essence of the paper was to explicit show that substantial improvement in the reliability of a drum production line can be achieved by using FMECA approach. Potential equipment failure modes, effects of the failure modes, as well as criticality of the effects were identified for the drum assembly line. The areas of improvement were identified and corrective and prevention measures were implemented resulting in the improvement of the reliability of welder and seamer. FMECA assisted the company to identify the potential failure modes and effects and identify the areas of improvement. After the risk analysis has conducted based on RPN the company managed to put in place corrective and prevention measures to prevent failures which improved the reliability of drum assembly line. The benefits derived from implementation of FMECA were reduction in the number of breakdowns as the MTBF of welder increased from 8 hours to 120 hours after the machine was put on preventive maintenance plan and MTBF of seamer has improved from 80 hours to 150 hours. Concurrently, the reliability of these machines improved, which improved the overall availability of the whole production line. It was also noted that most of the machines for the drum production line were not on the preventive maintenance plan and preventive maintenance should also be extended to these machines equipment or equipment components. It is recommended that the drum manufacturer commits itself to implementation of full scale RCM and TPM. On overall, FMECA can be the vital tool in the organisation and companies could adopt the same approach to optimise their performance.

REFERENCES

- [1] Blanchard, B.S. 2004. Logistics engineering and management. Prentice Hall.
- [2] Chen, X. 2012. Impact Of Business Intelligence and IT Infrastructure Flexibility on Competitive Advantage: An Organizational Agility Perspective.
- [3] **Bouti, A. and Kadi, D.A.1994.** A state-of-the-art review of FMEA/FMECA. International Journal of Reliability, Quality and Safety Engineering, 1(04), pp 515-543.
- [4] **Catelani, M., Ciani, L., Cristaldi, L., Faifer, M. and Lazzaroni, M. 2013.** Electrical performances optimization of Photovoltaic Modules with FMECA approach. *Measurement*, *46*(10), pp 3898-3909.
- [5] Lipol, L.S. and Haq, J. 2011. Risk analysis method: FMEA/FMECA in the organizations. International Journal of Basic & Applied Sciences IJBAS-IJENS, 11(05), pp.74-82.
- [6] Catelani, M., Ciani, L. and Paolilli, E.S. 2013. Reliability and availability analysis of an automatic highway toll collection system. *Instrumentation and Measurement Technology Conference (I2MTC)*, 2013 IEEE International, pp. 1594-1598.
- [7] Jardine, A.K. and Tsang, A.H. 2013. Maintenance, replacement, and reliability: theory and applications. CRC press.



- [8] **Vuza, S.S. 2012.** A systematic approach in product development of industrial processing equipment (Doctoral dissertation, University of Johannesburg).
- [9] **Pistofidis, P., Emmanouilidis, C., Papadopoulos, A. and Botsaris, P.N. 2014.** Modeling the Semantics of Failure Context as a means to offer Context-Adaptive Maintenance Support. *Second European Conference of the Prognostics and Health Management Society, PHME*, pp. 8-10.
- [10] **Carazas, F. and Souza, G. 2010.** Risk-based decision making method for maintenance policy selection of thermal power plant equipment. *Energy*, 35 (2): 964-975.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



UNDERSTANDING THE REAL COST OF POOR QUALITY: CASE STUDY AT BOSCH SOUTH AFRICA (AUTOMOTIVE INDUSTRY)

Tshepo Mpshe¹, Dr. Grace Kanakana², Prof. J. Trimple³

¹ Tshwane University of Technology <u>tshepo.mpshe@icloud.com</u>

²Tshwane University of Technology kanakanamg@tut.ac.za

³Tshwane University of Technology <u>TrimbleJA@tut.ac.za</u>

ABSTRACT

No company wants to lose money on what could have been prevented and there is no better place to highlight this than in the cost of poor quality. The automotive industry continues to see an increase in the cost of poor quality, costs it cannot justify nor sustain. This paper seeks to investigate the real costs of poor quality by using a case study at Bosch in South Africa, limiting it to one particular incident in November 2015 affecting numerous customers. It further seeks to highlight the importance of balancing productivity against quality in an industry that has a huge growth potential that is threatened by financial losses.

All spheres of management and all employees within organizations are affected by the cost of poor quality, but the worst inconvenienced are the customers who in turn choose to publish such inconveniences to the general public, thus reducing sales. A reduction in sales and loss of customers is perhaps the biggest nightmare the executives have to battle with as they attempt to bring escalating costs down to acceptable levels.

¹ Tshwane University of Technology, tshepo.mpshe@icloud.com

² Tshwane University of Technology, kanakanamg@tut.ac.za

³ Tshwane University of Technology, TrimbleJA@tut.ac.za



1. INTRODUCTION

The 2015 Takata airbag recall that is said to have affected some 34 million vehicles is perhaps the largest the automotive industry has seen to date [1]. It can be argued that the crisis has cost the company (Takata) far more than what it would have budgetted for as quality expenses, as per norm in the industry. Similar recalls have been seen in the recent past, e.g. Toyota's accelerator pedal recall of 2014 (4.4 Mio), Ford's parking gear recall of 2014 (21 Mio), General Motors'ignition switch recall of 2014 (5,8 Mio). Although the real costs of these recalls are kept a close secret by suppliers and manufacturers, it is a known fact that they are exhubitant and all stakeholders in the automotive manufacturing industry should strive to minimise and eliminate such unjustifiable costs.

Board of Directors and Executive Managers in the industry would be concerned about the escalating costs of sales, especially those that could have been prevented, e.g. overtime costs due to internal inefficiencies, and internal/external quality costs - the focus of this paper being on quality costs. These costs are eroding profit margins and denying investors and interested parties growth potential and quicker returns-on-investments. Managers across all levels in both supplier and original equipment manufacturers would be interested in understanding the nature and extend of these costs that are both avoidable and preventable. It is expected that in the application of "management by objectives," all managers will in a near future begin to be given quality costs reduction targets due to current unsustainable levels.

Arguably, the worst hit in the supply chain are the Tier 1 and lower tiered suppliers, as they often do not have the financial capacity to handle a costly automotive recall. The cost of replacing or repairing one component in a recalled vehicle can often amount to a thousand times what the original supplied product cost as would be seen in the case study herein. Data available seem to suggest that the bulk of recalls in the past five years till end of 2015 have been supplier related. If this is going to be a trend going forward, original manufacturers need to be worried as reputation is at stake and it means potential loss of sales and profits. Pressure is therefore on the suppliers to understand costs of poor quality and minimise them to negligible levels.

This paper uses a case study at Bosch South Africa on an incident from November 2015 affecting two of main customers (names withheld for customer protection).

2. RESEARCH OBJECTIVES, DESIGN, AND METHODOLOGY

It is the primary objective of this paper to investigate the real costs of poor quality in the automotive industry by using a case study at Bosch plant in Brits. This primary objective is supported by the following secondary objectives:

- Research and document cost of poor quality in general
- Identify causes of poor quality in the automotive industry
- Research and document classification of quality costs
- Research and document accounting methods for poor quality (defect) costs
- Recommend methods of improving classification and financial accounting of poor quality (defect) costs.

A number of empirical study methods are used by various researchers (Sila, 2006; Ugboro, 2000) [9-10] to solve practical problems; in this this paper, an "action research" methodology is employed. Cooper & Schindler (2011) [13] define an action research as "a study carried out in the course of an occupation to investigate an identified challenge and to improve methods and approaches of those involved." The challenge identified by researcher in this paper is the incorrect quantification and understanding of the real cost of poor quality.

This paper, in agreement with Cooper & Schindler [13], seeks to improve understanding of the problem, and improve cost calculation and classification methods.

The research further takes the positivism research philosophy, i.e. a belief already exists that a phenomena exists independently of a researcher and can be detected through direct observation. Sekaran & Bougie [14] defines the approach as follows, "The positivist or mainstream approach evidences the way to achieve the truth, believing that it is always possible to predict that world."

Data was sourced from an automotive manufacturing concern in South Africa. The company is a Tier-1 supplier to the seven Original Equipment Manufacturers (OEM) in the country and has export contracts. Period of analysis is a 30 days period from 15 November to 15 December 2015; this period is significant because the chosen quality incident falls within it. Except for this particular incident falling within the period, it has also



been a period in which a lot of customer complaints are received and it is, therefore, not surprising that a major quality problem from this plant occurred during the period.

3. LITERATURE REVIEW

3.1 Cost of poor quality

Besterfield [2] devoted a significant portion of his work to expand on available literature in defining the cost of poor quality as,

"... Those costs associated with the non-achievement of product or service quality as defined by the requirements established by the organization and its contracts with customers and society. Simply stated, it is the cost of poor products or services."

His view, supported by Foster [3] and Oakland [4], is that quality assurance or control should contribute to the bottom-line and that minimizing cost of poor quality will lead to an increase in organizational profits. In today's highly competitive and cost-driven business environment, management of any institution will have an interest in this simple method of improving productivity.

From a "Cost and Management Accounting" perspective, Niemand et al. state that the cost of poor quality are costs incurred due to the existence of poor quality [5].

3.2 Classification of quality failure costs

Available literature makes a distinction between two cost categories for quality, i.e. cost of conformance (COC) and cost of non-conformance (CON). Foster defines the COC as expenses incurred by an organization to assure quality, and CONC as expenses incurred due to poor quality [3]. Because organizations compete on costs and investors are interested in quicker returns, both cost categories need to be monitored to ensure a balance. This sentiment is backed up by Oakland when he states,

"The costs of quality are no different from any other costs. Like the costs of maintenance, design, sales, production/operations, and other activities, they can be budgeted, measured, and analysed" [4].

A popular cost model within the Quality Management Systems (QMS) field is known as the PAF model, an acronym for Prevention, Appraisal, and Failure. This model was developed by Joseph Juran (1951) and Arman Feigenbaum (1956) with an intention to categorize quality costs into three broad categories, which were further broken down into sub-categories according to the following tables [1-3].

Table	Table 1: Prevention costs						
Туре	Explanation						
Product/service requirements	Determination of requirements and the setting of corresponding specifications for incoming materials, processes, intermediates, finished products, and services.						
Quality planning	Creation of quality, reliability, and operational, production, supervision, process control, inspection, and other special plans.						
Quality assurance	Creation and maintenance of the quality system						
Inspection equipment	Design, development, and/or purchase of equipment for use in inspection work.						
Quality training	Development, preparation, and maintenance of training programs for operators, supervisors, staff, and mangers to both achieve and maintain canability						
	to both demete and maintain capability.						



Table 2: Appraisal costs						
Туре	Explanation					
Verification	Checking of incoming material, process-set-up, first- offs, running processes, intermediates and final products, and testing against agreed specifications.					
Quality audits	Costs incurred during checks to determine if the QMS is functioning satisfactorily.					
Inspection equipment	Costs of calibrating and maintaining equipment used in all inspection activities					
Vendor rating	Costs of assessing and approving of all suppliers, of both products and services.					

Table 3: Failure costs

Туре	Explanation				
Internal failures	Costs incurred when the results of work fail to reach design quality standards and are detected before products are transferred to the customers.				
External failures	Costs incurred when the results of work fail to reach design quality standards but are only detected <i>after</i> products have been transferred to the customers.				

Of the three cost categories according to the PAF model, prevention costs are deemed to be the most difficult to quantify due to their subjective nature [3]. Clear guidelines on defining or classifying quality costs have been attempted, but this has not taken away subjective nature of some of the types defined in literature [6].

Oakland developed a poetic definition of the three cost categories [4],

- Prevention costs:
- costs of doing it right the first time. Appraisal costs: costs of checking if it is right.
- -Failure costs:
- costs of getting it wrong.

Popular with the PAF model is the cost improvement model used to depict the relationship between quality cost and awareness levels within an organization [6]. This pictorial view is given in Figure 1.



Although the model is presented as a continuous improvement model, there can never a point in time where total quality costs are equal to zero. In order to keep failure costs at a minimum, some appraisal and prevention costs have to be incurred. In the automotive industry, requirements on ISO/TS16949 makes it



impossible to keep total quality costs at zero, e.g. quality planning, audits, inspections, calibrations, and continual improvements are mandatory requirements for which resources have to be made available [7].

An important message to be deduced from a model in Figure 1 is that as quality awareness & improvement are increased, the total cost of quality decreases. The model suggests that to improve profit margins through quality as a strategy, quality awareness and improvement should be increased. Using the TQM philosophy, all levels of employees should be covered in the quality awareness and improvement initiatives [4].

Alternative to the PAF, a process cost model was developed and published as a standard by the British Standards Institution [8]. The standards calls for a move away from traditional means of reporting on quality defects (rejection and defective materials reports) to a presentation of quality reports in financial terms. This paper will argue for this assertion but go beyond and call for correct classification/calculation of quality related costs.

In expanding on the Taguchi quality loss function, Teli and Majali [15] outlines the following key benefits:

- We cannot reduce costs without affecting quality
- We can improve quality without increasing costs, and
- We can reduce cost by reducing variations; when we do so, performance and quality will automatically improve.

Taguchi makes a significant contribution towards quality literature by stating an implied relationship between quality and costs, and proposing simple means by which quality and costs can be improved within organizations.

4. THE PROBLEM

4.1 Background

In most assembly processes, various subcomponents are joined together by a variety of methods, e.g. welding, glue application, riveting, and screwing. The joining process is an important product characteristic that requires proper process controls. In the automotive industry, various methods are applied to document and control the process parameters that have a bearing on the joining product characteristic. A Failure Mode Effect Analysis (FMEA) is used as an analytical tool to assess the impact of failures and potential failures on the functioning of a product and quantify these in a risk form using a Risk Priority Number or equivalent rating criteria [12]. Figure 4.1 is an example of an applied FMEA method.

Process			Sin Yon		e	B () ()	Curr	ent	Process					Action	Resu	lts		
Step Function	Requirement	Potential Failure Mode	Potential Effect(s) of Failure	Severity	Classificatio	Cause(s) of Failure	Controls Prevention	Occurrence	Controls Detection	Detection	RPN	Recommended Action	Responsibility & Target Completion Date	Actions Taken Completion Date	Severity	Occurrence	Detection	RPN
Op 70: Manual application of wax inside door panel	Cover inner door, lower surfaces with wax to specification thickness	Insufficient wax coverage over specified surface	Allows integrity breach of inner door panel	7	S. S	Manually inserted spray head not inserted far enough	None	8	Variables check for film thickness Visual check	5	280	Add positive depth stop to sprayer	Mfg Engineering by 0X 10 15	Stop added, sprayer checked online	7	2	5	70
			Corroded Interior lower door panels Deteriorated life of door leading to: -Unsatisfactory		ò	When the t	Pol toj	0	for coverage.			Automate spraying	Mfg Engineering by 0X 12 15	Rejected due to complexity of different doors on the same line				
			appearance due to rust through paint over time. • Impaired function of Interfor door hardware			Spray head clogged - Viscosity too high - Temperature too low - Pressure too low	Test spray at start-up and after Idle periods and preventative maintenance program to clean heads	5	Variables check for film thickness Visual check for coverage.	5	175	Use Design of experiments (DOE) on viscosity vs. temperature vs. pressure	Mfg Engineering by OX 10 01	Temp and Press Limits were determined and limit controis have been installed- Controi charts show process is in controi Cpk=1.85	7	1	5	35
						Spray head deformed due to Impact	Preventative maintenance programs to maintain heads	2	Variables check for film thickness Visual check for coverage.	5	70	None						

Figure 4.1: FMEA Example (Source: AIAG FMEA Manual, 4th Edition).

FMEA, after identifying what is critical and important, becomes an input for the Control Plan which "provides a structured approach for the design, selection, and implementation of value-added control methods for the total system" (AIAG, 2008) [13]. Figure 4.2 shows an example of a Control Plan of a set-up dominant process from the AIAG manual.



Destetue		-in 12	Deaduret	tion			INTROL PLAN					Door 4 of 22
Control Bion												Page 4 or 25
Consorran	NUMBER	001	1/26	2008	Date (New.)	12008						
Part Number	n atest Change Level			Core Team	0.1904101010-02	0.0000			Customer Engl	neering Approx	a/Date (If Reg'd.)	12000
David bilances (22521211/G 11-2-92			Conservation (D)	Product Develop	oment Tean	1 (E01) - See List		Customer Cust	in Annual/De	to (if Paneld)	
Part Namer	rescription			Olganization	ant Approvaribate	1			Customer Quar	ity Approvavoa	ie (ii Redu.)	
	Plastic Injection Molda	d Grill										
Organization	/Plant	Organization Co	de	Other Approval/	Date (If Reg'd.)				Other Approval	/Date (If Reg'd.)	
4-B (Srill Co. Plant #3	0123										
		MACHINE,		CHARACTER	etice				ETHODE			
PART/	PROCESS NAME/	DEVICE,		CHARACTER	131103	SPECIAL		M	ETHODS			
PROCESS	OPERATION	JIG,TOOLS		1		CHAR.	PRODUCT/PROCESS	EVALUATION/	SAM	IPLE		REACTION
NUMBER	DESCRIPTION	FOR MEG.	NO.	PRODUCT	PROCESS	CLASS	SPECIFICATION/	MEASUREMENT	SIZE	FREQ.	CONTROL	PLAN
							TOLERANCE	TECHNIQUE			METHOD	
3	Plastic Injection Molding	Machine No. 1-5	18	Appearance		*	Free of blemishes	Visual Inspection	100%	Continuous	100% inspection	Notify Inspector
	NUMBER	140.152		No hlemishes			flowlines	I st piece buy-off			Check Sheet	Adjust/re-check
							sink marks	1 st piece buy-off			Check Sheet	Adjust/re-check
		Machine		Mo atting hole						hav-off		
		No. 1-5	19	loc.		*	Hole "X" location	Fixture #10	1st piece	per run	Check Sheet	Adjustre-check
							25±1mm		5 pcs	he	_x-R chart	Quarantine and adjust
		Machine No. 1-5	20	Dimension		٥	Gap 3 ±.5mm	Fixture #10	1st piece	buy-off per run	Check Sheet	Adjust and re-check
		Fixture #10	21	Perimeter fit		*	Gap 3 ±.5mm	Check gap to fixture 4 locations	5 pes	hr	x-R chart	Quarantine and adjust
		Machine No. 1-5	22		Set-up of mold machine		See attached set-up card	Review of set-up card and machine settings		Each set-up)st piece buy-off	Adjust and re-check
											Inspector verifies settings	

Figure 4.2: Control Plan Example (Source: AIAG APQP/CP, 2nd Edition).

These two methods are herein referenced and explained because of their role in preventing failures within the automotive industry. When correctly applied as quality tools for identifying and limiting process failures, FMEA and Control Plan are considered to be at the forefront of a preventive approach in Quality Management Systems in accordance with ISO/TS16949 [7]

4.2 Torque Process - The Failure

This empirical research work focuses on the failure of a torque process on an alternator assembly during a 15 November - 15 December 2015. In an alternator assembly, herein referred to as an "alternator", between eight and ten torque processes are employed to join various components together according to defined specifications. Adherence to these specifications is an important process characteristic as it is desired that none of the components torqued together gets loose during operation.

For an optimal operation of an alternator, it needs to be appreciated that if functions within a vibrating environment and potential loosening of screws and nuts becomes a possibility to be dealt with by observing torque specifications.

Figure 4.3 depicts the products in question with an exploded view to create an impression of some of the subcomponents that requires joining together during an assembly process.



Figure 4.3: Alternator, with exploded view (Source: Bosch Automotive Parts)

In the assembly process, four throughbolts are used to join together the Drive-End and Collector-Ring End Shields. A simalteneous torque is applied to the four bolts by an automated torque driver (SE330 type) according to specifications. Figure 4.4 shows an overview of these four through-bolts.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 4.4: Alternator Assembly, with through-bolts

A required torque specification for each of the bolts is 4.00 to 4.6Nm according to design and manufacturing drawings. Durign the period under review, an average applied torque was 1.15Nm, almost a quarter of the specified torque.

4.3 What caused the failure?

Various tools are used in the automotive and other industries to find potential and real-causes of failures, most popular being the following: Ishikawa, 8-Dimensions (8D), 5 x Why Drill-Deep, IS/IS NOT.

An Ishikawa, commonly known as the fish-bone diagram is a good starting point in problem solving where a team can brainstorm around a problem. Potential causes of a problem (fish head) are identified and later tested within one or more of other techniques. For the loose through-bolts, identified potential causes are listed in figure 4.5.

One of the objectives of this paper is to highlight general causes of quality failures in the automotive industry and the example used herein reveals a number of these, e.g.

- Standard operating procedures not adhered to,
- Poor incoming quality (materials),
- Lack of management intervention,
- Ineffective quality checks,
- Poor maintenance,
- Incorrect use of equipment,
- Incorrect machine setting, and
- Ineffective training methods.

The list of possible contributors to poor quality is endless; the above examples are but the most common that come up in problem-solving efforts. Some quality defects are as a result of poor product/process design - they generally require a lot of in-process checks which increase the appraisal costs, otherwise the failure costs are in the increase.





Figure 4.5: Ishikawa - Loose Alternator Through-bolts

Potential failures are categorised according to the 6M classification method, i.e. Man, Method, Mother Nature (Environment), Method, Material, Management. Items highlighted in red were chosen by a team to be further analysed and interrogated.

From a brainstorming action using an Ishikawa, a drill-deep analysis was carried out using a 5 x Why technique, where a questioning technique is repeated until an answer leads to a real root-cause. The application of this technique seeks to ensure that symptoms are separated from causes and thus lead to permanent eradication of failures by implementing corrective actions that address causes and not symptoms. Correct usage of this technique, therefore, has huge benefits to the industry and any other sector which wants to do a proper problem solving.

For the problem under analysis, figure 4.6 details a $5 \times$ Why technique employed by a problem solving team. With this analysis, a team concluded that it was possible for the error to happen due to a lack of a system to prompt Maintenance Technician to ensure all interface plugs were replaced after maintenance activities were carried out on a torque driver. Same technique can be split into three sections, i.e.

- 1. Why did the failure occur?
- 2. Why was the failure not detected?
- 3. Why was the failure not prevented?

Figure 4.6 addresses the first of these 3 questions. On why was it not detected, analysis revealed that the torque verification process used by Product Auditors was ineffective. This is on the "management section" on Ishikawa.

Problem	Problem: Loose alternator through-boits					
Why 1:	Why were 4 alternator through-bolts torqued below minimum specification?					
Ans 1:	Torque driver operated with incorrect program					
Why 2:	Why was the torque driver operated with an incorrect program?					
Ans 2:	Interface bits for torque program selection missing (defaulted to program zero)					
Why 3:	Why was the interface bits for program selection missing?					
Ans 3:	Interface plug was removed and not replaced after maintenance activities					
Why 4:	Why was an interface plug removed and not replaced after maintenance activities?					
Ans 4:	Maintenance Technician forgot to replace interface plug					
Why 5:	Why did the Maintenance Technician forget to replace interface plug?					
Ans 5:	No system in place to remind Technician to replace interface plugs after maintenance					

Figure 4.6: 5 x Why Analysis - Loose Alternator Through-bolts



The impact of a failure in the company's maintenance and product audit functions caused a huge financial loss. This loss is a focus of this research paper. Products that were already delivered to products had to be contained, i.e. re-torqued manually to a defined specification. Some of the alternators were already assembled onto engines and these also required special actions as a risk of a completely loose alternator became evident.

Two customers were affected, names of which are not revealed to protect them. "Customer A" and "Customer B" shall be used as code names for the affected customers.

4.4 What was done?

Customer A required an external sorting/rework team. The amount of parts that had to be reworked was 300 and no parts were fitted onto engines/vehicles.

Customer B was severely affected and required a team to be despatched to its manufacturing location for sorting and rework actions. This was carried out in a period of 14 days with a team of 13 persons with skills levels including Technicians, Engineers, and Managers. A total of 554 alternators had to be checked and reworked, i.e. re-torqued to required specification in the process represented by flowchart 1.

A flowchart is herein added to indicate the effort it took to rework all 554 alternators, a task requiring approximately 13 minutes per part, thus equalling 120 hours of active rework time. What the flowchart is not indicating are the following activities:

- Daily security checks for team members before and after rework activities,
- Driving and walking to various locations (manufacturing areas, warehouses, and ship yards) to carry out rework activities, and
- Activities to locate defective products through the logistics chain.

There was a total of 110 parts that could not be contained as they were already in a ship for export and a risk assessment was carried out and the risk was found to be minimal. It was argued that in the event any or all of the through-bolts came loose in the engine compartment, the bolts holding an alternator to an engine block would be sufficient to hold parts together and prevent disintegration.



Flowchart 1: Alternator rework procedure.

5. RESULTS AND DISCUSSION

For any organization to spend resources on fixing a quality problem that is both preventable and avoidable, should be a source of concern. Although the rework activity carried out in the case study employed in this



paper prevented a recall activity and reduced risks to the customers, it left the manufacturer with a high external failure costs. An indication of costs that had to be accounted for is given below.

- 1. Travelling costs for 13 team members
- 2. Accommodation costs for 13 team members
- 3. Daily allowances for lunch incidental expenses
- 4. Labour costs for 13 members over 14 days

Although the internal directive of the company makes provision and requires all costs related to a rework activity to be accounted for, the available accounting tool excludes most of the costs identified above and focuses on the following:

- 1. Piece price, i.e. sales price per component if parts are returned, and
- 2. External (third) party labour costs.

In cases where failed parts are returned from customers and reworked in the plant, the labour and additional materials employed to carry out the activity are not accounted for. With this practice, it becomes impossible for real costs of quality to be understood and it results in a better projection of quality because of miscalculations. These miscalculations prevent management from intervening and curbing escalating quality costs. The weakness herein is a methodical failure since the available accounting policies make a distinction and correct categorization of different quality costs, but the available collected data used as an input for costing does not differentiate costs correctly.

From the popular model for depicting an optimum balance between prevention and failure costs (figure 4.7), it goes without a discussion that no organization will be able to achieve and maintain desired quality levels without investing in prevention costs. If the failure costs are increasing and the prevention costs remain the same, an organization will find itself in a difficult financial situation where the costs of manufacturing are increased and these cannot be recovered from the customers.



Figure 4.7: Optimum Quality Level - Cost of quality.

In management reports, the only distinction made is between internal and external failure costs, thus leaving out the other two costs classification of prevention and appraisal costs. In the case example for a given accounting period, a spike in the external failure costs as a result of the 14 days external rework using high skilled workers would have been observed. In the same case, there was an effort to inspect materials already produced (appraisal costs) but not delivered to the customers. In addition, there were process improvements initiatives (prevention costs) that had to be implemented in order to prevent the same defect occurring again; Machine Operators and other support staff had to be re-trained on the improved processes.

By making mention of a few examples of unaccounted quality costs in this case study, it is apparent that in any manufacturing plant, there will be many similar cases and a summation of these hidden costs has to be a source of concern for management.

In the example used, critical aspect such as waiting times due to increased inspection efforts whenever defective products are received from suppliers (internal and external), are not addressed.

Forster [3] correctly puts the challenge facing Quality and Accounting Professionals by stating,

"One of the impediments to the collection of quality cost data has been the lack of acceptable accounting standards for these costs. For example, the standard accounting



definition for quality is "meeting specifications." This narrow definition limits organizations desiring to quantify and measure customer requirements as a means of improving service to the customer. A reason for this is that accounting rules require definitions that are not open-ended or open to alternative interpretations".

Quality managers have to take note of the limitations brought by the accounting principles and realise that until viable accounting models are designed, a skewed picture of real quality costs will continue to be projected, and such a picture does not support quality improvements due to the ice-berg effect where majority of costs are both hidden and unaccounted for.

6. CONCLUSION

The case presented in this paper is one of many incidents that occur in the manufacturing industries and in automotive industry in particular, research shows an increase in this occurrences [15]. The absence of a good accounting model to capture real costs of poor quality makes it difficult to quantify the extent to which quality contributes to business objectives. It, therefore, has to be understood that if organizations do not make an effort to quantify the true cost of poor quality, jobs and growth are at risk as return on investments are threatened by escalating costs.

Managers and specialists tasked with reduction of quality costs and continuous improvements have a duty of care in the methods used to collect costs related to costs of poor quality. Stakeholders have to be aware that when costs of poor quality is not correctly accounted for, cost management within a manufacturing entity becomes too difficult and budgeting and provision of resources cannot be effectively carried out.

Organizations have to be aware of escalating costs of internal and external failure costs and make some concerted efforts to reducing these by increasing on preventions and appraisal costs. The P-A-F paradigm prevents a simplified model of categorising quality costs and can be a good starting point in making sure that almost all costs related to making sure customers are satisfied, are classified and accounted for.

The British Standard model [6] for reduction of quality costs through quality awareness and improvement initiatives can only be effective when such quality costs are adequately classified and accounted for.

REFERENCES

- [1] United Nations Senate, 2015. Danger behind the wheel: The Takata Airbag Crisis And How To Fix Our Broken Recall Process, Committee on Science, Commerce, and Transportation.
- [2] Besterfield, D. H. 2004. *Quality Control*. 7th Edition, Pearson International.
- [3] Foster, S. T. 2007. Managing Quality Integrating the Supply Chain. 3rd Edition, Pearson International.
- [4] Oakland, J. S. 2003. Total Quality Management. 3rd Edition, Butterworth-Heinemann.
- [5] Niemand, A. A., Meyer, L., Van Vuuren, S.J. 2007. Fundamentals of Cost and Management Accounting. 5th Edition, LexisNexis Butterworth.
- [6] British Standard. 1992. BS6134-1: Guide to the economics of quality Process cost model. British Standards Institution.
- [7] International Standards Organization. 2009. ISO/TS16949: Quality Management Systems Particular requirements for the application of ISO 9001:2008 for automotive production and relevant service part organizations. International Standards Organization.
- [8] British Standard, 1992. BS6143-2: Guide to the economics of quality Prevention, apparaisal and failure model. British Standards Institution.
- [9] Sila, I. 2006. Examining the effects of contextual factors on TQM and performance through the lens of organizational theories: An empirical study. *Journal of Operations Management*, 25, pp 83 109.
- [10] Ugboro, I.O., Obeng, K. 2000. Top management leadership, employee empowerment, job satisfaction, and customer satisfaction in TQM organizations: an empirical study. *Journal of Quality Management*, 5, pp. 247 272.
- [11] Automotive Industry Action Group. 2008. Potential Failure Mode and Effect Analysis. AIAG
- [12] Automotive Industry Action Group. 2008. Advanced Product Quality Planning (APQP) and Control Plan. AIAG
- [13] Cooper, D.R., Schindler, P.S. 2011. Business Research Methods, 11th Edition Mc Graw Hill International.
- [14] Lopes, I.T., 2015. Research methods and methodology towards knowledge creation in accounting. Contaduría y Administración, 60, pp.9-30



[15] Teli, S.N., Majali, V.S., Bhushi, U.M., Surange, V.G. 2014. Impact of poor quality in automobile industry. International Journal of Quality Engineering and Technology, 5, pp. 21 - 41.



EVALUATION OF WASTE-TO-ENERGY GRATE INCINERATION POWER PLANT DRIVERS AND BARRIERS FOR A SMALL SOUTH AFRICAN CITY: A SWOT ANALYSIS APPROACH

W. Maisiri^{1*} & L. van Dyk²

¹Department of Industrial Engineering North-West University Potchefstroom campus, South Africa wmlisper27@gmail.com

²Department of Industrial Engineering North-West University Potchefstroom campus, South Africa liezl.vandyk@nwu.ac.za

ABSTRACT

The global share of energy from waste is approximately 0.4% of global energy consumption. The average amount of municipal solid waste generated in South Africa is equivalent to 11 GWh wasted energy. This paper focuses on the drivers and barriers to the establishment of a waste-to-energy (WtE) grate incineration power plant for a small South African city in the North-West Province. It employs strength, weakness, opportunity and threats (SWOT) analysis to analyse the drivers and barriers to implementation of such a plant. Strengths and opportunities were acknowledged as drivers to the establishment of such a plant and weaknesses and threats as barriers. A holistic investigation using a SWOT analysis showed significantly more drivers than barriers. The study proved that SWOT analysis can be used as both a preliminary technology selection tool and an investment decision-making tool.

¹ The author was enrolled for an M Eng (Mechanical) degree in the Department of Mechanical and Nuclear Engineering, North-West University, Potchefstroom Campus.

² The author is the Director of Research and Industrial Engineering Program Manager at North-West University, Potchefstroom Campus.



1. INTRODUCTION

South Africa is currently facing power shortages due to increase in demand, failure to invest in additional capacity and limited maintenance. Simultaneously, over the years, municipal solid waste (MSW) generation and management problems have escalated [1, 2] in response to constant population growth and expansion of urbanization.

A significant amount of energy is wasted in South African landfill, as over 90% of MSW is disposed of in landfill and dumped illegally without energy recovery [2]. Landfill poses a number of environmental problems, such as greenhouse gases (methane and carbon dioxide) emissions, leakages into groundwater and inefficient space utilization [3].

The global trend is that the focus in waste management strategies has shifted to harmonize with the goal of sustainable development through electrical power generation and minimization of the adverse effects of landfill [4, 5]. MSW can now be viewed as a useful resource and can be used as an alternative renewable energy source.

Waste-to-energy (WtE) thermal technologies have the potential to meet the goal of sustainable waste management [6]. Developed countries that have implemented WtE thermal technologies have boosted their recycling rates, minimized the adverse impacts of landfill and increased the renewable energy generation level [4, 7].

South African cities are faced with power shortages, high unemployment rates and MSW generation and management problems, with landfill and illegal dumping dominating waste management strategies. WtE thermal technologies can be used to harness energy wasted in the landfill of South African cities, at the same time creating employment.

Maisiri [8, 9], performed technological and performance evaluation of four WtE thermal technologies and financial analysis of WtE grate incineration power plant. This paper is a continuation of this work and its objective is to carry out a SWOT analysis for a WtE grate incineration power plant for a small South African city in the North-West Province.

The paper is organized as follows: firstly, an overview on WtE grate incineration power plant is presented in section 2. Small city WtE grate incineration power plant SWOT analysis (Section 3) is discussed next. Small city internal drivers and barriers to WtE grate incineration power plant is deliberated in section 4. This is followed by a discussion on external drivers and barrires to WtE grate incineration power plant in section 5. A discussion on SWOT analysis results (Section 6) and conclusion (Section 7) are presented last.

2. WASTE-TO-ENERGY GRATE INCINERATION OVERVIEW

WtE grate incineration technology, commonly know as mass burn, is a conventional technology used to thermally treat MSW and other waste streams in a suststainble manner. Waste is treated with the objective of recovering energy inform of heat and electricty [10, 11, 12].

Oualitative and quantitative analysis evaluations carried by Maisiri [8, 9] proved that WtE grate incineration is the dominant and commercialy accepted technology over other WtE thermal technologies. The investigation further indicates that grate incineration remains the preferred and most economical thermal technology in view of its reliability, capability to incinerate assorted waste, lower operational complexity and higher power efficiency [13].

WtE grate incineration has been in use for more than 130 years and more that 90% of European WtE plants use grate incineration, and there are more than 1000 operational WtE plant installations across the world [9,14]. Figure 1 shows WtE grate incineration power plant process flow diagram. Selected existing WtE grate incineration power plants are presented in Table 1.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 1: WtE grate incineration process flow diagram [14].

Plant	Country	Discription	Capacity
			(Ton/year)
AEB,	The Netherlands	The largest plant in the Netherlands.	1,370,000
Amsterdam		Produces electricity with a total	
		thermal efficiency of 30%.	
Lakeside,	UK	A recently commissioned merchant	410,000
London		incinerator developed by a major UK	
		waste management company. The	
		plant processes residual MSW and	
		other waste	
Spittelau,	Austria	Relatively old conventional moving	250,000
Vienna, Austria		grate combustion plant. The first	
		facility that used architectural	
		treatment to gain public acceptance.	
Issyles	France	The newest and largest incineration	460,000
Moulineaux,		plant in France.	-
Paris			
Reno Nord,	Denmark	Modern incinerator in CHP mode and	160000
Aalborg		providing district heating to the local	(line 4)
		city.	

3. WASTE-TO-ENERGY GRATE INCINERATION SWOT ANALYSIS

SWOT analysis is a tool used for analysing internal and external environments in order to ensure a systematic approach and support in making decisions. Strengths and weaknesses are regarded as internal factors, while opportunities and threats are external factors [15].

In this study, SWOT analysis is applied by regarding WtE drivers and barriers for the small city case study as internal factors, while drivers and barriers outside the small city but within South Africa are regarded as external factors. Table 2 summarizes the strengths, weaknesses, opportunities and threats associated with establishing a WtE grate incineration power plant for a small city in the North-West Province.

Table 2: South small city WtE grate incineration	on power plant SWOT analysis [13].
--	------------------------------------

Small city internal WtE investment drivers and barriers						
Strengths	Weakness					
 Average MSW generation growth rate 3.04% per annum (Section 4.1.1) MSW characterized by more than 80% of combustible material (Section 4.1.2). 	 Dominance of wet months in a year in the small city (Section 4.2.1). MSW yearly tonnages fluctuations (Section 4.2.2). 					



 Significant percentage of metals promotes material recovery (Section 4.1.2). 98.3% dominance of landfill as MSW management strategy for the small city (Section 4.1.3). Increased illegal dumping in the small city (Section 4.1.3). 	 Absence of landfill diversion measures evidenced by no cost to dump waste at the landfill (Section 4.2.3). Local culture of non-payment for service especially in the African and colored settlements (Section 5.2.4).
Small city external WtE inve	estment drivers and barriers
 Opportunities 90% dominance of landfill as MSW strategy at national level (Section 5.1.1). National waste generation growth rate estimated 3.0% per annum (Section 5.1.2). 34% of non-recyclable combustible waste in the national MSW characterization (Section 5.1.2). Non-energy recovery methods used to treat health care risk waste (HCRW) (Section 5.1.2). National power crisis, promotion of renewable energy (Section 5.1.3) Intermittent nature of solar and wind energy (Section 5.1.3). Accumulation of waste vehicle tyre (WT) stockpile nationwide (Section 5.1.4). Anticipated implementation of carbon tax and offset incentives in the country (Section 5.1.5) 	 Threats WtE thermal technologies are not allocated generating capacity in the REIPPP (Section 5.2.1). Lack of enforcement of environmental legislation in the country hampers technological innovation in the waste sector (Sections 5.2.2). High capital cost that requires access to investment subsidy from government institutions (Section 5.2.3). Local culture of non-payment for service especially in the African and colored settlements (Section 2.4).5.

The small city internal and external drivers and barriers to investment in WtE grate incineration power plant are further explained in section 4 and 5.

4. SMALL CITY INTERNAL DRIVERS AND BARRIERS TO WASTE-TO-ENERY PLANT

An MSW management survey was performed for a small municipality in the North-West Province. Data were collected through interviews with representatives from the department of waste management. The collected data were analyzed and this section presents the drivers and barriers to implementation of WtE grate incineration plant identified from data analysis results.

4.1 Internal drivers to the implementation of WtE grate incineration

4.1.1 Municipal solid waste generation trends

MSW generation is on an upward trend due to the expansion of the small city and population growth. Figure 2 shows MSW generation trends for the small municipality. The average MSW generation growth rate was calculated at 3.04% per year [13].





Figure 2: Small city MSW generation trends [9].

4.1.2 Municipal solid waste characterization

Table 3 shows MSW characterization for the small municipality investigated. MSW is characterized with more than 80% of combustible material of which plastic makes the highest

percentage. The lower heating value (LHV) of MSW is estimated at 8.0 MJ/kg. There is significant percentage of metals which can promote metal recovery [13].

Table 3: Small m	nunicipality MSW	characterization	[9].
------------------	------------------	------------------	------

Waste Type	Percentage (%)
Builders rubble	2
Plastic	30
Paper	28
Organics, garden waste	8
Metals	6
Glass	13
Other (textile, disposable nappies, tyres, residue and miscellaneous)	13
TOTAL	100

4.1.3 Municipal solid waste management strategies

Landfill was identified as the dominant MSW management strategy for the small municipality investigated. On average 98.3% of the waste generated is landfilled while 1.7% is recycled. Figure 3 shows the contribution of landfill and recycling to the overall waste management strategy of the small city. Investigations proved that there is increase in illegal dumping in the small city [13].

4.2 Internal barriers to the implementation of WtE grate incineration

4.2.1 Municipal solid waste dampness

Data analysis results showed that the small city has five dry months in a year. Figure 4 shows the rainfall patterns recorded at the landfill. Damp MSW dominates because of wet months recorded [9].









Figure 4: Monthly rainfall trends for the small city [9].

4.2.2 Municipal solid waste generation trends

Figure 2 shows variations in MSW generation trends from the year 2011 to 2014. The trends show significant fluctuations in the amount of MSW generated for the state period. This presents uncertainty in the availability of the waste stream to run the WtE grate incineration plant for the small city [13].

4.2.3 Landfill diversion measures

In developed countries landfill diversion is a major driver of WtE technologies [4]. Measures such as high fees charged on waste managed through landfill have been instituted. WtE grate incineration power plants' gate fees are relatively low compared to landfill gate fees. This leaves waste producers with no other option than using the WtE grate incineration plant as an alternative waste management strategy.

Contrary to this, investigations by Maisiri revealed that no landfill gate fee is levied by the small city in the North-West province and in South Africa at large. This makes landfill the cheapest waste management option and waste producers consequently choose this method. Local residents only pay an insignificant flat fee, embedded in their property bills, to local municipalities. Refuse collection fees vary with residential areas.



5. SMALL CITY EXTERNAL DRIVERS AND BARRIERS TO WASTE-TO-ENERGY PLANT

5.1 External drivers to implementation of waste-to-energy grate incineration

5.1.1 Municipal solid waste management trends

Investigations reveal that landfill dominates other waste management strategies in South Africa, regardless of the amendment of the waste hierarchy in 2011 [2, 16]. Landfill accounts for approximately 90.1% of waste generated in the country. It was noticed that MSW accumulation in uncontrolled dumpsites is rampant in South African cities [2].

Surveyed literature indicate that WtE thermal technologies make no contribution to the country's current MSW management strategies [17]. The current MSW management trends in South Africa offer an opportunity to implement WtE thermal technologies.

Significant waste divergence from landfill can be achieved in the country through WtE thermal technology initiatives. Figure 5 reflects the experience of countries that have implemented WtE thermal technologies. Noticeable decrease in landfill usage and boost in recycling levels can be seen [6].

WtE thermal technologies can boost recycling activities in the country considerably. Countries that have implemented WtE thermal technologies attest that WtE facilities appreciably improve material recovery. WtE facilities promotes recycling through metals recovered from air pollution control and bottom ash residue and in the feedstock preparation stage [6].



Figure 5: Percentage contribution of WtE incineration, recycling and landfill in managing MSW in EU [7].

5.1.2 Municipal solid waste generation trends.

MSW national generation was at 59 million tonnes in 2011. The national waste generation growth rate is estimated at 3.0% per annum [1]. The MSW composition is dominated by combustible non-recyclable waste with a percentage contribution of 34%. Table 4 shows South Africa's general waste composition by 2011.

In South Africa health care risk waste (HCRW) generation was estimated at 45 232 tonnes per annum in 2011. Gauteng province made the highest percentage contribution to HCRW [1, 18]. HCRW treatment facilities are not decentralized and there is no energy recovery during incineration [18, 19]. Significant illegal dumping of HCRW has been witnessed, with Gauteng recording most HCRW illegal dumping incidents [18].

Waste type	Percentage contribution (%)
Glass	4
Plastics	6
Paper	7
Tyres	1
Combustible non-recyclable MSW	34

Table 4:	General	waste	composition	[1].	
----------	---------	-------	-------------	------	--



Organic waste	13
Construction and demolition waste	21
Metals	14

5.1.3 South Africa's energy scenario

In 2014, South Africa was hit with blackouts in its major cities due to a shortfall in electrical energy generation capacity [20]. The shortfall in generation capacity was a result of increased demand, failure to invest in additional supply and limited power plant maintenance.

The renewable energy mix in South Africa comprise wind, solar energy, small-scale hydro generation, biomass and bagasse. MSW is also listed as a potential source of energy in the country. It is estimated that MSW equivalent to 11 000 GWh is wasted in South African landfill per annum [21].

The wasted energy in the landfill can be harnessed through direct incineration and by using methane gas. WtE thermal technologies best suit urban areas where a large amount of waste is produced and the costs of landfill sites are relatively high [22].

International Energy Agency statistics point out that residual MSW has the potential of replacing 2% of fossil fuel in South Africa [4], thus WtE technology can make a significant contribution to the country's renewable energy mix.

South Africa's renewable energy sector is dominated by wind and solar energy. Increasing the share of wind and solar power in the national grid might result in a long-term crisis. A large amount of intermittent renewables in the grid causes grid crush [23]. This is due to their spasmodic nature, which is unable to supply power at all times and when it is needed most.

Unlike wind and solar energy, a WtE grate incineration power plant has load flowing characteristics and has no intermittency problem. A WtE grate incineration power plant can be classified among peaking technologies and can operate 24 hours a day. The capacity factor of a WtE power plant is approximated at 90%. Implementation of WtE grate incineration technology can significantly boost renewable energy and increase national grid security in South Africa.

5.1.4 Waste vehicle tyre management

Muzenda [16] maintains that South Africa is currently faced with problems in waste vehicle tyre (WT) management. It is stated that there are over 60 million WT in South Africa. The illegal dumping and disposal of WT poses significant environmental problems. Landfill and stockpiling are the commonly used WT management strategy in South Africa [16].

The millions of WT lying in stockpiles, landfill and scattered in residential, industrial and rural areas are posing serious environmental and health problems. WT stockpiling causes significant problems such as breeding of insects and rodents, as well as risk of uncontrollable burning, which cause air pollution [16].

Including WT as a supplementary feedstock to a WtE grate incineration power plant has the potential of improving its financial model performance. This is possible through improving the feedstock LHV, thus boosting power generated.

5.1.5 Carbon tax

The Department of National Treasury states that carbon tax will be introduced to facilitate transition to a low-carbon economy. The government of South Africa is moving towards cutting green-house emissions by 34% and 42% in 2020 and 2025 respectively [24]. The move is significantly subject to the availability of sufficient financial, technological and capacity-building support by developed countries.

In order to include the cost of pollution on goods and services, carbon tax will be used as instrument that will ensure polluters, both producers and consumers, held accountable [24]. To this end, major consumers of electricity such as municipalities will incur large carbon tax. The use of alternative green energy such as WtE grate incineration can assist in offsetting such huge amount carbon tax.

WtE grate incineration reduce green-house gas emissions by approximately 13 000 tonnes carbon-dioxide equivalent for every 100 000 tonnes of MSW processed. For combined heat and power systems, green-house gas emissions are lowered by roughly 23 000 tonnes carbon-dioxide equivalent [4]. This means for a 200 000 tonnes capacity WtE grate incineration plant, carbon credits equivalent to 26 000 tonnes carbon-dioxide equivalent can be earned.



5.2 External barriers to implementation of waste-to-energy grate incineration

5.2.1 Renewable energy independent power producer procurement program allocations

Renewable energy independent power producer procurement program (REIPPP) is competitive tender procurement process, designed to promote renewable energy to match global trends and uphold the goal of sustainable development [25]. Table 5 shows summarized results of RIEPPP generation capacity allocation in bidding windows 1 to 4. WtE thermal technologies were not allocated generating capacity in the REIPPP. There has thus been no participation in WtE thermal technologies [26].

	Total allocated and remaining generation capacity (MW) in bidding windows 1 to 4					
Technology	1	2	3	4	Total allocated	Total remaining
Solar photovoltaic	632	417	435	415	1899	626
Onshore wind	634	563	787	676	2660	660
Concentrated solar power	150	50	200	0	400	0
Small hydro (< 40MW)	0	14	0	5	19	116
Landfill gas	0	0	18	0	18	7
Biomass	0	0	16	25	41	19
Biogas	0	0	0	0	0	60
WtE thermal technologies	0	0	0	0	0	0

Table 5: RIEPPP total allocated and remaining generation capacity [27].

5.2.2 Legislative framework

National environmental management in South Africa facilitated the development of regulations that support the Waste Act. These regulations are aimed at supporting innovation and divergence of waste streams from landfill [2]. However, lack of enforcement of environmental legislation has hampered technological innovation in the waste sector.

Investigations show that the authorization of waste management licenses has been hindering novelty and implementation of new technologies in the waste sector. The authorization process is expensive and unrealistically long. Reports of corruption in the system have been received [17].

Enforcement of environmental legislation will encourage different stakeholders to implement alternative technologies in the waste management sector. An effective, efficient and transparent authorization process will attract innovation in the country's waste management sector.

5.2.3 Waste-to-energy thermal technologies high capital investment

WtE thermal technologies involves high capital investment, operational and maintenance costs. In comparison with other renewable energy technologies, the capital cost of a WtE power plant makes it an unfavourable investment option. South Africa has no operating WtE thermal technologies at present. The capital-intensive nature of WtE thermal technologies calls for long-term security to investors.

The capital cost of WtE technologies in South Africa has not been well researched [19]. The capital investment figures used are taken from the experience of other countries that use WtE thermal technologies. Because of the availability and low cost of material, the estimated capital cost of WtE thermal technologies in South Africa might be significantly lower.

5.2.4 Local culture

The financial viability of a WtE grate incineration power plant in developed countries is noticeably influenced by the MSW gate fees. A negative price on waste has resulted in the success of many WtE grate incineration power plant projects. This means that waste producers must pay a fee to the plant operator for their waste to be processed [4].

The local culture of non-payment for service, mainly encountered in African and colored communities [28], promotes failure of the waste negative price model. This adversely affects the financial viability of a WtE power plant.

The local culture might necessitate a WtE plant financial model that will have a positive price on waste. This means that the WtE plant operator will have to pay waste generators for their waste to be processed. This decreases the anticipated revenue streams and may cause financial instability in a WtE grate incineration power plant. Nevertheless, that model will grantee residual waste stockpile for the plant.



6. SWOT ANALYSIS DISCUSION

SWOT analysis was applied in this study by considering small city case study drivers and barriers to the establisment of WtE grate incineration power plant in South Africa as the strenghts and weakness respectively. Drivers and barriers outside the small city but within South Africa were regarded as oppotunities and threats correspondingly.

According to Table 2, the identified strengths are more than the weakness while the opportunities acknowledged are significantly more than the threats. Thus the drivers to the implementation of a WtE grate incineration power plant are considerably more than the barrriers.

The drivers identified include an average of 90% dominance of landfill over other MSW management strategies, average MSW generation growth rate of 3% per annum, national power shortages, the intermittent nature of solar and wind energy and non-energy recovery methods used in treating health care risk waste (HCRW).

However, obstacles to execution, such as the legislative framework, lack of landfill diversion measures, high capital investment involved and local culture were acknowledged.

7. CONCLUSION

The purpose of this paper was to perform a SWOT analysis for a WtE grate incineration power plant for a small South African city. The analysis proved that the drivers to WtE grate incineration power plant implememenation are significantly more than the barriers identified. Thus SWOT analysis results proved that investment in a WtE grate incineration power plant for a small South African city is feasible. The investigations in this study proved that SWOT analysis can be used as both a preliminary technology selection tool and an investment decision-making tool.

REFERENCES

- [1] South Africa. DEA (Department of Environment Affairs). 2012. National waste information baseline report. Pretoria.
- [2] Godfrey, L., Rivers, M. & Jindal, N. 2014. A national waste R&D and innovation roadmap for South Africa: Phase 2 waste RDI roadmap. Trends in waste management and priority waste streams for the waste RDI roadmap. Pretoria: Department of Science and Technology. (CSIR/NRE/GES/ER/ 2014/0016/A)
- [3] Meisen, P. & Morgan, P.I. 2010. Waste-to-energy plants. San Diego, CA: Global Energy Network Institute.
- [4] IEA (International Energy Agency) Bioenergy. 2010. Accomplishments from IEA bioenergy task 36: integrating energy recovery into solid waste management systems (2007-2009). UK: IEA.
- [5] Sethi, S., Kothiyal, N., Nema, A.K. & Kaushik, M. 2012. Characterization of municipal solid waste in Jalandhar city, Punjab, India. *Journal of hazardous, toxic, and radioactive waste*, 17(2):97-106.
- [6] Brunner, P.H. & Rechberger, H. 2014. Waste to energy key element for sustainable waste management. *Waste management*, 37:3-12.
- [7] Eurostat. 2009. Europe in figures: Eurostat yearbook 2009. Luxembourg: European Communities.
- [8] Maisiri, W., van Dyk, L. & de Kock, J. 2015. A technological and performance comparison of a waste-toenergy thermal technologies. Proceedings of the 23rd Southern African Universities Power Engineering Conference, University of Johannesburg, South Africa, 28-30 January 2015.
- [9] Maisiri, W., Van.Dyk, L., de.Kock, J. & Krueger, D. 2015. Financial analysis of waste-to-energy grate incineration power plant for a small city. Proceedings of the 12th conference on the Industrial and Commercial Use of Energy, Cape Peninsula University of Technology, Cape Town, 18-19 August 2015.
- [10] Lombardi, L., Carnevale, E. & Corti, A. 2015. A review of technologies and performances of thermal treatment systems for energy recovery from waste. *Waste management*, 37:26-44.
- [11] Martin, J.J., Koralewska, R. & Wohlleben, A. 2014. Advanced solutions in combustion-based WtE technologies. Waste management, 37:147-156.
- [12] Stantec Consulting Ltd. 2011. A technical review of municipal solid waste thermal treatment practices. Victoria, BC: Stantec Consulting Ltd. (No. 1231-10166).
- [13] Maisiri, W. 2016. A techno-economic evaluation of waste-to-energy grate incineration power plant for a small South African city. North-West University, Potchefstroom.
- [14] Whiting, L., Wood, S., Fanning, M. & Venn, M. 2013. Review of state-of-the-art waste-to-energy technologies. London. (No.31427).



- [15] Kurttila, M., Pesonen, M., Kangas, J. & Kajanus, M. 2000. Utilizing the analytic hierarchy process (AHP) in SWOT analysis - a hybrid method and its application to a forest-certification case. Forest policy and economics, 1(1):41-52.
- [16] Muzenda, E. 2014. A discussion on waste generation and management trends in South Africa. *International journal of chemical, environmental & biological sciences,* 2(2):105-112.
- [17] Godfrey, L., Strydom, W., Muswema, A. & Oelofse, S. 2013. Department of science and technology (2013). South African waste sector - 2012 An analysis of the formal private and public waste sector in South Africa. Pretoria: Department of Science and Technology. (CSIR/NRE/GES/IR/ 2013/0078/A).
- [18] Otto, K. & Clements, J. 2008. Survey of generation rates, treatment capacities and minimal costs of health care waste in the 9 provinces of RSA. Pretoria: Department of Environmental Affairs and Tourism.
- [19] Purnell, G. 2014. Trends and opportunities waste to energy (WtE). Presentation delivered at the Department of Science and Technology workshop, University of KZN, South Africa, 26 February. http://www.wasteroadmap.co.za/download/presentation_20140220_03.pdf. Date of access: 11 Sept.2014.
- [20] Wentworth, L. 2014. Creating incentives for green economic growth: Green energy in South Africa. Occasional paper no. 193. Johannesburg: South African Institute of International Affairs.
- [21] South Africa. DoE (Department of Energy). 2011. South African energy synopsis 2010. Pretoria.
- [22] Stengler, E. 2012. Waste-to-energy in Europe. Brussels, Belgium: Confederation of European Waste-to-Energy Plants.
- [23] Serfontein, E.D. 2014. Review of: Draft 2012 integrated energy planning report (IEP), released by the South African department of energy. South Africa: Nuclear Industry Association of South Africa.
- [24] South Africa. DNT (Department of National Treasury). 2013. Carbon tax policy paper: Reducing greenhouse gas emissions and facilitating the transition to a green economy. Pretoria.
- [25] Eberhard, A. 2014. South Africa's renewable energy IPP procurement program: Success factors and lessons. Washington, DC: Private Infrastructure Advisory Facility.
- [26] Rycroft, M. 2013. Summary of REIPPP round three projects. Nooitgedacht, Gauteng: Energize. http://www.ee.co.za/article/mike-rycroft-118-12-reippp-round-three-preferred-biddersannounced.html Date of access 15 Jul.2014.
- [27] South Africa. DoE (Department of Energy). 2015. Renewable energy IPP procurement programme: Bid window 4 preferred bidders' announcement 16 April 2015. Pretoria.
- [28] Fjeldstad, O. 2004. What's trust got to do with it? non-payment of service charges in local authorities in South Africa. *The journal of modern African studies*, 42(4):539-562.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



RESPONSIVE STRATEGIES AND SUPPLY CHAIN DECISION-MAKING FOR THE SOUTH AFRICAN WINE INDUSTRY

L. Knoblauch¹, J. van Eeden² & R. Edwards²

¹Department of Industrial Engineering University of Stellenbosch, South Africa 15660559@sun.ac.za

²Department of Industrial Engineering University of Stellenbosch, South Africa jveeden@sun.ac.za

ABSTRACT

Efficient supply chains have been the order of the day for most wine cellars in South Africa. However, not all wineries can benefit from the same strategy. For some, a responsive supply chain strategy may be considered in order to provide better returns on investment. This paper aims to have a look at how supply chain strategy choice has an impact on the strategic and tactical business decisions of a winery. Furthermore, it investigates which trade-offs need to be made between performance metrics within 'Inventory' and 'Sourcing' as supply chain drivers and how this impacts decisions and strategy.

¹ The author was enrolled for an M Eng (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch



1 INTRODUCTION AND BACKGROUND

1.1 Wine industry background

The South African wine industry is one of the oldest industries in South Africa, dating back to 1655. More than 350 years later, South Africa is the world's 7th largest wine producer [14] and gaining more acclaim for quality wines. The wine industry is a large contributor to the country's GDP (through wine sales and tourism) and employs close to 300 000 workers - unskilled and highly qualified. [18]

To be competitive in the modern business world is not about having the most sought after or cheapest product, but rather the most effective supply chain. These days it is supply chains competing with each other rather than companies. [12] This can be easily grasped when looking at the contemporary consumer. They want their products on-time, at the best price, with flexibility and without forfeiting quality. Supply chain management is therefore at the heart of the modern organization.

For wineries to be profitable and competitive, improving their supply chains could be beneficial. [17] The focus of this paper is to establish how wineries can deploy the correct supply chain strategies for their various supply chains. It investigates how these supply chain strategy choices influence strategic and tactical decisions. The research studies the performance metrics a winery should take into account and which trade-offs need to be made between them as a result of supply chain strategy choice.

Supply chain strategies and supply chain decision-making are not concepts widely found in wine industry literature. [21] The wine supply chain is one of the most complex chains considering the nature of the product, the vast amount of role players and their interaction. [6] Recently some wineries from other 'new world' wine countries (Australia and Argentina amongst others) realized what the supply chain field can offer the industry. They are expectant to see the benefits of implementing various supply chain management tools and frameworks. [13] [6] In the South African wine industry, however, little research has been done on supply chains. According to Van Eeden et al, most wineries do not fully recognize the existence of supply chains within their company and have little accurate data available to inform decision-making. [20]

Since 2003 PwC has been collecting financial data in the wine industry. Their aim is to do forecasting in the wine industry, investigate the outlook industry players have for the industry and identify opportunities and threats in the industry. Only recently has PwC collaborated with the University of Stellenbosch and CSIR to focus on supply chains within the wine industry. [17] The research has identified the lack of understanding and expertise causing a major drawback for the wine industry, as wine producers are already under financial pressure. [20]

1.2 Problem statement and objective

South African wine cellars are not as profitable as they could be. Neither are they competitive enough globally. Van Eeden et al [21] indicated inventory management and supplier partnerships as some of the key challenges and issues in the wine industry. As South Africa's wine quality is gaining worldwide acclaim, one would expect the wine bottles to fly off the shelves. This, however, is not the case. The lack of customer segmentation, proper value propositions, partnerships and supply chain strategy in the wine industry could be possible hurdles in wine producers becoming more profitable and competitive.

The focus of this paper will be to:

- 1) Investigate the wine industry, its supply chain strategies and supply chain decision-making through literature.
- 2) Compile case studies of supply chain decision-making and strategies within wineries.
- Determine a preliminary decision-making structure to take to wine producers. This decision-making structure will include strategic and tactical decisions to be made when applying a responsive supply chain strategy.
- 4) Discuss key performance metrics, measured in 'Inventory' and 'Sourcing' as drivers of the supply chain.

2 RESEARCH METHODOLOGY

This paper investigates supply chain strategies and key performance metrics by means of a literature study and case studies from the industry. Interviews and questionnaires gave the researchers clarity on the current state of supply chains in the wine industry. From there the researchers aimed to develop a decision-making framework for the wine supply chain, focussing on the supply chain strategy decisions, for the areas of inventory and sourcing.



Previous research was studied to understand what has already been done in this field. Not many previous studies on supply chains were found specifically in the wine industry. From literature, the researcher was able to identify the extent of the problem and possible ways of implementing different supply chain strategies.

Resources that were consulted include a search through online databases for journal articles, books and conference papers. These searches included key words such as "supply chain strategy", "supply chain management", "wine supply chain", "wine industry South Africa", "performance metrics and measurements" and "decision-making in supply chains" amongst others. Further informal knowledge and understanding of the matter was also gained by reading internal project documentation, magazine articles, online articles and conference proceedings. It is clear that the wine industry has not done much business or supply chain research, as the literature mainly consists of scientific studies to improve wine or grape quality. The South African wine industry is under financial pressure and some advice on streamlining supply chains would be beneficial. [17]

Gunasekaran [7] found a lack of case studies in the performance measurement field across all industries in 2004. Although there are more studies available now, the wine industry has not caught up with performance measurement yet.

The researchers investigated the supply chain decision-making and responsiveness through compiling case studies of three wine producers. Every wine farm had a unique supply chain challenge, from which they had to make strategic and tactical decisions. Data for the case studies were gathered through face to face interviews, telephone conversations, e-mails and questionnaires. Focus was placed on how the cellars developed a responsive supply chain strategy, if the wineries were successful in their implementation thereof and how this could be improved. From this process, the researchers deducted which metrics could help wineries perform better in terms of 'Inventory' and 'Sourcing' as supply chain drivers. The wine producers will stay anonymous throughout the study to protect their identity.

3 LITERATURE REVIEW

3.1 Wine supply chain studies

Limited studies have been done on supply chain optimization in the wine industry. Some research has been done on elements of supply chains, but not on supply chains as a whole, leaving room for improvement. *Figure 1* illustrates the elements in the wine supply chain. Such a supply chain should be drawn for every individual wine, because not all wines (products) can be modelled on the same supply chain and follow the same supply chain strategy. This makes the wine supply chain complex. [5] [1]



Figure 1: Wine supply chain [6]

South Africa produced 11 million hectolitres, 4.1% of the global wine production in 2015. [14] Wine production is increasing yearly and so are the export and import markets worldwide. The European market is now opening up to accept an additional 60 million litres of packaged wine tax free. [18] This means that South African wine producers would want to tap into that packaged export market more aggressively. Wineries need to focus on their supply chains and supply chain strategy in order to fulfil customer needs worldwide. [6] For South Africa to be more competitive than the other New World wine producers (Australia, New Zealand and the USA), the



focus needs to shift to optimizing the supply chain and aligning the different supply chains to the winery's business strategy.

For the local market, distribution should be an area of focus. [11] Since grape growing and production is site specific, distribution for the local market should be streamlined. *Figure 2* shows the grape production areas in the coastal Cape area. [19]



Figure 2: South African Wine Regions [19]

This makes responsiveness difficult to the northern part of the country. Responsive focussed wine cellars should invest in having distribution centres in other parts of the country. Alternatively, wine cellars can outsource their distribution to national wine distributors. It is, however, important that they choose a distributor with a similar footprint that has the same strategy, either being cost efficient or responsive.

According to PwC surveys, there are insufficient "industry best practices" or "performance indicators" against which cellars can measure themselves. They came to the conclusion that wineries willing to invest in a proper supply chain framework and strategy, will be leading the way in "new and innovative ways of doing business". [17] The development of such a framework to enhance decision-making could improve their return on investment, but requires a solid understanding of their supply chain strategy.

3.2 Business/ Competitive strategy

The company's business strategy identifies the customer needs and seeks to satisfy those needs through products or services, better than the competition. [3] If a customer segment buys wine and requires it to be delivered immediately, the winery's supply chain strategy should be set up for responsiveness in order to win the customer order. The winery will be able to charge a higher price on the product, because it is immediately available on customer demand. The ideal is to be as responsive as possible at a reasonable cost to this specific segment. [1] High inventory levels, company-owned facilities and effective information systems are some of the strategic decisions made by management in order to win these types of customers above the competition.

Business is governed by three tiers of decision-making: Strategic, tactical and operational. Each tier has an important function in management. This strategy should be aligned with their business strategy. Their business strategy can be described by "how the organization intends to create and sustain value for its shareholders." [9]

Strategic supply chain decisions are made by senior management, keeping all business areas in mind, advisably with leaders of each area involved. On the strategic level, company policies, competitive strategies, company financial plans and adherence to corporate goals are regulated by management. [7] Big decisions regarding the key logistical drivers are also considered strategic decisions, for example, whether or not to have the company's own facilities and equipment.

On a tactical level, decisions need to be made about keeping to the company's set strategic goal, adopting industry best practices and allocating resources in the most optimal way. [7] Tactical decisions often overlap with strategic decisions.



Operational decisions are short-term focussed and include decisions made every day on the job. These include adherence to and rescheduling of a production plan and the assurance that everything according to tactical level decisions gets implemented. [7] Operational decisions fall beyond the scope of this article.

3.3 Supply Chain Strategy

In Fisher's [4] very popular article "What is the Right Supply Chain for Your Product?" he lists the drivers of different strategies, shown in Table 1. There are major differences between the efficient and responsive strategies. The efficient strategy is usually implemented for functional products with low demand variation and profit margins. A responsive strategy is implemented for innovative products with high demand variation and profit margins. [4] [3]

	Physically Efficient	Market - Responsive		
Primary Purpose	Lowest cost	Ability to meet unexpected demand		
		High profit margins		
Manufacturing focus	Cost efficient	Quality of product		
	High utilization	Quality service		
	Full capacity	Excess capacity, utilization buffer		
Inventory Strategy	Make to order	Make to stock		
	Minimize inventory to cut storage	Invest in storage to be able to supply		
	cost	demand		
Lead-time focus	Shortest lead time for lowest cost	Short lead time extremely important		
		High investment		
Approach to choosing	Lowest cost at reasonable quality	Reliable suppliers, shorten lead-time,		
suppliers (Sourcing)				
Product-design strategy	Generic design to minimize cost	Customized wines		
	Little to no differentiation	Special requests can be fulfilled		

Table	1: Supply	Chain	Categorization.	(Adapted	from	Fisher	[41)
		••••••		(L 'J/

Table 1 addresses some of the 'Inventory' and 'Sourcing' aspects in the supply chain, which is a main focus of this paper. For the wine industry, inventory refers to raw materials, work-in-process wine and the bottled finished goods. Sourcing would refer to agreements with suppliers of materials to make and package wine, i.e. grapes and/or wine and dry goods. The wine making process cannot be rushed, thus the market variation needs to be buffered with finished goods or bottled ready wine and a fast labelling facility. In order for a winery to be efficient in terms of 'Inventory', it needs to minimize inventory by implementing a make-to-order strategy, lowering storage cost. The opposite is true for a responsive strategy where more stock is kept in inventory (make-to-stock strategy), which makes the winery able to supply unpredictable demand. In terms of 'Sourcing', the efficient winery will opt for the lowest cost option every time, even if it has some minor quality implications. The responsive winery should have reliable suppliers, which delivers the right quality, at the right time at the shortest lead time.

The electronics industry has some of the same attributes as the wine industry and will be used as another example to demonstrate how 'Inventory' and 'Sourcing' can be used as supply chain drivers. Electronics also has a short life cycle, with the technology changing ever so quickly ("new" wine products are made yearly due to climate differences). Premium Brands are innovative products, with uncertain demand. Thus, according to Fisher [4], niche wineries and electronics can be dictated by a responsive supply chain strategy. Helo [8] also mentions that competition in the electronics industry is time based - you have to be first and fast to win orders. Therefore, dry goods need to be kept on hand, with extra capacity for manufacturing. Too much finished goods stock can easily be redundant due to the fast changing technology in electronics. Similarly, older white wines can become redundant and have to be sold at a markdown price as soon as the newest harvest seasons become available. The suppliers also need to be reliable, with short lead times in order to be responsive to the market pull.

According to Chopra [3], the key to achieving strategic fit and strong financial performance across the supply chain is to structure the supply chain drivers appropriately, to provide the desired level of responsiveness at a reasonable cost. In **Error! Reference source not found.**, Chopra illustrates where strategic fit is to be found. Wine cellars need to segment their customer-product offering and then place each of their supply chains in the matrix in *Figure 3*. This will dictate which strategy they need to follow for each individual supply chain.





Figure 3: Achieving a fit in the supply chain [3]

According to Ambe "mismatched strategies are the root cause of problems that plague supply chains, and strategies based on a one-size-fits-all strategy often fail." [2] A common practice that wineries tend to neglect is segmentation. [21] Segmentation of customers and products into supply chains are crucial to the profitability of the winery, who need to understand the variety of customers they intend to serve. According to Osterwalder, every customer segment has different aspects they value. In order for the winery to win and keep customers, they need to approach these different segments to create value through applying the right supply chain strategy to every customer segment. [15] In this way all customer segments are satisfied by aligning the supply chain with the correct strategy. For example, when a customer wants the fastest delivery, irrespective of the cost - a responsive strategy will generate the most sales and potential profit.

3.3.1 Implementing an efficient supply chain strategy

Most wineries in South Africa implement an efficient supply chain strategy across all their supply chains. [21] Larger wine producers can make more profit selling high volumes. Some smaller wineries perceive this strategy to be successful and also tried it. With an efficient supply chain strategy, the focus is on economies of scale, cutting costs, sometimes compensating on quality. This strategy is suited for wineries that sell large volumes of wine, with a lower profit margin and a steady demand. The steady demand enables the company to reduce inventory and plan in advance for product replenishments with suppliers. Customers buy these products because of its low cost.

3.3.2 Implementing a responsive supply chain strategy

Perez suggests that where a fluctuating demand for products exist and market mediation costs are highly relevant, one of the three responsive strategies should be considered (agile, custom-configured and flexible). Perez reckons this will improve return on investment. The business provides capacity in response to changes in demand. [16] These supply chains are focused on the customer willing to spend a bit more when wanting a product immediately or with specification alterations. The manufacturer is geared to meet customer demand and might be able to increase profit margins in response to meeting customer specifications.

Wineries with innovative products and customers willing to pay a premium price should follow a responsive strategy. An **agile** strategy will be followed when customers value short lead times and when a company's business strategy is orientated towards high service levels and delivering on fluctuating demand. [16] Wineries on this strategy would compete in terms of product availability. The winery should have enough finished product inventory on hand. Or if they prefer to keep sufficient dry goods in inventory, they should ensure high reliability and extra capacity on their bottling/labelling line. Agility is needed in an unpredictable environment. [1]

A **custom-configured** strategy is uncommon in the wine industry. This could be because the opportunity does not exist to do this often. It probably would be more popular if a user-friendly system existed to "design" your own wine. On this strategy, a winery keeps no finished products in inventory, because of custom design at a premium price. However, to be responsive in the market, dry goods are kept in stock along with reliability from



dry goods suppliers for expensive or bulky items, and excess capacity is required in production. This strategy is geared towards blending (and bottling) wine according to customers' specifications. [16]

A flexible strategy would play a bigger role in the distribution of wines, since the ability to meet fluctuating demand in terms of delivery is extremely important. This means that the winery needs to keep sufficient finished product inventory on hand in order to supply the demand. In the wine industry, inventory can be considered as components or unlabelled bottles with sufficient capacity to quickly assemble and ship. They should also have reliability in terms of distribution services: If a third party distributor is used, it should be extremely reliable. If the winery uses its own distribution trucks, etc., it should always have extra capacity leading to lower asset utilisation.

3.4 Supply Chain Decisions

The aim of business is to improve Return on Investment (ROI). Growing the supply chain surplus is the ultimate goal of a supply chain and in effect, the companies involved as a whole. In order to achieve a better business performance, it is of importance to track the key performance drivers and achieve a strategic fit between the supply chain strategy and competitive strategy.

As illustrated in **Error! Reference source not found.**, a winery needs to start by designing its 'competitive strategy': how it intends to win or maintain market share in that specific supply chain. Thereafter a supply chain strategy will be chosen to align with the competitive strategy and all further decisions must be based on these strategies.



Figure 4: Supply chain decision-making framework [3]

Chopra [3] describes three logistical drivers: facilities, inventory and transportation. Furthermore, there are three cross-functional drivers that play a big role: information sourcing and pricing. All six of these key drivers encompass strategic decisions by management. The logistical drivers require decision-making about assets the company owns or outsources. The cross-functional drivers require decision-making about 'how' every logistical driver will be obtained, used, delivered, etc. All these decisions should be based on what the supply chain strategy entails. For example, if a winery intends to follow a responsive strategy, it should keep more inventory on-hand to meet unexpected demand. Alternatively, they could make use of reliable suppliers, with



shorter lead times, but which would also be more expensive. Or they could invest in sufficient facilities to quickly bottle and/or deliver products.

Key to the winery's success is to deliver what they promise. It is no use to advertise for delivering products on short notice, but keeping low inventory and having unreliable facilities or suppliers. It takes time to build responsive trust, but quickly to break it.

3.5 Performance Metrics

Organizational performance metrics have received a lot of attention from industry. The use of metrics has not been studied and implemented so widely in the wine industry. [7] Currently there is no implemented performance measurement framework in the South African wine industry. [10] The development and implementation of such a framework is currently being investigated by several Master students at the University of Stellenbosch. Several students have completed their studies and some of these outputs are referenced in this paper. Supply chain performance can be enhanced through supply chain control, which can be executed in part by measurement. [7]

3.5.1 The trade-offs between different performance metrics

A business is not able to perform simultaneously at all competitive levels and should, therefore, have a strong notion of how they intend to gain their competitive advantage. Key to planning operations and supply chain strategies should include an operations focus and trade-offs. [9]

Every supply chain metric inhibits a trade-off that directly affects the strategy of the firm. These trade-offs are applicable in the wine industry. The researchers considered 'average inventory' and 'supply lead time' as performance metrics.

A winery cannot have both high and low 'average inventory' levels. Inventory will be considered as raw materials, dry goods and finished products. Keeping higher average inventory levels, allows the winery to be much more responsive to market demand, but at a higher cost. The efficient supply chain strategy will try to keep the average inventory levels of finished goods as low as possible in order to save on storage costs. Supply chain strategy and customer segments thus play a big role in determining how much average inventory will be kept on hand.

Having a shorter 'supply lead time' can assist a winery in not keeping too many finished products in inventory. Suppliers with shorter lead times and reliability would be more expensive, but being responsive on short notice will ensure satisfying customers changing demand patterns and assist in keeping those customers. These customers are usually also willing to pay a higher price to have delivery on time. This is also useful in the electronics industry as less inventory needs to be kept in stock (with a possibility of the stock becoming outdated). Being able to quickly assemble a product if suppliers are reliable and have short lead times, enables the producer to respond to demand without carrying unnecessary inventory. The same is true for the wine industry in the case where white wine becomes redundant once the new harvest season's product is available, thus keeping finished goods inventory low is an important strategy consideration.

4 CASE STUDIES

The researchers considered three different wineries, each with a range of products. Critical decision-making is required in order for the winery to achieve optimal supply chain and business performance. The researchers considered which decision-making measures should be taken into account and which trade-offs need to be made in order to achieve strategic fit for specific scenarios at each winery. Due to the space limitations of this paper, only a few relevant scenarios related to inventory and sourcing will be highlighted for each of the case studies.

4.1 Winery 1

This winery produces a special wine in small volumes with high quality expectations from the customers. One year the production of this wine was impossible due to low quality grapes. The dry goods were already ordered, including an imported, long lead time bottle. Dry goods are expensive and ordering should be postponed until final volumes are known in efficient strategies. The winery had two major problems in terms of this wine not being good enough for the 'special' quality wine:

- 1) What will be done with the excess 'low quality' wine?
- 2) What will be done with the bottles that were ordered?


The wine could easily be blended into another lower quality blend the winery produces and take up the storage space the 'special' wine would have used. Since this wine uses another bottle, locally made, more could be ordered and delivered easily. The imported bottle cannot be used for any other wines produced by this winery. Thus a choice needed to be made whether to store the bottles until the next harvest or to return it. Both options would incur a cost. In this case, the trade-off is being proactive with inventory in order to be able to react responsively versus postponing decision-making to be cost efficient.

As the wine industry is cash-flow sensitive, having cash tied up in inventory for a year can lead to problems. In order for this winery to remain responsive in such situations the following metrics can be considered:

- 1) Supplier reliability and lead time
- 2) Possibility of designing and producing the same bottle locally, in order to reduce lead time (the average cost of this also to be measured)
- 3) Cost of inventory versus supplier reliability
- 4) The amount of safety stock to be kept in inventory to supply demand

These measurements would provide feedback to management and inform better decision-making related to the areas of inventory and sourcing within a responsive supply chain strategy.

4.2 Winery 2

This winery is a small producer of premium and super premium wines. Over the years they have had a constant production volume and done their forecasting accurately in order to meet customer demand. They are able to be responsive in terms of bottling and labelling their premium wines. With excess capacity on the bottling and labelling line, they can quickly have wine ready to ship according to changing customer demands.

One year, however, a lower yield was harvested from the farm. This led to insufficient grapes for their super premium wines. These super premium labels were discontinued and the wine back blended into the premium wines. This was done at the risk of losing listings at restaurants and on retail shelves permanently. This would be the same effect if a respected cell phone company could not release a new device for a particular year, and thus lose customer loyalty.

In terms of strategic decision-making, this winery could consider sourcing high quality grapes from elsewhere in order to produce their super premium wines. Estate title considerations should be discussed and decided upon. It should be weighed against the responsiveness of the cellar, meeting customer expectations, and losing customer loyalty.

4.3 Winery 3

This winery produces a variety of wines, including single varietals and blends for the export market. In recent years, demand for the blends has been growing and the demand for single varietals is decreasing. In order for this winery to be responsive to customer demand, some strategic decisions need to be made.

Their main considerations should be if they will buy in more grapes to make blends, plant their own vineyards, use grapes from the single varietal production for blends and the additional facilities that are needed to produce these wines. These are all strategic decisions to be made at a senior management level.

If this winery intends to be successful in their responsiveness, they can consider the following 'Sourcing' metrics:

- 1) Consider purchasing price of grapes over time against the planting of own grapes
- 2) The quality of the grapes bought in
- 3) The reliability of quality and timeliness
- 4) The quantity of available quality buy-in

This winery needs to consider strategic partnerships with grape suppliers in order to ensure the quality and quantity of grapes they require for their various blends.

4.4 Decision-making

Throughout the case studies strategic and tactical decision-making were crucial to the success of these wineries. When problems arise, critical decisions should be made with supply chain strategy in mind. Table 2 provides an overview of the types of decisions to be made by the aforementioned wineries when they consider implementing a responsive supply chain strategy. Wineries with similar situations can follow suit. The focus of this article is only on 'Inventory' and 'Sourcing' as supply chain drivers, and thus the list is limited to these drivers and the related performance metrics.

Performance measurements can assist strategic and tactical decision-making to be more responsive in the supply chain. As mentioned previously in the case studies, there are different situations in which these metrics



can be useful and a measure as to how responsive a winery can be. The trade-offs between efficiency and responsiveness should be considered when decisions are made so as not to implement conflicting strategies.

Drivers of supply chain	Relevant performance	Applicable to case		
	metrics	study		
Inventory:	Cash to cash cycle time	Winery 1, 2 & 3		
Cycle inventory	Average inventory	Winery 1 & 2		
Safety inventory	Inventory turns	Winery 1 & 2		
Seasonal inventory	Safety stock	Winery 1, 2 & 3		
Level of product availability	Order fill rate	Winery 2 & 3		
	Fraction of time of stock out	Winery 2 & 3		
Sourcing:	Average purchase price	Winery 1 & 3		
In-house vs outsourced	Range of purchase price Winery 1 & 3			
Supplier selection	Average purchase quantity	Winery 3		
Procurement	Supply quality	Winery 1, 2 & 3		
	Supply lead time Winery 1 & 3			
	Fraction of on-time deliveries	Winery 1 & 3		
	Supplier reliability	Winery 1 & 3		

Table 2: Proposed	performance measurements.	(Adap	ted from	Chopra	[3]
Tuble Et Troposed	periormanee measurements:	(/ Caup	ce a mom	enopra	1.01

As responsive strategies are often implemented where quality products are produced and demanded by specific customer segments, the quality aspects are important considerations. The cost of quality, shorter lead times and reliability are high, but worth the investment to keep market share in responsive supply chains. Inventory cost is also higher in order to be more responsive, but easy calculations can be made to ensure responsiveness at the lowest possible cost.

5 CONCLUSION

The literature on the wine supply chain, the South African wine supply chain specifically, is very limited. It is necessary to consult other industries with overlapping concerns and similar production environments, as the electronics industry only mentioned in this paper. In further research, supply chains of industries such as the textile industry can be considered to compare with the wine industry. The wine industry is similar to the textile industry in the light that both constantly produce new products launched on the market as seasons and trends change. It sounds outrageous, but a winery effectively has new products every year with the changing harvest climates. The responsive winery can in a similar way be responsive to the market 'pull' due to accurate forecasting as the textile industry also needs to forecast trends and market demand. However, it is also crucial that the industry sets its own benchmarks and develop the necessary industry best practices to take the industry forward. The recent interest in the South African wine supply chain by industry role-players and academics is a positive sign and could benefit the industry in improving to the next level in terms of supply chain performance and competitiveness.

The development of frameworks to measure the performance of the wine supply chain within wineries, especially regarding decision-making, should make wine producers aware of the different options within their supply chains. Due to the nature of the product and the complexity of the supply chain, it most probably will take several years to implement frameworks and improve supply chains.

Since South African wines have been performing extremely well in terms of quality recently, the demand for South African wine is rising. In order to fulfil this demand, supply chain management and strategies will need to improve and grow in the South African wine industry over the next decade. Maybe a more responsive strategy can improve agility and the ability to meet the potential changing customer demands.

The authors propose that wineries segment their customer markets carefully and develop value propositions for each of these markets. In doing this, wineries should consider the option of developing responsive supply chain channels to serve some of these markets where appropriate. The metrics provided in this paper could assist wineries in making the correct strategic and tactical inventory management and sourcing decisions.

6 REFERENCES

[1] Ambe, I.M., 2010. Agile supply chain: strategy for competitive advantage. *Journal of Global Strategic Management*, 7, pp.5-17.

[2] Ambe, I.M., 2012. Determining an optimal supply chain strategy. *Journal of Transport and Supply Chain Management*, 6(1), pp.126-147. Available at: http://uir.unisa.ac.za/



[3] Chopra, S., Meindl, P. 2013. Supply Chain Management: Strategy, Planning and Operation, 5th Edition, Pearson.

[4] Fisher, M.L. 1997. 'Finding the right supply chain for your product', *Harvard Business Review*, March-April 1997, 83-93.

[5] Garcia, F.A., 2009. Modelling and measuring logistics performance in the wine supply chain: The case of the Argentine wine industry (Doctoral dissertation, M. Sc. thesis, National University of Cuyo, Mendoza, Argentina).

[6] Garcia, F.A., Marchetta, M.G., Camargo, M., Morel, L. and Forradellas, R.Q., 2012. A framework for measuring logistics performance in the wine industry. *International Journal of Production Economics*, *135*(1), pp.284-298.

[7] Gunasekaran, A., Patel, C. and McGaughey, R.E., 2004. A framework for supply chain performance measurement. *International journal of production economics*, 87(3), pp.333-347.

[8] Helo, P., 2004. Managing agility and productivity in the electronics industry. *Industrial Management & Data Systems*, 104(7), pp.567-577.

[9] Jacobs, F.R., Chase, R.B. 2014. Operations and Supply Chain Management, 14th Edition, McGraw-Hill.

[10] Jooste, C., van Eeden, J., van Dyk, E., 2015. South African Wine Supply Chain Performance Measurement Framework. *Innovations and Strategies for Logistics and Supply Chains*, August 2015, 305-322.

[11] Jooste, C., 2016. Performance Measurement Framework for the South African Wine Supply Chain: An Investigation of Packaged Products in the Local Market.

[12] Martinez-Olvera, C. & Shunk, D., 2005. Comprehensive framework for the development of a supply chain strategy. International Journal of Production Research, pp. 4511-4528.

[13] Monday, A. and Wood-Harper, T., 2010, January. Exploring the supply chain of small and medium-sized South Australian wine producers. In *Supply Chain Forum: An International Journal* (Vol. 11, No. 1, pp. 16-26). Taylor & Francis.

[14] OIV, 2015. International Organization of Vine and Wine. www.oiv.int

[15] Osterwalder, A. and Pigneur, Y., 2013. Business model generation: a handbook for visionaries, game changers, and challengers. John Wiley & Sons.

[16] Perez, H.D., 2013. Supply Chain Strategy: Which one hits the mark? CSCM P's Supply Chain Quarterly.

[17] PwC. (2013). The South African wine industry, Insights survey 2013.

[18] SAWIS. 2015. 'SA WINE INDUSTRY RAISES ITS GDP CONTRIBUTION, GENERATES MORE JOBS', Press release. Available online: <u>http://www.sawis.co.za/info/download/Press_release_2015</u>.

[19] SAWIS. Available online, http://www.sawis.co.za/cert/download/Regions.

[20] Van Eeden, J., Goedhals - Gerber, L., Louw, J. 2012. Wine Supply Chain Survey 2012 - Findings and Challenges. PAPER PRESENTED AT SAPICS 35TH ANNUAL CONFERENCE AND EXHIBITION. www.sapics.org.za

[21] Van Eeden, J. Van Dyk, FE. Louw, JJ. Jooste, C. Smit, J. Roos, T. 2014. Development of a supply chain performance measurement framework for the South African wine industry at the *Proceedings of the Third International Workshop on Food Supply Chain. (Making Food Supply Chains Efficient, Responsive and Sustainable).* San Francisco, CA, 4-7 Nov 2014.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



DECISION SUPPORT HEURISTICS FOR COST ESTIMATION MODEL FOR INJECTION MOULDS

M.T. Dewa^{1*}, A.F. Van der Merwe², S. Matope³ and L. Nyanga⁴

¹⁻⁴ Department of Industrial Engineering University of Stellenbosch, South Africa

> ¹<u>17628598@sun.ac.za</u> ²<u>smatope@sun.ac.za</u> ³<u>andrevdm@sun.ac.za</u> ⁴nyangalu@gmail.com

ABSTRACT

Cost estimation of products in the tooling industry is a complex task requiring a lot of expert knowledge and sound judgement. Higher estimates may lead in losing orders or customer goodwill while lower estimation will affect business profitability. Hence accurate and timely cost estimation before tool, die and mould production is a key attribute for sustaining global competitiveness. Due to the skills shortages the South African tooling industry has experienced, a majority of Toolmaking firms take long times to quote a job. Furthermore results from benchmarking exercises of these firms have shown that a majority of quotes generated lack accuracy due to methods employed. In the paper, the key parameters to be considered when quoting the price of injection moulds are identified and ranked. The method of knowledge engineering was employed and heuristic data collected through interviews with five job quoting experts in the Western Cape Province tooling sector. Based on the information gathered, the Analytical Hierarchical Process (AHP) was used to rank the main parameters identified. The result of the survey can be used in the development of an expert system for quoting injection moulds in the South African Tooling industry.

Key words: Cost Estimation, Knowledge engineering, Injection moulds, Expert system

^{*} The author is currently enrolled as a Ph.D. student in the Industrial Engineering Department at Stellenbosch University, South Africa

^{*} Corresponding Author



1. INTRODUCTION

Tool-making firms adopt a crucial role in the manufacturing value chain, contributing significantly to quality, price and the delivery reliability of series production goods [1]. According to the International Specialized Tooling and Machining Association (ISTMA) statistics, up to 50% of any manufactured component's cost competitiveness is governed by tooling (Instimbi Annual Report, [2]).

The South African Tool, Die and Mould-making (TDM) industry plays a major role in the growth and existence of the local manufacturing sector. With rapid growth of globalization and fierce competition from tool-making firms in the East, timeous delivery of a quality tool, die or mould within the allowed budget in non-negotiable. Unfortunately, recent results from the benchmarking exercises conducted in the South African TDM sector have revealed that a majority of the firms suffer in the area of delivery lead times (Malherbe [3]) as shown in Figure 1. Though there are several reasons for this trend as explored by Dewa [4], the order processing function has been identified as one of the main bottlenecks in the Tool-making value chain for most companies observed.





Figure 1: Due Date conformance trends [Rittstieg and Garms, [5])

At the present moment the job quoting process is time consuming since it requires great expertise and knowledge, which is currently in short supply within the TDM sector (Instimbi Annual Report, [2]). Most companies take long lead times (usually weeks) to quote a job and in some cases, the tool-making project doesn't end within budget as shown in Figure 2. A potential solution to the dilemna can be the development of knowledge-based (expert) systems for job quoting which encapsulate the main parameters required in pricing a job.

Part of expert system development involves interaction with field experts to derive key heuristic rules utilized in decision-making, a process known as knowledge engineering. The paper presents the key parameters used by expert cost estimators in the tooling industry during pricing jobs, with specific focus on the quoting of injection moulds. Five different tool-making firms were used in the study. Cost estimation experts in these firms participated in interviews and the parameters were derived. The paper is structured as follows: Firstly, the complexity of the quoting process is highlighted before different approaches to cost estimation of injection moulds are outlined. Secondly, the Analytical Hierarchical Process (AHP) is explored from the literature. Finally, we present the data collection methodology adopted, firm demographics before we finally show the results of the study.







Figure 2: Orders completed within budget [Rittstieg and Garms, [5])

2. RELATED LITERATURE

2.1 Complexity of the Cost Estimation Process

The successful calculation of costs is an important requirement for the performance of injection molding companies. Higher estimates may lead in losing orders or customer goodwill while lower estimation will affect business profitability. However, determination of the overall costs of a tooling job is a complex process which requires great expertise and knowledge in tooling. According to Jones [6], the estimator needs to have sound judgement and knowledge in:

- Injection molding specifications and functions
- Reading and understanding of engineering drawings in both printed and Computer-Aided Design (CAD)
- Mould tool design
- Tool-making
- Mechanical engineering
- Basic physics
- Plastic material properties
- Behavior of polymer flow
- Injection moulding
- Basic electronics
- Robotics

A delay in quoting is bad practice. This is because research has shown that in the mould-making industry usually less than 10% of all offers turn into orders (Fonseca, M et al., [7]; Duverlie, and Castelain, [8] and Denkena et al., [9]). Hence accurate and timely injection mould quoting is key to survival in a cutthroat business environment. For timely quoting, the key elements to be considered in pricing the job should be known. The paper aims to identify the key parameters required in quoting an injection mould.

2.2 Cost Estimation systems

The overall goal of the cost estimation process is to ensure that all the costs related to a job are covered with a satisfactory profit realized. Furthermore, the final price of the product must be competitive on the market to avoid the risk of a lost order. To achieve this goal, injection moulding firms need to identify all the related costs incurred in manufacturing a mould, select an appropriate costing system and estimate the price based on the identified parameters and selected system. Manufacturing firms usually adopt the following costing systems; standard costing, absorption costing, marginal costing and costing based on machine hourly rates.

2.2.1 Standard Costing

Standard costing is a very common method used in the manufacturing industry. The method is based on established set standards for each manufacturing activity or process. These standards are derived from time study analysis and agreed upon within the organization. Usually, costs are classified broadly within the categories of direct, indirect and overhead expenses.



2.2.2 Absorption Costing

Absorption costing is an approach which ensures that all manufacturing costs are recovered by the products produced. The method is also referred to as full costing or complete absorption method. It involves identification and accurate assessment of all the costs a business incurs. Similarly, costs are broadly classified as either direct or indirect expenses. However, this method is best suited for determination of the selling price of products in a profitable way.

2.2.3 Marginal Costing

With both standard and absorption costing, the allocation of fixed and overhead costs can be complex to analyse. The costs behave in different ways as the production or sales volumes change. However, in marginal costing the behaviour of the associated costs is measured. In marginal costing, expenses are broadly categorised as either fixed or variable costs.

2.2.4 Costing based on machine hourly rates

Costing based on hourly machine rates is a system usually adopted where the majority of the production is done by industrial machinery. A machine rate per hour (MHR) is determined for each resource and used in determining the final cost of a job. Since the injection moulding process is heavily reliant on the use of Computer Numerically Controlled (CNC) machinery, this approach is usually the most appropriate for costing of injection mould jobs. Usually these systems are either applied using manual filing system or the estimator uses computerized packages to do the job [6].

2.3 Cost Estimation Approaches

According to Duverlie and Castelain [8], there are four different approaches estimators use when determining the price of a job. These are the intuitive, analogical, parametric and analytical methods. The intuitive method is employed when the estimator heavily relies on his or her experience to make decisions based on their knowledge. In most cases, gut feeling or rules of thumb are employed to make a decision using this approach. On the other hand, the analogical method uses similar jobs done based on historical records to approximate the cost of a new job. The parametric method seeks to evaluate the costs of a product based on specific parameters characterising the product. However, the product does not necessarily have to be fully described to follow this method. The analytical method allows for evaluation of the cost of a product from a decomposition of the work required into elementary tasks. This makes the analytical method to be mostly suitable for firms using a machine hourly rate costing system. The estimation process is complex in that so many different decisions are made at the same time. Hence, job estimation is a multi-criteria decision-making problem.

2.4 Analytical Hierarchy Process

The Analytic Hierarchy Process (AHP) is a multi-criteria decision-making methodology which was proposed by Saaty [10]. It uses a well-defined mathematical structure of consistent matrices and their Eigen vector to generate weights on the selection criteria [11]. To make decisions with AHP the following steps should be followed [12]:

- 1. The problem should be defined.
- 2. A decision hierarchy with the goal of the decision at the top followed by objectives from a broader perspectives, then intermediate level criteria is developed from the top level down to the alternatives.
- 3. A set of pair wise comparison matrices is developed with each element on an upper level used to compare the elements in the level immediately below it.
- 4. The priorities of the comparison the criteria in the upper level is used to weigh the criteria in the lower level immediately under it.

2.4.1 Problem definition

The facilitator interviews the decision-maker(s) to structure the problem and develop a hierarchical structure of the criteria which enables users to focus on specific criteria and sub-criteria when allocating the weights. Brugha [13] also proposed a complete guideline to structure a problem hierarchically and compiled hierarchies in different applications. To avoid large differences when decision making involves a large number of elements, the elements should be clustered [14].

2.4.2 Pair wise comparison

A pair wise comparison matrix is developed using a scale of numbers that indicates how many times more important or dominant one element is over another element, with respect to the criterion or property to which they are compared as shown in Table 1. The development of the matrix is discussed in section 4.5.



Intensity of Importance	Definition	Explanation
1	Equal Importance	Two activities contribute equally to the objective
2	Weak or slight	
3	Moderate importance	Experience and judgement slightly favour one activity over another
4	Moderate plus	
5	Strong importance	Experience and judgement strongly favour one activity over another
6	Strong plus	
7	Very strong or demonstrated importance	An activity is favoured very strongly over another; its dominance demonstrated in practice
8	Very, very strong	
9	Extreme importance	The evidence favouring one activity over another is of the highest possible order of affirmation
Reciprocals of above	If activity I has one of the above non-zero numbers assigned to it when compared with activity j, then activity j has the reciprocal value when compared to i.	
1.1 - 1.9	If activities are very close	May be difficult to assign the best value but when compared with other contrasting activities the size of the small numbers would not be too noticeable, yet they can still indicate the relative importance of the activities.

Table 1: Fundamental scale of absolute numbers [10]

2.4.3 Parameter ranking

The parameters investigated are then ranked by determining their priorities during the interview process. The priority rating of each parameter is obtained by finding the Eigen vector of the pairwise comparison matrix. This is achieved by raising the matrix to large powers and summing each row and dividing each by the total sum of all the rows [13]. Once the priorities have been determined, the principal Eigen value can be obtained from the summation of products between each element and the sum of the columns of the reciprocal matrix.

2.4.4 Checking for consistency

The final step in the process involves checking for consistency in judgement. This is done by measuring a Consistency Ratio (CR) as shown in equations 1 and 2.

$$CI = \frac{\lambda_{max} - n}{n - 1}$$

Where λ_{max} is the Principal Eigen value of the pair wise comparison matrix and n is the dimension of the matrix. The Consistency Ratio (CR) is given by:

$$CR = \frac{CI}{RI}$$
(2)

Where CI is the Consistency Index and RI is the Random Consistency Index given in Table 2. If the determined Consistency Ratio is smaller or equal to 10%, the inconsistency will be acceptable. However if the Consistency ratio is greater than 10%, they will be need to revise the subjective judgement.

(1)



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Table 2: Random indices [15]

n	3	4	5	6	7	8	9	10
RI	0.58	0.9	1.12	1.24	1.32	1.41	1.45	1.49

2.5 Cost Estimation Parameters

The costing of a product in the manufacturing industry can be broadly classified into production, packaging and distribution-related costs. According to Jones [6], production related costs include contributing factors like machine-based operations, materials used, assembly or manual tasks and support processes. For accurate quoting, these broad categories need to be further explored and broken down. The paper seeks to fill this gap by exploring how toolmakers in the South African tooling industry reckon these factors during quoting while also exploring how these costs elements compare in terms of their contribution to the total cost of the product. Knowledge engineering can aid in deriving the information of how the parameters are reckoned while the AHP approach assists in the ranking of the identified parameters.

3. RESEARCH METHODOLOGY

The method of knowledge engineering was selected as the best approach to answering the research question. According to Lucas and van der Gaag [16], knowledge engineering involves interaction with field experts to derive key heuristic rules utilized in decision-making. As such, job quoting experts in five selected firms within the Western Cape Tooling industry were visited at different times and interviewed. All selected firms accepted to participate in the study and were visited at different times. The purpose of the interviews was to determine the key parameters required when quoting an injection moulding job. The companies selected were specifically specialists in manufacturing of injection moulds. Furthermore, the method of observational note taking was employed to determine the approaches and systems of quoting jobs the experts employed. Part of the interview involved the expert demonstrating how the price of a job was reached from the design drawing specifications to the preparation of the job invoice. An interview guide was prepared for the interaction process. The guide involved both open-ended and closed-ended questions which required the expert to give information on:

- Estimation approach employed
- Job costing system used
- Average time it took the expert to quote a new or old job
- Average due date and budget conformance
- Key parameters used in quoting

Each interview lasted for approximately 45 minutes with the interviewee taking the researcher through files and estimation platforms.

4. RESULTS

4.1 Firm demographics

The five firms observed are injection mould design specialists within the Western Cape Province Tool, Die and Mould-making (TDM) and Plastics manufacturing industry. The mean number of employees for the observed population of companies was 10.4 with a mean annual income of less than 10 million ZAR. Hence all the firms observed are classified as Small, Medium to Micro Enterprises (SMMEs) according to the classifications defined in South African National Credit Regulator by Mahembe [17]. The specific number of toolmakers in each of the visited firms is shown in Figure 3.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 3: Number of employees in the tool room

Companies A, B, C and E are stand-alone Tooling firms while company D is a tool-room serving in-house operations of a larger organization.

4.2 Cost Estimation approaches and systems employed

Each of the visited firms was questioned on the job estimation processes and policies they adopted. Table 3 summarizes the job quoting approaches by the visited firms.

Table 3: Job quoting approaches						
Expert from Company	A	В	С	D	Е	
Job quoting approach	Geometric- based; parametric	Intuitive; Analytical	Intuitive; Analytical	Geometric- based; Parametric	Geometric- based; Parametric	
Software system	Excel	None	None	Excel	Excel	
Manual or Computerized	Computerized	Manual (file system)	Manual (no system)	Computerized	Computerized with back-up filing system	
Number of quoting experts	1	1	2	2	1	
Experience of the expert (years)	19	28	23	40	33	

Tal	ole	3:	Job	quoti	ng	appr	oac	hes

Three out of the five firms visited utilized an excel-based system for quoting based on the parametric and geometric based approach with the remaining two using the intuitive method. It was interesting to note that the firms using the intuitive approach did not have any historical records of jobs done or record keeping system.

4.3 Cost Estimation lead times, due date and budget conformance

Since prompt quoting of jobs is paramount importance, the survey also sought to investigate the average time the experts interviewed spent quoting both old and new jobs. Furthermore, the respondents were given the opportunity to cite reasons for delays and failure to end the project within budget. Cost Estimators from companies A and E had the lowest lead times in quoting a job while Estimators from Company C admitted to struggle in sending a customer invoice early as shown in Table 4.

Table 4. Job quoting lead times

Company	A	В	C	D	E	
Approximate number of days (New jobs)	2 days	3 - 5 days	7 - 14 days	3 - 5 days	2 days	
Approximate number of Days (Old jobs)	Less than 24 hours	2 days	3 - 5 days	1 day	Less than 24 hours	



All firms visited confirmed that the quote generated would never be accurate. In some cases, the firms experienced a profit or a loss. When queried on the major reasons for huge deviations between the quoted price and the actual cost of the job, the reasons shown in Table 5 were stated.

Expert from Company	A	В	С	D	E
Reason for failure to conform to budget	Excessive Labour hours	Inaccuracy in operation times estimation (labour hours)	Longer Operation times	 Wrong quote Customer specification changes Wrong quote Planning inefficiencies Labour costs 	Overtime

Table 5. Reasons for budget non-conform	mance

The main reason stated for budget non-conformance and delays was a failure to control labour time so that it is within the plan. However, the respondent from company D was honest to highlight that at times non-conformance meant the quote was wrongly prepared in the first place.

4.4 Derivation and classification parameters

The greater part of the time spent during the visits was on identification of the key parameters used when quoting a job. Each expert was presented with an open-ended question to specify what the parameters were. To validate the responses given, each expert demonstrated with an example of a previously quoted job. The main cost elements identified were material, labour and machining costs as shown in Table 6. However, in most cases the experts broke down the cost elements to smallest elements possible. Experts from Company E had a more elaborate system which specified the parameters in the strictest detail possible. This eventually makes the cost estimation process of injection moulds a multi-criteria decision making process, hence justifying the need to use the AHP procedure to rank each of the cost elements according to their contribution to the total cost of an injection mould job.



Table 6: Derived costing parameters					
Expert(s) from company	Cost Parameters derived				
	• Material				
	Cooling system				
٨	Machining				
2	Labour				
	 Tooling cutters 				
	Outsourcing				
	• Material				
	Labour				
В	Heat treatment				
Б	Design				
	 Subcontracting 				
	Outsourcing				
	• Material				
ſ	Labour				
c	Consumables				
	Heat treatment				
	• Material				
D	Labour				
	Heat treatment				
	Machining				
	Material and services				
	✓ Mould base				
	 Copper (electrodes) 				
	 Cavity core steel Heat treating and plating 				
	Manifold bot runner system				
	\checkmark Components				
	✓ Texture				
	✓ Tooling (cutters and inserts)				
E	✓ Sample material				
	✓ Tooling delivery				
	Labour				
	✓ Data management				
	✓ Mould design				
	 Computer simulations 				
	✓ Advanced engineering				
	✓ CNC machining				
	✓ General shop labour				
	✓ CMM parts				

Based on the data collected and presented in Table 6, the costs incurred during injection mould manufacture can be broadly classified as:

- ✓ Material costs
- ✓ Labour and machining costs
- ✓ Services (outsourcing)

4.5 Ranking of Parameters - Analytic Hierarchy Process

4.5.1 **Problem definition**

In the costing of an injection mould, the key parameters to be considered are the material, labour, machining and service cost elements as depicted in Figure 4 below. These elements are further decomposed into the common constituent elements as shown in Figures 5, 6 and 7.





Figure 4: Injection mould cost top hierarchy

The material cost can be further broken down into the costs incurred in securing the mould base, copper electrodes, cavity steel core, hot runner system and the tooling components as illustrated in Figure 5.



Figure 5: Material cost sub hierarchy

Simillarly, the labour and service cost components can be further broken down as illustrated in Figures 6 and 7 respectively.



Figure 6: Labour cost sub hierarchy





Figure 7: Service and outsourcing cost sub hierarchy

4.5.2 Parameter ranking

The methodology employed also included ranking of the parameters determined in terms of their contribution to the overall cost of a job. The purpose of the exercise was to identify the cost elements requiring careful estimation and stricter monitoring during the production phase since deviations carry huge consequences. To achieve this, three stages were followed which included:

1. Pair-wise comparisons

The cost estimator (decision maker) evaluates the relationships between the first level criteria i.e. the material, labour and service cost components. Questions were asked in the following format: "How important are labour costs for a product to the material costs when it comes to contribution to the overall cost of the job?" Based on the framework given in Table 1, on a scale of 1 - 9, a pairwise comparison value is assigned. Given that the L represents the labour costs, M the material costs and S the service costs, the pairwise comparisons derived are shown in Table 7.

	L	Μ	S			
L	1	6	9			
M		1	4			
S			1			

Table 7: Pairwise comparisons

All the experts interviewed identified labour costs as the most critical cost component requiring stricter reckoning, monitoring and control. One of the interviewees specifically stated that labour costs would always account for approximately 55 - 60% of the total mould cost. Hence, based on this analysis, labour-related costs were deemed to be having extreme importance compared to service related costs. Labour-related costs have a strong plus importance as compared to material-related costs while the cost of material is moderate plus with respect to service-related costs. The cost of the hot runner system and the mould base makes the material related costs relatively high in comparison to the service related costs.

2. Development of a pair-wise matrix

The pair-wise comparison matrix is completed by putting the inverse of the scales for the inverse relationships as shown in the Tables 8a and 8b respectively.

Table 8a: I	Pairwise	comparison	matrix

	L	Μ	S
L	1	6	9
M	1/6	1	4
S	1/9	1⁄4	1

Table	8b:	Column	totals

	L	Μ	S
L	1	6	9
Μ	0.1667	1	4
S	0.1111	0.2500	1
Column totals	1.2778	7.2500	14



3. Priority determination and ranking

The pair-wise comparison matrix is normalized by dividing the values in each column by the column sum as shown in Table 9. To determine the priorities of each criterion, the normalized principal Eigen vector is obtained by averaging across the rows.

	L	Μ	S	Priority	Rank		
L	0.7826	0.8276	0.6429	0.7510	1		
Μ	0.1305	0.1379	0.2857	0.1847	2		
S	0.0870	0.0345	0.0714	0.0643	3		

Table	g٠	Normalized	table
Tuble		normatizeu	LUDIC

Row averages = $\begin{pmatrix} 0.7510\\ 0.1847\\ 0.0643 \end{pmatrix}$

Hence, based on the results, Labour costs are ranked first with material and service costs ranked second and third respectively.

4. Checking for consistency

The final step involves checking for the consistency of the solution by determining the Consistency Index as shown below:

Weighted sums = $\begin{pmatrix} (0.7510 \times 1) + (0.1847 \times 6) + (0.0643 \times 9) \\ (0.7510 \times 0.1667) + (0.1847 \times 1) + (0.0643 \times 4) \\ (0.7510 \times 0.111) + (0.1847 \times 0.25) + (0.0643 \times 1) \end{pmatrix} = \begin{pmatrix} 2.4379 \\ 0.5671 \\ 0.193911 \end{pmatrix}$ Consistency vector = $\begin{pmatrix} 2.4379/0.7510 \\ 0.5671/0.1847 \\ 0.193911/0.0643 \end{pmatrix} = \begin{pmatrix} 3.2462 \\ 3.0704 \\ 3.0157 \end{pmatrix}$ $\lambda_{max} = \frac{(3.2462 + 3.0704 + 3.0157)}{3} = 3.111$ Using equation (1), $CI = \frac{\lambda_{max} - n}{n-1} = \frac{(3.1111 - 3)}{(3-1)} = 0.055$

Hence using equation (2) the consistency ratio, $CR = \frac{CI}{RI} = \frac{0.055}{0.58} = 0.096 = 9.6 \% < 10\%$

Since the CR < 10% the judgement can be accepted.

5. CONCLUSION

Cost Estimation is a crucial stage in the injection moulding industry. This process is handled by highly skilled inhouse experts. The paper helped derive the main parameters Cost Estimators in the Western Cape Province Tooling industry use when quoting jobs. The AHP method ranked the parameters and labour-related costs were identified as the most significant cost element when it comes to contribution to total costs. Hence, proper monitoring and control of the production shop-floor during injection mould manufacture is key to due date and budget conformance. The results of this study can be used in the development of Costing expert Systems for toolmakers who specialize in Injection mould manufacture.

6. DECLARATION

This paper was done in partial fulfilment of a Ph.D. study in Industrial Engineering on the Development of a Holonic Control System for the Tool, Die and Mould-making Industry in South Africa conducted by Mncedisi Trinity Dewa.



ACKNOWLEDGEMENT

The financial assistance of the National Research Foundation (NRF) towards this research is hereby acknowledged. Opinions expressed and conclusions arrived at, are those of the author and are not necessarily to be attributed to the NRF.

REFERENCES

- [1] Menezes, J. 2004. European Mould Making: Towards a New Competitive Positioning, in "Proceedings of the 4th International Colloquium Tool and Die Making for the Future", Apprimus, Aachen, pp. 24.
- [2] Instimbi National Tooling Initiative. 2013. Annual Report 2012 2013. pp. 5.
- [3] Malherbe, D.C., 2007.Benchmarking in the South African Tool and die manufacturing sector Master's Thesis Report, Stellenbosch University.
- [4] Dewa, M.T., Van der Merwe, A.F., and Matope, S., 2015. Towards a competitive South African Tooling Industry. International Journal of Social, Behavioural, Educational, Business and Industrial Engineering, Vol. 9:11, 2015, pp. 3533-3538.
- [5] Rittstieg, M. and Garms, F. 2011. South African tooling sector a way to international competitiveness. Proceedings of AfriMold conference, Johannesburg, South Africa.
- [6] Jones, P. 2009. Budgeting, costing and estimating for the injection moulding industry, pp. 161.
- [7] Fonseca, M., Henriques, E., Ferreira, A., Jorge, J. 2007. Assisting mould Quotation though Retrieval of Similar Data, Digital Enterprise Technology, , Session 5, pp. 527-534.
- [8] Duverlie, P. and Castelain, J. M. 1999. Cost Estimation during Design Step:Parametric Method versus Case Based Reasoning Method, The International Journal of Advanced Manufacturing Technology, 1512, pp. 895-906.
- [9] Denkena, B., Lorenzen, L.,E., Schürmeyer, J. 2009. Rule-based Quotation Costing of Pressure Die Casting Moulds, Production Engineering, 31, pp. 87-94.
- [10] Saaty TL. 1980. The analytic hierarchy process : planning, priority setting, resource allocation. New York, N.Y.: New York, N.Y. : McGraw-Hill International.
- [11] Saaty, T.L. 1994. How to make a decision: The Analytic Hierarchy Process. Interfaces, Vol.24, pp. 19 43.
- [12] Saaty, T.L. 2008. Decision making with the analytical hierarchy process, International Journal of Services Sciences, 1(1), pp. 83 93.
- [13] Brugha, C. 2004. Structure of multi-criteria decision-making. Journal of the Operational Research Society
- [14] Ishizaka, A. 2004. The advantages of clusters in AHP. In the 15th Mini-Euro conference, MUDSM.
- [15] Saaty T. L. 1977. A Scaling Method for Priorities in Hierarchial Structures, Journal of mathematical Psychology, 15, pp. 57 68.
- [16] Lucas, P. j. F. and van der Gaag, L. C., 1991. Principles of Expert Systems. Centre for mathematics and Computer sciences, Addison-Wesley, pp. 81.
- [17] Mahembe, E. 2011. Literature review on Small and Medium Enterprises' Access to credit and support in South Africa. *National Credit Regulator*, pp. 1 92.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



DEVELOPMENT OF A HEALTH REHABILITATION ACTIVITY MONITORING SYSTEM FOR TRANSTIBIAL AMPUTEES.

T. Garikayi^{1*}, D. Van den Heever² and S. Matope³

^{1,2}Biomedical Engineering Research Group Department of Mechanical and Mechatronic Engineering University of Stellenbosch, South Africa talon@sun.ac.za, dawie@sun.ac.za

³Department of Industrial Engineering University of Stellenbosch, South Africa <u>talongarikayi@gmail.com</u>, <u>smatope@sun.ac.za</u>

ABSTRACT

This paper presents a novel approach for a remote activity and mobility rehabilitation monitoring system for transitibial amputees. The proposed system has the capability of monitoring ambulatory and mobility related activities such as standing, walking, sitting and walking on stairs. The design utilizes single channel inertial measurement signals to determine amputee performance. The methods involved in the system design include development of an embedded data acquisition system, data storage system, signal processing algorithm and the user interface. The final module of the system includes a data processing and presentation algorithm for clinical applications. Design parameters were based on normal human gait. The hardware and software for context-aware ubiquitous computing applications was developed. Subjects were tasked to perform random activities for three hours and the data was collected and analyzed. The system classified motions as walking, standing, laying, sitting and walking on stairs. The classification accuracy was 93.3%, due to challenges on classifying between walking upstairs and walking downstairs. The system error was 6.7%.

Keywords: Transtibial; amputation; mHealth; inertial, classification

¹ Author is enrolled for a PhD (Mechatronic Engineering) degree in the Department of Mechanical and Mechatronic Engineering, University of Stellenbosch



1 INTRODUCTION

Transtibial (below-knee) amputees fitted with prosthetic limbs often face difficulties due to the loading of the amputated site, boredom, lack of exercise and stress imposed by their new reality. Transtibial amputees often have the capability to regain normal movement more readily due to the availability of the intact knee which offers easy movement. However, due to the load bearing characteristics of the prosthetic limb, the amputees often find it difficult to accept the use of artificial limbs as compared to the use of clutches. Regardless of the type of surgical procedure used for amputation there is always severe pain experienced during rehabilitation thereby affecting the gait of the amputee. In some cases this has resulted in total rejection of the limb. During the immediate post-surgical procedure, prosthetists focus on minimizing swelling and preventing muscle flexion contractures. As recovery progress, focus is shifted on resuming normal activities and hence introduces the use of prosthetic limbs. The physician assists the amputee during the initial days of rehabilitation. This include training the amputee on the possible activities and movements that can be achieved using the fitted prosthetic limb. It is however difficult to determine the usability of the prosthetic limb by simple visual inspection as the amputee performs the predetermined activities under the guidance of the physician in the laboratory. Due to the desire to be discharged and the will to perform normal activities in society, the patient can easily set his mind to achieve the expected results. However, this will be different as compared to when the amputee performs daily mobility activities at home. Lack of exercises and reduced daily mobility at home may lead to increased weight gain thereby increasing the load bearing of the artificial limb and ultimate rejection of the artificial limb. This is even worse for an amputee whose cause of amputation was sugar diabetes as lack of mobility increases the effects of the disease. There are more than 300 000 amputees in South Africa and 60% of lower limb amputees in South Africa are as a result of sugar diabetes. In the United States of America there is an average of 133 735 new amputees per year [1]. Therefore one in every 200 Americans has received an amputation of which lower limb amputation accounts for 86% [2].

Biomechanics laboratories equipped with very expensive equipment such as optical motion tracking systems and imbedded force plates mainly focus on tracking human motion and performing dynamic analysis of physical behavior of individual body parts [3]. Such environments are not convenient for continuous patient monitoring. It is well known that the behavior of a patient in a laboratory is quite different compared to when the patient is at home [4]. Therefore there is a need to acquire data on how the amputee utilizes the prosthetic limb on a daily basis at home or at work. The data can then be used to improve prosthetic designs and motivate the amputees to live a good healthy lifestyle.

The objective of the proposed design is to determine the usability and acceptability of the prosthetic limb by an amputee by determining the type and extend of activities carried out by amputees during rehabilitation under minimum to no supervision. The proposed design is categorized as a remote mobile activity monitoring tool due to its capability of being integrated into Mobile Health Systems on information recording, conveyance and processing capabilities as a remote health care monitoring tool.

The paper unfolds as follows: section 2 presents an overview of mHealth System, section 3 presents different types of signals that could be harnessed from the limb, section 4 presents the design considerations, section 5 presents the development of data acquisition unit, section 6 presents the results of the design and section 7 concludes the research.

2 OVERVIEW OF m-HEALTH

The world currently revolve around Internet-Of-Things (IoT), however, the applicability of IoT based technologies such as e-Health in developing countries is still a challenge as the accessibility of the internet is still low. The increased usage of smart phones have however facilitated the rapid increase in the use of mHealth (Mobile Health) systems. These smartphones can also be used to acquire, receive and send data. The three fundamental issues on the applicability of mHealth technology are acceptability, applicability and mostly adaptability [5]. The technology is mainly evaluated on its appropriateness to the task at hand and the user rather than by its modernity or ease-in-use for developers. The objectives of mHealth technology for amputee rehabilitation include promoting healthy lifestyles, improving decision making capabilities of health professionals as well as patients, and enhance healthcare quality by improving access to medical and health information. mHealth systems also facilitates instantaneous communication in places where this was not previously possible. Centralised health care facilities, limited experts and poor infrastructure has been the driving factors for the implementation of mHealth systems in developing nations. The mHealth system has been around since the inception of telemedicine, however the application of mHealth had been limited due to the availability of smart phones and telecommunication infrastructure. The rapid increase in telecommunication operators and improved network coverage supported by decrease in smart phone cost has resulted in the sudden growth in mHealth systems. For almost a decade mHealth has been utilized within health care systems for treatment compliance, data collection and disease surveillance, health information systems and point-of-care support, health disease prevention and emergency medical response. There are several advantages of utilizing mHealth systems for amputee patient rehabilitation, these include [6]:



- Provide platform for connecting the amputees and the physicians in remote areas
- Provide a reliable platform for early diagnosis and treatment of amputation based diseases
- Provide real-time data acquisition for online and offline analysis
- Decrease time gap between data collection in the field and viewing of data.

3 METHODS AND MATERIALS

According to [7], technology integration is divided into two general frames: *processes* and *outcomes* that lead to development. The processes are then divided into data acquisition, processing and communication. The systematic procedure used to achieve the objective of using mHealth to monitor transitibial amputees was as follows:

- An electronic hardware system with the IMU sensors and the processor was developed
- An inertial measurement unit was mounted on the artificial prosthetic limb
- A data acquisition algorithm was developed and deployed
- On-board filtering was implemented as firmware for embedded sensors
- Sensor fusion technique was implemented for IMU sensor
- Data from the IMU sensors were stored in an SD Card as text files
- The acquired data was preprocessed using Matlab by applying the low pass and high pass filters
- The filtered signal was then used to determine a feature set
- Activity recognition was then determined using a classifier

A data set was collected from five normal subjects. Each subject was asked to perform the six activities (standing, sitting, laying, walking, walking upstairs and walking downstairs).Each activity was performed 10 times for a period of five minutes per activity. During the testing phase the activities were performed in a random order for a duration of three hours. The data was collected using an SD card located within the signal acquisition circuit. The training data was labeled according to the respective activity being carried out. After developing, training and testing the system, raw data was used to test for system classification accuracy.

4 DATA ACQUISITION SYSTEM

The reliability of a mHealth system is largely dependent on the quality of data acquired. The main goal of the data acquisition module is to record and store the raw data for further processing by the physicians. However, there are several factors that affect the quality of recorded data. These include sensor calibration, sensor positioning and general environmental condition. According to [8], knee rotation angle during walking can be determined using two gyroscopes one at the thigh and another one at the shank. However bulky designs suffer from frequency detection errors and make the user feel uncomfortable. The proposed system utilizes only one IMU unit which was mounted below the knee. Sensor fusion techniques have been the key aspect of robust data acquisition hence the combination of accelerometer, gyroscope and magnetometer signals in this proposed device provided a more reliable data acquisition system. The Python algorithm was used to acquire the desired signals and stored as a text file on a SD card.

4.1 Inertial measurement unit (IMU)

The use of a single IMU unit largely depend on the level of intelligence of the algorithm implemented for data acquisition. Most amputees would not be comfortable wearing a lot of sensors on their body during rehabilitation therefore a 3 axis accelerometer, 3 axis gyroscope and a 3 axis magnetometer were used along with a temperature sensor as a 9 degrees of freedom single IMU unit. The advantage of using the microelectromechanical systems (MEMS) is that the devices are small in package size, low power consumption, high accuracy, high repeatability and high shock tolerance. Their application specific performance programmability present an avenue for easy integration with existing technologies. In this regard a MP6050 model was used and the orientation of the axes of sensitivity and the polarity of rotation is shown in figure 1.





Figure 1: Orientation of Axes of Sensitivity and Polarity of Rotation

The objective is to acquire the accelerometer, gyroscope and magnetometer values so as to formulate a data set which can easily be used for activity recognition. To achieve such a data set there is a requirement for proper filtering and sensor fusion techniques. The design made use of arbitrarily defined zero-identity orientations which makes perfect physical alignment unnecessary and reduces the difficulty in extracting desired outputs. The sensor was equipped with an internal Digital Motion Processor (DMP) which allows 3D motion processing algorithms. The human body is not uniform and thus presents the possibility of imperfect sensor placement, therefore the quaternion orientation output and the quaternion operations were developed and deployed to account for the human body's irregularities thereby achieving accurate forward and downward vectors. The algorithm for detection of angles also incorporated the Kalman filtering technique.

A quaternion, q, is a fourth dimensional vector that can be interpreted as a third dimensional rotation:

$$q = (ix, jy, kz, w)$$
(1)

$$\boldsymbol{q} = \left(\boldsymbol{u}\sin\left(\frac{\theta}{2}\right),\cos\left(\frac{\theta}{2}\right)\right) \tag{2}$$

$$\boldsymbol{q} = (\boldsymbol{v}, \boldsymbol{w}) \tag{3}$$

Where u is a vector defined as: $\mathbf{u} = (\mathbf{i}, \mathbf{j}, \mathbf{k})$ and $\boldsymbol{\theta}$ is the measured angle.

To compensate for the misalignment and sensor placement problems, the gravity vectors of the sensor were considered since the gravity vector can align with one of the Cartesian coordinate axes of the device reference frame when in the calibration mode (starting position). Assuming that **g** is the expected gravity vector, g_d is the gravity vector from the device and **a** is the unit vector that denotes the axis of rotation from g_d to **g** and **q**₀ is the rotational offset as a quaternion, then equation (3) is true.

$$\theta = \arccos(\boldsymbol{g}_{\boldsymbol{d}}, \boldsymbol{g}) \tag{4}$$

$$\boldsymbol{a} = \boldsymbol{g}_d \times \boldsymbol{g} \tag{5}$$

$$\boldsymbol{q}_{0} = (\boldsymbol{a}\sin\left(-\frac{\theta}{2}\right), \cos\left(-\frac{\theta}{2}\right)) \tag{6}$$

The ultimate goal was to offset g_d so as to line up with \mathbf{g} , this was achieved through the product of the filtering orientation and the rotational offset, \mathbf{q}_0 and the result was the tare orientation of the sensor. A *tare* orientation is the use of a certain orientation for the zero orientation instead of the orientation indicated by the reference vectors. Finally the forward vector of the sensor, v_F , was determined as in equation (7) taking into consideration the filtered tare orientation of the device, \mathbf{q}_t .

$$v_F = q_t q_0 v \tag{7}$$

Where v is the unrotated forward vector in the sensor's space. However the determination of the down vector, v_D is almost similar to the forward vector hence the up-vector, v_U is the negative vector of the down vector as shown in equations (8) and (9).

$$\boldsymbol{v}_{\boldsymbol{D}} = \boldsymbol{q}_t \boldsymbol{q}_0 \boldsymbol{v} \tag{8}$$



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

$$v_U = -v_D$$

(9)

4.2 Measurements

The determination of quantitative information for analysis was based on the standard normal gait analysis for humans. The quantities required by most physicians for the design to be clinically viable are *when*, *how long and how often* and *what* activity was being carried out as related to ambulatory and mobility activities. These can be translated to *walking*, *sitting*, *running*, *standing*, *walking upstairs and walking downstairs*. These activities can be determined by monitoring the accelerometer, gyroscope, and magnetometer readings to determine the roll, pitch and yaw of the limb. However, an inverted tilt measurement was also tested. Experiments were carried out using the Vicon Camera System to determine the joint angles of the lower limb during several activities with an emphasis on walking, sitting, running and standing. These values were used as design specifications. The acquired signals were then used to determine the design parameters and validation of the proposed design. The signals from the accelerometer, the gyroscope and the magnetometer were tested for variability during each individual activity. The data set contained the following information:

- Roll,Ø
- Pitch , θ
- Yaw, ψ
- Kalman Filtered Values (kalX, kalY, kalZ)
- Complimentary Filtered Values (compX, compY, compZ)
- Raw IMU data values (accX, accY, accZ) and (gyroX, gyroY, gyroZ)

It should however be noted that only the signal features from the above mentioned parameters were used as input data to the classifier and not the raw signal. The roll and pitch were calculated using the accelerometer while yaw was calculated using the magnetometer. The accelerometer had a full scale of $\pm 2g$ and a sensitivity of 16384 LSB/g. The gyroscope had a full scale of ± 250 °/s and a sensitivity of 131 LSB/ (°/s). The greatest set back with the use of IMUs is the sensor misalignment considerations, this is mainly defined as the angular difference between the sensor's axis of rotation and the system defined inertial reference frame (global frame). The roll, pitch and yaw rotation matrices can transform a vector such as gravitational field vector **g** under a rotation of the coordinate system about x, y and z resulting in R_x, R_y and R_z respectively.

$$R_{x}(\emptyset) = \begin{bmatrix} 1 & 0 & 0 \\ 0 & \cos \emptyset & \sin \emptyset \\ 0 & -\sin \emptyset & \cos \emptyset \end{bmatrix}$$
(10)
$$R_{y}(\theta) = \begin{bmatrix} \cos \theta & 0 & -\sin \theta \\ 0 & 1 & 0 \end{bmatrix}$$
(11)

$$R_{z}(\varphi) = \begin{bmatrix} \cos\varphi & \sin\varphi & 0\\ -\sin\varphi & \cos\varphi & 0\\ 0 & 0 & 1 \end{bmatrix}$$
(12)

However the yaw, $R_z(\varphi)$ is difficult to use as the accelerometer's z-axis is aligned to the earth's gravitational field. Therefore the roll and the pitch were more relevant.

4.3 Determination of the roll and the pitch

The roll and the pitch are also useful parameters which can be determined by using a single IMU sensor. In this regard raw accelerometer values were recorded and the roll and pitch was determined using equation (16) and (19) respectively. The Kalman filter and the complimentary filter were both applied to the signal under restricted pitch and unrestricted pitch conditions and the results are illustrated in figure 5. If G_p is the normalised accelerometer reading related to the roll (\emptyset) and pitch (θ) angles, then,

$$\frac{G_p}{\|G_p\|} = \begin{bmatrix} -\sin\theta \\ \cos\theta\sin\phi \\ \cos\theta\cos\phi \end{bmatrix}$$
(13)

$$\frac{1}{\sqrt{kal_x^2 + kal_z^2}} \begin{pmatrix} kal_x \\ kal_y \\ kal_z \end{pmatrix} = \begin{bmatrix} -\sin\theta \\ \cos\theta\sin\phi \\ \cos\theta\cos\phi \end{bmatrix}$$
(14)

$$\tan\phi_{xyz} = \left(\frac{kal_y}{kal_z}\right) \tag{15}$$



$$roll, \phi = actan\left(\frac{kal_y}{kal_z}\right) \tag{16}$$

$$tan\theta_{xyz} = \left(\frac{-kal_x}{kal_y sin\phi + kal_z cos\phi}\right) \tag{17}$$

$$\tan\theta_{xyz} = \frac{-kal_x}{\sqrt{(kal_y^2 + kal_z^2)}} \tag{18}$$

$$pitch, \theta = -actan(\frac{-kal_x}{\sqrt{(kal_y^2 + kal_z^2)}})$$
(19)

Where kal_x , kal_y and kal_z are outputs from the Kalman filter. However, the same roll and pitch can be determined using the *accX*, *accY*, *accZ* values which are the raw signals from the IMU before the application of either the Kalman filter or the complimentary filter.

4.4 Determination of tilt, ρ

The tilt angle, ρ of the lower limb from the original position when the subject is seated or standing in the absence of linear acceleration can be determined. The angle ρ , is between the gravitational vector measured by the accelerometer and the initial orientation with the gravitational field pointing downwards along the z-axis as shown in figure 2. In the algorithm this is used to determine the standing and seating position, hence distinguishing leg swing from a walking activity.



Figure 2: Tilt angle from vertical position.

Taking into consideration the accelerometer reading, $G_p(acc_x, acc_y, acc_z)$ in the absence of linear acceleration, the tilt angle is determined as shown in equation (20).

$$G_p \begin{bmatrix} 0\\0\\1 \end{bmatrix} = acc_z \tag{20}$$

$$G_z = |G_p| \cos\rho \tag{21}$$

$$cos\rho = \frac{acc_z}{\sqrt{(acc_x^2 + acc_y^2 + acc_z^2)}}$$
(22)

$$\rho = \arccos(\frac{acc_z}{\sqrt{(acc_x^2 + acc_y^2 + acc_z^2)}})$$
(23)

The algorithm was developed using Python to determine the roll, pitch and yaw angles. The minor changes of the mathematical expressions within the algorithm were handled by the libraries. Major challenges were on calibration and offset value detection.

5 ACTIVITY RECOGNITION

The activities were identified using both a threshold technique and pattern recognition system. According to [9], the clinical viability of the proposed system is largely dependent on the underlying recognition module's ability to detect a variety of activities that are performed routinely in many different manners by different individuals under different environmental condition. This however presents a bottleneck as the real world data is noisy and complex. The raw data acquired from the accelerometers, gyroscopes and magnetometer were stored as text files. Matlab was then used to filter the values so as to remove noise interference and consider only signals due to limb movement. Low pass filters and high pass filters were implemented in the algorithm. The pre-processing



stage was done so as to reduce the processing time and minimize the computational complexity of the system. The signal acquisition algorithm had several outputs which included raw accelerometer and gyroscope readings before applying the Kalman and Complimentary filter, magnetometer readings, roll, pitch and yaw readings. The data set had 21 signals. This provided a large data set for activity recognition. However only a set of features were used to formulate a feature set which was then used as an input to the classifier as shown in figure 3.



Figure 3: The Activity recognition Module

Although there are several signal features that can be used by signal classifiers [10], the most common ones are integrated EMG (IEMG), mean absolute value (MAV), modified mean average value (MMAV), variance of EMG (VAR), waveform length (WL) and Wilson amplitude (WAMP).

$$IEMG = \sum_{n=1}^{N} |h_n| \tag{24}$$

$$MAV = \frac{1}{N} \sum_{n=1}^{N} |h_n| W(x) = \begin{cases} 1, \ 0.25N \le n \le 0.75N \\ 0, otherwise \end{cases}$$
(25)

$$MMAV = \frac{1}{N} \sum_{n=1}^{N} |h_n| W_n \tag{26}$$

$$VAR = \frac{1}{N-1} \sum_{n=1}^{N} h_n^2$$
(27)

$$WL = \sum_{n=1}^{N-1} |h_{n+1} - h_n|$$
(28)

$$WAMP = \sum_{n=1}^{N-1} f(|h_{n+1} - h_n|) f(x) = \begin{cases} 1, & x \ge threshold \\ 0.5, & otherwise \end{cases}$$
(29)

Simple Square Integral,
$$SSI = \sum_{n=1}^{N} |h|^2$$
 (30)

Root Mean Square,
$$rms = \sqrt{\frac{1}{N}\sum_{n=1}^{N}h^2}$$
 (31)

$$\log detector, LOG = e^{\frac{1}{N}\sum_{n=1}^{N} \log|h_n|}$$
(32)

slope sign change,
$$SSC = \sum_{n=1}^{N} f \left[(h_n - h_{n-1}) \times (h_n - h_{n+1}) \right]$$
, $f(x) = \begin{cases} 1, & \text{if } h \ge threshold \\ 0, & \text{otherwise} \end{cases}$ (33)

These features formulated the input signal to the classifier. Each activity resulted in a unique feature vector. Therefore it was easy for the classifier to distinguish the activities.

6 SYSTEM ARCHITECTURE

The proposed system architecture consist of an IMU with an embedded Digital Motion ProcessorTM (DPM). The DPM provides an added advantage on an onboard processing unit before the signals are send to the main processing unit (MPU). This in turn reduces the computational time and complexity of the motion algorithms. The IMU and the microprocessor used I2C communication protocol thereby enabling a sensor fusion for all the four forms of signals from the accelerometer, gyroscope, magnetometer and the temperature sensor. The temperature sensor is inbuilt for monitoring the operational temperature of the device as shown in figure 4.





Figure 4: Proposed System Architecture

The main processing unit has a Bluetooth module which allows the uploading and downloading of files onto the personal computer where the GUI is resident. If internet is available on a mobile phone the files can be accessed using third party software. However, for those in remote areas where internet accessibility is limited or not present the SD card can be removed from the device and the information can be send via text file (data.txt).

7 RESULTS

The results presented in this section were recorded after calibration of the sensor so as to reduce the errors from offset and sensitivity mismatch. Tilt was determined by direct integral of the output of the gyroscope, however the error associated with the null bias stability affected the results as the signal period increased. This caused an apparent rotation even when the observed subject was stationary. Therefore this is not a recommendable method for a signal to be recorded over a period of hours or days. As a result the inclination angle was determined using accelerometer values under static conditions. The results were used to determine the movement of the lower limb when the amputee (subject) is seated or standing as illustrated in figure 5. Standing would produce a constant and almost steady acceleration at $10ms^{-2}$ in the x-axis (a_x), while the walking activity produces an oscillating signal around $10ms^{-2}$. Positions that can be monitored for such activity are the knee angle and the ankle angle during (dorsiflexion or plantarflexion) under static conditions.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 5: Standing vs Walking Accelerometer values (Where estimated resembles the labeled data points and actual is the real time performance of the classifier)

The gyroscope had notable drift under small periods, the features from the gyroscope was fused with accelerometer signals so as to reduce errors associated with the drift. The movement of the legs during walking or climbing stairs could both not provide convincing information for the adoption of raw gyroscope readings, as a result this reduces the classification accuracy. The accelerometer returned true values when the acceleration was progressive but suffered greatly from vibrations of the system. This has necessitated the implementation of the Kalman filter and complimentary filter as techniques to extract a useful signal. The two filters were both tested under the same conditions and the results are presented in figure 6.



Figure 6: Comparison Kalman Filter and Complimentary Filter (Walking Activity)

The results show that the Kalman filter provides a smooth and useful signal with a steady state error of zero. The Kalman filter is an optimal estimator [11]. It infers parameters of interest from indirect, inaccurate and uncertain observations. The underlying principle is that if all noise is Gausian, the Kalman filter minimizes the mean square error of the estimated parameters. It is regarded as the best tool to achieve optimal estimates for linear systems through online real time processing [12]. The implementation was based on the fact that prior knowledge was not necessary as the filter relies on the last state and covariance matrix that defines the probability of the sate being correct. The strength of the Kalman filter is on adaptation and robust sensor fusion through the use of the prediction and the update equations [13]. The prediction equations predict current position of the artificial limb based on the previous position (state). The update equations check for IMU outputs reliability and how reliable the predicted state is as illustrated in figure 7.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 7: The Kalman Filter algorithm

Where x and X- represents the state (accX, accY, accZ) of the limb (signals to be filtered), U is the action or the filtered signal (roll, yaw, pitch) (kalX, kalY, kalZ), P and p- confidence coefficients and it shows how the filter is confident with the solution, it is initiated as a diagonal matrix of the variance, Q is the covariance of the process noise, H is the sensor model and initialized as diagonal identity matrix and adjusted during design, R- indicates the confidents of the sensors (accelerometer and gyroscope), it is the key for sensor fusion, z- these are the outputs or measurements returned from the sensors at each position and estimated by the filter when missing. I is the diagonal identity matrix. A and B are model matrices that represents how the system changes from one state to the other. After initial calibration, the algorithm begin with the current state, x_{k-1} , the initial covariance matrix P_{k-1} and the current input u_{k-1} to get the predicted state X and predicted covariance matrix p. Then the achieved measurement was y_k and use the update equations to correct the initial predictions in order to get the new state matrix x_k and new covariance P_k . The process was then iterated in the algorithm.

The classification system employed the artificial neural network (ANN) with the determined feature set as input and activity as output. There was however a challenge on classifying walking upstairs and walking downstairs. Although the autocorrelation in figure 8 illustrates differences in amplitude, the signal frequency looked very similar. Principle component analysis was one of the feature extraction technique that was implemented so as to distinguish the walking downstairs and walking upstairs activity thereby reducing the classification error.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 8: Autocorrelation Comparison of Walking Upstairs and Walking Downstairs

The autocorrelation was useful for frequency estimation especially for the low-pitch frequencies such as accelerometer values. The magnetometer signal was very difficult to utilize for activity recognition as it was affected largely by the presence of metallic objects. Even after the use of a plastic casing the readings were still affected, it was impossible to use shielded or screened casing. There was however clear distinction between walking on a flat surface and walking upstairs as illustrated on figure 9. The contribution of low frequencies on the walking upstairs as compared to walking on a flat surface was then detectable during feature extraction and used as a distinct feature.



Figure 9: Power spectral density for walking and walking upstairs

The confusion matrix shown in figure 10 illustrates how the system was able to classify the activities. The system achieved 93.3% classification accuracy. The main contribution for the 6.7% error was largely due to classifying between sitting, standing and laying. There was also notable errors on the system on classifying between walking upstairs and walking downstairs. However to a larger extent the system was able to classify all mobility activities.



	Confusion Matrix							
2 2 5 6 6	263 17.0%	8 0.5%	3 0.2%	0 0.0%	0 0.0%	0 0.0%	96.0% 4.0%	
	4 0.3%	228 14.8%	5 0.3%	0 0.0%	0 0.0%	0 0.0%	96.2% 3.8%	
	6 0.4%	3 0.2%	190 12.3%	0 0.0%	0 0.0%	0 0.0%	95.5% 4.5%	
	0 0.0%	0 0.0%	0 0.0%	231 15.0%	35 2.3%	0 0.0%	86.8% 13.2%	
	0 0.0%	0 0.0%	0 0.0%	39 2.5%	254 16.4%	0 0.0%	86.7% 13.3%	
	0 0.0%	0 0.0%	0 0.0%	0 0.0%	0 0.0%	276 17.9%	100% 0.0%	
	96.3% 3.7%	95.4% 4.6%	96.0% 4.0%	85.6% 14.4%	87.9% 12.1%	100% 0.0%	93.3% 6.7%	
	1	2	3	4	5	6	-	
			Та	raet Cla	SS			

Figure 10: Confusion matrix

The confusion matrix shown in figure 10 illustrates that the system can reliably classify the activities. The designed electronic circuit and the proposed system architecture provided reliable information for the development of a Rehabilitation Monitoring System (RMS) for transitibial amputees who have received prosthetic limbs. The system was capable of recording daily activities responsible for mobility. The results can easily be used by clinicians to make informed decisions regarding the usage of the artificial limb. However, the developed system does not show the gait cycle parameters as it is only limited to activity recognition, length of activity and period of when the activity was performed. The activities can easily be visualized as illustrated in figure 11. However, the signal illustrated shows only one channel. The proper signal presentation will be implemented using Qt, which is a software development tool capable of running on several platforms including html, android and windows.



Figure 11: Classified signal

The results for hardware development and signal acquisition were not presented, the paper focused on results for activity recognition which determine the success of the hardware and signal acquisition system. The graphical

2581-12



user interface within Qt will include results such as type of activity, when the activity was done and duration of each activity.

8 CONCLUSION

The development of the mHealth System for rehabilitation monitoring of lower limb amputees was successful to a greater extent. However, there are other areas which we recommend still need attention such as data security as the system is supposed to provide high level security to personal information. The technology concept proved to be cheap and easy to integrate into already existing healthcare monitoring systems. One of the novelties of the system is its machine learning algorithm, sensor fusion and use of single nodal data acquisition system. The use of information from a single nodal point requires high level machine learning language as it is well known that information gained from multimodal sensors can offset the information lost when sensor readings are collected from a single location. However, in our case the need for a single sensor was to reduce the number of components mounted on the artificial limb, reduce system cost and introduce a new technique. The reliability of the decision made when a signal is lost is based on the capability of the Kalman filter to project a signal based on past signal inputs. Recommendations for further improvement include miniaturization of the architecture, full utilization of the embedded DMP, a standalone signal processing and presentation mobile application.

The proposed design will assist on the rapid implementation of mobile health systems in South Africa. The benefits which will be realised on the full roll-out of the system include drug performance monitoring, improved home based care system, reduced queues at public hospitals as Medical Doctors could now assess the patient via the mobile platform and the outbreak of common diseases in some remote areas can easily be detected during the initial stages. The proposed system is an add-on tool to an already existing functional platform. Much of the health care system is carried out by the Healthcare Systems Research Group at Stellenbosch, which focuses on telemedicine and its related technologies.

The on-going research will ultimately result in the development of hardware and software for context-aware ubiquitous computing applications.

9 REFERENCES

- [1] Ziegler-Graham, K., MacKenzie, E.J., Ephraim, P.L., Travison, T.G. and Brookmeyer, R. 2008. Estimating the prevalence of limb loss in the United States: 2005 to 2050. Archives of physical medicine and rehabilitation, 89(3), pp.422-429.
- [2] Adams, P.F., Hendershot, G.E. and Marano, M.A., 1999. Current estimates from the National Health Interview Survey, 1996. Vital and Health Statistics. Series 10, Data from the National Health Survey, (200), pp.1-203.
- [3] Inoue Y., Matsuda T., Shibata K.2003. "Estimation of vertical reaction force and ankle joint moment by using plantar pressure sensor," JSME, Symposium on human dynamics, pp. 57-62.
- [4] Karaulova I.A., Hall P.M., Marshall A.D.2002. "Tracking people in three dimensions using a hierarchical model," Elsevier Science, Image and Vision Computing, Vol.20, pp.691-700.
- [5] Mechael P., Batavia H., Kaonga N.2010. Barriers and gaps affecting mHealth in low and middle income coutnries: Policy white paper. Center for Global Health and EConomic Development Earth Institute, ColumbiaUniversity.
- [6] Hordacre, B., Birks, V., Quinn, S., Barr, C., Patritti, B.L. and Crotty, M. 2013. Physiotherapy Rehabilitation for Individuals with Lower Limb Amputation: A 15-Year Clinical Series. Physiotherapy Research International, 18(2), pp.70-80.
- [7] Heeks, R.2002. i-Development Not e-Development: Special Issue on ICTs and Development. Journal of International Development, 2002. 14(Special Issue: Information Communcation Technologies (ICTs) and Development): p. 1-11.
- [8] Jakob C., Kugler P., Hebenstreit F., Reinfelder S., Jensen U., Schuldhaus D., Lochmann M., and Eskofier B.2013. "Estimation of the knee flexionextension angle during dynamic sport motions using body-worn inertial sensors," in BodyNets, Boston, US.
 [9] Choudhury T., Consolvo S., Harrison B., Hightower J., Lamarca A., LeGrand L., Rahimi A.2008. "The
- [9] Choudhury T., Consolvo S., Harrison B., Hightower J., Lamarca A., LeGrand L., Rahimi A.2008. "The mobile sensing platform: An embedded activity recognition system." Pervasive Computing, IEEE 7, no. 2.pp. 32-41.
- [10] Phinyomark, A.; Limsakul, C. and Phukpattaranont, P.2009. A Novel Feature Extraction for Robust EMG Pattern Recognition, Journal of Computing, ISSN: 2151-9617, Vol.1, No. 1,pp.71-80
- [11] Anderson B. D. O.and J. B. Moore.2005. Optimal Filtering. New York: Dover.
- [12] Kalman, R. E. 1960. "A new approach to linear filtering and prediction problems," J. Basic Eng., vol. 82, no.1, pp. 35-45, Mar.
- [13] **Groves P. D.**2008. Principles of GNSS, Inertial, and Multisensor Integrated Navigation Systems. Norwood, MA: Artech House.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



IMPACT OF MANUFACTURING STRATEGY ON OPERATIONAL PERFORMANCE: A CASE STUDY OF ZIMBABWEAN MANUFACTURING INDUSTRY

B. Chindondo^{1*} and S. Chindondo²

¹Gordon Institute of Business Science University of Pretoria, South Africa <u>bchindondo@gmail.com</u>

²stvmoyo@gmail.com

ABSTRACT

The relationship between manufacturing strategy and operational performance was studied within the ambit of world class manufacturing. The study investigated the extent to which manufacturing strategies were being applied in the case of an alcohol manufacturer in Zimbabwe as well as in the Zimbabwean manufacturing industry as whole. The derivation of manufacturing strategies for successful manufacturers; and the linkage of these strategies to corporate strategies of the firms was assessed. The study evaluated the involvement of manufacturing managers in strategy development; and determined the manufacturing strategies being employed. Data from the alcohol manufacturer and a sample of forty members of the Confederation of Zimbabwean Industries' (CZI) manufacturing firms were used. Ninety percent of the manufacturers had a corporate strategy and operational performance. The alcohol manufacturer had no documented manufacturing strategy but considered two competitive priorities (cost and quality) in planning. The study also revealed that manufacturing firms used a combination of competitive priorities in coming up with manufacturing strategy, with corporate strategy being derived from manufacturing strategy. This study recommends that manufacturing priorities drive the strategic management process necessitating the presence and input of manufacturing technical managers in the strategy formulation process.

¹ Corresponding Author



1 INTRODUCTION

Globalisation has tightened competition in the Zimbabwean manufacturing sector with products from companies across the globe landing in Zimbabwe at almost half the price of the same product produced in Zimbabwe [1] & [2]]. The manufacturing sector is one of the four wealth generating sectors in Zimbabwe [2], contributing 30% to the country's GDP at its peak in the late 90s [3]. With this sector now contributing a meagre 2.3% to Zimbabwe's GDP as of 2012 [2]; the need to strategically position companies in this sector cannot be overemphasized. The bottom-line purpose of strategic planning in the manufacturing sector is to ensure long-term viability and growth, the cornerstones of which are quality to customers, returns to owners, and opportunities for employees. As noted by Platts and Gregory [4], manufacturing systems improvements have traditionally been the responsibility of technical personnel, cost reduction often being the primary aim with little awareness of, or attention being paid to other business factors including quality, flexibility and delivery period. Lacking in sound manufacturing strategies that are consistent with overall business strategies, such manufacturing systems have thus failed to contribute satisfactorily to the competitive position of their companies [4].

Manufacturing is an important component in corporate success, a view that has spurred the development of manufacturing strategies in companies seeking competitive advantage, yet there is a realisation that understanding of manufacturing strategies creating competitive capabilities and profits is weak [5]. According to Ehie [6], manufacturing strategy refers to the specific competencies in the areas of cost, quality, delivery and flexibility that are developed to achieve the competitive advantage in a firm. Traditionally, the manufacturing system has been limited to addressing technical issues of production [7]. However, as early as 1969, Skinner [8] noted that the manufacturing function can and should be employed as a competitive weapon with manufacturing strategy providing the long term guideline to plan, schedule and control value adding activities and decisions that over time, enable a business unit to achieve a desired set of specific capabilities [9]. The link between strategy and performance has been the focus in manufacturing strategy research with a positive relationship between manufacturing strategy and performance being reported [10]. The manufacturing success-business performance relationship is however indirect. To link manufacturing to overall business performance requires management to exploit manufacturing capabilities to operationalize manufacturing strategy [11].

1.1 Competitive priorities and Manufacturing capabilities

The early contributions of manufacturing strategy researchers have highlighted the strategic role of the operations and manufacturing function in establishing firm performance [7]; [8] & [9] Competitive priorities are the attributes of an organization that attract customers; the potential points of differentiation between an organization and its competitors. They define the set of manufacturing objectives and represent the link to market requirements [12]. Researchers consider these competitive priorities to be the link between market requirements and manufacturing [13]. The manufacturing function is expected to implement structural and infrastructural decisions that are consider low cost, quick delivery, flexible designs and superior quality [10]. While some authors suggest innovativeness and service as additional priorities [14] & [15], research and strategy theories consistently stress the four basic dimensions of cost, quality, delivery and flexibility [16].

Operationalization of manufacturing strategy occurs through a pattern of decisions, an observation acknowledging the influence of management on the development and performance of the system. Decisions within the manufacturing functions determine which resources to use and what routines or practices to employ and emphasise in order to achieve the manufacturing objectives. This set of practices, resources, and routines used, ultimately determine the operating characteristics of the manufacturing system, i.e. the manufacturing capabilities that together with competitive priorities influence the firm's manufacturing performance [17]. Thus, manufacturing capabilities can be viewed as the link between manufacturing strategy content and manufacturing performance. Manufacturing strategy has adopted the notion of capabilities from strategic management literature, particularly the resource-based view (RBV) of the firm proposed by Wernerfelt [18] and Barney [19]. The basis in RBV is that resources are not uniformly distributed across firms making the deployment of these resources a potential source of competitive advantage. The resources, to be of competitive advantage, must be assets, routines, or practices controlled by a firm that are valuable, rare, imperfectly imitable and unsubstitutable [19].

Corbett and Claridge [20] denote capabilities as the ability of a firm to apply resources to act or achieve a goal, and further states that capabilities form the primary basis for competition between firms. In manufacturing strategy literature, capabilities are conceptualised as a business unit's intended or realised competitive performance or operational strengths and are therefore assessed using measures of operational performance, which typically includes cost, quality, flexibility, and delivery measures [21]. Swink and Hegarty [22] suggest that this performance-based approach to capabilities is conceptually aggregated to clearly direct the proper use of manufacturing resources. Different from the performance-based approach to capability research that is dominating the manufacturing strategy literature is the routine based approach to explaining the heterogeneity of firms. Capabilities are identified as high-level routines or bundles of routines [23]. Compared to resources, routines and capabilities built on routines are embedded in the dynamic interaction of multiple knowledge



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

sources and are more firm-specific and less transferable. For instance, a firm may have machines, input material, financial resources, and operators available to manufacture their products. However, to facilitate efficient manufacture and to achieve superior performance effective routines need to be developed to make all resources work in harmony and to enable dynamic information and knowledge exchange among individuals. Again, this clearly implies that manufacturing practices constitute a very important part in the development of manufacturing capabilities.

1.2 Manufacturing Strategy and Corporate Strategy: the link

Researchers have called for the alignment of the manufacturing strategy with the corporate strategy as well as involvement of the manufacturing aspect in formulating the corporate strategy to improve business performance [24]. The role of the manufacturing system in the company's competitive strategy has increasingly been realised from a strategic planning point of view [7;10]. Manufacturing strategy, in addressing goals and strategic decisions to face competition, seeks to align the firm's management approach to manufacturing with the corporate strategy that ultimately ensures business success [7]. The resource based theory suggests that a business unit's ability to achieve its profit maximization goal is a function of its ability to assemble and deploy appropriate resources that will provide it with sustainable competitive advantage in the market place [25]. Manufacturing strategy focuses at least in part on these resource decisions, being the means by which competitive strategy is implemented, and thus a major contributor to firm success.

1.3 Capacity utilisation

Operational performance can be inferred from the firm's capacity utilisation. As Ray [26] notes, business decision and policy formulation mostly depend on economic indicators however manufacturing capacity utilization becomes a key indicator of economic performance by giving a measure of resource-use efficiency. Having determined the competitiveness of the manufacturing economy and the supply-side factors that affect manufacturing, the design capacity of the economy can be ascertained. This is essentially the volume that can be produced by the economy in different product categories. The average capacity utilization for the Zimbabwean manufacturing sector in 2012 was 44.9% [27]. The design capacity represents the output that the economy can optimally produce at any given time, and this does not necessarily correspond to the capacity that is actually utilised. Various constrains will affect capacity utilisation, and from the metrics relating to cost and time, optimum capacity utilisation can be determined. Klein and Summers in Nyaoga et al. [28] define productive capacity as "the total level of output or production that could be produced in a given time period". *The Statistics South Africa Manufacturing: Utilization of Production Capacity Survey* defines capacity utilisation - measured as a percentage - as;

"utilisation which can feasibly be attained by [an] establishment with its current fixed assets and without significantly increasing unit costs as a result thereof"

Note should thus be taken of the fact that capacity utilisation is economic in a given range, outside of which, the cost of production increases and reduces the ability of manufacturing to compete against other economies. The measurement of actual capacity utilisation should therefore be accompanied by the measurement of the range in which capacity utilisation is economic.

2 HYPOTHESES

The research sought to show how the manufacturing strategy links to the corporate strategy and ultimately, to operational performance. Three distinct hypotheses stemming from the main objective were proposed:

- i. The relationship between manufacturing strategy and corporate strategy;
- ii. The relationship between existence of a manufacturing strategy and; capacity utilisation; and
- iii. Employees involved in developing the manufacturing strategy.

2.1.1 The link between Manufacturing strategy and corporate strategy

The type of manufacturing strategy that a firm emphasizes depends on its chosen competitive priority or priorities. Linking manufacturing strategy with business strategy uses market requirements to establish the performance criteria against which manufacturing can be measured. This necessitates an orientation based neither on products/markets nor on manufacturing exclusively, but a strategy that embraces the market / production interface, so that the degree of fit between the proposed market strategies for the corporate coincide with manufacturing's ability to support these strategies at the business and corporate levels. The strategic role of operations has been emphasized as being important in light of increasingly fierce global competition [10]. The research question was to establish whether there was a link between manufacturing strategies in Zimbabwean manufacturing firms and the corporate strategies of the firms. The research hypothesised that:



(H1) There is a positive relationship between the manufacturing strategy and corporate strategy in Zimbabwean Manufacturing firms.

2.1.2 Existence of manufacturing strategy and capacity utilisation

Superior manufacturing improves the odds of relative business success, thus manufacturing success generates opportunity for overall business success [5]. Amoako-Gympah and Acquaah [10] indicate a firm able to achieve both high design and conformance quality, cost reduction and concomitant improvements in efficiencies, high volume and mix flexibility and reliable on-time delivery can expect to achieve competitive product pricing, ensure customer satisfaction, fast response to market changes and overall, potentially increase sales growth and market share (factors in business success). The link between manufacturing strategy and firm performance has already been established with manufacturing strategy literature expounding on the veracity of manufacturing strategy-firm performance relationship [10]; [25] & [29]. This study extends the view, hypothesising that manufacturing strategy influences a measure of firm performance - the operational parameter of capacity utilisation. In determining how manufacturing strategies for successful manufacturing firms were derived, the level of capacity utilisation was used as an indicator of operational performance. The research thus tested the hypothesis that:

(H2) There is a positive relationship between existence of a manufacturing strategy and capacity utilisation.

2.1.3 Responsible employees for manufacturing strategy development

Research has already noted how participation of manufacturing managers in the strategy development process has the potential for eliminating incongruent strategic goals thus aligning manufacturing needs with those of the corporate in realising strategy [30]; [31] & [32]. High levels of manufacturing managers' participation in strategic decision-making has already been shown to relate positively to performances as measured by growth in sales, return on total assets, market share and sales [29]; [33] & [34] . When strategy formation is approached as a process involving both operational and top management, top management intent embodied in corporate strategy effectively combines with operations activity to create realized strategy. The involvement of manufacturing plant managers in organizational level strategy development was assessed. The research sought to determine if employees responsible for manufacturing plants were involved in strategy development at corporate level for firms with manufacturing divisions in Zimbabwe. Hypothetically;

(H3) There is a positive relationship between the involvement of manufacturing plant managers in corporate strategy development and operational performance.

3 METHODOLOGY

The research focused on manufacturing firms in Zimbabwe and the case of one alcohol manufacturer with specific emphasis on its manufacturing approaches and polices as well their linkage to corporate goals and strategies. The survey and case study research design were used in the study. Information was obtained from manufacturing executives of the selected manufacturing firms in Zimbabwe both for the survey and case study.

3.1 Population and sampling

The target population for this study was manufacturing firms in Zimbabwe, the sampling frame being the Confederation of Zimbabwean Industries' list of manufacturing firms in Zimbabwe. Respondents were manufacturing executives of the manufacturing firms as well as the Alcohol manufacturer's top management team. Stratified random sampling was applied with the population being divided into five strata based on industrial location using Zimbabwe's five business hub (the provinces: Mashonaland, Manicaland, Matebeleland, Masvingo and Midlands). A random sample was then drawn from each province.

3.1.1 Data collection

Survey research is often associated with deductive theorizing and is popular in business and management research [35], allowing the collection of large amounts of research information from a sizeable number of people in a way that is economical, giving information that is easily comparable and easy to standardize [36]. For the survey, a questionnaire to manufacturing managers of manufacturing firms in Zimbabwe was used for data collection. Questionnaires were administered personally in order to increase the response rate and to increase contact with the respondents. The questionnaires were pre-tested on two firms before the final administration. Corrections were made to the questionnaire based on suggestions and recommendations from the pretesting. Fifty (50) questionnaires were administered to representatives from the manufacturing firms, 40 (80%) of the sampled population responded. For the case study 10 structured interviews were conducted with top management at the alcohol manufacturing firm for data collection.


3.2 Measurement of variables

The existence of corporate strategy (V1) and existence of manufacturing strategy (V2) as well as management commitment to and emphasis on the strategic approach were determined based on responses to each of the questions in Table 1. The strategic emphasis and thus preferred manufacturing strategy (V3) was evaluated based on the average emphasis the firm placed on each of the four competitive priorities of cost, delivery, flexibility, and quality. Respondents indicated their understanding of the derivation of manufacturing strategy (V4) and its link to corporate strategy by stating their view in terms of the manufacturing strategy- corporate strategy relationship. Involvement of technical personnel in strategy formulation (V5) was determined as presence or absence of technical managers in strategy formulation meetings while one measure of firm performance, capacity utilisation (V6), was used to determine operational performance.

Table 1: Measurement of variables

Variable	Variable Description	Question	Response
V1	Existence of Corporate	Has your firm articulated a	1-Yes
	Strategy	corporate strategy?	2-No
			3-Don't know
		What is the frequency of	1-Bi-annually
		review of the corporate	2-Annually
		strategy?	3-Two years
			4-Continuously
			5-Not sure
V2	Existence of Manufacturing	Has your firm articulated a	1-Yes
	strategy	manufacturing strategy?	2-No
			3-Don't know
		What is the frequency of	1-Bi-annually
		review of the manufacturing	2-Annually
		strategy?	3-Two years
			4-Continuously
			5-Not sure
V3	Manufacturing strategy	Indicate the factor(s) that	0. There is no manufacturing
		your firm emphasises in	strategy
		determining the	1. Quality,
		manufacturing direction of	2. Delivery,
		the organisation	3. Cost;
			4. Flexibility
			5. Combination of factors
V4	Derivation of Manufacturing	Rate the relevance of the	1. Extremely relevant
	strategy	strategic approach to	2. Very relevant
		organisational growth.	3. Relevant
			4. Not very relevant
			5. Not relevant at all
		How does the manufacturing	1. Manufacturing strategy is
		strategy of your organisation	derived from corporate strategy;
		relate to the corporate	2. Corporate strategy is derived
		strategy?	The two are not related:
			3. The two are not related;
			4. Manufacturing strategy doesn't
			5 Don't Know
V5	Manufacturing Management	To what extent are conjer	1 Never
٧J	involvement	no what extent are senior	2 Paroly:
	Involvement	involved in strategy	2. Narely,
		formulation meetings?	5. Attrays
V6	Capacity utilisation	What is the current canacity	1 0-25%
10		utilisation of your firm as a	2 76-50%
		nercentage?	3 51-75%
		percentage:	4 76-100%
			T. / U I UU/0

The study controlled for firm size and age. All sampled organisations were medium to large manufacturing operations with a minimum of 50 employees and having been in operation for over 5 years. Manufacturing firms responding to the survey comprised of 65% private organisations, 25% government departments and 10% public companies. The effect of this ownership was not factored in for this study. Respondents in the survey research



were manufacturing plant managers, i.e. technical managers responsible for managing the manufacturing firms or divisions in their firms (Table 2). Respondents for the case study were executive mangers responsible for crafting the firm's corporate strategy

Designation	Responses
Managing Director	2
Technical/operations Director	11
Plant Manager	10
Engineering Manager	8
Production Manager	7
Plant Engineer	2
Total	40

Table 2:	Profile of	respondents	(survey	research)
			`	,

3.3 Hypotheses tests

The *Cramer's V* statistic was used to determine associations amongst variables. In estimating relationships among variables, Cramer's V is independent of sample size. *Cramer's V* ranges between 0 (no relationship) and 1 (perfect relationship).

Cramer's
$$V = \sqrt{\frac{\chi^2}{N(L-1)}}$$
; where L = min (h, k) [1]

4 FINDINGS

All respondents in the survey research were manufacturing plant managers, i.e. technical managers responsible for managing the manufacturing firms or divisions of their firms. The highest number of respondents (27.5%) were technical/operations directors, then plant managers (25%), engineering managers (20%) and production managers (17.5%), plant engineers and managing directors making up five percent each.

Results indicated that 70% of the sampled firms are operating above 50% capacity utilisation with the greater number or organisations being in the 75-100% capacity utilisation band. 90% of the sampled organisations had corporate strategies in place while 72.5% had developed manufacturing strategies. Of those firms with corporate strategies, 75% had manufacturing strategies in place. The relationship between existence of corporate strategy and manufacturing strategy is significant (Cramer's V= 0.713; p = 0.000 for α = 0.05, n = 40) (Table 3), indicating that the existence of manufacturing strategy is related to the existence of corporate strategies in the organisation.

	-	Manufa	Manufacturing strategy exists		
		0	Yes	No	Total
Corporate strategy exists	Yes	0	27	9	36
	No	2	0	0	2
	Dont know	0	2	0	2
Total		2	29	9	40
Cramer's V	0.713	-	-	-	
Approx. Sig					

Table 3: Corporate strategy * Manufacturing strategy Cross-tabulation

All firms (100%) that did not have corporate strategies either did not have manufacturing strategies or did not know about manufacturing strategies while 75% of those that had corporate strategies had manufacturing strategies as well. Of the 36 firms that had corporate strategies, 42.5% derived manufacturing strategies from the corporate strategy, 25% derived corporate strategy from manufacturing strategy while only 5% indicated that



there is no relationship in their view between their firm's corporate and manufacturing strategies. 80% of the firms that derived corporate strategy from manufacturing strategy recorded 76-100% capacity utilisation, while only 47.1% of those deriving manufacturing strategy from corporate strategy were in the 76-100% utilisation band. (Table 4).

	-	Capacity Utilisation (%)					
		0-25%	26-50%	51-75%	76-100%	Total	%
Strategy derivation	Manufacturing from corporate strategy	0	3	6	8	17	42.5
	Corporate strategy from manufacturing strategy	0	0	2	8	10	25.0
	The two are not related	0	0	2	0	2	5.0
	Manufacturing strategy doesn't exist	4	3	0	0	7	17.5
	Don't Know	2	0	0	2	4	10.0
Total		6	6	10	18	40	100
Cramer's V			0.554	<u>.</u>		-	
Approx. Sig.			0.000				

Table 4: Strategy derivation * Capacity Utilisation (%) Cross-tabulation

The association between the existence of manufacturing strategy and capacity utilisation is significant (Cramer's V=0.595, p = 0.000 at α = 0.05) (Table 5). Successful manufacturing firms sampled (with capacity utilisation in the range 76-100%) reported that they have manufacturing strategies in place. 77.8% of firms that did not have manufacturing strategies were operating below 50% capacity utilisation while 89.7% of firms that have manufacturing strategies were operating above 50% capacity utilisation.

		Capacity	Capacity Utilisation (%)			
		0-25%	26-50%	51-75%	76-100%	Total
Manufacturing strategy exists Yes		0	3	10	16	29
	No	4	3	0	2	9
	Don't Know	2	0	0	0	2
Total		6	6	10	18	40
Cramer's V		0.595		<u>4</u>	<u></u>	
Approx. Sig.		0.000				

Table 5: Manufacturing strategy * Capacity Utilisation (%) Cross-tabulation

In determining the considerations when formulating manufacturing strategy, and thus determining the firm's emphasis on the competitive priorities, responses indicated that 79.3% of the firms that have manufacturing strategies consider a combination of factors including quality, delivery, cost, and flexibility in manufacturing strategy formulation with only 20.7% considering cost alone in the formulation of manufacturing strategy. Further, successful manufacturing firms (with 76-100% capacity utilisation) involve senior manufacturing managers in strategy formulation meetings (64% of the organisation in the maximum capacity utilisation category involve manufacturing managers in strategic planning meetings always). There is a significant (Cramer's V= 0.524, p=0.000, α =0.05) association between manufacturing management contribution in strategic planning meetings and by extension, operational performance (Table 6).



	-	Senior manut	Senior manufacturing management involvement			
		0	Never	Rarely	Always	Total
Capacity Utilisation (%)	0-25%	2	2	2	0	6
	26-50%	0	3	0	3	6
	51-75%	0	0	4	6	10
	76-100%	0	2	0	16	18
Total		2	7	6	25	40
Cramer's V	0.524					
Approx. Sig.	0.000					

Table 6: Capacity Utilisation (%) * Senior Manufacturing management involvement Cross-tabulation

4.1 Management approach for the Case

The management interview for the case focused on the existence of manufacturing strategy, its relationship to the corporate strategy as well as considerations in determining the manufacturing direction for the firm. The alcohol manufacturer, like other manufacturing firms in the country, has been affected by industry wide problems that include high production costs due to old machinery and skills flight. Yet the firm has enjoyed monopoly as the only Zimbabwean manufacturer of potable alcohol. Limited by its capacity (size), the firm has continued to lose potential customers to importers of potable alcohol who are able to land the import at the Alcohol manufacturer's cost of production posing a veritable threat if the organization pursues the same operating strategies into the future. From interviews with management at the alcohol manufacturer, evidence that managers realise the relationship between the existence of a deliberate manufacturing strategy and sustained operational performance exists. All the interviewees (100%) in the case study consented to the existence of a corporate strategy in the group, they all agreed that there is no documented manufacturing strategy at the alcohol manufacturer and highlighted the need for a structured process for establishing a manufacturing strategy as a necessary improvement to the firm's strategy process. The capacity utilisation for the case was in the 76-100% band thus classifying the firm as one of the few successful firms that do not possess a manufacturing strategy given 22.2% of firms in the survey that did not possess a manufacturing strategy were successful. In terms of competitive priorities in determining the manufacturing direction for the case, 60% highlighted that the approach employed was a quality based manufacturing approach, while 40% considered it to be a combination of cost and quality. The results showed that whilst the case did not have a documented manufacturing strategy, they were employing some of the competitive priorities used as basis for deriving manufacturing strategies.

5 RECOMMENDATIONS

The strategic approach to manufacturing is critical to operational performance. In particular, the existence of manufacturing strategies is essential for the success of manufacturing firms in Zimbabwe. Manufacturing firms are recommended to adopt corporate and manufacturing strategic planning approaches for the successful running of their operations. The research showed that manufacturing firms that had both corporate and manufacturing strategies recorded better performance. Further those firms that derived their corporate strategy from the manufacturing firms derive their corporate strategies from their manufacturing strategy. Thus manufacturing firms are recommended to adopt the strategic approach in which manufacturing priorities drive the strategic management process. Further, successful manufacturing firms involve senior manufacturing management in their strategy formulation process. From an analysis of operations at the case, the operation at full capacity (76-100% capacity utilisation) without a manufacturing strategy was possibly due to the prevailing situation in which the firm enjoyed a more or less monopolistic hold to its market, a situation that is not guaranteed to last with competition growing and the changing economy. Adoption of a deliberate and documented manufacturing strategy is recommended.

6 CONCLUSION

The research showed that almost every manufacturing firm had a corporate strategy and that there was a significant relationship between existence of corporate strategy, manufacturing strategy and operational performance. The research also showed that successful manufacturing firms had established corporate and manufacturing strategies and that the existence of a manufacturing strategy is positively associated with



operational performance. Manufacturing firms used a combination of competitive priorities (cost, delivery, flexibility, quality) in coming up with manufacturing strategy and those firms that derived their corporate strategy from the manufacturing strategy showed more success that those that derived their manufacturing strategy from the corporate strategy. The research verified that successful manufacturing firms involved senior manufacturing managers in strategy formulation meetings. There was a significant relationship between the presence of manufacturing management contribution in strategic planning and operational performance.

The research has shown that the strategic approach to running a manufacturing concern ensures success. Manufacturing priorities drive successful manufacturing firms through the corporate and manufacturing strategy. While the research shows that there is significant application of the strategic approach to running successful manufacturing industries in Zimbabwe, there is need to study the overall strategic management process of these firms from planning to implementation. Given Zimbabwe's unique economic, political and social environment the test would be to determine the applicability of conventional strategic management methods and processes. The success emanating from the implementation of strategic approaches to manufacturing, particularly as firms seek to achieve world class standards, the choice of strategy and its implementation will be all the more critical.

REFERENCES

- 1. Chindondondo, B. Nyanga, L. Van der Merwe, A. Mupinga, T & Mhlanga, S. 2014. Analysis of a time based and corrective maintenance system for a sugar producing company, *Southern African Institute for Industrial Engineering 26th Proceedings*, 14th 16th of July 2014, Muldersdrift, South Africa, 26 (1), pp 11701-14
- 2. Siwadi, P. & Pelser, T. G. 2015. Exploring technological capabilities to resuscitate the Zimbabwean manufacturing sector, The Journal of Applied business research, 31 (3), pp 1023-1036.
- 3. Chindondondo, B. Nyanga, L. Van der Merwe, A. Mupinga, T & Mhlanga, S. 2014. Development of a condition based maintenance system for a sugar producing company, *Southern African Institute for Industrial Engineering 26th Proceedings*, 14th 16th of July 2014, Muldersdrift, South Africa, 14th 16th of July 2014, Muldersdrift, South Africa. 26 (1), pp 11731-14
- 4. Platts, K.W. and Gregory, M.J. 1990. Manufacturing Audit in the process of strategy formulation, International journal of operations and production management, 10(9), pp 5-26
- 5. Roth, A. V. and Jeffrey G. M. 1992. 'Success Factors in Manufacturing', Business Horizons, July-August ,73-81.
- 6. Ehie, I. Muogboh, O. 2016. Analysis of manufacturing strategy in developing countries, Journal of Manufacturing Technology Management, Vol. 27 (2) pp. 234 260
- 7. Vivares-Vergara, J. A., Sarache-Castro, W. A., & Naranjo-Valencia, J. C. 2014. The content of manufacturing strategy: A case study of Colombian Industries, *Dyna*, 81(183); 140-147
- 8. Skinner, W. 1969. Manufacturing missing link in corporate strategy. Harvard Business Review, May June, 136 45
- 9. Hayes, R.H. & Wheelwright, S.C. 1984. Restoring our Competitive Edge: Competing through Manufacturing. Wiley, New York.
- 10. Amoako-Gyampah, K. and Acquaah M. 2008. Manufacturing strategy, competitive strategy and firm performance: An empirical study in a developing economy environment, *International journal of Production Economics*, 111 (2008), pp.575-592
- 11. Ward, P.T. and Duray. R. 2000. Manufacturing Strategy in Context: Environment, Competitive Strategy and Manufacturing Strategy; Journal of Operations Management, 18 (2), 123 -138
- 12. Greasley, A. 2006. Operations Management, John Wiley & Sons, Chichester, West Sussex.
- 13. Slack, N. and Lewis, M. 2002. Operations Strategy, Pearson Education Limited, Harlow, Essex
- 14. Krajewski, L. J., Ritzman, L. P. 1987. Operations Management, Strategy and Analysis, Addison-Wesley Publishing Company, Boston, MA.
- 15. Leong, G.K., Snyder, D.L., Ward, P.T. 1990. Research in the process and content of manufacturing strategy, *Omega*, 18 (2), 109-122.
- 16. Größler, A. & Grübner, A. 2006. An Empirical Model of the Relationships between Manufacturing Capabilities, International Journal of Operations and Production Management, 26(5), 458-485.
- 17. Tan, K.C., Kannan, V.R. and Narasimhan, R. 2007. The impact of operations capability on firm performance. *International Journal of Production Research*, 45 (21), pp 5135 -5156.
- 18. Wernerfelt, B. 1984. A resource-based view of the firm, *Strategic Management Journal*, 5(2), pp 171-180.
- 19. Barney, J. B. 1991. 'Firm resources and sustained competitive advantage'. Journal of Management, 17(1), pp 99-120.
- 20. Corbett, L.M. & Claridge, G.S. 2002. Key manufacturing capability elements and business performance, *International Journal of Production Research*, 40 (1), 109-131.
- 21. Schroeder, S, Flynn, B. B., Sakakibara, R.G., Bates, K. A & Flynn, E. J. 1990. Empirical research methods in operations management, *Journal of Operations Management*, 9(2), 250-284.
- 22. Swink, M. & Hegarty, W.H. 1998. Core manufacturing capabilities and their links to product differentiation, International Journal of Operations and Production Management, 18(4), 374-396.



- 23. Zollo, M. & Winter, S.G. 2002. Deliberate learning and the evolution of dynamic capabilities', Organization Science, 13(3), 339-351
- 24. Lee, W., Rhee, S. & Oh, J. 2014. The relationships between manufacturing strategy process, manufacturing-marketing integration, and plant performance: an empirical study of Korean manufacturers, *Operations Management*, 7, pp 117-133
- 25. Williams, F. P., D'Souza, D. E., Rosenfeldt, M. E. & Kassaee, M. 1995. Manufacturing strategy, business strategy and firm performance in a mature industry, *Journal of Operations Management*, 13, pp 19 -33
- 26. Ray, S. 2013. A close look into research studies on capacity utilization in India and abroad, *International journal of economics, finance and management*, 2 (1) pp 52-59
- 27. Katsvamutima, E. Jeevananda, S. 2014. An Investigation of the Business Level Strategies in Zimbabwe Food Manufacturing Sector (2006 -2013). International journal of Science and research, 3 (6). Pp 1052 1063
- Nyaoga, R.B. Wang, M. and Magutu, P.O. 2015. The relationship between capacity utilization and value chain performance: Evidence from Kenyan tea processing firms *African Journal of Business Management*, 9(9), pp.402-411
- 29. Swamidass, P.M., Newell, W.T. 1987. Manufacturing strategy, environmental uncertainty and performance: a path analytic model, *Journal of Operations Management*, 33 (4), pp 509 524
- 30. Hill, T. J. 1985. Manufacturing Strategy, Macmillan, London.
- 31. Voss, C.A. 1992. Manufacturing Strategy Process and Content, Chapman & Hall, London
- Floyd, S. W. & Wooldridge, B. 1992. Middle management involvement in strategy and its association with strategic type: a research note, *Strategic Management Journal*, 13 (Special Issue: Strategy Process: Managing Corporate Self-Renewal) pp. 153-167
- 33. Ward, P.T., Leong, G.K., Boyer, K.K., (1994). 'Manufacturing proactiveness and performance'. Decision Sciences 25 _3, 337-358.
- 34. Meredith, J.R., Vineyard, M., (1993). 'A longitudinal study of the role of manufacturing technology in business strategy'. International Journal of Operations and Production Management 13 (12); 4-24
- 35. Saunders, M., Lewis, P., & Thornhill, A. 2016. Research methods for business students. Edinburgh Gate, Harlow, England: Pearson Education.
- 36. Shaughnessy, J., Zechmeister, E., & Jeanne, Z. 2011. *Research methods in psychology*, 9th Edition, New York, NY: McGraw Hill



A SYSTEMATIC COMPARISON OF DONOR FUNDED SUPPLY CHAIN AND COMMERCIAL SUPPLY CHAIN CHARACTERISTICS

D. Lingervelder¹, L. Bam¹, W.G. Bam²

¹Health Systems Engineering and Innovation Hub, Department of Industrial Engineering, Stellenbosch University, South Africa dlingervelder@gmail.com & louzanne@sun.ac.za

> ²Value Capture Systems, Department of Industrial Engineering, Stellenbosch University, South Africa wouterb@sun.ac.za

ABSTRACT

Commercial supply chains and donor funded supply chains are driven by distinctive mechanisms and purposes, and the factors that make the operational environment of donor funded supply chains unpredictable, complex and dynamic, differ from those that cause this complexity in a commercial supply chain environment. Although literature contains several theories and frameworks to assist in the design, measurement and management of commercial supply chains, there is a significant research gap related to the formal study of donor funded supply chains. In this research, several examples of donor funded supply chains are investigated in order to systematically compare and highlight the key differences between donor funded supply chains and commercial supply chains.



1. INTRODUCTION

Since the introduction of supply chain management in the 1980s, commercial supply chains have been studied extensively, leading to the development of several theories and frameworks to assist in their design, measurement and management. There has, however, been less formal study of donor funded supply chains. Donor funded supply chains play an important role in relieving human suffering and addressing prioritised global health challenges and, as such, often function under circumstances that might make traditional commercial supply chains unviable. The factors that make the operational environment of donor funded supply chains unpredictable, complex and dynamic, may differ or be more severe from those that cause this complexity in a commercial supply chain environment [1]. As such, though commercial supply chains and donor funded supply chains share many similarities, there are distinct differences between the two that should be considered when developing management practices and principles for donor funded supply chains [2]. This article investigates several examples of donor funded supply chains, including medical-, relief-, agricultural- and nutritional supply chains, with reference to relevant characteristics, to systematically compare and highlight differences between donor funded supply chains and commercial supply chains. The characteristics in Table 1 represent the range of business functions and services included in a supply chain as defined by three popular¹ commercial supply chain focussed models:

- 1. the Mentzer Model [3];
- 2. the Global Supply Chain Forum (GSCF) Model [4]; and
- 3. the Collaborative Planning, Forecasting and Replenishment (CPFR) Model [5].

The characteristics in Table 1 are divided into two sections, those that are included in the comparison of donor funded and commercial supply chains in this article, and those that are not. All characteristics that are defined by at least two of the three models presented, are included in the discussion. Furthermore, three characteristics that proved exceptionally different for donor funded supply chains and commercial supply chains (namely trust, agility, and structure and components) are also included in the discussion.

Characteristics	Mentzer Model [3]	GSCF Model [4]	CPFR Model [5]		
Characteristics discussed in this article:					
Structure & Components	х	-	-		
Stakeholders	х	-	х		
Trust	х	-	-		
Goals & Objectives	х	-	х		
Finances	х	х	-		
Customer Service & Marketing	х	х	х		
Research & Development	х	х	-		
Demand & Forecasting	х	х	х		
Procurement	х	х	х		
Manufacturing	х	х	х		
Logistics	х	х	х		
Agility	-	-	-		
Characteristics not discussed in this article:					
Customer Relationship Management	-	х	-		
Supplier Relationship Management	х	-	-		
Returns Management		х	-		
Product Development & Commercialization	-	х	-		

Table 1: Perspectives on the range of characteristics included in a supply chain

¹ These models have been referenced frequently in supply chain management literature (including in: Naslund and Williamson [6], Amemba [7], Davis-Sramek et al. [8], Campuzano and Mula [9], and Tracey et al. [10]).



2. ANALYSIS OF DISTINGUISING CHARACTERISTICS

In this section, the characteristics of donor funded supply chains and commercial supply chains are described and contrasted. The discussion is structured in accordance with the distinguishing characteristics identified in Table 1. The focus is on highlighting structural differences between the supply chains.

2.1 Supply chain structure and components

A typical commercial supply chain encompasses the activities and processes related to the flow, transformation and transportation of goods [11], as illustrated in Figure 1.a. In donor funded supply chains, the supplies often flow through various shipments as illustrated in Figure 1.b. The supplies mainly comprise pre-positioned stock in warehouses, in-kind donations and supplies obtained from suppliers after manufacturing [12]. The supplies are transported, in most cases from several locations worldwide, typically to a primary warehouse, from where they are moved to a large secondary warehouse where they are stored, sorted and distributed to local distribution centres. These distribution centres deliver the supplies to the populations in need [13]. The structure and components of a donor funded supply chain will differ for each setting. For example, in relief chains the manufacturing activities and processes are not necessarily included in the supply chain, while in medical and nutritional supply chains a significant amount of time, effort and research typically goes into manufacturing activities [13, 14].



Figure 1: (a) A typical commercial supply chain. (b) A typical donor funded supply chain.

Furthermore, a supply chain comprises of an upstream (global) segment and a downstream (domestic) segment. The upstream segment includes activities such as manufacturing, procurement, financing, forecasting and warehousing; while the downstream segment includes the in-country storage and distribution to retailers and consumers. A key difference between donor funded supply chains and commercial supply chains, is that although these two segments are primarily managed as two coupled segments in a commercial supply chain, they are ordinarily decoupled from one another in a donor funded supply chain [14]. The global segment(s) is typically managed by international foundations and remains fairly uninvolved once the supplies reach the country. From there on the domestic segment is typically separately managed by the government or other local organisations to administer part of the local distribution and the last mile distribution of suppliers to beneficiaries, as seen in Figure 1.b. This is often due to the unsupervised nature of donor funded supply chains, since there is no individual organisation or primary stakeholders that governs the entire supply chain or has authority to instigate coordination between actors and activities [15].

2.2 Stakeholders

In commercial supply chains, the primary stakeholders (shareholders) are often seen as the owners of the supply chain [16] and their interests guide the firm's policy [17]. The primary stakeholders in donor funded supply chains are the host government and NGOs [16]. All actors, except national aid agencies, require approval from the host government, and sometimes the neighbouring government, in order to operate within the country [18]. NGOs



can include numerous actors such as international aid agencies as well as small organisations that are formed within the local communities [18]. Though commercial organisations also have multiple stakeholders such as its customers, employees, retailers and suppliers, each with different needs, meeting the needs of all these stakeholders rarely conflicts with the shareholders' and organisation's long-term goals and objectives [19]. In contrast, the principal goals, missions, interests, capacity and constraints of stakeholders in donor funded settings are often different and conflicting [20, 21]. Furthermore, the stakeholders involved in the donor funded setting, can be categorised as (i) actors that exist in the area and are linked to it, or (ii) international actors. These two groups normally take different perspectives and approaches to donor funded settings and will often perform operations differently, adding to the complex nature of donor funded supply chains [22].

2.3 Trust

Trust between the actors and stakeholders of the different supply chain activities and stages is crucial for positive long-term performance. The majority of the commercial sector spends a significant amount of time and resources to build a lasting relationship with their partners [23]. In several donor funded situations however, different actors form a network with a shared purpose of relieving the suffering of people, but there is often not sufficient time or resources for the opportunity to build trust before the network must begin to operate together [21].

Additionally, trust not only plays an important role between actors and stakeholders, but also between the organisation and the consumers [24]. For example, in Nigeria several mid-size Nigerian firms aim to produce nutrient-rich foods for undernourished populations. These products compete with foreign-produced products sold by donor-organisations. Consumers are doubtful about the quality of the locally produced products and trust is therefore low between the actors in the chain. This low-trust environment depresses innovation and creates prejudices against certain products (in this case those that are locally manufactured) [25].

2.4 Goals and objectives

Every supply chain has a goal towards which it operates. The fundamental goal of any supply chain is to "deliver the right supplies in the right quantities to the right locations at the right time" [13]. The main goals and objectives for commercial and donor funded supply chains, however, are quite different. The overriding goal for a commercial supply chain is financial profit [26]. These supply chains strive to make profits and provide financial returns to their shareholders [27]. In contrast, for donor funded supply chains, profits are not the goal; instead, typical goals include to produce value for money, be efficient and effective, ensure fair competition between suppliers, ensure accountability, and ensure procedures are executed ethically [13]. Furthermore, each organisation involved in the supply chain strives to achieve its own purpose and mission [19, 28].

Similar to commercial supply chains, donor funded supply chains are also concerned about their financial well being, since financial stability is an important aspect of their mission and sustainability. However, in contrast to commercial supply chains where the strategic objectives are based on the financial returns paid out to the shareholders and the value created through delivering high quality goods and services to the consumers [13], finances are viewed as constraints rather than objectives in donor funded supply chains [29]. For donor funded supply chains, the goal is to reduce human suffering, save lives and improving quality of life, within the financial constraints. The strategic objectives are therefore to ensure financial sustainability and to ensure effective flow and operations [19]. Both supply chain types consider customer service and cost in their goals and objectives, but they bring different dimensions to the concepts. For commercial settings the aspects are considered primarily from the perspective of maximising profits, while in donor funded situations, these are considered primarily from the perspective of maximising the wellbeing (or in some cases, minimising the mortality) of the population [30].

2.5 Finances

The life blood of an organisation is its finances since this ensures the organisation remains functioning. A notable difference in the finances of donor funded and commercial supply chains is the source of income. In commercial supply chains, the defining source of income is the income earned from the sale of goods and services. In donor funded supply chains, however, the recipients of the goods and services often pay nothing or very little and there is consequently no income, or at least no profit, from this source. Rather, the income to fund the operations of donor funded supply chains are sourced from individuals and organisations that expect no personal benefits in return (though these sources may of course expect benefits for the intended recipient population), where in commercial supply chains, revenue sources are customers that purchase goods and services for their own benefit [19, 31]. All of the transactions that occur between the different partners throughout the stages of the supply chain are referred to as the financial supply chain. Attention is often placed on the integration of the physical and financial supply chain, since failure to identify and improve any weaknesses within the integration can threaten both chains' liveliness [32]. In donor funded supply chains, the source of revenue can include government funding [A6] and charitable donations from individuals and organisations [13]. Ordinarily, donors provide the majority of the funds for the activities. Donations consist of free goods and/or services or financial contributions to support the operations [18]. Consequently, a functioning financial supply chain is essential in ensuring that financial contributions can be utilised [33]. Typical challenges experienced in donor funded financial supply chains include: lack of formal provisions with local financial institutions and suppliers that can cause difficulties with transfer processes [34]; and irregular funding cycles (especially for smaller agencies) due



to contributions being donated for particular operations rather than for the agency's activities in general [35]. Regular and effective financial flows are essential for the accurate functioning of any supply chain.

A last feature that frequently creates challenges in a donor funded supply chain's finances, is when the chain has two separate financial supply chains. One that is internationally financed by one or more global funds and one that is domestically financed within a given country [14, 36]. This corresponds with the decoupled upstream and downstream segments of some donor funded supply chains. A donor funded supply chain with different financial supply chains (international and domestic) faces a unique situation when it comes to price and demand elasticity. Demand in the internationally donor financed supply chain is ordinarily inelastic to small changes in price, when compared to the domestically financed portion. Therefore, when examining the overall market in this type of donor funded supply chain, the demand curve will be flat and unresponsive to large volume changes or small price changes. However, at a certain stage in the domestically financed supply chain, the volume will drastically increase when the price descends below a certain threshold [14].

2.6 Marketing and customer service

For commercial supply chains, the customers are any individual or organisation that buys and receives the product or service [13]. In donor funded supply chains the customers are often seen as the end-consumers that receive the intended aid. However, since donors play such an important and influential role in donor funded supply chains, a majority of NGOs instead view the donors as the customers. With this perspective, the NGOs deliver a service to donors; the service being providing aid to those in need [37]. Commercial supply chains view the final customer as the source of income for the entire supply chain, where in donor funded supply chains the final customer (the people who receive aid) rarely takes part in a transaction [38, 39]. In donor-funded supply chains, the customer often does not have a choice of product, while in most commercial supply chains the customer can analyse the market and choose which product, out of several options, to buy [39]. Therefore, where commercial supply chains market their product to the customer, donor funded supply chains must target the donors and convince them to provide some contribution [38].

In donor funded supply chains, an important issue concerning the distribution of the supplies directly to the consumers, is equity and fairness. When comparing the geographical spread of consumers with the availability of supplies over time and space, one population or group could easily be favoured over another [40]. In commercial supply chains, large amounts of time and money are typically allocated towards market segmentation with dedicated departments typically assigned to segment the consumers and market the product or service according to the segments [41]. This is not the case in donor funded supply chains, where the main goal, in terms of market segmentation, is to ensure the fair distribution of supplies [14]. In donor funded supply chains, there is often less choice with regards to which locations to distribute products to, with the overarching objective being to provide supplies and services to populations in need, regardless of their location.

2.7 Research and development

Research and development plays an important part in the improvement and growth of an organisation and its supply chain. It involves investigative activities that either lead to the development of new products or processes, or the enhancement of current products and processes. In donor funded supply chains, challenges can arise when donors provide funds only for specific activities. As an example, donors typically provide funding specifically for operational activities; therefore, strategic improvements activities, such as research and development, often have to be put on hold until funding that can be utilised for these activities becomes available [13]. In commercial supply chains, on the other hand, financial resources are typically allocated to both strategic and operational activities [42]. In many commercial settings, an organisation will spend large amounts of time and money on research and development, with some having a separate department focussing only on developing new or improving current products and processes [43]. Most donor funded supply chains, on the other hand, have to rely on global initiatives such as the World Health Organisation, UNITAID, The United Nations Children's Emergency Fund, etc. to develop new products and help with the improvement of processes [14, 44, 45].

2.8 Demand and forecasting

A principle difference between donor funded supply chains and commercial supply chains is the stability and / or predictability of the demand. Due to the nature of the types of challenges that donor funded supply chains typically seek to address, demand tends to be less stable and / or predictable than for commercial supply chains [46]. Typical differences in terms of demand elasticity, demand patterns and forecasting abilities are discussed.

In donor funded settings, there are often different types of demand items with different elasticity. For example, in disaster relief supply chains, some items are in demand only once (such as tents), others are recurring (such as water), some demand items can expire (such as food) while others may be carried over (clothing). Or in medical supply chains, where a patient follows a treatment regime specific to the patient's condition, out of a number of possible treatment regimes. Each regime is unique in that it can comprise of different quantities of different drug formulations to be prescribed for different lengths of time [14]. Due to the various types of demand, each often with different transportation requirements, there is a heightened risk of the demand overwhelming the distribution network's capacity [40].



Another difference is the demand pattern [13, 47]. For many commercial supply chains, demand occurs from recognised locations and orders are often placed in fixed quantities, in consistent time intervals from fixed suppliers [48], under these conditions, demand is steady and can be anticipated [47]. However, these conditions rarely occur in donor funded settings where demand is often unpredictable and occurs in irregular amounts and time intervals. Various factors contribute to this unpredictability including the impact of the situation, demographics, social and economic conditions of the area, as well as the type of supplies [38]. In some cases, the locations are unknown until the demand occurs [47]. For example, relief supply chains only know the demand location after a disaster has struck or in medical supply chains a disease or virus, such as TB or Ebola, can unexpectedly spread in a country creating a sudden large demand in that country.

An environment that would allow easy and effective forecasting involves predictable demand; unlike the demand characteristics in donor funded settings as described above. Donor funded supply chains often exist in a complex and irresolute environment, where time is imperative to save lives. The long distance between different stakeholders in the supply chain and the distance between them and the area of the incidents intensifies the uncertainty. In many instances, the correct information is not available or never makes it to the stakeholder(s) that need it [1]. When the correct information is unavailable, assumptions about the types and quantities are made [49]. Examples such as the global TB epidemic and the Haiti earthquake in 2010, illustrate the presence of uncertainty in donor funded situations [1].

2.9 Procurement

Procurement involves all the activities related to the acquisition of goods, services or works from the raw materials stage through to the end product [50]. In donor funded situations, procurement can often represent the biggest percentage of the expenses, depending on the situation [26] For example in medical supply chains, the procurement of diagnostics and drugs is very expensive, while in relief supply chains, logistics and distribution is often more expensive than the procurement of supplies. Donors provide the majority of the funding required for procuring goods and services [51]. The procurement process in a donor funded supply chain is typically exceptionally different from commercial supply chains due to the different environments in which they operate. For example, in relief chains, there is often pressure to immediately deliver goods or services, due to the desperate needs of the population, without knowing the exact quantities [26], while in some medical supply chains an order is placed several months before receiving any supplies. For the majority of donor funded supply chains, purchasing decisions are made through a competitive bidding process and short-term contracts or agreements are set up, whereas in commercial supply chains a lasting relationship is often built with their partners to set up long-term agreements and contracts [23]. Procurement coordination, which is often applied in commercial supply chains, is difficult to implement in donor funded supply chains due to the uncertainties in demand, characteristics of donor funding, limited information and the specific procurement processes of donor funded supply chains [20].

2.10 Manufacturing

Processes included in the manufacturing aspect of the supply chain are the manufacturing of the product, formulating and packaging, inspection, and shipping and transport. Under conditions of stable and / or predictable demand (as often occurs in commercial supply chains), the majority of manufacturing processes can be forecast-driven. In contrast, the unstable and / or unpredictable demand typically experienced in donor-funded supply chains represents a risk to manufacturers and causes some manufacturers to employ an order-driven process where the procurement and manufacturing process only begin once an order is received [14]. The manufacturers must therefore take on suboptimal inventory structures and manufacturing processes, which lead to higher prices and longer lead times [44, 45]. The limited actual demand and lack of accurate forecasting weakens the manufacturers' confidence to produce large volumes of supplies. Therefore, they manufacture the supplies in smaller batch sizes that are not cost effective. This occasionally necessitates manufacturers to ask a higher price for the finished product. Because of the longer lead times, countries are pressured to place orders months or even years in advance. In many cases, the inaccurate forecasting either leads to shortages, when needs are forecasted too low, or destruction of expired goods, when needs are forecasted too high [14].

In several cases, manufacturers have few incentives to invest and innovate for new products, demand is difficult to pool in order to negotiate price reductions, and forecasting to plan production and avoid product shortages is difficult [14, 52]. Additionally, the difficult nature of the products in specialised donor funded supply chains, such as medical or nutrition chains, leads to a low market attractiveness for manufacturers. For these reasons, there are often only one or small number of manufacturers per product, while in the commercial sectors there is rarely a manufacturer who has a monopoly in the market. This restricted market structure and lack of competition among manufacturers creates additional challenges for managing other supply chain aspects such as price and availability [14].

2.11 Logistics

An important function required to support the activities of the supply chain, is logistics [39]. Logistics, in general, incorporate a variety of activities throughout the different phases of the supply chain, such as transportation,



warehousing and customs clearance [53]. In commercial supply chains, logistics can be defined as the process of managing the product-, information-, and financial flows from the source to the final customers [22], while in donor funded supply chains it is defined as the process of managing these flows from the donors to the affected populations [54]. Logistics is one of the fundamental elements, if not the most important, in any supply chain [18], since it is crucial for effective and efficient delivery, is a typical expensive part of the supply chain and it provides feedback about all aspects of execution, such as supplier effectiveness, cost and timeliness of transportation etc. [53].

Transportation in donor funded settings are typically very different than in commercial settings [51] with the biggest logistical challenge in donor funded supply chains being the complexity of the conditions in which they operate [30]. For example, poor existing transportation infrastructure and geographical characteristics including mountains, islands, etc. may discourage some commercial sectors to target areas, but donor funded supply chains have to deliver supplies to populations in need regardless of these conditions. Another challenge is obtaining the appropriate information about the conditions of the infrastructure, since sufficient communication and information technologies may be unavailable in particular Low- and Middle Income Countries [20]. The majority of donor funded supply chains do not own and operate their own vehicle fleets, but instead rely on local or international logistic agencies. Although several commercial supply chains are also reliant on external logistics companies, the majority do invest in their own vehicle fleets. In some donor funded settings, especially in medical supply chains, the distribution is narrowed to only state run distribution systems that makes incentive structures difficult to create and implement. Nonetheless, even when distribution is done through private networks, the regulations and small market size can inhibit sufficient competition. In commercial supply chains, several distribution channels are often used in order to achieve expanded reach [52].

The resources required for distributing goods and services of a commercial supply chain, is often generic and widely available. Therefore, the majority of the consumer product companies in a commercial supply chain use horizontal collaboration, in order to create a distribution pool that will increase the delivery rate of shipments without an increase in cost [52]. The unpredictable demand patterns in donor funded supply chains would make scheduling and, consequently, collaboration exceedingly difficult [20]. In certain donor funded settings, such as nutrition and medical supply chains, the resources required for distribution are specific and require investment. However, by providing affordable goods and services, the revenue rates are relatively low, which results in poor investment. The delivery rates in donor funded supply chains are therefore lower, due to the inability to create distribution pools [52].

2.12 Agility

Agility, in a supply chain context, can be defined as the chain's ability to effectively perform in an unpredictable and changing environment [55]. In certain supply chain settings, the needs of the populations will change. Their need will either grow or decline, leading to changing priorities within the supply chain. Additionally, in donor funded settings, the dynamic nature of priorities and local conditions often require different responses, resources and capabilities than in a traditional commercial supply chain [1]. For example, when initially developing the supply chain there might not have been the necessary roads, ports, warehouses etc. and the supply chain would have been designed with these conditions in mind. However, if roads or ports are constructed over time, the supply chain should be redesigned to incorporate the changes in the conditions. There are several situations in which the responsibilities, workforce, routes and carriers and various other components of the supply chain had to be changed, adapted or redesigned to meet the changing conditions [38]. Commercial supply chains also change due to certain conditions, but rarely as rapidly (and sometimes frequently) as donor funded supply chains [1].

A supply chain with an agile strategy is continuously ready to change [56] and can promptly respond to changes in demand and the market [57]. The supply chain strategy, however, must be aligned with the business strategy and goals and tailored to meet customer requirements. Therefore, the unique characteristics of commercial and donor funded supply chains will affect their supply chain strategy and processes [13]. While both commercial and donor funded supply chains encourage agility throughout the chain, their reasons for doing so are often different. For commercial supply chains, agility is a key strategy that allows them to compete in the global market by improving delivery rates, operational performance and cost performance, among other things [58, 59]. Since the environment of donor funded supply chains is typically characterised by uncertainty and several actors [60], they practice agility in managing different relationships between donors and actors [61], evaluating impacts of distributed supplies and monitoring various ongoing activities [51]. The upstream activities and processes in donor funded supply chains are often acknowledged for its agility [38].

3. SUMMARISED COMPARISON OF KEY CHARACTERISTICS

In summary, the biggest differences between the two supply chains can be ascribed to the four following causes associated with donor funded supply chains:

• The nature of the demand, especially its instability and unpredictability.



- The overarching goals of reducing human suffering, saving lives and improving quality of life which leads to different actions and decisions than in profit-oriented commercial supply chains.
- The end-consumer is not a source of revenue and sometimes does not even take part in a transaction, instead the main source of income comes from donors.
- The often challenging terrain, regulations and infrastructure associated with some of the populations in need.

For the sake of clarity, the most salient differences between typical donor-funded supply chains and commercial supply chains, discussed in Section 2, are summarised in Table 2.

Characteristics		Commercial SC	DFSC (general)
Structure & Components		Upstream (global) segment and downstream (domestic) segment primarily managed as two coupled segments by primary stakeholders	Upstream (global) segment and downstream (domestic) segment are ordinarily decoupled from one another
	Primary Stakeholders	Shareholders (that guide the organisation's policies, goals and decisions)	Host Government and some NGOs
Stakeholders	Other Stakeholders	Several stakeholders with different needs - rarely conflicts with shareholders interests	Stakeholders have different, often conflicting, goals, missions, interests, capacity and constraints with different perspectives and approaches
Trust		Spend time and resources to build lasting relationships with their partners	There is often not enough time or resources to enable the building of trust
Goals & Objectives	Main Goal	To make profits and provide financial returns to shareholders	To produce value for money, be efficient and effective, ensure fair competition between suppliers, ensure accountability, and ensure procedures are done ethically
	Objectives	'Owners' of the supply chain share the organisations policies, goals and decisions	Each organisation involved in the supply chain strives to achieve its own purpose and mission
	As an objective	Strategic objectives are based on the financial returns paid out to the shareholders and the value created through delivering high quality goods and services to the consumers	Finances are seen as a constraint rather than an objective
rinances	Revenue	Income earned from the sale of goods and services	Government funding and donations from individuals and organisations
	Financial Supply Chain	Manages payment transactions and orders collectively	Often has a separate chain for international flows and domestic flows
	Customer characteristics	Any individual or organisation that buys and receives the product or service	Often seen as the end- consumers that receive the intended aid, but some view the donors as the customers
Customer Service & Marketing	Choice of product	End-consumer can analyse the market and choose which product, out of several options, to buy	End-consumers often do not have a choice of product
	Target	Market their product to the customer	Targets donors and convince them to provide some contribution
	Market segmentation	Large amounts of time and money is allocated towards market segmentation - in most cases dedicated departments are assigned to segment the consumers and market the product/service according to the segments.	Doesn't spend time and money to market product differently to market segments - since the consumers are in need of the product, the goal is to provide supplies and services to populations in need.

 Table 2: Summary of characteristics for commercial and donor funded supply chains



brance br
--

Research & Dev	elopment (R&D)	Financial resources are typically allocated to both strategic and operational activities with large amounts of time and money allocated for R&D	Typically, funding is provided specifically for operational activities and not R&D. Instead, they have to rely on global initiatives for their R&D.
Domand &	Stability	Relatively stable and predictable demand	Irregular amounts at irregular intervals
Forecasting	Demand Pattern	Occurs from recognised locations in fixed quantities at consistent time intervals	Often unpredictable and occurs in irregular amounts and time intervals
Procurement		Often build a lasting relationship with their partners and set up long-term agreements and contracts	Often use a competitive bidding process and short- term contracts or agreements are set up
	Processes	Processes can be forecast- driven due to the relatively stable demand patterns	Processes are mostly order- driven due to the unpredictable demand
Manufacturing	Volume	Typically uses economic batch sizes and order quantities. One manufacturer often has several customers and can produce larger volumes.	Supplies are manufactured in smaller batch sizes that are not cost effective, since the limited actual demand and lack of accurate forecasting weakens the manufacturers' confidence to produce large volumes of supplies.
	Market & competition	Manufacturers rarely have a monopoly in the market	The difficult nature of products and lack of incentives leads to a restricted market structure and lack of competition.
Logistics	Definition	Process of managing flows from source to the final customers	Process of managing flows from the donors to the affected populations
Logistics	Collaboration	Often implements horizontal collaboration, to create a distribution pool	Unpredictable demand patterns makes collaboration difficult
Agility		A key strategy that allows them to compete in the global market by improving delivery rates	Used in managing different relationships between donors and actors, evaluating impacts of distributed supplies and monitoring various ongoing activities

4. CONCLUSION

Several examples of donor funded supply chains were compared with commercial supply chains, with reference to several relevant characteristics. The characteristics that were discussed were primarily selected based on their identification in at least two well-known models, with three characteristics that do not meet these criteria also being discussed due to their salience. Although alternative characteristics are certainly discussed in other models and literature, those encountered and studied throughout the course of this research can confidently be categorised under the characteristics presented in Table 1.

This article mainly served as an exploratory high-level study, aimed at defining the field of donor funded supply chains in terms of how the main drivers and characteristics differs from commercial supply chains. Further value can be brought to the field by investigating how these differences should be taken into account when developing management practices and principles for donor funded supply chains.

REFERENCES

- Whybark, D.C., Melnyk, S.A., Day, J. and Davis, E. 2010. Disaster Relief Supply Chain Management: [1] New Realities, Management Challenges, Emerging Opportunities, Decision Line, May.
- Peng, M., Chen, H. and Zhou, M. 2014. Modelling and Simulating the Dynamic Environmental Factors [2] in Post-Seismic Relief Operation, Journal of Simulation, 8(1), pp 164-178.
- Mentzer, J.T., Keebler, J.S., Nix, N.W., Smith, C.D. and Zacharia, Z.G. 2001. Defining Supply Chain [3] Management, Journal of Business, 22(2), pp 1-25. Croxton, K.L., García-Dastugue, S.J., Lambert, D.M. and Rogers, D.S. 2001. The Supply Chain
- [4] Management Processes, The International Journal of Logistics Management, 12(2), pp 13-36.



- [5] Holmstrom, J., Framling, K., Kaipia, R., and Saranen, J. 2002. Collaborative Planning, Forecasting and Replenishment (CPFR®), Supply Chain Management: An International Journal, 7(3), pp 136-145.
- [6] Naslund, D. and Williamson, S. 2010. What is Management in Supply Chain Management A Critical Review of Definitions, Frameworks and Terminology, *Journal of Management Policy & Practice*, 11(4), pp 11-28.
- [7] Amemba, C.S. 2013. The Effect of Implementing Risk Management Strategies on Supply Chain Performance: A Case of Kenya Medical Supplies Agency, European Journal of Business and Management, 5(14), pp 1-16.
- [8] Davis-Sramek, B., Fugate, B.S. and Omar, A. 2007. Functional/Dysfunctional Supply Chain Exchanges. International Journal of Physical Distribution & Logistics Management, 37(1), pp 43-63.
- [9] Campuzano, F. and Mula, J. 2011. Supply Chain Simulation: A System Dynamics Approach for Improving Performance, Springer.
- [10] Tracey, M., Lim, J.S. and Vonderembse, M.A. 2005. The impact of supply-chain management capabilities on business performance, Supply Chain Management: An International Journal, 10(3), pp 179-191.
- [11] Shepherd, C. and Günter, H. 2011. Measuring Supply Chain Performance: Current Research and Future Directions, International Journal of Productivity and Performance Management, 55(3/4), pp 242-258.
- Ballou-Aares, D., Freitas, A., Kopczak, L.R., Kraiselburd, S., Laverty, M., Macharia, E. and Yadav, P. 2008. Private Sector Role in Health Supply Chains: Review of The Role and Potential for Private Sector Engagement in Developing Country Health Supply Chains.
- [13] Beamon, B.M. and Balcik, B. 2008. Performance Measurement in Humanitarian Relief Chains, International Journal of Public Sector Management, 21(1), pp 4-25.
- [14] Nicholson, A., English, R.A, Guenther, R.S. and Claiborne, A.B. 2013. Developing and Strengthening the Global Supply Chain for Second-Line Drugs for Multidrug-Resistant Tuberculosis: Workshop Summary, National Academies Press.
- [15] Stephenson, M. 2005. Making Humanitarian Relief Networks More Effective: Operational Coordination, Trust and Sense Making, *Disasters*, 29(4), pp 337-350.
- [16] Chandraprakailkul, W. 2010. A guiding framework for designing humanitarian relief supply chains A case study in Thailand, *Proceedings of the Production and Operations Management Society 21st Annual Conference*, pp 1-22.
- [17] Speckbacher, G. 2003. The Economics of Performance Management in Nonprofit Organizations. Nonprofit Management and Leadership, 13(3), pp 267-281.
- [18] Cozzolino, A. 2012. Cross-Sector Cooperation in Disaster Relief Management, Springer.
- [19] Moore, M.H. 2000. Managing for Value: Organizational Strategy in for-Profit, Nonprofit, and Governmental Organizations, *Nonprofit and Voluntary Sector Quarterly*, 29(1), pp 183-204.
- [20] Balcik, B., Beamon, B.M., Krejci, C.C., Muramatsu, K.M. and Ramirez, M. 2010. Coordination in Humanitarian Relief Chains: Practices, Challenges and Opportunities. International Journal of Production Economics, 126(1), pp 22-34.
- [21] Kumar, S. and Havey, T. 2013. Before and After Disaster Strikes: A Relief Supply Chain Decision Support Framework, International Journal of Production Economics 145(2), pp 613-629.
- [22] Kovács, G. and Spens, K.M. 2007. Humanitarian Logistics in Disaster Relief Operations, International Journal of Physical Distribution & Logistics Management, 37(2), pp 99-114.
- [23] Schliephake, K., Stevens, G. and Clay, S. 2009. Making Resources Work More Efficiently The Importance of Supply Chain Partnerships, *Journal of Cleaner Production*, 17(14), pp 1257-1263.
- [24] Bullington, K.E. and Bullington, S.F. 2005. Stronger Supply Chain Relationships: Learning from Research On Strong Families, Supply Chain Management: An International Journal, 10(3), pp 192-197.
- [25] Nwuneli, N., Robinson, E., Humphrey, J. and Henson, S. 2014. The Role of Businesses in Providing Nutrient-Rich Foods for the Poor: Two Case Studies in Nigeria, Sahel Capital Partners and Advisory Ltd., Institute of Development Studies.
- [26] Kamau, C.W. 2013. Humanitarian Supply Chain Management in Kenya.
- [27] Boland, T. and Fowler, A. 2000. A Systems Perspective of Performance Management in Public Sector Organisations, *The International Journal of Public Sector Management*, 13(5), pp 417-446.
- [28] Baruch, Y. and Ramalho, N. 2006. Communalities and Distinctions in the Measurement of Organizational Performance and Effectiveness Across For-Profit and Nonprofit Sectors, Nonprofit and Voluntary Sector Quarterly, 35(1), pp 39-65.
- [29] Kaplan, R.S. 2001. Strategic Performance Measurement and Management in Nonprofit Organizations, Nonprofit Management & Leadership, 11(3), pp 353-370.
- [30] Van Wassenhove, L.N. 2005. Humanitarian Aid Logistics: Supply Chain Management in High Gear, Journal of the Operational Research Society, 57(5), pp 475-489.
- [31] Henderson, D.A., Chase, B.W. and Woodson, B.M. 2002. Performance Measures for NPOs, Journal of Accountancy, 193(1), pp 63-68.
- [32] Kristofik, P., Kok, J. and De Vries, S. 2012. Financial Supply Chain Management Challenges and Obstacles, ACRN Journal of Entrepreneurship Perspectives, 1(2) pp 132-143.



- [33] Maon, F., Lindgreen, A. and Vanhamme, J. 2009. Developing Supply Chains in Disaster Relief Operations Through Cross-Sector Socially Oriented Collaborations: A Theoretical Model, Supply Chain Management: An International Journal, 14(2), pp 149-164.
- [34] Russell, T.E. 2005. The Humanitarian Relief Supply Chain: Analysis of the 2004 South East Asia Earthquake and Tsunami by Master of Engineering in Logistics.
- [35] Gustavsson, L. 2003. Humanitarian Logistics: Context and Challenges, Forced Migration Review, 18(1), pp 6-8.
- [36] Popa, V. 2013. The Financial Supply Chain Management: A New Solution for Supply Chain Resilience, Amfiteatru Economic, 15(33), pp 140-153.
- [37] Day, J.M., Junglas, I. and Silva, L. 2009. Information Flow Impediments in Disaster Relief Supply Chains, Journal of the Association for Information Systems, 10(8), pp 637-660.
- [38] Oloruntoba, R. and Gray, R. 2006. Humanitarian Aid: An Agile Supply Chain?, Supply Chain Management: An International Journal, 11(2), pp 115-120.
- [39] Howden, M. 2009. How Humanitarian Logistics Information Systems Can Improve Humanitarian Supply Chains: A View from the Field, Proceedings of the 6th International Systems for Crisis Response and Management Coference.
- [40] Afshar, A. and Haghani, A. 2012. Modeling Integrated Supply Chain Logistics in Real-Time Large-Scale Disaster Relief Operations, Socio-Economic Planning Sciences 46(4), pp 327-338.
- [41] Chen, F. 2001. Market Segmentation, Advance Demand Information, and Supply Chain Performance, Manufacturing and Service Operations Management, 3(1), pp 53-67.
- [42] Chomilier, B., Samii, R. and Van Wassenhove, L.N. 2003. The Central Role of Supply Chain Management at IFRC, Forced Migration Review, 18(1), pp 15-16. [43] Krasnikov, A. and Jayachandran, S. 2008. The Relative Impact of Marketing, Research-and-
- Development, and Operations Capabilities on Firm Performance. Journal of Marketing, 72(7), pp 1-11.
- [44] So, D.A. and Vickery, C.M. 2009. A Supply Chain Analysis of Ready-to-use Therapeutic Foods for the Horn of Africa.
- [45] Komrska, J., Kopczak, L.R. and Swaminathan, J.M. 2013. When Supply Chains Save Lives, Supply Chain Management Review, pp 42-49.
- [46] Balcik, B. and Beamon, B.M. 2005. Distribution Network Design for Humanitarian Relief Chains, Institute for Operations Research and the Management Sciences Annual Meeting Proceedings.
- [47] Haghani, A. and Afshar, A.M. 2009. Supply Chain Management in Disaster Response.
- [48] Hovhanessian, M. 2012. Coordination Barriers between Humanitarian Organizations and Commercial Agencies in times of disaster.
- [49] Long, D. 1997. Logistics for Disaster Relief, IIE Solutions, 29(6), pp 26-29.
- [50] Quinn, F.J. 1997. What's the Buzz?, Journal of Logistics Management, 2(36), pp 43-47.
- [51] Yadav, D.K. and Barve, A. 2015. Analysis of Critical Success Factors of Humanitarian Supply Chain: An Application of Interpretive Structural Modeling, International Journal of Disaster Risk Reduction, 12(1), pp 213-225.
- [52] Yadav, P. 2011. Always Cola, Rarely Essential Medicines: Comparing Medicine and Consumer Product Supply Chains in the Developing World, INSEAD Social Innovation Centre.
- [53] Thomas, A.S. and Kopczak, L.R. 2005. From Logistics to Supply Chain Management: The Path Forward in The Humanitarian Sector, Fritz Institute.
- [54] Ernst, R. 2003. The academic side of commercial logistics and the importance of this special issue. Forced Migration Review, 18(1), pp 5.
- [55] Maskell, B. 2001. The Age of Agile Manufacturing Insight From Industry. Supply Chain Management: An International Journal, 6(1), pp 5-11.
- [56] Goldman, S.L., Nagel, R.N. and Preiss, K. 1995. Agile Competitors and Virtual Organizations: Strategies for Enriching the Customer, Long Range Planning, 29(1), pp 131.
- [57] Jain, V., Benyoucef, L. and Deshmukh, S.G. 2008. A New Approach for Evaluating Agility in Supply Chains Using Fuzzy Association Rules Mining, Engineering Applications of Artificial Intelligence, 21(3), pp 367-385.
- [58] Yusuf, Y.Y., Gunasekaran, A., Adeleye, E.O. and Sivayoganathan, K. 2004. Agile Supply Chain Capabilities: Determinants of Competitive Objectives, European Journal of Operational Research, 159(1), pp 379-392.
- [59] Eckstein, D., Goellner, M., Blome, C. and Henke, M. 2013. The Performance Impact of Supply Chain Agility and Supply Chain Adaptability: The Moderating Effect of Product Complexity, International Journal of Production Research, pp 2-45.
- [60] Besiou, M., Pedraza-Martinez, A.J. and Van Wassenhove, L. 2012. The Effect of Earmarked Funding on Fleet Management for Relief and Development, INSEAD Social Innovation Centre.
- [61] Kovács, G. and Spens, K. 2009. Identifying Challenges in Humanitarian Logistics. International Journal of Physical Distribution & Logistics Management, 39(6), pp 506-528.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



AN EVALUATION OF WASTE AND COSTS AT A TEXTILE FACILITY: A CASE STUDY

K Ramdass¹

¹Department of Mechanical and Industrial Engineering University of South Africa, South Africa <u>ramdakr@unisa.ac.za</u>

ABSTRACT

The textile industry continues to face challenges in view of realignment to the context of globalization and competition. The industry has a myriad of issues with which it has to contend with, which are tortuously entwined and symbiotic. It is a vehicle of economic development and growth and a major contributor to the GDP. In addition, the textile industry is a major provider of employment and a vital component in the supply chain of the clothing industry. The aim of the textile industry in South Africa (SA) is to harness its potential in terms of human capital, natural resources and technological impetus in order to become a preferred international supplier. The textile and clothing industry accounted for 14 % of manufacturing employment and a major contributor of tax revenue. The textile industry offers a range of services in terms of its diverse production facilities - from natural fibres to synthetic fibres to non-wovens. This includes spinning, weaving, tufting, knitting, dyeing and finishing. The methodology that was most appropriate for the evaluation was a case study because it provides the ability to infiltrate aspects of waste and costs in the organisation. The study enabled the collection of important information and data that can be used to improve the current experiences.

Key words: textile industry, economic development, production facilities, continuous improvement

¹Dr. Kemlall Ramdass is a senior lecturer at the Department of Mechanical and Industrial Engineering, University of South Africa.



1. INTRODUCTION

The textile industry continues to play a major role in terms of employment, however formal employment decreased by approximately 40% in the last decade [15]. Some of the reasons for this are the increasing wage rates of the sector with an adverse decline in productivity. In addition, the application of inefficient work methods, overpriced raw materials that are normally imported from around the world and dependant on the rand/dollar exchange, aged equipment that require repair on an ongoing basis that ultimately result in the industries uncompetitiveness [2]. The global arena bolsters the impact on the industry through cheap imports. Labour laws are perceived to favour the employee, adding the burden of an already struggling industry[3]. It was decided that an evaluation be done in order to address internal inefficiencies as well as exploring cross-industry innovation to determine if the waste can be recycled into products for a different market[6]. In view of the clothing and textile industry protection from imports, the industry is now placed in a predicament, where the cost factor in terms of production facilities are priority. The industry is continuously focused on cost-cutting measures to improve productivity [10;11;13].

2. LITERATURE REVIEW

The textile industry requires competitive strategies in order to maintain its position in the global economy. These strategies provide significant impact on the financial viability in terms of the organisations manufacturing capabilities[1;4]. It may be used as a competitive weapon against national and international markets vying for the same products.

The textile industry in particular may implement particular dimensions of competitive strategies in the context of the textile industry. Ward et al [14]postulates five magnitudes which he believes are imperative for for global competitiveness. These are product flexibility, volume flexibility, price, quality, dependability in order of importance in the textile industry.

Glock and Kunz [7] mentioned that the "winning of orders" is vital in the textile fraternity. Their focus is similar to Peng, however they focus on price, product design and variety. On the other hand Hill postulates performance of the product as a qualifier criterion in the textile industry. The adds to the organisations pursuit for service excellence in manufacturing.

Krajewski and Ritzman and Malhotra [9] mention "a more detailed list by differentiating four different aspects namely, cost; quality; time; and flexibility. Their list then included low-cost operations, high-performance design, consistent quality, fast delivery time, on-time delivery, development speed, customisation, and volume flexibility."

Stevenson [12] on the other hand recommends three areas of competitive advantage which he believes would lead the organisation to global competitiveness:

- cost leadership lean pricing to attract the market;
- differentiation an exceptional product that is in demand;
- focus an exclusive marketplace based on cost leadership and differentiation strategy.

In order to achieve the above mentioned strategies Ramdass indicates that it is imperative that the textile industry defines its worker performance in terms of the prevailing working conditions. In this argument, the ergonomics of the workstation and job design as well as the physique of the operator are critical components especially in the textile industry [18].

2. METHODOLOGY

A case study methodology creates an enabling environment in terms of understanding complex phenomena in terms of three stages. The first and most important one is its alignment to the research question. In this case the research question is "what is happening in the textile industry in terms of improvement strategies?" This would provide an exploration of the current status on the industry through a descriptive analysis. The second reason for the choice of the case study method is that it is a real world phenomenon where data is collected in the natural setting. Thirdly the methodology is appropriate in that it evaluates the current situation for the purpose of improvement.



3. RESULTS AND DISCUSSION

The paper focuses on four critical areas in the thread manufacturing process. The reasons for this are multifold and include human resources in terms of training, efficiency, quality and productivity. The four areas that are investigated include spinning, twisting, dyeing and final winding departments that are the core components of thread manufacturing process.

3.1 Spinning

Introduction to the spinning process

The spinning process has been expanded due to the fact that it is a critical process in textile manufacturing, especially in the manufacture of thread for the clothing industry. After the sliver has passed through the different speedframe processes in the preparation department, the end product, which is called roving has to be transformed into yarn. This is achieved by means of ring spinning frames, giving the roving its draft and twist. The roving passes through the drafting area, the guide eye and the traveler onto the bobbin. The roving speed of the bobbin is very high, taking the traveler and roving with it. Every revolution of the bobbin puts one turn in the length of roving between the traveler and the front drafting rollers. Due to the friction of the traveler against the traveler ring and the traveler lags a little behind the speed of the spindle causing the yarn to be wound onto the bobbin. For a given count the amount of twist in the yarn can be varied; this is:

- By increasing or decreasing the speed of the drafting rollers the more material is given the less the amount of twist.
- By using lighter or heavier travelers the further it stays behind in comparison with spindle speed, the quicker the up winding, the less twist is inserted.

The amount of twist inserted into the yarn depends on its end use as twist affects the strength of the yarn. Warp yarns must have more twist than yarns of the same linear density used for weft knitting. The direction of the spinning twist is normally "z."

The first issue that was observed was the high levels of waste in the spinning area [5]. On analyzing the doff weights, it was found that there was an average of a 35% variation between doff weights of the same count as well as that the weight was below the expected weight.

The waste and inefficiency was the result of the following:

- Lack of operator patrol
- Spindles broken
- Faulty spinning
- Tight ring tubes
- Spindles not started after doff (changes on a regular basis)

Table 1: Ring Spinning Machine Analysis (waste)

MACHINE	NO.OF SPINDLES	NO NOT WORK	LOSS IN GRAMS	LOSS PER MONTH
Machine 1	420 spindles	15 (100 GRAMS)	1500 (120)	180000
Machine 2	420 spindles	25	2500	300000
Machine 3	420 spindles	20	2000	240000
Machine 4	420 spindles	16	1600	192000
Machine 5	420 spindles	45	4500	540000
Machine 6	420 spindles	35	3500	420000
Machine 7	420 spindles	37	3700	444000
Machine 8	420 spindles	41	4100	492000
Machine 9	420 spindles	34	3400	408000
Machine 10	420 spindles	43	4300	516000
Machine 11	420 spindles	33	3300	396000
Machine 12	420 spindles	26	2600	312000
Machine 13	420 spindles	37	3700	444000
Machine 14	420 spindles	25	2500	300000
Machine 15	420 spindles	27	2700	324000
Machine 16	420 spindles	10	1000	120000
				5628000 (5628)







Figure 2: Loss of production in grams





Figure 3: Loss of production per month

Human Resources

- In terms of the ergonomics of the spinning operation the following was identified: .
- The loading of machines required tall men (due to the weights that required lifting when loading the machines)
- The unloading of the machine required shorter men due to bending required when offloading spindles.
- Transport and weighing required bin with wheels as well as durable material for the construction of the bin.

Recommendation to HR

- Employment of men as per criteria for the different operations
- Development of a bin for the doffing of machines
- Pattern of patrol of machines •

Quality

The development of a quality plan that samples the output of each machines at regular intervals.

Machine maintenance

It was recommended that there be the repair and replacement of all worn parts as they occur and the total refurbishment at shutdown which is in December.

This evaluation was applied to all sections that were studied.

3.2 Grey Mill - Hamel stage 2 twisting

The twisting process generally involves threads being put together and twisted to form the respective specifications required by the customer. Twisting is generally satisfactory but machine wear in terms of the peeling of protective plating on both top and bottom rollers was a cause for concern.

Problems that were observed in this area are the following:

- Inefficient Operator patrol it was observed in the sample that too many spindles were not in operation.
- Cheese shape output cheese shape must be parallel variable tension would result in uneven shape .
- Mandrells are problematic and cradles must be in line and in good working order.
- Differing dimensions of stainless steel springs in use required correction.



3.3 Gilbos winding

Problems that were observed in this area are the following:

- Inefficient operator patrol It was observed in the sample that too many spindles were not in operation.
- Thread running through deep cuts in the tensions resulting in damaged thread.
- Oil deposits on outputs from leaking damper
- Knock-offs were not working properly, causing larger packages that pick up dirt.

3.4 Hamel down-twisting

Problems that were observed in this area are the following:

- Ceramic guides broken and damaging thread
- Thread was running against metal
- Thread not positioned behind guide rod
- Rusty rollers contaminate thread
- Incorrect distance between twisting bobbin and metal guide
- Variation of travelers used on one machine
- Dirty machinery

3.5 Dyehouse

To maintain high standards of quality and to meet the specified requirements, it is important that the 'fast dyed' threads have the highest colour fastness attainable. Colour laboratory recipes are formulated with this aim in mind and selection of dyestuffs restricted to those which are capable of producing the required levels of colour fastness. However certain fastness characteristics of dyed material are dependent on both the recipe formulation and the dyeing process, and routine fastness testing of bulk dyelots was necessary. In this regard the dyehouse is automated but relies for the correct proportions in terms of dyestuffs going into the dye vessel. Should there be colour problems the batch is redyed into black to prevent waste. At this stage, all machinery is maintained accordingly. However, if a lot is dyed and the colour is not correct, the entire lot is scrapped.

3.6 Final winding - Scharer Lubrication

An important stage in the final winding process is the lubrication process which is imperative for the clothing industry. During the sewing process, the needle becomes hot due to friction, therefore the thread requires lubrication to prevent thread breaks on a regular basis.

This is a messy process, therefore machine cleaning is required on a regular basis. The following was observed in the process:

- Regular thread breaks that lead to waste due to inefficient patrolling of operators
- Different types of empty centres were used in the lubrication process, one much thicker than the other and causing problems in terms of tension. This needs to be adjusted by the mechanic, alternatively use the same thickness of centres
- Machine checks on a daily basis due to the process itself
- Complete installation of new clearers
- Damaged centres causing excess breaks

Leeson winding

Problems that were observed in this area are the following:

- Missing ceramic eyelets above the creel peg
- Damaged tension disc springs
- Reposition bail roller arm where necessary
- Leaking wax troughs
- Ergonomics of label insertion onto vicone



3.7 SPT Cop winding

Problems that were observed in this area are the following:

- Make up and inspection need re-evaluation
- Machines are in good working order

SPT cone winding

Problems that were observed in this area are the following:

- Input creel peg needs attention
- Work-layout requires investigation in terms of ergonomics •
- Material handling to next stage
- Lubrication needs to be maintained
- New clearers installed
- Output centres are damaged

Hacoba vicone winding

Problems that were observed in this area are the following:

- New work methods required
- Programme settings for the machine on day shift only •
- Thread breaks were frequent caused by thread sloughing off machine •
- Optimum speeds require re-evaluation
- Cutting knife be moved forward to enable easier doffing
- Tensions on the thread need evaluation as diameter of cones vary.
- Quality of empty vicones are problematic and need investigation

Hacoba cone winding

- Plunger housing to be checked and replaced as necessary •
- Wing assemblies, that is the traverse guides need evaluation and replacement

ANALYSIS 4.

A disturbing issue that was observed was that a person was specifically hired to cut up waste. With reference to the Table 2 below the approximated waste per annum for the organisation was 7000 kilograms. This was done for over 10 years. The monetary value that was wasted in the form of waste was insurmountable. The graph shows that most of the waste was generated in the spinning department.

In the endeavour to reduce waste the following action was taken from an industrial engineering perspective:

- Define the problem generation of waste in each process of manufacturing •
- Identify the causes of waste input/process/output •
- Identify the principal contributor of waste
- Develop a strategy to address and curtail waste creation

The following table provides the waste per department

Table 2: Breakdown of waste and costs per department				
	SPINNING	TWISTING	DYEING	FINISHING
JANUARY	57 (16.50)	125 (18.00)	300 (25.00)	85(33.00)
FEBRUARY	128	400	235	56
MARCH	255	350	255	86
APRIL	85	352	145	120
MAY	120	285	186	95
JUNE	95	125	85	51
JULY	112	156	128	123
AUGUST	100	115	201	76

Table 2: Breakdown of waste and costs per depa	artment
--	---------



SEPTEMBER	68	125	98	102
OCTOBER	125	300	129	84
NOVEMBER	80	250	158	100
DECEMBER	138	96	100	63
	1363 (22489)	2523 (45414)	2020 (50500)	1042 (34386)



Figure 4: Waste per department per month

In terms of cost the dying department was a major contributor. In terms of quantity the twisting department was the major contributor of waste. Management decision on the matter was that that the project will start in the spinning department because the input from spinning is important for the other processes.

SPINNING	TWISTING	DYEING	FINISHING	
Filament damaged/cut	Incorrect twist	Incorrect colour	Operator error	
No cotton	Wrong ply	Machine malfunction	Machine error	
Defective material	Defective machine	Operator error	Quality standards not	
			met	
Defective rollers	Quality standards not	Quality standards not		
	met	met		
Worn rollers	Variance in spindle turns	Defective input material		
Defective input material	Operator error			
Quality standards not				
met				

Table 3. Breakdow	n of causes of waste

A programme was developed to address the above problems. Regular discussion on the 5s principles was implemented and drastic measures were taken to reduce waste throughout the organisation. The maintenance team was asked to investigate and correct all machine related problems. Human resources performed training and retraining on all operator-related problems and the work-study department provided standard operating procedures for all operations. The quality department performed quality tests on all input/output of materials on a regular basis. Weekly productivity and operator efficiency reports, together with the waste report were presented to executive management since inception. This was carefully scrutinised and each departmental manager was asked to report on the progress that was made using the 5s principles.



5. CONCLUSION

Commitment from all stakeholders is imperative in order to gain the rewards of this programme. Moreover, the rewards may be seen over a period of time and difficult to quantify. However, this case demonstrates the saving gained on the reduction of waste in the textile industry.

Discipline means instilling the ability to do things the way they are supposed to be done. This involves training and inculcating good habits and having everyone practise them, thus encouraging the continuity of good habits. As the final S in the 5-S, it propagates the 5-S practice by ensuring that the former 4-Ss are carried out conscientiously. Making the 5-S a daily routine helps to maintain orderliness at the workplace. Setting up the 5-S is useless if it is not followed through, as things will return to its previous state if they are not well maintained. Taking a further step, discipline also includes reviewing current practices and revising them to keep them relevant. It also means striving for kaizen, which is the Japanese equivalent for continuous improvement.

There is a definite relationship between worker performance and working conditions. They found that the problems with most workstations centred on two design-related problems - the physical components of the workstation and the nature of the job or job design. In the case of this study, it was important to hire personnel that are able to reach the top of ring spinning machines. Today's workforce is multiracial, male and female, and includes disabled and ageing workers. Body shapes, sizes, capabilities and limitations vary greatly, yet many workstations do not take these factors into consideration, as they are set up in a rigid structure. To compound these problems, workplace physical structures may be designed for a minority of workers.

A "good" workstation, (one that promotes machine-worker interface) therefore, should be developed with the anthropometric measurements of the majority of workers as a standard and should allow flexibility. In addition to these physical layout problems, the pace of work and the type of work has changed. Employees are expected to produce more output quickly, and, because of the faster access to information, employees are being asked to make more and more decisions around their tasks. These factors are potential stress builders [3;14], however the encouragement of staff to take ownership of their process creates an environment of collegiality that encompasses building relationships.

Lastly, it is recommended that machine maintenance and the measurement of productivity be implemented as this would be able to monitor all aspects in the manufacturing process.

REFERENCES

[1] Amdam, R. P., Lunnan, R., & Ramanauskas, G. (2015). FDI and the transformation from industry to service society in emerging economies: A Lithuanian-Nordic perspective. Engineering Economics, 51(1).

[2] Bruce, M., Daly, L., & Towers, N. (2004). Lean or agile: a solution for supply chain management in the textiles and clothing industry?. International journal of operations & production management, 24(2), 151-170.

[3] de Abreu, M. C. S., de Castro, F., de Assis Soares, F., & da Silva Filho, J. C. L. (2012). A comparative understanding of corporate social responsibility of textile firms in Brazil and China. Journal of Cleaner Production, 20(1), 119-126.

[4] Danskin, P., Englis, B. G., Solomon, M. R., Goldsmith, M., & Davey, J. (2005). Knowledge management as competitive advantage: lessons from the textile and apparel value chain. Journal of Knowledge Management, 9(2), 91-102.

[5] Depoers, F. (2000). A cost benefit study of voluntary disclosure: Some empirical evidence from French listed companies. European Accounting Review, 9(2), 245-263.

[6] Enkel, E., & Gassmann, O. (2010). Creative imitation: exploring the case of cross-industry innovation. R&d Management, 40(3), 256-270.

[7] Glock, R. E., & Kunz, G. I. (2005). Apparel manufacturing: Sewn product analysis. Prentice Hall.

[8] Kirner, E., Kinkel, S., & Jaeger, A. (2009). Innovation paths and the innovation performance of low-technology firms—An empirical analysis of German industry. Research Policy, 38(3), 447-458.

[9] Krajewski, L. J., Ritzman, L. P., & Malhotra, M. K. (2007). Operations management: processes and value chains.

[10]Nordås, H. K. (2004). The global textile and clothing industry post the agreement on textiles and clothing. World, 7(1,000).

[11]Porter, M. E. (2011). Competitive advantage of nations: creating and sustaining superior performance. Simon and Schuster.

[12] Stevenson, W. J. Operations Management, 2007. McGraw Hills, USA.

[13] Viotti, E. B. (2015). Technological learning systems, competitiveness and development.

[14] Ward, P.T. Durray, R. 2000. Manufacturing strategy in context: environment, competitive strategy and manufacturing strategy, Journal of Operations Management, 18, 2, 123-38.

[15] http://www.texfed.co.za/ Textile industry report



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



EMPLOYEE RETENTION STRATEGIES: FACTORS FOR GENERATION Y BURSAR GRADUATES

Bontle Tladi

Department of Industrial Engineering at University of the Witwatersrand

bontle.tladi@sasol.com

ABSTRACT

The nature of work is changing, and so are the elements which are necessary to manage these different transformations.

The focus of this study was to determine which factors were important to Generation Y engineering bursar graduates from a particular organisation; in particular those within their first 5 years of work. A theoretical framework from the literature was developed, and the Case Study research method was applied.

The results showed that *Job satisfaction* had the greatest influence as a retention factor. It was found that there was a difference between factors which influenced an employee to stay or leave.



1. INTRODUCTION

The nature of work is changing, and so are the elements which are necessary to manage the different transformations - such as workforce demographics, awareness of social and environmental corporate responsibility which are taking place. Some of the contributing factors to these changes are globalisation, technology, where and how people work, as well as workforce expectations [1] [2] [3] [4]. As a result of these transformations taking place in the workplace, the risk of a disengaged workforce is high; and hence the implication for employee retention strategies. The external factors (such as economic crises, the dynamics in the labour market, and even political instability), have a profound and direct impact on the internal changes which take place within individuals. Workplaces are changing from being areas of generally economic and social activity to being places where individuals are seeking meaning, value, support and purpose to their lives [5].

In light of the transformation which is taking place in the workplace arena, it is evident that financial incentives and/or financial security are not sufficient as sole "retention drivers"- there are other factors which need to be considered to drive and support organisational sustainability through these changes [6].

1.1 Problem Statement

The organisation which forms the focus of this study runs a bursary scheme program. This program aims at developing young talent, and providing an opportunity for that talent to be part of the organisation. After the completion of studies, the expectation is that the bursary holder "work back" the period over which they were sponsored.

The early loss of bursaried graduates has implications for the organisation in the long run. It has been shown that 'tenured employees drive far greater value than those who are "cycling through" the business' [6]. The investigation of the critical factors affecting this early loss of bursary graduates has the potential to benefit the development and application of certain strategies to address the retention, as well as the growth, of incoming talent to the business.

The purpose of this study, thus, was to determine what factors are essential in the development of retention strategies (those which respond to the workplace transformation(s) taking place); with specific regard to engineers within their first 5 years of work.

1.2 Research Question(s)/Hypotheses

The focus of the study was on employees from a particular organisation, and the research question was framed as:

What factors are important to the young engineer (within 5 years working experience) in the work context, which could contribute to the development of effective retention strategies for this segment of the workforce?

1.3 Research Objectives

The objectives of the research were to:

- Develop a conceptual framework from the literature incorporating relevant factors that may affect the retention of young engineers in the organisational workforce.
- Use the conceptual framework to develop survey questions, and to frame the responses from the interviews, so as to elicit key factors
- Establish which factors are essential in the development of employee retention strategies for young engineers who have joined the organisation through the organisation's bursary scheme (within 5 years working experience)
- Recommend how the current employee retention strategies need to accommodate the above factors
- Comment on how the findings relate to the role of spirituality in the workplace, and the consequent implications for employee retention strategies

1.4 Limitations and Constraints

The study and its findings were limited by the following:-



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE

- It was not possible to include employees who had left the company in the study, as their contact details were not available.
- The information (separation data) related to the employees who had left the company was limited to the history capacity of the data system being used to capture these details (i.e. it was not possible to access the historic data from before the system was implemented).
- The study was also limited to the number and the type of respondents that completed the survey (responses to the survey were optional to the recipients of the survey).

A Literature Review was conducted which covered the following areas:-

- The changing workplace
- Spirituality at work
- Knowledge workers
- Employee Retention Strategies
- Generation Y at work

2. DEVELOPMENT OF A THEORETICAL FRAMEWORK

A study by [7] developed a *Content Model for Employee Retention* from a review of the major theories over the last fifty years, which helped explain why employees stay with an organisation or leave. The reviewed literature from this study was used to generate a theoretically-derived set of factors.

The literature on (i) Generation Y, (ii) Knowledge workers, and (iii) Spirituality at work was surveyed to determine relevant factors that may affect the retention of young engineers. It was evident from the comparison of the area-specific factors that there is some overlap of factors in some of the areas. An Affinity Diagram approach was followed to group the various factors into broad themes. These consequent themes are presented in Table 1.

Table 1: Retention Factors used in the Survey

(Definitions of the factors from the *Content Model for Employee Retentions* study have been used as defined in the study [7]. The factors emanating from the Affinity Diagram were defined as suggested by the perused literature).

Retention Factor	Definition
Advancement opportunities	The amount of potential for movement to higher
	levels within the organization
Constituent attachments	The degree of attachment to individuals associated
	with the organization such as supervisor, co-workers,
	or customers
Extrinsic rewards	The amount of pay, benefits, or equivalents
	distributed in return for service
Flexible work arrangements	The nature of the work schedule or hours
Investments	Perceptions about the length of service to the
	organization
Job satisfaction	The degree to which individuals like their jobs
Lack of alternatives	Beliefs about the unavailability of jobs outside of the
	organization
Location	The proximity of the workplace relative to one's
	home
Non-work influences	The existence of responsibilities and commitments
	outside of the organization
Organizational commitment	The degree to which individual's identify with and are
	involved in the organization
Organizational justice	Perceptions about the fairness of reward allocations,
	policies and procedures, and interpersonal treatment
Organizational prestige	The degree to which the organization is perceived to
	be reputable and well-regarded
Growth	The prospect of growth in various facets (e.g.
	employee, human, personal); self-actualisation



Energising environment	An energising culture, in which aspects such as creativity, excitement, fun, inspiration, transformation are active
Learning organisation	An environment where continuous learning is encouraged; programs available such as mentorship, skills development, training, etc.
God in the organisation	An environment which is "soulful", which contains "spiritual" values, has an element of sacredness
Values	An organisation which is grounded in values such as morality, respect, authenticity, open communication, transparency; where a greater vision is communicated and shared
Nature of work	Work in which the job design is aligned with the employee's personal life interests, is challenging, knowledge-based work, appropriate job design
Flexibility	An environment which allows for autonomy, independence, ease of movement, work-play balance
Career path	An organisation which offers a clear career growth and development plan; a career path aligned with personal life interests; targeted development programs
CSR/ Civic activities	A caring company, which is active in community development/ has a reputable CSR program
Intrinsic reward systems	Being rewarded in a manner which results in personal satisfaction (e.g. recognition; sense of meaning, choice, competence, progress)
Meaning	Finding purpose, relevance, inner-connectedness, etc. in the work that is being done
Technology	An organisation which is technologically-savvy, has leading technology, innovative
Leadership/ Management	The type of organisation leadership/ management style e.g. a cool and laidback management style, servant leadership, etc.
Workplace relationships	An environment in which the workplace is a community; promotes teamwork, comradeship, networking, cohesion

3. RESEARCH METHODS

A Case Study of the Type 2 design [8] was used.

The following data collection methods were used (as indicated in Table 2):

Table	2:	Sources	of	Evidence
-------	----	---------	----	----------

Source of Evidence	Type of Evidence
Documentation The purpose of this evidence was to determine what employee retention strategies were employed by the organisation.	 Previous Graduate Development Programme (GDP) Policy Current GDP Policy Company's Talent Management (TM) Department documentation Company's website
Archival records	 Separation data (of bursar graduates who had separated from the company)
The purpose of this evidence was to determine the reasons for separation by previous employees	
<u>Survey</u>	• Survey results (Likert scale was used)
The purpose of this evidence was to determine which retention factors would cause an employee to stay and/or leave the organisation	



Interviews	Semi-structured interviews with:
The purpose of this evidence was to supplement the information from the perused documentation. Interviews were held with the custodians of the various elements of the organisation's employee retention strategies so as to better understand the context of the documentation.	 Acting Manager: Bursary Services Office Senior Manager: Talent Management (People and Organisational Effectiveness) Senior Manager: Graduate Development Programme Office Head of Wellness and Benefits

Validity and Reliability

To ensure research validity, the following were applied:

Construct validity:

- Multiple sources of evidence were used
 - Documentation
 - Archival records (Separation data)
 - o Survey
 - o Interviews
- A Case Study Database was created to provide a chain of evidence

External validity

• The results were related back to the Literature that was reviewed (in order to generalise the results to the broader theories)

Reliability:

- A Case Study Protocol was used
- A Case Study Database was created

4. RESULTS

The aim of this research was to determine which factors would be required in the development of retention strategies for young engineers (those between 0 - 5 years of work); in particular, graduates who had joined the organisation through the bursary scheme.

A conceptual framework of these factors was drawn up, incorporating retention factors which were mentioned in the Employee retention strategies, Knowledge workers, Generation Y and Spirituality at work literature. The framework was used to develop survey questions which were posed to the employees at the organisation, who had joined the company through the bursary scheme. All respondents were considered to be part of Generation Y (Gen Y) in this study as the year brackets are not strictly bound (the last generation with strict time-frames is the Baby Boomer 1946 - 1964 [9]).

The factors were tested for personal congruence (PC) ("what would make you stay with the company?") and personal incongruence (PI) ("what would make you leave the company?"). The collected data was compared according to:-

- Tenure (engineers who had been with the company for less than five years, and those who had been employed for more than 5 years)
- Gender

Analysis of data

The responses from the Likert scale data was organised to reflect the modes and respective percentages.

The respondents had the option to add comments in addition to indicating their level of congruence with the given factor. The nature of the received comments differed: some of the comments were related to the included factors, others were more general comments on the design and development of employee



retention strategies, some were commentary on the organisation's methods, while some had a more personal commentary (i.e. what factors were important for the particular individual).

Inferential statistics (Mann-Whitney U test) were applied to determine whether there was a difference between the populations which formed the focus of this study (employees who had been with the company for i) more, and ii) less than 5 years), as well as between the genders.

A significance level of 0.05 was applied; with the null hypothesis being that there was no significant difference between the responses. Both uni-directional and non-directional tests were performed (as indicated in Table 5 and Table 6).

The U-value was determined using an Online Calculator [10]; and the following formula was used to calculate the z-value.

 N_1 = sample size in population 1

 $N_2 = sample size in population 2$

$$z = \frac{U - (N_1 N_{2/2})}{\sqrt{\frac{N_1 N_2 (N_1 + N_2 + 1)}{12}}} [11]$$

The following null hypotheses were rejected under the Personal Congruence factors (Rejection of the null hypothesis implies that a significant difference existed between the two groups on the respective factors):-

- For the Tenure-based tests, the null hypothesis was rejected for the CSR/ Civic activities factor (non-directional test);
- For the Gender-based tests, the null hypothesis was rejected for the *Flexible working arrangements* and *Organisational Justice* factors (non-directional test)

No null hypotheses were rejected for the Personal Incongruence factors.

It is evident from Table 3 that a blend of different retention factors is required by young engineers; and that these factors do not significantly differ from those of longer-tenured engineers.

Personal Congruence	Overall	< 5	> 5	Personal incongruence	Overall	< 5	> 5
Job satisfaction	92%	95%	86%	Job satisfaction	60%	56%	67%
Energising environment	92%	94%	89%	Advancement opportunities	59%	61%	56%
Career path	90%	90%	89%	Extrinsic rewards	58%	55%	64%
Learning organisation	90%	95%	81%	Organizational justice	55%	52%	61%
Nature of work	89%	89%	89%	Non-work influences	54%	53%	56%
Advancement opportunities	87%	89%	83%	Career path	51%	45%	61%
Flexibility	87%	85%	89%	Growth	50%	48%	53%
Intrinsic reward systems	86%	85%	86%	Nature of work	50%	47%	56%
Growth	85%	89%	78%	Leadership/ Management	48%	47%	50%
Extrinsic rewards	84%	79%	92%	Intrinsic reward systems	47%	44%	53%
Workplace relationships	83%	85%	78%	Location	47%	50%	42%
Flexible work arrangements	83%	84%	81%	Flexibility	41%	35%	50%
Organizational justice	81%	82%	78%	Values	39%	42%	33%
Technology	78%	81%	72%	Learning organisation	39%	39%	39%
Meaning	77%	74%	81%	Meaning	39%	37%	42%
Leadership/ Management	77%	82%	67%	Workplace relationships	36%	34%	39%
Values	74%	74%	75%	Energising environment	31%	31%	31%
Location	69%	69%	69%	Flexible work arrangements	31%	27%	36%
Organizational prestige	58%	63%	50%	Organizational commitment	31%	32%	28%
Organizational commitment	56%	56%	56%	Lack of alternatives	31%	32%	28%
Non-work influences	50%	50%	50%	Technology	29%	29%	28%
CSR/ Civic activities	47%	53%	36%	Organizational prestige	26%	26%	25%
God in the organisation	45%	44%	47%	Constituent attachments	23%	24%	22%
Investments	44%	47%	39%	God in the organisation	22%	23%	22%
Constituent attachments	39%	40%	36%	Investments	18%	23%	11%
Lack of alternatives	37%	37%	36%	CSR/ Civic activities	14%	13%	17%

Table 3: Comparison of Tenures against Overall Results (Greater Influence)



Table 4: Comparison between Gender Results

	Greater i	nfluence		Greater influence		
Personal Congruence	Female	Male	Personal Incongruence	Female	Male	
Job satisfaction	100%	88%	Advancement opportunities	62%	58%	
Energising environment	97%	89%	Job satisfaction	62%	59%	
Organizational justice	97%	72%	Extrinsic rewards	56%	59%	
Intrinsic reward systems	97%	80%	Organizational justice	56%	55%	
Flexible work arrangements	94%	77%	Non-work influences	53%	55%	
Nature of work	94%	86%	Growth	50%	50%	
Advancement opportunities	91%	84%	Nature of work	50%	50%	
Learning organisation	91%	89%	Leadership/ Management	47%	48%	
Growth	91%	81%	Career path	47%	53%	
Career path	88%	91%	Location	47%	47%	
Extrinsic rewards	88%	81%	Intrinsic reward systems	44%	48%	
Flexibility	88%	86%	Flexibility	41%	41%	
Workplace relationships	85%	81%	Meaning	38%	39%	
Leadership/ Management	85%	72%	Organizational commitment	35%	28%	
Meaning	82%	73%	Flexible work arrangements	35%	28%	
Values	79%	72%	Constituent attachments	32%	19%	
Location	79%	64%	Values	32%	42%	
Technology	76%	78%	Workplace relationships	32%	38%	
Organizational prestige	68%	53%	Learning organisation	29%	44%	
Non-work influences	59%	45%	Energising environment	26%	33%	
CSR/ Civic activities	56%	42%	God in the organisation	26%	20%	
Investments	50%	41%	Technology	24%	31%	
Organizational commitment	50%	59%	Organizational prestige	24%	27%	
God in the organisation	47%	44%	Lack of alternatives	Lack of alternatives 24%		
Lack of alternatives	35%	38%	CSR/ Civic activities	12%	16%	
Constituent attachments	29%	44%	Investments	6%	25%	



Table 5: Inferential Statistics Results: Personal Congruence

	Gender			Tenure (T)					
	N1 =34	N2 = 64	z Den		N1 = 62	N2 = 36	z Den		
	Female	Male	133.9851		T < 5	T >5	135.6982		
	Gender, Non-directional			Tenure			Non-directional (+-1.96)	Uni-directional (+-1.65)	
Personal Congruence	U	z Num	Ζ*	Null Hypothesis	U	z Num	Z *	Null Hypothesis	Null Hypothesis
Non-work influences	893	-195	-1.4554	Retain	1109.5	-6.5	-0.0479	Retain	Retain
Constituent attachments	975	-113	-0.8434	Retain	1055.5	-60.5	-0.4458	Retain	Retain
Energising environment	834	-254	-1.8957	Retain	935	-181	-1.3338	Retain	Retain
Flexible work arrangements	804	-284	-2.1196	Reject	995.5	-120.5	-0.8880	Retain	Retain
Advancement opportunities	933.5	-154.5	-1.1531	Retain	1032.5	-83.5	-0.6153	Retain	Retain
Extrinsic rewards	1016	-72	-0.5374	Retain	935.5	-180.5	-1.3302	Retain	Retain
Organizational justice	758.5	-329.5	-2.4592	Reject	993.5	-122.5	-0.9027	Retain	Retain
Values	977.5	-110.5	-0.8247	Retain	1034.5	-81.5	-0.6006	Retain	Retain
CSR/ Civic activities	927.5	-160.5	-1.1979	Retain	833.5	-282.5	-2.0818	Reject	Retain
Investments	952.5	-135.5	-1.0113	Retain	1011.5	-104.5	-0.7701	Retain	Retain
Leadership/ Management	873	-215	-1.6047	Retain	1030	-86	-0.6338	Retain	Retain
Career path	909	-179	-1.3360	Retain	971	-145	-1.0685	Retain	Retain
Technology	1073	-15	-0.1120	Retain	1012	-104	-0.7664	Retain	Retain
Job satisfaction	831	-257	-1.9181	Retain	963.5	-152.5	-1.1238	Retain	Retain
Learning organisation	911.5	-176.5	-1.3173	Retain	932.5	-183.5	-1.3523	Retain	Retain
Organizational prestige	934	-154	-1.1494	Retain	964.5	-151.5	-1.1164	Retain	Retain
God in the organisation	913	-175	-1.3061	Retain	1090.5	-25.5	-0.1879	Retain	Retain
Growth	1009.5	-78.5	-0.5859	Retain	880.5	-235.5	-1.7355	Retain	Retain
Meaning	863.5	-224.5	-1.6756	Retain	1004	-112	-0.8254	Retain	Retain
Intrinsic reward systems	842	-246	-1.8360	Retain	1000	-116	-0.8548	Retain	Retain
Organizational commitment	1012.5	-75.5	-0.5635	Retain	1085	-31	-0.2284	Retain	Retain
Workplace relationships	1057	-31	-0.2314	Retain	1051.5	-64.5	-0.4753	Retain	Retain
Lack of alternatives	1058	-30	-0.2239	Retain	1056.5	-59.5	-0.4385	Retain	Retain
Location	833	-255	-1.9032	Retain	1075	-41	-0.3021	Retain	Retain
Flexibility	925	-163	-1.2166	Retain	916	-200	-1.4739	Retain	Retain
Nature of work	900.5	-187.5	-1.3994	Retain	994.5	-121.5	-0.8954	Retain	Retain


Table 6: Inferential Statistics Results: Personal Incongruence

		Gender, No	on-directio	onal		Tenure		Non-directional (+-1.96)	Uni-directional (+-1.65)
Personal incongruence	U	Num	Z *	Null Hypothesis	U	Num	Z *	Null Hypothesis	Null Hypothesis
Non-work influences	1038	-50	-0.3732	Retain	1002.5	-113.5	-0.8364	Retain	Retain
Constituent attachments	1069	-19	-0.1418	Retain	1087.5	-28.5	-0.2100	Retain	Retain
Energising environment	964.5	-123.5	-0.9217	Retain	1054.5	-61.5	-0.4532	Retain	Retain
Flexible work arrangements	1055	-33	-0.2463	Retain	999.5	-116.5	-0.8585	Retain	Retain
Advancement opportunities	959.5	-128.5	-0.9591	Retain	1017	-99	-0.7296	Retain	Retain
Extrinsic rewards	1026	-62	-0.4627	Retain	1023	-93	-0.6853	Retain	Retain
Organizational justice	1014	-74	-0.5523	Retain	941	-175	-1.2896	Retain	Retain
Values	894.5	-193.5	-1.4442	Retain	1087	-29	-0.2137	Retain	Retain
CSR/ Civic activities	1064	-24	-0.1791	Retain	1055	-61	-0.4495	Retain	Retain
Investments	916	-172	-1.2837	Retain	958	-158	-1.1643	Retain	Retain
Leadership/ Management	1073	-15	-0.1120	Retain	1005	-111	-0.8180	Retain	Retain
Career path	947	-141	-1.0524	Retain	976	-140	-1.0317	Retain	Retain
Technology	1051	-37	-0.2762	Retain	1095	-21	-0.1548	Retain	Retain
Job satisfaction	1047	-41	-0.3060	Retain	907.5	-208.5	-1.5365	Retain	Retain
Learning organisation	980	-108	-0.8061	Retain	1104	-12	-0.0884	Retain	Retain
Organizational prestige	991.5	-96.5	-0.7202	Retain	1104	-12	-0.0884	Retain	Retain
God in the organisation	895	-193	-1.4405	Retain	1025	-91	-0.6706	Retain	Retain
Growth	1078.5	-9.5	-0.0709	Retain	1077.5	-38.5	-0.2837	Retain	Retain
Meaning	1078	-10	-0.0746	Retain	1075	-41	-0.3021	Retain	Retain
Intrinsic reward systems	1052	-36	-0.2687	Retain	1008	-108	-0.7959	Retain	Retain
Organizational commitment	971	-117	-0.8732	Retain	1102	-14	-0.1032	Retain	Retain
Workplace relationships	1040.5	-47.5	-0.3545	Retain	1082	-34	-0.2506	Retain	Retain
Lack of alternatives	1074	-14	-0.1045	Retain	1077	-39	-0.2874	Retain	Retain
Location	1028	-60	-0.4478	Retain	1062	-54	-0.3979	Retain	Retain
Flexibility	1013	-75	-0.5598	Retain	862	-254	-1.8718	Retain	Retain
Nature of work	1054	-34	-0.2538	Retain	1005.5	-110.5	-0.8143	Retain	Retain



Table 7 shows the relation of each of the sources to the conceptual framework. The Survey results which had a greater influence of more than 50% were highlighted; and the influence on PC or PI was indicated in the Survey column. The factors from the Separation Data were marked with a PI in the *Sep Data* column as these factors contributed to employees leaving the organisation.

The relation of the sources (other than the Survey and Separation Data results) to each of the conceptual framework factors was tallied to indicate which factors were prominent in the perused information from the organisation (which related to the employee organisation's retention effort)

The counts of the factors which had both PC and PI scores above 50% are highlighted in the Tally column.

	Survey	Archival	Doc		Interv	iews		Tally
Conceptual Framework	Results (>50%)	Sep Data	TM, GDP	Bursary	тм	GDP	W&B	Count
Job satisfaction	PC, PI	PI						0
Energising environment	PC			*				1
Career path	PC, PI	PI	*		*	*		3
Learning organisation	PC	PI	*		*	*		3
Nature of work	PC		*					1
Advancement opportunities	PC, PI		*					1
Flexibility	PC					*	*	2
Intrinsic reward systems	PC	PI					*	1
Growth	PC, PI	PI	*				*	2
Extrinsic rewards	PC, PI	PI	*	*		*	*	4
Workplace relationships	PC					*		1
Flexible work arrangements	PC						*	1
Organizational justice	PC, PI							0
Technology	PC						*	1
Meaning	PC						*	1
Leadership/ Management	PC	PI					*	1
Values	PC					*		1
Location	PC	PI						0
Organizational prestige	PC		*		*			2
Organizational commitment	PC	PI					*	1
Non-work influences	PC, PI	PI				*	*	2
CSR/ Civic activities						*		1
God in the organisation								0
Investments						*		1
Constituent attachments								0
Lack of alternatives								0

Table 7: Relation of Study Findings to the Conceptual Framework



5. DISCUSSION

The study considered both the factors which would result in the bursar graduate staying with the organisation (PC factors), and those which would contribute to departure from the organisation (PI factors). These factors for bursar graduate engineers within 5 years working experience were found to be:-

PC factors

- Job satisfaction
- Learning organisation,
- Energising environment
- Career path
- Advancement opportunities

PI factors

- Advancement opportunities
- Job satisfaction
- Extrinsic rewards
- Non-work influence
- Organisational justice

Table 8: Top 5 ranking PC and PI factors

	Overall		Tenure < 5 years		Tenure > 5 year	rs
al nce	Job satisfaction	92%	Job satisfaction	95%	Extrinsic rewards	92%
oni	Energising environment	92%	Learning organisation	95%	Flexibility	89%
ers	Career path	90%	Energising environment	94%	Energising environmen	89%
C_0	Learning organisation	90%	Career path	90%	Career path	89%
	Nature of work	89%	Advancement opportunities	89%	Nature of work	89%
е	Overall		Tenure < 5 years		Tenure > 5 year	rs
al enc	Job satisfaction	60%	Advancement opportunities	61%	Job satisfaction	67%
inu on:	Advancement opportunitie	59%	Job satisfaction	56%	Extrinsic rewards	64%
ers ong	Extrinsic rewards	58%	Extrinsic rewards	55%	Organizational justice	61%
F	Organizational justice	55%	Non-work influences	53%	Career path	61%
	Non-work influences	54%	Organizational justice	52%	Non-work influences	56%

The below points will be discussed from the study's results:

- What do bursar graduates want from the organisation
- The organisation's approach to employee retention strategies
- Spirituality at work

What do Gen Y bursar graduates want from the organisation

It is evident from

This blend of different retention factors suggests that employees value both career pursuits (e.g. Career path, Advancement opportunities, Learning Organisation); as well as those which support personal satisfaction at work, as well as personal pursuits (e.g. Job satisfaction, Energising environment, Non-work influences).

Of the factors suggested by literature, those with the most influence appeared to be more highly reflective of knowledge workers retention factors, with factors such as *Career path* and *Advancement opportunities* appearing as great influencers for both PC and PI

that a blend of different retention factors is required by young engineers; and that these factors do not significantly differ from those of longer-tenured engineers



This blend of different retention factors suggests that employees value both career pursuits (e.g. *Career path, Advancement opportunities, Learning Organisation*); as well as those which support personal satisfaction at work, as well as personal pursuits (e.g. *Job satisfaction, Energising environment, Non-work influences*).

Of the factors suggested by literature, those with the most influence appeared to be more highly reflective of knowledge workers retention factors, with factors such as *Career path* and *Advancement opportunities* appearing as great influencers for both PC and PI

The respondents identified with some of the factors from the Gen Y literature as well; such as *Growth*, *Job* satisfaction, and *Meaning*. These factors also matched the motivational variables of "training and development", and "challenging and interesting work" [12, p. 413] (factors which were found to significantly influence employees in both public and private sector organisations [12]).

The factors which ranked high for PC were not the same as those which ranked high for PI (

). This illustrates that there is a difference in the reasons that would encourage an employee to stay and those which would cause an employee to leave the organisation. Again, the difference in the percentages of the top ranking PC and PI factors suggests that the factors used in the conceptual framework influenced the respondents' PC (what would make the employee stay with the organisation), more than they did the PI (what would make the employee leave the organisation) with the factors.

Job satisfaction was an important factor to take into consideration for both groups (overall PC greater influence of 92%, and overall PI greater influence of 60%; Table 3,

). This implies that more efforts are required to ensure a good organisation-individual fit. The use of psychometric tests could assist in this approach; however the realities of a shrinking labour market and urgent organisational needs could perhaps taint this as a more idealistic approach than a practical one.

If Employee Retention is defined as: "a process in which the employees are encouraged to remain with the organisation for the maximum period of time" [13, p. 80], the "maximum period of time" is a unit which needs to be reviewed. With the low influence that *Lack of Alternatives* had on the respondents' PC or PI, and that *"the current trend is that young employees typically transfer jobs after about 5 years..."* (PC General comments, Tenure < 5 years), the feasibility of an idea such as employee retention for bursar graduates is one that could possibly require greater consideration. Other elements would need to be considered, such as: How long is the "maximum period of time? What should it be? How is the answer to that determined? What makes that answer feasible from both an organisational and the individual's perspective (possibly the rate of return on investment)? These are questions that could be explored in future work.

The organisation's approach to employee retention strategies

Literature suggests that organisational approaches to employee retention are to have less of a "blanket" approach [7, p. 5]; such that the approaches to be adopted are more responsive to the needs of its workforce. The company has made an attempt at this by segmenting its talent pool, and by developing different processes to deal with each segment. However the type of segmentation could also be along demographics such as age and gender, as the results showed some differences in the factor ranking of the groups (Table 3,

Table 4).

The company has developed targeted development programmes for high-potential and high-performing employees (which can assist the organisation with dealing with the two-fold retention strategy challenges of not only how to minimise the possibility of losing good employees, but also how to identify the employees an organisation needs and wants to keep [14]). The development of such targeted programmes for other segmentation groups can also assist the organisation with the development of retention strategies which are relevant to the specific employee groups [7].

Targeted initiatives will have implications on the "Attract, Source and Develop" strategies, as these may need to be specific to the different groupings as well (e.g. Plant Support team members that feel under-developed after a certain period of time, as indicated by the PC comments under the *Advancement opportunities* and *General sections*).



A difference was found between the Talent Landscape elements and what is suggested by the EB/ EVP literature (e.g. there is less of a focus on the *collegial work environment* and *respect elements in the talent landscape*). There was also less of a focus on the 'softer', or more people-centric, elements in the building blocks of the Strategic Talent Planning Framework .There is thus potential to develop in these ' softer', or more personal aspects of EB and EVP (e.g. work relationships, how employees are treated, etc.). This could potentially also relate to *Organisational justice* (i.e. having policies and guiding principles to address these 'softer' elements); and thus contribute towards addressing the PI influence of this factor.

The survey results suggest that the company should focus on a balance of career-specific and personal development. This is in order to increase the retention of its younger employees, who could potentially contribute to the growth of the organisation through the accumulation of institutional memory. This requires that the management of talent take on a more future-looking approach to the design of people processes in order to be able to predict trends that may occur (which would be due to the changes which may take place in the current young generation, as well as to anticipate the needs of future young graduates).

There did not appear to be an indication of how graduates are supported post-completion of the GDP. Because of the reduced guidance and focused assessment of the graduate employee post the programme, this could be risky for the continued personal and professional development and retention of the individual. It is thus important that the "bridge" between the early career and professional categories of the EB/ EVP Attraction and Retention portfolio support the graduate's transition from the GDP.

There is a need to adequately capture the reasons for departure. This can be done through ensuring that exit interviews are conducted with each exit from the company (although the integrity of such data may not always be reflective of the actual reasons. The outcomes from those interviews can be used as inputs to refining development programmes and retention strategies. The reasons can be addressed through the design of appropriate interventions which can prevent these reasons from recurring in the future.

Employers need to consider how the success of their retention strategies will be measured. The Employer Brand aspect of the company's Attraction and Retention portfolio also cites "Employer of Choice" as an outcome. Literature also suggests that a successful retention strategy can be indicated by the degree to which an organisation is an employer of choice [15]. This could mean that the measurement of an effective retention strategy could be related to the degree or success of talent attraction efforts. Other potential measurements could be the amount of "intellectual capital" that a company has, and the return on the educational investment put into the bursary scheme.

Spirituality in the workplace

One of the objectives of this study was to determine how the above findings relate to the subject of spirituality at work. This was done through the inclusion of such spirituality at work related factors into the survey. The results showed that these factors do indeed play a role in employee retention, with some of these factors scoring overall above 70% for the PC factors: *Energising environment* (92%), *Growth* (85%), *Meaning* (77%), and *Values* (74%) - all which support Job satisfaction at work. This was reflected in expectations of *"servant leadership"* (*Leadership/ Management* PC comment) and in sentiments that *"finding one's place creates a field where performance is easier and fun"* (*Meaning* PC comment, Tenure < 5 years). *Respect* (one of the principles used to define spirituality [16] was also cited as a top driver for both attraction and commitment from the CEB study [17]. Organisational culture has a role to play in how employees experience spirituality at the workplace (*"Fear and blame is the culture currently being driven in my company. This stifles creativity, excitement, fun and inspiration" - Energising environment* PC comment, Tenure > 5 years). Creativity was also one of the seven principles which the Institute for Management Excellence used to define spirituality [16] as cited by [18].

Although *Extrinsic rewards* featured in the top 5 PI factors across the different categories, the Survey comments suggest that the sense of making a difference, of *Meaning*, is also required (e.g. "Money is not the main driver anymore for most engineers. Career development and a sense of making a difference is more important" (Extrinsic rewards PC comment, Tenure > 5 year)).

Despite the above, it was found that ideas of 'God' at the workplace had a low influence in PC and PI (as indicated by the low ranking of "God" in the organisation factor in both instances). Some respondents primarily identified this factor with Religion ("Please no gods" ('God in the organisation PC Comment, Tenure > 5 years); and "Religion has absolutely no place in a workplace, unless it is a religious organisation. I find the inclusion of religion in workplaces at best annoying, and at worst obnoxious and insulting" ('God in the organisation PC Comment, Tenure < 5 years). This is despite the definition that the factor was about "An environment which is "soulful", which contains "spiritual" values, has an element of sacredness").

The CSR/ Civic activities factor also displayed low ranking overall and within the different groups.



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE

The high-ranking of the Job satisfaction (both PC and PI) and Energising environment (PC) factors, and the theme of the importance of a work-life balance suggest that the idea of spirituality at work is a considerable aspect to explore within the area of employee retention strategies. (Albeit more of an 'elusive' and perhaps abstract idea to work with - unlike e.g. Career path/ Advancement opportunities, Intrinsic rewards, Flexible work arrangements). The application of such a factor definitely requires a more aware and astute understanding of workforce needs and expectations.

6. CONCLUSION

The design of employee retention strategies as a whole is a complex process; when taking into account the intricacies of human nature and the different life paths that people follow (be it out of personal choice or personal circumstance).

The results did not show any significant difference between the differently tenured groups. This could primarily be because of both populations being Knowledge Workers and generally belonging to the same generation (Gen Y). What was evident though is that individuals at different stages in their personal and professional lives would have different needs; and that a balance between these segments is of great importance.

This has implications for how employees are supported throughout their working career. Although a young graduate may have some support during the GDP phase, this can affect the continued growth and development of the individual after s/he exits from the GDP. This necessitates that the TM and HR strategy for the greater workforce be adequately structured to ensure the continued retention of the internal talent pool, and that leadership/ management is sufficiently equipped to enable the employee retention effort.

This need is compounded by a highly dynamic and competitive labour market landscape which is vying for the same talent. This requires that organisations reflect (and invest) more deeply into how their objectives and operations can be realised through human capital resources. There is a need for current employee retention strategies to be more employee-centric; i.e. that they need to respond to the needs of the individuals of the workforce. This can be achieved through the development of "highly targeted initiatives" which can be approached through more demographic segmentation of the workforce (e.g. aspects like age and gender). This will allow for the inclusion of the 'softer' elements of employee retention. In that way a more holistic framework to employee retention strategies at the organisation can be developed.

A difference was also found between the factors which came out of the interviews, but which were not indicated in the perused documentation. It is important that the employee retention documentation (as well as GDP documents) comprehensively reflect these (possibly) implicit elements; as well as the initiatives that the organisation has been place to support the retention of its employees.

The measurement of the effectiveness and success of an employee retention strategy (one that is more holistic and more reflective of employee-needs) needs to be considered. This entails developing ways to measure the outcomes of the EVP and EB strategies; and includes the investigation the measurement of investment on intellectual capital and efforts to attract the talent in the first instance.

Future work would include the measurement of the success of employed retention strategies; as well as alternative methods of retaining institutional memory. More research can be conducted on contrasting the needs of the Gen Y with those from earlier Generations, so as to more adequately account for these differences in the employee retention effort. The feasibility of the expected retention period (i.e. how long should employees be retained for; considering the changing nature of work, and the nature of knowledge workers) should be explored as well.

This research would also need to be expanded to other companies (multiple-case, embedded i.e. Type 4 Case study design).

The implications of the study's overall findings to the organisation's current employee retention strategies were discussed; and the consequent recommendations are summarised below.

Recommendations from the Research

In summary, it is recommended that:

- The refinement of employee retention strategies needs to take into consideration the mix of factors (between career and personal pursuits) which are required by employees.
- Organisation should be responsive and allow for some flexibility to be able to accommodate employees' personal situations.



- Extrinsic rewards do play a factor in employee retention; and the organisation should continue to maintain the benefits which are offered to employees.
- (Linked with the *Extrinsic rewards* factor) The organisation should ensure that "perceptions about the fairness of reward allocations, policies and procedures, and interpersonal treatment [7, p. 6] (i.e. Organisational justice) are managed and adequately addressed.
- Job satisfaction was a key factor for both PC and PI. This requires that the job-fit and organisationalfit of the individual be determined; to ensure that an individual is well-placed in a position that can add value to both the employee and the organisation.
- The employee retention strategies which are reliant, or based on the extended tenure relationship and loyalty that an individual may have with and towards an organisation should be reviewed.
- The organisation can consider segmentation of the talent pool along other factors such as demographics (e.g. age, gender, culture, etc.) or professional disciplines (staff in a particular working environment e.g. Plant Support engineers) to identify retention factors which may be specific to the particular segment. This will enable the development of targeted employee retention programmes which are relevant to the particular employee groups' needs.
- The preceding recommendation will also enable for the development of more individual-centric and holistic approaches to employee retention.
- The Employee Retention strategies should clearly be differentiated from the Employee Attraction strategies. Although the two may be concomitant in Workforce Planning; it is important to understand which of the strategies fulfil which purpose.
- Leadership/ Management should be trained and equipped to enable a working environment which supports employee retention.
- The organisation should also offer mentorship programmes which support the graduate postcompletion of the GDP. These mentorship programmes could also be of benefit to new and existing employees generally (so as to support the continuous development of the individual).
- The employee retention documentation needs to be fully reflective of all the initiatives and developments (social, recreational, etc.) that the organisation offers its employees. The development of comprehensive documentation on such retention strategies can help address any gaps, as well as ensure effective communication of such initiatives to the workforce.
- The information emanating from the exit interviews needs to be captured in detail, and utilised in the further development of employee retention strategies to address these gaps.
- The idea of spirituality at work is a considerable aspect to explore within the area of employee retention strategies. Albeit an almost abstract idea to work with, literature suggests some guiding principles, as well as methods with which this can be measured.
- The actual (quantifiable) measurement of an effective retention strategy needs to be considered.
- The developers of employee retention strategies should maintain a future-facing approach in the design of people processes. This is so that trends and possible changes in the talent landscape can be predicted and prepared for.
- It may be worthwhile to consider alternative methods of retaining institutional memory; as the knowledge worker attributes of Gen Y and the engineering profession could result in early loss of employees who move elsewhere for employment.
- Organisations should determine what a 'realistic' expectation of the period of employee tenure is; and have a method of measuring the feasibility of such a period.

REFERENCES

7. BIBLIOGRAPHY

- [1] J. M. Schabracq en L. C. Cooper, "The changing nature of work and stress," Journal of Managerial Psychology Vol 15 No.3, pp. 227-241, 2000.
- [2] R. J. Burke en E. Ng, "The changing nature of work and organisations: Implications for human resource management," *Human Resource Management Review 16*, pp. 86-94, 2006.
- [3] G. N. P. Konz en F. X. Ryan, "Maintaining an organisational spirituality: no easy task," Journal of Organisational Change Management, Vol 12 No. 3, pp. 200-210, 1999.
- [4] G. P. Nayar, "Overview," SCMS Journal of Indian Management Vol IX, No. III, p. 2, 2012.
- [5] B. Panahi en A. Abedinpoor, "Identification of Spiritual Organisations: Theories and Models," 2010. [Aanlyn]. Available: http://www.wbiconpro.com/721-Panahi.pdf. [Onttrek March 2014].
- [6] J. Bersin, "Employee Retention Now a Big Issue: Why the Tide has Turned," 16 August 2013. [Aanlyn]. Available: https://www.linkedin.com/today/post/article/20130816200159-131079-employee-retentionnow-a-big-issue-why-the-tide-has-turned. [Onttrek 20 February 2014].
- [7] J. Hausnecht, J. M. Rodda en M. J. Howard, "Targeted Employee Retention: Performance-based and Job-



related Differences in Reported Reasons for Staying," Cornell University ILR School, Ithaca, 2008.

- [8] R. K. Yin, Case Study Research: Design and Methods, Thousand Oaks: SAGE Publications, 2003.
- [9] P. Bump, "Here is When Each Generation Begins and Ends, According to Facts," 25 March 2014. [Aanlyn]. Available: http://www.theatlantic.com/national/archive/2014/03/here-is-when-each-generation-beginsand-ends-according-to-facts/359589/. [Onttrek December 2014].
- [10] Social Science Statistics, "Mann Whitney U-value Calculator," 2014. [Aanlyn]. Available: http://www.socscistatistics.com/tests/mannwhitney/. [Onttrek 17 February 2015].
- [11] P. Billet, "The MannWhitney U-test -- Analysis of 2-Between-Group Data with a Quantitative Response Variable," 2003. [Aanlyn]. Available: http://psych.unl.edu/psycrs/handcomp/hcmann.PDF. [Onttrek February 2015].
- [12] M. O. Samuel en C. Chipunza, "Employee retention and turnover: Using motivational variables as a panacea," African Journal of Business Management Vol 3, pp. 410-415, 2009.
- [13] L. James en L. Mathew, "Employee Retention Strategies: IT Industry," SCMS Journal of Indian Management Vol IX No. III, pp. 79-87, 2012.
- [14] B. J. Kreisman, "Insights into Employee Motivation, Commitment and Retention," Business Training Experts, Denver, 2002.
- [15] J. Gering en J. Conner, "A Strategic Approach to Employee Retention," *Healthcare Financial Management*, pp. 40-44, 2002.
- [16] Institute for Management Excellence, "Seven Principles of Spirituality in the Workplace," 26 May 2011. [Aanlyn]. Available: http://www.itstime.com/rainbow.htm. [Onttrek 16 September 2014].
- [17] Corporate Leadership Council, "Attracting and Retaining Critical Talent," Corporate Executive Board, 2006.
- [18] C. Litzsey, "Spirituality in the Workplace and the Implications for Employees and Organisations," Southern Illinois University Carbondale, Carbondale, 2003.



DESIGNING A MOBILE LABORATORY SOLUTION FOR RAPID TUBERCULOSIS TESTING BY MEANS OF THE CEPHEID® GENEXPERT® DEVICE

M Hattingh¹ and L Bam²

^{1 & 2} Health Systems Engineering and Innovation Hub, Department of Industrial Engineering, Stellenbosch University, Private Bag X1, Matieland, 7602, South

1 16493125@sun.ac.za

²louzanne@sun.ac.za

ABSTRACT

Tuberculosis (TB) poses a major health risk in South Africa. The problem is particularly pronounced amongst employees in the mining industry, where South Africa has the highest TB incidence rate in the world. Diagnostic laboratory testing is a crucial component of managing TB and multi-drug resistant (MDR) TB. Mobile laboratories that are housed in vehicles are one option for reaching communities to expand diagnostic coverage. The World Health Organisation endorsed, GeneXpert[®] platform can be used for the diagnosis of TB and MDR-TB and is sufficiently robust to be used as a point-of-care device. The study reviews the GeneXpert[®]-based TB diagnosis process, to accurately determine the operations that are to be supported by the mobile laboratory. Consequently, this technology has enabled the development of a mobile laboratory for TB testing. This article describes the development of a concept design for such a mobile laboratory solution. The concept design considers various functional, safety and storage aspects as well as ergonomic principles to ensure an effective testing procedure along with the health and safety of the laboratory technician. A costing estimate for the design is also provided. Owing to the high incidence of TB amongst employees in the mining industry, the design has been developed with deployment in peri-mining communities in mind.

Keywords: Design, diagnostics, ergonomics, GeneXpert[®], mobile laboratory, tuberculosis



1. INTRODUCTION

Tuberculosis (TB) along with the Human Immunodeficiency Virus (HIV) make up the two largest health challenges to South Africa. The World Health Organisation (WHO) rates South Africa as having the second largest estimated TB incidence rate (834 per 100,000 population) and the sixth largest incidence by absolute number (an estimated 450,000) in the world [1]. Furthermore, South Africa is categorised as a high MDR-TB burden country [1]. South Africa's mining industry has the highest infection rate worldwide amongst mine workers, with an incidence rate of 948 per 100,000 miners, compared to global- and Sub-Saharan incidence rates of 128 and 350 per 100,000 respectively [2]. Furthermore, it is estimated that 89% of the approximately 500,000 mineworkers in Southern Africa have latent TB infection [3].

In an attempt to combat tuberculosis amongst miners and their families, fifteen Southern African countries, including South Africa, Lesotho and Botswana, signed the 'declaration on tuberculosis in the mining sector' at a meeting of the Southern African Development Community (SADC) in August 2012. The declaration is to guarantee commitment by the member states to improve the lives of miners affected by tuberculosis [4]. Briefly summarised, the various governments pledge their commitment to address tuberculosis in the mining industry in accordance with SADC [5] protocols to eventually eliminate tuberculosis in the region. In addition the declaration also identified priority areas that require immediate attention.

This article describes the development of a concept design for a cost-effective mobile laboratory for testing sputum samples for the presence of tuberculosis. Central to this design, is the GeneXpert® MTB/RIF test endorsed by the WHO as a rapid and effective tuberculosis testing method. The GeneXpert® is a point-of-care device that can diagnose tuberculosis and drug resistant tuberculosis with an especially high sensitivity and accuracy, even in patients that are HIV-positive [6]. The compact and relatively robust (in comparison to traditional diagnostic laboratory instrumentation) nature of the GeneXpert® platform opens the opportunity for the development of more manoeuvrable and cost effective manners of reaching peri-mining communities, such as through the use of mobile laboratories. This in turn enables more effective monitoring and treatment of tuberculosis as well as the gathering of more comprehensive population health monitoring data.

The design, although proposed for testing in peri-mining communities, is relevant and replicable for diagnosis in other settings in the country or abroad. Furthermore, the majority of elements in the design consideration would be directly applicable to the development of mobile laboratories for other compact testing devices, aimed either at TB or other diseases. Finally, the approach that is followed in generating the design could be useful in designing other mobile laboratory solutions.

The article describes the methodology that was followed to develop the concept design. Background is given on why TB is particularly pronounced amongst mining communities in South Africa as well as on the processes of diagnosing TB in South Africa with the GeneXpert[®] device. The design considerations and recommendations are presented whereafter the concept design is described and illustrated. A costing estimate for the concept design is also presented.

2. METHODOLOGY

In order to comprehend the problem and acquire the necessary knowledge to develop a solution, two National Health Laboratory Service (NHLS) laboratories were visited (one in Cape Town and one in Port Elizabeth). Activity-flow diagrams were drawn up for the national algorithm (national prescribed TB-testing process); the testing process with the GeneXpert[®] as well as the proposed testing method in the mobile laboratory. From the knowledge gained during field visits and the literature review, a set of operational and engineering specifications were compiled which were used for guiding the concept design process. Three concept designs were developed and evaluated, and, after careful consideration, the final concept was selected and refined using Google SketchUp software. The final design is illustrated and discussed, after which the cost estimation for the mobile laboratory is presented.

3. BACKGROUND

This section briefly describes tuberculosis and the Southern African mining industry; similar mobile laboratories; the GeneXpert[®] device; and testing TB methodologies in South Africa.

3.1. Tuberculosis in Southern Africa's Mining Industry

The rate of TB-infections has increased extraordinarily from the beginning of the 21st century. A study performed in 2011 estimated that the mining industry is responsible for an additional 760 000 TB cases every year in African



mining communities [4]. These are, amongst other factors, the consequence of exposure to silica dust; crowed living conditions and, most prominently, the high HIV prevalence [4, 7].

According to the '*Stop TB*' organisation, the average infected miner spreads TB to an estimated fifteen people in their community [4]. This is a major problem owing to a ripple effect that is carried from the peripheral mining communities through to other countries such as Mozambique, Swaziland and Lesotho to which many migrant workers return, in some cases after having been exposed to silicosis, HIV and TB [8]. It is estimated that two million ex-miners live in Lesotho, South Africa, Mozambique and Swaziland [2]. The risk of contracting TB in a high disease burden area, is up to ten times higher for miners than for the general population; and miners that have contracted TB are 3.6 times more likely to die from it [3].

The mining industry in South Africa plays a significant role in the economy, contributing 18% (8.6% directly and 10% indirectly) to the Gross Domestic Product (GDP) in 2013 [9]. In spite of the recent instability in the mining sector caused by wide-spread labour unrest and fluctuations in global commodity prices, the sector experienced a growth of 3.5% between 2010 and 2015, still contributing 8% (directly) to the GDP in 2015 [10]. Approximately 500 000 mineworkers are employed in Southern Africa and an estimated 9.6 million workdays are lost due to tuberculosis infections [2]. TB-related reduced productivity in miners thus has an impact on the economy of Southern African countries, where mining can contribute up to 25% of the GDP [3].

South Africa has signed the 'declaration on tuberculosis in the mining sector'. The declaration states that the main challenges miners, previous miners and the peri-mining communities experience are the lack of access to basic social and health services, effective cross-border systems, and no treatment regimen, legal protection or compensation for respiratory problems developed due to working circumstances [11].

3.2. Similar Mobile Laboratories

There is limited literature available on similar mobile laboratory solutions and none could be found that use the GeneXpert[®] device or are solely for testing for TB. However, a few examples of mobile laboratories targeting other diseases are briefly discussed. Partec developed a CyLab[™] for HIV monitory, TB- or malaria testing. A four-wheel Mercedes Transporter was used and the laboratory is detachable from the vehicle [12]. Another design for testing and monitoring meningitis in West Africa, equipped a Renault Kangoo with a laboratory relying on solar power for the first rollout phase[13]. The second roll-out phase used a Toyota Hilux, with a detachable laboratory that does not rely on solar power. These are examples of small laboratories, similar to that being developed in this study. Larger mobile laboratories mounted on trucks are used frequently for various services.

3.3. The GeneXpert[®] Device

The Xpert[®] MTB/RIF test functions on a molecular level and incorporates modern deoxyribonucleic acid (DNA) technology to detect tuberculosis by testing a sputum sample [14]. The procedure is fully automated, easy and safe to use, does not require a conventional laboratory and is highly accurate in identifying both tuberculosis and Rifampicin-resistance in one test [15]. The test produces one of four outcomes, namely: (i) positive and Rifampicin susceptible; (ii) positive and Rifampicin resistant; (iii) negative; or (iv) an unsuccessful test. A particular advantage of the test in the South African context is the diagnostic accuracy in patients that are HIV positive [16].

Point-of-care (POC) testing occurs when a patient is tested near or on the same site where the results will be reported and the treatment is given. The four-module device is portable, robust and requires minimal training with regard to operating and maintaining the device - making it suitable for POC testing [17]. Limitations associated with implementing the GeneXpert[®] in point-of-care testing are the environmental conditions such as the temperature, and requirements with regard to uninterrupted power supply, regular maintenance and annual calibration.

The WHO compiled a document which outlines the requirements and actions that should be considered by countries before and during the roll-out and implementation phases. There are two checklists in this document, the first pertaining to the key prerequisites countries should meet before the roll-out phase and the second list containing key actions to be undertaken for implementing the test efficiently [18].

3.4. Testing Tuberculosis in South Africa

The implementation of the GeneXpert[®] assay does not eliminate the need for conventional TB culture, microscopy and drug susceptibility tests. The 'national algorithm' prescribes the steps that must be followed in NHLS laboratories for diagnosing TB. This section describes TB-testing processes which form part of the national algorithm.



3.4.1. National Algorithm

The algorithm starts with the initial GeneXpert[®] test. The results of the GeneXpert test along with the patient's HIV-status determine the next steps in the algorithm to be followed. The mobile laboratory, being equipped with a GeneXpert[®], will play a role in this initial diagnostic step. Two sputum samples are taken from the patient: one is used for the GeneXpert[®] test and the other is stored in the fridge. An unsuccessful test requires a retest. If there is still enough of the prepared sample left to retest, and the reagents have been added less than three hours prior, the sample can be retested, otherwise a new sample should be prepared and tested.

In the case of a positive GeneXpert[®] result and HIV-positive patient, the second sample is sent for TB culture and microscopy testing. Patients that are HIV-negative are initiated on to TB treatment following a positive GeneXpert[®] result.

3.4.2. GeneXpert[®] Procedure

The GeneXpert[®] procedure was observed at the NHLS laboratory in Port Elizabeth. Although some institutions indicate that the samples require incubation, it is not necessary and at this laboratory, the samples are shaken and put aside for a few minutes. The GeneXpert[®] test is genotypic, implying that the samples can be prepared at room temperature. In contrast, smear microscopy testing, being phenotypic, requires incubation of the sample at 37 degrees Celsius.

Samples are sorted upon arrival and receive a unique barcode. The barcode, along with the patient information is logged into the NHLS computer system. Sample reagents that come in plastic bottles containing about 8 millimetres of fluid, are added to the sputum sample in a 2:1 ratio. A bottle of sample reagent is used per sample, even though there might be an adequate amount of liquid left for a following sample. Next, using a pipette, 2ml of the solution is extracted and added to an empty GeneXpert[®] cartridge, which has a corresponding barcode. In the fixed site laboratory environment, these steps take place within a class-2 bio-safety cabinet.

The batch is then moved to the GeneXpert[®], where each cartridge is scanned into a GeneXpert[®] computer. The cartridge is inserted into the slot allocated by the computer system. After the results are printed, the laboratory technician checks for any unsuccessful tests. If a sample was unsuccessful, there is still enough prepared sample left to redo the test and the sample was prepared less in three hours ago, it may be added to a new cartridge and retested. The successful results, being either negative or positive, are added to the patient's file after which the national algorithm is followed to determine the next steps.

3.4.3. Mobile Laboratory Process

The mobile laboratory is designed to be used as diagnostic support for clinical partners that screen patients and take samples, typically in a tent. The mobile thus supports the GeneXpert[®] testing procedure similar to that followed at fixed-site laboratories and is displayed in Appendix A. Only one sample taken from a patient is sent for testing by the GeneXpert[®] and accordingly the laboratory design only allocated storage space, including refrigeration space, for one sample.

The only controversial aspect of the mobile laboratory design is the protection of laboratory personnel during the sample preparation procedure at fixed laboratory sites, GeneXpert[®] samples are typically prepared for testing in a biosafety cabinet. However, installing a biosafety cabinet in a mobile laboratory creates challenging requirements in terms of air vents and sensitivity to peripheral air flows. The standard operating procedure manual for the GeneXpert[®] MTB/RIF test indicates that, in instances where a bio-safety cabinet is not available, the sample should be heat inactivated in a hot air oven (20 minutes at 85°C or 30 minutes at 80°C). The inclusion of a hot air oven in the mobile laboratory proved infeasible, due to the large size of the required oven, the large required power supply and the increased processing time for samples. According to SMEs, adequate ventilation and the wearing of the necessary protective clothing, will be considered sufficient protection during sample preparation in mobile laboratories in South Africa. Therefore, the mobile laboratory design has been developed without a biosafety cabinet or a hot air oven. Options for providing additional protection to mobile laboratory personnel during sample preparation is discussed in more detail in Section 7.

The laboratory is designed to accommodate two four module GeneXpert[®] machines, allowing for roughly eight tests to be run each taking approximately two and a half hours. Taking setup and packing-up times into consideration, approximately sixteen samples can be tested each day. The mobile laboratory is designed to accommodate one laboratory technician.

4. DESIGN CONSIDERATIONS AND REQUIREMENT SPECIFICATIONS

This section describes the ergonomic and operational requirements that were taken into consideration during the design development. The South African Military standards 127 volume 1 (RSA MIL STD 127: Vol. 1) was



predominantly used for the ergonomic calculations. Where possible, dimensions are chosen so that the 95th percentile male will fit in and the 5th percentile woman will be able to reach.

4.1. The Vehicle

The required contents of the mobile laboratory were determined based on a detailed analysis of the mobile laboratory process, with consideration given to the design capacity (16 samples to be tested per day). The approximate required size of the mobile laboratory was calculated based on the the required contents. The Volkswagen Crafter was consequently chosen as the vehicle to accommodate the design. The vehicle has a long wheelbase (4.3m) and a person is able stand upright comfortably throughout the rear of the vehicle. An upgraded suspension and at least 80kW output are specified. A retractable awning is also to be added to the side door of the vehicle.

The floor is specified to be covered with 9mm thick protective cladding which is crack and scratch resistant, lightweight, anti-slip even when wet, and oil and chemical resistant. The walls are also specified to be covered in a lightweight composite cladding and the windows specified to have anti-theft protection.

4.2. Workspaces and Storage

The layout and dimensions of the workbenches, desk and seating as well as the equipment and components to be included in the laboratory are discussed in this section.

4.2.1. Layout and dimensions of the workbenches and aisle

A number of ergonomic principles were taken into consideration during the design of the workbench layout and height. Work must be kept as close to the body as possible to avoid outstretched arms and a twisted posture which place stress on the lower back [19]. Frequent work should be within 25cm of the body, with occasional work between 25cm and 50cm [20]. The height of work surfaces should be comfortable in order to avoid (i) shoulder fatigue, caused by surfaces that are too high; or (ii) strain on the back, caused by surfaces that are too low. In general this is achieved when the arms are hanging down naturally and the elbows are bent 90 degrees. For fine work, the work surface can be raised by up to 20cm above elbow height [19].

The elbow sitting height was used to determine the height of the desk workbench, incorporating the adjustment capability of the chair. It is not possible to ensure perfect height for all possible body types owing to people generally not falling in the same percentile for all the possible measurements (e.g. a 90^{th} percentile height, but a 60^{th} percentile elbow height). A desk height of 720mm is specified, where a shorter person can use a foot rest if required.

For the workbenches where the technician will be standing, the height is specified at 900mm which includes an added 100mm owing to the work requiring some precision. A minimum aisle space of 1.2m is recommended by the Occupational Health and Safety Act (OSHA). Owing to space being limited in the laboratory and only one technician working in the area implying no back-to-back work, a minimum aisle space of 850mm is specified for the mobile laboratory. The dimensions for all the devices and components were evaluated and accordingly, the workbench on which the GeneXpert[®] devices will stand, is specified to have a depth of at least 420mm (allowing for the accompanying UPS, barcode-scanner and laptop). The sink and fridge requires a workbench depth of at least 400mm and 490mm respectively. The workbenches are specified to hold the weight of all the equipment and to be of a composite stainproof and waterproof material.

4.2.2. Seating

Factors taken into consideration during the design of seating include: (i) the trade-off between under-bench space available for storage and space assigned to seated work; (ii) the estimated time to be spent on an activity; (iii) the risk of unsecured contents moving around during transit or being stolen from the vehicle. A chair bolted to the floor has limited adjustability and takes up floor space. A swivel chair for computer-based work is specified, whilst the sample preparation work is to be done in a standing posture. The chair is fastened to the side of the desk and the floor to secure it during transport, yet the distance from the desk and the chair height remain adjustable.

4.2.3. Components and Equipment

The laboratory is specified to have storage space for at least one month's supply of consumables such as GeneXpert[®] kits, gloves and N95 masks, calculated in accordance with a daily workload of at most 16 tests, adding safety stock for gloves, masks and bio-hazard bins. The sizes requirements were estimated by either measuring components at NHLS laboratories and suppliers or from online specifications.



Two four-module GeneXpert[®] systems are included in the design, each requiring a UPS, scanner and laptop. An independent laptop and printer are also required for desk work. The UPS is specified to supply at least 450 VA for 20 minutes.

A small fridge of 90 litres is specified to store at least two 16-sample holders. The laboratory requires water and accordingly a small sink with two 25 litre water canisters is specified. The canisters are located inside the cabinet, the supply-canister is equipped with a submersible pump to supply water and the drainage-canister allows for overflow to the outside of the vehicle. The sink and canisters were chosen to ensure the canisters fit in the cupboard, underneath the sink.

Specific bio-hazard bins with bags are used for any biomaterial waste. Space is allocated for one bin and storage space for three extra bins. The bins are from hard cardboard and use minimal storage space when folded up. Another bin with heavy duty plastic bags is also specified for general waste. This minimizes the amount of bio-hazard waste that is generated as this waste entails an expensive waste-removal procedure. Laboratory coat hangers and extra storage space is allocated in the laboratory along with a SABS approved fire extinguisher. Additional, but not essential items that are included are two folding-up chairs, a filing system, glove and hand-sanitiser fixtures, and a white board.

4.3. Lighting

The correct design of lighting is particularly important in laboratories, given the work hours, intensity and importance of tasks. Lighting intensity in laboratories can be up to twice that of general office environments [21]. The aim is to design the lighting to be high performing and energy efficient. Where possible the use of natural light should be encouraged as this reduces power consumption and is one of the most visually effective light sources.

Visibility, which is the clarity at which an object is seen, is dependent on illumination, contrast and visual angle. The colour rendering index (CRI) of a light source is a measure of its ability to duplicate various objects' colours in comparison with a natural (ideal) light source. The higher the CRI the better the visible light spectrum is reproduced, possibly reducing the illumination levels that would have be required in the same circumstances without a higher CRI. Natural light has a CRI of 100. A light's colour temperature is measured in degrees Kelvin (K). Lower colour temperatures (±3000 K) tend toward being more golden while colour temperatures of approximately 6000 K have a blueish tint [22].

The ceiling and walls should have a minimum of 80% and 65% reflectance respectively in order to ensure effective lighting [19]. The ceiling of the mobile laboratory has the most potential to contribute to light-distribution and accordingly the ceiling is specified to be matte-white which conforms to a recommended reflectance level of 0.8. A matte finish limits the reflectance of images from bright sources [21].

In laboratories the direct lighting aspect should be between 20% and 40% [19]. Lights are to be located directly above the edge of the workbench and parallel to the work surface. Alternatively the lights can be installed between two workbenches [21].

Fluorescent (T5 and T8), incandescent, light emitting diodes (LED) and low watt ceramic metal halide lamps were evaluated for inclusion in the laboratory. The South African National Standards 10114-1:2005 does not have guidelines specifically for mobile laboratories, but a minimum of 300 lux for laboratories can be derived [23]. A value between 75 and 90 can be approximated for the CRI resulting in an intermediate to cold colour appearance. Four 220V 5050 LED light strips were specified for the laboratory, located above the counter edges. The light strips take up little space, are energy efficient and provide the required CRI, luminance and temperature levels as mentioned previously. The two sets of lights are to be independently controlled and equipped with dimmers to save energy and ensure adjustability. The floor is specified to be dark grey which has low reflecting properties along with matte-white counter tops.

4.4. Temperature Regulation

A roof-mounted air conditioner is included to prevent ambient temperature exceeding 28°C (a requirement for the GeneXpert® cartridges). The temperature sensors are specified to be installed near the GeneXpert® devices. The most common indexes used to measure heat exposure are wet-bulb globe temperature (WGBT) and the metabolic load. These are also used to analyse the comfort level as well as how long an acclimatised and unacclimatised worker can function under given conditions. The work in the mobile laboratory is not strenuous and far from the lactic threshold. Metabolic heat calculations for males and females respectively indicate that they would be able to work comfortable even if the temperature is at the maximum of 28°C (permitted by the GeneXpert).



4.5. Electricity Requirements

The laboratory is specified to function from grid power as well as a generator. Lessons learnt from previous mobile laboratory solutions, such as the Kangoo laboratory previously mentioned, indicate that solar power solutions for mobile laboratories are unreliable and do not provide adequate power.

The generator must carry the maximum load which is calculated by using the equipment's maximum power (watt) or current drawn (ampere). The generator is specified to have an electric start and a wheel-set which should be custom made if not included, owing to all the generators providing power of above 5300 Watts weighing more than 80 kilograms. The item which has the the largest impact on the load that is to be carried by the generator is the air conditioner. The top four electricity consuming devices are displayed in Table 1. Insulation can be installed in to the mobile laboratory walls, but this is unlikely to make a meaningful contribution to temperature control.

Appliance	In	formation (ma	ximum)	# required	Safety	VA	Watt
	input A	input V	watt usage		factor		
Air-conditioner	12	230	3500	1		2760	3500
UPS	2	230	240	2	1.8	1656	864
GeneXpert	0.95	200	240	2		380	480
Laptop	4.62	19.5	90	3		270.27	270

Table 1: Top four electricity consuming devices

The vehicle is to have power points for at least all the electrical devices along with a deep cycle standby battery system and power socket for grid power.

4.6. Securing Equipment during Transit

Any equipment that may shift during transit is to be secured by means of loading trays, bolted to either the vehicle or workbenches and secured during transit with a nylon strap with a female-male connector clip. The trays should be made specifically to the equipment's size to ensure movement is minimised.

The loading ramps for the generator are to be secured with a storage bin bolted to the door, as well as straps for both the ramps (illustrated in Subsection 5.4). The generator is to be secured with a three meter ratchet strap, which is to be attached to fixtures submersed in the floor. The scanners are to be bolted to the workbench and all cupboard doors and drawers are to have self-locking mechanisms.

4.7. Noise and Vibration

The standard noise threshold limit, prescribed by OSHA for an eight hour period, is 85 A-weighted decibels (dB(A)). There is a correlation between the time of exposure and the noise level e.g. the daily limit is reached after only a hour's exposure to 89dB(A). Noise levels are reduced by decreasing sound transmission between the source and the receiver. Ear plugs, such as standard disposable earplugs (having a noise reduction rating (NRR) of 31) provide adequate protection for noise levels below 95dB(A) [19].

The GeneXpert[®] emits less than 69dB at a low frequency and will not produce irritating or harmful sounds. The specified generator is likely to emit approximately 99 dB(A) at 7m distance, which would therefore be harmful to a person. At a distance of 30m, the noise level from a generator would be approximately 86dB(A). The outdoor setting and the laboratory walls are likely to further lower the noise levels inside the mobile laboratory. A lead allowing for the placement of the generator at least 30m from the laboratory is included in the design providing sufficient noise-protection. Vibration from devices are unlikely to present a risk to health in the mobile laboratory.

5. CONCEPT DESIGN

As mentioned, the Volkswagen Crafter 50 2.0 Tdi is used in this design. Dimensions not available on the vehicle brochure were measured directly from the model. The wheel arc housings take up a significant amount of space, restricting design possibilities slightly owing to the space between housing being 978mm. A general view of the floor-area of the vehicle and laboratory can be seen in Figure 1.





Figure 1: Floor layout of the loading area and top view of the design

After evaluating various design alternatives, a layout with two countertops, one on either sides of the vehicles length, was selected. This leaves an aisle width of 860mm. The aisle width is deemed sufficient as there are no overhead storage compartments and no back-to-back work is to occur as the mobile laboratory. A general view of the mobile laboratory is displayed in Figure 2. To the front of the vehicle, in the corner opposite to the sliding door, a rounded corner desk is installed. The desk height is 720mm and the minimum counter depth 450mm, increasing to the corner of the vehicle. The height of the standing-workbenches is 900mm and all workbenches have a foot clearance of 100mm depth and 100mm height. The worktops have an overhang of 45mm for possible spills and general standing comfort.



Figure 2: General view of the mobile laboratory

5.1. Sample Preparation Workbench

The left side workbench, viewed from the back doors, has a depth of 500mm. This bench is purposed for sorting and preparing the samples and displayed in Figure 3. The fridge is allocated closest to the back doors. The opening beneath the countertop allows for a comfortable standing position as well as placement of the biohazard and general-waste bins. The general-waste bin consists of bracket fastened to the side of a cabinet, with a frame to secure a plastic bag with. The bin design takes up minimum space and allows for quick refill and disposal. The biohazard bin is secured with a strap, also partially fastened to the side of the cupboard. There is a three drawer cabinet installed above the wheelhouse enclosure, which can be used for storing the contents of an opened GeneXpert[®] kit (pipettes, sample reagents etc.) A paper towel dispenser is fastened to the wall, along with a glove box holder, a white board and to the right side, a 30mm towel rack and coat hanger. A two door cabinet with the sink and water canisters, both secured with straps, can also be seen. The clean and waste water tanks are clearly marked. There is also storage space for two extra folded up bio-waste bins.





Figure 3: Left workbench

5.2. GeneXpert[®] Workbench

The workbench on the right side is 420mm deep, with all the devices placed on the counter top as illustrated in Figure 4. The space in the middle is to place the prepared sample cartridges before transferring these into the respective GeneXpert® devices. The loading trays and straps are also displayed in the figure. In the one drawer there is enough space for the three boxes of gloves and one to two boxes of masks. The longer two door cabinet to the left, above the wheelhouse, is designed for storing seven to eight boxes of GeneXpert® kits. To the right of the cabinet is a SABS approved 2.5 kilogram fire extinguisher, fixed with a bracket to the side. There is a two door cabinet, split horizontally in the half, with storage space for a first aid kit and the thirty meter extension lead for the generator. Underneath, with a flip-down door, storage space is allocated to two camping chairs to be used outside by the mobile laboratory technician and driver during breaks times. The wheelhouse is not covered completely, to allow for more standing space, because the foot clearance here is already slightly less than 100mm.



Figure 4: Right workbench

5.3. Corner Desk

The corner desk worktop, displayed in Figure 5, offers space for the printer with its loading tray as well as the laptop. The desk has a two drawer filing cabinet as well as a smaller stationary drawer. The swivel-chair is also displayed in the figure.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 5: Corner desk with swivel chair

5.4. Transit and Generator storage

The transit view is displayed in Figure 6. The generator is loaded off and onto the laboratory by ramps which extend to two meters, but when stored are only about 860mm long. The steel storage bin is located on the lower part of the door to minimise the chance for lower-back injury. Each ramp weighs 6kg and is fastened individually with a strap.



Figure 6: Transit view (air-conditioner and lighting not shown)

6. COST ESTIMATION

The total cost for the mobile laboratory infrastructure was estimated at R659,420 (R202,917 without the vehicle cost) in 2014. A breakdown of the various cost components is given in Figure 7. Adjusting for inflation, the estimated 2016 costs are R731,883 and R225,214 respectively. The cost estimate includes the mobile laboratory infrastructure and excludes equipment that would typically be provided by the laboratory organisation using the mobile laboratory, such as the GeneXpert[®] devices, printers, laptops and consumables (such as test kits, gloves and masks).





Figure 7: Cost contribution summary - excluding base price of the vehicle (2014 figures)

7. ADDITIONAL SAFETY CONSIDERATIONS

There is potential space for the bio-safety cabinet in the vehicle, but the required additional power, air vents and the unreserved sensitivity to any external air flow influencing the effectiveness of the cabinet, pose challenges to the installation. To ensure the safety of the laboratory technician, additional concepts that will improve ventilation as well protection against airborne infections should be investigated further. Possible solutions to be evaluated include a custom made lateral airflow system which regulates incoming fresh air as well dispersing air to the outside of the vehicle. HEPA-filters can be installed in the sample preparation area and custom made protective masks and clothing should be evaluated. A custom made oven to heat only a few samples at 80°C can also be considered. It is imperative that a mask covers and encloses the technician's mouth and nose - specific attention should be paid to the fit around the nose bridge. An alternative solution may be a bio-safety helmet or mask which encloses the technician's head and shoulders, with an optional oxygen or air-filter system attached.

8. CONCLUSION

This article described the process of developing a concept design for a mobile laboratory to be used for diagnosing tuberculosis using the GeneXpert MTB / RIF test. The design was conceptualised with use in peri-mining communities in mind, though it could also be successfully implemented elsewhere. The concept design as well as a cost estimate were also presented.

The most important future design adjustment considerations relate to the protection of the laboratory technician during the sample preparation process, as well as to efficient air conditioning and the generation of power for the mobile laboratory. Alternative solutions for providing improved protection to laboratory technicians would include the installation of a customised lateral air flow system or oven. The generator poses a few challenges to the design, especially with regard to cost, noise and weight. The air conditioner contributes a significant portion of the required power load to be generated. If a more energy efficient air conditioner were to become available, this could significantly improve this aspect of the design. The use of renewable energy sources could contribute to extending the flexibility of the proposed mobile laboratory. An example of a mobile laboratory design where solar power solutions failed was cited and this motivated the decision to specify a generator in the concept design. However, renewable technology is developing rapidly and it would be worthwhile re-visiting this aspect of the design as new solutions become available.

9. REFERENCES

- [1] World Health Orginisation, 2015. "Global tuberculosis report 2015." WHO, Geneva.
- [2] Mogeni, T., 27 March 2015. "Addressing TB in the Mines: A Multi-Sector Approach in Practice Regional Dialogue: Portability of Social Benefits for Mineworkers in Southern Africa", p. 25
- [3] AERAS, 2014. "The Potential Impact of New Tuberculosis Vaccines on Mineworker Health and Productivity." [Online] Available at: http://smhttp.24906.nexcesscdn.net/80B0A2/aerascdn/img/uploads/attachments/437/aeras_mining_policy_brief_2014_2.pdf [Accessed May 3, 2016].
- [4] Baleta, A., 2012. "Southern African declaration targets TB in mining sector." *The Lancet*, Vol. 380, pp.1217-1218. [Online] Available at: http://www.thelancet.com/journals/a/article/PIIS0140-6736(12)61698-5/fulltext [Accessed January 29, 2014].
- [5] SADC, 2012. "Declaration on Tuberculosis in the Mining Sector." Mozambique, p. 9
- [6] WHO, 2013. "Diagnostics: Xpert MTB/Rif Test: WHO Recommendations." [Online] Available at: http://who.int/tb/features_archive/factsheet_xpert.pdf [Accessed March 3, 2014].



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

- [7] Stop TB, 2012. "Stop TB Partnership | Southern African heads of state join forces to end tuberculosis in the mining sector," p.2. Stop TB Partnership, Johannesburg. [Online] Available at: http://www.stoptb.org/news/stories/2012/ns12_053.asp [Accessed March 5, 2014].
- [8] Anon, August 2013. "Practical responses to TB in mining." Results.org. [Online] Available at: http://results.org.uk/sites/default/files/August 2013 BGS1 Responses to TB and Mining.pdf [Accessed March 5, 2014].
- [9] Kearney, L., 8 August 2012. "Mining and minerals in South Africa." southafrica.info. [Online] Available at: http://www.southafrica.info/business/economy/sectors/mining.htm#.VzA7n4R97IU [Accessed May 9, 2016].
- [10] Statistics South Africa, 2 March 2016. "Mining: winners and losers of 2015," [Online]. Available: http://www.statssa.gov.za/?p=6247 [Accessed: 09-May-2016].
- [11] Parker, F., 8 March 2012. Huge TB mission to combat the "silent" killer in SA mines. *Mail and Guardian*. [Online] Available at: http://mg.co.za/article/2012-03-08-huge-tb-mission-to-combat-the-silent-killer-in-sa-mines [Accessed March 5, 2014].
- [12] Partec, [n.d]. "The Complete Mobile Diagnostic Laboratory" p.3. [Online] Available at: http://www.presseportal.de/showbin.htx?id=15541&type=document&action=download&attname=cylab.p df [Accessed March 5, 2014].
- [13] Ouedraogo, R., Njanpop-Lafourcade, B.M., Jaillard, P., Traoré, Y., Mueller, J.E., Aguilera, J.F., Dabal, M., Tiendrébéogo, S.R., Goehde, W., DaSilva, A., Stoecke, P. & Gessner, B.D., 2008. "Mobile laboratory to improve response to meningitis epidemics, Burkina Faso epidemic season 2004." *Field Actions Science Reports*, 1(1), p.19
- [14] Müller, A., 2011. "The Gene Xpert". *TB Online*. [Online] Available at: http://www.tbonline.info/posts/2011/10/3/gene-xpert/ [Accessed March 5, 2014].
- [15] Darkdaily, "New Rapid Molecular Pathology Test for Tuberculosis Wins Favor with World Health Organization | Dark Daily," Darkdaily.com news, 2010. [Online]. Available: http://www.darkdaily.com/new-rapidmolecular-pathology-test-for-tuberculosis-wins-favor-with-world-health-organization-1218#axzz2v6WzKhAt. [Accessed: 05-Mar-2014].
- [16] D. Helb, M. Jones, E. Story, C. Boehme, E. Wallace, K. Ho, J. Kop, M. R. Owens, R. Rodgers, P. Banada, H. Safi, R. Blakemore, N. T. N. Lan, E. C. Jones-López, M. Levi, M. Burday, I. Ayakaka, R. D. Mugerwa, B. McMillan, E. Winn-Deen, L. Christel, P. Dailey, M. D. Perkins, D. H. Persing, and D. Alland, 2010. "Rapid detection of Mycobacterium tuberculosis and rifampin resistance by use of on-demand, near-patient technology.," J. Clin. Microbiol., vol. 48, no. 1, pp. 229-37 [Online] Available at: http://jcm.asm.org/content/48/1/229.full [Accessed February 24, 2014].
- [17] Cepheid, 2013. "The New GeneXpert® System." California. [Online] Available at: http://www.cepheid.com/administrator/components/com_productcatalog/libraryfiles (42/422-22-24/6/11/26/22/02-62/6/10-5-Com-27-2014). 2014
- files/434212e2ee4df414a6922108ef36f105-GeneXpert-Brochure-0112-08.pdf [Accessed March 3, 2014].
- [18] WHO, 2011. "Prerequisites to country implementation of Xpert MTB/RIF and key action points at country level" Checklist. WHO.
- [19] A. Freivalds, 2014. Niebel's Methods, Standars, and Work Design, 13th ed. Singapore: Mc Graw Hill, 2014.
- [20] CCOS, 08-Jul-2008. "OSH answers fact sheet" [Online] Available: http://www.ccohs.ca/oshanswers/ergonomics/standing/standing_basic.html. [Accessed: 18 October 2014].
- [21] Kozminski, K., Lewis, S. & Mathew, P., 2006. "Best practice guide: Efficient electric lighting in laboratories." Laboratories for the 21st century, pp.1-10. [Online] Available at: http://www.i2sl.org/documents/toolkit/bp_lighting_508.pdf.
- [22] CleaResult, 2013. General Lighting Recommendations, Available at: http://www.eeprograms.net/entergy/documents/CLEAResult_Gen_Lighting_Recs.pdf
- [23] SANS 10114, 2005. "SANS 10114-1:2005 Interior lighting Part 1 : Artificial lighting of interiors." Standards South Africa, Pretoria.
- [24] Van der Merwe, A. 2014. "Class notes: Ergonomic principles." Stellenbosch, 13-Mar-2014.



APPENDIX A

This appendix contains a diagram which displays the envisaged testing process for the mobile laboratory.





SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



REDUCING THE COMMUNICATION COSTS OF A REMOTE MONITORING AND MAINTENANCE SYSTEM FOR AN ENERGY SERVICES COMPANY (ESCo)

J.N. du Plessis^{1*}, J. Prinsloo² & J.C. Vosloo³

¹⁻³Center for Research and Continued Engineering Development (CRCED) North-West University, Pretoria, South Africa ¹jduplessis@researchtoolbox.com ²jprinsloo@researchtoolbox.com ³12317845@nwu.ac.za

ABSTRACT

Demand-Side Management (DSM) initiatives can result in substantial electricity cost reductions. Energy Service Companies (ESCos) typically perform remote monitoring and maintenance to ensure sustained performance of these DSM initiatives. Mobile network service providers offer the flexibility required to allow an ESCo to monitor DSM initiative that are widely distributed across South Africa. Mismanagement of the communication infrastructure and the potential for excessive data usage threatens the financial feasibility of remote DSM maintenance. Software was developed to promote improved management of this communication infrastructure, and to monitor the accessibility, data usage and other communication vitals automatically and remotely. Optimal management of the communication infrastructure ensures the lowest possible base cost, while prediction based on data usage information allows for predictive maintenance in order to avoid or reduce excessive data costs. This paper presents the process of reducing the communication cost of an ESCo that actively monitors and maintains more than forty DSM initiatives. Improved asset management and automated monitoring resulted in a 73% reduction in the monthly communication costs.

This work was sponsored by HVAC International

^{*}Corresponding author



1. INTRODUCTION

When implementing Demand-Side Management (DSM) projects on a mine and industrial plants, Energy Service Companies (ESCos) typically develop specialised control systems that implement DSM strategies. Van Heerden *et al.* [1] and Van Jaarsveld *et al.* [2] describe the development and implementation of such systems. Du Plessis *et al.* [3] showed that remote monitoring and maintenance is vital for sustaining DSM performance and the resulting cost benefit. Furthermore, the overall maintenance process can be optimised by introducing automated diagnostics that span across the industrial system targeted for DSM intervention; the system that implements the DSM strategy; and the performance assessment and reporting system.

This cascade of monitoring, diagnostics and remote maintenance systems creates a complex scheme that is highly reliant on communication infrastructure that facilitates remote maintenance. Interrupted communication impairs the ability of maintenance personnel to assess DSM performance and perform reactive maintenance based on automated diagnostics. This threatens the sustainability of DSM initiatives and could result in lost financial savings. Therefore, to make DSM maintenance financially feasible, the cost of maintenance must be kept well below the potential financial savings produced through sustained DSM performance.

2. BACKGROUND

Blumberg [4] analysed the customer calls of more than twenty service organisations and found that most service requests do not require on-site assistance. According to Blumberg [4], 80% to 90% of software-related difficulties and 10% to 40% of hardware-related difficulties can be corrected with remote assistance. Since the publication of this work in 1982, the opportunities for remote monitoring initiatives have increased, as discussed by Weppenaar *et al.* [5], due to the rapid growth of the information and communication technology field. As mentioned by Yang *et al.* [6], these systems provide large organisations with the ability to collect, store and analyse information from distributed sites.

Du Plessis *et al.* [3] and Du Plessis *et al.* [7] presented the concept for an automated diagnostic system designed to assist an ESCo to maintain the performance of DSM initiatives remotely, and provided the results obtained after implementing this system. Du Plessis *et al.* [8] elaborated on the use of this new system by presenting case studies detailing diagnostics of the DSM strategy of mine water reticulation systems; and cooling auxiliaries on different mines. These works highlighted the contribution of continuous remote maintenance towards sustained DSM performance. Furthermore, the obtained results showed the value of automated diagnostics of the DSM implementation in general; and the DSM strategy in particular. The diagram in Figure 1 shows the complexity of the DSM implementation, the diagnostics system and the communication system that allows for automated diagnostics and remote maintenance.

Hardware components are connected through different networks. The equipment and instrumentation are connected to the Programmable Logic Controller (PLC) and Supervisory Control and Data Acquisition (SCADA) system through an industrial network. The DSM server connects to the PLC and SCADA system via an Ethernet network, and uses an Open Platform Communication (OPC) connection to read values from field instrumentation, and to control industrial equipment according to a specific DSM strategy.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 1: Remote diagnostics and maintenance system (constructed from [3], [7], [8])

A mobile router connects the DSM server to the ESCo server via mobile networks. Mobile networks provide the flexibility required to monitor DSM initiatives that are widely distributed across South Africa [9]. The execution of the DSM strategy is not reliant on this mobile network, yet, the mobile network allows for prompt notification of irregular operation. Furthermore, this mobile network connectivity also allows maintenance personnel to access the DSM server remotely in order to perform corrective maintenance.

Interruptions in mobile network connectivity threaten the sustainability of DSM initiatives. Some geographic areas experience weak or no signal for a specific mobile network service provider, which makes it impractical to use one service provider exclusively. A private Access Point Name (APN) provides a cost effective method for linking Subscriber Identification Module (SIM) cards from different service providers to the same secure network [10]. This private network only facilitates connectivity between SIM cards that are provisioned with access to this APN. Figure 2 shows a private APN facilitating connectivity between SIM cards from different service providers.



Figure 2: Data flow through an APN

SIM cards are linked to active mobile lines that are procured from mobile service providers on a contract basis, with fixed monthly data allocation. This ensures continuous connectivity and reduces the overhead created by replenishing prepaid lines on a monthly basis. In turn, the APN service provider charges a fixed monthly fee for each mobile line that is provisioned on their APN.

3. MANAGING COMMUNICATION COST

Data contracts are a major contributor to an ESCo's monthly communication bill, illustrating the roll of optimal managing of mobile lines in reducing communication costs. Mismanagement of mobile lines can result in the over-procurement of data contracts. However, optimal management of mobile lines, linked to SIM cards that are distributed across the country has proven to be challenging. Table 1 lists four circumstances that can be avoided or quickly identified through comprehensive management. The table also shows what risk factors are involved for each of these circumstances. The first risk factor indicates the potential for excessive data usage and cost. The second risk factor indicates that the SIM card has become unavailable for use and therefore could result in procurement of unnecessary contracts.



	Risk f	actor	
Circumstance	Data	Lost	Description
	usage	SIM	
Unnecessary redundancy		Х	Routers support failover to a secondary SIM card for areas with weak signal reception. Improved reception on one of the SIM cards voids failover. Secondary SIM card becomes dormant.
Accidental redundancy		Х	Secondary router installed although the site already has a router that is not on the asset register.
Offline router		Х	Router is broken, removed or disconnected.
Theft or abuse	Х	Х	Site personnel use a SIM card for personal use.
Malfunction	х		Failed data transmission is mistakenly perceived as successful, resulting in continuous retransmission of the same data.
Malware	х		Certain malware probe the computer network to infect other machines. Probing involves continuous transmission of small data packets that amount to large data transmission on a monthly basis.

Table 1: Risks associated with mismanagement

Monitoring the cost of communication in general and the cost of individual lines in particular is the starting point to identify excessive data usage. The first option is to use information from the monthly bills to identify overspending based on a predefined budget. A monthly bill can however only be used to identify situations where excessive usage has already occurred. The bill will reflect the data cost of the previous month, or depending on the billing cycle and billing date, the cost of the month before the previous month. To illustrate this, Figure 3 shows the costs generated by a single mobile line over a five-month period in 2013. This figure also shows the cost reflected on the bill for each of these months. The first month shows the normal monthly cost of R 149. During the second month, this cost escalated to R 2 355, and excessive costs continued for the next three months.

If the billing date is set for the last day of every month, it does not allow sufficient time to consolidate the data usage for that specific month. Therefore, the bill received on the tenth day of Month 5 reflected the data cost generated in Month 3. At that stage, Month 4 already produced a similar data usage, which continued for the first ten days of Month 5. This resulted in a total overspend of R 58 185 during this five-month period. Therefore, using monthly bills to monitor communication cost does not significantly reduce the risk of excessive costs.



Figure 3: Delay between usage and billing dates

To improve the resolution of data checking, complete data usage reports were requested from the service providers on a weekly basis. These requests were occasionally serviced promptly, but other requests were delayed for up to a week. After escalating these requests week in and week out, it became apparent that this is not a sustainable method to monitor data usage figures. Most service providers grant access to account and billing information using an online portal. These portals allow for manual generation of data usage reports, which ensured that the reports could be generated consistently on a weekly or even daily basis. Importing this data on a daily basis remains a manual process that is prone to human negligence.



4. AUTOMATED MONITORING

Heyns *et al.* [11] emphasised the importance of performing a root cause analysis using appropriate information, before making decisions. Software was developed to request information from routers on a daily or even hourly basis, in order to promote access to the correct information, which allows for prompt reaction to identified risks.

4.1 Asset management

The developed software allows the user to import a detailed subscriber list holding information on the mobile lines registered at each service provider. In addition to this information, the software can import a list that links the private APN Internet Protocol (IP) addresses to the specific mobile line. Table 2 lists the information that is retained in the database of the developed software, and the reason why this information is required.

Information	Need			
SIM number	Unique serial number of the SIM card used for asset manage			
Subscriber number	Unique number used to administrate the contract			
Contract type	Indicates the size of the recurring monthly data allowance			
Contract expiry date	Date after which a contract can be reviewed			
Subscriber status	Indicates weather a subscriber is active or locked			
APN IP address	Address used to access the site remotely			

Table 2: Mobile line and APN information

The software allows the user to allocate a mobile line or multiple mobile lines to a site or personnel. In extension to this, the mobile line that is allocated to a site might be allocated to different DSM initiatives implemented at that specific site. The software therefore enables administrative personnel to keep an electronic record of SIM card allocations, including the responsible person for each DSM initiative. Furthermore, the software allows maintenance personnel to gain remote access to the onsite DSM server, using the information in Table 2.

4.2 Monthly budgeting

The developed software also allows maintenance personnel to import monthly invoices in order to analyse the actual cost against the budget for individual mobile lines. The aim is not to exceed the fixed monthly contract cost, and therefore this cost is used as the budget for each individual mobile line. Any mobile line that exceeds the budget is highlighting for corrective action. Furthermore, the developed software provides an overview of budget and actual costs for the last year, assisting maintenance personnel to identify mobile lines that exceed the budget on a regular basis.

4.3 Communication checks

The developed software periodically checks the level of communication to all the SIM cards installed in on-site routers. The level of communication is calculated based on the number of successful ping requests sent to each router. This information is then used to calculate an average daily communication level for each router, which is combined to form a monthly profile. Personnel use this profile as an indication of the stability of the connection to each installed router. Additional information about the communication channel is obtained by sending Hypertext Transfer Protocol (HTTP) requests to the router. The information that can be requested from the router includes the signal level, the current active SIM (primary or secondary) and router serial number.

Furthermore, the software initiates Unstructured Supplementary Service Data (USSD) commands on the router in order to query the data balance of the mobile line linked to the installed SIM card. This data usage is then represented as a data usage profile from the start of the current billing cycle, giving maintenance personnel an indication of the data usage tendencies of each site. The software can provide a snapshot of the latest data usage figures for all mobile lines. A prediction of the total data usage at the end of the current billing period allows the ESCo to benefit from preventative maintenance, as discussed by Vermaak *et al.* [12].

5. IMPLEMENTATION

At the beginning of 2014, the total number of SIM cards used for remote maintenance peaked at 187, with a combined monthly cost of approximately R 50 000. No spare SIM cards were available, therefore requiring a new 24-month data contract every time a new SIM was required. In mid-2014, an investigation was initiated to verify the allocation of all SIM cards.

5.1 Investigation

The list of existing SIM cards and the last known allocation of each SIM card was imported into the developed software database. Figure 4 shows an outline of the investigation process. The first objective was to request a list of SIM cards that have been dormant for more than three consecutive months. This list was imported into



the software and all the dormant SIM cards were marked for replacement. As a second objective, 44 personnel SIM cards were verified. The third objective was to verify SIM cards assigned to sites. Remote connections were used to verify 47 of these SIM cards, while 28 SIM cards required site visits for verifications and reinstatement of remote access. Installation records for portable data logger, that require voice contracts for dialup based data transmission, verified the locations of a further 22 SIM cards.



Figure 4: Investigation process

The itemised bills of the six remaining SIM cards were used to identify suspicious activity. Three of these SIM cards showed billing for SMS messages and voice calls to various unknown numbers. Voice calls to the three corresponding mobile lines verified that unauthorised individuals were using the SIM cards. The remaining three SIM cards could not be located and did not show significant data usage; therefore the SIM cards were replaced.

5.2 Automated data collection

This investigation resulted in an updated database of all the SIM cards owned by the ESCo, and a substantial increase in the number of spare SIM cards. The software was configured to start collecting information from all routers that are allocated to sites. Figure 5 shows the overview page of the developed software with a list of DSM initiatives and the responsible personnel. The APN IP address is used to request data from the router and the SIM card remotely. This data is categorised as indicated by the five blocks in Figure 5.

The software shows when communication to a site is lost for prolonged periods and will therefore assist to rectify problems promptly. In addition, the data usage profiles of sites are constructed and allows for reassessment of the contract type and additional bundling. This information is also vital before implementing risk-limiting strategies, in order to prevent necessary interruptions in communication.



Site	Project	Engineer	IP	Ping	Ping	Signal	Data (month)	Current SIM	SIM config	Number	Provider	SIM	Router ID
Rodens 12/38	Board-RC	Reynhardt	10.223.2.62			111		Primary	Primary	10.000.000	MTN	14.5286,7885	TSH000447
Boen Water	Webschept/P	Ru-El	10 123 2 251			att		Primary C	Primary	1.10000-0121	Vodacom	J J	TSH000448
Cooke	Code IP	Mome	10 123 4 231		5	att	0	Primary C	Drimary	e l	Vodacom	U U U	TSH000471
Cooke	CoskeO4N	Mome	10 123 4 238		ŭ	att	5	Primary	Primary	jq	Galacom	E 9	TSH000274
Dahaka	Develop	Janco	10.223.0.39		Jan Jan	att	5	Primary	Primary	2	.≌N	e d	TSH000276
Dahaka	Databaciliti	Janco	10.223.0.19		.≌ Š	all	<u>e</u>	Primary	Primary	5	tĕ	l l l l l l l l l l l l l l l l l l l	TSH000276
Deheko	Deser	Janco	10 123 3 1		ΞŪ	att	e l	Primary	Primary	<u>а</u>	Calacom	9 2	TSH000312
Scorkog(24)	Doorkop/248	Andries	10 123 4 68		5 5	at		Primary	Primary	e,	dacom	- 9	TSH000338
ini	Ave CM	Peach	10.123-8-22		EŬ	att	С1	Primary	rimary	9.	dacom	_ <u>_</u> <u>.</u>	TSH000412
Later	Applex .	Mariska	10.223.0.31		5	at	a	Primary	Primary	2	. <u>₽</u> N	1 1	TSH000456
God	Red CH	Mome	10 123 4 78		U U	all		Primary	Primary	e.	Vodacom	S S	TSH000273
God	No. 1930	Mome	10.123-6.32		a links deal	att		Primary U	nimary	0	Vodacom	As	TSH000277
Considers)	Kanadeth-Ch	JacoF	10 123 1 198			at		Primary	Primary	NUT NOW	Vodacom	Inclusion of the	TSH000333
<													>

Figure 5: Communication overview on developed software

5.3 Reducing the risk of overspending

The next step was to ensure that the risk of excessive usage is reduced to a minimum. A cost limit is required in order to prevent excessive data costs as discussed in section 3. Mobile network service providers might offer the option to prevent data usage in excess of the allocated monthly data allowance. One of the ESCo's service providers did not provide this same option, however, it offered a cost limit instead. The mobile line is locked once the cost exceeds the budget by more than the elected cost limit amount. A mobile line lock prevents further charges and is only removed at the start of the next billing cycle. Figure 6 shows the maximum cost of six mobile lines with active cost limit of R 100. Line 6 exceeded the limit by up to R 1 500. The service provider stated that the exact cost incurred before the mobile line is locked cannot be guaranteed.



Figure 6: Monthly costs exceeding the cost limit

Although this unguaranteed limiting is not ideal, it was found that only exceptional cases would result in the budget being exceeded by more than R 1 000. Therefore, the risk of excessive costs that compare to the costs presented in section 3 is eliminated. Budget violations are further reduced by actively monitoring the data usage with the developed software. As discussed in Section 4, the automated monitoring system generates a data usage profile and predicts future data usage. This mechanism allows maintenance personnel to identify potential budget violations and take immediate corrective action to prevent or minimise the extent of overspending.

5.4 Reducing the base cost

Following the investigation process, 46 SIM cards became available for reallocation. There is however no need to keep such a large amount of SIM cards as spares. Therefore, a substantial portion of the spare SIM cards was marked for termination upon contract expiry. Base costs can be reduced further by contract upgrades, contract downgrades, and bundle adjustments. The automated monitoring software discussed in Section 4 allows for a data usage profile to be constructed. This profile clearly illustrates the behaviour of the mobile line, which can be analysed in order to determine whether a contract adjustment is required. If the overspending is found to be more expensive than a contract upgrade or a recurring data top-up, the contract is upgraded to minimise overspending. When a mobile line utilises significantly less data than the monthly allocation, a contract downgrade is considered after the contract expires.

6. RESULTS AND DISCUSSION

The implementation process started with a three-month investigation in June 2014. In October 2014, a pilot version of the developed software was implemented and updated with all the verified data. The software



started to perform automated data collection. In subsequent months, the functionality of the software was expanded continuously.

6.1 APN cost reduction

Mobile lines could be cancelled immediately after the initial investigation because these contracts were still in place. Early contract termination results in a settlement cost comparable to the outstanding monthly costs until the contract end date. Figure 7 shows the monthly APN cost and the corresponding number of active mobile lines. Although the first mobile line cancellations only took place in May 2015, unnecessary APN subscriptions could be cancelled with a notice period of one month.



Figure 7: APN subscription cost and number of lines

In November 2014, the first 22 mobile lines were deregistered from the APN as shown in Figure 7. As mobile lines were marked for cancellation, APN registrations were further reduced to a total of 79 in February 2015. The number of APN subscriptions fluctuated over the following months as communication requirements change. An APN tariff increase caused the total cost of the subscriptions to increase in July 2015 even though the number of subscriptions was reduced. Despite this tariff increase, the total monthly APN cost was be reduced from R 2 440 in November 2014 to R 1 700 in April 2016.

6.2 Mobile line base cost reduction

The APN cancellations provided an immediate reduction in cost, but the proposed mobile line cancellations had a much larger potential cost reduction. Figure 8 shows the total monthly mobile line cost starting in January 2014, compared to the budget of each month. The monthly cost is calculated based on the total cost of voice contracts and the total cost of data contracts for the specific month. The bars in the figure also show the number of mobile lines allocated to each of these costs. It is clear that in January 2014, 67% of the costs were generated by 33% of the mobile lines. Many of these voice contracts could be replaced by data contracts, because dialup connections were not used any more.



All SIM cards of mobile lines that were marked for cancellation were made unavailable for allocation to sites. Active management reduced the base cost of communication from R 44 324 in January 2014 to R 11 996 in April



2016. This reduction was achieved using the developed software to show the real demand for mobile lines compared to the current asset allocation. This resulted in a decrease in mobile lines from 180 to 82. In addition, expensive voice enabled contracts could be grouped together and replaced with existing data contracts in order to reduce unnecessary costs. The 60 voice enabled contracts were reduced to 4, which expired in June 2016.

6.3 Mobile line overspend reduction

In addition to reducing the base cost, the developed system allowed for better data usage tracking on a daily basis, hence reducing the incidence and severity of overspending. In Figure 8, only three major incidences of overspending occurred. In January 2015, a voice-enabled contract with no data allowance was installed on site in error. The mobile line immediately started to use data at a cost of R 2 per megabyte. The cost escalated and the SIM card was not being actively monitored because it was not allocated to a site. In February 2015, one of the personnel SIM cards was stolen and it was only reported stolen the next day. The thieves generated a bill of R 1 693.19 before the limit lock blocked the SIM card. In May 2015 overspending was identified on a mobile line that was prohibited from exceeding the monthly data allowance. Furthermore, the developed software showed a substantial data balance at the end of the billing period. This contradictory information was as result of a billing error that was rectified on the next bill.

7. CONCLUSION

Effective management of communication infrastructure can minimise the cost of monitoring DSM implementations distributed across the country. Communication costs can contribute considerably to the ongoing DSM maintenance cost and can have a negative impact on the overall financial savings achieved by a DSM initiative. This paper shows how the communication cost of an ESCo can inflate maintenance costs. The cost of communication should therefore be reduced to a minimum in order to ensure cost effective DSM maintenance.

When using a mobile network for communication, mismanagement results in an increased base cost, while sporadic data usage spikes results in costs that exceeds the budget. Software was developed to improve this management process and to increase the resolution at which communication-related information could be retrieved; from a monthly period to an hourly period. By focussing on preventative maintenance procedures, the risk of overspending was significantly reduced.

REFERENCES

- [1] S. W. van Heerden, R. Pelzer, and J. H. Marais, "Developing a dynamic control system for mine compressed air networks," in 2014 International Conference on the Eleventh industrial and Commercial Use of Energy, 2014, pp. 1-8.
- [2] S. van Jaarsveld, J. N. du Plessis, and R. Pelzer, "A control system for the efficient operation of Bulk Air Coolers on a mine," in 2015 International Conference on the Industrial and Commercial Use of Energy (ICUE), 2015, pp. 133-137.
- [3] J. N. Du Plessis, R. Pelzer, and M. Kleingeld, "An automated diagnostic system to streamline DSM project maintenance," in *Proceedings of the 9th Conference on Industrial and Commercial Use of Energy (ICUE)*, 2012.
- [4] **D. F. Blumberg**, "Remote diagnostics for improving field service productivity," *Computer (Long. Beach. Calif).*, vol. 15, no. 11, pp. 70-77, Nov. 1982.
- [5] **D. V Weppenaar, H. Vermaak, and J. Kinyua**, "Utilising multi-agent systems to develop an intelligent maintenance management system based on the Drools-Planner," *J. New Gener. Sci.*, vol. 10, no. 1, pp. 170-190, 2012.
- [6] S. H. Yang, C. Dai, and R. P. Knott, "Remote maintenance of control system performance over the Internet," *Control Eng. Pract.*, vol. 15, no. 5, pp. 533-544, May 2007.
- [7] J. N. Du Plessis, I. M. Prinsloo, and H. J. Groenewald, "Results from implementing a remote diagnostic and maintenance solution on energy management systems," in *Proceedings of the 10th Conference on Industrial and Commercial Use of Energy (ICUE)*, 2013.
- [8] J. N. du Plessis, R. Pelzer, and J. C. Vosloo, "Sustaining the performance of diverse energy management systems through reactive maintenance," in 2015 International Conference on the Industrial and Commercial Use of Energy (ICUE), 2015, pp. 44-49.
- [9] H. J. Groenewald, M. Kleingeld, and J. C. Vosloo, "A performance-centered maintenance strategy for industrial DSM projects," in 2015 International Conference on the Industrial and Commercial Use of Energy (ICUE), 2015, pp. 50-53.
- [10] H. J. Zhou, C. X. Guo, and J. Qin, "Efficient Application of GPRS and CDMA networks in SCADA System," in *IEEE PES General Meeting*, 2010, pp. 1-6.
- [11] J. H. Heyns and P. J. Vlok, "CORRELATION AND CAUSATION: A POTENTIAL PITFALL FOR EFFICIENT ASSET MANAGEMENT," in Southern African Institute for Industrial Engineering 2014, 2014.



[12] H. Vermaak and J. Kinyua, "Multi-agent systems based intelligent maintenance management for a component-handling platform," 2007 Ieee Int. Conf. Autom. Sci. Eng. Vols 1-3, no. November 2015, pp. 709-714, 2007.



AN EVALUATION OF PROJECT TEAM WORK: TEAM LEADERS' PERSPECTIVES

A.T. Murray¹ & A.S. Lourens²

¹Department of Industrial Engineering Nelson Mandela Metropolitan University, South Africa andrew.murray@nmmu.ac.za

² Department of Industrial Engineering Nelson Mandela Metropolitan University, South Africa ann.lourens@nmmu.ac.za

ABSTRACT

Teamwork is an acceptable and structured process used by academics to assess progress in academic modules. Team leaders are usually elected to coordinate team activities required for the successful completion of projects. In the process of teamwork, team members and team leaders alike gain valuable skills that will prepare them for life and the world of work. In addition, management and teamwork are also an engineering programme requirement of the engineering accreditation body. This study reports on team leader interviews conducted on completion of a major project executed in an engineering module in an engineering programme. The research was conducted to investigate the effectiveness of team projects in terms of team member commitment, project execution and the associated skills practised. The research can contribute to the structuring of project teams, measures of fair assessment and improving the quality of team project experience.



1. INTRODUCTION

Cooperative learning, as an effective teaching method, has been defined as instruction that involves students working in teams to accomplish a goal while under conditions that include (Felder and Brent, 2004) [1]:

- elements of positive interdependence,
- individual accountability,
- face-to face interaction,
- appropriate use of collaborative skills
- team processing

Cooperative learning has been used by an Engineering department at a South African university to incorporate the teaching of engineering principles with team learning, communication development, leadership, initiative, and creativity (Lourens et al., 2015) [2], (Lourens et al., 2016) [3].

With the introduction of an engineering laboratory at the university, it was decided to introduce innovative teaching practices in a cooperative learning framework [2]. Accordingly, a project was developed for a specific engineering module where students utilised the modern facilities and equipment whilst introducing non-traditional methods of teaching and learning as based on the research of Froyd et al, 2012 [4], Kuri, 2014 [5], and Lenschow, 1998 [6]. The project brief required students to design and build a product that could be tested for speed and distance whilst meeting other specified criteria. Students were also encouraged to experiment with the size and length of all the parts on the product and to be as innovative as possible.

As the teamwork project for the specific module evolved over the years, several areas of interest and research were identified. Particular attention was paid to the design of the project [2] and the influence of compiling teams with varying learning styles [3]. This research reports on the feedback obtained from team leaders over the course of two years.

Therefore, the main aim of the research is to investigate how team leaders experience and manage teamwork. The investigation particularly referred to the Engineering Council South Africa's (ECSA) Exit Level Outcome 8 (Engineering Council South Africa, 2012) [7] which specifies individual and teamwork requirements for engineering programmes. This investigation could thus enhance the project brief or instructions issued to students. Furthermore, it is anticipated that this research can contribute to how team projects are designed, developed, managed, and assessed to improve the quality of the team experience for individual team members.

2. THEORETICAL FRAMEWORK

Five shifts have been identified in engineering education. and inlcudes [4]:

- an analytical emphasis in engineering science,
- outcomes based education,
- renewed emphasis on design,
- applying education, learning and social behavioural sciences research,
- the influence of information, communication and computational technologies on engineering education. [4].

The project associated with this research focused on design, learning and social science and the influence of information, communication, and computer technologies in engineering education. The project brief was discussed with teams prior to the commencement of the project. The explanation of the project requirements guided the students and highlighted the difficulties that could be encountered. The importance of teamwork, the required leadership, planning and time management was explained. This gave the students a better understanding of the required project outcomes and at the same time influenced the selection of the team members and team leaders by individual students.

Leadership and teamwork were thus emphasised right from the start of the project. Students were briefed on the importance and benefits of teamwork and its associated responsibilities, not only for the elected team leaders but also for individual team members.



2.1 Importance of teamwork

The emergence of team assignments, although not a new concept, represents a major trend in education [2], [4], [6]. This trend, according to Ford and Morice (2003) [8] could be the result of increased student numbers, which have forced academics to turn to team assignments to lessen the academic workload in certain modules. It could also be because employers value teamwork skills highly and seek development in university graduates in this area [8]. Furthermore, teamwork is also considered one of the Exit Levels Outcomes for an engineering qualification and therefore important to incorporate in the engineering curriculum.

The ECSA range statement for Exit Level Outcome 8 (individual and teamwork) is described by ECSA as follows [7]:

- i. The ability to manage a project should be demonstrated.
- ii. Tasks are discipline specific and within the technical competence of the graduate.

Management principles include:

- i. Planning: set objectives, select strategies, implement strategies, and review achievement.
- ii. Organising: set operational model, identify and assign tasks, identify inputs, delegate responsibility, and authority.
- iii. Leading: give directions, set an example, communicate, motivate.
- iv. Controlling: monitor performance, check against standards, identify variations, and take remedial action

Given the importance of meeting ECSA requirements, a team project in an engineering module would be the ideal opportunity for students to develop individual and teamwork skills. Furthermore, the practice of educational objectives (specifically engineering, leadership, and teamwork related) gave students a better understanding of the outcomes to be achieved. Furthermore, it forced them to utilise problem-solving processes during the project, thus increasing their readiness for employment.

2.2 Benefits of teamwork

Team assignments provide an opportunity for students to engage in peer-to-peer learning which will enhance their ability to clarify and share knowledge, and develop their problem solving abilities and leadership skills (Almond, 2009) [9], (Johnson & Johnson, 2005) [10], (Burdett and Hastie) [11]. Other benefits could include higher-level learning, better communication, conflict management, and greater understanding [9]. At the same time, according to Candy et al., (1994) [12] skills that are transferable to the work environment such as teamwork, time management and interpersonal skills could be developed.

Burdett and Hastie (2009) [11], found that students do not always regard team assignments positively because dissatisfied students could inhibit and impede the performance of other team members and this could result in a poorer outcome. However, because students need a solid knowledge of fundamental principles, a domain of refined strategies for problem-solving, capacity to think and act in independent ways and to work and learn in teams, teamwork is a valuable approach [5].

Because of the associated benefits, effective learning associated with teamwork could lead to more competent practising engineers. This is especially important in light of more service and production industries indicating *competence* as the most important factor for competitiveness [6].

3. METHODOLOGY

A compulsory semester module in an Engineering programme was selected as a module that lends itself to a cooperative learning approach with a focus on developing teamwork and leadership skills. A project brief explaining all requirements was discussed with students prior to commencement of the project. As part of the teaching and learning objectives of the module, students were required to work in teams. These teams could consist of between four to five students and teams elected team leaders [3].

On completion of the project, team leader interviews were conducted. These team leaders were asked questions to determine not only their team members' commitment and involvement in the project, but also other issues that had an effect on the completion and success of the project. Team leaders were also asked to recommend changes for future team projects.



3.1 Interviews

Over a two-year period, team leaders were interviewed to obtain feedback regarding their experience as team leaders. Interviews were scheduled after the final assessment had been completed on the projects. It was not compulsory for team leaders to attend or partake in the interview session.

Year	Number of teams	Number of team leaders interviewed
Year 1	11	11
Year 2	8	5

3.2 Interview protocol

Year 1

All eleven team leaders attended the interview session. Fifteen questions were asked during the interview. Responses were coded into eleven categories. Table 1 shows the categories of the questions.

	Table 1: Qualitative interviews					
	Categories assigned to the project process	Corresponding interview question number				
1.	Selection of team members	11				
2.	Team leader election	8 and 10				
3.	Change of leadership	9				
4.	Project analysis	3				
5.	Coordination of tasks	2				
6.	Time management	1				
7.	Delegation	4,6,14				
8.	Delegation of tasks and leadership	5				
9.	Conflict	12 and 13				
10.	Combination of males and females in the team	7				
11.	Project improvements	15				

All the team leaders attended the interviews. Certain team leaders were present but did not participate in the interview session.

Year 2

-

Five team leaders participated in the session, although eight teams participated in the project. Four extra questions were included in Year 2 that led to two new categories. The new category numbers are 6, and 13 as listed in Table 2.

	Table 2: Qualitative interview	/S
	Categories assigned to the project process	Corresponding interview question number
1.	Selection of team members	5
2.	Team leader election	6, 12,18
3.	Change of leadership	11
4.	Project analysis (size of the project)	4
5.	Coordination of tasks	2
6.	Planning (project)	3
7.	Time management	1
8.	Delegation	8, 9, 15
9.	Delegation of tasks and leadership	7
10.	Conflict and getting rid of team members	13, 14
11.	Team dynamics and gender in the team	10
12.	Project improvements and improvement of the quality of the	16, 19
	finished project	17
13.	Effect of lab technician	18

Feedback was then further categorised into planning, organising, leading, and controlling.


4. FINDINGS AND DISCUSSION

For the purpose of this research, the responses from team leaders in Year 1 and Year 2 of the project have been combined and are reported below. Feedback was summarised into management principles identified in the ECSA range statement for Exit Level Outcome 8 (The ability to manage a project should be demonstrated). Accordingly, feedback is reported in the following categories as defined by ECSA [7]:

Planning: set objectives, select strategies, implement strategies, and review achievement.

Organising: set operational model, identify and assign tasks, identify inputs, delegate responsibility, and authority.

Leading: give directions, set an example, communicate, motivate.

Controlling: monitor performance, check against standards, identify variations, and take remedial action.

Selection of the teams and team leaders

Students selected their own teams. The students that were not assigned to any teams were either teamed together or placed in a team that had capacity. No student was forced into a team. Over the two years, only one team leader was self-elected while other team leaders were elected by their teams.

The feedback captured in the interviews showed that the students prefer to choose their own teams and select their own leader. Some team leaders indicated that previous team leaders had been changed during the course of the project because of the following perceived issues: lack of self-esteem, lack of confidence, inability to guide the team or the level of maturity.

This indicated that some of the students displayed the necessary maturity in dealing with potentially awkward situations that could lead to conflict and loss of friendships. Furthermore, it indicated that they viewed the successful completion of the project in a serious light.

Planning: Coordination of tasks and time management

In most cases, the team leader was responsible for planning all the activities associated with the project execution. Setting priorities and developing action plans at the beginning of the project helped some teams to manage the project. Delegation of responsibilities assisted teams in achieving their goals. However, underestimating the complexity and time required executing the project, the amount of input, team involvement, time management, and last minute changes to the finished product were issues that all teams indicated as obstacles for successful project completion.

The feedback indicated that team leaders in general were aware of their responsibility of managing their teams to complete the project. It also became apparent that team leaders needed regular support and guidance from the lecturer.

Organising: Project analysis and delegation

A unanimous response from team leaders was that they underestimated the size and duration of the project. From the interviews, it became apparent that team leaders did not always organise tasks and deadlines effectively. Poor quality of work resulting in teams falling behind schedule placed pressure on team leaders. If team leaders did not schedule time with the laboratory technician to use certain of the equipment, it also resulted in bottleneck situations in the laboratory, which led to tasks not completed on time.

Delegation of tasks by team leaders, irrespective of age or gender, did apparently not pose any problems. However, team leaders felt that they worked harder than their team members did. This is because the majority of the team leaders indicated they had to redo either work delegated to others, or that they had to do the work themselves from the start. Certain teams did comment that age and gender had an effect on who was appointed as the team leader. They felt that the older students with work experience might be better leaders.

The feedback from the team leaders thus indicated certain shortcomings in their organisation of work. Possible reasons are that they were unfamiliar with the processes and the associated required laboratory time.

Leading: Conflict

All teams interviewed, apart from one team, experienced conflict in their team. Conflict was thus a common occurrence in all the teams. In general, team members because of the following reasons caused conflict:

- non-attendance at the briefing session
- not understanding the project requirements
- not pulling their weight
- not taking the project seriously



- a negative attitude towards the project
- not willing to get involved or unwilling to be a team player.

Any conflict was handled within the teams; this was done by scheduling team meetings to discuss the issues at hand. In extreme cases, the team leaders had no option but to expel team members not willing to cooperate. All the team leaders reported that the responsibility to complete the project fell on the team leader's shoulders.

The majority of the teams thought it good to have a mix of both males and females in a team, but they also reported that the gender of the team members was not important. The team leaders felt that the way team members contributed to the completion of the project in terms of management, team dynamics, capability, and other skills was important from a teamwork perspective.

Controlling: Project improvements and quality of the finished project

All the project requirements as stipulated in the project brief were explained in class. Some team leaders indicated that a lack of understanding of the brief was because either students were not in class or they did not pay attention during the briefing.

The team leaders indicated that the closer the due date, the more responsibility fell on them to complete the project. This resulted in team leaders having to complete tasks allocated to team members. Team leaders realised that this was due to poor time management, organisation, and control of the project schedule.

Laboratory technician

In Year 2, a laboratory technician was employed and overall improvements in the final products produced from Year 1 to Year 2 were clearly visible. The laboratory assistant, experienced with all the laboratory equipment required for executing the project therefore had a positive effect on the products produced in Year 2. Team leaders indicated that the laboratory assistant was very involved in assisting and guiding teams and displayed the relevant knowledge of the equipment utilised in executing the project. They were appreciative of this dedication and felt that a laboratory technician was definitely beneficial to the successful outcome and completion of the project.

Feedback from the team leaders clearly indicated the relevance and need for dedicated technical assistance in executing their projects.

5. **RECOMMENDATIONS**

Based on the study's findings, recommendations are made that mostly pertains to improving students' general and project management and leadership skills.

The team leaders felt that they did not require any extra training in managing projects. However, it is recommend that leadership training prior to a major teamwork project could assist to reduce the amount of conflict encountered throughout the duration of the project. Leadership training could assist team leaders to understand the tasks and duties associated with being a team leader. Training can include the stages of team formation, motivating the team, setting deadlines, planning, and assessing team members' strengths and weaknesses to allocate tasks accordingly. The process of getting to know team members and organising tasks according to individual strengths would teach valuable management, self-knowledge, and interpersonal skills to both team leaders and team members.

A major shortcoming on the part of team members and team leaders appeared to be a lack of time and project management. Teams need to understand the concept of breaking jobs into manageable parts and assigning realistic time values to these tasks whilst planning for contingencies and rework. They need to understand that constant feedback, communication, and control are requirements for successful project execution. Detailed project briefings and assisting teams in planning and setting control measures such as peer review meetings could assist in this regard. Team leaders recommend that more emphasis is placed on understanding the project brief and that it should be made compulsory for all students to attend the project briefing.

Furthermore, it became apparent that a peer review system for individual parts of work executed by team members could lead to improved quality of work and therefore meeting deadlines. Regular peer reviewing could also minimise the workload of team leaders and the resultant pressure from redoing team members' work.

Team leaders recommend a bi-weekly report back session with the lecturer for a progress report from team leaders. Team leaders would then obtain feedback from the lecturer and peers on their progress and guidance on dealing with difficult situations. Team leaders would then provide feedback to their teams. Team leaders



felt that they needed support from the lecturer so that the importance of timeous work and scheduling would be emphasised to team members. The feedback meetings will also allow team leaders to compare their project progress and serve as an indicator for teams if they were falling behind schedule.

The assessment rubric included peer assessment marks for team member participation. This proved successful where students had the courage to assign lower marks to team members who did not meet their deadlines or successfully execute their tasks. It is this recommended that students understand the importance of this measure and that peer marks are allocated honestly, fairly and responsibly.

The introduction of a laboratory technician had a positive effect on teamwork and project execution. The teams who enjoyed the benefit of a dedicated laboratory technician produced higher quality projects and felt that the laboratory technician played an important role in successful project execution.

6. CONCLUSION

General benefits that can be attributed to this team project are the understanding of productivity and performance, skills development within the team, better self-knowledge and improving teamwork and project management skills. The aim of this research was to investigate how team leaders experienced and managed teamwork. The investigation particularly referred to the Engineering Council South Africa's (ECSA) Exit Level Outcome 8 (Engineering Council South Africa, 2012) that specifies individual and teamwork requirements for engineering programmes. Based on the feedback from team leaders, they practised and enhanced their planning, leading, organising, and controlling skills in the execution of this project in an engineering module. Even if many further opportunities for development were identified, due to a possible lack of maturity and experience, students were given the opportunity to work in teams and develop skills associated with being part of a team or leading a team.

This investigation could thus enhance future project briefs or instructions issued to students by including regular scheduled progress reports with teams. This could assist the team leader to understand, develop, and improve their leadership skills. Furthermore, it is anticipated that this research can contribute to the way team projects are designed, developed, managed, and assessed to improve the quality of the team experience for individual team members. This can result in improving their readiness for life and the world of work.

7. REFERENCES

[1] Felder, R.M, Brent, R. 2001. Effective strategis for cooperative learning. J. Cooperation & Collaboration in College Techning. 10(2). 69-75.

[2] Lourens, A.S., Snyders, P.J., Murray, A.T., Joubert, J.P. & Hempel, J. 2015.'A cooperative learning approach to teaching engineering and complementary skills'. Poster presentation: South African Society for Engineering Education (SASEE), Third Biennal Conference: Retaining the Best and Brightest in Engineering Education and Profession. 4 June-5 June 2015, School of Engineering, University of KwaZulu Natal, Durban, South Africa.

[3] Lourens, A., Hempel, J., Joubert, C., Snyders, J., Murray, A. & Dolley-Reynevelt, M. 2015. 'Incorporating individual learning styles of engineering students for effective teaching of manufacturing engineering techniques', *Unpublished*.

[4] Froyd, J.E., Wankat, P.C. & Smith, A.K. 2012. 'Five major shifts in 100 years of engineering education', *Proceedings of the IEEE*, 100.

[5] Kuri, N.P. 'Kolb's learning cycle: an alternative strategy for engineering education', *Imeerweb.osanat.cz*. Accessed 26 Aug.14

[6] Lenschow, R.J. 1998. 'From teaching to learning: a paradigm shift in engineering education and lifelong learning', *European Journal of Engineering Education*, 23(2).

[7] Engineering Council of South Africa: Standards and procedures system. (2012). Qualification Standard for Bachelor of Engineering Technology: NQF Level 7. Available: www.ecsa.co.za /E-02-PT_Rev1. Sighted 11 November 2015.

[8] Ford, M. & Morice, J. 2003. 'How fair are team assignments?', *Journal of Information Technology*, 2 pp 367-378.

[9] Almond, R.J. 2009 'Team assessment: comparing teams and individual undergraduate module marks', Assessment and evaluation in Higher Education, 34(2) pp 141-148.



[10] Johnson, D.W. & Johnson, F.P. 2005. Joining together: Team theory and team skills, 9th Edition, Pearson.

[11] Burdett, J. & Hastie, B. 2009. 'Predicting satisfaction with team work assignments', *Journal of University Teachinf and Learning Practice*, 6(1) pp 61-71.

[12] Candy, P.C., Crebert, G. & O'Leary, J. 1994. Developing lifelong learners through undergraduate education, Australian Government Publishing Services.



Development and implementation of an advanced mobile data collection system

I.M. Prinsloo^{1*} & J.N. du Plessis² & J.C. Vosloo³

¹North-West University, CRCED-Pretoria, Pretoria, South Africa nprinsloo@researchtoolbox.com

² North-West University, CRCED-Pretoria, Pretoria, South Africa jduplessis@researchtoolbox.com

³ North-West University, CRCED-Pretoria, Pretoria, South Africa 12317845@nwu.ac.za

ABSTRACT

This paper describes the development and implementation of a generic data collection tool. The Mobile Information Collection and Verification System (MICAVS) provides users with a generic platform to create application specific data collection questionnaires. During the data collection phase, the system uses an Android^M application to display relevant questionnaires and interfaces. The application thus utilises modern smartphone technology to aid data collection, ensure data traceability, perform tests on said data and guide users. After collection, the collected data are exported to a range of reporting systems. Case studies of successful system implementations in numerous industries are also discussed.

This work was sponsored by HVAC International

¹ Mr. I.M. Prinsloo holds a Masters degree in computer and electronic engineering and is currently enrolled for a PhD at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.

² Dr. J.N. du Plessis is a registered professional engineer and holds a PhD in computer/electronic engineering from the North-West University. He is currently a post-doctoral student at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.

³ Dr. J.C. Vosloo is a registered professional engineer and holds a PhD in electrical engineering from the North-West University. He is currently a lecturer at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.



1. INTRODUCTION

1.1 Challenges within South Africa's industrial sector

Many large industries in South Africa rely on electricity as primary energy source. Historically South Africa enjoyed low electricity tariffs, which led to a high reliance on electricity. However, by 2016 the cost of electricity has risen to 527% of what it used to be in 2006 [1], [2]. This caused a sudden increase in operational expenses and has a significant impact on the profitability of large electricity consumers. Industries affected by this increase include, but are not limited to: cement manufacturing, steel producing, water utility and mining.

Companies within these industries stand to gain significant benefits by improving energy efficiency. Energy service companies (ESCOs) are therefore often contracted to assist these companies with energy related projects. Typical services offered by these ESCOs during energy interventions include energy investigations, funding arrangement, infrastructure upgrades and project management. Selected ESCOs also offer maintenance programs, in order to ensure that system performance is sustained for an extended period of time.

Meetings with ESCO clients in various industries revealed the need for a cost effective mobile data collection system. These clients described their data requirements and unique challenges they face with collecting relevant data. Details surrounding these challenges are provided in the following section.

1.2 Mining industry challenges

One of the primary industries that ESCOs focus on is the mining industry. South Africa has some of the deepest gold and platinum mines in the world, including the world's deepest gold mine, with a current depth exceeding 3800m [3]. Mining at these depths presents many unique challenges with regard to service provision and maintaining of safe working conditions. These services require advanced systems that increase energy consumption and associated expenses.

Growing operational expenses reduce the profitability of mining activities. Furthermore, the increased operational expenses coupled with declining commodity prices and ore grades, raise the need for mines to investigate cost effective methods to reduce expenditure per ton of product delivered. This creates the demand for energy efficiency improvement and ESCO services. Reduced energy consumption will result in reduced energy expenses and will produce significant drop in overall operational expenses.

Studies have shown that proper maintenance offers great efficiency improvements that offer cost savings in return. However, ESCO projects and funding initiatives require large amounts of data. Infrastructure expansion to gather data is often expensive and do not guarantee a return on investment. Therefore the developed system will provide mining companies with a cost effective data collection tool.

1.3 Water industry concerns

In addition to escalating electricity prices, large industries are placed under pressure by several resource constraints. One limited resource is water. Water forms a crucial part of operations in many industries. South Africa is currently in the early stages of a water crisis. Some parts of the country are already burdened with an inadequate water supply [4].

Dr. J. Dabrowski [4] stated that over 98% of South Africa's available water resources have already been allocated across various sectors. Furthermore, estimations indicate that by 2025 the country will have a water deficit of between 2% and 13% based on expansion and economic performance. This is a major concern for the country and creates the need to implement proactive water management strategies.

Water suppliers rely on large pump sets and piped networks to transport the water. Baird [5] states that, a failure in such a piped network will have significant financial, environmental and social impacts. Brent and Haffejee [6] further expand on this by raising the concern that not only the water supplier will be affected, but also the consumers within the service area.

In addition to water loss concerns, the large pump sets used in the industry are high electricity consumers, which relate to high electricity expenses [7]. On average 657 kWh electricity is consumed per Mega litre of water supplied [8]. Due to the heavy reliance on electrical pumps, water distribution can be considered an energy intensive industry. Some ESCOs have thus started offering their resource management strategies to large water networks.

A reduction in system losses offers the supplier many financial advantages by reducing electricity and water treatment expenses. Additionally, the environmental impact of operations is reduced and precious water resources are preserved. These advantages motivate the need for improved maintenance structures and



accurate maintenance information. Energy efficiency and maintenance projects on water systems require the collection of large amounts of data and therefore supports the demand for a mobile data collection system.

1.4 Maintenance data management

ESCO services typically include projects targeted at reducing energy consumption. These projects typically include infrastructure upgrades and improved control systems. In addition to this, ESCOs provide performance reports to the end user to ensure that the intervention operates as intended.

It was found that proper maintenance helped achieve target energy savings [9]. Furthermore, reducing wastages in pumped water schemes and on compressed air networks offer great energy saving opportunities. Complete and accurate data is however required to construct finding reports and identify system inefficiencies. Thus, the maintenance personnel responsible for these systems need tools to gather and manage maintenance data.

Maintenance staff needs effective methods to manage maintenance data. However due to economic constraints industrial users may also have difficulty funding these tools. A cost effective data collection system will therefore help industrial users to collect the data used to compile finding reports that will aid management to develop better maintenance structures.

1.5 Paper contents

The challenges described in the preceding sections indicate that clients in a range of industries share similar data requirements. Therefore, the data collection needs for a range of industry applications can be addressed by a single generic data collection system. A novel data collection system was consequently developed to satisfy the requirements described by ESCO clients. The following sections describe how this need was addressed through the investigation, development and implementation of a new data collection system. The following topics are addressed in ensuing sections of the paper:

- Evaluation of existing data collection methods;
- Design considerations for a mobile data collection system;
- Data validation structures;
- User guidance functionality;
- Case studies in the South African industrial sector;
- Concluding findings and comments.

2. EVALUATION OF DATA COLLECTION METHODS

2.1 Data collection systems

Digital data capture enables the collection of metadata, which is used to authenticate data, without placing additional responsibility on users. Computerised systems have the ability reduce data faults caused by human error. Furthermore, computerised systems offer users a range of features which ensures complete data sets, as well as validation data. Data collection systems exists, but present many shortfalls or restrictions. In this section three data collection methods are considered. These methods are: Supervisory Control and Data Acquisition (SCADA) systems, manual data recording and mobile applications.

2.2 SCADA systems

SCADA systems are the preferred method used for system control and data collection since these systems reduce the required amount of human intervention. Furthermore, these systems are utilised to achieve maximum cost saving potential offered by energy efficiency interventions by ensuring reliable system operation [10]. SCADA systems typically rely on PLCs for control and data acquisition at remote locations in order to ensure reliable operation. A Programmable Logic Controller (PLC) is a computational device and is used for industrial control and data collection in industrial applications [11].

Unfortunately the cost of implementing a PLC system is very high. In 2006 D. le Roux [12] estimated the implementation cost of a PLC system between R250 000 and R800 000. PLC installation costs are further escalated by software development and commissioning requirements [12]. The range of hardware components that form a complete PLC system includes:

- cabling for both power and communication purposes;
- junction boxes;
- a uninterrupted power supply (UPS);
- a human machine interface (HMI);
- circuit breakers.

Industrial applications, particularly mining, also require the use of high quality cables and instrumentation to withstand the extreme environmental conditions. These communication networks often span across large



areas. This creates the need for very long cables and further increases the network cost. An additional downside of a SCADA system is that the system only monitors a rigid collection of variables. System expansions are required to add new sensors and add them in the system. These expansions are costly and also serve only targeted purposes. Additionally, these expansions have integration problems due to different suppliers and systems.

2.3 Mobile applications

The development of handheld computing devices such as smartphones and personal digital assistants (PDAs) provide new data collection opportunities. Many applications have therefore been developed to collect data. Each of the listed applications presents a unique combination of features, as well as its own list of shortfalls. Therefore, none of the other applications satisfied the combined set of requirements that this study is based on. Table 1 provides a summary of the existing applications and their features.

Table 1: Data collection application comparison

		Momento database [13]	Momento [14]	Goformz [15]	Nexticy [16]	Formetize [17]	Pushforms [18]	Devicemagic [19]	Magpi [20]	Doforms [21]	Flowfinity [22]	Epicollect plus [23]	Gis_cloud [24]	Poi_mapper [25]	Prontoforms [26]	Keel [27]
	Authorised user accounts		٠			•	•	٠	•	•	٠		٠	•	•	٠
_	Authorised user linking				•	•		•	•	•	٠			•	•	٠
ion	Data structure versioning													•	•	
ırat	Auto update user configuration								•		٠			•	•	
figu	Auto update data structure configuration	•		•	•	•	•	•	•	•	•		•	•	•	
Con	Offline configuration management	•												•		
	Kiosk mode device limitation					٠										
	Licencing and authorised access to group data					•				•	٠			•	•	
	Customisable input linking	•		•		•	٠	•	•	•	٠	٠	•	•	•	٠
on	Code scanning	•				•	•	•	•	•	•	•	•		•	٠
ecti	Photograph with annotations					•		•		•	٠				•	
coll	Document scanner															
ita (Signature capture	•		•	•	•	•	•		•	•				•	٠
Da	Sketch pad			•		•		•		•	٠				•	٠
	Sound level measurement													•	•	
	Historic data access on device	•	•								•		•	•		
on	Geo location capture	•	•	•		•	•	•	•	•	•	•	•	•	•	٠
cati	On device calculations	•		•	•	•	•		•	•	•		•	•	•	
rifi	Customisable data validation tests	•					•				•	•		•		٠
Ve	Validate according to historic data										•					
	Customisable summary		٠	٠											•	
	User guidance on input form	•	•			•	•			•						
JCe	Automated data set detection															٠
staı	Workflow support via conditional sub forms	•		•	•	•			•	•	•	•	•	•	•	
issi	Manual task assignment	•	•	•		•	•	•	•	•	٠				•	٠
er a	Customisable map overlays												•	•		
Us	Review previous readings offline	•	•								•					
	Customisable validation messages					•										
e	Customisable report integration	•	٠	•		٠			•		٠		٠	•	•	•
Jata	Data encryption	•	•			•		•	•		٠				•	•
	External system integration	•	٠			•	•	•	•	•	٠		•		•	•



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

2.4 Manual human recordings

Manual data recording is practiced in many industries. This method of data collection allows great amounts of flexibility and does not require expensive infrastructure upgrades. In some applications users are guided by forms they have to complete. However, in other cases employees do not have formal data collection structures.

Therefore, although this method is quick and easy to implement, there are significant shortfalls. The largest shortfall is inaccuracy due to human error, since during this process humans interact with data on various levels and each point of interaction creates a possible fault from human error. Possible errors can be made while: reading the meter, writing the meter reading down, reading handwritten notes, typing values into a computer and creating custom reports. It is thus clear that manual data collection has numerous human interactions and that each interaction can be considered an integrity risk.

Clients who made use of manual data collection identified additional system flaws during meetings. During manual data collection exercises employees often visit harsh environments and there are many difficulties using a pen and paper in these non-ideal conditions. These difficulties include insufficient paper support, stationary failure and harsh environmental conditions.

3. DESIGN CONSIDERATIONS FOR A MOBILE DATA COLLECTION SYSTEM

3.1 Introduction to a generic data collection system

Due to the shortfalls of the traditional data collection systems, a new system was developed. The system is designed to provide a generic data collection platform that is usable across multiple industries and applications. The primary component of the Mobile Information Collection and Verification System (MICAVS) is therefore a mobile application.

The application provides infrastructure that allows users to create custom data collection sets. These data sets can then be customised from within the application and exported to other devices without the need for further development. During configuration ingress activities are linked to items within the data set and stored in the database. During execution, the application dynamically creates user interfaces based on database entries.

When data sets are created, the set of collection variables can be personalised to address the user's specific needs. A range of data ingress activities have been created to allow the collection of various data types. During configuration, a specific data ingress activity is linked to each item in the data set. This aids users to enter appropriate data entries for each data field that they defined. Data fields can be marked as required to ensure that records are complete.

3.2 Operating system selection

The core of the MICAVS relies on an Android application which is used to create and manage log entries. Android was chosen as the best suited operating platform for the system due to the following factors. The principal factor was that Android has significant market penetration in South Africa, consequently many consumers already own Android devices [28]. These users are thus able to make use of the application without investing in new hardware.

Many ruggedized mobile devices are also commercially available. These devices are better suited for harsh operation conditions. Rugged devices can withstand risks including drops, as well as dust and water exposure factors better than standard mobile devices. Furthermore, selected rugged devices are certified for use in potentially explosive areas.

Most of these commercially available ruggedized devices employ either the Android or Windows mobile operating systems. No Apple® devices are available in ruggedized packages and are therefore not suitable for this application. Device availability and market penetration were thus considered and revealed that Android is the preferred platform for the MICAVS application.

3.3 Database

Android supports SQLite databases and therefore, the developed application utilises a SQLite database to manage data. Without root access, the database is isolated by Android so that only the associated application can access and modify the database. MICAVS uses the relational database to provide structure to the application. The database can thus be summarised in three categories namely: constants values, configurations settings and recorded data.



Constants are defined during compile time and cannot be modified on the device. These values are used to provide structured options and default values. Although these entries generate fixed outputs that offer users access to structured options, it offers developers a platform to quickly access and update crucial application options.

Configuration table values are used to manage variable interface options and predefined lists. These values can be configured on the device or using a configuration tool. During execution the application uses these tables to dynamically construct interface elements and populate lists as defined by the administrator.

Lastly data tables are used to manage recorded data. The database structure allows users to manipulate input options, by adapting the configuration tables. These changes affect the storage tables that accommodate unique configurations and thus stores data accordingly.

3.4 Interfaces

One of the key elements of the MICAVS is customisable user interfaces. Various interface elements have therefore been created. These elements include layouts for interfaces such as the main menu and sub-menus. The next and arguably most important element is the display cards. Lastly, various data ingress interfaces with fixed layouts were also created.

Interfaces are constructed dynamically from the database when the application is executed. Menu cards items including images, headings and support text are constructed from database entries. A collection of user access tables contains links between user access options and menu items. Therefore, only privileged users have access to specialised menu options such as the management interface. The MICAVS main menu with privileged user access is shown in Figure 1-A.

When creating a new log the application detects the appropriate mode and then compiles a list of data fields assigned to that particular mode. Card elements are then created based on the compiled list. Figure 1-B, shows the data input interface with a card for each data set entry. These cards contain links to the appointed data ingress interfaces. To add data the user must press on the appropriate card and use the data ingress interface to enter the relevant data. Figure 1-C, shows the numeric data ingress dialog.



Figure 1: MICAVS customised interfaces

After all the required data fields are completed the user is allowed to finish the log. A log summary is then created by selecting *Finish* from the data input menu. This summary contains the logged data along with additional data including GPS location, timestamp and verification test results. Some data fields also have support interfaces that allow users to view the captured data. One example is the map view that shows a map with a pin where the log was created.

Test results displayed in the summary are created dynamically. If no tests are linked to the data field, a normal summary card will display only the captured data. Conversely, if tests are linked to a specific data



field a detailed summary card will be displayed. These tests are discussed in more detail in section 4. The user is also notified if the captured value exceeds the test parameters.

Figure 2 shows examples of review interfaces in the MICAVS application. Figure 2-A shows an example of a summary interface. In this case an average test was executed on *Meter Reading*. The field is emphasised by displaying a highlighted card item. In this case the test failed and an exclamation is displayed. A failure message can be seen by selecting the *Meter Reading* card. Figure 2-B, shows an example of the customisable failure message dialog.

Certain selected data ingress types have support interfaces. For instance, *Location* entries can be viewed on a map. Figure 2-C shows an example of the reading location as a pin on a map. The map type can also be modified to show either a satellite image, normal map, hybrid map or terrain view. Images can also be reviewed in a similar interface. These support interfaces are invoked by selecting the grey buttons on the right hand side of the interface shown in Figure 2-A.



Figure 2: MICAVS review interfaces

3.5 Data transfer

The ability to transfer data is required in order to extract the full potential of the data collection system. Existing log entries can be filtered and viewed on the device. To allow customisable reports, the MICAVS exports data to remote servers in order to process the collected data. The MICAVS has been integrated with both internal and third party data processing systems to satisfy a range of user requirements.

The MICAVS was initially developed for an ESCO that provides various energy services to clients across multiple industries. To enable this, many internal systems were developed to receive, process, store and report on data collected on site. The MICAVS was therefore integrated with internal systems. The application sends structured data files to these systems via an internet connection both automatically and upon user request.

The integration enables the internal ESCO systems to receive data from the MICAVS application. The received values are then archived along with supporting data for traceability purposes. Furthermore, this integration allows the construction of custom reports and provides a platform to perform additional tests on data based on the client's needs.

A specific ESCO client also has an internal system used to track maintenance requests. The developer therefore collaborated with a team from the client in order to develop a data transfer strategy between the two systems. This strategy was conceived and implemented successfully. This allows the MICAVS application to communicate directly with the client system and enables automatic maintenance work order creations without the need for additional human intervention.



3.6 Data verification

The MICAVS provides users with a unique data verification platform. This platform provides clients with the ability to perform different data verification tests at three different stages of the data collection process. Firstly, there is a generic data validation component in the application. This component allows users to select one of multiple verification tests and apply it on a data field of their choosing. Unique test parameters can then be defined for each test. The application also has the ability to select historic data entries connected to the data field in question and compare the new entries with said historic data.

During the second verification stage, recorded data that are uploaded to the server are saved alongside metadata such as pictures that prove certain data elements. This stored data can then be used for verification for audit purposes by requesting raw data from the ESCO.

Lastly, the exported data are received by a reporting system which generates reports which can be customised according to the client's needs. This allows users to manually verify the data against previous collections or predefined values. A Microsoft® Excel template also allows users to perform advanced mathematical calculations on the dataset and use these values in automated reports.

3.7 User guidance

One of the unique functions of the MICAVS is user guidance. When the researcher joined ESCO clients during various field tests, the client made use of a note sheet that shows previous reading. New readings were compared with previous readings in order to verify that readings fell within acceptable limits.

After viewing these tests, the application was expanded to include this functionality. Users now have the ability to view previous readings while making new log entries. Users can therefore perform a manual verification that the values fall within an acceptable range.

When the user selects *Finish*, a log summary is also generated and the MICAVS performs linked tests on the data. A warning flag is displayed if any of the tests exceed the specified parameters. The user can then select the summary item to see a detailed description of the test result and verify that the entered values are correct before saving the log entry.

4. SYSTEM IMPLEMENTATION

The MICAVS system was implemented at three client sites. Client 1 is a large water supplier in South Africa and used the system to manage maintenance data. The second client is in the mining industry and used the application to manage water meter readings. Lastly, the application was used by ESCO personnel to perform an audit on a compressed air system in a goldmine.

4.1 Implementation 1: Water network maintenance

A large water board in South Arica relies on an extensive pipeline network to transport water to clients. The majority of this pipeline network is underground and totals to a combined length of 3600km [29]. These pipelines are subject to corrosion and other factors that decrease its life expectancy. According to the American Water Works Association similar pipelines have a typical lifespan of 70 years [5]. The network in question can therefore be considered mature which increases the risk of failure.

Due to the size of the network it is impractical to replace large sections of pipelines that do not present immediate risks. However, it is difficult to predict when and where pipelines will fail. According to Kahn [29], a pumped water loss between 4% and 5% was recorded for the network. This equates to 210 million litres of treated water that is lost on a daily basis. Furthermore, Kahn [29] raises the concern that the exact location of these losses are unknown, due to the fact that pipeline leaks are not accurately recorded.

These losses are a serious concern due to limited water resources in South Africa. Furthermore, these losses have significant impacts on other systems and the environment. Most of the lost water is potable water. It can therefore be deducted that 210 million litres of water received treatment without cause. The client therefore had the need for advanced data collection to record pipeline failures within their network.

Previously they relied on a manual system whereby the general public would call and report water leaks. A maintenance representative from the district was then sent out to investigate the report. If the leak was found on their network the investigator would report the fault and open a work order request.

The client had two powerful software systems, but did not utilise their potential as loads of manual labour was required to open work requests. In addition to this, according to maintenance personnel leak descriptions were not accurate. Furthermore, the collected data was inconsistent and varied according to each responder's



personal preference. Reporting structures did not support structures used to collect Global Positioning System (GPS) data or photographic evidence.

The first of the software systems was used to manage maintenance requests. After a maintenance issue was created the system generated a work order and assigned a responsible person. The responsible person then had to find the leak with limited information available and perform the required maintenance with limited prior details of what the scope of work would entail.

The second system was used to manage Geographic Information System (GIS) data. Data stored in this system has the ability to show regions where frequent maintenance is required. Unfortunately this system was not utilized, because responders did not have access to a tool that can capture GPS locations and photographs and support data. Furthermore, no automated structures were in place to process inspection findings.

When the MICAVS was provided to the client, special data import functionality was added to the MICAVS to enable communication with the client's system. Additionally, a custom data set was defined for the client. The data set formed an inspection checklist which contains user specified inputs in combination with photographic evidence, GPS locations.

After initial implementation the client distributed the MICAVS system to maintenance representatives in each operation region. The regional representatives use Android devices equipped with the MICAVS application to create detailed issue logs. These logs are then uploaded to the client servers where they are processed. The client servers receive the log data and automatically open work orders. Detailed descriptions thus provide maintenance staff with a better indication of work required.

In addition to this the GIS system could be utilised for marking locations where leaks occurred. High risk areas can therefore now be identified based on geographical similarities. This enables maintenance management to identify problem pipelines and sections and replace only limited sections of the pipe network. This improved information structure therefore enables the client to make more informed maintenance decisions and reduce expenditure.

4.2 Implementation 2: Water meter readings

In this case the client was responsible for gathering water meter reading data. To do this the client traditionally relied on manual pen and paper methods and would physically visit a predetermined list of sites and write down readings. Most of these locations were outdoors; therefore weather conditions had a serious impact on operations.

The client was supplied with the MICAVS application and a unique setup was created according to the user's needs. After the system was implemented trial tests were performed and data verification tests were developed and the system was updated with an improved configuration. Part of this improved configuration included adding additional users. Due to security concerns, selected meter readings had to be assigned to personnel who could take the readings while escorted by security personnel.

The MICAVS system is capable of detecting specific components and user modes based on predefined identification codes. These codes can be in the form of barcodes or QR codes and must be unique. The client therefore obtained barcode tags and assigned the codes to specific meters. By utilising this functionality the possibility of selecting the wrong meter was removed.

Currently meter readings are recorded on a monthly basis. The application allows the user to access previous readings during data collection inspections. A detailed summary after every log contains on device calculations that indicate possible data errors and alert the user of any discrepancies while still on site. The users record data on site and then synchronises the collected data with the ESCO's server. Next the data is backed up for traceability purposes and custom reports are generated according to user specifications.

4.3 Implementation 3: Compressed air network audit

As previously mentioned South African gold mines often operate at depths exceeding 3800m [3] which presents many unique challenges. Specialist systems have therefore been developed to allow operation in these conditions. To illustrate this, consider the compressed air network used in mines. Compressed air generation can make up between 20% and 50% of a mine's electricity consumption [30], [15]. Botha [31] continues that compressed air generation expensive and one of the most inefficient means of energy distribution in mining operations.

Part of the ESCO services includes energy audits on client premises during which ESCO personnel identify inefficiencies and system faults. These results need to be recorded accurately in order to be used to construct finding reports. Potential saving opportunities or maintenance requirements can typically be addressed if these reports are used.



In this case the ESCO used the MICAVS system to perform an audit on a mine compressed air network. Recordings were made if any leaks or misuse of compressed air systems were observed. During this investigation 18 significant air leaks were noted on a single level within the mine.

These leaks have a major impact on system performance and induce significant cost implications since the compressed air networks rely on large electrical compressor systems. If leaks are managed properly less supply from the compressors is needed and electricity costs can be reduced.

5. CONCLUSION

This paper presented a generic data collection and verification system that was developed and implemented as part of a study. The developed system relies on an Android application for data collection and integrates with various support systems. The integrated systems offer users a unique platform to create custom data gathering structures.

The data requirements for the system were therefore obtained through a combination of discussions with clients and previous ESCO experience. This prompted the development of a new data collection system. The new system allows users to create multiple custom data collection interfaces accompanied by verification and data tractability functions.

The collected data is captured using a fully customisable Android application. The application offers user guidance and data verification tools to assist users to collect accurate data. After collection the data are synchronised with support systems which are used to store the data, generate custom reports and perform additional verification tests.

The system was successfully implemented on three separate industrial sites in South African. The users reported great advantages from using the system. Custom reports have also been generated, based on the client's specific needs. These reports aid clients to make better informed decisions. Additionally, recorded data and metadata are backed up and are available for further processing and audit purposes upon request.

REFERENCES

- [1] **ESKOM** 2006 'Tariffs and Charges', no. Megawatt Park, Maxwell Drive, Sunninghill, Sandton.
- [2] **ESKOM** 2016 'Tariffs and Charges', no. Megawatt Park, Maxwell Drive, Sunninghill, Sandton.
- [3] Manzi, M. 2014 '3D Seismic Imaging of the Ghost-Carbon Leader Reef of the World's Deepest Gold Mine -Mponeng Gold Mine, South Africa'.
- [4] Daws, D. 2016 'Scientists say there is no debate that SA is experiencing a water crisis' Engineering News. pp. 1-3.
- [5] Baird, G. M. 2011 'Money matters--the epidemic of corrosion: Part 1, Examining pipe life' J. Am. Water Work. Assoc., vol. 103, no. 12, pp. 14, 16-17, 19-21.
- [6] Haffejee, M. and Brent, A.C. 2008 'Evaluation of an integrated asset life-cycle management (ALCM) model and assessment of practices in the water utility sector', Water SA, vol. 34, no. 2, pp. 285-290.
- [7] Rand Water 2013 'Rand Water Integrated Annual Report 2013-2014', no. November. p. 33.
- [8] Rand Water 2010, 'Rand Water Annual Report 2009-2010', p. 24.
- [9] Van Tonder, A.J.M. 2011 'Sustaining compressed air DSM project savings using an air leakage management system', 2011 Proc. 8th Conf. Ind. Commer. Use Energy, no. November, pp. 133-137.
- [10] Osareh, A. R. and Pan, J. and Rahman, S. 1996 'An efficient approach to identify and integrate demand-side management on electric utility generation planning', Electr. Power Syst. Res., vol. 36, no. 1, pp. 3-11.
- [11] Yilmaz, E. and Katrancioglu, S. 2011 'Designing Programmable Logic Controller (PLC) Experiment Set with Internal Experiment Blocks', Procedia Soc. Behav. Sci., vol. 28, pp. 494-498.
- [12] Le Roux, D. 2006 'A new approach to ensure successful implementation and sustainable DSM in RSA mines', Ph.D thesis, North-West University, Potchefstroom, South Africa.
- [13] Mementodatabase.com, 2016. [Online]. Available: http://mementodatabase.com/. [Accessed: 05- Aug-2016].
- [14] Momento.com, 2016. [Online]. Available: http://momentoapp.com/features/. [Accessed: 15- Aug-2016].
- [15] Goformz.com, 2016. [Online]. Available: https://www.goformz.com/features. [Accessed: 05- Aug-2016].
- [16] Nexticy.com, 2016. [Online]. Available: https://nexticy.com/about. [Accessed: 05- Aug- 2016].
- [17] Formitize.com, 2016. [Online]. Available: http://formitize.com/en/features/full-feature-list. [Accessed: 06- Aug- 2016].
- [18] Getpushforms.com, 2016. [Online]. Available: http://www.getpushforms.com/#features. [Accessed: 04- Aug- 2016].



- [19] **DeviceMagic.com**, 2016. [Online]. Available: http://www.devicemagic.com/features. [Accessed: 06-Aug-2016].
- [20] Support.magpi.com, 2016. [Online]. Available: http://support.magpi.com/support/home. [Accessed: 15- Aug- 2016].
- [21] **DoForms.com**, 2016. [Online]. Available: 2016. [Online]. Available: http://www.doforms.com/wpcontent/uploads/2016/05/doForms-Overview.pdf. [Accessed: 05- Aug- 2016]. [Accessed: 04- Aug- 2016].
- [22] Flowfinity.com, 2016. [Online]. Available: http://www.flowfinity.com/kb/. [Accessed: 06- Aug- 2016].
 [23] Epicollect.net, 2016. [Online]. Available: http://www.epicollect.net/start.html. [Accessed: 05- Aug-
- 2016]. [24] Ciscland com 2016 [Online] Available: http://www.epicollect.net/statt.ntnk. [Accessed: 05" Aug-
- [24] Giscloud.com, 2016. [Online]. Available: http://www.giscloud.com/apps/mobile-data-collection. [Accessed: 06- Aug- 2016].
- [25] **PoiMapper.com**, 2016. [Online]. Available: http://www.poimapper.com/benefits/. [Accessed: 05- Aug-2016].
- [26] **Prontoforms.com**, 2016. [Online]. Available: https://www.prontoforms.com/product/mobile-form-app. [Accessed: 05- Aug- 2016].
- [27] Keel Industrial Mobility, 2016. [Online]. Available: http://mobility.keelsolution.com/data-collectorapp/. [Accessed: 05- Aug- 2016].
- [28] **Potgieter, A.** 2015 'The mobile application preferences of undergraduate university students: A longitudinal study', SA J. Inf. Manag., vol. 17, no. 1, p. 6 pages.
- [29] Khan, F. 2015 'Capturing critical pipeline failure data for optimal maintenance management of a water supply network: a rand water proposition'.
- [30] Howells, M. I. 2006 'The targeting of industrial energy audits for DSM planning', J. Energy South. Africa, vol. 17, no. 1, pp. 58-67.
- [31] Botha, M. 2011 'The state of the power system', Energize, pp. 6-8.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



ENGINEERING THE FACE OF THE FUTURE FACTORY

J. Durbach¹, D. Hartmann² & T.S. Hattingh³

School of Mechanical, Industrial and Aeronautical Engineering University of the Witwatersrand, Johannesburg. <u>1jarred.durbach@wits.ac.za</u>, <u>2dieter.hartmann@wits.ac.za</u>, <u>3teresa.hattingh@wits.ac.za</u>

ABSTRACT

People are the next critical link in world class manufacturing facilities within South Africa. There needs to be a complete paradigm shift in terms of how we train our engineers who will design and manage these facilities as well as how we upskill operators and technicians within the facilities. This combined effort needs to focus on an active learning environment centred around problem solving and continuous improvement. This paper describes a very unique partnership between a university and industry partner which aims to revolutionise the way in which undergraduate engineers are trained and operators are developed. This paper will discuss the structure of this exciting new programme, the anticipated benefits and challenges experienced during its first year.



1. INTRODUCTION

The training of excellent engineers always has been, and remains, a partnership between industry and academic institutions

Academia without industry is paralysed. Industry without academia is blind.

It is often mentioned that graduate engineers have little practical experience, particularly of working in a factory environment. This is largely a consequence of engineering graduates seeking "clean" office jobs; with those organisations that expose graduates to real operating environments tending to experience high turnover and difficulty in recruiting engineers. As such graduates are rarely exposed to the shop floor environment, and certainly not on a sustained basis, at the very least because the cost to company of a graduate does not justify long term deployment to the shop floor.

Out of this emerge graduates who have little understanding of the practicalities associated with the operating environment, including low levels of familiarity with plant, and also the human factors that affect working environments.

To overcome the post graduate stigma associated with very valuable practical experience, Wits¹ entered into an innovative internship scheme with Unilever's newly commissioned Homecare Liquids factory, Khanyisa in Boksburg.

This partnership was set up as a long-term strategic programme to develop young engineering leaders in South Africa with first hand operational experience and excellent academic education. Simultaneously, this project aimed to function as a medium for transferring best practice to the operators at the factory.

Thirty engineering students were recruited out of their second and third years of study, to spend a full year, working as operators on the line as operators at Khanyisa. During this time, students would be exposed to elements, such as shift work, overtime, repetitive work and interacting with the workforce on the shop floor.

2. LITERATURE SURVEY

The predominant theories that underpin this initiative are the philosophy of a learning organisation supporting continuous improvement as well as the importance of problem-based learning within an engineering curriculum. A learning organisation supports the idea of developing and uplifting the base skills of the workforce while work-integrated learning is used to justify the benefit of the programme to students. These theories will be unpacked briefly in this literature survey.

2.1 A Learning Organisation

A learning organisation is introduced by Senge [1] as one where people are "continually learning to see the whole together". He believes that the only competitive edge that companies have is their ability to learn faster than their competitors. Ballé et. Al. [2] frame Toyota's famous production system as *The Thinking Production System*, emphasising the importance of thinking people in making organisations competitive. Ballé et al [2] introduce four frames which guide this Thinking Production System. Three of these frames speak directly to the importance of developing and involving people: creating a problem awareness culture, teaching people how to solve problems and developing people. They also discuss the importance of individual and team effort, aligned towards a common goal and constantly learning, challenging and experimenting.

2.2 Problem based learning (PBL) and Work-Integrated learning (WIL)

The concept of problem-based learning (PBL) was first introduced in the field of medical education; proposing that students learn best through immersion in a discipline-based problem [3].

In traditional learning, students are taught the fundamentals and are then given a problem to solve making use of this material. The learning objectives and principles for the problem are given to the students. In this form of learning, students often only learn what is required to pass [4]. In contrast, PBL makes use of a problem in which students are required to learn the fundamentals to find a solution to the problem. By making use of relevant problems, students' retention of learning can also be improved [5].

¹The University of the Witwatersrand (Wits) School of Mechanical, Industrial and Aeronautical Engineering. hereinafter referred to either as "the School" or "the University", as appropriate.



PBL can be team based or individual (self-directed). It has been shown that a team based approach results in better learning [6]. Polanco et al [7] introduce three pillars for the effective implementation of PBL: student questioning, interdisciplinary method and teamwork. Savage et. al. [4] discuss the importance of integrating teamwork and systems- level thinking throughout their PBL projects.

To address the concerns associated with poor student performance in higher education and to improve the attributes of graduates, the Council on Higher Education [8] has recommended a participative and real-world approach to learning. They term this approach, work-integrated learning (WIL). The main driver behind WIL is improving student learning. Other benefits include improved academic performance, motivation to learn, improved teamwork, communication and collaborative skills as well as the development of positive work ethic and values and clearer career clarification [8].

Literature strongly supports the idea of giving students an opportunity to learn in a work-integrated and problem-based environment. There are many thoughts on how best to integrate these and it would be worth exploring how this approach aligns with other approaches in future work. The idea of developing shop floor employees to support continuous improvement is also well supported by literature. This unique approach by incorporating student learning and shop floor learning will lead to results that can enhance thinking in this space.

3. DESIGN OF THE PROGRAMME

The programme was designed as a multiyear agreement between the University and Unilever. Annually thirty students are to be recruited from the second and third year streams within our School (See Footnote 1). The programme has support from a dedicated lecturer and the students enter the internship on a full time basis. During this time, the students, the support lecturer and other staff of the School will be involved in process improvements at the Khanyisa factory. It is intended that these savings will fund the programme.

3.1 Consultation

We consulted widely in designing the programme. This included focus groups with undergraduate students, and discussions with Unilever and University staff.

The focus groups were conducted with forty-five randomly selected students from the School registry. This gave valuable insights into the student needs. The major emerging theme was fear; fear of delaying career entry, fear of not graduating with friends and fear that a "year off" would cause them to become "unfit" as students.

These themes were included in the programme design as far as possible.



3.2 Timelines and approach

The programme is intended to run over five years, with an initial one year pilot to establish baseline feasibility. After five years, it is hoped that the structure will have stabilised and the programme will become self-sustaining in the long term. Figure 1 shows a rough plan of the management elements needed in the various years.

Work Package						of Project		
			1	2	3	4	5	
0 Integrated Process Planning			٠	0	0	0	•	
1 Market Research			٠		0		0	
2 Marketing			٠	٠	٠	0	0	
3 Recruitment			٠	٠	٠	0	0	
4 Mentoring			0	٠	٠	٠	•	
5 Setting up and managing internship projects			٠	٠	0	0	0	
6 Supervision of projects on site				٠	٠	٠	•	
7 Ad-hoc short term projects			0	0	0	0	0	
8 Monitoring of the transfer of problem solving and improvement skills	s from students to	operators		0	٠	٠	•	
9 Ongoing monitoring and improvement of the programme			0	0	٠	٠	0	
10 Provide input on global best practice, trends and expert analysis.			0	0	•	•	•	
	Ke	y						
	0	Refresher Study						
	0	, Minor Activity						
	•	, Maior Activity						

Figure 1: Annual Plan

3.3 Academic programme support.

It was deemed impossible to run this programme without dedicated support, for which the University recruited a staff member tasked with managing and curating the Unilever partnership. This included being the official liaison between the partners, supervising student projects and coordinating the functions shown in Figure 1.

3.4 Funding

Apart from initial seed funding, the cost of the program is derived from savings generated as a result of student projects and other interventions supported by the University. A proportion of these savings is allocated to the University at quarterly review meetings. These funds are used amongst others to fund the position of the Unilever lecturer. The funding model aims ensure cost neutrality, a priority for the sustainability and replicability of the programme.

The Chemical Industries SETA (CHIETA), subsidises the student bursaries, thus assisting with the financial sustainability of the program. Student salaries are funded from the factory's operational budget as students fill required roles in the factory.

3.5 Students

3.5.1 Recruiting

Recruiting takes place from July in the year preceding the deployment of the students to Unilever. This includes several marketing events and roadshows, leading to one on one mentorship and guidance, which often also involves discussions with parents or existing bursary companies.

3.5.2 Cohort

The group consists of thirty students selected from applicants within the School. Students who have completed third year are preferred; however (with just a year to go) these students are difficult to recruit, so numbers are made up from students who have completed second year.

Returning students retain programme support, and as such by year three, there will be as many as ninety students in the system continuously - the current year of deployed students, and the returned students from the preceding cohorts. For greater clarity, see Figure 2.



	Year 1	Year 2	Year 3	Year 4
Activities				
Preparatory Work				
Recruitment				
Internship				
1 st year post				
4 th Project Prep				
2 nd year post				
Students in System	30	30	60	90
Staffing				
Unilever Lecturer 1		ĺ		
Unilever Lecturer 2				
ohort 1			-	
ohort 2				
ohort 3				

Figure 2: Phasing of people on the programme

3.5.3 Financial Support

During the year of deployment, students on the programme are paid a standard operator's wage. Upon completion of their year's deployment, the student returns to University where the remainder of their anticipated student fees are paid for. Students furthermore continue to enjoy the mentorship of the Unilever lecturer during their remaining years at University.

There is no obligation for the company to employ any of the students upon completion of their degrees, however top performers may be recruited.

3.5.4 Pre Programme

Prior to deployment to the shop floor, the students are sent on a third party work readiness programme, which is designed to bridge the academic programme and the working world. This is a rigorous boot camp, that includes shift work, a heavy load of assignment deadlines, and emotional and self-awareness training.

3.5.5 Working as Operators

Students spend the year working as operators in budgeted roles. During this year, it is a priority that the students integrate into the work teams and the workforce and are not separate in any way. Additional programme activities are designed as "after hour" activities, as student absence from the workplace is problematic.

3.5.6 Improvement Projects

Students working in the factory are tasked with identifying, investigating, and implementing improvement projects within their work area and the organisation. While these projects will be facilitated by the University, the expectation is that students will find and execute improvement projects independently.

3.5.7 Formal Learning

Formal learning time, including safety training, standard operating procedures, and technical operator skills will be provided by Unilever.

Recognising the student need to remain "intellectually fresh" the University provides formal learning material, lectures and assignments, to support improvement, and ensure a smooth return to university on completion of the program.

3.6 Benefit

The programme is designed to meet three key objectives; of creating a cohort of future engineering leaders with an awareness of the practical side of Engineering, who will excel at their remaining studies as a result of their exposure, and whilst deployed, transfer their skills to the existing factory workforce, creating an organisation, that in Senge's [1] words *keeps learning together*, which is the face of the future factory.



3.6.1 To Unilever

Unilever may leverage the University, her access to skilled professionals and strong students to assist with process improvement and generate savings. Unilever further gains early access to potential future leaders. Key as part of the relationship is the skills transfer from students to existing operators, essentially upskilling the workforce through these inputs.

3.6.2 To Wits

The University benefits by having students, of whom it is hoped, will graduate faster. An experienced cohort of students may be able to relate abstract academic concepts to a working environment, increasing their chance of success in future studies. It is also hoped that they will act as leaders amongst students, creating a stronger qualification. In addition to this, the funding contributions made through this scheme are of considerable benefit to the institution.

3.6.3 To the students

Problem based learning gives students the ability to relate concepts learned at University into practice and back again. This gives them "hooks to hang ideas on", hopefully enabling them to graduate faster and with deeper more embedded insight. Students additionally benefit considerably from the financial security provided in part by the year of working as well as from the guarantee on study fees in their remaining years of study. Students further learn independence, grit and a structured work ethic.

4. INTERIM OBSERVATIONS AND DISCUSSION

4.1 Educating the future engineers

Poor pass rates and high dropout rates currently plague the higher education system with engineering being highlighted as one of the degree programmes that are worst affected [9]. This shift into higher education is also often compounded by financial and engagement challenges faced by students, but academic factors are seen as being the key to improving student success [9]. The poor success rates at South African Institutions is affected by poor student preparedness specifically related to key learning transitions that are required throughout the degree [9]. The program is designed to address these challenges while simultaneously preparing engineers for the currently unknown face of the future factory.

The primary educational objective for the work experience year is not the instruction of core engineering topics, but rather the provision of ancillary knowledge (such as; people skills) and the context in which engineering knowledge is required. For example, no mechatronics will be taught, however, student exposure to automation and control systems may provide context to link theory and practice once they return to university.

Participation in the programme is not graded. This aims to instil students with accountability to decide how much experience, exposure and benefit they choose to obtain. The students could just work as operators for a year, or they can take full advantage of their access to the organisation and show initiative to add value through improvement projects.

4.1.1 Student placement in the factory

Students are allocated to machine operator roles where tasks are fairly repetitive yet their presence is required to ensure continuous production. Examples of student roles include; packaging product, offloading raw materials onto the line and the operation and monitoring of automated equipment. Students are, for the year, not students but a fundamental part of operations.

The issues that students experience while performing operator roles should provide deeper understanding of work. It is believed that the hard work required by repetitive manual tasks may inspire student engineers to design better systems in their future careers. In addition, this provides a platform for bottom-up continuous improvement.

The operator's role given to each student for their year of work plays a role in how much a student *perceives* they are learning. Students believe the manual, labour intensive roles they are performing are not preparing them for the completion of their degree or a career as an engineer. This is however a critical step in developing the face of the future factory, where future management has not only witnessed but experienced the plight of shop floor work and are therefore better equipped to design, manage and improve the systems of the future; and this, through the experience gained here, in consultation with those doing the work.



4.1.2 Improvement Projects

Practically realising improvement projects within the factory has been a challenge. This can be attributed to three key factors

- 1. Time and Motivation
- 2. Knowledge
- 3. Self-Empowerment

Students are responsible for their jobs as operators. These shift roles take up the work day yet students are also expected to complete improvement projects, expose themselves to other areas of the factory and study in their own time. The adjustment to shift work is a challenge for students who feel fatigued, especially when they are on night shift. Students are also traditionally motivated by marks, and the risk of failing. On the program students have to adopt an internal locus of control, motivated by their own learning and ambition. There are no rewards or incentives for completing improvement projects. Other assignments asked of the students have also been poorly executed or have low submission rates, owing, in part to the non-educational environment the students are in and the absent threat of failure.

Students participating in the program have had limited exposure to engineering application and are at different stages in their degrees. For example, students who have completed 3rd year Industrial Engineering are in a far better position to autonomously scope and execute improvement projects than a student who has only partially completed the second year.

Knowledge however is not as critical a challenge as student's lack of self-empowerment. Students are positioned at the bottom of the organisation's hierarchical structure and do not feel empowered to take action, make suggestions or illicit support from individuals at higher levels. Students, just like other employees, need to be empowered if they are going to make a positive contribution to improving an organisation.

The progress of student projects presents a major opportunity to students' learning, as this is the space in which they can take their experience as operators and apply themselves as engineers and problem solvers to improving that experience.

4.1.3 Formal Learning

This component keeps students academically engaged to ease their return to full time study. Classroom time together also allows students to reflect on workplace challenges and their learning. The formal learning component has experienced challenges in scheduling, given that students are on different shifts. Without the threat of failing, it is difficult to ensure all students engage in lecture material, and the associated assignments. This is compounded by operational demands of the factory which often requires overtime and the full focus of student operators, detracting from the learning objectives of the program.

4.2 Skills transfer to shop floor

The program encourages students to learn from operators, who are now their colleagues, many of whom have more than 10 years of experience in their roles.

The program is working towards skills transfer from students to operators. The first step towards this; is shown in the strong relationships that have already been built between student and traditional operators. Operators claim to enjoy having the students around and appear to appreciate the contribution the students are making.

Apart from the transfer of skills and knowledge, it is expected that students will share their career ambition (privileged to them by being in an operator role for one year and not for life and their future prospects as engineers) with their co-workers. Students make up approximately a third of the operational roles at the factory. The cross pollination between age and education levels is expected to encourage operators to continue learning and believe that they can break the 'glass ceiling' of their roles. In the long term this is expected to enable the shop floor to play an active role in decision making and improvement and allow the organisation to promote from with-in, enabling a bottom-up management approach and creating a learning organisation.

4.3 Measuring success

The primary goals of the program are to prepare engineers (who are more successful at their studies) for industry, simultaneously uplifting the workforce of the organisation. It is difficult to quantify success in these three areas. Moreover, the concept and partnership is a new one and participants still need to complete their year of work experience and return to university and finish their degrees before joining industry. It would be



difficult to quantify programme success without a longitudinal study on participants' careers. We intend to monitor the cohorts whilst completing their degrees. Quantitative analysis of students' academic performance controlled against their peers. Qualitatively, students, independence, grit and ability to cope with academic work will be tracked through interviews and surveys.

The success of the program will also be shown through the overall financial benefit the students and academic partnership bring to the factory through improvement initiatives and shop floor engagement. With significant cost savings it is hoped that it can be proved that: *The right thing to do, is also the profitable thing to do*.

4.4 Summary

The program's three learning components together with the theory that each component relates to is summarised in Table 1. The current benefits and challenges being experienced are also detailed.

Program Related Component Theory		Benefit	Challenges			
Students Working as Operators on 8 hour shits	Learning organisation Work- integrated learning	 Students are a critical part of operations Students 'live' the operator experience Exposure to operational problems Information and knowledge share between students and operators Students develop a deep understanding of process Exposure to working world 	 Repetitive work makes student tired and demotivated Lack of time for learning, reflection and improvement initiatives Students do not make a clear link to engineering or degree Students are used to fill labour gaps, being treated as general workers 			
Continuous Improvement projects	Problem based learning	 Learning continuous improvement Partnership sustainability (add real value) Employee engagement, promoting improvement from shop floor Development of operators 	 Time for projects after shift Lack of incentive Lack knowledge and experience to deliver results Students do not feel empowered to take actions, lacking self- belief 			
Lectures and assessments	Formal learning	Keep students academically engaged Support improvement projects Close knowledge gaps	 Limited time for lectures after shifts Industrial Engineering focused Operation issues take over Poor assignment submission 			

Table 1: Learning Components, Theory, Benefit and Challenges

5. CONCLUSION

We believe that South Africa's future depends on the development and growth of the country's industries. Sustainable and real growth comes not only from great, highly schooled leaders, but also from an engaged, committed, respected and collaborative workforce.

This initiative aims to strengthen both: training good engineers and uplifting the skills and motivation of shop floor workers, making a start towards creating a learning organisation.

While the programme is in its infancy, positive feedback and signs are already evident. This includes the strong relationships between student and traditional operators and some simple improvement projects. Areas for future improvement include improved metrics, and refining the formal learning elements.

In designing the face of the future factory, Wits and Unilever are changing the way in which students become engineers and leaders.



REFERNCES

- [1] Senge, P.M. 2006. The Fifth Discipline: The Art and Practice of the Learning Organisation. Doubleday. New York.
- [2] Ballé, M., Beauvallet, G., Smalley, A., & Sobek, D. 2006. The Thinking Production System. Reflections. 7, pp 1-14.
- [3] Barrows, H.S. 1986. A Taxonomy of Problem Based Learning Methods, *Medical Education*, 20, pp 481-486.
- [4] Savage, R. N., Chen, K. C., & Vanasupa, L. 2007. Integrating project-based learning throughout the undergraduate engineering curriculum. *Materials Engineering*, 1.
- [5] Woods, D.R. 1994. Problem-based Learning: How to gain the most from PBL, Woods Publishing, Waterdown.
- [6] Felder, R., Woods, D., Stuce, J. & Rugarcia, A. 2000. The Future of Engineering Education: Teaching Methods that Work. *Chemical Engineering Education*, 34, pp 26-39.
- [7] Polanco, R., Calderon, P. & Delgado, F. 2004 Effects of a Problem Based Learning Program on Engineering Students' Academic Achievements in a Mexican University. *Innovations in Education and Teaching International*. 41(2), pp 145-155.
- [8] Council on Higher Education. 2011. Work Integrated Learning: Good Practice Guide. HE Monitor No. 12.
- [9] Council on Higher Education. 2013. A proposal for undergraduate curriculum reform in South Africa: The case for a flexible curriculum structure. Pretoria. CHE.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



DEVELOPMENT OF A MODEL FOR ROAD TRANSPORT FUEL MANAGEMENT

M. van der Westhuizen^{1*} & A.J. Hoffman²

School of Electrical, Electronic and Computer Engineering North-West University, Potchefstroom, South Africa alwyn.hoffman@nwu.ac.za

ABSTRACT [1] [2] [3]

Road transport is responsible for 76% of cargo movement in South Africa; at the same time transport cost in Sub-Sahara Africa forms a much higher fraction of the total cost of landed goods compared to the rest of the world. Fuel represents the single biggest operational cost for road transport operators; efforts towards improved fuel efficiency are therefore a priority within this sector. As fuel usage depends on many factors, including engine size, vehicle fabrication, driver behaviour, payload, traffic conditions and route inclinations, it is not a trivial exercise to create accurate consumption benchmarks for a specific operation. This paper investigates various factors that are known to impact fuel utilization with the aim of quantifying the relative importance of the contribution of each. Fuel usage data was collected for a representative set of trucks covering all major routes in South Africa and for various cargo categories over a 3 year period. This data was filtered based on different criteria, including driver identity, route and vehicle model. Comparisons were drawn between consumption figures derived from manually recorded refuel events and figures derived from measurements that are automatically performed by on-board vehicle sensors. It was concluded that driver behaviour and the possible siphoning of fuel from vehicles seem to be a major factor and would justify further actions towards curbing fuel losses. At the same time route inclination, payload and vehicle model also play an important role and should be incorporated into costing models used to determine how different routes and trips are priced.

¹ The author is enrolled for an M Eng (Computer and Electronic) degree in the School of Electrical, Electronic and Computer Engineering, North-West University

² The author is professor at the School of Electrical, Electronic and Computer Engineering, North-West University *Corresponding author



1. INTRODUCTION

Road transport forms a backbone of the African economy, due to the limited availability of rail infrastructure [1]. Road transport is responsible for 76% of cargo movement in South Africa [2]. Compared to the rest of the world the cost of transport in Africa is however much higher as a fraction of the total cost of delivered goods - transport represents as much as 18% of GDP in many African countries, compared to a global average of less than 10%. Fuel cost is the single biggest contributor to the cost of road transport operations, not only due to the high price of diesel but also due to unacceptable levels of fuel theft experienced by transporters: practical figures obtained from industry players indicate that this figure may be as high as 15 to 20% of total fuel consumption if not more [4] [3]. At the same time those factors that cannot be avoided, like payload and route inclinations, must be properly accounted for when costing specific routes and trips to ensure that all routes and customers are serviced on a profitable basis; the alternative would be that some customers are unknowingly subsidizing others. Previous research in the field of fuel consumption modelling identified the primary factors that impact on consumption [5] [6] [7]. In order to restore the competitiveness of the African road transport industry it is therefore essential to find solutions, not only for efficient fuel management, but also for the effective prevention of fuel theft.

Figure 1 below displays a histogram of fuel consumption collected over more than 280,000 trips undertaken within the boundaries of South Africa. It can be seen that the consumption varies quite widely, from around 0.5 km/l up to more than 5 km/l. It is furthermore noticed that the statistical distribution seems to contain more than one mode, which is indicative of more than one independent underlying cause for the observed variations. The first mode centres around 2 km/l, which is generally regarded as the average consumption rate that should be expected for long haul road transport; the other is much higher at around 4 km/k, which seems unlikely to be caused only by normal factors. This provides additional motivation for a study of this nature.



Figure 1 Histogram of fuel consumption over approximately 280,000 vehicle trips

Effective fuel management requires a system solution that can operate at different tiers: tier 1 comprises of sensing capabilities reflecting the current status of the vehicle and its fuel (e.g. GPS coordinates, valve open status, fuel tank level, etc.); tier 2 consists of local intelligence on the vehicle to collect and interpret sensor data and communicate with the home base if necessary (typically in the form of a tracking unit with different communications capabilities); tier 3 will be the management system that collects data from a number of vehicles and provides supervisory functions (e.g. for fuel consumption benchmarking).

In a previous paper we demonstrated that it is possible to deploy a telematics solution for improved road freight transport fuel management, and that with limited savings achieved (typically 1% of current fuel consumption) such a system will generate a positive return on investment, even if it requires the installation of new hardware for that purpose [8]. The obvious question from the perspective of a road freight transport operator is what level of improvement, compared against current fuel costs, can realistically be achieved should additional measures be imposed. The answer to this question will largely depend on the underlying causes for currently excessive consumption, and whether such causes could be eliminated. A variety of underlying reasons contribute towards non-ideal fuel consumption; these include driving practices (e.g. speeding), fuel theft, deviating from prescribed routes, excessive idling at points of loading and offloading, spending significant time in congested traffic and others. The identification of the actual reasons for excessive consumption and the quantification of



the contribution of each is not a trivial problem to solve; even if a telematics system is in place that captures raw data related to fuel consumption the answers are typically hidden within stacks of collected data that still requires detailed interpretation.

The focus of our current research work is to create relevant benchmarks for fuel consumption that can be used by road freight transport operators to evaluate their own fuel performance. To achieve this objective we need to firstly quantify the actual fuel consumption for a sufficiently large number of trips over a representative set of routes within Southern Africa. This data set must include observations that cover all of the suspected primary determinants of fuel consumption as listed above. Secondly we must develop a methodology that can be applied to mine large sets of consumption data in order to identify the primary factors that explain differences in fuel consumption between different trips and to quantify the contribution of each. This will allow the realistically achievable improvements to be quantified and benchmarks to be created at levels that will challenge the status quo but that will still be practically achievable.

In this paper we describe the process that was followed to create a fuel consumption database and to categorize fuel consumption for different scenarios that represent the impact of some of the primary determinants of fuel consumption. We identify the obstacles that were encountered and that complicated the correct interpretation of the data, and describe the approaches and methods that we developed in order to overcome these. We then proceed to illustrate typical results that were obtained in terms of average consumption within specific categories, the difference in consumption between categories and the relative contribution made by each of a number of explanatory variables that impact upon fuel consumption.

The rest of the paper is structured as follows: section 2 details existing literature with relevance to our study, whilst section 3 provides an overview of the process that was followed to collect a representative set of fuel consumption data and section 4 describes the different routes that were covered by the available data set. In section 5 we draw a comparison between manually recorded refuelling figures and the consumption figures as collected by on-board vehicle sensors. We then proceed to investigate the impact of driver behaviour in section 6. Section 7 focuses on the impact of vehicle model on consumption, while section 8 describes a case study that investigates the impact of route inclination and payload. In section 9 we conclude and make recommendations for future research work.

2. LITERATURE STUDY

An investigation into fuel consumption in the commercial vehicle industry will be of value to both the logistics sector (from a cost reduction and performance management perspective) as well as for environmental agencies (from an emissions control perspective). It has been found that road transport is responsible for as much as 93% of carbon emissions in urban areas, while at the same time fuel costs represent between 25 and 40% of overall logistics transport costs [9]. From both perspectives the primary objective is to reduce the amount of fuel that is required to achieve a specific economic objective; in the case of the freight logistics industry this is the transportation of goods from points of production to points of consumption.

In order to pursue the above objective it is necessary to firstly identify the primary determinants of the fuel consumption of commercial freight vehicles (mostly trucks) when engaged in economic activities; the activities of all vehicles in the population must be sufficiently similar to those of other vehicles in the population to allow comparisons and the generation of performance benchmarks. Once all the primary determinants have been identified and their relationships with fuel consumption and emissions have been characterized it will be possible to develop a mathematical model that can be used in scenario analysis to study various possibilities towards improvement of the status quo. Previous research in the field of fuel consumption modelling identified the primary factors that impact on consumption [5] [6] [7]. These include vehicle engine type and size, payload, road inclination, wind strength and several others. In this study we focus on those factors that can be extracted from the available set of data reflecting the trip plans for a fleet of vehicles as well as their GPS tracking coordinates and fuel consumption per trip.

The determinants of fuel consumption can be broadly divided into three categories: vehicle specific attributes (those that determine the fuel efficiency of a specific model), the characteristics of the assignment to be completed (the route distances and inclinations and the times of day of week when goods must be delivered), and driver specific attributes (how effectively the vehicle is controlled by the driver and the accuracy of execution of the assignment by the driver). While each of these categories of factors have been studied separately before we could find no literature where factors from all three categories were studied collectively.

The first set of factors is fairly well understood as explained by [6] and many other research papers that focus on the fuel efficiency of different vehicle makes and models. The second set of factors have been studied from the perspective of characterizing the nature of commercial vehicle movements through the national and provincial road networks [10] [11]. It has furthermore been demonstrated that freight traffic can be simulated under different traffic conditions and policy measures [12]; such results can be combined with a fuel usage model to predict the amount of fuel that vehicles will use under different circumstances.



Other studies [13] derived personal trip data from GPS devices embedded into Smartphones; these techniques are however limited to the characterization of the movements of different sectors in the population without knowing the mode of transport that was used or the impact of behaviour on fuel consumption. While focusing on the establishment of models for vehicle movements in order to support traffic simulation models, the above studies did not address the impact of vehicle activity chains on fuel usage.

The third set of factors relate to the behaviour of truck drivers from a fuel efficiency perspective. Previous studies [8] have shown that within the African context not only inefficient driving styles but also illegal practices like the siphoning of fuel play a large role in determining the overall cost of fuel to the transport operator. While only some of these behaviours, like speeding and excessive idling, directly impact emission level they all add to the overall cost of the logistics operation, hence contributing towards the cost of goods delivered to the consumer [1][4].

3. OVERVIEW OF THE COLLECTION OF FUEL CONSUMPTION AND RELATED DATA

The purpose of our fuel usage data collection exercise was to ensure that we cover all the aspects to be investigated in this study. In order to simplify the process we started off with a fleet of approximately 400 vehicles that covers most of the major routes in South Africa. This would allow us to generate a significant amount of statistics on routes that include widely ranging inclinations (e.g. relatively flat from Johannesburg to Cape Town versus uphill and downhill from Durban to Johannesburg and back where passes over the Drakensberg range have to be crossed). This fleet also includes more than one vehicle make, allowing consumption of different engine designs to be compared.

As the primary long term objective of our research is the creation of a reliable set of fuel consumption performance benchmarks, it is important to collect data in a manner that support direct comparisons between different sets of data. This requires the available data to be subdivided into subsets per major routes that are covered. Different fuel performance levels can be expected to be achieved on different routes based on the number of expected stops, the likelihood of encountering congested traffic and the average incline. As the fleet of vehicles did not cover a defined set of standard routes the major routes had to be derived from the GPS tracking data itself. This is discussed in more detail in section 4 below.

Our initial focus was to reconcile fuel dispensed to vehicles according to refuelling records and fuel actually used by vehicles based on on-board sensing devices. For this reason we collected the refuelling records per truck over a period of one calendar year that covers all seasons, which is important as the road transport industry tends to be cyclic with peaks reached in the months leading up to the Christmas season, and a slump in business early in the new year. As the vehicles involved in the study are not always refuelled once per trip but as needed, we had to reconcile fuel dispensed with fuel used over periods of time than covered more than a specific trip.

The GPS tracking system used by these vehicles collects fuel usage data via the CAN bus system. While fuel usage is measured almost continuously by way of a flow meter, most of the installed units of this system were configured to only store and communicate the aggregate consumption as from when the engine was switched on until it was switched off again; this is done mostly to save on communication costs. The actual fuel usage reports were matched with the fuel dispensed reports by using the vehicle registration numbers that were present as vehicle identifiers in both lists.

It is important to also match trips with drivers, as driver behaviour is one of the primary determinants of effective fuel consumption, both in terms of driving skills and the tendency toward fuel shrinkage. In this case drivers were identified in two ways: in the trips plans drivers should be identified; this field was however missing for 44.8% of recorded trips. In addition drivers have to identify themselves by way of an ID tag that is inserted before a truck is started; this field was available for a larger fraction of trips (76.7%). The missing 23.3% may be ascribed to the fact that driver identification is not enforced on some vehicles, or can be overridden by a manager.

Another important aspect to take into consideration is the impact of route inclination on fuel usage. As could be expected fuel consumption is much higher on inclines compared to flat portions of the route; given the limit on the speed of a truck on downhill stretches this increased consumption level is not all made up on the downhill sections. In order to quantify the impact of inclines integrated over the entire route the inclines were extracted from Google maps by using the route descriptions as defined by the set of GPS coordinates representing each route [9]. While there may be other sources of inclination data the inclines obtained from Google maps were sufficiently accurate for the purpose of this work.

The final determinant of fuel usage that was investigated is payload. The load carried by a vessel has a major impact on its fuel usage over a specific route, specifically for routes that include steep inclines. The vehicles that were available for this study were however not always weighed before departure; we therefore had to use another approach to characterize the impact of payload for at least a subset of the total data set. For this purpose we analysed data on a two-way trip from a depot to a sugar mill where trucks travelled empty in one



direction and full in the other. This allowed us to measure the impact of payload for the same set of vehicles.

4. IDENTIFICATION OF MAJOR ROUTES TO BE USED FOR COMPARISON PURPOSES

In order to achieve the objectives of this study it was necessary to identify routes that were regularly travelled by a significant number of vehicles and that would reflect the impact of all the major determinants of fuel consumption as identified above. As the number of locations where weigh bridges are present is quite limited it was decided to use the availability of a frequently visited static scale as one of the primary considerations in this choice. This would allow us to incorporate payload as determinant for fuel usage in the analysis, at least for a subset of vehicles that visited the weighbridge during the trip.

Table 1 below lists the Sanral weighbridges where vehicle from the fleet were observed from time to time; this list was compiled by searching the entire database of weighbridge data for the list of vehicle registration numbers present in the fleet. The most frequently visited weighbridge by fleet vehicles is Heidelberg just south of Gauteng on the N3 on the road corridor between Durban and Johannesburg; this is also known to be the busiest road freight corridor in South Africa. We therefore decided to firstly consider vehicles that use this route.

Site identifier Name		Number of Occurrences
GAUTHEID	Heidelberg	1463
MPUMFARR	Farrefontein	700
KZNLMIDW	Midway	561
MPUMMDLE	Mid-East	555
MPUMMACH	Machado	523
KZNLMKON	Mkondeni	276
MPUMMDLW	Mid-West	138
МРИМКОМА	Komati	54
GAUTDONK	Donkerhoek	48
MPUMMDWT	Polokwane Zebediela	9

Table 1 Sanral weighbridges where fleet vehicles were observed

Figure 2 below displays the trip start positions for all trips included in the available data set. It can be seen that many of these are present along the N3 between Johannesburg and Durban, which confirms our choice. Figure 3 below displays the map for the N3 route as well as the elevation profile. It can be clearly seen that several steep inclines have to be mounted from Durban to Johannesburg; once the route has passed Harrismith it is relatively flat till the final destination is reached. For this route it will therefore be possible to characterize the fuel consumption over a relatively large sample of vehicles of which a percentage were weighed en route and that travelled across upwards and downwards inclines as well as flat sections.

We proceeded to refine the search for the most commonly used route by listing the loading and offloading points. Firstly we listed the most frequent loading points and selected from the list those that were in close proximity to the N3. For this most frequently visited loading point the list of most frequent offloading points were determined. In this way we identified the most frequently travelled single route from Redhill, Durban to Illovo, Johannesburg for which 323 trips were available within the available data set. We also identified the list of vehicles that had planned journeys between these points; 14 vehicles were identified that complete this route at least twice during the period covered by the data set.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 2 Trip start positions for all trips in the study



5. COMPARISON BETWEEN REFUELING AND FUEL CONSUMED FIGURES

As mentioned above, one of the primary objectives of the study was to compare fuel dispensed to vehicles against fuel actually used by the engine. This part of the investigation would serve several purposes:

- It can be used to calibrate the fuel flow gauges used inside the engines. It is quite possible that these gauges may have a calibration error which need to be corrected before the results can be used for comparison purposes. It can be expected that for a significant fraction of trips the fuel dispensed is correctly recorded. In such a case the fuel dispensed vs fuel consumed can be used to verify if the fuel usage gauge produced accurate results.
- It would indicate what fraction of fuel dispensed to a vehicle was not used in the engine if both
 measurements were accurate it would imply that the remaining fuel was removed (in all likelihood
 stolen). Once it is known during which trips most fuel disappeared in this way it could be determined
 where trucks made unplanned stops that could potentially be associated with efforts towards fuel theft.
- It would indicate how accurately fuel dispensing records were maintained if there are many cases where the engine used more fuel than what was dispensed to the vehicle based on fuel records it would imply that some records of fuel dispensed are not accounted for this can potentially be correlated with the responsible driver to determine whether negligence played a role.
- If however the general tendency is for fuel dispensed to exceed the amount of fuel used by vehicles, and at the same time manually recorded distances exceed distances based on the automatically sensed odometer reading, it may imply that these records are deliberately manipulated to hide the fact that fuel is siphoned from the system and that recorded distances are overstated to show acceptable consumption figures.



Figure 3 Map and elevation profile of the N3 Johannesburg-Durban route

The fuel records show that for the majority of trips vehicles are refuelled at the main fleet depot and that vehicles are most of the time filled up rather than partially refuelled. E.g. for one specific vehicle it was refuelled 137 times during the course of a year, 89 times at the main fleet depot with only 5 partial refuels. From the fuel records, which provide the date and time of the activity, and the GPS tracking data, it can be determined exactly where the fuel pump was located, as the vehicle had to be stationary for a period of time at this location.

In order to allow direct comparison between refuel and fuel consumption records we selected calendar periods during which vehicles always refuelled only at the main fleet depot, as this also corresponded with the end of the current trip and start of a new trip. Some instances of such comparison displayed discrepancies of up to 15% where the fuel dispensed was less than the fuel consumed - this indicated missing refuel records; this was



confirmed by comparison of the distance logs as in the same case the vehicle odometer recorded a 19% longer distance covered compared to the distance extracted from the refuel records.

Two further case studies are displayed in Table 2 below for vehicles that were refuelled only at the main depot. OBC refers to information available from the on-board computer based on measurement of distance and fuel consumed made available via the CAN bus system of the vehicle [8] [9] [10], while manual corresponds to records maintained during refuelling of the vehicles.

Item	Trip 1	Trip 2
Start date / time	2016-02-02 22:29:25	2016-02-01 09:55
End date / time	2016-02-04 14:44:21	2016-02-02 22:19
Distance travelled	905 km	1221 km
Fuel consumed	450 l	640 เ
OBC distance / manual distance	99.8%	99.84%
OBC fuel / manual fuel	88.8%	100.82%

Table 2 Case studies: refuel vs fuel consumed records

From the above experiment we can make the following conclusions:

- In all likelihood the good correspondence between distance covered and fuel consumed for Trip 2 provides proof of the fact that the OBC measurements are very accurate and that discrepancies due to faulty gauges are limited to a fraction of 1%.
- In the case of Trip 1 there is accurate correspondence between the distances but 11.2% less fuel was consumed than dispensed in all likelihood this represents a case of fuel theft.

We performed a more complete comparison between fuel dispensed and consumed for a particular vehicle. In order to make accurate comparisons we excluded data where there were suspected missing refuel records. While some partial refuel records were encountered, this would average out over a sufficiently large number of observations. Table 3 below displays a comparison between OBC and manually recorded data over a full year of data.

Table 3	Comparison betwe	en OBC and manua	l fuel records over	one year
---------	------------------	------------------	---------------------	----------

Vehicle FMID	Refuel distance travelled (Km)	OBC distance travelled (Km)	Refuel fuel dispensed after trip (L)	OBC fuel consume d (L)	OBC Km / Refuel KM %	OBC L / Refuel L %	OBC vs Refuel accuracy %
1340	86 891.00	84 937.70	43 415.07	42 312.74	97.75	97.46	99.70

As can be seen from the above table, the refuel logs indicate that the vehicle travelled slightly further that the OBC recorded and that slightly more fuel was dispensed than consumed. This may indicate that some trips were not properly logged by the OBC, which will be investigated further. However, taking the ratio between the fuel dispensed based consumption rate and OBC fuel consumed based consumption rate (both in km/l) yields 99.70% correlation, indicating that good accuracy is being achieved in the fuel dispensed and fuel consumed categories.

Extending this to all vehicles, the average results over 148 vehicles are shown in Table 4 below. A few further conclusions can be drawn from this comparison:

- In general manually recorded distances are overstated; this may be to try and make up for dispensed fuel that was not consumed by the vehicle but stolen.
- Significantly more fuel is dispensed to vehicles than actually consumed by the engines; this difference is close to 20%.
- Even when accepting the possibly overstated distance travelled there is still a discrepancy of almost 10% between the fuel consumption rate in l/km between OBC and manually recorded fuel levels this can be accepted as the minimum estimate for fuel shrinkage.

Table 4 Averages of comparisons between OBC and manual fuel records over 148 vehicles

Total refuel distance travelled (Km)	Total OBC distance travelled (Km)	Total refuel fuel dispensed after trip (L)	Total OBC fuel consumed (L)	OBC Km / Refuel KM %	OBC L / Refuel L %	OBC vs Refuel accuracy %
69 432.92	63 524.19	35 985.61	30 161.00	90.6	82.9	91.3



6. IMPACT OF DRIVER BEHAVIOUR

The results from the previous section show that major discrepancies exist between distances and volumes of fuel consumed as measured by the vehicle sensors and similar figures as recorded by drivers and fuel pump attendants during refuelling of vehicles. It confirms that driver behaviour, and more specifically manipulation of recorded figures by drivers, most likely in order to hide the siphoning of fuel, justifies more in-depth investigation.

For this purpose we calculated the following figures across all drivers and all routes:

- 1. The fuel consumption rate in km/l based purely on OBC measurements calculated over the full calendar year for each vehicle.
- 2. The fuel consumption rate in km/l based on manually recorded figures during refuelling events, for the same calendar period.
- 3. The fuel consumption rate in km/l calculated as OBC distance divided by dispensed fuel this is a figure that in many cases will be used by transporters to determine their average fuel efficiency levels, should they use the aggregate odometer readings for the entire fleet over a year as well as the total fuel expense of the company.
- 4. OBC distance as fraction of recorded aggregate distance during refuel events, to provide an indication of the overstating of actual travel distances as recorded by drivers.
- 5. OBC fuel consumed as fraction of fuel dispensed, to provide an indication of the fraction of fuel dispensed but not consumed by the vehicle, i.e. the best estimate of the level of fuel theft.
- 6. Standard deviations for all of the above, to provide an indication if larger deviations occur within the manually recorded figures compared with the OBC recorded figures if the manually recorded figures were manipulated it could be expected that larger deviations would occur within those sets of figures opposed to the OBC based figures, as it would be difficult to always siphon the same proportion of fuel between each refuel event.

These figures were calculated both as normal averages over all drivers as well as by weighting the contribution of each driver based on the OBC distance associated with that driver. The use of weighted averages will prevent large deviations that may occur within the figures for drivers that drove only small distances, from skewing the overall figures. We furthermore removed outliers that were beyond the 3 sigma limits and where the distance driven by that driver was very low, as it could be assumed that, in the case of drivers for whom only a few refuel events are available, one unintentional mistake in the recording of a refuel event can result in a totally unreliable figure. Amongst the 203 drivers 10 observations were removed for this reason; this represented approximately 5% of the total distance travelled by all drivers.

	OBC Km / Refuel km %	OBC I / Refuel I %	Refuel based km/l	OBC based km/l	OBC km / Refuel I	Refuel km/l vs OBC km/l %
Average	86.027	81.516	2.010	2.117	1.723	95.211
Weighted Average	90.574	82.251	1.942	2.156	1.754	90.828
Standard						
Deviation	18.384	20.637	0.389	0.241	0.463	15.500
Variance	337.958	425.889	0.151	0.058	0.214	240.243

Table 5 Comparisons between OBC and manual fuel records over 193 drivers

From the above set of figures we derive the following conclusions:

- 1. Distances travelled are overstated by approximately 10%, in all likelihood to hide the fact that fuel is siphoned.
- 2. Consumption in km/l based on recorded figures is 5% lower than actual consumption based on OBC figures drivers claim, based on their own recordings, to achieve approximately 1.94 km/l, whereas the OBC average is 2.16 km/l. The trucks therefore actually perform better than claimed by drivers, the latter figures once again manipulated to hide the fact that fuel is removed from the system.
- 3. The consumption figure that the fleet owner will arrive at by only using overall distance travelled based on odometer readings and overall fuel paid for is even worse (1.75 km/l) than what is effectively claimed by drivers.
- 4. The argument that manually recorded figures are manipulated is strengthened by the standard deviation figures: for the OBC based consumption, which is mainly impacted by factors like route inclination,



payload and idling times, the standard deviation is 0.24 km/l, whereas for the figure based on OBC distances and dispensed fuel the standard deviation is 0.46 km/l. For stochastic processes where different independent causes contribute to the same outcome, the relative importance of each are normally measured in terms of variances, as for normally distributed processes the total variance can be calculated as the sum of the variances caused by the contributing factors. If we therefore compare the variance within the OBC only km/l set with the variance where dispensed fuel is used, the latter has a variance that is between 3 and 4 times larger; differently stated the factors that should explain variance in fuel consumption in a system that is not manipulated (inclinations, payload, etc.) explain only 27% of the observed variation - the remaining 73% is caused by inaccurate recording and/or fuel theft.

5. For both normal and weighted averages the OBC fuel consumed is approximately 18% less than recorded fuel dispensed - this confirms that fuel theft is a significant problem, and that it is most likely the single biggest determinant of deviation between actual and ideal fuel cost.

7. IMPACT OF VEHICLE MODEL ON FUEL CONSUMPTION

The choice of vehicle model is a further important consideration for transport operators as it is known that not all models perform equally well in all conditions. For this reason we separately calculated the average OBC based consumption of each vehicle model over all routes and all drivers. In Error! Reference source not found. below we show the histograms of fuel consumptions for two different vehicle models. The average figures for a larger number of models appear in

Table 6 below; we also show the number of trips over which the calculations were performed. A large difference can be seen for different models; this may be partly explained by the fact that different vehicle models may be used over different routes, a factor that has not yet been removed from these figures.



25000 20000 15000 count 10000 5000 0 0.00 2.00 4.00 6.00 8.00 10.00 12.00 Km/L (b) Fuel consumption for Model 3

(a) Fuel consumption for Model 1


Figure 4 Histograms of fuel consumption for different vehicle models

	Number of trips	Weighted Average km/l
Model 1	23140	1.76
Model 2	19876	2.53
Model 3	215611	2.51
Model 4	10206	3.07
Model 5	17952	2.62

Table 6 Fuel consumption for different vehicle models

8. CASE STUDY: IMPACT OF INCLINATION AND PAYLOAD FOR A SPECIFIC ROUTE

Section 6 and 7 provided evidence that factors that can be managed to reduce overall fuel consumption indeed play a very significant role. This does however not mean that other factors, including route inclinations and payload, should not be further investigated. In a system where the unwanted phenomena have been largely eliminated, these factors should be the primary determinants of fuel cost, and should therefore be used not only to measure the performance of vehicle models and driver proficiency, but also to correctly price each route and each trip based on actual cost.

In order to investigate the impact of inclination and payload a route was selected that displays large difference between inclinations between the two directions of travel, and where the payload also differed significantly as trucks travelled empty in one direction and fully loaded in the other. The route selected was the Pietermaritzburg - Noodsberg route where trucks moved between a sugar refinery and an offloading point for processed sugar. In Figure 5 below the inclination of the route is displayed, showing on average uphill from Pietermaritzburg to Noodsberg but with mostly empty trucks and downhill the other way but with loaded trucks.

When studying the routes travelled by trucks between these destinations is was observed that for a portion of the journey two different routes were used; for the purpose of the analysis only trips that took the most travelled route were included. When the average OBC based consumption was calculated for the two direction the rather surprising result was obtained that the unloaded trucks traveling mostly uphill on average achieved a figure of 2.14 km/l, while the loaded trucks traveling mostly downhill could only achieve 1.59 km/l - a difference of more than 25%. As these figures were not based on recording of fuel dispensed but purely on OBC measurements it could be assumed that they reflect the true average performance of trucks on these routes.

This provides evidence that factors such as inclination and payload do play a very significant role in fuel consumption and should be incorporated into a pricing model on a per trip basis. Should in this case different customers be serviced in each direction, and should the uphill leg have involved a larger payload compared to the downhill leg the applicable fuel consumption for each leg would have differed by an even larger amount than the 25% recorded above. Given that fuel cost represent approximately 40% of the overall cost of a road transport operation, and that margins are often only a few percent, the implication is that the differential fuel cost for each customer specific trip can make the difference between running at a healthy profit or a significant loss.





Figure 5 Map and elevation profile of the Pietermaritzburg - Noodsberg route

9. CONCLUSION

In this paper we investigated different factors that impact upon fuel consumption of trucks operated on a variety of routes around South Africa. Fuel consumption and related data was collected from a fleet of approximately 400 vehicles over a calendar year during which more than 280,000 trips were completed. We found that large discrepancies exist between distances travelled and amounts of fuel consumed as recorded by the on-board sensors of the vehicles versus those that were manually recorded. While some of these may be due to negligence, the errors are mostly skewed towards larger amounts dispensed than what the vehicles actually consumed. More detailed investigations into cases where deliberately introduced mistakes were most likely absent indicates that the on-board sensors produce errors of less than 1%, leaving us with the conclusion that in most cases where significant discrepancies occur the manually recorded figures are manipulated in order to hide the levels of fuel siphoning that appear to be present. Based on a comparison between fuel dispensed and fuel consumed it would appear that the level of fuel theft is approximately 18%, in line with estimated figures that were obtained from various industry players. This finding suggest the need to improve the level of control over the way in which fuel figures are recorded as well as improved measures to physically prevent fuel theft from vehicles. If these figures are calculated and reported to managers on a per trip basis it should be possible to act against the offenders immediately after the completion of a trip.

We further investigated other factors that influence fuel consumption, and found that vehicle model also has a significant impact, justifying current practices to select specific vehicle models for specific types of routes based on average inclination. It was also demonstrated that payload and inclination plays a big role and should be incorporated into costing models to determine suitable pricing for specific routes and for vehicles carrying specific payloads. The primary application areas for the results of this work would be the evaluation of the performance of truck drivers and of different truck models, as well as the accurate costing of trips based on a combination of the distance to be covered, the characteristics of the route in terms of inclination and rural vs urban, as well as the payload to be carried.

Further work will refine this study by determining the trips and geographical locations where most fuel siphoning appears to occur, and to match these incidents with suspicious stops that are made in unauthorized locations along such trips. This can provide further evidence of the prevalence of fuel theft and provide motivation for the implementation of additional preventative measures. Weights obtained from Sanral weighing scales will be combined with vehicle GPS and consumption data to determine the impact of payload on consumption for similar vehicle models on similar routes. Routes will furthermore be characterized in terms of their average inclinations and the results incorporated into a fuel consumption model that can estimate the expected consumption per route, supporting more accurate benchmarking of fuel efficiency and assisting efforts towards more accurate costing of specific routes.



REFERENCES

- [1] A. J. Hoffman, "The use of technology for trade corridor management in Africa," in *NEPAD Transport Summit*, Sandton, Johannesburg, South Africa, 2010.
- [2] J. Havenga, "10th Annual State of Logistics Survey for South Africa, ISBN number: 978-0-7988-5616-4, p. 9.," Stellenbosch, South Africa, 2013.
- [3] J. Naidoo, "Transport White Paper: The South African Cross-Border Industry, THRIP Research Report," North-West University, Potchefstroom, South Africa, 2013.
- [4] "Stopping the flow of illegal black gold," Transport World Africa, p. Vol. 12 No. 3, June 2014.
- [5] J. d. Weille, "Quantification of Road User Savings," World Bank Occasional Papers No. 2, World Bank, Washington D.C., 1966.
- [6] D. Biggs, "ARFCOM Models For Estimating Light to Heavy Vehicle Fuel Consumption, Research Report ARR 152," Australian Road Research Board, Nunawading, 1988.
- [7] I. G. a. C. Bennett, "HDM-4 Fuel Consumption Modelling, Preliminary Draft Report to the International Study of Highway Development and Management Tools," University of Birmingham, Birmingham, 1995.
- [8] M. v. d. Westhuizen and A. J. Hoffman, "An Investigation into the Economics of Fuel Management in the Freight Logistics Industry," in IEEE Intelligent Transportation System Conference Proceedings, Qingdao, PRC, 2014.
- [9] J. M. Francke and P. Wouters, "Policy options for carbon efficient heavy vehicles".
- [10] Q. van Heerden and J. W. Joubert, "Commercial vehicle behaviour: analysing GPS records," in *CIE42 Proceedings*, Cape Town, South Africa, 2012.
- [11] Q. van Heerden and J. W. Joubert, "Generating intra and inter-provincial commercial vehicle activity chains," *Procedia Social and Behavioural Sciences*, vol. 125, pp. 136-146, 2014.
- [12] S. Schroeder, M. Zilske, G. Liedtke and K. Nagel, "Towards a multi-agent logistics and commercial transport model: the transport service provider view," *Procedia - Social and Behavioral Sciences*, vol. 39, pp. 649-663, 2012.
- [13] L. Gong, T. Morikawa, T. Yamamoto and H. Sato, "Deriving Personal Trip Data from GPS Data: A Literature Review on the Existing Methodologies," in *The 9th International Conference on Traffic & Transportation Studies*, Beijing, China, 2014.
- [14] "SAE J1939-21 Data Link Layer, Surface Vehicle Recommended Practice," April 2001.
- [15] "SAE J1939-71 Vehicle Application Layer, Surface Vehicle Recommended Practice," December 2003.
- [16] "FMS-Standard," 19- June 2014. [Online]. Available: http://www.fms-standard.com.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



DEVELOPMENT OF A DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY OF CAPITAL EQUIPMENT FOR A SOUTH AFRICAN RETAILER

H.S. Swart¹, Dr J.L. Jooste² & D.L. van Blommestein³

¹Department of Industrial Engineering University of Stellenbosch, South Africa Hesterswart34@gmail.com

²Department of Industrial Engineering University of Stellenbosch, South Africa wyhan@sun.ac.za

³Department of Industrial Engineering University of Stellenbosch, South Africa donvanblom@gmail.com

ABSTRACT

In this paper a methodology is presented for determining an optimal run, repair or replace policy for capital equipment in the retail industry. A decision support model is developed to assist the retailer's decision making process. The model incorporates both dynamic programming recursion and discounted modelling which assist in selecting the optimal sequence of decisions over a period of ten years which results in the lowest accumulated cost over the time period. A needs gap analysis is also performed to determine decision criteria currently not measured by the retailer.

¹ The author was enrolled for a B Eng. (Industrial) degree in the department of Industrial Engineering, University of Stellenbosch.



1. INTRODUCTION

The constant growth in the local retail industry has led both large and small South African retailers, to seek competitive advantages in order to minimise cost and maximise profits. Asset intensive retailers have recently identified that monitoring costs incurred with respect to capital equipment (CE) may lead to an increase in profits.

This paper originates from the need of a retailer with substantial investment in CE (such as generators, airconditioners and refrigeration systems) to reduce expenditure in the pursuit of competitive advantage in their industry. Currently, the retailer's decisions to run, repair or replace (RRoR) CE are based on past experience and intuition. Since this decision making process is predominantly subjective, there is a likelihood that the current RRoR policy is not optimal with respect o minimum cost incurred and minimum time called. Decisions about CE should not be based on intuition but rather on scientific economic policies [3].

To improve the RRoR decisions regarding CE, this paper addresses the retailer's need for a decision support model (DSM). The objective of the (DSM) is to minimise resources (cost and time) invested by the retailer over the decision time horizon.

To successfully address the need of improved RRoR decisions certain assumptions are necessary in order to formulate a mathematical model for optimising the retailers RRoR policy. These assumptions delimit the scope of the project, but ensures focus on the intended purpose. The development of the DSM is subject to the following assumptions:

- The modelling only considers the decision to either run, repair or replace CE. As these are the decisions the retailer is face with when it comes to their CE.
- Explicit focus is placed on the *monetary value* of the CE with respect to the cost incurred by the retailer over the planning horizon. This assumption is made based on the goal of research done.
- The unit of measurement used in the DSM is years and a total time period of 10 years is considered. The measurement and period chosen is based on the preference of the retailer.

2. LITERTURE REVIEW

2.1 Capital Equipment Costs and Run, Repair or Replace policies

A RRoR policy is implemented by an organisation or by an individual to ensure optimal use of CE which is determined from the optimum economic life of the CE. The optimum economic life of CE, also known as optimum replacement age, is the time at which the total annual cost of the CE is at a minimum value [2]. In addition, the operating costs, as well as the purchase price of the new CE are required to determine the optimum economic life of CE [2].

Two errors should be avoided when determining RRoR policies:

- 1. Repairing or replacing the CE too soon, and
- 2. Repairing or replacing the CE too late.

These errors are avoided by making the right decision (whether to run, repair or replace) at the right time (period).

The outcome of making one of the two aforementioned errors is illustrated in Figure 2.1. Additionally, the figure depicts the influence that ownership, operations and fixed costs have on the optimum replacement age. These three costs add up to the total cost and the optimal replacement age is the pont in time when the total cost curve reaches it's minimum. In other words the optimum replacement age is where the gradient of the total cost curve is zero.





Replacement Age (years)

Figure 2.1: The classic economic life conflict of CE

Adopted from [2]

When the economic life of the CE increases, the operations cost per unit rises while the ownership cost decreases [2]. Ownership cost of CE is defined as the purchase price minus the resale value at the replacement time, divided by its economic life [2].

Operating cost consist of standing and running cost. In this paper standing cost is defined as depreciation and running cost is defined as maintenance costs.

Fixed cost is made up of costs such as insurance and licensing. These costs are omitted from the DSM development since these costs have no effect on the economic life of the CE due to it being independent of the age at which the CE is replaced.

2.2 Models for Determining Run, Repair or Replace Policies

Various models for determining repair or replace policies are available in literature. These models vary in application and complexity and include the *limiting maintenance cost*, *limiting accounting cost* and *historical data* models. Each of these models have its own limitations [12].

There exists a relationship between the cost of performing maintenance and the value of the equipment at any point in time [4]. Furthermore, a limit should be determined for the cost of maintenance, the so-called *limiting maintenance cost* model. If this limit is met or exceeded, no further repairs should be performed and the CE should be replaced [4]. This cost limit will become a lower value as the CE ages while the probability of replacing the CE will increase. The maintenance cost limit is not ideal for determining a RRoR policy due to the assumption that maintenance cost is a function of time rather than a function of utilisation [5].

The *accounting limit* model calculates an optimal point of replacement by taking the cost of maintaining the CE as well as the resale value into account [5]. The model considers the point in time where the cost of maintenance intercepts the resale value as the minimum cost point. This point is considered a suitable point for replacement and occurs where the total cost is at a minimum value [5].

The *historical data* model considers historical data for determining an optimal replacement policy [6]. By using historical data, the effect of the CE's age and usage can be determined. This model focuses on identifying a replacement policy that replaces the theoretical limits of ownership which are set initially. This model is applicable to both the age and utilisation of CE. For example, a retailer can set a policy (based on historical data) which states the CE should be replaced when it is five years old or when it has been in use for a specific number of hours or years. The outcome of this policy is dependent on which limit is reached first.

2.3 Factors Influencing Run, Repair or Replace Policies

Various studies and approaches have been documented suggesting factors which should be considered when approaching the critical issues of determining optimal RRoR policies. For the purpose and scope of this paper the following factors are considered: *resale book value, age, maintenance and repair* and *inflation*.

According to Jardine an Tsang [2] the *resale book value* of CE is a function of time and decreases as time increases. The actual income incurred from selling the used CE depends on the age, utilisation and the general condition of the CE.



The age of CE is an important factor to take into consideration with resale. The resale value of CE can be defined as the current market rate of the depreciated CE value.

The decision to replace CE cannot be accurately predicted and therefore making use of a repair limit should not be the sole indicator of the RRoR policy for CE. Estimates of the replacement time can be made based on the predicted routine maintenance costs. Nonetheless the instance of the occurrence and the costs associated with the non-routine maintenance cannot be predicted.

Inflation is known as one of the factors that cannot be controlled to achieve a desired result in the national economy. Inflation is taken into account when determining operating costs to ensure realistic overall cost is calculated over the specified time period. In general, investigations and calculations that incorporate in historical costs have to be adapted for inflation to provide reliable outcomes.

It is evident from the background information presented that the recording of appropriate data and the quality of such data are important considerations for determining a RRoR policy for an organisation's CE. As part of this case study a need gap analysis is performed to identify limitations in the data collection practices of the retailer at which the case study is performed, followed by the development of a DSM for the retailer.

3. NEED GAP ANALYSIS FOR AN OPTIMAL RUN, REPAIR OR REPALCE POLICY

A need gap analysis along with the principles of maturity modelling were used to identify data that are required for the RRoR DSM, but were not previosuly collected or measured by the retailer.

A need gap analysis is defined as a method of assessing the difference in current and desired performance [7]. The gap between the potential and actual performances is referred to as the area between "where one wants to be" (future state), and "where one is" (current state). Figure 3.1 illustrates the procedure for constructing a need-gap analysis [7]. The need-gap analysis consists of three steps:

- 1) Future state: Identify the objectives needed to be achieved.
- 2) Current state: Analyse the current state for each of the identified objectives.
- 3) Describing and bridge the gap: Comparison of the two states in order to identify gaps. If there are gaps, they are described together with specific solutions to remove the gaps.

In summary the three steps can be summarized as:





Figure 3.1: An overview of the three steps of constructing a need-gap analysis

Adopted from [6]

Maturity models are defined as: "descriptive stages through which systems, processes or organisation's progress evolve as they are defined, implemented and improved." [8]. Therefore, a maturity model is a method used to scale the maturity of the processes or areas of a business. The concept of levels is a fundamental building block of a maturity model. A level is considered as an extent, measure or degree of intensity [8], [9]. A maturity model consists of different maturity levels, which are used in this study to assist in identifying the different areas of improvement for closing the gaps which the need-gap analysis will reveal.

3.1 Constructing the Need-Gap Analysis

The ideal future state is identified by incorporating the factors that typically influence a RRoR policy and the objective of the model of minimising the resources (cost and time) invested by the retailer.

For this study, the analysis of the current state is performed by means of a questionnaire which the retailer completed. The questionnaire, shown in Figure 3.2, was created according to the procedure devised by Diem [9].



Gap analysis questions for Capital Equipment (CE)

Please complete the table of questions below by ticking the appropriate answers (Yes/No).

Ques	tions	Yes	No
Mainte	enance Cost: costs incurred to keep an item in good condition and/or working	order.	
1.	Maintenance costs are tracked and documented on a monthly basis.		
2.	Maintenance is done on scheduled dates.		
3.	Maintenance is done based on CE failure.		
4.	A work order is completed before maintenance is done.		
5.	Unscheduled maintenance is documented (documentation of date).		
6.	The date(s) of the scheduled maintenance are documented.		
7.	The date(s) on which the scheduled maintenance are completed are documented.		
8.	The costs incurred due to the parts/material needed to complete maintenance are documented.		
9.	The labour costs incurred for completing maintenance are documented.		
Consec	quence Cost: costs incurred due to CE being replaced or idle.		
1.	Replacement costs are tracked and documented on a monthly basis.		
2.	Reason(s) for replacement is documented.		
3.	Documentation is made of the date on which the replacement is done.		
4.	Documentation is made of the date on which the replacement is completed.		
5.	Cost of new CE is documented.		
6.	The labour costs involved for the replacement are documented.		
7.	Downtime costs are tracked and documented.		
8.	The failure date(s) of the CE are documented.		
9.	Documentation is made of the downtime period.		
10.	Documentation is made of the production loss (costs).		
Repair	Cost: cost of labour and material needed to fix the broken CE and get it to e	operate	again.
1.	Repair costs are tracked and documented on a monthly basis.		
2.	Reasons for repair(s) are documented.		
3.	The repair period(s) are documented.		
4.	The performance of the production of the CE is documented.		
5.	The costs incurred due to the parts/material needed to complete the repairs are documented.		
6.	The labour costs incurred for completing repairs are documented.		
Operat	ting Cost: the cost incurred of working the CE.		
1.	Operating costs are tracked and documented on a monthly basis.		
2.	The costs of consumptions (e.g. oil and fuel) are documented.		
3.	Operating costs are documented under descriptive/categorised headings.		

Figure 3.2: Need-Gap analysis questionnaire

With the questionnaire completed by the retailer, an analysis could be performed in order to determine the current state. This was done by means of a percentage scheme. For example, for each "yes" answer for questions one to nine of the maintenance cost section in Figure 3.2, a 11.1 percent contribution is recorded in the current state. This means that if the retailer answered "yes" three times, the current state is calculated as 33.3 percent compared the ideal future state which is represented by 100 percent. The same method is used for the consequence costs (10 percent per "yes" answer), repair costs (16.16 percent per "yes" answer) and finally the operating costs (33.3 percent per "yes" answer). As can be seen in the percentages, each section's answers are independent from the other sections, yet each section's final contribution is dependent on the individual questions within the respective section.

An example of a completed need gap analysis is shown in Figure 3.3. It can be seen that for the maintenance cost there is no gap and the current state meets the future state. This results in the fact that no actions are needed for further collection of maintenance related data. For the consequence, repair and operating costs there exist gaps between the current state and the future state. Actions are required to close these gaps. The amount of detail of the need-gap analysis is based on preference and may be altered to show more or less information. The consequence cost is broken up into replacement cost and downtime cost. This is done to provide more detail of where the gap is within the consequence cost.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

		Ne	ed-Gaj	p Analysis			
Data Required	Current State	Future State	Gap	Action needed			
Maintenance Costs:	100%	100%	No gap	No action needed			
Preventative	20%						
Condition Based	0%						
Corrective	100%						
Consequence Costs:	-	-	-	-			
Replacement Cost	0%	100%	100%	Process implemented and documentation needed			
Downtime Cost	60%	100%	40%	Documented needed			
Repair Costs:	40%	100%	60%	Documentation needed			
Operating Costs:	0%	100%		Process implemented and documentation needed			
			Cost D	efinitions			
Maintenance Costs:	costs incurred	to keep an i	item in goo	d condition and/or working order.			
Consequence Costs:	costs incurred	due to CE 1	being repla	ced or idle.			
Repair Costs:	cost of labour	and materia	l needed to	o fix the broken CE and get it to operate again.			
Operting Costs:	the cost incur	red of working	ng the CE.				

Figure 3.3: Completed Need-Gap analysis

Using the output of the need gap analysis, the input data required for developing the DSM is determined. In addition to, the actions needed to obtain the input data are also highlighted.

4. DECISION SUPPORT MODEL FOR AN OPTIMAL RUN, REPAIR OR REPLACE POLICY

A DSM was developed to determine an optimal RRoR policy for the retailer. The DSM incorporates dynamic programming recursion (DPR) and discounted modelling (DM).

4.1 Mathematical Model

DPR is used to determine an optimal RRoR policy. DPR can be used to solve optimisation problems [1]. In addition a DPR problem is said to be in a given *stage* that simply consists of all the possible *states* that the system can occupy.

The stages are defined as the different time periods t, t+1, t+2,...t+9 and the states are defined as the different decisions being evaluated as shown in Figure 4.1.



Figure 4.1: The Transformation of the evaluated states (decisions) over the different stages (current or new)

The application of DPR is aimed at finding the shortest, or longest, path which joins two points in a network [1]. This means that the shortest, or longest, path is determined by comparing the lengths of the paths. With the objective of minimising costs incurred and time spend, the shortest path will result in the optimal solution. Each decision made has an impact on the path, thus each decision should be compared to one another for the optimal sequence of decisions to create the shortest path.

A model for the optimal replacement interval of CE with the objective of minimising total discounted costs are used for developing the DSM, and are formulated as 4.1:



$$f_{t}(n) = min \begin{cases} \frac{A_{c}r^{i} + \sum_{i=1}^{n} O_{i}r^{i}}{1 - r^{i}} \\ \frac{(R_{i} + A_{c}) + \sum_{i=1}^{n} O_{i}r^{i}}{1 - r^{i}} \\ \frac{(A_{c} - S_{i} + A_{n})r^{i} + \sum_{i=1}^{n} O_{i}r^{i}}{1 - r^{i}} \\ \end{pmatrix} \dots replace$$
(4.1)

where A_c is the initial acquisition cost of the current CE. A_n is the initial acquisition cost of the new CE. O_i is the operating and maintenance costs, r is the discounting factor, R_i is the repair cost at time i and S_i is the resale price of the CE at time i.

4.2 Using the Decision Support Model

The DSM was developed in Visual Basic for Applications (VBA) in Microsoft Excel. The model consists of three sheets which are available to the user. The worksheet labelled "READ ME" informs the user of all constraints and requirements regarding the DSM, as shown in Figure 4.2. Figure 4.3 shows the worksheet labelled "Historical Data". This worksheet allows the user to submit historical data (costs incurred) of the considered CE for the different time periods. It is intended that this worksheet be updated with historical data when such data becomes available. Finally, the optimal RRoR policy is displayed in the worksheet labelled "RRoR policy" as depicted in Figure 4.4.



Figure 4.2: The general layout of the first worksheet labelled "READ ME" related to the DSM

4.2.1 Assumptions, Constraints and Requirements

Some assumptions, constraints and requirements were considered during the development of the DSM. The assumptions pertain to the costs incurred and the decisions considered in the DSM (Figure 4.2). Due to the limited amount of historical data available (which was identified through the need-gap analysis), it is assumed that new CE incurs costs following a similar trend as in the past. This trend can be rationalised if more data becomes available or when updated costs are recorded in the model. The second assumption considers the decisions



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

evaluated as shown in Figure 4.4. The decisions evaluated are to run, repair or replace the CE. The number of states (decisions) at each stage (point in time) increases with a factor of the state to the power of stages. In other words, the number of decision routes considered at each point in time is calculated as the three to the power of the point in time. For example, in year two; 9 decision routes are evaluated. The DSM is developed to compute an output of infinite periods, however, Microsoft Excel does not have the capacity to compute decisions for periods of more than 10 years. Up to the tenth year 59 049 ,(3¹⁰) possible decision routes are evaluated for the optimal RRoR policy.

Users should first familiarise themselves with the assumptions, considerations and requirements to ensure they understand these before attempting to use the DSM. After reading the sheet labelled "READ ME", the user may choose to either submit data or execute the DSM algorithm.

F	ile	Hom	e	Insert	Draw	Page Lay	out Forn	nulas D	lata Review	View [eveloper)	♀ Tell me w						
Pa:	a み	- C		<u>u</u> - }	- 10 -	А° А́ • <u>А</u> •	= = =	&>- ≣ ≣	Wrap Text	Genera	। %,∣ :	ت انگ انگ Formatt	ional Formatas	Cell In Styles •	sert Dele	te Format →	itoSum ▼ A, I▼ Z sor ear▼ Filt	▼ 🔎 t& Find & er • Select
Clip	board				nt			Alignme			Number		Styles		Cell		Editing	
	•	-			. / F .													
Ра	9				√ Jx													
_																		
1																		
3																		
4																		
ы												H					М	N I
	<u>^</u>		`ani	tal Equ	linment	1			<u> </u>					· · ·	<u> </u>			
-		_	Japi		apment			_									Maintenance	
	Perio	ds		Acuis	ition Cost		Labour (and	Downtime	Operating	Maintenance Costs			Repair Costs			Operating Costs	Costs (after	
2	(Year	's) –					installation)	Costs	Costs (no repairs done)			Resale Price	Downtime			(after repairs)	repairs)	
3				Acquisitio	on of new C	E				Material	Labour		Costs	Material	Labour		Material	Labour
4	t					931,262	7,500.00	-	668,001	17,800	-	5,000.00	-	569,193	-	545,255	1,568	-
5	1+1								802,531	34,734	-	5,000.00	-	1,085,891	-	711,151	3,635	-
~	1+2								732 626	21,425	-	5,000.00	-	1 001 382	-	44,004	20,849	-
8	t+4								708 223	27 687		5.000.00	-	706 371		667 222	17 919	
9	t+5								1,328,643	20,392	-	5,000.00	-	1.009,772	-	27,654	5,170	-
10	t+6	1							980,239	18,549		5,000.00	-	751,275		52,771	11,341	
11	t+7								980,463	26,302	-	5,000.00	-	733,661	-	601,228	2,715	-
12	t+8								1,035,140	26,123		5,000.00	-	1,128,537	-	259,677	17,491	-
13	t+9						-		1,012,123	19,068	1.1	5,000.00	-	940,392	1.1	776,290	17,114	
14																		
16																		
17																		
18																		
19								112.		.								
20								<u>HI</u>	storical Da	te:								
21																		
22								Ple	ease compl	ete and s	ubmit 1	he require	d data an	d select:				
24										E C			-					
25											Ever	uto DSM						
26											Exec	ute DSM						
27																		
28											_				_			
29																		
31																		
32																		
33																		
34		_	_	_														
			DE/	DME	Historica	Data	PPoP pol	iou	A									

Figure 4.3: The general layout of the first worksheet labelled "Historical Data" related to the DSM

4.2.2 Submitting and updating Historical Data

The user can:

- 1. Update/enter data in the worksheet and execute the DSM, or
- 2. View the data and then execute the DSM.

Figure 4.3 shows the data required for the execution of the DSM for determining an RRoR policy.

Table 4-1: The data required for the DSM

Acquisition Costs This cost can entail purchase price of new CE, installation costs, downtime costs and even commission.

Resale Price

This is the price at which the CE will be sold.

Repair Costs

The costs incurred to repair a component or components of the CE.

Operating Costs

Two forms of operating costs are needed, the first are operating cost if no repair is done and the second is operating costs after repairs are done to the CE.

Maintenance costs

These costs can comprise of material and labour. As with the operating costs, two similar forms of maintenance costs are also required.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 4.4: The general layout of the first worksheet labelled "RRoR Policy" related to the DSM

4.2.3 Output of DSM

The optimal RRoR policy as determined by the DSM is presented in the worksheet labelled "RRoR policy", as shown in Figure 4.4. Additionally, the user is given the option to alter the decision at the different periods. The impact of these alterations is shown in the form of a graph created by the DSM (shown in Figure 4.4). The user may create up to 10 custom or user RRoR policies. After the user has generated the custom policies, a graph is created facilitating comparison between the optimal and custom RRoR policies. The optimal RRoR policy is shown in black and the custom policies in various colours (Figure 4.4).

The DSM was validated by means of face validation and a hypothetical case study. The face validation was performed in the form of a questionnaire completed by the involved retailer. The effectiveness of the DSM was judged through the following questions to the retailer. Following the questions, the responses given by the retailer are presented.

- 1. Considering the methodology developed to determine an optimal RRoR policy for CE, what is your opinion on the potential of the DSM developed?
 - "The DSM has potential, but the heavy dependence on data makes the DSM cumbersome and unlikely to be used in practice."
 - "Significant improvements in data capturing will need to be made for the model to successfully determine a replacement strategy."
 - Although the DSM has many favorable attributes, in the short to medium term there is little probability that the model could be employed successfully."
- 2. In your opinion what are the strong points of the DSM developed?
 - "Ability to handle exception price scenarios. The model can be run for a cheaper unit available immediately."
 - "The model has flexibility in that users can generate their own policy and compare the costs to that of the optimal solution. This will be useful in instances where future cash flow restrictions are known, or refurbishments are scheduled and down time can be minimized."
- 3. In your opinion what are the weak points of the DSM developed?
 - "The DSM is data intensive."
 - "There is not an option to compare similar types of equipment with one another."



- "The DSM does not consider depreciation."
- 4. Please comment on the following aspect of the DSM by giving a rating of 1 to 4; 1 being *poor*, 2 being *fair*, 3 being *good* and 4 being *very good*:
 - Graphical modelling approach: "2"
 - Presentation of results: "3"
 - Asset management and retail terminology used: "3"
 - Model consistency: "2"
- 5. Base on the comments given previously, in question 4, how do you think it is possible to improve the DSM for determining an optimal RRoR policy?
 - "Expand the model to compare different types of equipment."
 - "Simplify the model by reducing the data requirements."
 - "Increase the processing speed of the model by either reducing the computational expense, or use computing methods to improve the calculation speed."
 - "Include reliability considerations."
- 6. Please comment on the overall impression of approach to solving the optimisation problem (determine what decision to take at each period in time) presented by the student.
 - "The model provides an interesting perspective on the run, repair or replace decision."
 - By being able to view the decision at discrete points in time in offers flexibility and facilitates modification of the optimal strategy based on the company's needs."
 - "However, the strong theoretical basis of the solution with multiple decisions in the future makes the model fairly complex to unpack."
 - "Finally due to the lack of actual data, it is difficult to gauge the validity of the decisions."

Insufficient data was available to test DSM at the retailer, therefore a hypothetical case study is created to assess the validity of the DSM. The cost incurred in the hypothetical case study are justifiable as the retailer assisted in formulating this case study along with the assumptions made regarding the CE. The assumptions made with respect to the case study is tabulated in Table 4-2.

Table 4-2: Hypothetical case study assumptions

1. CE Considered

A 600KVA generator with a prime rating set at 70%

2. Downtime Cost

It is assumed that the installation and repairs are done after trading hours which means the retailer does not incur any downtime costs.

3. Acquisition Costs (installation of new CE and commission)

Installation costs consists of ZAR 5,500 for delivery and rigging plus ZAR 2,000 for commissioning when a new generator is installed.

4. Resale Price

The generator is sold for its value in kilograms of metal. This value is assumed at ZAR 5,000 plus inflation.

5. Repair Cost

The most severe failure that occur is motor failure. The assumption is made that if a part of the generator would fail and repair is needed, this would be the motor. The motor will be replaced While the alternator and switchgear will not be replaced. It is assumed that ZAR 569,193 is incurred when the decision to repair is considered at a point in time.

6. Operating Cost

It is assumed that the cost for operation is ZAR 668,001 for period t.

7. Maintenance costs (Repair Cost)

An annual service period of t is required, at a cost of ZAR 17,800 which is incurred for maintenance.

8. Interest Rate

The average interest rate is assumed to be 9% for the period t to t+9.

The resale price for the case study is constant throughout the full time frame and is only influenced by the average inflation. The installation costs, maintenance costs, operating costs and repair costs for periods t+1 to t+9 for the second to tenth generator considered for repairing or replacing is created by means of a Microsoft Excel random number generator.

The output of the model with the input data given as detailed for the case study is depicted in Figure 4.4. The optimal RRoR policy for the case study for the time period t to t+9 is to run, repair, repair, repair, run, repair, run, repair, repair, repair, run, repair, repair, repair, run, repair, row the respective points in time with the total incurred cost being ZAR 70,396,645 as seen in Table 4-3. This is a table of the results displayed in Figure 4.4.



Period	t	t+1	t+2	t+3	t+4	t+5	t+6	t+7	t+8	T+9	Incurred cost (ZAR)
Optimal	run	repair	repair	repair	run	repair	run	repair	repair	run	70,396,645
User 1	replace	run	run	run	run	run	run	run	run	run	95,168,667
User 2	replace	repair	repair	repair	run	repair	run	repair	repair	run	89,746,730
User 3	run	run	run	run	run	run	run	run	run	run	75,321,442
User 4	repair	run	repair	run	run	repair	run	replace	repair	run	77,849,310
User 5	replace	run	run	run	repair	run	run	run	repair	run	95,970,218

Table 4-3:Optimal RRoR policy generated by DSM along with five user policies with total incurred cost per policy

Five other user policies are considered and none deliver a lower accumulated cost than the optimal policy produced by the DSM.

5. CONCLUSION AND RECOMMENDATIONS

Determining an optimal RRoR policy is not a trivial task. Alternative decisions need to be considered, while ensuring the final decision is optimal. This decision is influenced by various factors which differ from retailer to retailer and are determined by the retailer's business objectives.

Each CE group exhibits its own unique trends and relationships when it comes to cost incurred over a period of time. This is due to different operating conditions and maintenance strategies. It is important to take these conditions and strategies into account when designing a RRoR policy. To account for these variations a run, repair or replace decision is required to be considered at the end of each period. The DSM makes provision for evaluating each decision made at each period, then selects the sequence of decisions with the lowest accumulated cost over the total time period.

In order to select the decision with the lowest cost, quality data is needed. Good quality data will ensure a higher degree of accuracy and reliability.

The need gap analysis identifies the data required for the model. As part of the study the retailer completed the need-gap analysis for determining how readily available the required data is from industry asset owners. The results of the need-gap analysis show that only some of the data is available. The problem is not so much the collection of data but rather the type of data which are recorded. Emphasis is placed on the financial aspects of CE and not on the condition, utilisation and lifetime data of the CE. Although only one need-gap analysis was performed it is expected that similar limits exist in the wider retail industry.

The following limitations are associated with the DSM in its current format and should be considered in the future use of the DSM:

- The DSM is dependent on the quality and availability of the historical data. Assumptions would need to be made if the required data are unavailable. If the quality of the data is poor, the optimal RRoR policy produced by the DSM would be compromised.
- The user of the DSM needs to have a basic understanding of RRoR policies and what they entail in order to interpret the computed results.

It is important to be aware of these limitations when using the DSM or when making run, repair or replace decisions.

The following recommendations should be considered for future research in the field of RRoR decision making:

- Investigate other factors. The factors investigated in this project are resale book value or resale price, age, maintenance and repairs and inflation. It may be useful to investigate decrease in value of CE over time (depreciation), as well as the reliability of the CE.
- Additional features. Currently the DSM does not include the option to compare similar types of CE for the lowest acquisition costs. Investigating and incorporating this may lead to a different optimal RRoR policy to be derived with a lower accumulated cost.
- *Limitations*. Investigating the limitations (DSM dependency on quality data of the DSM and user having a basic understanding of RRoR policies) may decrease the limitations of the DSM and create more opportunities for the application of the DSM.



Researching and investigating these recommendations will lead to the improvement of the DSM and consequently more opportunities and assistance in run, repair or replace decision making.



6. **REFERENCES**

- [1] **Douglas**, J. (1975). *Replace for more profit. Construction methods and equipment.*
- [2] Jardine, A.K and Tsang, A.H. (2006). *Maintenance, replacement, and reliability*.
- [3] Drinkwater, R and Hastings, N.A. (1967). An economic replacement model. Operations research pp. 121 138.
- [4] Kritzinger, J.L. (1990). Optimale Voertuigvervanging in 'n Vlootomgewing. Masters of Business Administartion Thesis. Stellenbosch University, Stellenbosch.
- [5] Pedraza-Martinez, A.J and Van Wassenhove, L.N. (2013). Vehicle replacement in the international committee of the red cross. Production and operations Management vol. 22 no.2, pp. 365 376.
- [6] Brown, S.W and Swartz, T.A. (1990). A gap analysis of professional service quality. The Journal of Marketing, pp. 92 -98.
- [7] Clark, T and Jones, R. (1999). Organisational interoperability matirity model for c2. In proceedings of the 1999 Command and Control Research and Technology Symposium.
- [8] Dictionary.com . (2015). [Online]. Available at : http://www.http://dictionary.reference.com/browse/level. [Accessed 2 September 2015].
- [9] **Diem, K. (2004).** A step by step guide to developing effective questionaires and survey procedures for program evaluation and research.
- [10] Winston, W.L. (2004). Operations research: Application and algorithms. Brooks/Cole.ISBN 0-534-42362-0.
- [11] O'Keefe, R.M. (1989). The evaluation of decision-aiding systems: Guidelines and methods. Information & Management, vol. 17, no 4, pp. 217 - 226.
- [12] Mitchell, Jr. Z.W. (1998). A statistical analysis of construction equipment repair costs using field data and the cumulative cost model.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

EVALUATING LEAN IMPLEMENTATION IN SOUTH AFRICAN CASTING FOUNDRIES

Y.N. Mawane

Department of Mechanical and Industrial Engineering Technology

University of Johannesburg, South Africa

yvoniem@uj.ac.za

G. Muyengwa

Department of Mechanical and Industrial Engineering Technology

University of Johannesburg, South Africa

gmuyengwa@uj.ac.za

ABSTRACT

Manufacturing companies implement lean to enhance their competitiveness and meet the ever changing demands of the globalized markets. Literature suggests that when lean tools are correctly implemented, companies experience better productivity, improved worker morale and profits. However not much research has been done to evaluate lean implementation and the identification of best practices. This research paper evaluates and studies the implication of lean implementation in South African foundry environment.

Multiple case studies were carried out to evaluate lean implementation. These case studies involved a survey, where questionnaires were distributed to the foundries. The study focused on lean tools, techniques and strategy that companies adopt in the implementation of lean. The research showed that foundries have introduced lean manufacturing. The results revealed that foundries benefit from lean implementation. Some of the noticed benefits were improved quality, production, profitability and customer service. Barriers that hindered lean implementation were weak top management involvement, poor worker attitude and the non-availability of adequate resources.

This paper contributes to the body of knowledge in lean philosophy. To lean practitioners the paper highlights some of the critical factors that needs to be monitored for a successful lean implementation.

Keywords: Manufacturing, lean tools, lean implementation, foundry and productivity.



1. INTRODUCTION AND BACKGROUND

In today's competitive environment, companies are required to enhance their operations and strategically implement and design procedures which assist in improving product quality, reducing costs and improving delivery performance, [1]. Many industries have increasingly used lean manufacturing in order to improve performance, although, lean techniques were initially applied only to industrial operations, [2]. Womack, [3], describes lean as the most powerful tool available for creating value while eliminating waste in any organization. The implementation of lean tools has a significant impact on various industries, [4]. Lean tools have potential in assisting productivity improvements and reducing waste, [5]. According to Forbes and Ahmed, [6], manufacturing industries have greatly improved their competitiveness through the use of lean methods. Studies by Womack, [3], reveal that other organizations that have implemented lean manufacturing successfully have gained a competitive advantage. However, there are pitfalls for organisation's that attempt to use this powerful strategy without a thorough knowledge and understanding of the critical success factors of lean, [1].

Lean production emerged in the automobile industry, [1]. It originates from the Toyota Production System, [7]. The Toyota production system provides the basis of what is known as lean thinking by Womack, [3]. Many companies envy the way Toyota does its production, [1]. The Toyota system is really about growth and profit, [1]. Consequently, the successes of lean manufacturing principle were never questioned because Toyota had taken control of the market share in automobile production, [8]. A serious weakness with this argument, however, is that if lean is such a powerful approach and Toyota models it so well, why is it that few organizations are able to implement lean successfully. This implies that few companies succeed in the way Toyota did, [1].

The aim of lean production is to reduce waste using several tools and techniques. Different organisations have implemented lean manufacturing; therefore, lean principles can be applied in any industry, [5]. The key problem with this statement is that is it possible to apply lean in various industries where the traditions are different, [9]. The latter point has been critiqued by Womack, [3]. However he argued that it is essential to try. Chakravorty and Atwater, [10], defines lean manufacturing as "a philosophy, strategy or concept that depends on a set of practices or tools that are implemented to minimize waste in order to improve company performance".

Casting foundries in South Africa are challenged by highly regulated energy intensive processes, [11], however, there have been several investigations into reducing energy consumption. Lean tools and techniques have proven to be effective in improving casting foundry sustainability. As the foundry industry is an industry where efficient production and environmental impacts are strongly linked, Torielli and Abrahams, [11], suggests that foundries can become economically and environmentally sustainable businesses with the systems approach offered by implementing lean. He further suggested that, in an operation industry like casting the purpose of implementing lean is to reduce the production cost thereby eliminating non-value adding activities. Foundry plants which are small, medium and large in size should also be a place for lean implementation, [7].

Foundries in general face various barriers in lean implementation. Several authors have highlighted various barriers faced by foundries such as: top management and workers resistance, [12]; [13], lack of resource availability, [14]; [15], cultural differences, [16]; [17], lack of clear communication, [17], lack of top management involvement, [16], [18], lack of formal training for managers and workers, [14], [19]. These challenges are the motivation of this study.

Although success factors indicate that lean intervention in South Africa organisation has been successful, [1], literature suggests that many organisations are however finding that lean implementation is not easily achieved, [20]. Kanakana,[8], cited in Vermaak, [1], stated that, more than half of South African companies that have tried to implement lean have not even reached the standard of industry average. Some organizations adopt lean operations principles either as defensive strategy to stay competitive or as an offensive strategy to move ahead of competitors, [21].

This paper evaluates lean implementation in 3 casting foundries. Consequently in an attempt to eliminate nonvalue adding activities within the casting process, the paper investigates barriers faced by local foundries when implementing lean.

1.2 Research Objectives

The objectives of this research are:

- To evaluate how South Africa (SA) foundries have implemented lean.
- To identify lean tools that can help the foundries eliminate waste.



1.3 Research Questions

The research study seeks to answer the following questions:

- To what extent has SA foundries implemented lean?
- What are the barriers to lean implementation?
- To what extent has foundries benefited from lean implementation?

2 Literature Review

2.1 The Concept of Lean

Lean is a concept comprising a set of principles, practices, tools and techniques which, when implemented by following a systematic approach would improve resource utilization, quality and delivery with respect to products and services, [6]. It is a production philosophy which considers that any activity which consumes resources but not create value for the end customer is wasteful and therefore should be eliminated, [3],[13].). Consequently, lean production was developed in the 1950's to eliminate waste, [22], [1]. Waste in this context refers to overproduction, waiting time, excess inventory, defective goods or any other factor that can disrupt the even flow of goods along the transformation process, [23]. Lean production has an underlying philosophy that, by eliminating waste, quality can be improved and costs reduced, [22].

2.2 Lean Manufacturing Tools and Techniques

Lean manufacturing is carried out through a set of lean practices, [5]. The implementation of these tools and techniques identifies the elimination of numerous wastes which exists, [24]. In order to reduce waste lean manufacturing tools and techniques should be employed, [1]. The foundation of lean manufacturing includes the following tools and techniques:

5S

• This tool is a systematic method for organizing and standardising the workplace, [25]. It's one of the lean simplest tools to implement, [21]. The activities are (1) Sort out (2) Set in order (3) Shine/Cleanliness (4) Standardize (5) Sustain, [24].

Kanban

• A kanban system is an information system that controls the required parts at the required time, [5].

Kaizen

• Kaizen is the Japanese name for continuous improvement and it is central to lean operations, [17]. It is the process of identifying and eliminating wastes as quickly as possible at the lowest possible cost, [24].

Continuous improvement

• Continuous improvement can be defined as the planned, organized and systematic process of ongoing, incremental and company-wide change of existing practices aimed at improving company performance, [15].

Work Standardized

• Operations are organized in the safest, best known sequence using the most effective combination of resources, [24]. The job is broken down into elements.

Value Stream Mapping

• Value stream mapping is employed to identify the areas that need to be improved and to decide the wastes to be eliminated, [2], [5]. It provides a graphical flow of the process and supports the related value stream analysis tool. It also provides the scope for the project by defining the current and future states of the system, [24]. [5], agrees that, it facilitates a communication flow from customer to supplier and a resource flow from supplier to customer.

Total Productive Maintenance

• Total Productive Maintenance (TPM) is a maintenance program, which involves a newly defined concept for maintaining plants and equipment. The goal of the TPM is to increase production, increase employee morale and job satisfaction, [26].

Total Quality Management

• This is a management system used to continuously improve all areas of a company's operations, [2].



2.3 Barriers in Lean Implementation

Implementing lean manufacturing is not an easy task for any organisation, [18]. Despite the benefits from lean manufacturing, however, many companies are not successful in implementing lean, [27]. According to Jadhav et al., [12], numerous problems and issues regarding the implementation of lean manufacturing has been reported.

In general foundries face barriers in lean implementation. Several authors have highlighted these barriers as, [12], [18], [27]:

Top management and workers resistance

• The primary reasons for resistance are often a lack of clarity and uncertainty of the change, pressure and the challenge to learn something new, [28]. Furthermore, resistance from employees might be due to the fear factor of losing their jobs as Lean implementation eliminates non-value adding activities, [13].

Lack of formal training for managers and workers

• According to Miller, [29], the aim of training is to impart new knowledge, skills and attitudes on employees for the sole purpose of performance improvement. This was supported by Houshmand and Jamshidnezhad, [30], who reported that lack of training hampers the improvement process. Miller, [29], further argues that training must enable workers to adapt to the fast changing global competitive environment. Eswaramoorothi et al., [14], suggested that a lack of lean training in foundries leads to low levels of successful lean implementation.

Cultural differences

• The cultural barrier introduces challenges in implementing lean techniques, [9], [12]. Hines et al., [31], maintains that while organizations have introduced lean techniques relatively easily, they still find it difficult to achieve the organizational culture and mindset that provides the foundation of lean. Robinson and Schroeder, [9], reported that the workforce is usually difficult to engage in creating a culture of lean improvement. However, culture is deep-seated and difficult to change, but leaders can influence or manage an organization's culture, [12]. Respectively, defining lean culture is important in foundries.

Lack of top management involvement

• For any organisation to succeed in lean implementation, top management must take full responsibility in empowering all employees in lean activities. Lack of commitment may lead to limited access to resources, lengthy decision making processes and communication breakdowns, [32].

Lack of resource availability

• Lack of resources (financial, technical, and human) is a common barrier for implementing lean, [15]. According to Tracy, [23], some organizations avoid implementing lean as they believe that they do not have sufficient resources to invest in a lean program.

Lack of clear communication

• In foundries, employees need to be properly informed of the changes that are being implemented, [17]. Barriers to communication may lead to the termination of a lean project, [32].

2.4 South African Foundries

A foundry is a factory that produces metal castings from either ferrous or non-ferrous metals including copper, brass, bronze, aluminum, zinc, lead, nickel, and all their various alloys, [33]. They offer services such as, mould-making, melting, pouring, heat treating, surface cleaning and other finishing operations. Many foundries also offer welding and assembly services to achieve semi or finished products ready for use by customers or other manufacturing operations.

The Executive Director of the Aluminum Federation South Africa reported that, the foundry sector is one of the foundation stones for all manufacturing and tool and die making. About 80% of manufactured products have castings in them. Yet, South African foundries have been in decline for more than a decade, though there are still some centers of excellence, [34]. The local foundry industry has been earmarked as one of the manufacturing sectors with a significant potential for growth, in both the domestic and export markets. However, the South African Institute of Foundry men (SAIF) believe that growth opportunities for the industry lie in embracing technological development that will create new market opportunities.



The South African foundry industry is predominantly made up of small to medium sized companies. According to a study by National Foundry Technology Network, [35], smaller foundry companies that employ less than 100 people comprise 80% of the South African foundry industry. Foundries serve many markets and industries, including automobile, construction, agriculture, forestry, mining, pulp and paper, heavy machinery, air craft, municipal services, military, railroad, chemical processing and energy production. Hence a study on casting foundries' productivity improvement using lean methods is relevant for the South African economy.

3 Methodology

A case study methodology was adopted for this paper, [36]. The case studies focused on three Casting Foundries in South Africa. The case study consisted of a mixed methodology which included a detailed literature review on lean implementation, a mini-survey with a structured questionnaire and some interviews. From literature a questionnaire on lean implementation was adopted from the work of Mehta et al., [16], Wan and Hou [37] and Akugizibwe and Clegg [38]. The questionnaire focused on evaluating lean implementation in South African Casting Foundries by examining factors such as tools and techniques, strategies and barriers in the foundries. The questionnaire is attached as Appendix 1. An analysis of the lean implementation is discussed below. The analysis was done through a comparison of agree and disagree as shown in the analysis graphs. The respondents included managers, production supervisors and shop floor workers of the three studied foundries. The responses were analysed as percentages of disagree and agree.

4 Findings and Discussion of Results

A total of 18 questionnaires were distributed amongst managers, production supervisors and shop floor workers of the three casting foundries. 18 responses were received and used for analysis, Company A had 5, Company B had 7 and Company C had 6 responses respectively. 3 managers from the three foundries were interviewed. The results obtained from the research is presented in this section. All the data was grouped together and graphs were drawn showing the percentage of those who agreed or disagreed with the lean implementation themes covered in this paper: aims of implementing lean; tools and techniques used in implementing lean; strategies used to engage staff in lean projects and progress made in implementing lean. The data was used to analyse lean implementation in casting foundries.

4.1 Aims for implementing lean

Aims for implementing lean in the casting foundries were identified as cost reduction, quality improvement, customer satisfaction, quality improvement and reduction in lead time respectively. For example on cost reductions 17 responses, which gave 95 % as indicated in figure 1, agreed that cost reduction was achieved through the attempts made in removing the 7 wastes, [24]. All three foundries indicated a reduction of lead time that was achieved through improved manufacturing planning and the implementation of quality management systems. These efforts have had an effect of improved customer satisfaction, [26], and increased worker motivation, [15]. Information given by the three managers through interviews indicated that there was improvement in production due to quality improvements and cost reductions.



Figure 1: Aims for implementing lean



4.2 Tools and techniques used during implementation of lean

The foundation of lean incorporates set of tools and techniques into the business process. Vienazindiene and Ciarniene, [25], further stated that, these tools and techniques lead to improvement in five dimensions. These dimensions are elimination of waste, continuous improvement, continuous flow and pull-driven systems, multifunctional teams and information systems, [25].

The results in the figure below show effective tools and techniques that assists casting foundries in their sustainability. The most common tool used in the casting foundries was Kaizen, followed by Total Quality Management and Workplace Organisation in third place. Kaizen appears to be very significant. Mehta et al., [16], reported that "Kaizen can play a leading role in making lean process a successful one as it helps the organization to meet the challenge of doing more with the same or fewer resources". Interviews revealed that the casting foundries are also using other quality management tools in their plants other than lean tools. The managers indicated that they were familiar with quality improvement tools such as check sheets, Pareto, cause and effect diagram and both variable and attribute charts.

TQM implementation has helped management to deliver on organisational goals, targets and strategy. Worker empowerment was noticeable during the study period's visits to the foundries. This empowerment has improved team work, communication and commitment to organisational transformation. All the foundries are using fact based decision making through the appropriate use of data collection and analysis methods. Barriers to TQM implementation were avoided through education and training. Study interviews indicated that worker suggestions are now an acceptable norm by management. 5S has improved all the foundries workplace organization.





4.3 Strategies used to engage staff in the lean project

Lean manufacturing can be described as bundles of practices, [25]. It engages people in the act of continually increasing value creation. Hence, all production employees must work together. The most strategies which were used in the casting foundries to engage staff in the lean project were emphasis on training and skill building, employment engagement in identifying solutions and incentives. Through interviews, the research established that before lean implementation efforts, employees were not properly empowered to make contributions to their working environment. Lack of empowerment causes risks such as lack of morale, [29]. This has a negative impact on the lean project success. Interviews revealed that the staff was exposed to sufficient training and to appropriate lean education. Incentives of production bonuses served as a motivation to the adoption of lean in two of the studied foundries. Training that was offered to workers improved their knowledge, skills and attitudes. This helped in process improvements efforts. Training and skills building helps subordinates to understand their responsibilities, authority and accountability, [19], as they contribute in achieving the objectives and goals of the organization, [3].



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 3: Strategies used to engage staff in the lean project

4.4 Progress made in implementing lean manufacturing

The study highlighted the progress that has been made in implementing lean in casting foundries. According to the empirical research conducted by Vienazindiene, [25], statistics shows that the most significant progress is seen in areas such as return on assets, on time delivery, machine availability, machine set up reduction, reduction in floor space and inventory reduction. The managers reported that the cycle time improved and lean implementation has been beneficial for their casting foundries.





4.5 Barriers of implementing lean manufacturing

The barriers which were noted as preventing or delaying lean manufacturing implementation are shown in Figure 5. Barriers may differ depending on the sector of economy and specific company; however, 90% of the responses indicate that the main barrier was backsliding to old ways of working, [25]. This may be caused by the fact that, it was hard for employees to adapt to the new system and procedures. The managers indicated that the lean philosophy must be clearly understood before implementation and there must be ongoing commitment from both employees and management. The managers also indicated that lean project funding was a major constrain. They



reported that the funding was not always adequate and undercapitalization of projects is a risk. Wan and Hou, [37], reported that undercapitalization was due to inaccurate cost estimates and exchange rate variability. The research also established that some of the resistance to changes was caused by lack of clear communication. This may lead to termination of lean project. However sustained efforts in education and skilling reduced the impact of most of the barriers.



Figure 5: Barriers of implementing lean

5 Conclusion

The success of lean implementation can only be achieved through people's efforts. It is the people that bring the concept of lean to life. The interaction between top management and workers is important. All production employees must work together to eliminate waste. Senior managers in Casting Foundries were encouraged to directly involve and motivate all staff and provide necessary resources in the training and implementation of lean. Lean is a long-term process which includes many difficulties and requires commitment and resilience. Hence, senior managers were also encouraged to create the right conditions that assist openness and transparency that will result in good team building and better cohesion of team members. The research found that, the rate at which lean techniques are implemented depends on the circumstances of individual foundries, however, the most significant progress seen in both Casting Foundries were reduction of production costs which increased profitability and the development of employees and work environment. This proves that the lean concept can be applied in the casting foundry industry, thereby improving productivity.

6 Limitations

Only three casting foundries were studied making it difficult to genaralise these results. Financial figures before and after lean implementation were not obtained for analysis, hence detailed analysis of business results was not done. The effect of new technology and impact of new product development were not part of this study, but all these could give different results.



7 References

- 1. Vermaak, T.D. 2008. Critical success factors for the implementation of lean thinking in South African manufacturing organizations. PHD Thesis.
- 2. Bhasin, S. 2008. Lean and performance measurement. Journal of Manufacturing Technology Management, 19(5), 670-684.
- 3. Womack, J.P. 2003. Lean thinking. New York. Simon and Schuster.
- 4. Vienazindiene, M. and Ciarniene, R. 2013. Lean manufacturing implementation and progress measurement. Economics and Management: 18(2), 366-373.
- 5. Vasudev, K.C. and Mallika, P. 2013. Comparative study of lean manufacturing tools used in manufacturing firms and service sector. Proceeding of the World Congress on Engineering. International Journal of Services and Operations Management, 1(2), 175-208.
- 6. Forbes, L. and Ahmed, S. 2004. Adapting lean construction methods for developing nations. Second LACCEI International Latin America and Carribean Conference for Engineering and Technology (LACCEI' 2004) "Challenges and Opportunities for Engineering Education, Research and Development" 2-4 June 2004, Miami, Florida, USA.
- 7. Jezierski, J. and Janerka, K. 2013. The lean manufacturing tools in polish foundries. Archives of Metallurgy and Materials, 58(3), 937-940. Doi: 10.2478/amm-2013-0105.
- 8. Kanakana, M.G. 2013. Lean in service industry SAIIE25 Proceeding, 9th-11th of July 2013. Stellenbosch. South Africa.
- 9. Robinson, A.G. and Schroeder, D.M. 2009. The role of front-line ideas in lean performance improvement, Quality Management Journal, 16(4), 27-40.
- **10.** Chakravorty, S. and Atwater, J.B. 1995. Do JIT lines perform better than traditionally balanced lines? International Journal of Operations and Production Management, 16(6), 91-108.
- 11. Torielli, R.M., Abrahams, R.A., Smillie, R.W. and Voigt, R.C. 2011. Using lean methodologies for economically and environmentally sustainable foundries, China Foundry, 8(1), 74-88.
- 12. Jadhav, J.R., Mantha, S.S. and Rane, S.B. 2014. Exploring barriers in lean implementation. International Journal of Lean Six Sigma, 5(2), 122-148.
- 13. Wong, Y.C., Wong, K.Y. and Ali, A. 2009. A study on lean manufacturing implementation in the Malaysian electrical and electronics industry. European Journal of Scientific Research, 38(4), 521-535.
- 14. Eswaramoorothi, M., Kathiresan, G.R., Prasad, P.S.S. and Mohanram, P.V. 2011. A survey on lean practices in Indian machine tool industries, International Journal of Advance Manufacturing Technology, 52(1/2) 1091-1101.
- **15.** Achanga, P., Shehab, E., Roy, R. and Nelder, G. 2006. Critical success factors for lean implementation within SMEs. Journal of Manufacturing Technology Management, 17(4), 460-471.
- 16. Mehta, R.K., Dr. Mehta, D., and Dr. Mehta, N.K. 2012. An exploratory study on implementation of lean manufacturing practices (With special reference to automobile sector industry). YÖNETİM VE EKONOMİ Yıl:2012 Cilt:19 Sayı:2
- **17. Cudney, E. and Elrod, C. 2010.** Incorporating lean concepts into supply chain management. International Journal of Lean Six Sigma and Competitive Advantage, 6(1/2), 12-30.
- Nordin, N., Deros, B.M. and Wahab, D.A. 2010. A survey on lean manufacturing implementation in the Malaysian automotive industry. International Journal of Innovation, Management and Technology, 1(4), 374-379.
- **19.** Schein, E.H. 2004. Organisational Culture and Leadership. 3RD Edition, Jossey-Bass, San Francisco, CA.
- **20. Van der Merwe, K.R., Pieterse, J.J. and Lourens, A.S. 2014**. The development of a theoretical lean culture casual framework to support the effective implementation of lean in automotive component manufacturers. South Africa Journal of Industrial Engineering, 25(1), 131-141.
- 21. Nicholas, J. and Soni, A. 2006. The portal to lean production. Boca. Auerbach.
- 22. Kempton, J. 2006. Can lean thinking apply to the repair and refurbishment of properties in the registered social landlord sector? Structural Survey, 24(3), 201-211.
- 23. Tracy, R. 2007. Driving lean through your supply chain: the cultivation of a more efficient supplier network. White Paper, Intek Plastics.
- 24. Kumar, K.R., Dr. Reddy, Y.V.M., Dr. Gowd, B.U.M. and Baswaraj, D. 2013. Improving the performance of manufacturing sector-lean methodology and its impact. International Journal of Engineering Science Invention, 2(3), 37-41.
- 25. Vienazindiene, M. and Ciarniene, R. 2012. Lean manufacturing: theory and practice. Economics and Management, 17, 732-738. Doi:10.5755/j01.em17.2.2205.
- 26. Bisen, V. and Srivastava, S. 2009. Production and Operations Management. India Global Media.



- 27. Balle, M. 2005. Lean attitude-Lean application often fail to deliver the expected benefits but could the missing link for successful implementations be attitude? Manufacturing Engineer, 84(2), 14-19.
- 28. Axelsson, B., Rozemeijer, F. and Wynstra, F. 2005. Developing sourcing capabilities: Creating strategic change in purchasing and supply chain management, John Wiley and Sons, West Sussex.
- 29. Miller, K.D. 2009. Organisational risk after modernism. Organisation Studies, 30,157-180.
- **30.** Houshmand, M. and Jamshidnezhad, B. 2006. An expected model of design process of lean production systems by means of process variables, Robotics and Computer-Intergrated Manufacturing, 22(1), 1-16
- **31.** Hines, P., Holweg, M. and Rich, N. 2004. Learning to evolve. International Journal of Operations and Production Management, 24(10), 994-1011. www.werc.org/assets/1/workflow_staging/publications/726. PDF. Accessed 15 April 2016.
- 32. Scherre-Rathje, M., Todd, A.B. and Patricia, D. 2009. Lean, take two! Reflections from the second attempt at lean implementation, Business Horizons, 52(1), 79-88.
- **33.** Beely, P. 2001. Foundry Technology 2nd Edition, Butterworth-Heinemann
- 34. Du Preez, W.B. 2006. Product development within the framework of a National Casting Technology Center. Journal for New Generation Sciences. 4 (1), 57-63
- 35. National Foundry Technology Network, 2009. South African Casting Industry Casting Benchmarking Report
- 36. Yin, R.K. 2009. Case Study Research, Design and Methods, Fourth Edition. Thousand Oaks, CA: Sage Publications.
- 37. Wan, J.P. and Hou, J.J. 2012. Research on SAP Business One Implementation Risk Factors with Interpretive Structural Model. Journal of Software Engineering and Applications, 5, 147-155. http://dx.doi.org/10.4236/jsea.2012.53022.
- **38.** Akugizibwe, A.M. and Clegg, D.R. 2014. Lean Implementation: An evaluation from the implementers' perspective. International Journal of Lean Enterprise Research, 132-161.



APPENDIX 1

LEAN QUESTIONNAIRE

EVALUATING LEAN IMPLEMENTATION IN SOUTH AFRICAN CASTING FOUNDRIES

The purpose of this study is to evaluate Lean implementation in South African Foundries. This questionnaire will gather information on how South Africa (SA) foundries have implemented lean.

The information provided will be used for research purposes only.

1. AIMS OF IMPLEMENTING LEAN

What were the main aims for implementing lean? On a scale of 1 - 5, kindly indicate by inserting X against the level of importance of the lean project implemented.

Scale: 1-Strongly disagree;

Cost reduction	1	2	3	4	5
Customer satisfaction improvement	1	2	3	4	5
improvement in flexibility	1	2	3	4	5
Increase staff motivation	1	2	3	4	5
Increase staff contribution in decision making	1	2	3	4	5
Reduction in the lead time	1	2	3	4	5
Reduction of workforce	1	2	3	4	5
Quality improvements	1	2	3	4	5

2. TOOLS OR TECHNIQUES USED TO IMPLEMENT LEAN

What tools and techniques were used during the implementation of the Lean? Please indicate the tools and techniques which were either used or not used during the lean project implementation.

Value stream mapping	1	2	3	4	5
Kanban	1	2	3	4	5
Six sigma	1	2	3	4	5
Visual management	1	2	3	4	5
Total Productive Maintenance	1	2	3	4	5
Error proofing	1	2	3	4	5
Workplace Organization (5S)	1	2	3	4	5
Standardized operation	1	2	3	4	5
Total Quality Management	1	2	3	4	5



Kaizen	1	2	3	4	5
Just-In-time	1	2	3	4	5

3. STRATEGIES USED TO ENGAGE STAFF IN THE LEAN PROJECT

What strategies were used to engage staff in the lean project? Kindly indicate your rating by inserting a number of your choice from 1 to 5 as guided by the scale below how successful they were.

Employee engagement in identifying solutions	1	2	3	4	5
Employee empowerment	1	2	3	4	5
Incentives	1	2	3	4	5
Emphasis on training and skill building	1	2	3	4	5
Employee ownership	1	2	3	4	5
Improved flexibility	1	2	3	4	5

4. PROGRESS MADE IN IMPLEMENTING LEAN MANUFACTURING

What progress has been made in implementing lean in your organisation? Kindly indicate your rating by inserting a number of your choice from 1 to 5 as guided by the scale below how successful they were.

Utilization of man power (indirect labour)	1	2	3	4	5
Reduction of defects per year	1	2	3	4	5
Productivity(sales/employee per year)	1	2	3	4	5
Increase in the current capacity	1	2	3	4	5
Quality improvement	1	2	3	4	5
Utilization of man power(direct labour)	1	2	3	4	5
Reduction of cycle time	1	2	3	4	5
Delivery lead times	1	2	3	4	5
Inventory reduction	1	2	3	4	5
Reduction in floor space	1	2	3	4	5
Machine setup reduction	1	2	3	4	5
On time delivery	1	2	3	4	5
Machine availability	1	2	3	4	5
Return on assets	1	2	3	4	5



5. BARRIERS OF IMPLEMENTING LEAN MANUFACTURING

What have been the main barriers of implementing lean manufacturing in your organisation? On a scale of 1 - 5, kindly indicate by inserting X against the barriers of implementing lean manufacturing in your organisation.

Backsliding to old ways of work	1	2	3	4	5
Lack of implementing know-how	1	2	3	4	5
Employee resistance	1	2	3	4	5
Middle management resistance	1	2	3	4	5
Not easy to implement	1	2	3	4	5
Lack of time	1	2	3	4	5
Lack of labour resources	1	2	3	4	5
Lack of a crisis to create sense of urgency	1	2	3	4	5
Lack of capital funds	1	2	3	4	5
Lack of idea generation/innovation	1	2	3	4	5
Failure of past lean manufacturing projects	1	2	3	4	5



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



TECHNOLOGY TRANSFER TO UPCOMING COMMERCIAL - COTTON FARMERS IN THE MAKHATHINI REGION

John Mashala

(UNISA) University of South Africa <u>mashajj@unisa.ac.za</u>

ABSTRACT

Abstract - Makhathini Farmers have produced cotton for over 10 years at subsistence level despite utilizing related technology extensively; the farmers' mission remains to produce cotton at commercial level. The technology mentioned in the study is available worldwide, as well as locally. The technology accessed is readily available to the local farmers. Against the above background the research problem was formulated as: "How is technology deployed to upcoming commercial - cotton farmers in the Makhathini region to support sustainable cotton production in this specific area. A stratified random sample was drawn from the upcoming commercial - cotton farmers based in Makhathini. To ensure representivity of data and to allow the researcher to generalize findings to upcoming commercial - cotton farmers in South Africa, factors such as farm size were considered. Representatives of 13 Famers' Associations were interviewed and a response rate of 13/13 (100%) was obtained. The findings indicate overwhelming utilization of technology by the farmers. Yet an issue that is unresolved is why farmers are not producing at a commercial level despite the aid of technology. Cronbach's Alpha is an instrument used to measure reliability. Calculating alpha has become widespread practice in comparison to other index of reliability. The alpha was used in SPSS programme.

'n Profiel van die katoen tegnologie gebruik in die Makhatini streek word aangespreek in die navorsings artikel. Die fokus van die artikel is gerig op die mate van ondersteuning beskikbaar aan die boere in die streek. Die klein boere in die streek verbou katoen die afgelope 10 jaar. Die verbouing van katoen is meer gerig op oorlewings boerdery maar dit is ,n oogmerk van die boere om in die toekoms katoen op ,n komersieële vlak te verbou. Die tegnologie waarna verwys word in die artikel is globaal sowel as plaaslik van toepassing in die verbouing van katoen. Die tegnology is vrylik beskikbaar aan die boere in die streek. Gestruktureerde onderhoude is met boere van 13 plaaslike Boere Verenigings gevoer. Die navorser het daarin geslaag om n 100% reaksie behaal met vraeluste se beantwoording. Die bevindinge dui daarop dat boere in die streek tegonologie gebruik in die verbouing van katoen. Die onbeantwoorde vraag is dan waarom die verbouing van katoen nie op n komersieële vlak plaasvind nie. Cronbach se Alpha kofficient instrument was gebruik om die betroubaarheid te meet. Dit is algemene gebruik in navorsing om die Cronbach Alpha aan te wend om vergelykings vir betroubaarheid te toets. Die Cronbach Alpha was gebruik deur middel van die SPSS program.

Key words: Accessibility of technology; adoption of appropriate technology; technology diffusion theory and transfer.



1. INTRODUCTION

The literature review will focus primarily on types of technology available in the cotton industry in South Africa. Comparison will be made of technology used in different countries. Modalities of technology transfer, Challenges or obstacles to utilizing technology faced by cotton farmers, and support mechanism for the formation of sustainable small and medium-sized for farmers to reach fully commercialized business will be outlined.

Cotton farmers in the South African Development Community, including the South African cotton farmers, face a common problem having to compete internationally. The advent of free-trade resulted in countries like China and India taking advantage of the market gap that exists

Cotton S.A has revealed its plans to support upcoming commercial - cotton farmers to produce 30% of the South African crop by 2014. Verryne[19], reports on ARC-IIC a five-point plan that has been drawn up:

• to familiarize upcoming commercial - cotton farmers with the nature and characteristics of cotton plant;

• to demonstrate best practice for farming management;

• to device a practical method of transferring information to all the farmers even if they are illiterate;

• to prepare the incumbent farmers with the basic knowledge for second level training which involves Sector Education and Training Authority courses in cotton cultivation and management Verryne[20] adds that farmers should be encouraged to move away from subsistence cotton farming to commercialize cotton farming.

2. THEORETICAL FRAMEWORK

This chapter will focus on theories that seek to explain the reason for utilizing technology within the context of upcoming commercial - cotton farmers among whom the empirical study was conducted. The results of the study conducted indicate that the farmers use technology, yet they are still farming at subsistence level. The theories will include among others: Theory of technology; Classification of technology; Framework for adopting technology and Comparative cotton technology. Application will be drawn from the following countries: United State of America, China, India, and South Africa.

2.1 THEORY OF TECHNOLOGY

Rijk[15], states that the need to improve agricultural labour productivity has in recent years become a forceful reality. Even more so, mechanization of agricultural labour productivity impacts on various aspects, namely:

• increased labour productivity where labour is substituted by machinery to facilitate cultivation of a larger area or serviced by the same labour force;

• increased labour productivity where there is faster turn-around time to achieve higher cropping intensity; and

• decreased costs of production where the machine introduced has a longer run at lower cost which offsets initial capital outlay.

2.1.1 Classifications of technology

Technology can be classified according to the following categories:

2.1.1.1 Codified technology

Business has to convert technology capabilities into saleable units that meet customer needs Pries & Guild[14]. For the benefit of the organizations, tacit knowledge can be packaged as technological terms of manual Hall[8]. Knowledge based on know-what, know-how and know-why form the domains of technology which is seen as standardized content which could be shared or used to develop and train personnel Hall[8].

2.1.1.2 New technology

According to De Coster & Butler[5] new technology refers to technology that is utilized for the first time by a firm or organization, even though it may have been in existence for long time. Its utilization impacts on the old way of achieving the company's objectives. Although it may have been on the market, it is new to the organization. Often its utilization impacts on cost and time-based competitiveness.

2.1.1.3 High technology



High technology companies generally focus on deploying an overall technology approach to business. They constantly seek to develop new technologies that are superior to those of their competitors. They have diverse technology portfolios. In particular, these organizations pursue technology with high intensity to gain technology-based competitive superiority Lichtenthaler[11].

2.1.1.4 Emerging technology

Emerging technology refers to technology that has not yet been utilized to its optimal potential or not commercially exploited. The use of it could be limited by certain government policy restrictions. The level of its utilization depends on the impact of knowledge proximity and economic stability of the country with emerging technology. The European Union has developed technology framework that is based on cognitive science, health-ageing and emerging knowledge-based economy. The policy makes provision for healthy ageing, namely extended healthy life years, and improved health care support system, improvement of quality of life of the elderly and reactivation of social participation Rijker-Defrasne et al[15].

2.2 APPROPRIATENESS OF TECHNOLOGY

According to Duan, Deng & Corbitt[6], appropriate technology is a concept more prevalent in Small and Medium Enterprises. These Small and Medium Enterprises different widely and have distinctive characteristics in the adoption of technology. They are generally more prone to taking risks in the adoption of technology rate of failure after adopting is high.

2.3 COMPARATIVE COTTON TECHNOLOGY UTILISATION

2.3.1 Technology utilization in United States of America

As the United States of America is a highly developed country in terms of technology utilisation, many countries benchmark use to evaluate areas that are similar. The level of technology utilization in the United States of America puts it at a competitive level in all the spheres.

2.3.1.1 Use of organic cotton

Pantagonia is a manufacturer, designer and distributor of top-level quality outdoor wear and sportswear. The firm was founded by a mountaineer called Yvon Chouinard. It is based in Ventura California in the United States of America. It sells sportswear for mountaineering, rock climbing, skiing, fishing, kayaking, long-distance running (marathons) and mountain biking. In 1994 the company decided to manufacture apparel from organically produced cotton cloth. In 1996 the firm used well certified cotton Casadesus-Masanelm et al[3].

2.4 TECHNOLOGY UTILISATION IN CHINA

2.4.1 Impact of use Bacillus theringieusis on cotton

2.4.1.1 Molecular marker technique

According to Preetha and Raveendren[13] molecular markers have played an important role in cotton-breeding research. There are various marker techniques. Some of them include: Restriction Fragment Length Polymorphism (RFLP), Random Amplified Polymorphic DNA), Amplified Fragment Length Polymorphism, Simple Sequence Repeats, Inter Simple Sequence Repeats and Sequence Related Amplified Polymorphism.

This technique refers to variations among individual cotton lengths of Deoxyribonucleic Acid clusters that are produced by restricted enzymes that cut Deoxyribonucleic Acid at selected sites.

2.5 TECHNOLOGY UTILIZATION IN INDIA

Application of bacillus theringieusis crops has helped to reduce chemical pesticide utilization. The application of this technology has had positive environmental, economic and health results Krisna & Qaim[10]. Emerging countries have experienced a decrease in marginal returns due to yield that has stagnated should look to adopting organic agriculture. The organic technology has been successful in India, Eyhorn, Ramakrishma & Mäder[7].



2.6 TECHNOLOGY UTILIZATION IN SOUTH AFRICA

2.6.1 Introduction of genetically modified cotton in South Africa

In South Africa, **genetically modified** cotton was introduced in 1997 Verryne[19], some African countries saw it as a panacea to ensure sustainable development. Reaction to argument ranges from outright rejection to enthusiastic acceptance. This technology has lively discussions all over the world. According to Yang[22] scholars in this field are widely optimistic about the benefits that GM technology holds for the agricultural industry. Sceptics deny the benefits could hold for developing countries, let alone small-scale farmers Clover [4]. Scepticism is unavoidable, but what does seem certain is the fact that the world population is increasing by 81million people per year. The total population is expected to increase from the present figure of 6.6 billion to 9 billion by 2050 Morse & Mannion[12].

2.7 FRAMEWORK FOR ADOPTING TECHNOLOGY

According to Sousa and Voss[18] technology is adopted for a number of reasons inter alia: products have reached maturity; the products are at the stage where sales are stagnating or close to becoming obsolete; inadequate technologies as well as market resources or access constraints; excessive product development costs; absence of sales techniques and market research; and an information gap between marketing and production. Singh, Garp & Deshmukh[16], postulate that market conditions usually dictate the parameter used to decide on strategies and priorities to be employed.

2.7.1 TECHNOLOGY TRANSFER MODALITIES

2.7.1.1 Mentorship

The mentorship programme for the Cotton industry was launched in 2006. Since then Cotton SA has been the accredited presenter of on-going skills training programmes to small-scale farmers. It does this to support the National Cotton Strategy (NCS). Accreditations of courses are granted by the Education and Quality Assurance (ETQA) body. Effective and fair standards are regulated by South African Qualification Authority (SAQA). The AgriSETA is responsible for funding the training programme.

2.7.2 Extension officers

The ARC-IIC has introduced a strategic plan to improve competitiveness among upcoming commercial - cotton farmers. This curriculum is aimed at training of the trainers. This means that the cotton extension officers and field technical advisers have to be trained to in turn train the farmers. Knowledge of pests, diseases and Integrated Pest Management (IPM), Best management practice, waste reduction, market access and strengthening of relationships with local stake holders. The aim is to improve efficiency, utilize resources efficiently and to become profitable Bruwer[1].

2.7.3 Training programme

The training programme entails basic unit standard of 10 units representing 43 credits. The units are grouped into four modules according to the growing season. Module I: covers: pre-planting activities, such as financial planning and management of agri-business. In this module, topics covered include: acquisition of production credits; basic business plan; cash flow management and record keeping.

Module II covers field preparation such as the establishment of cotton plant; application of fertilizers and weed control within six weeks after the first growth. Module III covers pest management, cotton and symbiotic insect life cycle, as well as symptoms and control of pests. Module IV covers preparation for harvesting, harvesting methods, limitation of contamination, grading and marketing. Bruwer[2] reports that up to 724 small farmers have received formal skill development certificates between Makhathini and Tonga which is situated in Mbombela. The credits involved certain number of units at NQF level 1.




Figure 1: A model of strategic planning for transferring technology Weiwei[21]

Weiwei[21], indicates that goals that are set should include not only current conditions but also Technology Organizing Environment (TOE) framework as a tool that could be considered for identifying factors influencing the adoption of technology to during the strategic planning process. Three factors are identified namely: technological factors, environment and the organization itself. The technological factors are those technologies that are relevant to the company. The environment is the macro area in which the company trades, competes and is subjected to government regulations.

3. RESEARCH DESIGN AND METHODS

This study intends to investigate the problem statement: "How is technology deployed to upcoming commercial - cotton farmers in the Makhathini region to support sustainable cotton production?" Based on the problem statement, a secondary question is phrased as:

To what extent are respondents aware of cotton technologies and information?

The principal objective is to make a theoretical and practical contribution to the situation of the upcoming commercial - cotton farmers and other stakeholders in the cotton supply chain with regard to using technology deployment as a means of facilitating quicker reaction to customer needs, meeting expectations and improving profitability.

3.1 SIGNIFICANCE OF THIS RESEARCH

The research is particularly important because it aims to identify factors hindering the deployment of technology to upcoming commercial - cotton farmers in the Makhathini region and to propose solutions to them. The solutions will be benchmarked against factors or models that have made other commercial farmers successful.

3.2 HYPOTHESIS

The first hypothesis test whether respondents are aware of technologies related to cotton picking, planting genetically modified cotton, use of integrated pest control and latest technologies applied by the cotton industry. The generic hypothesis can therefore be formulated as:



H₀: No significant difference exists between small-scale upcoming commercial - cotton farmers who are aware of cotton technologies and small-scale upcoming commercial - cotton farmers unaware of cotton technologies.

H1: More small-scale upcoming commercial - cotton farmers are aware of cotton technologies than those that are unaware.

The second hypothesis will test access to technology in terms of picking machines, modified seed, pest control systems and equipment.

H₀: No significant difference exists between small-scale upcoming commercial -cotton farmers that have access to technology and small-scale upcoming commercial-cotton farmers that do have access to technology.

H₁: More small-scale upcoming commercial - cotton farmers have access to technology than those that do not have access to technology.

The Cronbach Alpha reliabilities for technology awareness, and utilization were all above 0.85, the outcomes of the measuring instrument proved to be reliable. The reliability of the qualitative data was ensured by ensuring that the chairperson of each Farmers' Association was always present during the fieldwork to confirm or to provide a better perspective of respondents point views. The chairperson also played a useful role of interpreting areas of the questionnaire that were not clear to farmers.

3.3 RESEARCH DESIGN AND METHODOLOGY

The research will follow a survey research design to describe the conditions of the farmers. Based on the empirical research findings the researcher will be able to explain the farmers' level of preparedness to transform their operations from subsistence cotton production to commercial cotton production.

3.3.1 Description of population and region

Makhathini Cotton is a black economic empowerment firm with 65% black empowerment holders. In 2002 this firm invested about R50 million in a cotton ginnery. Currently there are approximately 90 to 120 small-scale farmers involved in this project. In 2002 the Kwazulu-Natal provincial government committed itself to providing production inputs to approximately 2 100 cotton farmers.

Smale et al[17], state that an estimated 3 000 farmers are organized into 33 farmers associations). It can therefore be assumed that the population size of upcoming commercial - cotton farmers in Kwazulu-Natal province is in the vicinity of 2 500. In the Makhathini region there are 30 cotton farmers' associations representing 2 000 farmers. As this study focuses on the Makhathini region a sample will be selected from this population.

3.3.2 Sample Frame

During the planning phase of the research, and with the support of the President of the Farmers' Association of the Makhathini region, the researcher was able to obtain access to the contact details of 13 Farmers' Associations representing a sample frame of 1 210 small-scale farmers' eligible for selection.

3.3.3 RESEARCH ASSUMPTIONS

If it is established from the findings that of the 121 farmers that were available to answer the questionnaire, 50% plus 1 utilized technology, a conclusion can be drawn that more farmers utilize technology in growing cotton.

4. RESEARCH FINDINGS

Figure 2, indicates that the majority of the respondents obtained their cotton seed from Cotton SA (n=56; 46%), whereas 39% (n=48) and 12% (n=15) both forming 51% representing the extent of technology (or genetically modified seed) that the respondents are getting from Monsanto and Vencan respectively. Figure 2, further indicates that the Kwazulu-Natal Cotton Growers Association is responsible for soil management and IPM (n=35; 29%).



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa $\ensuremath{\mathbb C}$ 2016 SAIIE



Figure 2: Access and assimilation of technology

The kind of training provided by both institutions named above is shown in Figure 2, and relates to the extent to which technology is utilized. Monsanto seems to be responsible for providing the greater extend of technology in the form of cotton seed, while the Kwazulu-Natal Cotton Growers Association provides guidance on soil management. Cotton SA serves as a link to pass on information technology. Figure 2, further indicates training provided by both the Kwazulu-Natal Cotton Growers Association and Cotton SA. However, 98% of the respondents have less than 8 years of formal education. This does not seem to be viable.

The overwhelming majority of the respondents (n=113; 93%) indicated to be in possession of cotton farming equipment. This gives an indication of the extent of self-reliance and availability of collateral security. There seems to be consensus among the respondents that the Government or Cotton SA should assign mentorship programmes to upcoming commercial - cotton farmers in Makhathini (Figure 2).

Response	Frequency	Percentage	Valid Percentage	Cumulative Percentage
Valid 1.00	117	96.7	96.7	96.7
2.00	3	2.5	2.5	99.2
4.00	1	.8	.8	100.0
Total	121	100.0	100.0	

Table 1: Utilisation of biologically modified seed

Table 1 shows that an overwhelming majority of respondents are utilising biologically modified seed, with frequency of (n=117; 96.7%). Relatively few almost negligible respondents still use conventional seed with a frequency of (n=3; 2.5%)



Variable	χ²	Sig.	χ² (Critical value)	Df	Ho
Use information and communication technology	0.997	0.05	7.82	3	Accept
Use mechanical cotton picking machines	3.1788	0.05	7.82	3	Accept
Plant biologically modified seed	1.000	0.05	7.82	3	Accept
Follow an integrated pest control strategy	0.999	0.05	7.82	3	Accept
Use mechanical devices to farm cotton	0.999	0.05	7.82	3	Accept
Process seed cotton	1.3751	0.05	7.82	3	Accept
Have the knowledge and skill to grade and classify cotton	7.2995	0.05	7.82	3	Accept
Access information on market prices on a regular basis	7.6101	0.05	7.82	3	Accept
Make use of training opportunities to support knowledge of farming	0.999	0.05	7.82	3	Accept

Table 2: Chi-square test for utilization of technology

Results in table 2 indicate that the H_0 hypothesis should be accepted, implying that small-scale upcoming commercial - cotton farmers do utilize technology to improve their cotton yield and the profitability of their farms.

5. CHALLENGES OR OBSTACLES TO UTILIZING TECHNOLOGY FACED BY COTTON FARMERS

> It was indicated that the current technology deployment process is constrained by an industry benefitting approach instead of a real focus on a commercialization transition that needs to take place at individual farming business level.

> An information gap exists between Cotton S.A sales forecasts and the Ubongwa Association currently active in the Makhathini region. Association leaders need to be taught forecasting techniques used and to understand how they can be used to provide reasonably accurate production estimates. Volume and market penetration take considerable time to develop.

 \succ Training of more Association leaders is recommended in order that these leaders could monitor sales trends and recommend the right time to sell cotton to the gins as international prices are used. At the moment there is only one trained person in this area.

6 SUPPORT MECHANISM FOR FORMATION OF SUSTAINABLE SME'S

Familiarize upcoming commercial - cotton farmers with the wonderful creation that the cotton plant is. Explain all the components of the plant and the development of the crop.

Devise a methodology to demonstrate best practices of cotton production and how each activity contributes to the yield that can be obtained by the producer. Device a methodology that not only literate, semi-literate as well as a person that did not have the opportunity to attend formal teaching would be able to understand. Preparation of the farmers with basic knowledge of cotton production to be ready to attend SETA courses in cotton cultivation and management. Lastly demonstrate good quality by differentiating between bad and good quality of cotton with samples.

7. CONCLUSIONS AND RECOMMENDATIONS

At this point, it is necessary to revisit the research questions posed in section 3. The principal research question was: "How is technology deployed to upcoming commercial - cotton farmers in the Makhathini region to support sustainable cotton production in this specific area?" Based upon the principal research question, the secondary question was formulated namely:

To what extent are respondents aware, have access and utilise cotton technologies?



An overwhelming majority of the respondents were aware and had access to available technology as per figure 2. However, the goal of commercialization of cotton eluded them. In the view of the researcher, too much emphasize was placed on the internal conditions of upcoming commercial - cotton farmers while too little emphasize is placed on the synchronization of internal and external conditions in which the farmers operate. Failure to synchronize these conditions could impede the transition of upcoming commercial - cotton farmers. This view is also supported by the theories put forward by Knott[9] discussed in the theory section. It is therefore recommended that deliberate plans be put in place to equip farmers for bigger transitional space to commercialisation. A paradigm shift in approach be adopted in which interventions focus on commercialization of individual upcoming commercial - cotton farmers to enable them to compete successfully in the competitive internal and external commercial market.

9. REFERENCES

[1] BRUWER, H. 2008. Control of cotton bollworms through intercropping. *Journal of cotton South Africa*, 11:22-24.

[2] BRUWER, H. 2009c. Cotton - an essential plant-based fibre. *Cotton South Africa*, July/Sep.:3-27.

[3] CASADESUS-MASANELL, R., CROOKE, M., REINHARDT, F. & VASISHTH, V. 2009. Households' willingness to pay for "green" goods: evidence from Patagonis' introduction of organic cotton. *Journal of economics and management strategy*, 18(1):203-233.

[4] CLOVER, D. 2010. Is BT cotton a pro-poor technology. A review and critique of the empirical record. *Journal of agricultural systems*, 10(4):482-509.

[5] DE COSTER, R. & BUTLER, C. 2005. Assessment of proposals for new technology ventures in UK: characteristics of university spin-off companies. *Journal of technovation*, 25:535-543.

[6] DUAN, X., DENG, H. & CORBITT, B. 2012. Evaluating the critical determinants for adopting emarketing in Australian small-and-medium sized enterprises. *Journal of management research review*, 35(3):289-308.

[7] EYHORN, F., RAMAKRISHNAN, M. & MÄDER, P. 2007. The viability of cotton-based organic farming systems in India. *International journal of agricultual sustainability*, 5(1):25-38.

[8] HALL, M. 2006. Knowledge management and the limits of knowledge codification. *Journal of knowledge management*, 10(3):117-126.

[9] KNOTT, P. 2009. Integrating resource-based theory in practice-relevant form. *Journal of strategy* and management, 2(2):163-169.

[10] KRISNA, V. V. & QIAM, M. 2012. Bt cotton and sustainability of pesticide reductions in India. *Journal of agricultural systems*, 107:47-55.

[11] LICHTENTHAILER, U. 2008. Role of corporate technology strategy and patent portfolios in low, medium and high technology firms. *Journal of research policy*, 38:559-569.

[12] MORSE, S. & MANNION, A. M. 2009. Can genetically modified cotton contribute to sustainable development in Africa? [Online]. Available from: <u>http://pdj.sagepub.com/content/9/3/225.refs.html</u> [Accessed on 02/08/2011].

[13] PREETHA, S. & RAVEENDREN, T. S. 2008. Molecular marker technology in cotton. *Journal of biotechnology and molecular biology review*, 3(2):32-45.

[14] PRIES, F. & GUILD, P. 2010. Commercialising inventions resulting from university research: analyzing the impact of technology characteristics on subsequent business models. *Journal of technovation*, 10:499-600.

[15] RIJK, A. 2011. Agricultural mechanization strategy. [Online]. Available from <u>http://www.anapcaem.org/publication/cigr-apcaem-website.pdf [Accessed on 19/10/2011.</u>

[16] SINGH, R. J., GARP, S. K. & DESHMUKH, S. G. 2008. Strategy development by SMEs for competitiveness: a review. *Journal of benchmarking: an international journal*, 15(5):525-528.

[17] SMALE, M., ZAMBRANO, P., GRUERE, G., FALCK-ZEPEDA, J., MATUSCHKE, I., HORNA, D., NAGARAJAN, L., YERRAMAREDDY, I & JONES, H. 2009. Measurung the Economic Impact of Transgenoc Crop in Developing



Agriculture during the First Decate: proceedings of Internationa Food Policy conference held in USA, on 20-23 May 2009. Washinton: D.C.

[18] SOUSA, R. & VOSS, C. 2008. Contingency research in operations management practices. *Journal of operations management*, 26:697-713.

[19] VERRYNE, T. 2005. Sharing experiences. Journal of cotton South Africa, 8(3):4-34.

[20] VERRYNE, T. 2006. Flowing from cotton. *Journal of cotton South Africa*, 9(2):4-34.

[21] WEIWEI, W. & BO, Y. 2008. Strategic planning for management of technology of China's high-tech enterprises. Paper presented at the PICMET proceedings, held in Cape Town, on 27-31 July, 2008. South Africa.

[22] YANG, L., PAN, A., ZANG, K., YIN, C., QIAN, B. &, CHEN, J., 2005. Qualitative and quantitative PCR methods for event-specific detection of genetically modified cotton Mon1445 and Mon531. *Journal of transgenic research*, 14:817-831.



PREVENTATIVE MAINTENANCE IN PUBLIC HOSPITALS

H. Steenkamp

Department of Mechanical and Industrial Engineering Technology

University of Johannesburg

rsteenkamp@uj.ac.za

ABSTRACT

This is a summary of eight research projects done by students as required for partial completion of their B-tech qualification. The purpose was to determine the manner in which maintenance was performed and to suggest possible improvements. Students evaluated radiology and diagnostics departments. The maintenance technicians were interviewed to determine the maintenance strategy and the effectiveness thereof. Nurses and a hospital representative were also interviewed in this regard. The purpose of maintenance is to keep equipment in working order and to prevent equipment failures. The maintenance strategy has a significant impact on the functioning of a hospital. Literature suggests that about one third of maintenance costs are wasted. Unexpected failure of equipment has a large impact on a hospital. The Government has a maintenance plan in place for public hospitals and every year a number of public hospitals are evaluated for compliance. The research indicated that most maintenance was managed by suppliers of equipment and that equipment was generally in good working condition. The current system can probably be improved on by using mobile devices for work orders, daily checks and uploading information onto a database.

Keywords

Preventative Maintenance, Public Hospitals, web-based systems, mobile devices



1. Introduction

In South Africa public health care is marked by an inadequate supply of medicine, deteriorating infrastructure, equipment failures, shortage of staff and misuse or misallocation of funds [1]. The quality of public health care differs greatly between the various hospitals and within certain areas. Rural areas in South Africa typically have the greatest constraints in all aspects but also in availability of equipment [2]. Benatar [3] claims that South Africa has the capacity to train skilled, caring practitioners who provide quality care with the highest professional standards. According to an article in Health 24 [4], the Free State lost 177 doctors due to difficult working conditions in 2015. One aspect that has a significant impact on operation of public health care is the inadequate functioning of equipment. It also has an impact on the working conditions of the staff. This paper focuses specifically on maintenance of equipment which will ensure that the necessary equipment is available when required.

An average to large sized hospital has 5 000 to 10 000 different types of medical equipment [5]. Statistics accumulated by The Joint Commission (TJC) [[6] Feb 2016) in America and Canada showed that faulty or absent medical equipment was involved in patient incidents that resulted in serious injuries or death. No information could be found with regard to the correlation between equipment failure and patient health in public hospitals in South Africa. It is critical to ensure that equipment is maintained for a reliable medical service [7]. The purpose of maintenance is to keep equipment in working order and to prevent equipment failures [8]. Careful management of the associated cost is especially important in public hospitals because of severe financial constraints

It is important to note that the cost of maintenance over the life span of medical equipment often exceeds procurement costs. Many modern medical devices are complex and sophisticated which further escalates their maintenance costs [9].

2. Objective

The purpose of this paper is to evaluate the manner in which maintenance is performed in public hospitals in Gauteng and suggest ways of possible improvement.

3. Literature Review

Rastegarie and Salonen [10] define maintenance as "the combination of all technical, administrative and managerial actions during the life cycle of an item intended to retain it in or restore to, a state in which it can perform the required function". Maintenance strategies can be based on several alternatives such as failure based, time-based, risk based and condition based [11]. Typically these strategies can be divided into two main categories, preventative or corrective maintenance [10]. Figure 1, illustrates the relationship between these two main categories. Organisations need to select a maintenance strategy that is optimal to the functioning of the company. Selecting the correct maintenance strategy can improve productivity and profitability.

Maintenance strategy selection can be considered as a complex multiple-criteria decision making problem [11]. Failure data needs to be collected for each piece of equipment in order to determine the effectiveness of the maintenance strategy. In addition consideration must be given to the available resources for these strategies.

It is important to measure maintenance performance so that it is possible to determine the value of maintenance performed [12]. Literature has identified [13] several different maintenance performance measures, these measures can be divided into three categories:

- Measures of equipment
- Measures of cost and
- Measures of process performance





Figure 1: Maintenance types [10]

3.1. Corrective Maintenance

One main approach to maintenance is corrective maintenance. Corrective maintenance is an approach whereby equipment is allowed to run to failure and then repaired. Corrective maintenance is also called reactive maintenance. This maintenance strategy is sometimes supported because it can be argued that no maintenance cost is incurred until the point of failure. Often organisations do not shut down equipment for repair as they have the perception that they do not have time for preventative maintenance. We will never be able to completely eliminate corrective maintenance but it needs to be minimised. [14].

It is important to ensure that the technicians responsible for reactive maintenance are well prepared. This includes ensuring that they have the correct tools, replacement parts, training and transport [15]. It is easier to manage reactive maintenance if technicians are assigned to do the maintenance of specific equipment. Consideration must also be given the process of dispatching work orders. It is also worth spending time to decide how work orders will be dispatched. Work orders should provide a detailed description of the work required and the priority of the work.

The consequences of failure should influence the decision of whether or not corrective or preventative maintenance is utilised for a specific piece of equipment [16]

3.2 Preventative Maintenance

Preventative maintenance is the alternative approach to maintenance is; see Figure 1. This type of maintenance requires maintenance to be completed before failure **occurs**. Most preventative maintenance is based on a predetermined interval irrespective of the condition of the equipment. These intervals can be time based or based on usage [11].

The maintenance cost is less than with reactive maintenance [17]. The extent of the damage of equipment that has failed is usually significant and the unplanned downtime can incur other unexpected costs. Preventative maintenance is however more difficult to coordinate because it necessitates careful planning.

There has been criticism of preventative maintenance as it includes implementation of unnecessary maintenance. The weakness is that it is seldom done at exactly the right time.

Advantages and disadvantages of preventative maintenance according to Hansen [17]:

Advantages:

- Cost effective in many capital-intensive processes
- Flexibility allows for adjustment of maintenance periodically
- Increased component life cycle
- Energy savings
- Reduced equipment or process failure
- Estimated 12% 18 % cost savings over reactive maintenance program



Disadvantages:

- Catastrophic failures still occur
- Labour intensive
- Includes performance of unnecessary maintenance
- Potential for incidental damage to components in conducting maintenance

It is therefore important to choose a successful maintenance program. Technicians should be well trained and scheduling should be done effectively. Preventative maintenance Includes condition based maintenance.

3.2.1 Condition Based Maintenance

Condition based maintenance can be defined as a set of maintenance processes and capabilities derived from real-time assessment of the equipment condition from various sources e.g. embedded sensors and external tests. The goal of condition based maintenance is to only perform maintenance when there is evidence of a need to do so and thus increasing the availability of the asset [18]. According to Rastegarie and Salonen [10] condition based maintenance falls under Preventative maintenance.

3.3 Improving Maintenance Effectiveness

Concepts for increasing the effectiveness of maintenance strategies that are frequently discussed in the literature are:

- Reliability Centred Maintenance (RCM) and
- Total Productive Maintenance (TPM).

3.3.1 Reliability Maintenance

Reliability centred maintenance, analyses the functions and potential failures of equipment with a focus on preserving system functions, rather than preserving the equipment. Reliability centred maintenance is used to develop scheduled maintenance plans that will provide an acceptable level of operation, with an acceptable level of risk, in an efficient and cost-effective manner [19]. The consequence of breakdowns is taken into account.

3.3.2 Total Productive Maintenance

Total Productive maintenance is designed to maximise equipment effectiveness by doing maintenance with as little as possible interruption of production, preferably doing maintenance during production. With this approach everyone is involved in the maintenance process, from top management to shop floor workers. [20]

3.3 Biomedical Technicians

In South Africa there are biomedical technicians and technologists as well as medical equipment maintenance professionals. These individuals are trained to maintain medical equipment. These careers are similar to biomedical engineers, clinical engineers or "medical equipment maintainers" these technician are referred to in other countries s

Malkin and Keane collected [21] evidence in a number of countries, to support the need for maintenance of laboratory and medical equipment in resource-poor settings. Engineering or engineering student volunteers visited 60 resource poor hospitals to gather data on out of service equipment in these hospitals. Of these hospitals only seven had some planned preventative maintenance and only three had complete preventative maintenance programs. The volunteers had a basic toolkit; they had access to internet and had to purchase replacement parts locally for no more than US\$50. They were allowed to **travel to get parts as long as they could use public transport.**

When analysing the repair request data it was determined that eighty nine percent of the repair requests were for medical equipment. The engineering volunteers were able to put seventy two percent of equipment back into service. This contradicts the belief that getting hold of spare parts is one of the major constraints as to why medical equipment is out of service in resource poor areas [21].

On analysis of the skills required to fulfil the repair requests an opportunity for a new curriculum for individuals to repair medical equipment (within different fields and different levels of complexity) was identified. Basic skills were defined as that of a person who could read, write and do math through fractions. These individuals could learn what is required for the successful completion of the desired task within 1-2 hours without them having any prior knowledge of medical equipment. It was found that 107 basic skills were used to put 66% of the medical equipment back into service in resource poor settings [21].



4. Methodology

Eight students from the University of Johannesburg went to five different medical institutions to research this topic of maintenance in public hospitals. The research project is required for partial fulfilment of their B-tech qualification. Differing methodologies were employed. Students used either a survey or studied available data or interviewed staff. Students surveyed the maintenance technicians. Others surveyed staff working with equipment or looked at maintenance records. Most students used a quantitative research approach. One student went to the radiology department at the University of Johannesburg. The remaining students went to four different public hospitals and a clinic. A radiology department, diagnostic equipment and other critical equipment were investigated. Of the equipment scrutinised some were maintained internally by maintenance technicians and others externally by the supplier. The capture and analysis of maintenance data was also investigated.

5. Findings

5.1. Maintenance of Equipment and Facilities in South African Public Hospitals

The maintenance of equipment and facilities in South African public hospitals is assessed according to the "Hospital Standards for Accreditation". This is a checklist completed by an inspector in order to determine whether or not hospitals comply with their requirements. The inspector is employed by the government in order to fulfil the Public Service commission (PSC) Act, 1997, Section 9[22]. After inspecting a hospital the inspector can suggest improvements or decide to continue or cancel the hospital's licence to operate. One of the aspects covered in this checklist is the maintenance plan of the hospital [23]. There are six main questions asked regarding maintenance:

- 1. The hospital has a maintenance plan.
- 2. The maintenance and repair manuals are in one location.
- 3. There is a master record of each piece of equipment in the hospital.
- 4. There is an action record that details the scheduled maintenance to be performed for each piece of equipment on a regular basis or schedule.
- 5. Defective parts for electrical systems are routinely inspected and replaced if indicated.
- 6. There are routine inspections, oiling and replacement of defective parts of mechanical systems.

5.2. Radiology Equipment

In the radiology departments, checklists need to be completed daily, monthly, quarterly, bi-annually and annually. Fluoroscopy has visual checks that have to be performed daily, monthly, quarterly and annually and sonography equipment has visual checks that have to be performed daily monthly and annually. The checklists are completed and kept as proof. If problems are detected during these checks necessary adjustments are made by the operator. For problems that the operator cannot correct, the maintenance technicians are contacted. In both radiology departments investigated equipment was regularly maintained as per the manufacturer's specifications. In interviews the radiographers in these departments indicated that they had never had a complete breakdown but they had experienced minor breakdowns that were repaired quickly and did not cause equipment to be out of service for any length of time. Several public hospitals are using their suppliers to maintain the equipment and this function is negotiated as part of the sale.

5.3. Medical Technicians and Assistant Technicians in Public Hospitals

Based on an interview with a medical technician at one of the public hospitals it became apparent that the hospital had a preventative maintenance program and the maintenance team was responsible for this program. The purpose of this program was early detection of problems and routine maintenance in the diagnostic department. The maintenance department set up the schedule and generated work orders; the completed work orders were then filed.

Most equipment was under a maintenance contract for maintenance by professional inspectors. The diagnostic imaging equipment had the most downtime due to load shedding. Not all public hospitals had an in-house maintenance department responsible for the proper continued operation of medical equipment in the hospital.

In the survey of hospital technicians it was identified that daily, weekly and monthly inspections were conducted. During these inspections technicians needed to determine whether there was malfunctioning of equipment. Data was captured manually which made it difficult for technicians to monitor life cycle and maintain traceability of equipment.



Operators' satisfaction was compared when machines were maintained in-house or by external contractors. Operators seemed to be able to make adjustments to equipment that was maintained in-house. The student found that operators were not satisfied with the maintenance whether it was done internally or externally.

5.4. Data Capturing and Planning

Most data was captured manually and filed. Daily, monthly, quarterly, bi-annually and annually checks were completed on forms and filed. None of the students found an electronic system in place. Reporting of maintenance problems was done by a phone call. The biomedical technicians were of the opinion that there would be many benefits when using an electronic system.

It could be possible to implement an electronic preventative maintenance program to assist hospitals in providing good quality services. To set up an electronic system to manage preventative maintenance the following would be required:

- Complete equipment inventory: The medical technicians in the hospital can assist with the identification of equipment that needs to be included in the inventory.
- Identification: It must be possible to distinguish equipment by its function. Identification could be developed in line with electronic system needs.
- Data of maintenance requirements as per supplier's instructions.
- Data of performance of equipment and failures that do occur.

In the public sector some clinics in Tshwane provided nurses and community healthcare workers with mobile devices sponsored by Vodacom. These devices cannot be used as a phone but has applications and data. An application was downloaded onto the device and this was linked to a database with all of the patients' information. When a healthcare worker examines a patient, the database can be accessed. The database provides some broad guidelines, e.g. this patient should be checked for TB. The healthcare workers can consult with the nurse via the system and make appointments to see the nurse for advice. When patients are released from hospital there is also a person who needs to inform the nurse that the patient is going home and that a healthcare worker needs to see them. Currently some problems have been encountered with this system as not all individuals are computer literate but it remains a very good system.

6. Conclusion

Of the public hospitals and clinics evaluated, preventative maintenance was used by most of them. In some areas the operators were not satisfied with the maintenance whether performed by internal or external personnel. Many of the institutions had an agreement with the suppliers to tend to the maintenance of the equipment.

There is currently no system to do maintenance performance measurement to determine the value of the existing maintenance strategy.

There are many benefits that can be realised from a web-based system specifically for maintenance of medical equipment. The cost can be minimised by looking for sponsors as was done with the healthcare workers. Training is extremely important. It cannot be assumed that all staff, whose responsibility it is to report faults of medical equipment, is computer literate. The employees need to be trained in the use of the web-based and mobile applications.

In conclusion it seems beneficial to develop a new curriculum for medical technicians that are not as extensive as the current curriculum requires and use these individuals to tend to the need in rural areas.

7. Recommendation

It is recommended that a system be developed or an existing system is bought to manage and maintain medical assets and equipment. Several local and international companies are developing systems that use mobile devices with web-based applications and databases to manage their assets. It would be possible to use similar systems with the same mobile devices that healthcare workers are using currently.

In such a system it will be easier to set up a maintenance schedule and assign tasks to the relevant technicians, depending on their skills and the task requirements. The technicians should download their tasks to their mobile devices and perform the maintenance. As maintenance is completed the task results must be uploaded. All failures and maintenance information (e.g. time taken to perform each task) will be available to determine whether the best asset management strategy is being followed. In this way the value of the maintenance can be determined.



A new curriculum could possibly be developed for medical technicians responsible for maintenance of medical equipment. Specialist advice could be made available via an electronic system. Hopefully these individuals could tend to some of the needs in the rural areas

A maintenance performance measurement system could be implemented. Such a system would also be easier to manage if an online system exists with maintenance data. This system should monitor value added by maintenance.

Further study should be done on the cost of using internal or external maintenance technicians and whether it truly is the best strategy to follow the manufacturer's guidelines for maintenance.

Further study should be done as to how the maintenance strategies in public hospitals can be improved and possibly standardised as there are quality issues with regard to maintenance.

8. Acknowledgement

The following students are acknowledged M. Kedibone; K. Khumalo; C. Makwela; N. Maibela; T.G., Sibanyoni; M. Sigidane; K. Swate and M.J. Tsukudu

9. Reference List

- Section 27 (2013): Monitoring Our Health: An analysis of the breakdown of health care services in selected 1. Gauteng facilities: A report for the period January- December 2012, March 2013
- 2. Health 24 (2012): http://www.news24.com/MyNews24/Healthcare-dilemma-in-South-Africa-20120507
- Benatar SR 2013: The challenges of health disparities in South Africa, South African Medical journal Vol 103 3. . No 3.
- 4. Health 24, 2015: http://www.health24.com/News/Public-Health/The-Free-States-public-healthcare-is-incrisis-20151021
- Stiefel R. H. (2009). Medical Equipment Management Manual 7th edition ed. AAMI: Annapolis Junction, MD 5.
- February 2016 Sentinel statistics 6. Joint Commission (TJC) Event quarterly https://www.jointcommission.org/assets/1/18/2004-2015_SE_Stats_Summary.pdf
- Kachieng'a M.O. (2002) : "Technology Management In The Public Health Sector: Professional View From 7. Equipment Maintenance Experts": South African Journal of Industrial Engineers, Vol13, No 1, 2002
- 8. Maletic, D., Maletic, M., Al-Najjar, B. & Gomiscek, B. (2014), The role of maintenance in improving company's competitiveness and profitability: a case study in a textile company, Journal of Manufacturing Technology Management, vol. 25, no. 4, pp. 441-456.
- 9. Jamshidi A, , Rahimi S.A. , Ait-kadi D; Bartolome A.R. (2014), Medical devices Inspection and Maintenance; A Literature Review; Proceedings of the 2014 Industrial and Systems Engineering Research Conference: 3895-3904
- 10. Rastegari, A., Salonen, A. (2013) Strategic maintenance management: Formulating maintenance strategy. COMADEM 2013. Helsinki
- 11. Shafiee M 2015: "Maintenance strategy selection problem: an MCDM overview", Journal of Quality in Maintenance Engineering, Vol 21 Iss: 4, pp. 378-402
- 12. Prajapati A., Bechtel J., Ganesan S. (2012) "Condition based maintenance: a survey" Journal of Quality in Maintenance Engineering, Vol 18 Iss:4 pp384-400
- 13. Parida, A., Kumar, U., (2006) Maintenance performance measurement (MPM): issues and challenges. Journal of Quality in Maintenance Engineering 12(1):7-20
- 14. Kumar, U., Galar, A., Stenstrom, C., Breges, L., (2014) Maintenance performance metrics: A State of the Art Review, Sweden
- 15. Cowley, M. (2013) "Embracing a reactive Maintenance management plan" Biomedical Instrument Technician Journal 22:76-85
- 16. Moubray, J. (1997) "Reliability-Cenetered Maintenance 2nd edition" Industrial Press Incorporated, New York 1997
- 17. Hansen, R.C., (2002), Overall equipment effectiveness. A powerful production/maintenance tool for increased profits. [Electronic] USA: Industrial Press Inc., U.S
- 18. Bengtsson, M., (2002) Condition based maintenance in technical systems, Mälardalen University, Eskilstuna, Sweden
- 19. Siddiqui, W.A., Ben-Daya, M. (2009)," Reliability Centered Maintenance. Handbook of Maintenance Management and Engineering", Springer Dordrecht Heidelberg, London, New York 20. Muyengwa, G. Marowa Y.N. (2015) "Analysing adoption of maintenance strategies in Manufacturing
- companies", IAMOT conference proceedings 2015
- 21. Malkin R, Keane A (2010) "Evidence based approach to the maintenance of laboratory and medical equipment in resource-poor settings" Medical Biology Engineering Comput (2010) Iss 48:pp. 721-726
- Commission 9. 22. Republic South Public Service 1997, Section of Africa act. http://www.psc.gov.za/documents/docs/legislation/act46-97.pdf



- Protocol on announced and unannounced inspections, August 2007, THE PUBLIC SERVICE COMMISSION (PSC)
 Vaisnys, P., Contri, P., Rieg, C. & Bieth, M. (2006) "Monitoring the effectiveness of maintenance programs through the use of performance indicators" European Commission Directorate-General Joint Research Centre, EUR 22602 EN



AN EVALUATION AND COMPARISON OF BUSINESS SIMULATION SOFTWARE

Jacobus Krige Visser

Department of Engineering and Technology Management, University of Pretoria, South Africa

krige.visser@up.ac.za

ABSTRACT

Various simulation programs are available today to perform business related simulation. A simulation model is useful for making decisions, e.g. in project, maintenance and operations management, or to understand the dynamic behaviour of business systems. This paper presents the results of a project to evaluate and compare a number of simulation programs. The output of four Monte Carlo simulation programs and two discrete event simulation programs were evaluated with the same input data and number of trials. The difference in output from the four simulation programs was found to be quite small. Engineers and managers can therefore use any these programs with confidence for similar type of problems as investigated in this study.



1. INTRODUCTION

1.1 Background

Various software packages are available today to perform business related simulation, e.g. @Risk [1], Crystal Ball [2], XLSim [3], SimVoi [4] and RiskAMP [5], or standalone software like Arena [6], GoldSim [7], BlockSim [8] and Raptor [9]. Simulation models are useful for making investment and other financial decisions, for optimising engineering designs or to understand the dynamic behaviour of business systems. These programs are often used in projects and in maintenance and operations management.

This paper presents the results of a study to evaluate and compare a number of simulation programs mentioned above. The functionality of simulation programs differ with more costly programs that usually have more functions for the user. The low budget options like RiskSim [4] and XLSim provide a limited number of probability distributions whereas @Risk and RiskAMP provide more than 40 built-in probability distributions.

Simulation programs can be applied to various business related problems and a comprehensive comparison of simulation programs would require a study of the performance and output of various problems. For this study only two cases were selected, i.e. the simulation of the total duration of a project when a number of tasks are performed in series or parallel and secondly a simulation of the availability of a system comprising a number of components in a series or parallel configuration.

1.2 Objective

The primary objective of this research study was to determine whether a number of simulation programs or add-ins for MS Excel provide similar outputs for the same inputs and number of trials. The use of spreadsheets with simulation add-ins to perform Monte Carlo simulation (MCS) to model project cost and schedule uncertainty has become quite popular in project management, risk management and financial management. It was therefore decided to model the total duration for a project comprising a number of individual activities to achieve the main object stated above. The following proposition was formulated.

The distribution for the total project duration if performed using the XLSim, RiskAMP or GoldSim simulation programs do not differ significantly from the distribution for the total project duration obtained with the RiskSim add-in software.

Two discrete event simulation programs, i.e. Raptor and BlockSim, were also applied to determine the availability of a multi-component system and the following proposition was defined.

The characteristics of the interval availability of a system as determined with the BlockSim software does not differ significantly from the availability as determined with the Raptor program when the same Weibull distributions are used as the failure distributions of the individual components.

2. LITERATURE

2.1 Monte Carlo Simulation

Monte Carlo methods were initially practiced under the names 'statistical sampling' or 'probability simulation'. Fermi used the method in the 1930's to calculate neutron diffusion in nuclear reactions. The name 'Monte Carlo Simulation' was used by Ulam and Von Neumann (Eckhardt, [10]) who modelled nuclear fission in the 1940's during the Manhattan research and development project. MCS is a 'static' simulation that uses random sampling to evolve an overall outcome distribution. It has become popular in project management to incorporate the uncertainty in cost or schedule in quantitative risk analysis (Schuyler [11], Grey [12]). The output in this application is typically a total cost or schedule distribution for the project.

MCS for modelling cost and schedule in projects has been used and applied by many project managers and researchers, e.g. Raydugin [13] explains how MCS could be used to determine a project cost reserve as well as a project schedule reserve or buffer. He also focuses on the value of MCS in decision making in projects. Cooper et al. [14] also discuss in some detail how the uniform and triangular distributions can be used to model general uncertainty in cost and schedule estimates as well as the impact of specific events on the cost and schedule distributions. He also explains how correlation can be included in simulations.

2.2 Discrete Event Simulation

A discrete event simulation models a system for which some variable 'changes only at a discrete set of points in time' (Banks et al. [15]). A repairable system could typically be in an operational or running state and when a failure occurs, it changes to a stopped or down state. When a repair or replacement is completed the system changes to a running state again. The interval availability of a repairable system is a function of reliability and maintainability and is determined by means of the uptime and downtime over a specific interval, e.g. one month or 720 hours of operation. A discrete event simulation (DES) is useful to model failures and repairs and



to determine the availability from the failure times and repair times. Two discrete event simulation programs were used to model a repairable system, i.e. Raptor [9] and BlockSim [8].

2.3 Probability distributions

2.3.1 Uniform distribution

The uniform or rectangular distribution is continuous and bounded on the left and right. It has two parameters, the lower bound, a, and the upper bound, b, and the range is therefore $a \le x \le b$. Random variates for the uniform distribution can be determined using the Excel function RANDBETWEEN(a;b) or the simulation software's built-in distribution function for the uniform distribution.

2.3.2 Triangular distribution

The triangular distribution is bounded on the left and right and can be symmetric or skew depending on the values of the three parameters that determine the shape of the distribution. The three parameters of the triangular distribution are typically:

- a = lower bound (minimum)
- m = most likely value (mode)
- b = upper bound (maximum)

The random variates for the triangular distribution can be determined from formulas provided in literature (Evans et al. [16]).

2.3.3 Normal distribution

The normal distribution, also known as the Gaussian distribution, is a continuous and symmetric distribution defined by two parameters, i.e. the mean value, μ , and the standard deviation, σ . The density function of the normal distribution is defined over $-\infty$ to $+\infty$ and care should be taken when using this distribution in simulation where the variable is only defined from 0 - ∞ , e.g. duration of an activity which cannot be negative. Random variates for the normal distribution can be calculated with the NORMINV(RAND(), μ , σ) function of Excel.

2.3.4 Lognormal distribution

The lognormal distribution is a continuous, non-symmetric distribution that is often used to model the duration of activities or tasks. It applies mostly to novice artisans or workers that have to perform non-standard and complex tasks. These tasks often take longer especially when something goes wrong. It has two parameters, i.e. the mean value, μ , and the standard deviation, σ . Random variates can be calculated using the Excel built-in function LOGNORM.INV(RAND(), μ , σ).

2.3.5 Gumbel distribution

The Gumbel distribution is continuous and also known as the Smallest Extreme Value (SEV) distribution (Type I) as well as the logWeibull distribution. It has a location parameter $\mu > 0$ and a scale parameter $\beta > 0$. A random variate, T, for the Gumbel distribution was calculated using equation 1. This distribution is mainly used in the analysis of extreme values, e.g. flood levels and earthquake damage.

$$T(\rho, \mu, \beta) = \mu - \beta \cdot LN\left[LN\left(\frac{1}{\rho}\right)\right]$$

(1)

(2)

where ρ is a random number.

2.3.6 Weibull distribution

The Weibull distribution is well known for modelling reliability of physical assets and humans due to its versatility with regard to failure rate. The 'shape' parameter, β , determines whether an item exhibits a decreasing, constant or increasing failure rate. The second parameter, η , is known as the 'characteristic life' when used in reliability applications.

The Weibull distribution is also useful to model task or activity duration in projects since it is one of a few distributions that is skewed towards the left, i.e. a negative skewness factor, if the shape parameter is larger than 1. The general assumption is that the distribution of task duration is skewed to the right and therefore some historical data is needed on specific task durations to warrant the use of the Weibull distribution. A random variate, T, for the Weibull distribution was calculated using equation 2.

$$T(\rho,\beta,\eta) = \eta \cdot \left[LN\left(1-\frac{1}{1-\rho}\right) \right]^{\frac{1}{\beta}}$$

where ρ is a random number.



2.3.7 Gamma distribution

The gamma distribution is continuous over the range 0 - ∞ with scale parameter, b > 0, and a shape parameter, c > 0. Random variates can be determined using the Excel built-in function GAMMA.INV(RAND(),b,c) or the simulation software's own defined function if available.

2.4 Comparison of simulation programs

Willis [17] performed a study in 2000 to compare a number of simulation programs for reliability and availability simulation. Ten capabilities that were regarded as important for mission reliability simulation were defined and nine simulation programs were tested against these criteria. This study is quite old now and many of the simulation programs might not be available any more. Some of the popular software that that are still available today are AvSim, BlockSim, Relex RBD and Raptor. The software was not compared in terms of the output of simulations.

3. METHODOLOGY

3.1 Overview

A Monte Carlo simulation can be done with MS Excel without additional software or add-ins using standard functions provided, e.g. RAND() for generating random numbers between 0 and 1 and FREQUENCY(Array1,Array2,) to produce a histogram of random variates. The variates can be sorted from smallest to largest and cumulative distribution values can be determined. However, the simulation add-ins or standalone software that are commercially available automate the simulation process and is more convenient and faster to use.

Two project activity networks were selected for Monte Carlo simulation study, i.e. a 'series' network where all activities are performed in sequence and a 'parallel' network where some activities take place in parallel.

3.2 Schedule simulation for a series network

Eleven activities that are performed in sequence were selected and the parameters used for the different distributions are shown in Table 1.

Activity	Triangular		Unif	orm	Normal		Gumbel		
ACTIVITY	а	m	b	а	b	μ	σ	μ	β
А	5	7	14	5.325	12.008	8.667	1.929	7.798	1.504
В	5	6	12	4.990	10.344	7.667	1.546	6.971	1.205
С	6	8	16	6.258	13.742	10.000	2.160	9.028	1.684
D	5	7	14	5.325	12.008	8.667	1.929	7.798	1.504
Е	4	5	10	4.060	8.606	6.333	1.312	5.743	1.023
F	6	7	14	5.918	12.082	9.000	1.780	8.199	1.387
G	6	8	16	6.258	13.742	10.000	2.160	9.028	1.684
н	7	10	18	7.646	15.687	11.667	2.321	10.622	1.810
J	5	6	12	4.990	10.344	7.667	1.546	6.971	1.205
к	6	7	14	5.918	12.082	9.000	1.780	8.199	1.387
L	7	10	20	7.520	17.147	12.333	2.779	11.083	2.167

Table 1. Distribution parameter values for activity duration for a series network

The 'most likely' or mode values (m) for the triangular distribution were randomly selected between 5-10 time units (typically days). The lower bound (a) and upper bound (b) were chosen to provide a right skewed distribution. This is quite typical for the subjective estimates of task duration in projects since tasks sometimes take much longer than the average and would very rarely take much quicker to complete.



- SAIIE27 Proceedings, 27^{th} 29^{th} of October 2016, Stonehenge, South Africa $\mbox{\sc c}$ 2016 SAIIE
- 3.3 Schedule simulation for a parallel network

The project network with parallel activities that was used in this study is illustrated in Figure 1 below.



Figure 1. Activity network for parallel case investigated

The dependency of activities that are performed in sequence and parallel should be accommodated in the Excel spreadsheet. For example, the start time for activity G is the maximum finish time of activities E and F. Similarly, the project is only finished when the activities G, J and L are all finished.

The output of the four simulation programs were obtained for four different input distributions. The mode values for the triangular distribution were also selected randomly between 7-15 time units and values for the parameters a and b were chosen to give a right skewed distribution. The values used for the other distributions were calculated from the mean and standard deviation values as shown in Table 2.

Activity	Т	riangula	ar	Logno	ormal	Weibull		Gam	nma
ACTIVITY	а	m	b	μ	σ	β	η	b	С
А	13	15	18	15.333	1.027	15.783	18.450	222.737	0.069
В	8	9	12	9.667	0.850	10.035	13.910	129.385	0.075
С	6	8	10	8.000	0.816	8.351	11.896	96.000	0.083
D	7	9	13	9.667	1.247	10.194	9.285	60.071	0.161
Е	9	12	17	12.667	1.650	13.363	9.192	58.939	0.215
F	11	13	17	13.667	1.247	14.206	13.377	120.071	0.114
G	6	7	10	7.667	0.850	8.030	10.905	81.385	0.094
н	9	12	18	13.000	1.871	13.783	8.265	48.286	0.269
J	5	7	10	7.333	1.027	7.764	8.505	50.947	0.144
к	13	16	20	16.333	1.434	16.955	13.932	129.784	0.126
L	10	13	18	13.667	1.650	14.368	9.964	68.612	0.199

Table 2. Distribution parameter values for activity duration for a parallel network

3.4 Discrete event simulation

A system with 20 components was defined to test the two software programs that performed the discrete event simulations, i.e. Raptor and BlockSim®. The reliability block diagram for this system is shown in Figure 2 below.



Figure 2. Reliability block diagram for 20 items



A Weibull distribution was selected to describe the failure behavior of all 20 components. The shape parameter for the Weibull distribution, β , was randomly selected between 2 and 6 and the characteristic life, η , was randomly selected between 5000 and 15000 hours. A normal distribution with a mean replacement time of 50 time units and standard deviation of 5 was chosen to model the repair behavior of all the components.

4. RESULTS

4.1 Overview

The RiskSim and GoldSim software provide the raw data, i.e. the random variates calculated for the total project duration, as well as percentiles at 5% intervals and other statistical data. The XLSim and RiskAMP software do not provide the raw output data but the percentiles at 5% intervals and some statistical data are provided. The values of the total duration at 5% intervals of the percentiles were therefore used as comparison of the output of these software programs.

Most software also provides some statistical data of the output distribution. RiskSim and GoldSim also provide graphical output, typically the probability density and cumulative distribution function of the output distribution.

The results are provided separately for the two cases, i.e. the series network and the parallel network. The descriptive statistics are given for the RiskSim add-in only since GoldSim does not provide all this detail. The P80 and P90 duration values are compared against the triangular distribution to assess the effect of the tails of the input distributions.

4.2 Distributions provided by software

Simulation programs provide a number of distributions to be used for uncertain inputs, e.g. the cost or duration of an activity in a project. Table 3 shows the functions included in the software and that were used in this research project. The different distributions in the standalone GoldSim software are just selected by name from a drop-down menu.

Distribution	RiskSim	XLSim	RiskAMP	GoldSim
Uniform	RandUniform	gen_Uniform	UniformValue√	\checkmark
Triangular	RandTriangular	gen_Triang	TriangularValue	\checkmark
Normal	RandNormal	gen_Normal	NormalValue	\checkmark
Lognormal	RandLognormal	gen_Lognormal	LognormalValue	\checkmark
Weibull	х	х	WeibullValue	\checkmark
Gamma	х	х	GammaValue	\checkmark
Gumbel	х	х	х	\checkmark

Table 3: Distributions provided by software

For those distributions not provided by the specific add-in software, the built-in functions of Excel were used with the function for the uniform distribution to provide a random number between 0 and 1. For the Gumbel distribution the formula in equation 1 was used to calculate the random variates.

4.3 Simulation software for project duration

4.3.1 Project with activities in series

The descriptive statistics for the total duration as provided by the simulation software are given in Table 4 for the triangular distribution only.



			-	
 Parameter	RiskSim	XLSim	RiskAMP	GoldSim
 Mean	101.027	101.042	100.951	101.108
Std Dev	6.527	6.560	6.510	6.515
Skewness	0.133	N/A	0.126	0.182
P25	96.488	96.469	96.445	96.533
Median	100.879	100.937	100.791	100.824
P75	105.432	105.423	105.350	105.408

SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE Table 4. Statistics for total duration distribution for triangular distribution inputs

The values for the four software programs agree very well and the only noticeable difference is the skewness that varies from 0.126 - 0.182. This could be due to different random number generators used by the software or the interpolation algorithm. The skewness value is not provided by the XLSim software.

The total project duration using the RiskSim add-in was used as reference and the difference in duration was determined with the XLSim, RiskAMP and GoldSim software at 5% intervals of the percentiles. A graphic result for the triangular distribution as input is given in Figure 3.



Figure 3. Difference in total duration with triangular distributions as input

From Figure 3 it is seen that the difference in total duration is very small in the 20-80% region, typically less than about 0.1%. In the regions 5-20% and 80-95%, the difference is larger but still less than about 0.3%. This was expected since different simulation software use different algorithms for random number generation as well as different interpolation and extrapolation algorithms. The difference in total duration for the other distributions was also less than about 0.3% over the 5-95% range. The RiskSim and GoldSim programs produce the values of the random variates and a two-sample t-test with unequal variance was performed on this data to compare the output distributions for the triangular distribution as input. The output distributions are so similar that a p-value of 1.0000 was obtained with the t-test. This indicates, at a 95% confidence level (p-value > 0.05), that the two distributions are not significantly different.

The P80 and P90 values are of particular interest to the project or risk manager and are often used for reporting the certainty duration values at various stages of a project. The comparison of P80 values for the series network is shown in Figure 4 and the comparison of P90 values in Figure 5.





Figure 4. Total duration for 80% certainty and different distributions as input

The total duration at P80 for the four software programs and the triangular and uniform distributions are very similar. There is slightly more variation for the normal and Gumbel distributions. It should be remembered that the normal distribution is a symmetric distribution while the input distributions were non-symmetric.



Figure 5. Total duration for 90% certainty and different distributions as input

The total duration at P90 shows more variation than at P80 as could be expected. The GoldSim software gave a noticeably longer duration than the other three programs but only 10000 trials were used with GoldSim while 20000 trials were used with the other programs. It is interesting to note that the duration values were very similar for all four programs when the normal distribution was used.

4.3.2 Project with activities in parallel

The simulations were repeated for the parallel network and the descriptive statistics for the total duration distribution are given in Table 6 for the triangular distribution only.

Parameter	RiskSim	XLSim	RiskAMP	GoldSim
Mean	55.013	55.031	54.969	55.010
Std Dev	2.721	2.724	2.718	2.724
Skewness	0.103	N/A	0.038	0.079
P25	53.096	53.137	53.040	53.049
P50	54.954	54.962	54.918	54.966
P75	56.873	56.884	56.849	56.926

Table 6: Statistics for total duration for parallel network with triangular distribution inputs

The values obtained with the four different simulation programs are in good agreement, except for the skewness that varies from 0.038 - 0.103. The skewness value is not provided by the XLSim software.



The total project duration using the RiskSim add-in was used as reference and the difference in duration was determined for the XLSim, and GoldSim programs at 5% intervals of the percentiles. The graphic result for the triangular distribution only is given in Figure 6.



Figure 6. Difference in total duration with different software with triangular input distributions

As was noticed for the difference between software programs for the series network, the difference for the parallel network is also quite small, i.e. less than about 0.1% for the 20-80% region. For the 5-20% and 80-95% regions, the difference is larger but still less than about 0.3%. The difference in total duration for the other distributions was also less than about 0.3% over the 5-95% range.

The comparison of P80 values for the project with some parallel activities is shown in Figure 7 and the comparison of P90 values in Figure 8.





Figure 7. Total duration for P80 with different software and different distributions as input

The total duration values for the four simulation programs investigated are quite similar with the largest differences noted for the lognormal and Weibull distributions. It is also interesting to note that one specific program did not have the largest or the smallest values for all four distributions.





Figure 8. Total duration for P90 with different software and different distributions as input

The P90 values agree remarkably well for this case. Even the difference in total duration is quite small for the different input distributions although there is more variation than for the P80 values which is to be expected. The best agreement between the different software and different distributions is obtained in the 20-80% range. One might have expected that the different skewness and tails of these distributions would cause a notable difference in total duration.

The results for the series and parallel networks support the proposition that the output distribution for the XLSim, RiskAMP and GoldSim programs do not differ significantly from the distribution for the RiskSim program for all the different distributions investigated.

4.4 Software for reliability and availability of a system

The Raptor and BlockSim programs were used to perform 1000 trials at time intervals of 720 hours which was taken as one month of operation. One output of both programs is the average availability since the start of operations. This was used to calculate the interval availability at each time interval of 720 hours. Only the Weibull distribution was used for each of the 20 components of the total system. The interval availability for the total system for both software programs is shown in Figure 9.



Figure 9. Availability comparison of Raptor and BlockSim simulation programs

The interval availability plots for the two simulation programs agree fairly well but there are some minor differences. The zig-zag pattern observed with the Raptor software was not as prominent with the BlockSim software. At times the maxima and minima for the two plots coincide, e.g. at 32, 38, 41 and 52 months for the



maxima. However, there are examples where the minima and maxima do not occur at the same time, e.g. at 30, 44 and 54 months. This could be due to different random number generators used by the two programs.

Descriptive statistics of the availability plots were determined for the steady state period of 18-56 months and are shown in Table 7.

Parameter	Raptor	BlockSim
Ao (average)	0.929	0.929
Std Dev	0.0077	0.0057
Ao (minimum)	0.914	0.917
Ao (maximum)	0.943	0.943

Table 7. Statistics for availability values obtained with Raptor and BlockSim software

The average, minimum and maximum values for availability for the two software programs agree very well and the only major difference seen in Table 7 is in the standard deviation. The standard deviation with the BlockSim software was 26% less than the standard deviation as determined with the Raptor software. The variation of the interval availability values around the mean value when steady state was achieved were used to perform a two sample t-test with unequal variance and a p-value of 0.83 was obtained. This supports the proposition that the results with the two simulation programs are not significantly different.

The random variates for the Weibull distribution are calculated by means of the inverse of the cumulative distribution function (equation 2) and both software programs would use the same formula. The difference in standard deviation might be due to different random number generators used by the software. The BlockSim software uses the Mersenne Twister pseudo random number generator (Matsumoto & Nishimura [18]). The random number generator algorithm for the Raptor software is not known.

5. CONCLUSION AND RECOMMENDATIONS

5.1 Conclusion

The distribution for the total project duration for a project with eleven activities in series as well as a project with some parallel activities was determined using four different Monte Carlo simulation programs and seven different input distributions. The results indicate that there is very little difference in the outputs of the four simulation programs even though only 10000 trials were used for the GoldSim program as opposed to the 20000 trials used for the RiskSim, XLSim and RiskAMP add-ins. The values for total project duration differed by less than about 0.3% over the 5-95% percentile range. Over the 20-80% percentile, range the difference in total duration is less than about 0.1%. The difference in total project duration for the 95-100% percentiles was not determined since extrapolation from a limited number of data points is inaccurate and should not be used by project and risk managers.

The results confirm the proposition that different simulation add-ins for Excel or standalone Monte Carlo simulation software can be used with confidence by project and risk managers for project cost and schedule simulation. The subjective estimates by experts for the parameters of input distributions are far more uncertain than the small differences obtained for different simulation software.

Two discrete-event simulation programs were also compared by determining the interval availability at intervals of 720 hours for a multi-component system. A two sample t-test for the difference between the availability as determined with the simulation and the theoretical availability confirmed the proposition that the results of the two programs are not significantly different.

5.2 Recommendations

Two test cases, one for a series and one for a parallel network were used to test the proposition. These test cases should also be investigated with highly skewed distributions and perhaps higher standard deviations. Right skewed triangular distributions where the value of (b-m) is 4 to 5 times the value of (m-a) could be assumed for the input and the output of the software compared.

There are also other add-ins or standalone simulation programs that could be tested, e.g. @Risk, Crystal Ball and RiskSolver. It could also be interesting to run a larger number of trials, e.g. 50000 for the spreadsheet simulation and 5000 for the discrete event simulation, and the output compared for the difference software.



REFERENCES

- [1] Palisade, @Risk, http://www.palisade.com/risk/. (Accessed 31 May 2016).
- [2] Oracle, Oracle Crystal BallTM, http://www.oracle.com/us/products/applications/ crystalball/overview/index.html (Accessed 31 May 2016)
- [3] XLSim Simulation Software, http://software.informer.com/search/analycorp+xlsim (Accessed 24 May 2016).
- [4] Treeplan Software, SimVoi[™] Monte Carlo Simulation Add-in, http://simvoi.com (Accessed 24 May 2016).
- [5] RiskAMPTM Simulation Software, https://www.riskamp.com (Accessed 31 May 2016).
- [6] Arena[®] Simulation Software, https://www.arenasimulation.com/ (Accessed 24 May 2016).
- [7] GoldSim Technology Group, GoldSim[™], Monte Carlo Simulation software for decision and risk management. http://www.goldsim.com/Web/Products/ (Accessed 31 May 2016).
- [8] Reliasoft, BlockSim[®], http://www.reliasoft.com/BlockSim/index.html (Accessed 24 May 2016).
- [9] Booz|Alan|Hamilton. Raptor Reliability Simulation Software, http:// www.boozallen.com/consulting/products/raptor (Accessed 30 May 2016).
- [10] Eckhardt, R. 1987. Stan Ulam, John von Neumann and the Monte Carlo method. *Los Alamos Science*, Special Issue.
- [11] Schuyler, J. 2001. *Risk and Decision Analysis in Projects*. 2nd Edition. Project Management Institute.
- [12] Grey, S. 1995. Practical Risk Assessment for Project Management. John Wiley & Sons, Chichester, England.
- [13] Raydugin, Y. 2013. Project Risk Management: Essential Methods for Project Teams and Decision Makers, John Wiley & Sons, New Jersey.
- [14] Cooper, D., Bosnich, P., Grey, S. Purdy, G., Raymond, G., Walker, P. and Wood, M. 2014. Project Risk Management Guidelines. Managing Risk with ISO 31000 and IEC 62198. 2nd Edition, John Wiley & Sons, United Kingdom.
- [15] Banks, J., Carson, J.S., Nelson, B.L. and Nicol, D.M. 2005. *Discrete event system simulation*. 4th Edition, Prentice-Hall.
- [16] Evans, M., Hastings, N and Peacock, B. 2000. Statistical Distributions, 3rd Edition, John Wiley & Sons, New York
- [17] Willis, R. Society of Reliability Engineers, Comparison of Reliability-Availability Mission Simulators. www.sre.org/pubs/simul8a.doc (Accessed 31 May 2016).
- [18] Matsumoto, M. & Nishimura, T. 1998. Mersenne Twister: A 623-dimensionally equidistributed uniform pseudo-random number generator. ACM Transactions on Modeling and Computer Simulation 8, pp. 3-30.



IDENTIFYING BARRIERS FACED BY KEY ROLE PLAYERS IN THE SOUTH AFRICAN MANGANESE INDUSTRY

H.J. van Zyl^{1*}, W.G. Bam² & J.D. Steenkamp³

¹Department of Industrial Engineering University of Stellenbosch, South Africa 16593588@sun.ac.za

²Department of Industrial Engineering University of Stellenbosch, South Africa Wouterb@sun.ac.za

> ³Pyrometallurgy Division MINTEK, South Africa joalets@mintek.co.za

ABSTRACT

South Africa has abundant manganese ore deposits. Yet, it fails to capitalize on this resource and reach possible levels of economic benefit in this commodity's industry. This paper aims to identify the barriers to economic growth for key role players in the manganese industry. The manganese value chain is mapped out to determine the different manufacturing processes, role player dependencies and factors shaping the value chain. Representatives from the various sectors in the value chain were interviewed in order to identify, categorise and rank the barriers according to their impact on economic growth. This enables areas for improvement to be identified in order to promote the development of the local manganese industry.

¹ The author was enrolled for an M Eng (Engineering Management) degree in the Department of Engineering, University of Stellenbosch.

² The author was enrolled for a D Phil (Engineering Management) degree in the Department of Engineering, University of Stellenbosch.

³ The author holds a Ph.D. in Metallurgical Engineering conferred by the University of Pretoria. *Corresponding author



1. INTRODUCTION

The manganese industry is an integral part of South Africa's well-established mineral and commodity sector. It is well endowed with natural reserves and has developed local technical expertise in producing value-added products i.e. manganese ferroalloys since 1937 [1]. Together with many other well-established mineral industries in South Africa, the manganese industry has served as a catalyst for the advancement of an extensive physical infrastructure in the country and similarly made significant contributions to research and development, technologic advancements and market relationships throughout the various sectors in the manganese value chain.

South Africa's vast mineral wealth has long served as an attractive option for large capital investments. This natural wealth is the driving force for the development of an integrated mining industry, due to the sharing of infrastructure and the overlap of shareholder interests. The mining industry has formed a dominant part of the country's economy and is responsible for revenues earned through trade in global markets [2]. The establishment of mining-related activities, together with the promotion of downstream beneficiation by government through various policies and strategies, has placed focus on unlocking economic value further downstream from mining to other activities, such as further processing, refining and fabrication of high value products [2]-[4].

The steel and manganese industries are closely related and regularly influence each other. The outlook of the steel industry is essential for manganese for the following four reasons:

- 1. Manganese is primarily used in steel manufacturing, which accounts for approximately 90 % of the total manganese demand [5]-[8].
- 2. It is used as an alloying material.
- 3. There are no suitable alternatives for manganese in steel production [5], [8], [9].
- 4. Vertical integration in the global manganese value chain (which means leading steel companies are to some degree involved in manganese production) [5], [10].

South Africa is a leading player in the international manganese ore and manganese ferroalloys industry. It is the largest producer and exporter of manganese ore and a major role player in ferromanganese production and global trade of these alloys [1], [11]. The abundance of natural resources, relatively low electricity costs and cheap labour are historical factors that have contributed to this dominant position in the manganese value chain [1], [12].

There are numerous factors that have progressively grown during the past couple of years which has had detrimental effects on the manganese industry and threatened the country's dominant position. These factors include [1], [12]:

- 1) New legislation affecting the mining and minerals industry;
- 2) A shortage of local suitable reductants;
- 3) Depletion of surplus electricity generation capacity with resultant threat of escalation of electricity prices, and;
- 4) Pressure resulting from the recent fall in the manganese commodity market.

Furthermore, the manganese ore and alloy production capacity added during the commodity boom cycle (between 2001 and 2007 when the manganese commodity cycle was at a peak [13]) has declined, resulting in the current market remaining oversaturated and in ample overcapacity of manganese ore and alloys [14]. As a repercussion, manganese is presently one of the worst performing commodities with prices at its lowest point, not seen since the 90's [14].

It is evident that the manganese industry like so many other metal and mineral industries, has entered an extremely difficult period. This has led to the industry shifting focus to address challenges not only in terms of short term gain, but to ensure long-term viability [1]. These measures include appealing for a review of specific legislative policies in the mining and mineral market to improve support from government, improvement of operational efficiency, development of port and rail infrastructure, as well as initiatives to expand electricity generation and coke production capacity [1].

1.1 Research objective

The aim of this study is to establish the importance of various factors that impede economic growth throughout the value chain in the South African manganese industry. This aim is supported by three research objectives, namely:

- 1. Determine the different stages and branches of the South African manganese value chain;
- 2. Identify the barriers faced by the role players within each of these stages and branches with regards to economic growth, and;
- 3. Rank these barriers according to their impeding effect on economic growth.



1.2 Literature gap to be addressed

The published literature pertaining to the South African manganese industry provides a detailed layout of the manganese value chain and how the country relates to the global market [1], [5], [10], [12]. Much of the literature, however, is slightly outdated, relying on sources published before the sharp rise in electricity tariffs were implemented, the regular occurrence of unreliable power supply, the manganese commodity cycle facing a record low point, increased labour disputes in the mining sector, and government policies implemented to support beneficiation initiatives. All of which gained traction in the mid to late 2000's.

Furthermore, there is a void in this research pertaining to the major problems and constraints faced by the different role players in this industry with regards to economic growth and the impact that these constraints have on their businesses. There also is a lack of information pertaining to the impact that these role players exert one another and how these constraints affect their relationships. At present, there is no research available where the main barriers to economic growth of individual role players in the manganese value chain is identified and ranked according to their impact on the business. This paper addresses these issues by providing an update of the current state of affairs of the South African manganese industry and its position in the global value chain, as well as identifying the current constraints faced by the various local role players and the impact these constraints have on their economic growth.

2. INDUSTRY BACKGROUND

South Africa hosts about 75 to 80 percent of the world's identified manganese resources and approximately 24 per cent of the world's reserves [12], [15]. Over 90 per cent of the reserves are located in the Kalahari Manganese Fields (KMF) which is situated in the Northern Cape and has an estimated 4 billion tons of manganese reserves [12]. There are two main types of manganese ore present in the Kalahari deposit, namely low-grade primary sedimentary Mamatwan-type ore and high-grade Wessels-type ore [12], [13]. The Mamatwan type contains between 20 to 38 per cent (less than 40%) of manganese in its ores, while the Wessels type, which only makes up 3 per cent of the total ore body, contains 45 to 60 per cent manganese [16], [17]. The manganese ores of the KMF are characterized by their low phosphorus content, which makes it a suitable feedstock for the steel industry.

The country produces high-grade ore, features increasing mining operations and is a significant producer of manganese alloys. The manganese supply and demand closely follows the iron and steel market trends due to manganese's primary use in steel manufacturing, but is also used in numerous other applications. Manganese is the 12th most abundant element in the earth's crust and the 4th most abundant of the metals in commercial use [9].

Manganese ore is obtained in the form of manganese oxide and must undergo numerous forms of processing in order to be converted into alloys. The most important manganese alloys are high carbon ferromanganese (HC FeMn), refined ferromanganese (RF FeMn) and silicomanganese (SiMn) alloys [5], [9], [12]. Manganese has properties which makes it an ideal input for alloy manufacturing and thus plays a significant role in the steelmaking process as an alloying element [7]. Therefore the three industries of manganese, iron and steel are intricately linked [5]. There is no adequate substitute for manganese in steel which contains all of its technical benefits and its relatively low cost [9]. The second most important application of manganese is portable dry cell batteries [7].

2.1 The manganese value chain

It is clear that the world demand for manganese and ferromanganese products have a direct dependence on the outlook of the steel industry, which in turn is driven by housing construction, the automobile industry and general infrastructural constructions [5]. To understand South Africa's position in this industry, the context of the country's role, where the barriers lie to economic growth, the global value chain of manganese, and manganese related products are examined. According to Gereffi [18], for many countries, especially low-income countries, the ability to effectively insert themselves into the global value chain is a vital condition for their development. The manganese value chain is closely related to the beneficiation process. According to Maia [2], beneficiation can be defined as the transformation of minerals, through chemical or metallurgical processes, to higher value added products for domestic or export markets. Different stages in the value chain are defined in Table 1, using examples applicable to the manganese value chain.

Through inspection of the manganese global value chain, it is evident how the industry is organized by examining the structure and dynamics of the different role players involved. Since this mineral commodity, like so many others, is globally integrated with complex industry interactions, examining the value chain is a useful tool to trace the shifting patterns of global production. It is also convenient for associating geographically dispersed activities and role players, and to determine the role they fulfill [18]. The global value chain focuses on the



sequences of value addition within the industry and examines the technologies, standards, regulations, products, processes, and markets which provide a holistic view of the global industry.

Table 1: The different stages of beneficiation, as defined by Maia [2], with examples applicable to the manganese value chain

Stages	Raw material	Stage 1	Stage 2	Stage 3	Stage 4
Metals	Ore	Smelted or refined products	Fabrication alloys and metals	Semi manufactured articles	Fabricated articles
Example	Manganese ore	Ferromanganese	Stainless steel	Long and flat products	White goods, pumps, valves

2.2 South African manganese industry

The first two stages of Maia's beneficiation process are dominant in the South African manganese industry namely, the mining and procurement of the manganese ore and the fabrication of alloys, metals and other intermediate products. The manganese value chain does, of course, stretch much further than only these two sectors (see Figure 1); but these are the main aspects directly pertaining to the current South African manganese production activities. South Africa's position in the global production and consumption of these manganese products are summarised in Table 2.

		Production		Consumption		
Product	Global	Volume	% of Total	Global	Volume	% of Total
	rank	(mt)	% 01 10tat	rank	(mt)	% 01 10tat
Mn Ore	1	4,640	24.9	9	325	1.8
HC FeMn	3	457	10.1	28	27	0.56
Ref FeMn	5	102	5.9	31	10	0.59
SiMn	14	134	1.0	30	30	0.22
Steel	21	7,220	0.45	-	-	-

 Table 2: South Africa's global production and consumption of manganese products [11]

Where Vol = Volume (000 mt Mn Units), % = Percentage of global total, Rank = Global ranking

2.3 Local role players of the manganese industry

It is evident from Figure 1 that the manganese industry consists of various sectors, each utilizing specific types of ore, intermediate products, processes, upstream inputs and role players. This summary was created to provide the reader with a better understanding of the context of the production scale of the various manganese products and to present the different sectors of the manganese value chain. There are currently five major manganese mining companies in South Africa, namely [5], [19]: South32 (formerly BHP Billiton/Samancor Manganese), Assmang Limited, Kalagadi Manganese, Tshipi Manganese and United Manganese of the Kalahari (UMK). BHP Billiton is the world's largest manganese producer while Assmang Limited is fourth. Together these two companies dominate the local manganese production market in South Africa [5]. Smaller producers include Kudumane, Metmin and National Manganese Mines, as well as many other BBBEE companies that have entered the market [19].

The country has four manganese alloy producers that are classified as stage 2 role players in Maia's beneficiation process. These ferroalloy producers are: Metalloys (South32), Assmang, Transalloys and Mogale Alloys [12], [19]. Of these four, Transalloys is the largest producer of silicomanganese (SiMn) in Africa. Mogale Alloys a smaller producer of SiMn and the other two producers supply ferromanganese. With the increase in electricity tariffs and unreliable supply thereof in recent years, together with the oversaturated market, many of these suppliers have drastically slowed down production or halted their operations altogether [20]. South Africa also has producers of electrolytic manganese dioxide (EMD) and electrolytic manganese metal (EMM).

The latter generally accounts for between 6% to 10% of the total manganese ore usage [21]. All of South Africa's manganese resources are located in the Northern Cape Province in a zone stretching northwards over a distance of 150km, from south of Postmasburg to the Wessels and Black Rock Mines north of Hotazel, known as the Kalahari Manganese Field (KMF) [10]. It is the largest single manganese depository in the world and accommodates all of the country's manganese mines. The manufacturers of manganese related products, present higher in the value chain, however, are situated closer to the eastern coast in industrial areas in the Mpumalanga, Gauteng and Kwa-Zulu Natal provinces.





Figure 1: South African manganese value chain [5], [8], [21], [22]



3. RESEARCH METHODOLOGY

Due to the exploratory and descriptive nature of the study, a qualitative research approach was followed as shown in **Error! Reference source not found.** A qualitative approach predominantly affirms an inductive approach to the relationship between theory and research, in which the emphasis is placed on information generation [23]. Since the focus of the study is on providing insight on the multiple role players in the manganese value chain and the various barriers that they face to economic growth, a qualitative study, is suitable for this research.

3.1 Research Approach

The principle data required for this study, is the ranking of the main barriers that different role players in the South African manganese industry face to economic growth. The research methodology followed for this study consists of 3 phases, with the first entailing the identification of data sources and method sampling. This phase addresses research objective 1 by providing an in-depth overview of the local value chain. The second phase is data collection, which includes drafting an interview guide, conducting interviews with relevant company's respondents, and reviewing reports and other forms of secondary data. This phase addresses research objective 2 and 3 through the identification of barriers by industry experts and ranking the barriers according to their responses. The final phase is the analysis and validation of the collected data and results. The phases and underlying activities are shown in Figure 2.

3.2 Data collection method

Multiple data collection techniques were used in the research methodology to reinforce the triangulation of data and therefore establishing the qualitative outcomes of the research. Various forms of data triangulation are employed through the use of qualitative research, the collection of specific operational data (such as electricity usage, when operational delays occur, productivity of the workforce, etc.) to corroborate the qualitative data gathered from the barrier discussions in the interviews. Furthermore, it supports the credibility, reliability and validity of the findings through cross verification of information [24]. This study incorporates triangulation with the following two approaches:

- 1) **Data source triangulation:** Evidence from different data sources (primary and secondary research) was collected. This includes interviews with relevant companies' respondents, questionnaires, company documents, public records, literature review and observations.
- 2) Methodology triangulation: Multiple methods to gather data was combined and utilised. This included conducting both conversational interviewing and semi-structured question interviewing to determine the barriers faced by companies. The results are compared with barriers gathered during company reports, news articles and relevant publications on the manganese industry.

The applied data collection tools for primary data collection includes document reviews and semi-structured interviews with the help of an interview guide which allowed for specific topics to be covered, yet allowed for questions to be added in accordance with responses provided by the interviewees. Open questions were used to ensure that the interviewee had leeway in how to reply and to allow for the interviewee's levels of knowledge and understanding to be tapped in order to explore new areas of research which has previously been limited [23]. The questions were exploratory in nature, prompting the interviewee to identify barriers that specifically affect them and which potentially have not yet been mentioned in public records. The interviews gave stakeholders the opportunity to voice their opinion on the level of interaction between different industry players. The questions also allowed for their perspectives to be shared on the future of the industry and where they believe the major barriers for economic growth lie.

Secondary data was also collected in the form of literature reviewed throughout the study, company publications such as annual and other reports, and publicly available data which is relevant to role players and barriers in the manganese value chain. In order to obtain a broader opinion on the various constraints faced by the role players, a questionnaire was prepared and emailed to numerous experts in the fields of mining, ferroalloy production and other production of manganese products (see Table 3).

The interview and questionnaire covered issues including current operations, logistics, technology, electricity and energy usage, raw materials and pre-treatment thereof, environmental issues, relevant legislation and policies, by-products, economic aspects of operations, exports and sale of products. An interview guide was sent to the respondents before the interview was conducted, presenting an overview of the various aspects that will be covered. It also explicitly requested the biggest barriers that they face in terms of economic growth. After the interview was conducted, a follow-up questionnaire was sent for soliciting specific information on content areas in order to identify and elaborate underlining barriers from the respondents.





Figure 2: Research phases of the study

3.3 Sampling method

Semi-structured interviews were conducted with the aim of gaining comparable views of the most pressing issues in the industry faced by various role players in different sectors of the manganese value chain. It was thus necessary that potential respondents from the different sectors in the value chain were approached for the study. The input from these various industry experts have allowed for the constraints to be ranked according to severity which in turn makes it possible to assign a level of priority to each constraint. All interview responses were used to identify major operational barriers and the scope of its impact.

The list provided by the DMR [19] of manganese mining companies and manganese-related product manufacturers, as well as relevant companies listed in literature [5], [12], [25], identified 23 companies in the South African manganese value chain. Seven of these companies were disregarded, since their operational focus on manganese were negligible or they could not be reached for comments nor were any of their company documentation available. The remaining role players varied in business size, operation field, time in market and size of their market share. From the relevant candidates, three types of role players could be identified, namely those in mining, alloy production and lastly manganese related product manufacturers such as EMM and EMD producers. In order to cover the majority of the local value chain, it was important to have representatives from each sector participating in the study. Smaller sectors such as the specialized usage of manganese in chemical applications, which only accounts for approximately 1% of manganese usage, is not as crucial for this research purposes and were therefore excluded from the study.



Interviews were conducted with representatives of two of the largest manganese mining companies in the world, two of the four South African manganese alloy producers, gaining perspectives in both ferromanganese and silicomanganese alloys operations, and the world's only non-China based producer of electrolytic manganese metal and Africa's only producer of premium-quality electrolytic manganese dioxide.

Role player Type	Interview	Questionnaire	Public records	Could not reach	Identified role players in sector
Mining	2	2	6	10	16
Alloy production	2	2	4	0	4
Other	2	2	2	1	3
Total sources	6	6	12	11	

Table 3: Sampling and sourcing of data

Company	Role / Sector	Operation phase	Interaction type
1	Mine	Running	Interview / Questionnaire / Publications
2	Mine	Running	Publications
3	Mine	Running	Publications
4	Mine	Running	Interview / Questionnaire / Publications
5	Mine	Running	Publications
6	Mine	Running	Publications
7	Alloy	Production decreased	Publications
8	Alloy	Production decreased	Publications
9	Alloy	Production decreased	Interview / Questionnaire / Publications
10	Alloy	Production decreased	Interview / Questionnaire / Publications
11	EMM	Running	Interview / Questionnaire / Publications
12	EMD	Process of closing	Interview / Questionnaire / Publications

Table 4: Companies from which the respondents were surveyed

3.4 Data analysis techniques and interpretations

Barriers are initially identified through observations in company reports and other literature. It was accepted and elaborated upon further through primary sourcing in the form of questionnaires and interviews conducted with experts in the manganese field. The barriers were ranked according to the sectors they occurred in the value chain by the respondents' feedback on each barrier's respective impact on economic growth. A score was assigned to each ranking as follows:

- Ranked 1st = 5 points
- Ranked 2nd = 3 points
- Ranked 3rd = 2 points
- Mentioned = 1 point

The barriers are ranked according to the highest accumulative score between the respondents in the same sector in order to identify the major barriers faced by role players in the manganese mining, alloy production and other related product manufacturing.

Examining the results can help determine how the barriers influence each business and where bottlenecks occur which hinders progression. This analysis thus reveals which barriers cause the greatest restriction on economic growth. The results from the respondents were compared to secondary data from published company reports to determine if there is a degree of consensus. The literature corroborated the findings.

4. RESULTS

Through primary and secondary sourcing, numerous barriers in the manganese industry were identified, stretching from lack of proper infrastructure and energy supply to labour issues and the implementation of new policies by the government. It is also clear that the barriers provided by role players in the same sector had a strong correlation, but sometimes differed from role players in other sectors. An example is the unreliable energy



supply identified in the energy-intensive alloy, EMD and EMM production sector, but does not have as a significant impact in the mining sector. In some instances, some barriers were ranked together by the respondents (such as both the unreliable supply and rising cost of electricity), in which case these barriers shared the respective ranking. From the responses and literature review, a total of 31 barriers were identified. In order to protect the anonymity of the various companies and their representatives, the barriers are not explicitly linked to specific individuals.

MINING SECTOR					
Ranked barriers	1 st	2 nd	3 rd	Other	SCORE
1) Low market price	1	1			8
2) Industry fragmentation	1				5
 Lack of rail capacity 			2		4
Electricity (unreliable supply)		1			3
4) Electricity (rising cost)		1			3
MANGANESE ALLOY SECTOR					
Ranked barriers	1 st	2 nd	3 rd	Other	SCORE
1) Low market price	2				10
Electricity (unreliable supply)		2			6
2) Electricity (rising cost)		2			6
Low productivity of workforce			1	1	3
Volatility of workforce			1		2
5) Cost of labour				1	1
EMD SECTOR					
Ranked barriers	1 st	2 nd	3 rd	Other	SCORE
1) Oversupply of product	1				5
2) Anti-dumping duty		1			3
Electricity (unreliable supply)			1		2
3) Electricity (rising cost)			1		2
EMM SECTOR					
Ranked barriers	1 st	2 nd	3 rd	Other	SCORE
1) Electricity (unreliable supply)	1				5
1) Electricity (rising cost)	1				5
2) Lack of rail capacity		1			3
3) Lack of government support			1		2

Table 5: Scoring of barriers

4.1 Identified barriers in the manganese value chain according to sector

The top three barriers of each sector is ranked according to the impact it has on the economic growth of the role players in the respective sector.

4.1.1 Barriers in the manganese mining sector

1) Low market price

Commodity prices are often affected by external factors which many times cannot be controlled by producers. All commodities are subject to wide fluctuation, especially minerals used for alloy and steel manufacturing. Manganese supply and demand are closely dependent on the iron and steel market with all manganese products following a similar trend to these resources. Price volatility has an adverse effect on a company's operating results, asset values and cash flows. If commodity prices remain weak for sustained periods, it can lead to growth projects no longer being perceived as viable options. China's dominance in the steel market also determines many trends in the industry.

The added manganese ore and alloy capacity during the commodity boom years is now in the market, resulting in considerable overcapacity in the market. As a consequence, manganese is one of the worst performing commodities today [14]. Companies in some instances have rejected additional rail capacity to transport their ore when prices were too low for export. The weak market conditions for manganese products, has caused mining companies to lower production and only consider the cheapest forms of transport or risk making a loss.

2) Fragmentation / rise in internal competition

The fragmentation of the local industry structure has lead to a significant rise in internal competion. The barriers to entry in the manganese ore market have traditionally been relatively high, which allowed for a limited number of players in this sector, but this changed after the government started issuing



new mining licenses to numerous companies. This has caused the mining sector to be condensed with numerous companies all competing for the same resources in the same market.

When the market consisted of less ore producers, rivalry was contained and the price could be regulated by these few entities. At present, many mines are situated together competing for the same rail and port capacity, customers and other resources. The rise in competition between mining companies which is constantly on the rise, has lead to decreased profit margins in order to remain in the market and attract customers. The market conditions are thus becoming less attractive for ore producing companies.

3) Rail capacity

Ports and the freight system in South Africa is currently suffering from inefficiencies rendering most of the manganese ore incapable of being optimally distributed to domestic and international markets. Since the country's transport infrastructure has been found inadequate of supporting higher export volumes to the international market, greater efforts are to be made to improve the efficiency of South Africa's ports and rail utilities. Accelerated economic growth and lack of adequate maintenance and upgrading, however, have rendered the transport system in urgent need of corrective measures.

In both interviewed cases, the mining company's production rate is higher than what can be transported via rail to its various destinations, causing a bottleneck to occur. When the market price for manganese ore was still high, it was profitable to transport the ore via road transport such as trucks, but with the current weak economic position the commodity faces, this is no longer the case. Many mining companies, situated in close proximity to one another, have to vie for rail allocation to transport their ore. If enough rail capacity cannot be provided, then mining production is lowered. The logistic costs of transporting ore to ports for export, as well as from the ports to the final customer, are the highest expenditure for these companies. All interviewed mining companies have scaled down production in order to match their rail allocation. If more is produced, the ore needs to be transported via trucks on the road, which greatly increases the cost per ton and will lead to a loss of profit.

4.1.2 Barriers in the manganese alloy sector

1) Weak market conditions

Currently all manganese ferroalloy producers are shutting down or are in the process of shutting down production, primarily due to the weak market conditions. The market demand is too low to justify operating the energy-intensive furnaces. There was a time when a bottleneck occurred between the mines and the smelters where the supply of ore did not fulfil the demand, but this is no longer the case.

The current commodity recession has consequently lead to a downward trend in the demand for manganese ore and alloys. Steel manufacturing is inherently a primary economy which drives the demand for alloys and is currently in decline. It was not expected that this commodity slump would have continued for so long, which resulted in many producers having continued with operations while hoping for the conditions to change. All along enduring the difficult economic conditions that it entailed.

2) Unreliable electricity supply and rising cost (ranked together)

The second major constraint faced by alloy producers is electricity availability and rising cost. With production lowering due to the weak mineral economy, the supply has become sufficient, but at normal production rate the interrupted supply causes many operational setbacks and the rising tariffs are cutting into profits. Where South Africa used to have very inexpensive electricity, the country has lost its international competitive advantage in this regard. Furthermore, it seems that these prices may still be on the rise in the coming years, while other countries prices remain relatively consistent [26]. In recent years, power has seen a massive jump in price which had a significant impact on operation costs of all alloy producers.

Energy-intensive alloy producers also regularly experience power containments where energy consumption in the company must be lowered for a duration of time in order to lighten the burden on the national electricity grid as per agreement with Eskom. Consistency is key for these producers and their furnaces need to remain continuously running for as long as possible to attain ideal operating conditions. Power containment and interrupted supply, drastically decreases efficiency and increases the operating costs of the furnaces.

3) Volatile and unproductive labour force

South Africa's weak economic growth, rising costs, high unemployment and numerous socio-economic challenges have resulted in many problems and unrest in labour cost and efficiency. This extreme financial pressure has been evident in extended strikes in metal industries. Unrest among workers has led to decreased labour productivity which undermines companies' profitability and threaten the sustainability of the business.


Strikes also regularly occur with employees demanding larger wages, which further decrease productivity. According to respondents, similar facilities abroad are operated by a workforce much smaller in size than locally, with the same work being done by a smaller group of people. This is due to better qualified staff, as well higher productivity of workers abroad. Labour costs are on the increase, adding to the already large operational costs.

4.1.3 Barriers in the EMD sector

1) Oversupply in EMD market

EMD is a niche market with various producers all over the world. Due to the large number of EMD producers and a reduction in demand, especially local demand in Africa, and the market is in oversupply. It is thus likely that a global producer must close down operations. Since local demand is much smaller than abroad, it is likely that South African producers' market share will shrink. This was the case for Delta EMD.

2) Anti-dumping duty

South African producers have not always found it economic to exploit rising demand for manganese products due to its distance from the markets, despite the existing excess production capacity to do so. Competitors that are located closer to the markets have traditionally had a competitive advantage over producers from abroad. Anti-dumping claims have been made against South Africa by companies in Europe stating that South Africa is dumping EMD products abroad and selling it for less than the production cost. This to the South African company being forced to pay a 20% duty fee in order to export its products to Europe which was not a viable option. The same company lost its share in American markets, when the USA decided to terminate its contract and rather support a local producer.

3) Unreliable electricity supply and rising cost (ranked together)

Electricity was another major barrier for this industry. Load-shedding which unpredictably took place and halted operations is a major concern. Many of the machinery had to be prepared more than two hours in advance to ensure that no damages occur to the equipment and the product. If the necessary precautions were not made beforehand, the entire batch being prepared was ruined and ended up as waste. Since an electrolytic process is used, a constant electricity supply is required for optimal working conditions. The sharp annual increases in electricity tariffs in recent years, has placed immense pressure on these companies and drastically increased their overall production costs.

4.1.4 Barriers in the EMM sector

1) Unreliable electricity supply and rising cost (ranked together) (See Barriers in the EMD sector)

2) Lack of rail capacity

The freight system and ports in South Africa are currently suffering from inefficiencies rendering most of the manganese products incapable of being optimally distributed to domestic and international markets. Since the country's transport infrastructure has been found inadequate of supporting higher export volumes to the international market, greater efforts are to be made to improve the efficiency of South Africa's ports and rail utilities. Accelerated economic growth and lack of adequate maintenance and upgrading, however, have rendered the transport system in urgent need of corrective measures.

3) Government support

Companies are making substantial changes to align their businesses with new policies implemented by the government to promote beneficiation, yet these companies do not receive the support or advantages, such as subsidies, as they expected by the government. Furthermore, the constant delays in national infrastructure developments adds to the mismanagement factor constraining economic growth of the businesses involved.

5. CONCLUSIONS AND RECOMMENDATIONS

This study has affirmed the current state of affairs of each sector which comprises the manganese industry of South Africa. An in-depth overview of the manganese value chain was presented, which provides context to how South Africa's local industry fits in with the global market. Furthermore, first-hand data was collected in order to identify the current major barriers faced by role players in different sectors of the value chain, as well as ranking the top three barriers in each sector. The results provide the barriers to economic growth that cause the greatest impedance on the various role players in the mining, alloy manufacturing and EMD and EMM production sectors.



These barriers were up to now, not explicitly identified and ranked according to severity by various role players in different sectors of the South African value chain. This provides significant insight by experts in the field of mining, alloy manufacturing, as well as the production of other manganese related products such as EMM and EMD on where operational bottlenecks in the manganese industry occurs. This study allows for further investigation as to how these major barriers to economic growth can optimally be addressed in South Africa, as well as forecasting the possible risks that these barriers could have on the different role players in the value chain.

Despite all of the research objectives being met, there are still some recommendations for future related work, including expanding the list of identified barriers to include more barriers. A severity score for each barrier could also be added, in order to quantify the extent of the impact of each of the barriers as experienced by each of the respondents.

REFERENCES 6.

- J. Basson, T. R. Curr, and W. a. Gericke, "South Africa's Ferro Alloys Industry Present Status and Future [1] Outlook," in Infacon XI: Innovation in Ferroalloy Industry, 2007, pp. 3-24.
- [2] J. Maja. "Unlocking economic value through the development of mineral value chains in South Africa the context (Presentation at conference on mineral value chains, National Research Foundation)," 2015.
- R. Baxter, "Facilitating further Minerals Beneficiation in South Africa," in Presentation to GIBS Forum, [3] 2013, pp. 1-22.
- Department of Mineral Resources RSA, "A beneficiation strategy for the minerals industry of South [4] Africa," pp. 1-23, 2011.
- O. Gajigo, E. Mutambatsere, and E. Adjei, "Manganese Industry Analysis: Implications for Project [5] Finance," Working Paper Series, vol. 2534, no. 132, 2011.
- UBS, "Manganese 101," Australian Resources Weekly, no. October, 2013. [6]
- "Applications," [7] International Manganese Institute, 2014. [Online]. Available: http://www.manganese.org/about-mn/applications. [Accessed: 20-Jun-2015].
- International Manganese Institute and RPA, "Manganese: The Global Picture A Socio Economic [8] Assessment," Report for the International Manganese Institute, 2015.
- S. E. Olsen, M. Tangstad, and T. Lindstad, Production of Manganese Ferroalloys. Trondheim, Norway: [9] Tapir Academic Press, 2007.
- [10] M. W. Bonga, "An Overview of South Africa's Iron, Manganese and Steel Industry for the Period 1986 to 2006," 2008.
- International Manganese Institute, "2013 Annual Report," vol. 10. pp. 1-158, 2014. [11]
- [12] J. D. Steenkamp and J. Basson, "The manganese ferroalloys industry in southern Africa," Journal of the Southern African Institute of Mining and Metallurgy, vol. 113, no. 8, The Southern African Institute of Mining and Metallurgy, pp. 667-676, 2013.
- K. Ratshomo, "South Africa's Manganese Industry Developments 2004-2011," 2013. [13]
- [14] International Manganese Institute, "Essential Manganese: Annual Review 2015," Essent. Manganese, 2015.
- [15] K. Ratshomo, South Africa's Mineral Industry 2012/2013. 2014.
- N. Van Averbeke, A. Harding, J. Duval, P. Mwape, and J. Perold, "South Africa's Mineral Industry," pp. [16] 1-187, 2005.
- B. Cairncross, N. Beukes, and J. Gutzmer, The Manganese Adventure the South African Manganese [17] Fields. Associated Ore & Metal Corporation, 1997.
- G. Gereffi and K. Fernandez-Stark, "Global Value Chain Analysi: A Primer," 2011. Department of Mineral Resources RSA, "Operating Mines," 2015. [18]
- [19]
- M. Creamer, "Across-board ARM review as earnings plunge 56%," Mining Weekly, 2015. [20]
- RPA, "Socio-Economic Importance of Manganese," pp. 1-2, 2012. [21]
- C. Callaghan, "Mineral Resource Based Growth Pole Industrialisation Growth Poles and Value Chains," [22] 2013.
- [23] Bryman, Bell, Hirschsohn, Dos Santos, Du Toit, Masenge, Van Aardt, and Wagner, Research Methodology - Business and Management Contexts, 2nd ed. Oxford, 2014.
- [24] Write, "Data Triangulation: How the Triangulation of Data Strengthens Your Research," 2016. [Online]. http://www.write.com/writing-guides/research-writing/research-process/data-Available: triangulation-how-the-triangulation-of-data-strengthens-your-research/. [Accessed: 19-Apr-2016].
- C. Callaghan, "Mineral Resource Based Growth Pole Industrialisation Ferrous Metals Report," pp. 1 -[25] 179, 2014.
- C. Fripp, "South Africa's Electricity Pricing Compared to the Rest of the World," 2015. [Online]. [26] Available: http://www.htxt.co.za/2015/06/26/south-africas-electricity-pricing-compared-to-the-restof-the-world/. [Accessed: 22-May-2016].



SUPPLY CHAIN MANAGEMENT: PERFORMANCE BENCHMARKS FOR SOUTH AFRICAN COMPANIES

Liezl Smith

Department of Engineering and Technology Management, Graduate School of Technology Management, University of Pretoria.

Liezl.smith@up.ac.za

Elma van der Lingen Department of Engineering and Technology Management, Graduate School of Technology Management, University of Pretoria. <u>elma.vanderlingen@up.ac.za</u>

ABSTRACT

In a fast-paced competitive business environment, it is essential to be able to gauge performance, not only for companies to track their progress over time, but also their relative performance to others in their own environment. Understanding where you are in terms of performance, can set the stage for improvement and developing competitive capabilities. A study was conducted to determine firstly which performance measures are considered important by South African businesses and secondly if this is an acceptable way to measure performance for benchmarking purposes. The objective of this project is to construct and test a tool to benchmark performance in the context of the South African supply chain environment. The benefit of determining these differentiating factors will be that clearer direction can be given to companies in terms of supply chain factors to improve their chances of success. This could be used as a guide to companies on a continuous improvement journey. A process of designing those measures was done by a focus group involving seven industry experts, where after a rough survey was conducted to test the reaction of the industry to those measures. The research shows a selection of 16 measures that the industry experts consider important to measure resulting from thorough debate. It also includes some segmentation questions that the experts considered important. The research also indicates which measures are used and which people from industry are willing to respond to. These form the basis of a better scorecard to measure supply chain performance, from the three perspectives of input, output and process. This will enable businesses to gauge their performance in relation to their competitors comparing the same indicators. It is recommended that these tested measures be rolled out into industry so that South African businesses can benefit from this research. It will require a mechanism that could form the basis of further research.

Keywords: Supply Chain, benchmarking, performance, measurements, South Africa



1 Introduction

Global competition has forced all businesses to become more competitive and there is a growing need for global performance standards to be understood [1, p. 452]. South African companies are competing with cheap imports from Eastern countries like China and Vietnam, while struggling to hold onto a dominant position in Africa. According to the Global Competitiveness Report [2, pp. 340-341], South Africa is performing relatively well in the African region, although there are certain areas like education, health and labour relations that need serious attention. This indicates that South Africa does have the capability to compete globally, but has special circumstances and challenges to deal with.

One of the key challenges of South African businesses is that they do not have benchmarking information of similar businesses to compare themselves with in the same environment. They can compare themselves with other countries in Europe or the USA, or even the other BRICS countries, but there is no exclusively South African study to show what the other manufacturing businesses are experiencing. An example of an international tool for benchmarking is the Baldridge Criteria for Performance Excellence that has been in existence since 1987. Companies in the US use it to improve practices, capabilities and results, facilitate communication and serve as a working tool for improvement [3, p. ii]. It is a long questionnaire that is self-assessed and only takes American requirements into account.

The reason why differentiation is needed between South African businesses and those in other parts of the world is that South Africa deals with a number of unique challenges that make the situation different to other regions. Even with the same type of product, value, mission, strategy and objectives, performance in different parts of the world will be different [4, p. 45]. Some of those challenges are the labour laws, the large pool of unskilled labour, the distances to other markets, long procurement lead times from other countries, our legislative environment, connectivity and bandwidth limitations, and preferential procurement policies. These differences are not necessarily supply chain related, but have a profound influence on the way companies do business, their results and performance.

Companies are often looking for other similar companies in the same environment to compare themselves to, especially with reference to the operational measures that ensure successful operation. One such example is forecast accuracy. It is well known that accuracy differs according to product type and group, but what are the accuracies associated with a specific group?

There are many surveys that attempt to provide answers, like the State of Logistics survey [5], done by the CSIR and Imperial Logistics, and the Supply Chain Foresight survey [6] run by Barloworld Logistics. None of these focus on specifically operational measures, the challenges they face and benchmark information they can compare themselves to. Most companies are still focused mainly on financial measures [7, p. 611] when they measure themselves.

No similar research to this study could be found. The closest was a study done in 2003 titled "Benchmarking Supply Chain Management Practice in New Zealand" by Basnet et al. [8]. South Africa shares a few of it challenges with New Zealand, like the geographical remoteness, and the relative small size compared to world markets [8, pp. 63-64], but cannot be compared directly. This was also an older study that has lost some relevance in the light of the global economic developments of the last 10 years.

2 Literature and Conceptual Method

In the current business environment, companies are in competition with each other not only horizontally, but also vertically in the same supply chain. Often misalignment of common goals can negate the best effort of those in your same team. They remain dependent upon one another in the supply chain for cooperation and understanding of common issues and challenges and collaborative partnering for best business results. This is called "competitive interdependence" by Gulledge & Chavusholu [9, p. 754]. Thus it is not necessary to just be concerned with what your competition is doing, but also those in your own chain as they may have an effect on your performance, which in turn will affect competitiveness with outside chains. Lai [10] performed a study evaluating performance measures for extended enterprise (EE) in China. He found that extended enterprises are more focused on customer processes, thus place more emphasis on collaboration with customers than with suppliers [10, p. 701], and are more likely to benchmark the associated activities.

Although businesses may produce or distribute vastly different products, the processes they employ may be similar and a possibility for benchmarking. As such, it needs not be industry specific, and may be less likely to be seen as a threat from a competitor wanting to spy. Codling [11, p. 26] uses the example of two concert pianists performing the same work, but with different interpretation. A process will similarly be executed with a



company's own interpretation. It is very seldom that processes can be copied exactly between companies, even as competitors. Each will add its own methods and culture to its interpretation.

Moffet et al. [12, p. 377] found that the benefits of benchmarking extends to improvement in strategy, customer service, performance and processes. According to Randall & Farris [13, p. 450], being able to capitalise on opportunities for better management of supply chains, both within and between companies, effective performance measurement and benchmarking are required. Through successful measuring and benchmarking, performance, decision making and accountability can be enhanced in the supply chain [14, pp. 35-36]. A benefit of benchmarking found in a study of Randall & Farris [13, p. 456] was that measuring the right metric can result in pinpointing the cause of the problem and give direction for improvement. There are significant operational and financial benefits to being best in class [15, p. 12]

The drivers of supply chain management need to be understood and prioritised. There may be some that will be more important to benchmark than others [4, pp. 55,60]. The main drivers include facilities, inventory, transportation, information, sourcing and pricing. According to Golec [16], Table 1 outlines the categories of tactical decisions that will help an organisation achieve its goals.

Tactical decision category	Plan Areas		
Capacity	Type, size, balance, forecasting, technology		
Facility	Layout, capacity, location, focus		
Process technology	Type, integration, productivity, flexibility		
Process management	Goal, effectiveness, efficiency, competitive		
Manufacturing infrastructural	Analysis, design, planning, control, integration,		
	inventory, quality, information		
Human resources	Competence, motivation, recruit, deployment		
New product launch	Design, innovation, customisation		
Supply chain management	Inventory, timing, quality		
Customer relationship management	Timing, delivery, service, satisfaction		

Table 1: Categories for making tactical manufacturing decisions [16]

One of the shortcomings of supply chain benchmarking in the past, has been the lack of integration of performance measures to form a holistic view [17, p. 34]; [18, p. 72]. The respondents in the interviews done by Camerinelli also indicated that alignment between functions is lacking [19, p. 58], and that there is a need for a model to integrate them. The Data Envelopment Analysis (DEA) research aims to level the field and create a single index. [20] [21]

The major weakness of benchmarking in general is that not all aspects are quantifiable and comparable between different businesses. In the Balanced Scorecard Impact Model (BSC -IM), they attempt to overcome this limitation by making use of Impact Charts and Impact Matrices [22, p. 796]. A major challenge in benchmarking is finding a single number index of performance, and most methodologies cannot cater for comparison between companies [23, p. 36]. To overcome this, ratios are sometimes used, but it is still a challenge to combine ratios into a single measure [23, p. 38]. It is the opinion of Li et al. [24, p. 2919] that it is unrealistic and not beneficial to try and measure complete supply chains, but to rather identify those smaller components to use for measuring and benchmarking as they exist in complex competitive and collaborative relationships and cannot be reduced to having one measurable outcome. The common purpose of a supply chain is of course to satisfy the end customer requirement, but how can you compare different aspects if only that is measured? Li et al. [24, p. 2919] thus concludes that a supply chain can be compared to an ecological system or community with symbiotic relationships and abiotic factors and cannot be reduced to only one measure.

Madu & Kuei [20] focus mainly on the efficiency of the conversion processes and take the view that relative efficiency is necessary for comparison. They employ a linear programming-based approach called Data Envelopment Analysis (DEA) to describe the relationship between inputs and outputs. Some benefits to this approach [20, p. 322] [21, p. 5098]:

- Individual companies can be assessed, so not dependent on sample size.
- It produces a single measure for a company.
- Multiple input and -output scenarios can be accommodated.
- There is no restriction on the form of inputs and outputs.
- No weights or costs need to be assigned.
- It does not focus on central tendencies.
- It can create a threshold level for performance.
- Qualitative and quantitative measures can be assessed simultaneously.

Two basic orientations exist - input reduction and output augmentation [20, p. 322]. This is similar to the manufacturing measure of productivity or the financial measure of Return on Investment (ROI). This could be



used to rank companies on a scale to identify the best performer in the benchmarking context. Shafiee et al. [21, p. 5093] also used this concept and combined it with the Balanced Scorecard to give a more integrated view.

Numerous attempts [25], [19], [26], [27], [21], [28] have been made to classify, quantify and model these processes of which The Oliver Wight Class A Checklist for Business Excellence, Balanced Scorecard (BSC), The Supply Chain Operations Reference Model (SCOR) and The Economic Value Add (EVA) Model are some. These were selected to studyfor this research. Although there are many such models available in the literature, there is not enough research on the link between performance measures and the objectives and motivations in entities in the supply chain [29, p. 703].

2.1 Oliver Wight Class A checklist for business excellence

In 1983, Oliver Wight and George Plossl, two manufacturing gurus first introduced the checklist as a map to manufacturing excellence. The current Oliver Wight Class A checklist for Business Excellence [27] is currently in its sixth edition. It contains nine chapters with ten main subjects in each chapter with a number of definitions of excellence each. It works towards world class performance, and advocates 19 milestones along the way. Each milestone has a number of definitions with driver and support measures to score. The status at each milestone is reported and used to indicate the areas to be targeted for improvement. It aims to give a checklist for companies to see where they are in the road to excellence. However, if all companies adopt this, it could be useful to compare, but the cost and time required to go through this exercise is just too large to be seen as worth the result. It usually needs guidance from the Oliver Wight consulting team, which is very expensive. The other criticism of this methodology is that it is not quantitative, and as such based on a subjective assessment of the level of activity of a business aspect. This is very susceptible to bias and manipulation.

Although a very well accepted and useful tool, for the purposes of this study, it is too long, complicated and expensive to adopt in its entirety. The search continued for a model closer to the goals of this study. This brought us to the Balanced Scorecard which is discussed next.

2.2 Balanced scorecard

As originally published by Kaplan and Norton [30], the Balanced Scorecard methodology is probably the most popular and cited of all measurement methodologies today as it not only measures financial aspects, but "balances" the functional perspectives to give better integration for full business performance. It takes the balanced view of not just regarding the tangible measures important, but also those that are considered intangible.

According to Shafiee et al. [21, p. 5093], the biggest benefit of the Balanced Scorecard is to provide a link between strategies and processes, illustrating cause and effect. A number of studies based their further development work on the Balanced Scorecard as discussed below.

Bhagwat & Sharma developed a balanced scorecard for supply chain management because simple financial measures are not giving meaningful direction for improvement in modern day supply chains, and proper measures for success may be of a more intangible nature [25, p. 44]. They based a large part of their research on the work of Gunasekaran et al. [18] who constructed a very thorough framework for measuring the performance of a supply chain. They classified measures into categories: Strategic, Tactical and Operational as well as financial and non-financial [18, p. 83]. They also aligned their framework to the SCOR model processes but added the outcome of customer satisfaction and some measures for that [18, p. 85]. In their conclusion they [18, p. 86] argue that people should be held accountable for the success of the business as a whole, and not just the area they work in. This paved the way for other models that aim to be more integrated. Shafiee et al. [21] conducted further studies combining the Balanced scorecard with Data Envelopment Analysis, another method not in the scope of this study.

Another way to view the Supply Chain through the Balanced Scorecard model, is as proposed by Franceschini et al. [22, p. 788] specifically for manufacturing firms. Their reference model is called the Balanced Scorecard Impact Model (BSC- IM) and is expanded into analysis criteria and summarised in Table 2 below:

Basic dimensions	Analysis criteria	
Financial	Sales	
	Revenue	
	Cost of human resources	
	Cost of raw materials, goods and external services	
	Other costs	
	Investments	
	Amount of debt	
Customer	Communication	

Table 2: Balanced Scorecard Impact Model [22, p. 790]



	After sales service		
	Perception of final product or service		
	Organisation image		
Internal business process	Quantitative production level		
	Cycle time		
	Qualitative production level (final products)		
	Qualitative production level (incoming products)		
	Delivery		
	Stock level		
	Capacity utilisation		
	Expansion		
	Satisfaction of human resources		
	Productivity of human resources		
	Security of human resources		
	Environmental impact		
Learning and growth	Product variety		
	Research and development on products		
	Research and development on process		
	Competitiveness		
	Response time		
	Conformity to customer requirements		
	Rationality in setting and development of projects		
	Education, training and qualification of human resources		
	Self-learning		

Lai [10, p. 695] developed a performance measurement model for the extended enterprise, looking at performance from the perspective of the customer, supplier, management and employee, to measure aspects of finance, technology, operations and the environment [10, p. 695], in a manner similar to the BSC.

All the above are attempts at finding a more balanced and complete view of business. However, like the Oliver Wight checklist, it had a largely functional focus. The search continued and led to investigating the SCOR model, which has more of a process focus.

2.3 The Supply Chain Operations Reference (SCOR) model

The SCOR model was developed in 1996 and endorsed by the Supply Chain Council, now part of APICS, The Association for Operations Management. The precursor to the SCOR model was the series of benchmarking studies done between 1992 and 1994 by a company called PRTM (Pittiglio, Rabin, Todd & McGrath), subsequently acquired by PriceWaterhouseCoopers (PWC). It evolved into a complete model developed by industry for industry. It was selected for inclusion in this study because of its long history and industry wide recognition of being a valuable tool.

The APICS Supply Chain Council already has a number of best practices and metrics in place that can be used as a basis for benchmarking these processes. The basic framework for SCOR measurements is shown in Table 3 below:

Performance attribute	Definition	Level 1 metrics
Delivery reliability	The performance of the supply chain in delivering the correct product to the correct place at the right time, in the right condition and packaging, in the correct quantity with the correct documentation and to the right customer.	Delivery performance. Fill rates. Perfect order fulfilment.
Responsiveness	The velocity at which a supply chain provides products to a customer.	Order fulfilment lead times.
Flexibility	The agility of a supply chain in responding to marketplace changes to gain or maintain competitive advantage.	Response time. Production flexibility.
Costs	Costs associated with operating the supply chain.	Cost of Goods sold. Total SC management costs. Value-added productivity.

Table 3: SCOR metrics [28]



		Warranty/ returns processing costs.
Asset management	The effectiveness of an organisation in managing assets to	Cash-to-cash cycle time.
efficiency	management of all assets: fixed and working capital.	Asset turns.

This model has proven extremely useful, as a number of these measures comply with our requirement for simplicity and ease of calculation. Those measures will be included in the study. But the search continued to build the full picture. This led us to the EVA model.

2.4 The EVA (Economic Value Add) model

The SCOR model has been criticised to lack integrative synchronisation capability [17, p. 36]. In other words, one area may be operating very well according to the performance measures, but has a detrimental effect on a next process. In an attempt to overcome this, Camerinelli & Cantu' [19] draws an Economic Value Add tree and add performance metrics based on the SCOR model to indicate where each will contribute to the financial performance of a business. Below is an adapted version of their tree (Figure 1: EVA model with detail in Table 4).

From the below can be seen that a number of core measures give an indication of financial performance. As the goal of a business is to add economic value, these assist in translating operational measures into financial impact, not only of process performance, but also of proposed changes [19, p. 42]





Table 4: Performance measures associated with EVA model elements	[19,	p. 57]	
--	------	--------	--

EVA element	Performance measures
Sales	Perfect order fulfilment
	Order fulfilment cycle time
	Return on Supply Chain Fixed assets
	COGS
	Return on working capital
	SCM cost
	Percentage in stock
	Forecast accuracy
Total costs	Perfect order fulfilment
	Order fulfilment cycle time
	Return on Supply Chain Fixed assets
	COGS



	Return on working capital
	SCM cost
	Percentage in stock
	Forecast accuracy
	Capacity utilisation
Fixed capital investment	Order fulfilment cycle time
	Return on Supply Chain Fixed assets
	Capacity utilisation
Working Capital	Perfect order fulfilment
	Order fulfilment cycle time
	Return on working capital
	Percentage in stock
	Forecast accuracy
	Capacity utilisation
Inventory	Perfect order fulfilment
	Order fulfilment cycle time
	COGS
	Cash-to-cash cycle time
	Return on working capital
	Percentage in stock
	Forecast accuracy
	Capacity utilisation
Accounts Receivable	Perfect order fulfilment
	Order fulfilment cycle time

	Order fulfilment cycle time Cash-to-cash cycle time Return on working capital
	Percentage in stock
	Forecast accuracy
Accounts Payable	Perfect order fulfilment
	Order fulfilment cycle time
	COGS
	Cash-to-cash cycle time
	Return on working capital
	Percentage in stock
	Forecast accuracy

The literature study was used to create a list of measures with classifications into three groups; Input, Output and Process measures. Table 5 shows how the measures were classified into these groups. These indicate the possible measures to be included in the study. They also form the basis for the proposal to the focus group.



Table 5: Possible measures from the literature



The conclusion from the literature is that if business follows the basic input-process-output construct, the inputs (lead) and outputs (lag) can be benchmarked through specific metrics and the process can be benchmarked through practices. It is also clear that there is no right or wrong answer and that the level of detail and number of methods and metrics is up to the specific party conducting the research. In other words, it can be made as simple or complicated as is desired. All of the methods and measures have advantages and shortcomings, but the challenge is to keep it simple and have a balanced view that gives an accurate assessment of performance, free of bias and manipulation to serve a specific agenda.

3 Research process

It is not clear to companies which measures are important for comparison with competitors and related performances. The following research questions were asked:

- What supply chain measures are important for companies to benchmark themselves against others?
- Can the accepted list be validated?

The research process is described by the diagram in Figure 2 and discussed below.





Figure 2: Synthesis of the research process

A thorough literature review was done to examine the existing theory of benchmarking and previous research done in this field. A number of methodologies were examined and a list of possible measures was compiled to serve as suggested measures to adopt. A selection of willing industry representatives was involved in a focus group initiative to explore the best selection of measures to do a study of competitive nature. The next step was to create a survey and get the survey to as many people as possible to test the validity of those measures. It was sent out to the databases of three industry organisations and a number of responses were received. The 10 most complete responses were selected for analysis. The population is the total business community in South Africa, while the frame was approximately 5000 high-level individuals in various positions in the supply chain space. The response rate was expected to be low due to the nature of the survey and the amount of information required, but the focus was on good quality feedback rather than quantity at that stage. A number of industry bodies were approached for distribution and support of the survey. Further analysis was done on the responses and analysed for results.

After the results were analysed, the findings were compared to the research questions and conclusions were made.

4 Results

During the literature survey, a list of measures were identified that could be presented to the focus group for discussion. A number of senior supply chain practitioners and consultants were invited to the focus group of which seven accepted the invitation to participate. Below a brief profile of the participants:

Job title	Company type	Short description
General Manager:	Explosives	Responsible for strengthening the Sales & Operations
Business Process		Planning process through improved Supply Chain visibility
Optimisation		as well as tactical and strategic decision support tools and
		processes.

Table 6: Profile of focus group participants



Consulting engineer	Transportation Engineers	Specialises in transport economics, logistics and supply chain management as well as education and training in those fields.
Warehouse specialist	Warehouse consultants	He has designed and implemented several hundred warehouse facilities in South Africa, Europe and the Middle East, and is seen as the leading authority in warehouse facilities design in South Africa.
GM Supply Chain	Machinery supplier	Design, implementation and management of optimisation projects across the supply chain. Expert in system enablement, contracting, procurement, maintenance, production planning, inventory and Sales & Operations planning.
GM Transport Operations	Leading retailer	Optimisation and management of retail network transport operations.
C00	Logistics consultants	Experience in consulting, systems integration and information technology. Former Director of Operations for a cosmetics company.
CEO	Bar-coding and warehouse enablement	Company specialising in enabling efficient warehousing solutions through hardware and software deployment.

During this session, there was a lot of debate around the measures, as one of the goals was to keep them as easy and simple to answer as possible. Due to the profile of people in the focus group, a lot of experience was utilised to arrive at the final list. The list was also refined to include the method for calculation that respondents should use. This would ensure that all measured something in the same way and that performance could be compared.

It was decided to include segmentation information for the purpose of industry comparison. Those questions included information on the industry sector, competitive priorities, their supply chain position, their number of employees and the number of stock keeping units (SKU's) in their system. Other background questions include the respondent's job title and whether they as a company have a sustainability strategy. At the end of the survey, respondents are asked to rate the usefulness of the survey as well as space to leave their contact details and comments.

A final list of 16 measures resulted and we could prove that a list of measures can be created. See the list and calculations below in

	Position	Measure	Suggested calculation	Information required	Bucket
		Forecast	MAPE (Mean Absolute	Calculated MAPE (Mean	
1	Input	accuracy	Percentage Error)	Absolute Percentage Error)	Year
		Inventory			
		value/			
		turnover	Average inventory value/	Average inventory value,	
2	Output	percentage	turnover X100	turnover	Year
				Total number of customer	
			Number of Customer Orders	orders received, number of	
			Delivered On Time and in	customer orders delivered	
			Full / The Total Number of	without any mistakes,	
		OTIF (On time	Customer Orders x 100	shortages or late in relation	
3	Output	in full)	(first customer commit)	to promise date.	Month
			Number of items returned/		
		Percentage	number of items sold X100	Number of items returned,	
4	Output	returns	(finished goods)	number of items sold	Year
		Percentage			
		salary bill			
		spent on	Training spend/ salary bill		
5	Input	training	X100	Training spend, Salary bill	Year
		Percentage			
		late or back	Number of Late or back	Number of late or back	
		orders/ total	sales orders/ Number of	orders, total number of	
6	Output	orders	total sales orders X100	orders	Current
		Percentage		Young products, Total	
		new products	Number of products	number of products in item	
7	Input	in the	younger than a year/ total	master	Year

Table 7: List of measurements and calculations



		portfolio (younger than a year)	number of products (finished goods)		
	Outrout	Charle terms	Cost of Goods sold/ average	CoGS, Average inventory	Veee
8	Output	Stock turns	inventory value	Value	Year
		as a percentage of	Outbound Logistics costs/		
9	Output	turnover	turnover X100	Logistics cost, turnover	Year
-	Carpar	Supplier			
		performance -			
		Percentage	Orders purchase orders	Number or purchase orders	
		orders	received on time in full/	placed, number of purchase	
		delivered on	number of purchase orders	orders received on time and	
10	Process	time	X100 (initial commitment)	in full	Year
		Percentage			
		active			
		products on			
		master	Number of items moved in	Items with transactions	
		(movement in	the last 12 months/ Total	against them in the last 12	
		the last 12	number of item number	months, total number of	
11	Process	months) SKU	records (finished goods)	item records	Year
		Percentage			
		Inventory	Number of stocked		
		accuracy at	products with accurate	Total number of item	
		last stock	records/ total number of	number records, Last month	
12	Process	count	stocked products (clarify)	number of accurate records	Last count
			Number of positive stock		
		Demonstration in	values on active items/	Number of positive stock	
12	Output	Percentage in	Number of active items	values on active items,	Last count
13	Ουτρατ	SLOCK		Average days for sustemars	Last Count
14	Process	Debtors days	Days for customers to pay	to pay	Actual
		Percentage		Value of material sourced	1
		local to total	Local / total sourcing value	locally, value of all material	
15	Process	sourcing	X100	sourced	Actual
				Average days to pay	
16	Process	Creditors days	Days to pay suppliers	suppliers	Actual

In Table 8 below the responses are summarised. They are ranked according to the number of responses received per measure.

Of the sample, the response was analysed to understand which measures were worth including in future studies and which not. If at least half the respondents answered a question, it was considered well supported and worth keeping. In our sample, two questions had 80% response, five questions had a 70% response, and six questions had a 60% response rate. The remaining three only had a 40% response that indicates that they are not well supported in industry. The actual values are not important at this stage, as only the benchmarking instrument was tested. From these results it can be seen that 13 out of the 16 measures had a high response rate, indicating high use and acceptability in industry, while the other three had a low response rate, indicating low use and/ or acceptability in industry. For the purposes of this study, it is concluded that those three measures either have to be discarded from future benchmarking exercises, need to be reworked or explained better, or developed to gain better buy-in from industry. This resulted in proving that the list of measures can be validated.

	Position	Measure	Number of responses received	Average value
1	Process	Creditors days	8	40
2	Output	OTIF (On time in full)	8	93%
3	Process	Debtors days	7	37
4	Process	% local to total sourcing	7	72%



CAUEDZ Dressediars 27th	20th of October 201/ Ctobel	and Couth Africa @ 201/ CAUE
SAILEZ / Proceedings, Z/*	- 29 th of Uctober 2016, Stone	ienge, south Africa © ZUTO SAILE
.......		

5	Output	% late or back orders/ total orders	7	11%
6	Input % salary bill spent on training		7	6%
7	% new products in the portfolio (younger Input than a year)		7	2%
8	Process	% active products on product master (movement in the last 12 months) SKU	6	83%
9	Process	% Inventory accuracy at last stock count	6	98%
10	Output	% returns	6	8%
11	Output	Stock turns	6	15
12	Output	% in stock	6	83%
13	Input	Forecast error	6	20%
14	Process	Supplier performance - % orders delivered on time	4	48%
15	Output	Inventory value/ turnover percentage	4	22%
16	Output	Logistics costs as a % of turnover	4	11%

The research shows which measures are used often and are thus well accepted by industry, and which ones are new to them. These form the basis of a better scorecard to measure supply chain performance, from the three perspectives of input, output and process.

5 Conclusions and recommendations

We can conclude from the literature that if business follows the basic input-process-output construct, the inputs (lead) and outputs (lag) can be benchmarked through specific metrics and the process can be benchmarked through practices. It is also clear that there is no right or wrong answer and that the level of detail and number of methods and metrics is up to the specific party conducting the research. In other words, it can be made as simple or complicated as is desired. All of the methods and measures have advantages and shortcomings, but the challenge is to keep it simple and have a balanced view that gives an accurate assessment of performance, free of bias and manipulation to serve a specific agenda.

The characteristics and features of the ideal performance management system are described by Gomes et al. [31, p. 523] as inclusiveness, completeness, timeliness, universality, measurability, consistency, integrity, flexibility and ethical.

It was found that certain measures are important to the supply chain community to measure (Research Question 1). Results were obtained for these measures indicating how many companies use them, and how viable they are in a benchmarking study such as this (Research Question 2). A small sample was used for this purpose, but it provided a great testing ground for the viability of future research in this area and the reception of something like this in the market. Some measures were responded well to, and this indicates that they are accepted and widely used. This could also indicate the level of measurement sophistication that the respondents have reached.

Others were not responded well to, and this can indicate that either the company is not aware of the measurement, not skilled at taking the measurement or does not understand the importance of measuring that aspect of supply chain performance. All three these scenarios present an opportunity for improvement.

Some measures present better value in a benchmarking exercise than others. Only forecast accuracy needs to be reconsidered in terms of how it is measured as it does not mean the same to everyone, and as such does not present good value to a benchmarking exercise. See Table 9 below for a summary.

Position	Measure (Research Question 1)	Supported/ accepted (Research Question 2)	Good benchmarking value
Input Forecast error		Yes	No
la a st	Percentage new products in the	N	N
Input	portfolio (younger than a year)	Yes	Yes
Input	Percentage salary bill spent on training	Yes	Yes
Output	Inventory value/ turnover percentage	No	Yes

Table 9: Benchmarking measures assessed



	Logistics costs as a percentage of		
Output	turnover	Νο	Yes
Output	OTIF (On time in full)	Yes	Yes
Output	Percentage in stock	Yes	Yes
Output	Percentage late or back orders/ total orders	Yes	Yes
Output	Percentage returns	Yes	Yes
Output	Stock turns	Yes	Yes
Process	Creditors days	Yes	Yes
Process	Debtors days	Yes	Yes
Process	Percentage active products on product master (movement in the last 12 months) SKU	Yes	Yes
Process	Percentage Inventory accuracy at last stock count	Yes	Yes
Process	Percentage local to total sourcing	Yes	Yes
Process	Supplier performance - Percentage orders delivered on time	No	Yes

6 Implications for and contributions to theory and practice

This research has long-term and far-reaching implications on theory and practice. The results of the survey can give an overall indication of the "health" of a supply chain, and scope exists to build further on this idea of developing a "health check" tool to make it easier to self-assess and interpret, not only based on individual measures, but also their relationship and interaction between each other. There is also scope for some sort of predictive analysis based on correlation of variables.

As this tool has been developed and tested during this research, it can be expanded to include a wider audience. This can form the basis of further research.

If used correctly, the measures can give the following value to organisations:

Table 10: Value of measurements

Position	Measure	Value to the company measuring
Input	Forecast error.	Indicator of analytical ability and quality of market information.
Input	Percentage new products in the portfolio (younger than a year).	Indicator of pace of innovation in the company, indicator of life cycle of products and product risk.
Input	Percentage salary bill spent on training.	Indicator of commitment to employee development, could influence staff retention and morale.
Output	Inventory value/ turnover percentage.	Indicator of throughput productivity.
Output	Logistics costs as a percentage of turnover.	Indicator of transport efficiency.
Output	OTIF (On time in full).	Indicator of discipline and precision in the warehouse, an indicator of forecast and planning quality.
Output	Percentage in stock.	Indicator of lost sales potential.
Output	Percentage late or back orders/ total orders.	Indicator of how well order fulfilment is managed, indicator of cash recovery and cash flow, indicator of capacity management quality.
Output	Percentage returns.	Indicator of quality of processes and discipline in warehouse, indicator of quality of order management, indicator of product quality, indicator of security and dispatch checking.
Output	Stock turns.	Indicator of forecast accuracy, indicator of planning quality.
Process	Creditors days.	Information about cash flow, indication about control over payment procedures, indicator of risk around credit ratings.



		Information about cash flow, indication of control
		over debt collection procedures, indicator of risk of
Process	Debtors days.	write-off.
	Percentage active products on product	Cross-indicator of stock turn and aging, indicator of
	master (movement in the last 12	stock management quality, forecast accuracy,
Process	months) SKU.	production planning and procurement competence.
	Percentage Inventory accuracy at last	Indicator of discipline and quality of processes in the
Process	stock count.	warehouse, indicator of system integrity.
		Information about the amount of local industry
		supported, indicator of transportation and sourcing
		risk, indicator of commitment to local industry
Process	Percentage local to total sourcing.	development.
	Supplier performance - Percentage	
Process	orders delivered on time.	Indicator of supplier reliability.

The table above is a valuable tool to understand the value that these measurements can provide in a business. It can provide the departure point for goal setting when a measurement is not rendering a good value. These areas also indicate the areas to go look for improvement in the supply chain.

The health of a supply chain cannot be assessed through only one or two measures, but a combination of measures. This can form the basis for further research, not only which measures influence each other, but also the strength of correlation between them.

Another implication is to structure a "best practice" set of KPI's for current supply chains in order to guide companies towards a more productive and useful scorecard to manage their supply chains.

It is recommended that the tool developed in this research be used in future benchmarking surveys, and include a formal structured program for annual benchmarking in South Africa.

7 Works Cited

- [1] P. Hong, S. W. Hong, J. Jungbae and R. K. Park, "Evolving benchmarking practices: a review of research perspectives," *Benchmarking: An International Journal*, vol. 19, no. 4/5, pp. 444-462, 2012.
- [2] World Economic Forum, "The Global Competitiveness Report 2014-2015: Full data edition," Geneva, 2014.
- [3] National Institute of Standards and Technology (NIST), "2013-2014 Criteria for Performance Excellence," Gaithersburg, 2013.
- [4] G. Soni and R. Kodali, "Internal benchmarking for assessment of supply chain performance," *Benchmarking: An International Journal*, vol. 17, no. 1, pp. 44-76, 2010.
- [5] CSIR, "The 10th State of Logistics Survey for South Africa," 2013.
- [6] L. Barloworld, "Supply Chain Foresight Report," Johannesburg, 2012.
- [7] A. Chia, M. Goh and S.-H. Hum, "Performance measurement in supply chain entities: balanced scorecard perspective," *Benchmarking: An International Journal*, vol. 16, no. 5, pp. 605-620, 2009.
- [8] C. Basnet, J. Corner, J. Wisner and K.-C. Tan, "Benchmarking supply chain management practice in New Zealand," *Supply Chain Management: An International Journal*, vol. 8, no. 1, pp. 57-64, 2003.
- [9] T. Gulledge and T. Chavusholu, "Automating the construction of supply chain key performance indicators," *Industrial Management & Data Systems*, vol. 108, no. 6, pp. 70-774, 2008.
- [10] I. K. Lai, "Benchmarking performance measures for extended enterprise in China," *Benchmarking: An International Journal*, vol. 17, no. 5, pp. 692-704, 2010.
- [11] S. Codling, Best Practice Benchmarking: The Management Guide to successful implementation, United Kingdom: Industrial Newsletters Ltd, 1992.
- [12] S. Moffett, K. Anderson-Gillespie and R. McAdam, "Benchmarking and performance measurement: a statistical analysis," *Benchmarking: An International Journal*, vol. 15, no. 4, pp. 368-381, 2008.
- [13] W. S. Randall and M. T. Farris, "Utilising cash#to#cash to benchmark company performance," *Benchmarking: An International Journal*, vol. 16, no. 4, pp. 449-461, 2009.
- [14] S. E. Griffis, T. J. Goldsby, M. Cooper and D. J. Closs, "Aligning logistics performance measures to the information needs of the firm," *Journal of Business Logistics*, vol. 28, no. 2, pp. 35-56, 2007.
- [15] T. M. Simatupang and R. Sridharan, "A benchmarking scheme for supply chain collaboration," *Benchmarking: An International Journal*, vol. 11, no. 1, pp. 9-30, 2004.
- [16] A. Golec, "A relationship framework and application in between strategy and operational plans for manufacturing industry," *Computers & Industrial Engineering*, 2014.



- [17] W. Peng, W. Kuan and Y. Wong, "A review on benchmarking of supply chain performance measures," *Benchmarking: An International Journal*, vol. 15, no. 1, pp. 25-51, 2008.
- [18] A. Gunasekaran, C. Patel and E. Tirtiroglu, "Performance measures and metrics in a supply chain environment," *International Journal of Operations and Production Management*, vol. 21, no. 1/2, pp. 71-87, 2001.
- [19] E. Camerinelli and A. Cantu', "Measuring the value of the Supply Chain: A Framework," Supply Chain Practice, vol. 8, no. 2, pp. 40-59, 2006.
- [20] C. N. Madu and C.-H. Kuei, "Application of data envelop analysis in benchmarking," *International Journal of Quality Science*, vol. 3, no. 4, pp. 320-327, 1998.
- [21] M. Shafiee, F. H. Lofti and H. Saleh, "Supply chain performance evaluation with data envelopment analysis and balanced scorecard approach," *Applied Mathematical Modelling*, vol. 38, pp. 5092-5112, 2014.
- [22] F. Franceschini, M. Galetto and E. Turina, "Impact of performance indicators on organisations: a proposal for an evaluation model," *Production Planning & Control: The Management of Operations*, vol. 25, no. 9, pp. 783-799, 2014.
- [23] W. P. Wong and K. Y. Wong, "A review on benchmarking of supply chain performance measures," *Benchmarking: An International Journal*, vol. 15, no. 1, pp. 25-51, 2008.
- [24] X. Li, X. J. Gu and Z. G. Liu, "A strategic performance measurement system for firms across supply and demand chains on the analogy of ecological succession," *Ecological Economics*, vol. 68, pp. 2918-2929, 2009.
- [25] R. Bhagwat and M. K. Sharma, "Performance measurement of supply chain management: A balanced scorecard approach," *Computers & Industrial Engineering*, vol. 53, pp. 43-62, 2007.
- [26] S. K. Jakhar and M. K. Barua, "An integrated model of supply chain performance evaluation and decisionmaking using structural equation modelling and fuzzy AHP," *Production Planning & Control: The Management of Operations*, vol. 25, no. 11, pp. 938-957, 2013.
- [27] Oliver Wight International, The Oliver Wight Class A Checklist for Business Excellence, 6 ed., W. Goddard, Ed., New Jersey: John Wiley & Sons, 2005.
- [28] Supply Chain Council, Supply Chain Operations Reference Model Revision 11.0, 2012.
- [29] J. Thakkar, A. Kanda and S. Deshmukh, "Supply chain performance measurement framework for small and medium scale enterprises," *Benchmarking:An International Journal*, vol. 16, no. 5, pp. 702 723, 2009.
- [30] R. Kaplan and D. Norton, "The balanced scorecard: Measures that drive performance," Harvard Business Review, vol. 70, no. 1, pp. 71-99, 1992.
- [31] C. F. Gomes, M. M. Yasin and J. V. Lisboa, "A literature review of manufacturing performance measures and measurement in an organisational context: a framework and direction for future research," *Journal of Manufacturing Technology Management*, vol. 15, no. 6, pp. 511-530, 2004.
- [32] U. Bititci, S. Firat and P. Garengo, "How to compare performances of firms operating in different sectors?," *Production Planning & Control: The Management of Operations*, vol. 24, no. 12, pp. 1032-1049, 2012.
- [33] G. Stewart, "Supply chain performance benchmarking study reveals keys to supply chain excellence," *Logistics Information Management*, vol. 8, no. 2, pp. 38-44, 1995.
- [34] A. Gunasekaran and B. Kobu, "Performance measures and metrics in logistics and supply chain management: a review of recent literature (1995-2004) for research and applications," *International Journal of Production Research*, vol. 45, no. 12, pp. 2819-2840, 2007.
- [35] C. Atilgan and P. McCullen, "Improving supply chain performance through auditing: a change management perspective," *Supply Chain Management: An International Journal*, vol. 16, no. 1, pp. 11-19, 2011.
- [36] K. Anderson and R. McAdam, "A critique of benchmarking and performance measurement," *Benchmarking: An International Journal*, vol. 11, no. 5, pp. 465-483, 2004.





DEVELOPMENT OF A WAREHOUSE MANAGEMENT MATURITY MODEL FRAMEWORK FOR THE SOUTH AFRICAN WAREHOUSING ENVIRONMENT

Justine Pillay¹ Sara Grobbelaar² Konrad von Leipzig²

¹Department of Industrial Engineering Stellenbosch University, South Africa justine.pillay@pragmaworld.net

²Department of Industrial Engineering Stellenbosch University, Centre of Excellence in Scientometrics and Science, Technology and Innovation Policy (SciSTIP), South Africa ssgrobbelaar@sun.ac.za

> ¹Department of Industrial Engineering Stellenbosch University, South Africa <u>kvl@sun.ac.za</u>

ABSTRACT

Warehouses play an active role in today's fast-paced and competitive global supply chains. They are utilized from collection of raw materials through to final dispatching of finished products. An efficient warehouse is necessary to achieve an efficient supply chain network with streamlined processes, quick response to market demands and overall high customer satisfaction. An inefficient warehouse on the other hand could result in high labour costs, duplicated and unnecessary processes, high inventory holding costs and even slow response to market demands compounding low customer satisfaction.

This paper aims to develop a Warehouse Management Maturity Model (W3M) framework applicable to the South African environment that will assist warehouse managers in achieving greater warehouse efficiency through improved quality of the various warehousing processes from receiving to dispatching of goods. To develop the W3M framework, three maturity models are studied via a literature review and a Delphi Method survey is conducted with warehousing experts to finalise the features of the framework.

¹ The author was enrolled for a Masters (Industrial) degree in the Department of Industrial Engineering, Stellenbosch University, South Africa.

 $^{^2}$ The co-authors are lecturers at the Department of Industrial Engineering, Stellenbosch University, South Africa



1. INTRODUCTION

In highly competitive business landscapes, companies are under pressure to compete for customers through mechanisms such as product features, price, quality and speed-to-market. In many instances, their survival strategy involves decreasing operational costs, with logistics costs related to warehousing and transportation along the supply chain often a key priority [1]. The South African logistics industry has not been immune to the pressure to compete for customers and has faced a number of challenges ranging from availability of supply chain skills and lack of overall supply chain strategy and tactics to ineffective processes and systems [2].

Rushton, et al. [3] states that with warehousing's key position in supply chains ranging from the distribution of raw materials, work in progress through to finished goods and spare parts storage, its efficiency is of high importance and therefore should not pose any risk to the functioning of the supply chain.

Inefficient processes in a warehouse could easily debilitate the ability of an organisation to fulfil its purpose thus risking the functioning of the rest of the supply chain. This research paper attempts to find answers to the following questions:

- 1. How can a warehouse be made more efficient?
- 2. How can warehouse staff be guided in the identification of strengths and weaknesses in the management of their warehouse?

The literature review in this research focused on maturity models which are a suitable tool for identifying strengths and weaknesses of processes as well as providing benchmarking information. It consists of a framework for assessing the quality of processes within an organisation by having its methods and processes assessed against best practices amongst similar businesses. Albliwi, et al., [4] firmly believes that the presence of maturity models in an organisation plays an important role in providing a platform for assessing their process maturity and making improvements. They also highlight the need for a process maturity instrument that is adaptable and ready to use.

The key factors and processes which are vital to achieving efficiency in a warehouse were identified using a Delphi Method survey. Once these key factors were validated in the iterative Delphi Method survey, a Warehouse Management Maturity Model (W3M) tool was constructed to provide a guide to achieving greater warehouse efficiency in the warehousing environment through improving the quality of the various processes.

2. LITERATURE REVIEW

The literature review presented the opportunity to deconstruct common maturity models to produce the framework and characteristics of the W3M. The intention of this research was not to radically innovate maturity models but rather draw on the success factors, structures and characteristics of practising maturity models to construct a tool which addresses the specific needs of warehouses in South Africa.

2.1 Maturity Models as a benchmarking tool

One may consider a maturity model as an auditing tool to aid organizations in implementing effective processes in a given management discipline so that they can evolve and become more efficient. Using industry best practices to improve and enhance the performance and improve efficiency of one's organisation is not a new concept and is applied to numerous fields [5].

Oyomno [6] describes the concept of "maturity" as *inspiring a sense of progress and growth* and infers that a maturity assessment is viewed positively to suggest a direction in which to grow and improve. Albliwi, et al., [4] states that the presence of maturity models in an organisation play an important role in providing a platform for assessing their process maturity and making improvements. They also highlight the need for a process maturity instrument that is adaptable and ready to use.

Judgev & Thomas [7] present a number of negative aspects of the practical use of maturity models, ranging from its inflexibility with respect to change management to its inability to solve problems and lack of consideration of the human element. However, they also clarify that despite the negative aspects, maturity models have made a considerable contribution to creating competency awareness. This is in the context of this study a very useful aspect of maturity models that allows one to see where weaknesses are present within an organisation in addition to defining the sequence in which they should be tackled.

2.2 Maturity models in industry

Maturity models were originally developed for the software industry in the 1970s, however in recent years they have been modified and applied in a multitude of different fields ranging from project management, telemedicine service maturity [8], accountancy maturity [9] and even government capabilities for e-government [6].

The two most common maturity models actively used in industry include the Capability Maturity Model[®] Integration (CMMI[®]) and Prince2 [®] Maturity Models (P2MM). Following the lack of benchmarking tools in the



Warehousing industry specifically, Dr Jeroen van den Berg, a well-known expert in warehouse management, proposed a Warehouse Maturity grid where he describes four stages of maturity within which companies can fall.

The most common characteristic of all maturity models is that they have levels of maturity ranging from bad to good. Figure 1 describes the characteristics observed in organisations in each of the 5 levels adapted from Crosby's Quality Management Maturity Grid in 1979 [10]. Each level is described by a particular state ranging from uncertainty to certainty.



Figure 1: General structure and characteristics of a maturity model adapted from Crosby [10]

Table 1 describes the characteristics of three Maturity Models discussed in the review of these models. Similar to Crosby's levels, each level has its own characteristics and requirements. The common features of the three models are also summarised in Table 1.

Maturity Model	Capability Maturity Model Integration [11]	Prince2 [©] Maturity Model [12]	Warehouse Maturity Model [13]		
Common name	СММІ	P2MM	WM2	Common feature	
Industry	Software development	Project Management	Warehouse Management		
Maturity Levels	5 levels	5 levels	4 Levels	5 appears to be the most common number of levels	
Features associated with each	Level 1: Initial Adhoc management of processes which are not defined. Control is lacking.	Level 1: Awareness of process Processes are not habitually documented. Organization is over- committed. Processes abandoned during a crisis, resulting in no repeatability of successes.	Level 1: Reactive This is the baseline level. Processes are unstructured and ill defined. Management reacts to same event that occur in the warehouse daily or even hourly. Individual heroics make things happen in this stage.	Processes are not in place. Crisis mode is the norm.	
level	Level 2: Managed Project planning, monitoring and control are exercised. Quality is important.	Level 2: Repeatable process Process discipline is not an organisational rule but basic management practices exist. Basic training in project management principles.	Level 2: Effective Management of the warehouse is standardised. The organisation is systematically structured. Goals are set. Cost and performance levels are identified. There is transparent analysis of bottlenecks.	There is an awakening to quality and processes which must be implemented. The WM2 model does not demonstrate an awakening stage	

T-L.		and all a second a set of a second	a Charles and a second second second		a second the set the standard terms
I ADIE 1	• Nimmarv	and comparison	OT KEV Characteristic	s of maturity model	s applied in industry
Tuble	. Summary	und companion	or ney characteristic	s of macuricy models	supplied in modeling



Maturity Model	Capability Maturity Model Integration [11]	Prince2 [©] Maturity Model [12]	Warehouse Maturity Model [13]	Common feature
				where there a slow transition into implementing processes.
	Level 3: Defined The focus on processes is strong. Process definitions are available. Integrated project and risk management are in place.	Level 3: Defined process Processes are documented and standardized. Improvements are planned and controlled, and based on assessments. Training programme in place to support employees.	Level 3: Responsive Information Technology is used to increase performance. New planning and control principles respond in real- time to events.	Processes are defined and planning is the focus.
	Level 4: Quantitatively Project management is performed using quantitative techniques and process performance is tracked.	Level 4: Managed Process Measurement data is collected to improve overall performance.	Level 4: Collaborative The role of the warehouse in the supply chain is reconsidered. The warehouse is looked at as a part of the supply chain and not a separate entity. The focus is on improving the performance of the entire supply chain through better collaboration.	Quantitative techniques are used to measure performance.
	Level 5: Optimizing The organization has established organizational performance management.	Level 5: Optimized process Strong focus on optimization and forecasting. Processes are quantitatively managed.	There is no 5 th level in Dr Jeroen van den Berg's model.	The focus is on continuous improvement. The WM2 does not have a 5 th level.

A review of these three maturity models (CMMI, P2MM, the Warehouse Maturity grid) is presented in the subsections to follow. This critical review provided the basic framework for the conceptual Warehouse Management Maturity Model.

2.2.1 CMMI framework

CMMI model consists of 5 levels of maturity which provides an incremental step approach for making process improvements. An important feature of CMMI is that it uses 16 process areas to provide focal points for improvement. These areas segment the entire software development process into smaller more manageable and accountable areas which can be improved with greater precision. Another desirable characteristic of the CMMI framework is that it is adaptable as more process areas can be added to the model [8].

Höggerl & Sehorz [14] highlights the need for intensive training on the various terminologies and concepts required to conduct an audit using the CMMI framework which requires a considerable amount of effort to implement and often a shift in culture and attitude [15]. The CMMI framework is complex, multi-faceted and cannot be implemented by a layperson. It needs a team of appraisers with very specific skills and experience requirements to commit time to the appraisal.

Another characteristic of CMMI maturity levels is that all specific goals of the process areas and all generic goals for the respective level have to be achieved in order to advance to the next maturity level [11]. This holds some merit in that organisations are motivated to finalise all improvements in a certain level so they can progress to the next level. However it may also inhibit an organisation from focusing on potential improvements which are quick wins from the next level since they are so focused on the previous level as highlighted by Van den Berg [13]. It is suitable for large organisations with adequate resources to implement



and appraise. The W3M model must steer clear of such requirements as it needs to be simple and easy implementable.

2.2.2 Project Management Maturity Model

Demir & Kocaba [16] indicate that maturity models in project management are becoming popular due to their versatility and ability to control time and cost parameters efficiently. While there are over 30 different variants of Project Management Maturity Models (PMMM) in the market today, each with their own focus, the most prominent one is the PRINCE2 Maturity Model (P2MM) which is a 5-maturity-level framework that organizations use to assess their current implementation of the PRINCE2 project management method. PMMMs are typically aligned with national or international project management bodies of knowledge like the Project Management Body of Knowledge (PMBOK) or PRINCE2. Some consulting firms and project management experts have also offered their versions of a PMMM. One such consulting firm is PM Solutions which combines Software Engineering Institute (SEI) type maturity measurement and the Project Management Institute's PMBOK guide industry standard in identifying key areas of project management to be addressed [17].

Organizations can obtain a P2MM assessment of their maturity either by self-assessment or a formal review conducted by consultants [7]. Self-assessment can be done by simply reviewing the model and assessing the organization using the attributes, or by using the P2MM self-assessment tool to gain a better insight [12]. Organizations have the option of obtaining a P2MM assessment of their maturity either by self-assessment or a formal review. Smaller organisations with limited funds can also use Project Management Maturity assessments.

2.2.3 Warehouse Maturity Model

Dr Jeroen van den Berg [13] describes his Warehouse Maturity grid as a growth model which guides the (warehouse) manager along the four stages in the maturity grid. Each stage in the maturity grid signifies the state of how processes are working. Completing each stage enables the warehouse to act successfully in the next stage as the quality of the processes improves [13]. No critical reviews of this model were available despite a thorough literature search. This may be due to the fact that the model is fairly new and not widely used in industry, let alone South Africa.

The warehouse maturity grid offers breakthrough improvements rather than incremental improvements. This is characterised by the four levels of improvement presented as opposed to the more common five. The step change between level 1 and 2 appears very dramatic. From having no process management to having standardisation and performance management in place is a big leap. These big leaps have a high probability of producing unpredictable results. Each preceding level should essentially act as a primer for each subsequent level. It is for this reason that this breakthrough improvement approach is not favoured in this research.

It is not a rule that you must advance from one level to the next. Dr Jeroen van der Burg structured this model such that opportunities from a higher level perceived as quick wins, could be seized in a lower level. This presents a chaotic approach to implementing improvements as it is not staged. More often than not, an improvement depends on another improvement. Again this offers unpredictable results.

3. RESEARCH METHODOLOGY

This research used the Delphi Method Survey to gauge the views of experts in the supply chain field. The Delphi Method survey is an organised research technique which uses an iterative process to reach consensus on survey responses [18]. The key outcomes of the Delphi Method Survey were to identify the organisational needs of the warehouse per level of the Maturity Model and build the features of the W3M.



Figure 2: Objectives of each Delphi Method round of surveys



Two rounds of the survey were conducted. In order to be selected for the survey, participants were required to have been actively involved in warehousing for a minimum of 5 years. The level of seniority of the participant also played a role, in that the participant should have decision making power and the ability to drive change within the warehouse. Participants directly involved in daily operations of warehousing functions included Operations Directors, Warehouse Managers and Operations Managers. Professional Supply Chain Analysts and Industrial Engineers with experience in overall supply chain activities were also selected to gain an academic perspective on warehousing.

The objective of round 1 was to identify organisational needs for efficiency in the warehouse environment. An invitation to participate in the survey was sent out to 20 experts with working knowledge of warehousing. The 20 experts were from a selection of work colleagues as well as from participants who responded to a LinkedIn advert in the Warehouse Professionals group and the South African Institute of Industrial Engineers group. Of the 20 who were invited, 16 participants responded with completed surveys. Participant statistics are reflected in Table 2.

The results from round 1 were summarised and used to develop a round 2 questionnaire where 14 of the round 1 experts responded with their views on which need should be satisfied in which maturity level. The objective of the round 2 Delphi Method Survey was to categorize each efficiency need according to the W3M maturity levels. Participant statistics for each round are reflected in Table 2.

	Round 1		Round 2	
Designations	Total Number	Combined years of experience	Total Number	Combined years of experience
Operations	2	50	2	50
DC Manager	5	52	3	40
Operations	4	55	4	55
Supply Chain	2	45	2	45
Industrial	3	20	3	20
Total	16	222	14	210

Table	2.	Kov	nartici	nant	ctatictics	for	Dolphi	Mathad	c
I adle	Ζ:	ĸey	partici	pant	statistics	TOL	Delphi	method	survey

4. RESULTS

4.1 Round 1 results

With the round 1 survey structured to find out what key decision makers in warehousing viewed as requirements for efficiency, participants were asked to answer two questions.

- 1. What does efficiency in warehousing mean?
- 2. What are the top 10 most important needs in warehousing to make the warehouse function efficiently?

The responses to the first question in the first survey highlighted how each participant had a different priority in terms of achieving efficiency. Responses which are listed in no particular order include the following:

- Performing and operating on a day-to-day basis with the least wastage of resources (labour, time, money, equipment etc.)
- Being able to store, pick, pack and dispatch items on time, in full and without any safety incidents in the given cycle time.
- Receiving the right product and dispatching the right product to the customer, in the right condition, at the right time using the least amount of effort and resources.
- Productive, happy staff members driving customer satisfaction
- The ratio of output vs. input. The input for warehousing should be looked at from an end-to-end perspective in terms of all resources (including time as a resource) vs. output (units/cartons). Often this can be measured as cost to process a unit/carton and lead time.
- Doing more with less, e.g. Using the least resources to get the highest output.
- Being able to store, pick, pack and dispatch items on time, in full and without any safety incidents in the given cycle time.
- Deliver customer orders on time, complete, without damage and with accurate documentation

Table 3 summarises the responses to the second question which yielded a comprehensive list of organisational needs which must be satisfied in order to achieve warehouse efficiency. In this research, each need was categorised according to its fit into one of the 5Ms (i.e. Manpower, Machine, Material, Method and Money) and then into Management Focus Areas. The result of matching efficiency needs with productivity elements and Management Focus Areas separately is that it is then possible to define productivity elements in terms of Management Focus Areas. Nine Management Focus Areas were identified. They include:

- 1. Demand Management,
- 2. Infrastructure Management,
- 3. Management Control,



- 4. Performance Management,
- 5. Process Management,
- 6. Project Management,
- 7. Quality Management,
- 8. Resource Management, and
- 9. Risk Management.

Management Focus Areas were carefully matched against each efficiency need. This was done to allow for easy allocation of appropriate person/s within the warehouse to own the improvement opportunities within that Management Focus Area. For example opportunities related to improving Risk Management within the warehouse will fall under the Risk Manager's responsibilities. The intention was to spread the responsibility of efficiency improvement opportunities amongst all staff within the warehouse. Improvements should never fall on the shoulders of a single individual as it could result in myopic solutions which have smaller impacts on efficiency.

Table 3: Round 1	results per	Productivity	parameter ((5Ms) auestion 2	1

Productivity element	Organisational Needs from Delphi Method Survey				
Manpower	 Effective resource management Adequate staff required to get the job done Flexibility in resources for peak / off-peak periods Motivated Staff Incentives for reaching targets / goals - recognise and reward Training and equipping staff with new skills / multi-skilling 				
	 2. Management Style Correct management structure to implement and supervise the process Leadership that inspires and listens Positive culture Collaboration workshops with clients to improve service delivery 				
Machine	 Adequate and appropriate equipment and tools Material Handling Equipment Hardware for WMS Warehouse space 				
Material	 The right documentation WMS systems Real time visibility and forecasting Data is used to enhance the warehouse capability to meet customer requirements Stationery available 				
Money	 Money to make improvements Faster decision making on spending of funds for improvements 				
Method	 Agile problem solving Planning, Monitoring and Control Communication / Meetings / Feedback Control, planning, project management (Supervisors/Managers) Planning, Scheduling of key tasks (Fail to plan is a plan to fail) Clear goals and targets, including vision and mission of warehouse KPI's for Managers/Supervisors to effectively manage processes Management reports to measure performance and track progress Continuous feedback on the progress made and checks on whether targets that have been met. Process management Standard Operating Procedures (SOPs) in place Sufficient process flow that works for the system Embracing of lean principles 				



Productivity element	Organisational Needs from Delphi Method Survey		
	 4. Risk Management Risk analysis - corrective action Clear risk and safety procedures documented in a SHEQ policy Good housekeeping Preventative Maintenance 		

4.2 Round 2 Results

Round 2 consisted of two questions which evolved from a summary of the first round results. The survey described the characteristics of the levels in the W3M and asked the experts to give their view on what efficiency need must be met in a particular level in the W3M, for that level of maturity to be achieved. A summarised list of efficiency needs was provided from the first round answers. Participants were asked to pick their answers from the list and add new answers which they thought were missing. They were then asked if the characterization of the different levels were adequate.

The responses from participants are summarised in Table 4. Participant's had a fairly good understanding of efficiency needs required at each level based on the characteristics of that level. The difference in responses from the 14 participants was marginal.

Maturity Level	Characteristics of Maturity Model (provided to experts in survey)	Survey Responses: Categorisation of warehouse characteristics per maturity level
Level 1	Leaning towards adhoc management of processes resulting in chaos when the unexpected occurs. Organisation lacks control and planning. Focus is on just surviving the day.	 Inadequate resources to get the work done (staff, equipment) No standard processes or SOPs No Project management principles employed Lack of role clarity High absenteeism Lack of leadership/ guidance No clear understanding of process flow in warehouse No strategy Targets not set Staff is untrained and no training plans in place Communication is ineffective
Level 2	Organisation is awakened to the need for planning and control as well as effective process management. It is not organisation wide but on a department level and co-ordination is lacking. Basic training of staff.	 Adequate resourcing to get to job done SOP's for high priority processes Basic upskilling/ training of staff KPI's for managers & supervisors Align warehouse goals with the Company mission and vision statement Effective communication Planning instead of random implementation
Level 3	Strong emphasis on effective process management. Processes are standardised and documented across the organisation. Risk Management is in place. Organisation realises the benefits achievable from standardised processes, planning and control. The right people are doing the right tasks.	 SOPs for all tasks Planning is a key requirement Work targets are put in place Risk assessments introduced Measuring and reporting is emphasised Training is commonplace Communication is emphasised Feedback loops installed/ Regular meets Lean culture is encouraged
Level 4	Performance measurement is a focal point to further improve the organisation. Management is committed and actively involved in improvement initiatives. Information is a valuable asset for the organisation.	 Key performance indicators introduced to measure the outcome of all actions being taken Lean and improvement culture embedded in leadership and senior supervisors Implement software to enhance warehouse capability for tracking and data reporting Data is used to enhance the warehouse capability to meet customer requirements Roll-out lean training to leadership and supervisors



Maturity Level	Characteristics of Maturity Model (provided to experts in survey)	Survey Responses: Categorisation of warehouse characteristics per maturity level	
Level 5	The focus is on optimisation. Performance management, forecasting and the use of technology is instrumental in the organisation's optimisation.	 Collaboration workshops with clients to improve service delivery Lean and improvement culture embedded in workforce Real time visibility and forecasting Knowledge of what equipment can be used in which areas and what benefits can be achieved Zero Waste culture Using staff to brainstorm Continuous Improvement initiatives 	

The second question tested the participants' level of agreement with the characteristics associated with each level. The level of agreement used a Likert Scale ranging from strongly agree, moderately agree and neutral, moderately, to disagree, and strongly disagree

As illustrated in Figure 3, the response from participants was very positive with 43% strongly agreeing that the characteristics of the model levels were adequately described and 50% moderately agreeing. This implies a 97% acceptance of the provided model characteristics. Only 7% were neutral in their responses. There were no disagreements with the description of the levels.



Figure 3: Responses from survey participants for Round 2, question 2

Conducting the Delphi Method Survey presented the opportunity to identify key needs for efficiency in a warehouse based on industry expert's understanding of efficiency in warehousing. This served as a reference point for what efficiency needs must be achieved in order to advance from one level of maturity to the next. It became evident that productivity elements could be represented as a function of Management Focus Areas, thus further allowing for a productivity score to be determined per productivity element. The benefits of this scoring are multi-fold in that:

- 1. This would give the warehouse manager an indication of how productive they are in using a certain element to achieve efficiency.
- 2. It will be possible to define a scoring dashboard which can be used for progress tracking, benchmarking; and
- 3. It could highlight the specific element which is not performing well in the warehouse and which management focus areas are weak.

There was a 97% agreement amongst the industry experts with the description of each maturity level and the overall framework presented (i.e. 5 levels of maturity ranging from level 1 where there is chaos to level 5 which is focused on continuous improvement).

5. CONCEPTUAL MODEL

There are predominantly 5 levels of maturity ultimately describing each state adapted from Crosby [10]. These 5 levels are

- Level 1: Uncertainty,
- Level 2: Awakening,
- Level 3: Enlightenment,
- Level 4: Wisdom, and
- Level 5: Certainty



Maturity level	Question asked by organisation	Characteristics of organisation
Level 1: Uncertainty	"Why are we not coping?"	The organisation is leaning towards adhoc management of processes resulting in chaos when the unexpected occurs. The organisation lacks control and planning. Focus is on just surviving the day.
Level 2: Awakening	"What can we do to survive?"	The organisation is awakened to the need for planning and control as well as effective process management. It is not organisation wide but on a department level and co- ordination is lacking. There is also basic training of staff evident.
Level 3: Enlightenment	"What can we do to cement our survival?"	There is strong emphasis on effective process management. Processes are standardised and documented across the organisation. Risk Management is in place and the organisation realises the benefits achievable from standardised processes, planning and control. The right people are also doing the right tasks.
Level 4: Wisdom	"How do we know what we are doing is good or bad for us?"	Performance measurement is a focal point to further improve the organisation. Management is committed and actively involved in improvement initiatives. Information is a valuable asset for the organisation.
Level 5: Certainty	"What can we do to improve our organisation?"	The focus is on optimisation. Performance management, forecasting and the use of technology is instrumental in the organisation's optimisation.

Table 5: Round 2 results with warehouse management characteristics per maturity level

Advancement from one level to the next without achieving the targets of the current level is not recommended. Knowledge areas are useful in categorising the different areas of management within a particular field. In the W3M framework, these knowledge areas will be known as Management Focus Areas.

Including the 5Ms used in root cause analysis as productivity elements will assist warehouse managers in pinpointing the specific resource area which is lacking, be it people, process, information, equipment or money related. In addition to the 5 productivity elements, there are nine management focus areas which are composed of specific attributes which must be met, namely:

- 1. **Demand Management** concerned with planning of resources to match demand profile of warehouse
- 2. Infrastructure Management focusing on appropriate design and layout of the warehouse and equipment upkeep
- 3. **Management Control** of warehouse in terms of decision making, support and guidance to staff, including setting of targets that are aligned with the company's strategy
- 4. **Performance Management** covering Monitoring and controlling of performance including tracking and reporting
- 5. **Process Management focused on** process mapping and analysis as well as ensuring that Standard Operating Procedures and times for key processes are adopted
- 6. **Project Management ensuring that a** structured approach to managing scope, schedule, budget and quality is in place
- 7. Quality Management focusing on Continuous improvement initiatives
- 8. **Resource Management managing** labour (training, well-being, role clarity), equipment, money (budget) and tools (e.g. WMS and MEH)
- 9. Risk Management ensuring that risk assessments and proactive mitigation of identified risks is conducted

6. W3M ASSESSMENT AND REPRESENTATION OF RESULTS

In order to evaluate the maturity of the warehouse processes in a warehouse, a W3M assessment must be carried out by the warehouse manager. As with any assessment, certain rules should be applied to allow for a fair and logical conclusion to the assessment. A W3M Assessment Tool will be used to determine the level of maturity of each Management Focus Area in a warehouse. The expected outcomes of each Management Focus Area's efficiency needs were identified in the Delphi Method Survey. To obtain the W3M Assessment Tool described in this section, it is suggested that the lack of that outcome is defined. Incremental improvements can then be devised to get to the next level of maturity.

Figure 4 depicts the W3M radar chart, which clearly shows the nine Management Focus Areas with the specific 5M productivity elements impacted. The W3M radar chart is able to provide a bird's eye view of the strong and weak Management Focus Areas in a process. The graphical W3M radar chart can be placed on a visible notice board in the warehouse and incorporated into meetings with staff so they are aware of the state of the warehouse and what needs to be done to improve its efficiency.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 4: W3M radar chart (author's representation)

7. CONCLUSION

The research consisted of two objectives, namely identify efficiency needs in the warehousing environment and a method of guiding warehouse staff in the identification of strengths and weakness in the management of their warehouse. The first objective was achieved via the Delphi Method survey. The second objective was achieved by constructing a Warehouse Management Maturity Model (W3M) tool help warehouse staff assess the quality of processes in their warehouse, and ultimately highlight what deficiencies or gaps must be filled to achieve efficiency. By combining the maturity model frameworks from the Literature survey with the efficiency needs from the Delphi Method survey it was possible to categorise the various efficiency needs required per maturity level.

The W3M tool is composed of 5 productivity elements and there are nine Management Focus Areas to pin point the aspect of the warehouse that requires improvement.

The following conclusions may be made about the conceptual model:

- 1. An assessment will be required to determine the level of maturity of the warehouse.
- 2. A defined assessment process must be in place incorporating continuous improvement methodology.
- 3. Assessment results must be visually pleasing and reflect the maturity of the various Management Focus areas at a glance. A W3M radar chart will serve this purpose as illustrated in Figure 4.
- 4. A W3M Productivity score must be calculated to highlight the specific productivity element which is not performing well in the warehouse.



8. RECOMMENDATIONS FOR FUTURE ADVANCEMENT OF THE W3M TOOL

With the scoring technique adopted, similar warehouses can select process areas by which to evaluate their maturity. They can then compare their maturity. Warehouses can be evaluated against the W3M in tender processes by manufacturers and retailers.

The W3M has a very flexible structure that can be further developed to add more efficiency needs and possibly more Management Focus Areas. Some examples include Customer Service Management and Information Technology Management to address high customer satisfaction and access to information efficiency needs respectively.

CMMI calls for process assets to be produced to validate the claim that a certain level is true of the organisation. The W3M lacks this aspect and may make the model very subjective. A list of process assets should be compiled for each Management Focus Area's expected outcomes.

Each warehouse process can be evaluated to determine its maturity. Despite being time-consuming, breaking down warehouse management into the six processes identified will simplify the implementation of improvement initiatives as the efforts will be more focused. The process area team leaders/managers will conduct assessments for the process areas which they are responsible for, thus allowing the gaps to be pinpointed to a specific process area. Each warehouse process will have its own W3M radar chart.

A possible 6th level of maturity may be added to the framework involving collaboration of the warehouse with the rest of the supply chain. The impact on reducing costs can be assessed once the model is adopted and measured for success.

9. **REFERENCES**

- [1] M. Christopher, Logistics and Supply Chain Management, Fourth ed., Dorchester, Dorset: Pearson Education Limited, 2011, p. 247.
- [2] Council for Scientific and Industrial Research (CSIR), "10th Annual state of Logistics for South Africa," CSIR, Pretoria, 2013.
- [3] A. Rushton, J. Oxley and P. Croucher, The Handbook of Logistics and Distribution, 2nd ed., Glasgow: Bell & Bain Ltd, 2001.
- [4] S. Albliwi, J. Antony and N. Arshed, "Critical Literature Review on Maturity Models for Business Process Excellence," *Industrial Engineering and Engineering Management (IEEM)*, no. 10.1109/IEEM.2014.7058604, pp. 79-83, 2014.
- [5] B. Walley, Efficiency Auditing, London: The Macmillan Press Ltd, 1974.
- [6] G. Z. Oyomno, "Towards a framework for assessing the maturity of government capabilities for 'e-government'," *African Journal of Information and Communication*, no. 4, pp. 77-97, 2003.
- [7] K. Judgev and J. Thomas, "Project Management Maturity Models: The silver bullets of competitive advantage?," *Project Management Journal*, vol. 33, no. 4, pp. 4-14, 2002.
- [8] L. van Dyk, "Development of a Maturity Model for Telemedicine," South African Journal of Industrial Engineering (SAJIE), vol. 23, no. 2, pp. 61-72, 2012.
- [9] Confederation of Asian and Pacific Accountants (CAPA), Maturity Model for the Development of Professional Accountancy Organisations, Kuala Lumpur: CAPA, 2014.
- [10] P. B. Crosby, Quality is free: The Art of Making Quality Certain, McGraw Hill, 1979.
- [11] Software Engineering Institute, "CMMI for Services Version 1.3," Carnegie Mellon University, 2010.
- [12] Axelos Limited, "PRINCE2® Maturity Model (P2MM), v 2.1," Axelos Limited, 2013.
- [13] J. P. van den Berg, Highly Competitive Warehouse Management, North American ed., Distribution Group, 2012.
- [14] M. Höggerl and B. Sehorz, "An Introduction to CMMI and its Assessment Procedure," Salzburg, 2006.
- [15] F. Backlund, D. Chronéer and E. Sundqvist, "Project Management Maturity Models A Critical Review: A case study within Swedish engineering and construction organisations," in 27th IPMA World Congress, Sweden, 2014.
- [16] C. Demir and I. Kocaba, "Project Management Maturity Model (PMMM) in educational organisation," Procedia Social and Behavioral Sciences Journal, vol. 9, pp. 1641-1645, 2010.
- [17] PM Solutions, "PMO Services," 2015. [Online]. Available: http://www.pmsolutions.com/services/pmoservices/project-management-maturity-advancement/. [Accessed 01 November 2015].
- [18] C.-C. Hsu and B. A. Sandford, "The Delphi Technique: Making sense of consensus," *Practical Assessment*, *Research and Evaluation*, vol. 12, no. 10, p. 1, 2007.



SOCIAL MANUFACTURING BAMBOO BIKES FOR AFRICA

J.F. Oberholzer¹, G.A. Oosthuizen², P. de Wet³, M.D. Burger⁴ & C.I. Ras⁵

¹Department of Industrial Engineering, STC-LAM University of Stellenbosch, South Africa <u>16499700@sun.ac.za</u>

²Department of Industrial Engineering, STC-LAM University of Stellenbosch, South Africa <u>tiaan@sun.ac.za</u>

³Department of Industrial Engineering, STC-LAM University of Stellenbosch, South Africa <u>17074975@sun.ac.za</u>

⁴Department of Industrial Engineering, STC-LAM University of Stellenbosch, South Africa <u>17005019@sun.ac.za</u>

⁵Department of Industrial Engineering, STC-LAM University of Stellenbosch, South Africa <u>16452550@sun.ac.za</u>

ABSTRACT

Manufacturing methods constantly change to adapt to societal needs. However, there is a new method aimed towards value creation emerging. This method or manufacturing paradigm is described as social manufacturing, which combines manufacturing with open design platforms. This paper discusses a Bamboo Bikes for Africa case study where the open design process is used through an online community using social manufacturing techniques. The best design is chosen, manufactured and compared to benchmark process chains. This study proved that using crowd sourcing and co-creation, designs can be developed and improved to generate a significant amount of innovative designs.

^{1,3,4 & 5}The author(s) was/were enrolled for an M Eng (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch



1. INTRODUCTION

The activity in which the form of raw materials is changed to create products is defined as manufacturing, and it ensures value creation from natural resources [1]. The evolution of manufacturing technologies has caused a paradigm shift and changed its emphasis from mass production and product variety a number of times since the industrial revolution [2]. Societal changes, as well as change in market imperatives are the major driving factors behind the manufacturing paradigm shift. We currently find ourselves at the brink of a new manufacturing revolution, called social manufacturing, and it includes the shared creation, distribution, trade, production and consumption of goods, resources and services by different people and organizations throughout the industrial and social spectrum [3]. Figure 1 illustrates these changes in manufacturing paradigms, with regards to economics of scale and scope, showing when the emphasis changed [4]. It also illustrates how social manufacturing links with the bamboo bicycle project sketch. With the invention of assembly lines, manufacturers were concentrated on producing large volumes of products, with low customisability. This was the era of craft production and mass production, however emphasis on producing larger volumes of products continued until the market became saturated around the 1970's. Many products were mass produced and society demanded greater variety, giving birth to the era of flexible production. This trend continues until around the year 2000, where manufacturing changed to a greater emphasis on customisation and personalisation. However social manufacturing is the new manufacturing revolution.





An industrial revolution, driven by the Internet of Things, has given rise to various manufacturing strategies worldwide [5]. Examples are Catapult in the United Kingdom, SIP in Japan, Industry 4.0 in Germany and NNMI in the United States [6]. Kagermann, et al. (2013) decribe this revolution as the convergence of the virtual world and the physical world in the form of Cyber-Physical-Systems (CPS). Changes in production methods, customer expectations and value creation will occur as a result of this era of manufacturing. This open design platform enables customers, local suppliers and small companies to develop products and social manufacturing helps manufacturers to produce these products.

A case study is required in order to understand the business model of social manufacturing. An experiment can be conducted where the community or industrial cluster is used within a manufacturing process. An industrial cluster is the "social community and economic agents" [7] that collectively strives to produce a superior product and/or service. A social community is an ever-changing body of people. Thus, by using a social media platform, their idea creation, knowledge and niche-spotting capacity could be harnessed to address seemingly overwhelming problems.

Bamboo is used, because it is a sustainable material that will contribute significantly to the manufacturing industry and local communities in Africa. This project does not aim to only give these people a means of transport, however the aim is to enrich their lives and businesses, therefore adding 'Swiss army knife' like technology to these bicycles to improve their way of living.

Because the rural parts of sub-Saharan Africa have a high poverty rate and declining road conditions, transportation issues in these parts impose a challenging environment to find sustainable solutions to uplift the economic and social conditions of individuals living in this strenuous environment. The ability of an individual to access jobs, healthcare and education directly affects their livelihood [8]. Transportation also provides access to various social networks throughout a country from where future job opportunities may emerge. By developing the skills of a population, sustainable economic growth can be reached through the alleviation of poverty and the reduction of inequality. This is proof that human development and economic growth is linked and mutually reinforcing [9].

Finding a way to address the transportation issues while reducing unemployment through human development will provide a sustainable solution that will diminish poverty and increase economic growth. Manufacturing and



selling bicycle frames made mainly from bamboo could be a sustainable solution to provide rural Africans with a method of transportation, while facilitating skills development and job creation.

This paper illustrates the use of social media as a design platform. It uses a case study of a final year engineering class, tasked to design bamboo bicycle frames. An effective transport system is one of the major problems in Africa, therefore using this Bamboo Bikes for Africa Project to create bicycles for rural Africa is essential and it can change the face of transport in Africa.

2. SOCIAL MANUFACTURING

According to Koren [10], personal and social networking relationships can create value to organisations, by allowing them to utilise the resources within a network for their benefit. This is the premise on which social manufacturing is built. Social manufacturing is described as a new kind of networked manufacturing mode, integrating many distributed, socialised resources, and bundles enterprises into manufacturing communities [11]. This results in the formation of manufacturing communities by initial clustering and self-organisation. An enabler of this social manufacturing, or crowd sourced manufacturing, is web 2.0 technologies [12].

The ability to generate new knowledge can play a significant role in staying competitive. Open design platforms use this ability to change the manner in which knowledge is constructed around manufacturing, leading to new and faster methods to solve problems through co-creation [13] and [14]. Social manufacturing is predicted to be in use by 2020. Table 1 shows the intended business model, which implements a pull system (sale-produce-assembly). The resulting value chain causes a sustainable conscious society, with a demand for personalised products. The internet of things allows social manufacturing to be driven forward and can thus be seen as an enabling technology. Self-organising systems can be seen as a key technology, while information and knowledge processing is based on a cyber-physical system.

Paradigm	Social Manufacturing		Existing Social Manufacturing companies
Societal Needs	Personalised products on demand – Sustainability conscious		Opendesk
Market	Global production – demand fluctuation		Local Motors
Business Model	Pull (sale-produce-assemble)		Shapeways – 3D printing
Enabling Technology	Internet of things		WindowFarms
Key technology	Self-organizing systems		Blender
Information & knowledge processing	Cyber-physical systems		OpenStructures

Social manufacturing has the ability to create opportunities for internal related work or work with corporate partners. This is done by providing all involved parties with access to the relevant information, allowing them to transfer and share documents, and to automate tasks which were traditionally done manually. This results in an accelerated process and decision making.

While traditional manufacturing companies rely on internal design, social manufacturing companies allow any interested party with internet access to submit designs and ideas. An open database allows these designs and ideas to be shared, where other people can contribute by suggesting improvements or adding it themselves. The design period can be shortened by using increasing amounts of crowd sourcing. This is done by identifying patterns from emerging synthesis quicker, assisting in developing customer demanded products faster.

Furthermore, in social manufacturing, the manufacturing is done by the user or the market. Manufacturing capabilities are embedded within the online community platform database. This allows social manufacturing companies to both design and prototype products faster, utilising more human resources, at a reduced cost. This is illustrated in figure 2. Therefore, identifying and comparing the different business elements of social manufacturing and generating a generic business model for social manufacturing is essential to incorporate social manufacturing within current manufacturing companies.





Figure 2: Comparing efficient utilization of social- and traditional manufacturing methodologies (not to scale - for illustrative purposes only) [4]

Ras et al [1] created a development lifecycle as seen in Figure 3. This lifecycle or process summarizes all the factors and at what phase of the product lifecycle these factors have to be taken into consideration.



Figure 3: Social manufacturing product development lifecycle.

3. BAMBOO MANUFACTURING

3.1 Standardisation

Within the species great variation is observed in the diameter and the height of the culms. The lower the density of the culms the greater the diameter-at-breast-height (DBH) will be, but the lower the total biomass per unit area [15]. A high culm density will result in a reduced DBH and higher total biomass. Hence the culm density influences the diameter of the culms. The most effective agricultural methods, which facilitates easy harvesting must also be pursued [16]. An irrigation schedule will also be followed especially during the dry season.

Bamboo reaches its maximum strength when the culm is mature. This depends from specie to specie but is usually from the age of 3 years. If bamboo is older than 6 years, it might have suffered damage caused by insects [16]. Lopez suggests that the location of the bamboo has an influence on the strength properties of bamboo [3]. Research done on *Dendrocalamus* showed that the finest bamboo of this species grew in drier areas.

South Africa's only indigenous bamboo is *Thamnocalamus tessellatus*. The Bamboo source in South Africa that can be utilized is *Bambusa balcooa*, it was introduced to the country in the 1660's for paper pulp production and has since naturalized to South Africa's climate, although it's natural habitat is located in more tropical climate areas [17]. The plant itself can reach a height of 12m to 20m and a diameter of 6 cm to 15 cm [18]. In some circles *Bambusa balcooa* is also referred to as giant bamboo [17]. This offers a distinctive challenge when it



comes to bicycle manufacturing which generally requires thinner diameter tubes especially at the chain and seat stays at the rear section of the bicycle.

Bamboo is harvested by selecting the most suitable culm, certain factors that are important to consider in the harvesting process are: the felling cycle; the intensity of the felling; the method of the felling as well as the transportation of the felled bamboo to the factory [15]. As previously argued, identifying mature bamboo is important. In year four, which is a good age to harvest, *Bambusa Balcooa* has the following characteristics. The culm covering is absent, branches have less leaves, most of the auxiliary and secondary branches transform into thorn like structures. Thicker branches are usually dead and shed, leaving scars on the culm. The culm has a smooth surface and is dark green. Black or rotten adventitious root ring can be present on the basal 1 to 2 nodes [16].

In the morning bamboo transports starch from the roots to the leafs as due to photosynthesis. The best time to harvest bamboo is early in the morning before sunrise from 12am and 6am, most of the starch is then in rhizomes at the roots [15]. The following advantages arise from harvesting bamboo in this manor: The bamboo is less attractive to insects, the bamboo weighs less for transporting purposes and the bamboo will dry faster.

The starch (sugar) content in bamboo is also high in the growing season considering this the best season for harvesting bamboo is after the rainy season. The culms will then be more resistant to biological degrading organisms [16].

Bamboo has less natural durability than most woods because of the absence of certain chemicals. The starch in the bamboo attracts beetles and fungi. Another factor that contributes to the bamboo's low natural durability is the hollowness of the bamboo. If an insect or fungi destroys 2mm of the outer layer of wood it is still in fair condition. Considering bamboo 2 mm is the quarter of the thickness in some instances. The hollow inside of the bamboo serves as a good hiding place for these agents of destruction [19].

Untreated bamboo typically has the following life, subject to the environment that it is in: 1-3 Years in the open air and in contact with soil, 4-6 years when it is under cover and not in contact with soil and 10- 15 years under very good conditions. There exist chemical and non-chemical methods to treat bamboo. The non-chemical methods can be conducted by unqualified villagers without technical equipment and with little cost. The non-chemical treatment can significantly increase the resistance against fungal and beetle attack. The real cost saving benefit of these methods with regards to long-term usability must be carefully evaluated [19].

Water- based chemical solutions are divided in non-fixing and fixing to the bamboo tissue and uses organic or inorganic salts. When the water from the solution evaporates the salts are left, which are either fixing or non-fixing. Non -fixing preservatives wash away in rainy conditions and are hence only suitable for use under cover in-doors or in dry conditions. There are different methods to apply chemical preservatives [16].

The drying of the bamboo is a step in the manufacturing process that improves the structural properties and the desirable appearance. Two major drying methods are: air drying and kiln drying.

- i. Air drying: The moisture is removed by exposing it to atmospheric conditions. Stacking bamboo upright dries the bamboo in half the time when compared to stacking them horizontally. The drying time can range from several weeks, to several months. It goes without saying that air drying is quite dependent on the weather. A moisture content below 12% is required [16].
- ii. Kiln drying: It is more efficient than air drying, the bamboo can be dried to the required moisture content in a much shorter time. Kiln drying produces high level results. It is a basic process of stacking bamboo culms or splits in a chamber where the air circulation and the temperature are maintained and controlled so that the moisture content can be reduced to the required level. Drying schedules are used to control the temperature and the relative humidity to the dry the bamboo in the shortest possible time with the least occurrences of degrades. Each bamboo specie has a unique drying behavior and therefore requires a unique schedule [16].

The identified process chain the bamboo follows before it is used for manufacturing are depicted in figure 4. The different strategies to standardize bamboo are listed below each step.





Figure 4: The four stages before the bamboo can be used for manufacturing and the accompanying standardizing strategies [15].

3.2 Manufacturing bamboo bicycles

The Bamboo Bicycle Club is situated in London and they help people who are passionate about cycling to build and ride their own bicycles. In addition to provide each person with all the necessary parts and tools to create their own bicycles, they also offer workshops to teach each person how the process works. The club generally uses one of the two most common species of bamboo for building bicycles, Mos or Tonkin, and source the product from an importer in the UK **[20]**. Both of the species, Mos and Tonkin, have good environmental performance and excellent strength properties, which makes them perfect for use in building a bicycle. Even though the bamboo is hand selected and treated at the source, they still recommend that it be coated with a protective layer to limit the movement of moisture and to protect the frame against the elements.

One of the most important steps in building a bamboo bicycle frame is to know which pieces of bamboo need to be used on which parts of the bicycle. After selection, the different pieces of bamboo are cut into the correct length according to the design that has been chosen. A hand drill with a circular hole cutter is used to cut the bamboo. This allows each piece of bamboo to fit snugly over the other. After cutting, a hand file and box cutter is used to complete the final shaping process.

The most important step of the secondary processes involved is the fiber and resin. This step determines the strength of the bamboo bicycle and is what keeps the frame the together. This could literally make or break the bamboo bicycle.

HERO bikes follows a contemporary approach to create a sustainable, socially conscious small business through the utilization of a locally grown bamboo [10]. HERO bikes cater for the DIY enthusiasts who want to manufacture their bicycles on their own. Not only do they sell bamboo bicycle kits that the customer assembles on their own, but they also have workshops which provides the customer with the opportunity to build their own bicycle in the HERO bike shop under the supervision of technicians. Included in the purchased kit is the jig used to assemble the bicycle. For the purpose of this study, only the building of the physical bicycle is considered, not the assembly of the jig itself.

A project was completed by the Engineering Materials Development Institute of Nigeria where they attempted to develop an eco-friendly bamboo bicycle by following these three steps: 1) outsourcing of the intricate parts; 2) heat treating the bamboo; and 3) jig design and assembly [21]. Since all the joints were outsourced, the manufacturing processes of these parts are overlooked. The drying process of the bamboo consisted of heating the bamboo evenly and slowly until the color transitioned from green to light brown, and then from light brown to dark brown. After heat treatment, the cutting process ensured that the bamboo was the required length according to the design. The third step is to mate the bamboo tubes with the brackets that were outsourced in step one. A large end mill was used to miter the tubes to roughly the size of the brackets they would be mated


to, then a dermal was used to miter these tubes to perfection. After ensuring that all the bamboo fit perfectly with the joints, all the components were assembled in the jig, awaiting the fastening process.

4. RESEARCH METHODOLOGY

Figure 5 summarises the methodology that was followed to complete this project. Using the literature study to gain an understanding of social manufacturing and bamboo manufacturing, respectively, a validation experiment will be set up to use social manufacturing to develop various different bicycle frame designs and the best design will be chosen to manufacture a bicycle frame. The standardizing procedure and the manufacturing process chain of the bicycle will be evaluated and compared to the results obtained through the literature study.



Figure 5: Research process steps followed in this study.

5. EXPERIMENTAL RESULTS AND DISCUSSION

Social Media will be the platform for the online community to simulate the social manufacturing experiment. In order to use Social Media data gathering is done by using social manufacturing in an industrial cluster, which is the 4th year students in the Advanced Manufacturing 414 course tutorial of the Industrial Engineering Department of Stellenbosch University. The tutorial is made up of 2 sections which should be finished in 2 weeks where the students are divided into 10 groups. The validation experiment followed the generic framework for social manufacturing.

The first week is the plan and concept generation phase from. In this phase the students must design free hand sketches of Bamboo Bicycles and upload these designs onto Social Media (Facebook, Instagram, Twitter and Dropbox for monitoring purposes). These designs must include attachments (Swiss army knife technology) that will improve the everyday life of someone living in rural Africa. These designs must then be promoted on Bamboo Bikes for Africa's Facebook-, Instagram- and Twitter pages. The design from each group with the most 'Likes' must then be used for the second week. This will act as the validation and marketing of the designs.

The second week the students will 3D CAD model the best design from week 1. This will act as the design phase (detailed design). These CAD models only include the frame of the bicycles along with the attachments (Swiss army knife technology). The students will then have to make a video around their design as to what makes it great. Both the video and 3D CAD model will then be posted onto Social Media to create awareness. These designs will then be used to manufacture the bicycles at Stellenbosch University and donate it to our local communities. The manufacturing process will be the production and support phases, however this is not in the scope of this case study.

A questionnaire was given to the students to determine their perspective and experience of social manufacturing. The questionnaire can give an indication to whether people would get involved with a social manufacturing company. The rest of the results captured in this study are based on the data from Facebook, Instagram, Twitter and Dropbox.

The next step is to analyse the data obtained from the students with regards to their designs and the data from the various social media platforms. From the 10 groups of students, they delivered 151 bamboo bicycle designs in one week. This meant that each group designed 15 bicycles on average.

The next step is to analyse the different Social Media platforms. The students were requested to promote all their designs on Facebook, Instagram and Twitter to see on which platform they could get the most Likes. Facebook got a total of 8472 Likes, Instagram 2013 Likes and Twitter 68 Likes. From these results we can see, because Facebook and Instagram focuses more on photos rather than text like Twitter, people reacted more to the photos rather than text.

In order to establish the amount of views of the page and posts the total reach is important. As illustrated in Figure 6, in 7 days the Bamboo Bikes for Africa posts reached 10552 people. This is significant as the students were only 89 people which transformed into a total reach of more than 10000. The total post reach proves the power of using Social Media as it is easy to connect to a very large online community in a short period of time.





Figure 6: Total post reach on Facebook page [1].

Week 2 of the project was used for the students to 3D CAD model, on Autodesk Inventor, the design from each group with the most likes from week 1. The CAD process needs to only show the frame of the bicycle along with the attachments that gives the bicycle its 'Swiss army knife' like technology.

The best design was chosen for the final Bamboo Bicycle that was manufactured. This manufactured bamboo bicycle is seen in Figure 7. The entire bicycle frame was manufactured from bamboo and the joints were made of a combination of steel and fiberglass.



Figure 7: Manufactured bamboo bicycle [22].

In Table 2 the standardizing strategies identified in the four steps of the sourcing process chain are rated according to the influence it has on the diameter, wall- thickness and the shrinkage. These three factors must be standardized and acceptable variances must be adhered to in order to maximize the usefulness of the bamboo when it is used for manufacturing of the bamboo bicycle. The summarized comment explains the reasoning behind the allocations.

Fable 2: The standardizing strategies ar	linked to the standardizing factors [15].
--	---

Strategy	Diameter	Wall thickness	Shrinkage	Comment
1. Good management practices	Low	Low		
2. Ecological and site condition	Medium	Low		Drier conditions lead to a smaller diameter [3].
3. Culm density	High	Medium		The density of the culms influences the diameter [16].
4. Allow to reach optimal mechanical and structural properties. (3-5 years.)		Medium		The age has an effect on the primary tissue, which influences the strength of bamboo [3].
5. Harvest at the right time of day and season.				This is related to the starch content which effects the preservation [16].
6. Establish allowable diameter range.	High			An allowable diameter variance for each tube.



7. Identify mature culms.	Me	dium		Bamboo reach their maximum strength when they are mature [3].
8. Low starch content				This effects the preservation of the bamboo.
9. Correct treatment for desired application				
10. Drying uniformly		Me	ledium	If it dries uniformly like losing moisture on one side it will contract unevenly [29].
11. Seal product after drying.		Hi	igh	Bamboo must be properly sealed off to prevent change in the moisture content [29].

The processes used to manufacture the bamboo bicycle was subject to available materials and machines at the University of Stellenbosch. It is therefore necessary to compare the process steps used with benchmark processes. The process steps utilized by each of the different process chains is displayed in table 3.

Table 3: Process steps utilized by each of the process chains (Adapted from [2] and [22]).

Process Steps	Bamboo Bikes for Africa 01	Bamboo Bikes for Africa 02	Bamboo Bicycle Club	Engineering Materials Development Institute	HERO Bikes
Heat treat the raw bamboo				x	x
Treat the bamboo with insecticide	x	х			
Dry bamboo	х	х		x	x
Bamboo selection	x	х	х	x	x
Prep bamboo for cutting	x	х	x		
Cut bamboo	x	х	x	x	х
Shape bamboo			x	x	
Rough bamboo					x
Remove bamboo material				x	
Outsource the joints				x	
Remove necessary joints from old bicycle	x				
Weld the parts to create the joints	x				
Sandblast the Joints	х				
Create joints using SLS		х			
Treat the joints	х				
Attach the lugs to the frame					x
Preliminary assembly	х	х	х	x	x
Shape the lugs					x
Assemble lugs					х
Apply fiber	x	х	х	x	х
Apply resin	x	x	x	х	x
Lug Preparation					х
Carbon-wrap					х
Filler	x				
Sanding	x	x	x	х	
Install brake bridge					x
Painting	x				
Varnish	x	x			x
	1				

Figure 8 shows the comparison of the manufacturing times between the different companies. It shows that HERO Bicycles and the Bamboo Bicycle Club have a very similar manufacturing time, while the first prototype of Bamboo



Bikes for Africa took substantially longer to complete. This can be attributed to the fact that the team had to remove the necessary components from an old bicycle before they could start the actual manufacturing process.



Figure 8: Comparison of manufacturing times between the different process chains [2]

There is a substantial cost difference between the Bamboo Bikes for Africa prototype and the rest of the process chains, as seen in Figure 9, even though the team had to purchase a bicycle to source all of the necessary components. This could be attributed to the fact that there is no markup added to the BBFA bicycle.



Figure 9: Comparison of manufacturing costs between the different process chains [2].



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

As the waste process could not physically be measured, the student distinguished between high or low waste during the manufacturing processes by using his own discretion through the knowledge he gained during the literature study. Because only the necessary components are used during the manufacturing process of the first prototype, the waste is high when manufacturing the joints. The 3D printing process has the lowest waste of all the stakeholders because every component is manufactured exactly to specifications. Figure 10 represents the material waste during the manufacturing processes of the different process chains.



Figure 10: Comparison of material waste during the manufacturing processes of the different process chains [2].

6. CONCLUSION

The business model elements of social manufacturing were explored and compared to traditional manufacturing methods. The research and conceptual design phase of social manufacturing when compared to traditional design phases were much shorter. This case study proved that by using social manufacturing, the design phase could produce more designs in a shorter amount of time than a traditional design method. Through analysing the designs, it was shown that an even higher amount of quality designs was generated by improving the first example design that was shown to the online community. The case study showed that by using an online community, a social manufacturing company can involve significantly more people than with a traditional manufacturing process. A bamboo bicycle was designed and manufactured, which is competitive with existing approaches. This paper indicates the business benefit of crowd sourcing and using an online community in the design phase of a manufacturing company.

7. REFERENCES

- [1] Ras, C.I. et al. Social Manufacturing Bamboo Bikes for Africa. IAMOT. [S.I.]: [s.n.]. 2016.
- [2] Oberholzer, J.F. et al. Evaluation of Resource Efficient Process Chains for Primary Manufacturing Processes of Bamboo Bicycles. 14th Global Conference on Sustainable Manufacturing. Stellenbosch: Elsevier. 2016.
- [3] Lopez, O.H. Bamboo. Bogota: [s.n.], 2003.
- [4] Ras, C.I. et al. Social manufacturing business model elements to support local suppliers. RAPDASA. Pretoria: RAPDASA. 2015.
- [5] Burmeister, C.; Lüttgens, D.; Piller, F.T. Business Model Innovation for Industrie 4.0: Why the "Industrial Internet" Mandates a New Perspective on Innovation, v. 0, p. 1-31.
- [6] Tao, F. et al. CCIoT-CMfg: Cloud computing and internet of things-based cloud manufacturing service system. IEEE Trans. Ind. Informatics, v. 10, n. 2, p. 1435-1442, 2014.
- [7] Oosthuizen, G. et al. Open Community Manufacturing. SAIIE, v. 1155, p. 1-14, 2014.
- [8] Porter, G.; Blaufuss, K.; Acheampong, F.O. Youth, mobility and rural livelihoods in sub-Saharan Africa: perspectives from Ghana and Nigeria. Africa Insight, p. 420-431, 2008.
- [9] Poverty and Inequality in South Africa. [S.l.]. 1998.
- [10] Koren, Y. Globalization and Manufacturing Paradigms. Glob. Manuf. Revolut. Prod., p. 1-40, 2010.
- [11] Zhang, F. et al. Modeling and analyzing of an enterprise collaboration network supported by serviceoriented manufacturing. roc. Inst. Mech. Eng. Part B J Eng. Manuf., v. 226, n. 9, p. 1579-1593, 2012.



- [12] Vukovic, M.; Kumara, S.; Greenshpan, O. Ubiquitous crowdsourcing. Proc. 12th ACM Int. Conf. Adjun. Pap. Ubiquitous Comput.- Ubicomp '10, p. 523, 2010.
- [13] Day, J.; Zimmerman, H. The OS1 Reference model. [S.l.]: Proceedings of the IEEE. 1983. p. 1334-1340.
- [14] Kagermann, H. et al. Recommendations for implementing the strategic initiative INDUSTRIE 4.0, n. April, p. 82, 2013.
- [15] Burger, M. D. et al. Strategies to Standardize Bamboo for Manufacturing Process Chains. 14th Global Conference on Sustainable Manufacturing. Stellenbosch: Elsevier. 2016.
- [16] Liese, W.; Kohl, M. Bamboo, The Plant and its Uses. [S.l.]: Springer International Publishing, 2015.
- [17] BAMBOO Plant for Sale, 2016. Disponivel em: http://www.brightfields.co.za/bamboo-farming/>.
- [18] Schroder, S. Bambusa Balcoa. Guadua Bamboo, 2016. Disponivel em: http://www.guaduabamboo.com/species/bambusa-balcooa.
- [19] Jansen, J.J. Designing and Building with Bamboo. Technical University Eindhoven. [S.l.]. 2000.
- [20] Bamboo Bike Supplies. Bamboo Tube Set. Bamboo bike supplies, 2016. Disponivel em: http://bamboobikesupplies.com/products/bamboo-tubeset- Acesso em: 13 April 2016.
- [21] Efford, M. Choosing the bamboo for your frame. Bamboo Bikes Australia, 26 July 2010. Disponivel em: http://www.bamboobikes.com.au/2010/07/26/choosing-the-bamboo-for-your-frame/>. Acesso em: 19 April 2016.
- [22] De Wet, P. et al. Resource Efficient Processchain Development of Secondary Manufacturing Processes. Stellenbosch: [s.n.]. 2016.
- [23] Brightfields. Bamboo plant for sale. Brightflieds, 2016. Disponivel em: http://www.brightfields.co.za/bamboo-farming/. Acesso em: 15 April 2016.
- [24] HERO BIKE. Harvest your bamboo. Hero Bike, 2016. Disponivel em: http://herobike.myshopify.com/pages/step-1. Acesso em: 21 April 2016.
- [25] Garay, A.C.; Amico, C.; De Oliveira, F.H. The use of bamboo culms in bicycle frame construction. 7th International Symposium on natural polymers and composites ISNAPOL 2010. Sante Andre: [s.n.]. 2010.
- [26] Arnone, J. et al. Engineering a Bamboo Bicycle. Worcester. 2015.
- [27] Beehive, C. Bamboo for Bicycles PART 2. Cozy Behive, 7 February 2008. Disponivel em: http://cozybeehive.blogspot.co.za/2008/02/bamboo-bicycles-part-2.html. Acesso em: 5 May 2016.
- [28] BNC. Bamboo Mountain frame. BNC, 2015. Disponivel em: http://bncbicycle.com/bamboo_frame_mountain.html>. Acesso em: 5 May 2016.
- [29] Calfee, C. Building bamboo bicycles. Bamboosera. Disponivel em: http://www.bamboosero.com/technical/building-bamboo-bikes/>. Acesso em: 13 April 2016.
- [30] The Wood Database, 2015. Disponivel em: http://www.wood-database.com/lumber-identification/monocots/bamboo/. Acesso em: 21 April 2016.
- [31] Nyberg; Timo. Workshop Sustainable and ocial Manufacturing. Global Cleaner Production and Sustainable Consumption. [S.l.]: Elsevier. 2015.
- [32] Steenkamp, L.P.; Ras, C.I.; Oosthuizen, G.A. Emerging Synthesis of Social Manufacturing. Stellenbosch: COMA. 2016.
- [33] Ukoba, O.K.; Ogunkoya, A.K.; Soboyejo, W. Development of an Eco-Friendly Bamboo Bicycle. The Pacific Journal of Science and Technology, v. 12, n. 1, p. 102-108, 2011.



FAILURE ANALYSIS OF LOCOMOTIVE ALTERNATORS USING THE SIX SIGMA METHODOLOGY

A. Mekwa¹, M.G. Kanakana² and K. Mpofu³

¹²³Department of Industrial Engineering Tshwane University of Technology, South Africa <u>Mekwa.Amos@yahoo.com</u>, <u>KanakanaMG@tut.ac.za</u>, <u>MpofuK@tut.ac.za</u>

ABSTRACT

The series 6E diesel electric locomotives use an alternator to produce power to the traction motors that are mounted to the bogies to drive the locomotive. Company XYZ is the largest repairer of the locomotive alternators which refurbishes an average of 675 locomotive alternators per annum. Locomotives are coupled with wagons to supply commodities such as iron and ore in South Africa. Company XYZ faced high failure rates due to locomotive alternators defects. These defects led to service issues and revenue losses. Therefore, the company embarked on a project to investigate strategic initiatives to reduce failure the rate. Six Sigma DMAIC methodology was identified as an initiative that focused on improving the quality of the locomotive alternator by reducing failure the rate. The current Six Sigma level was at 3.8 sigma with the defect per million (DPMO) of 11481 at 675 units annually. Of the 675 refurbished locomotive alternators produced annually, 18% had a failure rate. Through the implementation of the DMAIC methodology, insulation on an armature was the highest type of failure. A resolution was proposed that each locomotive alternator that had operated more than 12 months in-service must undergo complete armature insulation. This proposal reduced the failure rate to 6%, thus yielding improved Sigma level of 4.2 with a DPMO of 3457 at 675 units.

Key Words: Six Sigma, Product Quality, Variation, Product Defects

¹The author was enrolled for the M-Tech in Industrial Engineering at the Tshwane University of Technology

^{*}Corresponding Author



1. INTRODUCTION

Mittle [25] opined that, in simple terms, a locomotive alternator or electrical alternator is an electrical machine whose main purpose is to convert electrical energy into mechanical energy. This locomotive alternator is also known as a synchronous electrical machine that belongs to the electrical rotating machine family which includes; generators, convertors, motors and transformers [5]. A locomotive alternator is a critical assembly component that results in the stoppage of a locomotive whenever it experiences failure.

The main parts that form a locomotive alternator are the; armarture, frame and end-shield, where the armarture is the rotating part of the alternator that has a shaft, and the stationary part of the alternator comprise the frame and end-shield as illustrated in Diagram 1 below.



Diagram 1: Locomotive alternator assembly

Whenever the failure occurs with a locomotive, the customer updates the failure on the non-conformance report system. Due to a high prevalence of locomotive failures recorded on the non-conformance report system, data was extracted from the year 2014 and analysed. It was discovered that the failures of a locomotive alternator were the cause of such occurrences. Through this data a project was initiated to analyse the failures of locomotives alternator by use DMAIC methodology. Data collection was carried out though a visit to the company XYZ production plant.

The refurbishing process followed was analysed and consultations with Quality Assurance officers were conducted, wherein data of monthly recorded failures was sourced. This data was later used in data analysis phase. Therefore, Analyses was conducted and recommendation was made to improve the failure rate.

2. LITERATURE

This study was characterised by the analysis of the failures of locomotive alternators using the Six Sigma and DMAIC methodology. The author focused on Six Sigma, product quality, variation and defects when reviewing the relevant literature.

2.1. Definitions

Six Sigma is known as a methodology or a philosophy that rigorously focuses on quality improvement by reducing the variation and measurements of defects per million opportunities [12]. Any variation from customer specifications on either a product or a process is categorised as a primary problem to be solved using this methodology [14]. Six Sigma is also defined as a philosophy that is data driven to eradicate defects in the process of a product or service. In the early 1980's the methodology was developed and recognised by the



Motorola company [3]. A product defect refers to the failure of a product in its purpose and the product is not fit for use and does not conform to specification [2]. Product quality refers to the ability of a product to satisfy the expectations of a customer [29]. Variation is the term that means a change in difference or a change in a form, position or condition [3].

2.2. Quality of a product

Over the years, quality has been viewed as one of the primary order winners of why customers choose a particular product or service [31]. Many companies have witnessed their corporate image being damaged by a reputation of poor quality where some organisations use quality as one of their marketing strategic initiatives to capture the clients. Many organisations view high quality as "more money", although literature reflects "doing things right the first time" as a definition of quality [2]. It is therefore imperative for Company XYZ to produce good quality service to avoid reworking that will lead to high non-conformance costs, as well as increase the availability of locomotives to move the tonnages that will meet customer specifications.

2.3. Six Sigma and DMAIC methodology in manufacturing organisation

In manufacturing organisations, Six Sigma contributes to improving the processes and adds value to the product. It reduces the variation of problems that are complex to solve and could lead to high risks if left unresolved [14]. The DMAIC methodology is not very different from other common process improvements methodologies. However it has always been known as a proven methodology that delivers results by using data and statistical analytics. The five step systematic approach of DMAIC (Define, Measure, Analyse, Improve, Control) constitutes the classic Six Sigma problem-solving process. Traditionally, the approach is to be applied to a problem within an existing, steady-state process or product and service offering [12]. The Six Sigma philosophy has been adopted worldwide in manufacturing companies to improve their financial status as well as to improve quality and process performance in order to ensure that the product conforms to specifications. This methodology increases the process performance and measures process capability by means of statistical measurement [11].

There are several studies in the literature regarding the application of Six Sigma to reduce the failure rate or defect of a product; for example, amongst others, a study undertaken to "reduce the failure rate of the Screwing process with Six Sigma approach" The aim of the research conducted by Nihal Erginel was to reduce the failure rate of the screwing process, where the failure rate was decreased from 30% to 14%. Nihal Erginel followed five sequential phases known as DMAIC to conclude the study [13].

A study on the "reduction of welding defects using Six Sigma techniques" was conducted. This study focused on identifying the root causes of failure in the welding process through the implementation of Six Sigma to eliminate the defect rate in the final product. This project aimed at improving customer satisfaction and product quality. It was also noted that in order for one to understand and follow Six Sigma disciplines, the SIPOC (Supplier, Input, Process, Output, and Control) identification needed to be carried out in order to document the welding process from beginning to end [15].

Another Indian company has successfully undergone a project to implement Six Sigma "to reduce defects of a grinding process". This project was instigated in an automotive company. The application of the methodology resulted in decreasing the defects from 16.6% to 9%. The project also yielded positive results in the reduction of scrap costs and the number of employee hours spent on rework and as well as an increment in production output. Savings of approximately \$2.4million was reported as a result of this project. Common statistical tools such as a cause and effect diagram, process capability, ANOVA etcetera, were used in the study [18].

3. RESEARCH METHODOLOGY

3.1. Problem statement

Company XYZ had been experiencing a high failure rate of locomotive alternators. The quality assurance department saw a breach in service level agreements set at 6% rising to 18% and sudden spike in incurred maintenance costs. Data sourced from the non-conformance report system revealed that for each train set that could not operate, a loss of potential income of R7m was realised. Subsequent to this problem, a refurbished locomotive alternators would now needed to be re-serviced and thus the company incurs a maintenance cost of R65 000 per locomotive alternator. These problems affected the entity's profitability and customer service.

The problem statement was introduced through the investigation of non-conformance report system by tapping into the data that was used as feedback from the customer regarding quality issues of locomotive alternators. Through the analysis of 2014 data, it was noted that there are various root causes that explain why Company



XYZ was facing high failure rates of locomotives alternators. The project was therefore initiated to analyse the failures of the locomotive alternators by employing the Six Sigma and DMAIC methodology.

3.2. Objectives

The objectives of this study were to;

- Determine the root cause of failure in the locomotive alternators;
- Improve the failure rate from 18% to 6%;
- Improve the Sigma level from 3.8 to 4.2 sigma and
- Implement process control to ensure sustainability of the improvements.

3.3. Collection of data

To critically comprehend the problem statement, it was imperative to collect data regarding the failures of locomotive repairing process, and the locomotive alternator specifications as well as the non-conformance reports. The DMAIC methodology was followed to ensure a systematic problem solving approach.

3.4. Methods and dispersion of data

Six Sigma and DMAIC methodology was employed in command in order to better comprehend the failures experienced with the locomotive alternators. Six-Sigma comprises a rigorous, focused and highly effective implementation of proven quality principles and techniques [10]. The systematic methodology was employed in the following phases:

Define Phase: In this phase the project charter that comprises the project goals, scope, timeline and estimated benefits were defined together with the project leader, coach, sponsor and champion. The project team consisted of the engineering manager, quality controller, industrial engineer and a Six Sigma specialist (Black belt). A Pareto diagram was constructed in order to indicate the high contributors to failures. To establish a clear scope of the project or research, SIPOC (Supplier input Process Output Customer) aimed at summarising the inputs and outputs of a process of repairing and locomotive alternator AS IS process was defined in order to detail the high level mapping of the current process of repairing locomotive alternators.

Measure Phase: A data collection plan sheet was used to develop a sound data collection plan. To determine the baseline performance, graphical tools such as a control chart and histogram were used. Calculation of Sigma Performance (Defect per Million Opportunities) was performed to determine the Sigma level.

Analyse Phase: Once the sigma level was determined, the project team needed to identify the potential root causes of the locomotive alternator failures. The data were statistically analysed in order to obtain confirmation of the root causes and in turn to identify the contributors to the failure of the locomotive alternators. Tools such as fishbone diagram (cause and effect), FMEA (Failure Mode and Effects Analysis), and control charts, normality tests and correlation analysis were employed. Software such as Microsoft Office Excel, Six Sigma Excel and Mini-Tab were used as tools to graphically display the data.

Improve Phase: The main aim of this phase was to a identify solution that would prevent the failures of the locomotive alternators. During the brainstorming and generation of possible solutions, techniques such as using six thinking hats were employed by the project team. The proposed solutions were prioritised and selected by means of solution prioritization and selection matrix. In order to reduce the locomotive alternator failures, the To-be Process was defined by introducing more quality hold points and the FMEA (Failure Mode Effect Analysis) focused on the high priority items.

Control Phase: The purpose of the control phase was to implement the control mechanisms of process procedures to ensure that the improvements of the process are sustained. This includes the development of a process monitoring plan called the failure early detection system. Computation of the Sigma level was performed in order to compare the baseline Sigma level with the new Sigma level after the improvements has been implemented.



4. DISCUSSION OF RESULTS

4.1. Define Phase

4.1.1. Project charter

In order to provide a shared understanding from a bird's eye view for all the main stakeholders, a project charter was constructed with the main aim to outline and identify the scope, objectives, deliverables, constraints, expected benefits and main stakeholders of the project. Figure 1 comprises the project charter that detailed how the project was scoped with the estimated benefits, timelines and project goals.

Project Charter

Project Overview

Business Case			Project G	oals		
Locomotive alternator con outputs significantly and c hauling tonnages are impo	tributes to the performance of Lo consequently company XYZ object erative to ensure the economic gr	comotive ctives of owth of the	Reduce failure rate from 18% to 6%			
country. The business has electric motor alternator a	s to meet the customer requireme nnually to be assembled without	nts of 675 anv defects.	Estimated	d Benefits (R)		
Currently the business ha	s failure rate of 18% annually. Th	e 124	Nen confe	remember and the D7.9		
average cost of R72k per	motor. As a result of the deficit, t	s at an he business	Non conic	innance cost reduced by R7.0		
losses on average R744k	per month.		Project S	соре		
			Area of Fo Includes: Excludes: Start Dain	ocus: Locomotive alternator Re Armature Inspection, Shaft rep Commutator repair and End-s	epairing Process pairing Section shield Section	
Problem Statement			Start Point: Stripping Section Stop Point: Assembly Section (Fitting armature into frame)			
Failures on locomotive alt XYZ meeting customer ex The current failure rate is	Failures on locomotive alternator impedes the achievement of company XYZ meeting customer expectation of 675 electric motors per annum . The current failure rate is 18% annually			Expected Completion Date		
			Measure	End Feb 2016		
			Analyze	End Mar 2016		
Project Resources			Control	End May 2016 End Jun 2016	-	
		Core Tear	n Members:	Subject Matte	er Experts (SMEs):	
Project Leader (s):	Amos Mekwa	Bongani I	Nhlapo	Andile N	lqodi	
Project Coach:	Tebogo Makhafola	Matome I	Makgoba	Edgar N	lothapo	
Project Sponsor (s):	Sondaha Maivha	Mpho Tla	di			
Project Champion:	Chris Fourie					

Figure1: Project Charter

The project charter is a crucial part of any Six Sigma project to ensure that the project does not deviate from the expected outcomes and goals. The main aim of the initiated project was for the identified project team to reduce the failure rate from 18% to 6%.

4.1.2. Pareto

In order to focus on the highest occurring failures and graphically summarises the failures by categories, a Pareto diagram was initiated. This graphic assisted the project team in analysing which problems required immediate attention and illustrated which failures exerted the greatest effect on the locomotive alternator failures by applying the 80/20 principle. Diagram 2 indicates that insulation failure had the highest failure rate compared with other failure rates. Thus the focus was on the insulation failure that occurs during the armature repairing process.





Diagram 2: Pareto graph

Diagram 3 depicts armature that has a failed insulation. The photograph was taken by a quality officer during the investigation period.



Diagram 3: Picture of Insulation failure

4.1.3. SIPOC

The SIPOC (Supplier Input Process Output Customer) identification process was conducted with a view to summarising the inputs and outputs of the process of repairing locomotive alternator (high level process map). The focus fell on the armature repairing process that failed insulation quality. Figure 2 depicts the SIPOC process in this regard.



SIPOC

Define Phase

Process Start: Stripping Section					
Process End: Assembly Section (Fitting Armature into a Frame)					
Suppliers	Inputs	Process	Outputs	Customers	
Stripping Section	Defective Armature	- Rewind - Pull Down & VPI - Balancing - Shaft Inspection & Repair	Good Armature	Assembly Section (Fitting Armature into a Frame)	

Figure 2: Suppliers Input Process Output Customer identification

4.1.4. As-Is Process

Diagram 4 depicts the high level process of repairing a locomotive alternator. The first step of repairing the locomotive alternator is to receive the defective locomotive alternator. Thereafter, it is necessary to clean the component, inspect, strip, inspect the armarture, fix any defects, clean the frame, repair the coils, inspect the end-shield, repair any defects, fit the armature into the frame, test the locomotive alternator, paint it, and finally perform quality assurance.



Diagram 4: As-Is Process

The main focus fell on the process of repairing the armature where the insulation failure occurs, as highlighted in red. A lean event was also conducted in the form of a Just-Do-It project in order to remove all the non-value adding activities that are not critical to customer specifications.



4.2. Measure Phase

4.2.1. DPMO

The DPMO is one of the significant Six Sigma metrics employed to determine how effective the process is. This process was significant to this study in order to establish a baseline level performance. After the process improvements had been initiated, the DPMO was recalculated to gauge performance gains. The current defect per million (DPMO) was determined at 11481 with 675 units annually as shown below at the Six Sigma level of 3.8 sigma. Out of average total of 675 refurbished locomotive alternators annually, a failure rate of approximately 18% was experienced.



DPMO

Sigma Level

11481

3.8

Defects	124
Opportunities	675
Defect Opportunities per unit	16

Six Sigma Table:

1	690,000
2	308,000
3	66,800
4	6,210
5	320
6	3.4

Calculation of DPMO

$$DPMO = \frac{124}{675 \times 16} \times 1,000\ 000\ \dots\ \dots\ \dots\ (1)$$
$$DPMO = 11481\ \dots\ \dots\ \dots\ (1)$$

Therefore, the Sigma level was calculated by taking into consideration the calculated DPMO using Sigma Excel. The Sigma level of 3.8 indicates that about 6.7% of the time defects are transpired during the process of repairing locomotive alternators.

4.2.2. Histogram

To evaluate the current situation and to study the outcomes of the improvements, a histogram was used to display the numeric data. The histogram shape assisted in deciding how to improve the current situation of failures. Figure 3 indicates that most of the data falls on the left side and fewer on the right side. The mean in Figure 3 indicates that an average of 10.33 failures per month occurred over a 12 month period.



Figure 3: Histogram of the number of failures of locomotive alternators

The data in the histogram is relatively skewed to the right where it indicates that the mean failure is greater than median failure of 8.5.



4.2.3. Normality test

To avoid statistical errors and curb the assumption that the data would follow a normal distribution, a normality test was conducted by means of the Anderson-Darling Test. This test was recommended due to its precision and high power when testing whether the data followed a normal distribution or mean. The normality test was conducted to validate the data in order to determine whether it is normal or not (refer to Figure 4 which demonstrates that the data are normally distributed to the mean).



Figure 4: Normality test of locomotives alternator failures

The p-value was calculated at 0.0829, which is larger than 5%. Due to a lack of evidence that data in the graph is not normally distributed, the decision was to fail to reject the null hypothesis.

4.2.4. Control chart

A control chart was adopted to portray how the processes or failures change over time. The computed control chart depicts that there was a special cause variation in January where the point fell above the upper control limit. Further analysis and investigation were conducted regarding this special cause variation to ensure that it is permanently eliminated from the process. Figure 5 indicates the number of failures per month where there were more failures in January than other months.



Figure 5: Control chart of the number of failures of locomotives alternator per month

It is commonly known that the failures occur mostly during the hot summer weather and the rainy season, meaning that the increase in number of failures is dependent on the weather conditions. The project team was gathered to craft a way forward on why a special cause occurred. A cause and effect diagram was used to further investigate the causes of the increase in failures of the locomotive alternators.



4.3. Analyse Phase

4.3.1. Fishbone

To analyse the root causes of the insulation failure, the team used the fishbone diagram below, which also indicates that an increase in the number of failures occurred during spring and summer. The results of the fishbone graphic in Diagram 5 indicate that both methods and mother-nature were the main root causes of why the locomotives alternators experience insulation failure. The two main root causes that contributed most to the increase in the locomotive alternator failures to being 18% were further evaluated using Failure Mode and Effect Analysis to prioritise the root cause that exerted such a major impact.

Fishbone Diagram

Analyze Phase



Diagram 5: Cause and effect of locomotives alternator failures

The transfer function Y=f(x) was used to identify all the possible causes (x) that impact Locomotive alternator insulation failure (Y) and to classify a relationship between the output and input, The table in Figure 6 displays all the x data that were evaluated as well as their relationship.

	X1	X2	X3	X4	X5	X6	X7	X8	X9
Possible causes	Weak insulation material	Positioning of insulation	Too much insulation	Wrong Voltage applied	Wrong connection	Foreign object between armature & coils	Moisture	excessive temperature	Wrong dimenstions on insulation
Relationship	Μ	L	м	м	M	M	н	н	M

L= Weak Relationship

M= Moderate Relationship

H=Strong Relationship

Figure 6: Y=f(x)

Through this transfer function, it was found that one variable has a weak relationship. The variable 6 has a moderate relationship and 2 has a strong relationship. Therefore, moisture and excessive temperature were the root causes to be addressed in a solution.



4.3.2. FMEA (Failure Mode and Effects Analysis)

Through an evaluation to prioritise which root cause has the main impact, it was observed that mother nature was the main cause of failure owing to excessive temperature and moisture, as highlighted in red in the Figure 7. The results of the FMEA depict that excessive temperature was a root-cause which ranked highly at 288.

Deadwet Trees	Possible	E-iluna Marda	Causes of		Ranking (1 – 10)		Risk Priority Number	Recommended			
Failure	Failure	Failure Mode	Failure	[O]Occurrence	[S]Severity	[D]Detection	Ox SxD	Actions			
Locomotive Insulation failure on Alternator armature		Positioning of insulation material	Man	3	7	5	105	Implement QA hold point			
		Too much insulation on coils	Man	3	7	3	63	Implement QA hold point			
	Insulation failure on armature	High/Wrong voltage applied during testing	Man	4	7	3	84	Follow Standard Operating Procedure			
		Insulation failure on armature	Insulation failure on armature	Insulation failure on armature	Foreign object between armarture and coils	Loose objects	2	8	2	32	Early Detection
			Moisture	Rain	5	8	6	240	Early Detection		
		Experies	Operating Condition	6	8	6	288	Early Detection			
				temperature	Process (over bake) of armature	5	8	6	240	Follow Standard Operating Procedure	

Figure 7: Failure Mode and Effects Analysis

4.3.3. Correlation analysis

In the graph below, the increase in failures being dependent on weather conditions is validated. Since the evaluation in Figure 7 above indicated that moisture and excessive temperature are the main root causes of the increase in failure rates rising to 18%, indeed the excessive temperature cause an increase in the failures. Figure 8 indicates that in summer, when temperatures are high; there is an increase in the failures of the locomotive alternators.



Figure 8: Relationship between failures and temperature

To establish whether there is a relationship between two variables, namely failures and temperature, correlation analysis was performed. The graph in Figure 9 portrays a straight line with a positive slope where the Pearson correlation coefficient (r) was used as a measurement to determine the significance between these two variables.





Figure 9: Correlation analysis of temperature and number of failures

Calculation of Pearson correlation coefficient (r)

$$\boldsymbol{r} = \frac{1}{n-1} \left\{ \frac{\sum x \sum y \, (x - \bar{x})(y - \bar{y})}{s_x s_y} \dots \dots (2) \right\}$$

where;

n = 12 $\bar{x} = 10.33$ $\bar{y} = 26.92$ $s_x = 5.58$ $s_{xy} = 3.65$

The r is calculated at 0.6, which indicates a strong positive correlation where r is close to +1.

4.4. Improve Phase

4.4.1. Solution prioritisation matrix

Three solutions were suggested by the project team, namely each locomotive alternator that has been in service more than 12 months must undergo complete re-insulation, increase number of quality check point and perform regular audits on a process to ensure adherence, change the current insulation type. All the solutions were ranked using solution prioritization matrix. It was therefore found out that solution that each locomotive alternator that has been in service more than 12 months must undergo complete re-insulation was recommended due to the fact that it addresses he problem at hand and it is a quick win that provides long-term solution. The quality hold points will also be added to prevent the armature from being over-baked during the repair process.

Solution Description	Root Cause Addressed	CTQ Impact (H, M, L)	Implementation Effort (H, M, L)	Quick Win (Yes, No)	Technology Required (Yes, No)
Re-insulate Alternators> 12 months in service	Excessive temperature- Operating condition	н	L	Yes	No
Change/Improve insulation material	Excessive temperature- Operating condition	L	н	No	Yes
Increase Quality Hold Points	Excessive temperature- Process Over bake armature	н	L	Yes	No

Figuro	۰.	Colution	prioritization	matrix
rigure	9:	Solution	prioritization	matrix

To overcome the problem of insulation failure by performing re-insulation and adding quality check points, the project team proposed that each locomotive alternator that has been in service for more than 12 months must undergo complete insulation repair rather than only outer insulation. This process would serve as preventative



maintenance to avoid unscheduled breakdown. Locomotive alternator change-out or replacement was also recommended to be included on scheduled service maintenance of the locomotive provided that the alternator had been in service for more than 12 months.

4.4.2. DPMO (Improved)

After implementing the solution, data were collected over months with data points in order to calculate an improved DPMO and Sigma level to gauge the success and check whether improvement solution addressed the problem at hand. A new Sigma level was calculated at 4.2 sigma level. This demonstrates an improvement of 12% of the number of failures experienced.



The Sigma level of 4.2 indicates that only defects are transpired about 0.62% of the time during the process of repairing the locomotive alternators.

Calculation for DPMO

$DPMO = \frac{42}{675 \times 18} \times 1,000\ 000\ \dots\ \dots\ (1)$ $DPMO = 3457\ \dots\ (1)$

Therefore, the Sigma level was calculated by taking into account DPMO using Sigma Excel. Through this solution, the failure rate was significantly reduced to 6% and the improved Sigma level was 4.2 Sigma level with DPMO of 3457 at 675 units.

4.5. Control Phase

The refurbishment process was required to include the re-insulation on the armatures whether it is for minor or major repair. Quality checks also needed to be performed on the process followed during refurbishment, and lastly, regular audits were required to check adherence to the refurbishment process and ensure that he corrective measures that were implemented are monitored and remain controlled. A regular trigger reminder needed to be implemented through a system that will enhance the tracking process of ensuring that each locomotive alternator undergoes intensive refurbishment service after it has been in service for more than 12 months.

The main aim of the system was to early detect and trace each and every alternator that was repaired and has been operating in a locomotive for more than 12 months. All those locomotive alternators mainly minor repaired ones were identified as stood a chance of the insulation failure if during the first repair process they did not undergo complete re-insulation.

The system utilised data from the ERP system, where the mathematical model was designed to check the last date of the repair of a production order completion date and date of delivery to the customer by matching the serial number of the alternator. As a result, the mathematical model enabled the system to produce a list of alternators for which insulation failure is a risk. The system user was informed by an early warning message trigger, to start repair process of such locomotive alternators.

The process of repairing the alternator was improved. An early detection system was implemented with the aim to prevent insulation failures of alternators before they occur. The new system enabled the organisation to prevent failure from occurring by determining the reliability rate which can be used as a filter for the preventative measure. This ensures that all the alternators are of high quality and are reliable before fitment.





Diagram 5: Locomotive alternator failures early detection system

5. RECOMMENDATION

Company XYZ had experienced a high failure rate of locomotive alternators that led to potential income losses and re-incurred maintenance costs. Through the Six Sigma DMAIC methodology, it was experienced that during the refurbishment process, the insulation on the armarture was neglected.

Furthermore, the refurbishment process required to include insulation on the armatures whether it is for minor or major repair. Quality checks need to be performed on the process followed on refurbishment, and lastly, the regular audits were required to check adherence to the refurbishment process. A regular trigger reminder needed to be implemented through a system that will enhance the tracking process of ensuring that each locomotive alternator undergoes intensive refurbishment service after it has been in service for more than 12 months.

Implementation of the above recommendations had inherently improved the entity's customer service by being able to provide the client with good quality locomotive alternators, thus ensuring income. The service level agreement was also improved and the failure rate was significantly reduced to 6 %. The maintenance cost incurred was greatly minimized as the process rejuvenation curbed the problem of failures of locomotive alternators.

6. CONCLUSION

Six Sigma aims to achieve virtually error free business performance. This project was aimed at improving customer service and product quality. It was critical to follow the DMAIC methodology step by step in order to arrive at a solution to resolve the failure rate of locomotive alternator. Furthermore, there was a need for Six Sigma DMAIC methodology because all the decisions that were made were data driven.

The following findings were observed after piloting this research on a failure analysis of locomotive alternator using the Six Sigma and DMAIC Methodology:

- The current Six Sigma level was determined at 3.8 sigma with a defect per million opportunities (DPMO) of 11481 at 675 units annually.
- It was recognised that insulation failure was the main contributor failure due to mainly moisture and excessive temperature.
- Excessive temperature was determined as the root causes for the problem.



- To overcome this problem, the project team proposed that each locomotive alternator that has been in service more for than 12 months must undergo complete insulation rather than outer insulation and two quality check holding points were added.
- Therefore, a quality assurer assigned to monitor the age of insulation using an early detection system was found to be an effective control measure for the problem in the control phase.
- The improved Sigma level was 4.2 sigma with a DPMO level of 3457 at 675 units
- Through this solution, the failure rate was reduced significantly from 18% to 6%. Although these results do not yield zero failures, other root causes of failure such as bearing failure, commutator failure, mechanical integrity failure, winding failure and vibration failure can be eradicated by means of the same methodology.

7. ACKNOWLEDGEMENT

The author would like to express sincere gratitude to the project team members, electrical engineering specialist and Tshwane University of Technology. Their contribution and support is duly acknowledged.



8. REFERENCES

- [1] Kuncar, D., Erturk, U. and Spenhoff, E. 2003. Six Sigma, DMAIC Guide Book.
- [2] Juran, J.M. 1999. Juran's Quality Handbook, 5th Edition, McGraw-Hill Education, New York.
- [3] **Pyzdek, T.** 2003. *The Six Sigma handbook*, A Complete Guide for Green Belts, Black Belts, and Managers at All Levels, McGraw-Hill Education, New York.
- [4] Wolff, D.E., Thomas, P. and Ralph, J. 1987. Quality Improvement Process and Cultural Change, IMPRO 1987 Conference Proceedings, Juran Institute, Inc.
- [5] Jasper, I., Satia, H.S. (2012-11-05). Electrical Machine, http://www.slideshare.net/fikakhamis/advantages-and-disadvatages-of-acdc-motor
- [6] Toliyat, H.A. and Kliman, G.B. 2004. Handbook of Electric Motors, 2nd Edition, CRC Press.
- [7] Gottlieb, I. 1997. Practical Locomotive Motor Handbook, Newnes, Oxford.
- [8] **Rao, J.S.** 2011. History of Rotating Machinery Dynamics, 1st Edition, Series Volume 20, Springer Netherlands.
- [9] **Crosby, P.B.** 1979. *Quality is free*, The art of making quality certain, <u>www.freequality.org/documents/Training/Quality%20is%20free.ppt</u>
- [10] Chowdhury, S. 2002. The power of Design for Six Sigma, Kaplan, Inc., Florida.
- [11] Snee, R.D. 2004. Six Sigma: The Evolution of 100 years Business Improvement Methodology, <u>http://www.inderscienceonline.com/doi/abs/10.1504/IJSSCA.2004.005274</u>
- [12] Fursule, N.V., Bansod, S.V. and Fursule, S.N. 2012. Understanding the Benefits and Limitations of Six Sigma Methodology, Vol.2, pp 1 -7.
- [13] **Erginel, N.** 2014. *Reduce the Failure Rate of the Screwing*, Proceedings of the 2014 International Conference on Industrial Engineering and Operations Management Bali, Indonesia.
- [14] Valles, A., Sanchez J., Noriega, S., and Nunez, B.G. 2009. Implementation of Six Sigma in a Manufacturing Process: A Case Study, International Journal of Industrial Engineering, Vol. 16, pp 171 181.
- [15] Soni, S., Mohan, R., Bajpai, L. and Katare, S.K. 2013. *Reduction of welding defects using Six Sigma techniques*, International Journal of Mechanical Engineering and Robotics research, Vol. 2, pp 404 412.
- [16] **Berty, E.** 2011. *Cigarette reject rate reduction using Six Sigma approach*, Master Thesis Project, Department of Production Engineering and Management.
- [17] **Deepak, Garg, R. and Arya, S.** 2015. *Reduction of reducing rate using six sigma*, International Journal of Exploring Emerging Trends in Engineering, Vol. 02, Issue 04, pp 202 209.
- [18] **Gijo, E.V., Scaria, J. and Antony, J.** 2011. *Application of Six Sigma Methodology to Reduce Defects of a Grinding Process*, A case study. Quality and Reliability Engineering International.
- [19] **Titus, R.J.** 2009. Six sigma: success and failure stories, LEHIGH University.
- [20] **Creswell, J.W.** 2003. Research design Qualitative, Quantitative and mixed methods, 2nd Edition, SAGE Publication, London.
- [21] Kothari, C.R. 2004. *Research methodology methods and techniques*, 2nd Edition, New Age International Publishers, India.
- [22] Frerichs, R.R. 2008. Simple Random Sampling (SRS).
- [23] **Perez, R.X.** 2008. Using fault trees to determine root cause of failures, proceeding of the tenth International pump users symposium, pp 185 193.
- [24] Marshall, J. 2012. An Introduction to Fault Tree Analysis, Product Excellence using Sigma Module, The University of Warwick, pp 1 18.



- [25] Mittle, V.N. 2005. Design of electrical Machines, Standard Publishers Distributors, India.
- [26] **Tavner, P., Ran L. and Howard S.** 2008. *Condition monitoring of rotating electrical machines*, The Institution of Engineering and Technology Publication, London.
- [27] Bonnett, A.H. 2000. Root Cause AC Motor Failure Analysis with a focus on Shaft Failures,<u>https://www.researchgate.net/publication/3170995_Root_cause_AC_motor_failure_analysis_with_a_focus_on_shaft_failures</u>
- [28] Smith, M.J. 2014. *Statistical Analysis Handbook*, A comprehensive handbook of statistical concepts, techniques and software tools, <u>www.statsref.com</u>
- [29] Owen, D.G. 2002. Manufacturing Defects, University of South Carolina, Columbia
- [30] Zairi, M. 1991. Total Quality Management for Engineers, Woodhead Publishing Limited, India.
- [31] Brito, E.P.Z., Aguilar, R.L.B., Ledur, L.A. 2004. Order Winners, Order Qualifiers, and Quality Perception in Service Operations, pp 1 -17, <u>http://www.anpad.org.br/diversos/trabalhos/EnANPAD/enanpad_2004/GOL/2004_GOL1546.pdf</u>



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



STRUCTURING DATA FOR A RSA SECTION 12L ENERGY EFFICIENCY TAX INCENTIVE APPLICATION

HM Janse van Rensburg^{1*} & W Booysen² & SW van Heerden³

 ¹ CRCED Pretoria, North-West University, South Africa mjvrensburg@rems2.com
² CRCED Pretoria, North-West University, South Africa wbooysen@rems2.com
³ CRCED Pretoria, North-West University, South Africa swvanheerden@rems2.com

ABSTRACT

South African industries are under pressure to remain internationally competitive. By investing in energy efficiency, companies can lower operating costs. Section 12L of the Income Tax Act, (Act No 58 of 1962) has been implemented to incentivise energy efficiency. To receive the benefit, an application must be compiled by an independent South African National Accreditation System (SANAS) accredited Measurement and Verification (M&V) team. M&V teams are not necessarily familiar with the details surrounding a sites initiative implementation, assistance from industry is required. This article investigates the M&V requirements to understand the complexities involved in analysing facility energy consumption. Different data management techniques are reviewed to identify approaches to handle the large volumes of data generated. The developed methodology is split into two parts. The first part assists in reducing facility complexity to enable the selection of a measurement boundary. The second part streamlines the collection, organising and processing of the measured data and supporting documents. The outcome of the methodology was validated by means of a single complex case study. In this case the methodology was applied to identify Section 12L compliant measurement boundaries. The transparent output of the collected data and supporting documentation illustrated the definitive auditability of the results.

This work was sponsored by HVAC International.

¹ The author is enrolled for an PhD degree in the Department of Mechanical and Nuclear Engineering, North-West University.

² The author is a registered professional engineer and holds a PhD in electrical engineering from the North-West University.

³ The author is a registered professional engineer and holds a PhD in computer engineering from the North-West University.



1 INTRODUCTION

1.1 Overview

South African energy intensive industries are under pressure to remain internationally competitive. The implementation of energy efficiency initiatives have the potential to reduce energy consumption while sustaining the same amount of production output. By investing in these initiatives, companies can lower costs and their carbon footprint [1].

Unfortunately, energy efficiency initiatives have been met with a number of barriers such as lack of upfront capital, ignorance regarding energy use and higher production priorities which have hindered energy efficiency investment [2]-[6]. This reluctance to invest is linked to the level of uncertainty involved with energy efficiency investment [7]. This phenomenon is known as the energy efficiency gap [8]. Research has indicated that consumer behaviour and decision-making processes can have an effect on the energy efficiency gap [5], [6]. To alter consumer behaviour and decision-making, government policy has to include financial incentives to promote energy efficiency initiatives and discourage energy inefficient practice [9].

South Africa is considering the implementation of a disincentive in the form of carbon tax [10]. This disincentive can increase energy intensive industry's priority to invest in energy efficiency initiatives. However, it still does not address the financial barriers surrounding energy efficiency investment. Consequently, the South African government will assist companies with a number of financial incentives [11].

Section 12L of the Income Tax Act, (Act No 58 of 1962) has been implemented to reward energy efficiency savings [9]. It allows companies a tax deduction of 95 c/kWh for quantified energy efficiency savings [12]. To receive the benefit, an application, which quantifies the initiative impact, must be submitted to the South African National Energy Development Institute (SANEDI). This application needs to comply with stringent requirements as set out in the Section 12L Act, Section 12L promulgated Regulations and the SANS 50010:2011 Standard. It is therefore mandatory that the application be compiled by an independent South African National Accreditation System (SANAS) accredited Measurement and Verification (M&V) team. Proof of compliancy in the form of supporting documents must be supplied with the application [9], [13].

Since the M&V team is not necessarily familiar with the technical details surrounding a site's initiative implementation, assistance from industry is required [9]. Effective collaboration between industry and the M&V team is therefore important to ensure that the Section 12L application can be effectively compiled [9]. In this paper, a methodology is presented to assist industry to compile the necessary data and documentation. The methodology followed should not taint the independency of the M&V team auditing process.

1.2 Calculating energy efficiency savings

Large-scale industrial energy intensive industries in South Africa mainly consume non-renewable commercially available energy sources generated mainly by fossil-fuels (such as coal, natural gas and petroleum products) [14]. These energy sources can be used to generate secondary energy sources for production related purposes for example compressed air and steam. Both primary and secondary energy sources will be referred to as "energy carriers". The consumption of energy carriers is driven by specific production outputs. These product outputs will henceforth be referred to as "energy drivers"[15].

The quantity of energy carriers consumed per energy driver output is defined as operational energy intensity. An energy efficiency initiative will reduce the consumption of the energy carriers while maintaining the same amount of energy driver output [16]. This indicates that the process has become more energy efficient, which results in an operational energy intensity reduction and energy efficiency savings. In addition, operational energy driver output reduction [9]. Figure 1 below demonstrates this principle:





Figure 1: Determining energy savings (adapted from [9], [17])

The calculation of the energy efficiency savings must be conducted within the guidelines of the Section 12L Act, promulgated Regulations and the SANS 50010:2011 Standard [9].

The Regulations limit the type of energy efficiency savings eligible for the benefit. The following are claimable:

- 1. Energy awareness and conservation The energy efficiency saving achieved by the implementation of energy management systems can be claimed. Energy awareness initiatives like training, switching off non-essential equipment, etc. which promote energy conversation savings are also eligible [9].
- 2. Modify equipment Equipment, structures and/or process are replaced or modified to improve the energy efficiency of the equipment, structures and/or processes [9].
- 3. Combined heat and power Waste heat recovery systems and/or co-generation systems [13].
- There are four energy efficiency initiatives that are not claimable [13]:
 - 1. Captive power plants with less than an 35% energy conversion efficiency
 - 2. Renewable energies
 - 3. Power purchase agreements
 - 4. Projects that have been funded form any sphere of government or any public entity as listed in Schedule 2 or 3 of the Public Finance Management Act, (Act No. 1 of 1999)

Within the M&V SANS 50010:2011 methodology it is first necessary to identify a measurement boundary for the quantification of the energy efficiency initiative [9]. This is to determine which parameters and variables need to be measured for the calculations. However, the mandatory Section 12L requirements make it necessary to first identify measurement points that are compliant before selecting the measurement boundary.

According to the Standard there are four types of boundaries namely [9], [17]:

- 1. Retrofit isolation with key-parameter measurement: The energy efficiency initiative is isolated from the rest of the facility. The energy efficiency savings is calculated by identifying the key performance parameters that will influence the savings calculations.
- 2. Retrofit isolation with all-parameter measurement: same as 1, the savings calculation requires the measuring of all parameters which determines the energy usage of the equipment in the retrofitted isolation.
- 3. Whole facility: The measurement boundary is drawn around the entire facility or sub-facility level. The energy performance of the entire facility is assessed for the savings calculations. This option utilises the facilities existing utility and auxiliaries' meters. Energy invoices billed consumption amounts can be used with this option.
- 4. Calibrated simulation: When important data is missing or not available for extended periods, a calibration simulation approach can be used to predict the energy saving that would have been achieved. The calibrated simulation approach makes use of computer software to develop predictive simulation models.

After selecting the measurement boundary the selected measurement points' data and supporting documentation can be collected. Data accuracy plays a crucial role in the Standard's quantification process. Therefore the Standard requires that the energy carriers and energy drivers' data accuracy be verifiable by calibrated measurement equipment [15]. The calibration methodology used must however conform to national and international standards [15], [17]. This means that the calibration certificates should be supplied as proof



of accuracy. Alternatively, supplier invoices are also an acceptable source of energy data. However, these supplier invoices are only acceptable if they are based on actual measured data [15].

1.3 Data organisation and structuring

Industry will have to make the data and supporting documentation available for the M&V team to conduct their inspection and calculate the energy efficiency initiative's impact. This data can have multiple sources and data formats. Poor data accuracy, quality and integrity can have a significant impact on the energy efficiency savings calculations [9]. Therefore, ensuring data traceability is important.

The data and documentation collected will be in a combination of hard copy and electronic data files. The hard copy files can be scanned to create electronic copies. The electronic files will need to be stored on a centralised server. However, an organisational structure will need to be developed to make the data and documentation easily accessible and traceable. A simple folder hierarchy can be used to organise the data [9].

Whittaker, S *et al.* identified two types of folder hierarchies namely broad and shallow or deep and narrow [18]. A broad and shallow hierarchy will allow the user faster access to the required files, however, the time to find the correct folder increases [9]. The deep and narrow option allows for a faster folder finding time, but the user will need to access more folders to find a file [9]. Whittaker, S *et al.* concludes that the researched group tended to use a wide and shallow approach to folder structuring [18].

All the collected data will need to be processed into the same unit for the energy efficiency savings calculations. A flat-file database table can be used for this purpose. A database is an organisational structure for a collection of information which can easily be accessed, updated and managed [9]. Figure 2 shows a simple database table.

		Field		
Label	Departure date	City	Car make	Name
	01-Jan-15	Bloemfontein	Nissan	Pieter
Record	02-Jan-15	Johannesburg	Toyota	Andries
	03-Jan-15	Pretoria	Fiat	Charl
	05-Feb-15	Durban	Nissan	Pieter

Figure 2: Example of a flat-file database (from [9])

As shown in Figure 2 a database table has three main properties, namely a field, a label and a record. A record will be related to a specific entity's information. The database table record entry will consist out of a number of fields. Each of these fields is related to an independent property of the specific entity information. The database table label identifies the different field's independent properties. A flat-file database is simply a single database table which can have an unlimited number of fields and records [19]. A good example is a Microsoft Excel spreadsheet [9].

2 METHODOLOGY

2.1 Overview

The objective of this section is to develop a methodology to assist industry to reduce system complexity, identify Section 12L compliant measurement points and select a measurement boundary. Thereafter the necessary data and supporting documents can be collected, organised and processed for the M&V team. This methodology will only focus on industries' responsibility to construct a traceable, transparent and compliant folder structure and processed flat-file database that can be used by the M&V team to calculate the energy efficiency initiative impact [9].

2.2 Measurement boundary selection

2.2.1 Overview

Janse van Rensburg, HM et.al. & Janse van Rensburg, HM developed measurement boundary selection methodologies specifically for selecting boundaries for holistic implemented energy efficiency initiatives for the mining industry [9], [15]. These methodologies will be reviewed in the following sections. However the methodologies will be optimised to be suitable for different large-scale industrial industries. Figure 3 illustrates the main steps which will be undertaken.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 3: Key steps to identifying a measurement boundary [from [9]]

2.2.2 Understand

The first step is to understand the operations of the company [9]. A basic production flow can be used. This diagram must indicate all the facilities and/or components involved in production. The facilities can be based in the same premises or different regions. External companies operating on the same facility should also be indicated, especially if they share energy resources (for example electricity). The energy drivers for each stage of production should be indicated after each operational facility. Each production stage will have specific associated energy carriers. These energy carriers can also be collectively indicated on the production flow [9]. Figure 4 illustrates a simple example.



Figure 4: Understanding the operational facility

2.2.3 Identify

The methodology requires the identification of the incoming energy carriers and the outgoing energy drivers measurement points. Energy suppliers can supply multiple Points of Deliveries (PoDs). Therefore, it is necessary for the company to indicate all the facilities and/or components that are included in the supplier bill/invoice. Most of the energy consumption will be measured in weight or volume. If this is the case the specific energy or also known as calorific value (CV) sample point should also be indicated on the layout. These CVs are then used to calculate the energy (MJ) produced by the weight of energy carrier consumed. This CV data will also require proof of accuracy. Limitation projects should also be indicated [9]. Figure 5 illustrates how the invoice boundaries and measurement instruments need to be indicated.





Figure 5: Identification of measurement points and limitations

2.2.4 Simplify

The key to this step is to simplify the complexity of the production flow, while not losing vital information. The Standard requires that the measurement of variables should be obtained from calibrated instrumentation or supplier invoices [9]. Compliance indication can be at three different levels namely;

- Not available There is not sufficient information available for the measurement point
- Available The measurement points indicate data available, but the data compliance is not known
- **Compliant** The data available has sufficient supporting documentation to ensure the data compliance

Figure 6 shows below how compliance can be indicated on the production flow:



Figure 6: Measurement point compliance identification

2.2.5 Select

After the meter compliance is indicated, the user can start to identify possible measurement boundaries. Section 12L tax deductions can only be claimed for energy efficiency savings. It is therefore necessary to determine where energy efficiency initiatives were implemented. There are many types of initiatives that can be implemented on facilities. These interventions can range from major infrastructure projects to energy management programmes [9]. The interventions can then be indicated on the production flow, as in Figure 7. After all the necessary information has been shown on the production flow, the user can decide on the final measurement boundary.





Figure 7: Measurement boundary selection

As shown in Figure 7, for this example there are three possible compliant measurement boundaries. However, each of the boundaries will have its advantages and disadvantages. There should be a balance between costs and time, for example, with option 1 there will be no further calibration costs for available meters. Yet, if a whole facility approach is used; both energy efficiency initiative savings are included which can increase the benefit. This approach will require a larger number of measurement points to be audited which can increase time and costs.

2.3 Data collection and structuring

2.3.1 Overview

When the measurement boundary is selected, each of the measurement points' relevant data and documentation must be collected, structured and processed [9]. Figure 8 illustrates the three steps to be developed to handle the large volumes of data generated.



Figure 8: Data collection and structuring steps [from [9]]

2.3.2 Specify

Within the measurement boundary phase only high-level details regarding the measurement points and boundaries are indicated. A table summary should be constructed to specify these details. Figure 9 provides an illustration of a selected measurement boundary summary. The table should list the measurement point's energy group and necessary supporting documentation required. Energy supplier invoices which were relevant should also be specified. Only the energy driver measurement points chosen to represent the production process need to be summarised [9].





Figure 9: Data and supporting documentation summary

2.3.3 Collect

There may be numerous data sources where raw data and supporting documentation can be collected. The collection process will have to be done in systematically to ensure traceability. The relevant data and documentation is collected with the measurement boundary summary, before continuing to organise the information in a structure [9].

The data should first be tested for compliance. The compliance testing should ensure that all the supporting documentation was collected. However, if the data and documentation fails this test, a new measurement point will have to be identified. This may lead to the selection of a new boundary [9].

It is critical to have a dataset of acceptable quality⁴. After all the data and supporting documentation has been collected, data quality checks can be done. For the data, these checks should include error warnings identification, correct tag names, continuous period intervals, etc. The supporting documentation validity should be ensured by the quality checks. This will include checking the expiration date, the error percentages, meter numbers, account numbers etc.[9].

The SANS 50010:2011 Standard emphasises data accuracy within its requirements. Consequently, if the data is compliant with the Standard, the data accuracy is assured. Data integrity⁵ however, will need to be verified within the context of the methodology. This can be done by comparing two independent data sources which contains the same relevant measurements. An example of this type of verification is comparing standard compliant utility electricity bills (e.g. Eskom) with the company's internal electricity metering system [9].

When data confidence is assured, the data and documentation can be structured. It is important that the structure displays the data and documentation in a transparent and straightforward manner. A folder hierarchy structure will be used to develop the storage location [9]. Figure 10 gives a breakdown of the structuring approach.

⁴ Data quality refers to if the data is "fit for use". This will include ensuring that the data is relevant, complete, error-free and representative [20].

⁵ Data integrity is the prevention of unauthorised alterations to data. These alterations can occur when data is retrieved and stored or when the data is transferred from one format to another [21], [22].





Figure 10: Folder hierarchy for collected data (adapted from [9])

2.3.4 Process

The information collected will not always be in a format which can be used to do calculations. It is therefore necessary to process the data to deliver useful information. For the energy efficiency savings, it is required that all the energy carriers have the same energy equivalent. This unit can either be in kWh or GJ. For energy carriers that are not in the required unit, a transformation formula will be required. An example of this is a multiplication formula with the meter data and the energy carrier's CV value [9]. Figure 11 shows the steps of this process.



Figure 11: Processed flat-file database structure (adapted from [9])

Using a flat-file database, an energy equivalent table can be used to process all the energy carriers into the same unit. The sample data within this table should also be normalised to the same period interval. This can be in days, weeks, months or even years [9].

As seen in Figure 11 the first label should identify the energy category. The next label should identify if the data input was a calculation, or a direct input from the folder structure. For practical reasons the selected energy driver data should also be normalised to the same period interval and entered into the tables. The third level will be the normalised records [9].



2.3.5 Share with the M&V team

The final step of the methodology will be to share all the layouts, data and documentation with the M&V team. This will enable the team to audit the work and calculate the energy efficiency savings for the Section 12L application. There are three items to be shared with the team. These include the measurement boundary selection process, the folder hierarchy and the processed flat-file database. To assist the M&V team to understand the measurement boundary selection process, a report can be written [9].

3 CASE STUDY

The following section applies the methodology developed in a case study in industry. The case study presents operational production flow, measurement points, energy drivers and energy carriers which are based on actual operations [9]. This information was collected from interviews with personnel, operational layouts and the company's metered data. Due to strict confidentiality agreements, the company presented within this section will be referred as "the ferrochrome company".

3.1 Measurement boundary selection

3.1.1 Company operational overview

The ferrochrome company main production focus is the mining and smelting of chromite ore. The company has three main processing plants. The plant selected for this case study has three main energy sources which are electricity, coal and methane rich gas. The furnaces use electricity to generate extreme temperatures (exceeding 1500 °C) for smelting. Coal is mainly added to the chromite ore to not only generate heat but also carbon monoxide gas. This gas reduces the chromite ore to ferrochrome. Methane rich gas is used to dry the incoming raw materials and for other heating applications.

The main energy driver of the facility is the ferrochrome produced. The energy driver is weighed and sold to the market. The production flow consist of three separate smelting operations, namely the furnace section 1, furnace section 2 and furnace section 3. However, furnace section 3 was decommissioned in 2015, since the demand of the product decreased.

3.1.2 Measurement points compliance

Each furnace is supplied by a specific Eskom point of delivery, which are all invoiced individually. This means that the electricity consumption of each furnace type can be accurately determined.

Methane rich gas is supplied by an external supplier. There is no on-site storage facilities of the carrier and gas invoices are available. Although not fully compliant, additional metering is available for specific gas users. This means that these meters can be calibrated for the purpose of the application. The supplier invoices specify the CV (GJ/m^3) of the gas supplied.

The plant has a variety of coal suppliers from different mines. Coal is a natural energy source, therefore each coal mine will have a different coal composition. Due to this source variety, regular samples are taken and analysed to monitor coal quality and characteristics. Coal samples are analysed by a SANAS certified testing laboratory. The analysis results indicate the CV of the various coal types. Coal is weighed with a compliant device before it enters the furnace at the batching section of each furnace.

At each smelting section the ferrochrome production is measured in the weight of alloy produced. The alloy produced by the both the furnace section 1 and furnace section 2 is weighed using a calibrated weighbridge.

The pelletising and sintering plant (PSP) produces chromite ore agglomerations in the form of pellets. This plant however only provides pellets to the furnace section 2. When the PSP was not running, pellets were imported from one of the other plants. Since mid-2015, the plant started producing its own pellets again, and also exported to the other company sites. Pellets imported and exported will only be taken into consideration when assessing furnace section 2, or the whole facility.

Figure 12 demonstrates this in the final stage of the methodology. The company recently installed additional raw material drying equipment and improved furnaces section 1 raw material added ratios. This resulted in significant energy efficiency savings that are claimable. The company also participates in a Public Entity's funding programme, which is a possible concurrent project. The participation is only on furnace C & furnace D.





Figure 12: Case study measurement boundary selection

As is shown Figure 12 there are two measurement boundaries that can be selected. Because of the complexity involved with a whole facility approach, regarding the PSP usage, possible concurrent project and an increased number of variables. The best option will be to have a retrofitted isolation with all parameters measured around furnace section 1.

3.2 Data collection and structuring

The retrofitted isolation is drawn in more detail in Figure 13. Electricity, coal, methane rich gas and ferrochrome production measurements will have to be specified, collected and structured. Each dataset requirements is specified in the table summary, see Figure 13

3.2.1 Compliance verification and data quality

The necessary data and supporting documentation quality checks were done on the information collected. These checks indicated that the all the required data and documents are of sufficient quality. The comparison of Eskom invoice values and internal electricity meter data resulted in an average aggregated deviation of 0.057% over a period of 24 consecutive months.

Certified test weights are used to calibrate the coal batching bins in regular intervals. This assures the data accuracy as required by the Standard. Unfortunately, the data integrity could not be tested due to the lack of extra data sources.

The methane rich gas only 12L compliant data source for measurement is the facility supply invoices. This is the result of the gas meters specifically to the furnace section 1 not being calibrated. However the total gas usage is a minor portion of facilities total energy usage, 1.99%. Although it is a minor portion, it will be important to account for the increase in gas usage due to the installation of additional equipment in the raw material drying facility. It was therefore concluded that the total usage of the supplier invoice can be used for the purpose of this application.

The ferrochrome weighbridges are calibrated by a SANAS certified verification laboratory. The calibration is carried out every two years in accordance with the relevant Standard.





Figure 13: Data collection table summary

3.2.2 Folder hierarchy

The verified datasets and supporting documentations was structured in a folder hierarchy. The structured framework is shown in Figure 14. Level 1 categorises the energy carriers and energy drivers. The preceding levels are a more detailed breakdown of the information collected. These levels should align with the specification for the table summary.



Figure 14: Case study 1 folder hierarchy

3.2.3 Processed data table

To calculate the operational energy intensity reduction a normalised dataset is required. This dataset should be in the same unit of measurement and all data points must be aligned with the period interval. The energy carriers' unit of measurement chosen for this case study is kWh. The period interval will be on a monthly basis. This means that all the energy consumed in a specific month must be normalised to a single monthly data entry. The energy driver should also be normalised to monthly values [9].

The Eskom invoices' dataset requires no processing and can be directly entered into the processed database table. The coal raw datasets are measured in kilograms. The average CV values for a month is multiple by the total coal usage for that same month. This multiplied value will be in a gigajoule (GJ) unit. Therefore to obtain a kWh value the GJ value should be divided by 0.0036. The methane rich gas supplier invoices the ferrochrome company in the energy content of the gas supplied (GJ). Therefore the monthly value can be also divided by 0.0036 to convert the value to kWh. Figure 15 shows the final processed database labels and fields.




Figure 15: Case study 1 flat-file processed database table

CONCLUSION 4

The Section 12L tax incentive rewards companies for quantifiable energy efficiency savings. However, to claim the tax incentive a strict application process and requirements must be followed to ensure the quality, traceability and integrity of the quantified energy efficiency savings. These strict requirements together with system complexity require effective collaboration between the M&V team and industry. It will also be required by industry to assist in collecting, organising and processing the data and supporting documentation for the M&V team [9].

To facilitate this process a clear methodology was developed to reduce system complexity, identify Section 12L compliant measurement points and select an appropriate measurement boundary. Thereafter, the specified measuring point's data and supporting documentation was collected and organised in a folder hierarchy to ensure transparency. To further assist the M&V team, the energy carriers and energy driver's data was processed to the same measurement period and unit [9].

The outcome of the methodology was validated by means of a case study. In this case study the methodology was applied to identify Section 12L compliant measurement boundaries. The transparent and traceable output of the collected data and supporting documentation illustrated the ultimate auditability of results. The practical application and validation of the methodology confirmed that the original problem statement was sufficiently addressed [9].

REFERENCES

- KPMG International. 2013. "The KPMG Green Tax Index 2013," Amstelveen. [1]
- [2] Cagno, E. Trianni, A. Abeelen, C. Worrell, E. and Miggiano, F. 2015. Barriers and drivers for energy efficiency: Different perspectives from an exploratory study in the Netherlands, Energy Conversion and Management, 102, pp. 26-38.
- Shi, H. Peng, S.Z. Liu, Y. and Zhong, P. 2008. Barriers to the implementation of cleaner production in [3] Chinese SMEs: government, industry and expert stakeholders' perspectives, Journal of Cleaner Production, 16(7), pp. 842-852.
- [4] Painuly, J.P. Park, H. Lee, M.K. and Noh, J. 2003. Promoting energy efficiency financing and ESCOs in developing countries: Mechanisms and barriers, Journal of Cleaner Production, 11(6), pp. 659-665.
- Sardianou, E. 2008. Barriers to industrial energy efficiency investments in Greece, Journal of Cleaner [5] Production, 16(13), pp. 1416-1423. Zilahy, G. 2004. Organisational factors determining the implementation of cleaner production
- [6] measures in the corporate sector, Journal of Cleaner Production, vol. 12, no. 4, pp. 311-319.
- Klemick, H. and Wolverton, A. 2013. Energy-Efficiency Gap, Encyclopedia of Energy, Natural [7] Resource, and Environmental Economics, 1, pp. 74-81.
- [8] Gillingham, K. and Palmer, K. 2013. Bridging the energy efficiency gap: Insights for policy from economic theory and empirical analysis, *Resources for the future*, January, pp. 1-31.
- Janse van Rensburg, H.M. 2016. Structuring mining data for RSA Section 12L EE tax incentives, Masters [9] dissertation, North-West University.
- Republic of South Africa. 2013. Carbon Tax Policy Paper, Government Printer, Pretoria. [10]
- Republic of South Africa. 2011. National climate change response white paper, Goverment Printer, [11] Pretoria.
- [12] Republic of South Africa. 2015. Taxation Laws Amendment Bill No 29 of 2015, Government Printer, Pretoria.
- Republic of South Africa. 2013. Regulations in terms of section 12L of the Income Tax Act, 1962, on [13] the allowance for energy efficiency savings, Government Gazette, 186(38541), pp. 1-4.
- South African Goverment Department of Energy. 2012. Energy Statistics, The Energy Balances -[14]



Consolidated Aggregated Historical Energy Balances per commodity, [Online]. Available: http://www.energy.gov.za/files/energyStats_frame.html. [Accessed: 02-Sep-2015].

- [15] Janse van Rensburg, H.M. Maneschijn, R. and Booysen, W. 2015. Selecting mearsurement boundaries for Section 12L energy efficiency tax applications: A mining case study, in Proceedings of the 11th conference on the Industrial and Commercial Use of Energy.
- Boyd, G.A. 2007. Estimating the distribution of plant level manufacturing energy efficiency with [16] stochastic frontier regression, Durham, United States.
- [17]
- "Measurement and verification of energy savings," SANS 50010, no. 1, pp. 1-22, 2011. Bergman, O. Whittaker, S. Sanderson, M. Nachmias, R. and Ramamoorthy, A. 2010. The effect of [18] folder structure on personal file navigation, Journal of the American Society for Information Science and Technology, 61(12), pp. 2426-2441.
- Doyle, S. 2000. Understanding Information Technology, 1^{ste} Edition, Nelson Thornes. [19]
- Wang, R.W. and Strong, D.M. 1996. Beyond accuracy: What data quality means to data consumers, [20] Journal of Management Information Systems, 12(4), p. 5.
- Zviran, M. and Glezer, C. 2000. Towards generating a data integrity standard, Data and Knowledge [21] Engineering, 32(3), pp. 291-313.
- [22] Ghaeb, J.A. Smadi, M.A. and Chebil, J. 2011. A high performance data integrity assurance based on the determinant technique, Future Generation Computer Systems, 27(5), pp. 614-619.



DESIGN OF A CONDITION-BASED ANALYTICAL DECISION SUPPORT TOOL

L. Nyanga^{1*}, A. Van der Merwe², N, Masanganise³, S Matope⁴ & M. T. Dewa⁵

^{1, 2, 4 & 5} Department of Industrial Engineering University of Stellenbosch, South Africa <u>nyangalu@gmail.com</u>, <u>andrevdm@sun.ac.za</u>, <u>smatope@sun.ac.za</u>, <u>mnce2009@gmail.com</u>

> ³Department of Industrial and Manufacturing Engineering National University of Science and Technology, Zimbabwe nmasanganise@yahoo.com

ABSTRACT

As manufacturing companies thrive to achieve global competitiveness, effective plant monitoring and short decision making time has proven to be a big contribution for them to achieve their goal. To cut operational costs and offer a competitive edge, the companies need not only to capture equipment failure but to predict the failure occurrence. Relevant decisions also have to be made in the shortest possible time in the case of a failure. The paper discusses an analytical decision support tool for monitoring a generator thrust bearing of a hydro power station. The system consists of a distributed Multi-Agent System (MAS) comprising of diagnosis, decision alarm, database and distribution agents. The diagnosis agent is responsible for obtaining the parameter readings from the sensors using a Siemens S7-1200 PLC and Kepware OPC server. The decision alarm agent uses a Java Expert System Shell (JESS) rule base to monitor the parameters obtained from the diagnosis agent and make decisions on equipment availability using the Maintenance Free Operating Period (MFOP) concept. The desired action and equipment performance trends are then conveyed to the user by the distribution agent. The decision support tool is developed using the Java Agent Development framework (JADE). The proposed system showed an increase in machine availability and reduction in the maintenance scheduling time and parts inventory.

¹ The author was enrolled for a PhD (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch

 $^{^{2}}$ The author is a researcher in the Department of Industrial Engineering, University of Stellenbosch

³ The author was enrolled for a B Eng (Hons) (Industrial) degree in the Department of Industrial and Manufacturing, National University of Science and Technology

⁴ The author is a researcher in the Department of Industrial Engineering, University of Stellenbosch

⁵ The author was enrolled for a PhD (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch



1. INTRODUCTION

The information age has come with a change in focus of the manufacturing and service from being only product oriented to being customer oriented. The customers now have more variety of suppliers to choose from as the internet has made them to be no longer limited to the suppliers in their geographical area. The customer now demands more customized goods and more information on the state of their order. To meet the demands of the customer the decision makers in companies should be able to get more information from the state of their plant. The power sector industry in Zimbabwe still employs traditional methods of condition monitoring whereby the system is dependent on a centralized master computer for monitoring. However, the environment is now dynamic and technology can substitute the existing master-slave monitoring system with a more flexible intelligent system. This paper focuses on the development of a distributed approach to condition monitoring using a multi-agent system as a way of increasing plant flexibility. The structure of this paper is as follows, the AUML methodology for agent development is discussed, followed by the establishment of a knowledge base. We then move on to the design of a multi-agent system using the Java Development (JADE) toolkit and lastly the integration of the MAS with the control environment is shown as the result.

2. PROBLEM STATEMENT

During the year of 2013, a hydro-power station in Zimbabwe experienced several planned and unplanned outages. This was due to factors ranging from off-line maintenance, internal machine faults such as high bearing temperature, to external faults such as over-frequency. The number of failures for each month are shown on a graph in Figure 1.



Figure 1: Planned and unplanned outages for the plant

With a proper decision support tool in place, fault predictions that are likely to occur on the machine based on its current condition can be computed in advance. This has the advantage of reducing scenarios where a forced outage due to high bearing temperature occurs. Traditional methods of decision support still exist for condition based maintenance and usually these influence how long maintenance will take. Mangina [1] carries out a case study for an agent monitoring the partial discharge of a Gas Insulated Substation. The parameters were measured online whilst analysis of the captured data was carried out using intelligent agents which base decisions on Artificial Neural Networks. The software tool mainly specialized in recognizing partial discharge sources and offered decision support based on Case Based Reasoning. This differs from the tool designed in this paper in that the designed tool uses a Rule-Based algorithm as an artificial way of manipulating data. The algorithm has two parts, the left hand side comprising of the condition and the right hand side comprising of the action Negnevitsky [2]. Furthermore, efficiency is increased by using JESS to develop the rule base and JESS implements the Rete Algorithm were only new facts are compared against any rule Friedman-Hill [3]. Another agent-based system that was designed for monitoring and control was discussed by Deshmukh et al [4]. The system was implemented on power systems with limited power availability so that the agents could determine the most suitable power from the source. The system, however differs from the tool discussed in this paper in that it comprises of a simulation agent which executes dual communication between the measurement and simulation where the data is used to customize the simulation hence the results from the simulation are used for decision support. The tool discussed in this paper however, has distributed multi-agents able to offer decision support based on the real time data from the sensor environment without simulating. Moubray [5], points out that it is important that condition based maintenance is applied where it is appropriate, not as an overall policy. This is because condition based maintenance techniques are expensive, hence it would not be cost effective as an overall policy.



3. **METHODOLOGY**

Dewa et al [6] mention in their survey that there are several methodologies that are available for building MAS and as a result methodology selection is a challenge for MAS developers. Hence a methodology used for MAS development should not only be suitable for the intended application, but should also show a distinction between the analysis and design stages. Out of the many agents-oriented design methodologies there exist one methodology which is suitable for application in a power generation setup. The Agent Unified Modeling Language (AUML) methodology approach was used in developing the agents of the system. An agent is defined by Russel & Norvig [7], as an entity that can be viewed as perceiving its environment through sensors and acting upon its environment through effectors. Rudowsky [8] points out that the most critical decision of an agent is to determine the most suitable action to perform in satisfaction to its design objectives. Agent UML is an extension of the Unified Modeling Language (UML) and it attempts to bring together research on agent-based software methodologies and emerging standards for object-oriented software development Bauer & Muller [9]. The methodology comprises of two phases, namely the analysis phase and the design phase.

3.1 THE ANALYSIS PHASE

The analysis phase as mentioned by Nikraz et al [10] describes the problem without necessarily developing the solution to the problem. The analysis phase comprises of the following seven steps;

- **Requirements Analysis**
- **Requirements Modeling**
- Initial Agent Types Identification
- Roles and Organisation Identification
- Acquaintances Identification
- Agent Refinement
- Agent Deployment Information

3.1.1 Step 1: Requirements Analysis

The requirements were captured based on the business requirements, maintenance feedback and plant performance from the mechanical, operations and electrical maintenance departments. Holt [11], pointed out that the requirements should be classified in a way that enables assessment of their status at any point in the project. Hence he suggests that requirements be classified in terms of defined features or attributes, generally an attribute represents the properties of a requirement. After sufficient requirements engineering, a summary of the attributes is compared against the requirements and the comparison is shown in Table 1.

Table 1: Requirements Analysis						
Attributes						
Requirement	Source	Priority	V/V	Ownership		
Increase plant availability	KSPS plant objective	Essential	Internal/External Plant Trips Statistics	KSPS		
Increase efficiency	KSPS plant objective	Essential	Key performance indicators	KSPS		
Monitor System	Maintenance report	Essential	Operational Trends Decision Support Alarm Manager			
Get readings	Maintenance report	Essential	Listing parameters to be monitored Sensor Readings Provider			
Decision and alarm status	Maintenance report	Essential	Established rule base Decisi Suppo Alarm Manar			
Send notification	Maintenance report	Essential	Email or SMS to the user User Notificati Provider			
Maintain ISO 9001:2008	Quality report	Desirable	Quality Audits	Performance and Risk Officer		
0% accident rate	Operations report	Desirable	Safety Inspections	Performance and Risk Officer		
Store readings	Maintenance report	Desirable	Well established database	Readings Storage Provider		

Table 4. Demoissante Analysis

From the analysis, focus is shifted onto the user requirements and the constraints on them rather than with the high-level business requirements.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

3.1.2 Step 2: Requirements Modeling

Siau & Lee [12] mention in their research that subjects perform better in model understanding when both the Use Case diagram and the Class diagrams are used together. Therefore, for the analysis phase a Use Case diagram was used whilst in the design phase discussed later, a Class diagram is used. Figure 2 shows the requirements encircled whilst the providers are represented by the actor symbol and the human user is shown on the external of the system boundary. Interaction between the use cases and providers is depicted by means of the association symbol.



Figure 2: Use Case diagram for the system

The main requirement as shown is to monitor the system which includes sub-requirements fulfilled by the relevant providers.

3.1.3 Step 3: Initial Agent Types Identification

The third step dealt with identifying the main agent types and representing them on an agent diagram. Nikraz et al [10] suggests the following rules when identifying agents;

- 1. Add one type of agent per user/device.
- 2. Add one type of agent per resource.

3.1.4 Step 4: Roles and Organisation Identification

A role was identified for each agent type basing on the requirements noted on the Use Case diagram in Step 2. The artefact resulting from this process is the role table. For the condition monitoring system, the following rules were applied;

- 1. Deriving the initial set of roles from the Use Case diagram.
- 2. Considering the agents were these roles are clearer first and delay the identification of roles for other agents to later steps.

With the roles identified, they were then grouped together into an organisation. An organisation is a collection of roles that stand in certain relationships to one another, and that take part in systematic, institutionalised patterns of interactions with other roles.

3.1.5 Step 5: Acquaintances Identification

Interactions between agent types were identified in this step, hence adding the agent types to the role table. With the condition monitoring system the Diagnosis Agent interacts with the Decision Alarm Agent which in turn interacts with the Database Agent and the Distribution Agent. The Distribution Agent interacts with the Database Agent and the Human User. The role table is then reviewed to cater for the added agent types and is shown in Table 2.



Agent Type	Roles	Organization
Diagnosis Agent	 Monitoring variations of hydro bearing parameters Ensure operation of sensors Estimating values in case of sensor failure 	• Input
Decision Alarm Agent	Triggering alarmsOffering decision support	Administration
Distribution Agent	Pushes alarms raised to the userPushes decision support to the user	• Dispatch
Database Agent	 Communicating with the external database for storage of measurement readings 	Administration

Table 2: Reviewed role table

3.1.6 Step 6: Agent Refinement

Modification of the agent types identified in Step 3 was done using the following considerations;

- 1. **Support:** what supporting information agents needs to accomplish their roles, and how, when and where is this information generated/stored.
- 2. Discovery: how agents linked by an acquaintance relation discover each other.
- 3. **Management and Monitoring:** is the system required to keep track of existing agents, or the starting and stopping of agents on demand.

The given considerations were applied to the condition monitoring system and it was noted that the agent types in the main domain required, the services of the Yellow Pages Agent in helping to discover the Diagnosis Agent. As for the management and monitoring, no new agent types were required to manage or monitor the condition monitoring system.

3.1.7 Step 7: Agent Deployment Information

Agents in their respective domains were represented on an agent deployment diagram. Poggi et al [13] mentions that the deployment diagram is useful in systems that require visualization of the current topology and distribution of components and agents. The methodology adopts an AUML approach by representing the system artefact as nodes which are connected through communication paths to create network systems of arbitrary complexity.

3.1.8 Overall System Architecture after Analysis Phase

The condition monitoring system after the analysis phase comprised of the agent types, users, organisations, the environment, external resources and also it showed the interactions between the mentioned artifacts. Figure 3 shows a clear diagrammatic view of the system architecture for the MAS.



Figure 3: Architecture of the multi-agent system



The parameters that were monitored by the condition monitoring system included;

- Generator Thrust Bearing Oil Film Thickness
- Generator Thrust Bearing Temperature
- Generator Thrust Bearing Vibration
- Cooling Water Temperature
- Guide Vane Opening
- Generated Power

3.2 THE DESIGN PHASE

Nikraz et al [10] point out that the design phase is concerned with specifying a solution to be implemented by JADE. The approach used in the design phase was based on the AUML language which represents agents as they are, not as objects, while employing state chart diagrams to model their behavior and extended interaction diagrams to model their communicative acts. Basically the steps involved in the design phase are;

- 1. Static View Representation of Agents
- 2. Internal Agent States
- 3. Agent Interaction Specifications
- 4. Control Flow Overview of Agents

3.2.1 Step 1: Static View Representation of Agents

An Agent UML class diagram is used to describe the static views of the agents. An agent class according to Huget [14] is denoted by the stereotype <agent> and a unique agent name. The agent classes are then combined together with the behavior class diagrams and other class diagrams in relation with the agents. This gives an overall class diagram shown on Figure 4.



Figure 4: Overall agent class diagram

The various relationships are clearly shown and these include associations, dependencies, and aggregation and composition relationships. It can also be seen that there is an existing relationship between the Diagnosis Agent and the environment. Malaal & Addou [15] emphasize the importance of the environment in a class diagram as it influences the way information is perceived by the agents and also the appropriate action in response to the information supplied.



3.2.2 Step 2: Internal Agent States

An agent has different states that it occupies during its execution lifetime. Hence these states have to be identified so as to see how an agent reacts to events, together with its responses and actions. A State Diagram was used to represent the internal state of an agent and it also provides a graphical way of representing discrete behavior through finite state transition systems.

3.2.3 Step 3: Agent Interaction Specifications

Gomez-Sanz & Pavon [16] note that the UML sequence diagram is extended so as to specify FIPA interaction protocols. The agent interaction diagram used observes the FIPA interaction protocols as shown on Figure 5.



Figure 5: Agent interaction diagram

There are two parts that can be considered in sequence diagrams: a frame, which delimits the sequence diagram and the message flow between roles through a set of lifelines and messages as suggested by FIPA [17]. According to the diagram, the user can request for the current readings at any time and these requests are processed the Distribution Agent. In this kind of flow the Distribution Agent will carry out a FIPA Query Protocol for the measurement readings that can be obtained by the Database Agent. The Database Agent will in turn query the external database so as to obtain the current trends or measurement readings. The Database Agent will inform the Distribution Agent which in turn informs the user of the current measurement. The Distribution Agent subscribes to the Decision Alarm Agent using the FIPA Subscribe Interaction Protocol so that it is notified whenever an alarm or decision support advice is available. The Decision Alarm Agent executes a FIPA Request Protocol to the Diagnosis Agent to get the measurement values and execute the rule base required to inform the Distribution Agent of any alarms or decision support. The Diagnosis Agent will reply with an "INFORM-RESULT" which includes the sensor readings in its message.

3.2.4 Step 4: Control Flow Overview of Agents

The control flow overview of the agents is shown by means of an Activity Diagram that shows the internal activities executed by each agent in its lifecycle. These activities prove that an agent is autonomous and show that data is decentralized Sycara [18].

4. SYSTEM DEVELOPMENT

System development covered areas such as the JADE system development, development of the sensor environment and the JESS rule-base development. Singh et al [19] carried out a comparison between various agent developments toolkits and it can be noted that JADE used in this paper was more advantageous as compared to other toolkits. Nicolae et al [20] also did a comparison on expert system development toolkits which favoured Jess.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

4.1 Sensor Environment Development

The development of the sensor environment was carried out using the Siemens S7 as a controller and the KEPServerEX 5.16 as the server. The S7-1200 models and the Windows-based programming tool gives you the flexibility you need to solve your automation problems as pointed out by Siemens [21]. The main aim was to create an environment where the agents can access and monitor the station parameters by obtaining values from the sensors in the environment. Five analog inputs and 10 digital outputs were used to create a tag table. The server and the PLC were placed on the same subnet and an S7 connection was established between the server and the PLC. This connection will allow any client to manipulate data from or to the sensors without any challenges being faced. The client in this project is JEasyOpc which consists of Java code that is able to connect and register to the OPC Group hence making it possible to read/write tag values. Diaconescu & Spirleanu [22] strongly believe that JEasyOpc can be used in control structures comprising of the JADE multi-agent platform and the software to connect to a PLC.

4.2 JADE Multi-Agent System Development

Wooldridge & Jennings [23] review the first agent programming languages such as TELESCRIPT and PLACA that were used to manipulate agents. However, technology has advanced greatly and there exist user-friendly software programs like JADE. JADE is a Java written agent-oriented middleware system that supports add-on modules Bellifemine et al [24].

Four agents were developed and these are;

- 1. Diagnosis Agent
- 2. Decision Alarm Agent
- 3. Database Agent
- 4. Distribution Agent

Each agent comprises of a behaviour that determines how it executes its functions. A behaviour represents a task that an agent can carry out and is implemented as an object of a class that extends the jade.core.behaviours.Behaviour class. Caire [25] pointed out that an agent can execute several behaviours concurrently. It is important to notice that the scheduling of behaviours in an agent is not pre-emptive but cooperative.

4.3 JESS Rule Base Development

Using Jess, you can build Java software that has the capacity to "reason" using the knowledge you supply in the form of declarative rules. Jess is small, light, and one of the fastest rule engines available. Jess was originally conceived as a tool for building expert systems. Cardoso [26] elaborates that in the multi-agent systems world, Jess can be used as a decision component of an agent, which is implemented in a declarative way. The rule base for this tool was dependent upon two factors namely;

1. The defined allowed maximum and minimum operating parameters of the station units. 2. The MFOP period.

The MFOP period helps in predicting availability of the power station unit, Nyanga et al [27] developed a resource agent that was capable of predicting machinery availability however in a manufacturing setup. Shaalane & Vlok [28] also voiced the improvement and enhancement of equipment if the MFOP concept is applied. The rule base is integrated together with JADE, so that the message content used in the ACL message format of JADE is manipulated by JESS. The JESS reply meant for the agent will be sent via the same path after asserting the facts and comparing the left-hand-side to the right-hand-side of the facts and rules.

5. RESULTS

The results of the system were mainly those from the Multi-Agent System. Analog inputs were used for the sensors so as to obtain a respective engineering value after normalizing and scaling the analog value.

5.1 Results from the Multi-Agent System

Once a connection had been established with the server, the agents obtained the tag values via the JEasyOpc toolkit. Hence the multi-agent system consisted of autonomous agents communicating via ACL messages.



E EssyOpc 25	tributi 💶 🗖 🗙 n for Stoppage			• × × • •	
Masters Presentation [Java Application] C: Program Files Day Connectin INFO: Kar Generator Tronde		(0:20:37 AM)	Message	h Temperature	<u>^</u>
INFO: Gen earing Tempe: ==> f-10 (MAIN::ACLMessage (communio ==> Activation: MAIN::high-temp : f FIRE 1 MAIN::high-temp f-10	Enter	: DatabaseAgen		ОК	lsionAlarmAgent@Con
<pre>ins i main: main: main: active communicative act CONFIRM) (sender DecisionAlarmAgent@ConditionMonitor) (receiver DatabaseAgent => Activation: MAIN::send-a-message : f-1, f-11 <= f-10 (MAIN::ACLMessage (communicative_act CPP) (sender DatabaseAgent@ConditionMonitor) (receiver DecisionAlarmAgent@ConditionMonitor))</pre>					
<pre>Ress has executed 1 passes INFO: Decision Alarm Agent notified Connecting to Kariba South Database DecisionAlarmAgent@ConditionMonitor is checking if there is a message FIRE 1 MAIN::send-a-message f-1, f-11</pre>					

Figure 6: Diagnosis Agent retrieving an item value

Figure 6 shows the Diagnosis Agent is displaying the generator bearing temperature reading of the sensor and also an interface used to enter the reason for stoppages when the unit is down.

5.1.1 Communication between the Condition Monitoring Multi-Agents

The sniffer agent in JADE has sniffing features that allow us to track every message directed to or from an agent in a group of agents. The communicative acts are shown in Figure 7 for each agent and this is a sign of communication between the several agents.



Figure 7: Sniffer agent showing agent communication

It can be seen from the diagram that the agents adhered to the FIPA Interaction Protocol diagram that was discussed earlier. Basically it discussed about how agents communicate using the standard communicative acts.

5.1.2 User Interface

A user interface based on the Distribution Agent was created so as to communicate with the user. The user interface shows the current status of the plant as well as any other condition-based advice that the user might require. The user is able to select the option to view the unit trends of the selected unit. Figure 8 shows the designed user interface.



2	KSPS Plant			×
	WELCOME TO THE ANALYTICAL DECISION		OOL	
	GENERATED POWER :		122	MW
				,
	BEARING TEMPERATURE :		77	°C
				1
	COOLING WATER TEMPERATURE :		38	°C
	BEARING VIBRATION :		1.02	
			1.02	mm/s
	MFOP DAYS LEFT		15	
	UNIT STATUS :	Warning!! Hig	h Bearing Tempera	ature on Unit 1
	TREND ANALYSIS	;		

Figure 8: User interface

The space for unit status shown on the diagram indicates any alarms that the user has to be notified of.

5.1.3 Parameter Trends

The tool was able to trend the system parameters such as generator bearing temperature upon request from the user. Figure 9 clearly shows a snapshot of the online trends produced by the Distribution Agent.



Figure 9: Temperature trends

After requesting for system trends, the tool is also capable of storing an image file of the system trends, which can be used for offline analysis.

5.1.4 Database Storage and Retrieval of Parameter Readings

An added feature of the tool is that it has a database where it can store and query data upon request. Two databases were created using MySQL, the first database was for storing parameter readings whilst the second database was for storing MFOP data that were to be used for MFOP calculations and analysis. Figure 10 shows the database created for MFOP decisions.

mysql	nysql> select * from mfop1;					
id	date	time	stoppage_reason	state	l unit	ł
1 2 3	2015-04-2 2015-04-2 2015-04-2 2015-04-2	4 10:05 4 10:16 4 10:21	:10 High Temperature :41 High Temperature :43 High Pressure	Failure Failure Suspension	3 2 2	+



Figure 10: Database created for MFOP decision making

The database is accessible both online and offline depending on what the user wants. The database for MFOP logging was created in such a way that the user enters the data manually on an available GUI.

6. FUTURE WORK

The future work of this tool is to integrate the self-predicting maintenance free operating period function into the multi-agent system so as to increase the efficiency of the tool. This will imply creating an agent that is responsible for calculating the MFOP predictions automatically as soon as plant fault data is entered into the database. Agent communication is greatly improved as agents will now be in the same agent platform, meaning messages are sent and responded to in the shortest possible time. The applicability of the tool can also be implemented in areas such as the tertiary institutions, manufacturing industries and health sector. So much time is taken in analyzing data for students, manufactured products and patients yet there are agents who can manipulate large amounts of data in the shortest possible time.

7. CONCLUSION

The condition-based analytical decision support tool was designed so as to have a positive impact on the decision making process of the company under study. The positive impact mentioned here is that the tool has the ability to save unnecessary breakdown costs for the company. Engineers at the company under study supported the design and assured that the implementation of the tool in a real industrial setup will help cut the time taken in decision making. Therefore the tool can be regarded as an emerging technology that supports condition-based maintenance in that it reduces maintenance costs as it tends to detect failure before it occurs.

Several benefits that are realized upon implementation and some of these benefits include improved generating capacity. The tool comprises of user-friendly functions that can help the shift charge engineer to monitor the plant with relative ease and understanding since it eliminates the manual compilation of large quantities of data. Hydro-power plant equipment is delicate and requires continuous monitoring so as to reduce catastrophic failure and this paper introduced a tool capable of monitoring plant equipment.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

8. REFERENCES

- [1] Mangina, E., 2005. Intelligent Agent-Based Monitoring Platform for Applications in
- Engineering. International Journal of Computer Science & Applications, 2(1), pp. 38-48.
- [2] Negnevitsky, M., 2002. Artificial Intelligence: A Guide to Intelligent Systems. Harlow: Pearson Education Limited.
- [3] Friedman-Hill, E. J., 2008. Jess: The Rule Engine for the Java Platform, New Mexico: Sandia National Laboratories.
- [4] Deshmukh, A. et al., 2005. Multi-Agent system for diagnostics, monitoring and control of electric systems. Arlington, VA, Proceedings of the 13th International Conference on Intelligent Systems, pp. 201-206.
- [5] Moubray, J., 1997. Reliability-Centred Maintenance. 2nd red. New York, USA: Industrial Press Inc.
- [6] Dewa, M. T. et al., 2014. An Analysis of Methodologies for Agent-Based Modeling for E-Manufacturing Systems in South Africa. International Journal of Advances in Mechanical and Automobile Engineering (IJAMAE), 1(1), pp. 1-6.
- [7] Russel, S. & Norvig, P., 1995. Artificial Intelligence: A Modern Approach. Eaglewood Cliffs, USA: Prentice-Hall.
- [8] Rudowsky, I. S., 2004. Intelligent Agents. *Communications of the Association for Information Systems*, 14(1), pp. 275-290.
- [9] Bauer, B., Muller, J. P. & Odell, J., 2001. Agent UML: A Formalism for Specifying Multiagent Interaction. In: P. Ciancarini & M. Wooldridge, reds. Agent-Oriented Software Engineering. Berlin: Springer-Verlag, pp. 91-103.
- [10] Nikraz, M., Caire, G. & Bahri, P. A., 2006. A Methodology for the Analysis and Design of Multi-Agent Systems using JADE. International Journal of Computer Systems Science & Engineering, 21(2), pp. 99-116.
- [11] Holt, J., 2007. UML for Systems Engineering: watching the wheels. 2nd red. London: The Institution of Engineering and Technology.
- [12] Siau, K. & Lee, L., 2001. Role of Use Case Diagram in Requirement Analysis. Boston, Americas Conference on Information Systems (AMCIS).
- [13] Poggi, A. et al., 2004. Modeling Deployment and Mobility Issues in Multiagent Systems Using AUML. In:
 P. Giorgini, J. P. Muller & J. Odell, reds. Agent-Oriented Software Engineering 2003. Melbourne: Springer Berlin Heidelberg, pp. 69-84.
- [14] Huget, P. M., 2002. Agent UML Class Diagrams Revisited. Liverpool, Proceedings of UKMAS Workshop.
- [15] Malaal, S. & Addou, M., 2011. A new approach of designing Multi-Agent Systems. International Journal of Advanced Computer Science and Applications (IJACSA), 2(11), pp. 1-10.
- [16] Gomez-Sanz, J. & Pavon, J., 2004. Methodologies for Developing Multi-Agent Systems. Journal of Universal Computer Science, 10(4), pp. 359-374.
- [17] FIPA, 2003. FIPA Modeling: Interaction Diagrams, Geneva: FIPA.
- [18] Sycara, K. P., 1998. Multiagent Systems. Al Magazine, 19(2), pp. 1-14.
- [19] Singh, A., Juneja, D. & Sharma, A. K., 2011. Agent Development Toolkits. International Journal of Advancements in Technology, 2(1), pp. 158-164.
- [20] Nicolae, O., Giurca, A. & Wagner, G., 2007. *On Interchange between JBossRules and Jess*, Craiova, Romania: Brandenburg Technical University at Cottbus.
- [21] Siemens, AG, 2012. S7-1200 Programmable Controller: System Manual. Nurnberg, Germany: Siemens AG.
- [22] Diaconescu, E. & Spirleanu, C., 2012. Communication Solution for Industrial Control. Craiova, International Conference on Applied and Theoretical Electricity (ICATE) Proceedings, pp. 1-6.
- [23] Wooldridge, M. & Jennings, N. R., 1995. Intelligent Agents: Theory and Practice. The Knowledge Engineering Review, 10(2), pp. 115-152.
- [24] Bellifemine, F., Caire, G. & Greenwood, D., 2007. *Developing Multi-Agent Systems with JADE*. 1st red. Chichester: John Wiley & Sons Ltd.
- [25] Caire, G., 2003. JADE Tutorial: JADE Programming for Beginners. 2nd red. Boston, USA: Free Software Foundation.
- [26] Cardoso, H. L., 2007. Integrating JADE and JESS, Porto, Portugal: University of Porto.
- [27] Nyanga, L., van der Merwe, A. F., Mutingi, M. & Matope, S., 2013. Development of a Resource Agent for an E-Manufacturing System. Stellenbosch, 25th Annual South Africa Institute for Industrial Engineers.
- [28] Shaalane, A. & Vlok, P. J., 2013. Application of the Aviation Derived Maintenace Free Operating Period Concept in the South African Mining Industry. South African Journal of Industrial Engineering, 24(3), pp. 150-165.



DEVELOPMENT OF A MULTI AGENT SYSTEM FOR PART TYPE ALLOCATION TO MACHINES

L. Nyanga^{1*}, A. Van der Merwe², T. Musiyarira³, S Matope⁴& M. T. Dewa⁵

^{1, 2, 4 & 5} Department of Industrial Engineering University of Stellenbosch, South Africa <u>nyangalu@gmail.com</u>, <u>andrevdm@sun.ac.za</u>, <u>smatope@sun.ac.za</u>, <u>mnce2009@gmail.com</u>

> ³Department of Industrial and Manufacturing National University of Science and Technology, Zimbabwe tpmusiyarira@gmail.com

ABSTRACT

Multi Agent Based system (MAS) is a technology that has become more popular in E-manufacturing and is being used in a wide range of control applications including scheduling and planning. It enables a manufacturing company to be highly flexible, agile, and versatile. In the paper a MAS is used to allocate parts to machines in a company with high excessive stock in which work in progress is accumulating in between work stations with an aim of improving scheduling of jobs, utilisation of machines and production throughput. The MAS was developed using Multi Agent System Engineering (MASE) methodology with Analytical Hierarchy Process (AHP) used to develop the decision making model for best machine selection. Each machine was assigned priority value according to time, cost and quality as the decision criteria for determining the best machine to perform the intended operation. Java Agent Development Environment (JADE) was used for the deployment of agents. Simulation of the existing production set up shows that the proposed MAS model will reduce the lead time of developing work order by 83%, machine utilisation is increased from 73% to 84% and prioritization of machines is increased from 0.193 using a human expert to 0.197 using the MAS.

¹ The author was enrolled for a PhD (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch

 $^{^{2}}$ The author is a researcher in the Department of Industrial Engineering, University of Stellenbosch

³ The author was enrolled for a M Eng (Manufacturing Systems & Operations Management degree in the Department of Industrial and Manufacturing, National University of Science and Technology

⁴The author is a researcher in the Department of Industrial Engineering, University of Stellenbosch

⁵The author was enrolled for a PhD (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch



1. INTRODUCTION

Today's manufacturing businesses are facing immense pressures to react rapidly and robustly to dynamic fluctuations in demand distributions across products and changing product mix. Efficient and practical methods for scheduling and optimization technology are the key to improve the productivity and efficiency of a manufacturing plant as stated by Ayton and Lawley[1].According to Shen[2] the manufacturers' success is no longer measured by their ability to cost-effectively produce a single product butits now measured in terms of flexibility, agility, and versatility.

Local manufacturing companies are aiming at investing in new technologies to produce high quality products more economical than competitors so as to increase market share. Multi Agent Based system (MAS) is the new technology that has become more popular in E-manufacturing and is being used in a wide range of control applications including scheduling and planning as indicated by Colombo[3] andZhang[4].A MAS automates decision making normally done by human beings hence enabling a total integration of thefactory to the supply chain. An E-manufacturing framework for the South African industry has been proposed in Nyanga and Van der Merwe[5].The main aim of the E-manufacturing framework is to increase machine utilisation and factory productivity by sharing resources using an E-manufacturing platform. In this paper, MAS for allocating parts to machines in different sections of a single factory is proposed. A transformer manufacturing plant is used as a case study. The paper is structured as the following: section 2 reviews related literature, sections 3contains the Multi Agent System Engineering (MASE) methodology that was used in modelling thesystem and section 4 is the deployment phase. The paper finally ends with the conclusion.

2. RELATED LITERATURE

Russel and Norvig [6] define an agent as an entity that can be viewed as perceiving its environment through sensors and acting upon its environment through effectors. Wooldridge and Jennings [7] specifies that software agents are programs that engage in software based computer system displaying the properties of autonomy, social adeptness, reactivity and pro-activity. Agent's actions are based completely on both its own experience and the built-in knowledge used in constructing that agent for the particular environment in which it operates.Wooldridge [8] extends the definition of an agent to an intelligent agent by extending the definition of autonomy to flexible autonomy. A multiagent based system consists of many agents interacting together with the aim to achieve a negotiated common goal.

In agent-based manufacturing process planning and scheduling systems, bidding-based negotiations or marketlike approaches are commonly used. In systems of this kind, the applied agent negotiation protocols require individual agents to reply to the incoming offers, to compete, and to negotiate or to bargain with other agents. As a result, rich knowledge bases and powerful learning and reasoning mechanisms are very important. Each agent should have at least knowledge of the *capability, availability, and cost of the physical resource* (e.g, a machine) represented by it. Some sophisticated agents need to have knowledge of other agents in the system, the products to be manufactured, and the know-how (historical experience, successful cases)as indicated by Gordillo and Giret [9].

2.1 Methodologies for Multi Agent System (MAS)

With the rapid growth and promise of the agent technology, a number of methodologies for developing MAS (denoted as "MAS methodologies") have been proposed in recent years as discussed by Quynh-Nhu and Low [10]. So far no standardized design methodology has been recognized for Multi Agent System as stated by Omiciniand Molesini[11]. There are many methodologies for analysis and design of multi-agent systems and some examples of the common existing methodologies are Multi-agent systems Software Engineering (MaSE), Gaia, Jade/Gaia, JADE, SODA (Societies in Open and Distributed Agent spaces) and MESSAGE.

Multi-agent systems Software Engineering (MaSE) is a start-to-end methodology that covers from the analysis to the implementation of a MAS as stated by DeLoach [12]. The main goal of MaSE is to guide a designer through the software life-cycle from a documented specification to an implemented agent system, with no dependency of a particular MAS architecture, agent architecture, programming language, or message-passing system.

The Gaia methodology views the system as a society or organization, with the elements of that society defined by roles. In Gaia, roles are initially captured in a prototypical role model, which are incrementally expanded and fully elaborated by the end of the analysis phase. These roles have direct correspondence to roles and role model defined in MaSE developed by Deloach [13]. This methodology gives the possibility to design MAS using an organizational paradigm and to traverse systematically the path that begins by setting out the demands of the problem and to lead to a fairly detailed and immediate implementation as shown by Florea et al[14]. Gaia permits to design a hierarchical non-overlapping structure of agents with a limited depth. From the



organizational point of view, agents form teams as they belong to a unique organization, they can explicitly communicate with other agents within the same organization by means of collaborations, and organizations can communicate between them by means of interactions. If inter-organization communication is omitted, coalitions and congregations may also be modelled as stated by Isern et al [15].

However, Isern et al [15] goes on to state that the GAIA methodology is somewhat limited since we can describe MAS with different architectures of agents. The strength of MaSE compared to Gaia is completeness of the methodology; all the stages of extracting the goals from functional requirements of the system, to definition of roles up until the agents are created are so clearly described in MaSE. It has a strong software and methodology support, unlike Gaia. The architecture that describes the internal behaviour of the agent is included in MaSE but in Gaia it's not well adopted.

Gaia methodology can be used in the development of MAS combined with JADE. Gaia methodology and the JADE develop an environment that enables implementation of a real multi-agent system as shown by Moraitis, and Petraki[16].JADE is a software development framework fully implemented in JAVA language aiming at the development of multi-agent systems and applications that comply with FIPA standards for intelligent agents. JADE provides standard agent technologies and offers to the developer a number of features in order to simplify the development process. Gaia methodology is an easy to use agent-orient software development methodology that however presently, covers only the phases of *analysis* and *design*.

The Societies in Open and Distributed Agent spaces (SODA) methodology developed by Omicini and Molesini [17] is an agent-oriented methodology for the analysis and design of agent-based systems. SODA focuses on inter-agent issues, like the engineering of societies and environment for MAS. It adopts Agents and Artifacts (A&A meta-model) as building blocks for MAS development. SODA introduces a simple layering principle in order to manage the complexity of the system description and adopts a tabular representation. MESSAGE developed by Caire and Leal[18] tries to integrate different methodologies. It is built on five viewpoints that are described with meta-models as UML extensions: organization, agents, goals/tasks, interactions and domain. Itadopted the Unified Process and centred on analysis and design phases of development.

2.2 Multiple Criteria Decision-Making (MCDM)

Multiple Criteria Decision-Making (MCDM) refers to making decisions in the presence of multiple and usually conflicting objectives. It is divided into Multi-Objective Decision-Making (MODM) and Multi-Attribute Decision-Making (MADM) as discussed by Wang, and Chin[[19]. MODM consists of a set of conflicting goals that cannot be satisfied simultaneously whilst MADM deals with the problem of choosing an alternative from a set of candidate alternatives which are characterized in terms of some attributes.

There are different Multiple Criteria Decision-Making models, among these are Linear Goal Programming (LGP), Multi Attribute Utility Theory (MAUT), Merit Point System (MPS) and Analytical Hierarchy Process (AHP). The Analytical Hierarchy Process (AHP) is perhaps the commonly used method for prioritisation of decision alternatives as proposed by Ababutain[20]. In this researchpaper, AHP is the method that was used in prioritisation of machines in the production system.

2.2.1 Analytic Hierarchy Process (AHP)

Analytic Hierarchy Process (AHP) developed by Saaty [21] is a multi-criteria decision-making approach which is used to solve complex decision problems. It uses a multi-level hierarchical structure of objectives, criteria, sub-criteria and alternatives. The top level of the hierarchy reflects the overall goal objective of the decision problem. The factors influencing the decision are represented in the intermediate levels. The lowest level comprises the decision alternatives. Nyanga et al [22] implemented AHP in the development of a multi agent system for machine selection.

3. SYSTEM MODELLING USING MULTI AGENT SYSTEM ENGINEERING (MASE)

MaSE was used in modelling of the system. The analysis phase consists of three steps which are capturing goals, applying use cases, and refining roles. The Design phase has four steps which include creating agent classes, constructing conversations, assembling agent classes, and system design. Unified Modelling Language (UML) was used to support the modelling of the system. The development of the designed Multi Agent System (*system design*) is considered in section four and it was implemented using Java Agent Development Environment (JADE) software.



3.1 System Analysis

At the analysis phase, a set of roles whose tasks describe what the system has to do to meet its overall goal were identified from various departmental functions which work hand in hand in manufacturing of the transformer as highlighted in the next paragraph. Each role played by every personal from coordinated sub systems of the enterprises were analysed on how they are directed and focused in meeting the overall goal. A role describes an entity (sub system) that performs some function within the system.

"The Sales Department captures the requirements of the customer and transfers them to the Production Engineering where the needs of the customer are translated into technical characteristics and specifications. The Production Engineering function develops initial plan of how the customers' requirements are going to be met (scheduling of jobs). It interacts with the Stores Department for provision of raw materials required in producing the product."

3.1.1 Capturing Goals

In capturing of goals, the following two sub-steps were considered; identifying goals and structuring goals. Primary and secondary data was gathered straight from the manufacturing source through questionnaires, meetings with the relevant people and from company records during the industrial visits. Industrial expects and the management were asked about the overall goal of the company in the transformer manufacturing business, the importance of Competitive Performance Objectives (time cost and quality) in the production. These Competitive Performance Objectives (CPOs) were compared to each other and ranked according to the fundamental 1-9 standard scale by Saaty [21] using a pair wise principle. A checklist was used as a guide to confirm if all aspects relating to area of study were looked at through structured interviews and observations.

The goals were identified from the functional requirements of the system obtained from the gathered data are listed as follows:

- Capture customers' requirements
- Translate requirements to specifications
- Plan for processing of the customers' requirements
- Supply of materials to the production system.
- Execute planned work (manufacture the product)

The identified goals are then structured into a Goal Hierarchy Diagram as shown inFigure 1. The highest-level goals (design, plan and supply of materials) are summarized to create an overall system goal which is to manufacturea transformer and the high level goals became sub-goals of the system goal. The sub goals were further decomposed into new sub goals supporting the parent goals.



Figure 1 Goal Hierarchy Diagram

3.1.2 Applying use case diagrams

The system works cooperatively with respect to other departmental functions having different roles in order to satisfy the demand or order placed. To represent how the system interacts with the environment as it captures the requirements of the system, use case diagrams were used. The use case showing the overall transformer manufacturing system (all involved departments) is represented in Figure 2.





Figure 2 Use case showing the overall transformer manufacturing system

The customer places an order at Sales Department which then captures requirements of the customer and *extends* them to the Production Engineering where planning is done. Planning *includes* requesting of materials from the Stores Department and scheduling of job. The Stores Person supplies materials upon receipt of a request from the Planning Department. After planning, the process *extends* to task execution and this is done by the Artisan.

3.1.2 Sequence Diagrams

The sequence of events that are transmitted between roles identified from use cases were represented (from the moment a customer places an order to loading of machine) as shown in the sequence diagram Figure 3. The roles or classes are laid across at the top and time is assumed to flow from the top of the diagram to the bottom represented by the lifelines in the diagram.



Figure 3Sequence Diagram showing flow of events between different roles

3.1.3 Refining Roles

The structured goals and roles represented in the sequence diagram were further refined by combining some roles into a single role and decomposing other roles in the case that it has been noted that there are tasks that cannot be carried out by the same role. Roles are the building blocks used to define agent's classes and capture system goals during the design phase. *Database* class have been added in this role refining stage and intend to add *user interface* as well at design phase.

Customer and sales roles were not considered during the refinement stage, more focus was put on the actual production of the product. These roles were not taken into account in the detailed design phase of agents and the role modelling. The developed MaSE role modelling is shown inFigure 4



 Image: Sector

SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 4 MaSE Role Model

When an order is placed, the production planner access the resource and manufacturing information from the database about the product that needs to be manufactured and develops materials schedule and work order. The materials schedule is used to request for materials from the stores and the work order guides the artisan in executing the planned work.

It can be noted that the MaSE role model is not really showing how the jobs are being allocated or scheduled to machines, which is the overall objective in this context. To show that, Figure 5 is a further refinement of roles and tasks to represent how the researchers intend to cost effectively assign planned jobs to machines. English auction where the bidder with the highest bid is the winner is used in bidding and allocating jobs as illustrated in Nyanga et al.[22]Once a customer has placed an order the production planner will propose bidding for jobs to the machines using contract net, a JADE default standard interaction protocol compliant of FIPA-ACL language for communication between agents, the initiator and responder. The machines will request the database to search for details of the job so that they will be able to prepare bids and submit to the production planner. The production planner will now compute bids using the English Auction protocol in order to determine the winner who will undertake the proposed job. This is going to be explained in detail at the design stage when these roles are being mapped to agents.



Figure 5 Refined roles showing how jobs are going to be allocated to machines

3.2 System Design

In the systems design stage roles were assigned to specific class of agent types by creating agent classes. The conversations between agents were developed showing how they interact, coordinate and negotiate to achieve the desired goal of effectively assigning part type components to machines in the production environment. A winner determination criteria (AHP) was developed to select the best machine to process the part type component by comparing the required machine specifications and the specifications of the machines in the factory as shown in Nyanga et al [23]. The agent's classes were then assembled with the internal architecture, reasoning processes and behaviour of the agent classes also designed. Finally, in the last step of System Design, the designer defines the actual number and location of agents in the deployed system and in this paper it is addressed in the next section (section four).

3.2.1 Creating Agent Classes

Roles are the foundation upon which agent classes are designed. Since roles correspond to the set of system goals defined in the Analysis phase, they form a bridge from what the system is looking forward to achieve to how that is going to be achieved in the design phase. The agent class diagram is shown in Figure 6.





SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 6 Agent Class diagram

Two Database Agents were included since the Engineering departmental function has got its own server different from that of stores department. There are other agents considered as system resources, such as databases and user interface in order to have complete interactive classes that associate to achieve a common goal. Conversations in which different agent classes must participate were identified and are shown in Table 1.

STEPS	AGENTS CONVERSATION PROTOCOL/ROLES
1.	 The USER triggers the PA to initiate a conversation by announcing a part type arrival to the
	PPA the moment an order is placed.
2.	 PPA identifies the part type ID and request for resource and manufacturing information
	from DBA1 for the particular part type component.
	 DBA provides the requested information to the PPA.
3.	 Based on the resource information retrieved by DBA1, PPA request for raw materials from
	the SPA.
	 SPA request for availability of raw materials from DBA2.
	 The DBA2 retrieves the requested raw materials details to the SPA and the SPA returns
	required materials information details to the PPA and notifies the USER about availability of
	the materials.
	 If the raw materials are unavailable, the PPA informs the PA and PA announces a new part
	type to the PPA is in <i>step 1</i> .
	 The USER acknowledges the receipt of the message about the availability/unavailability of
	the raw materials and waits for part to machine allocation.
4.	 PPA asks for submission of bids from MA for processing the part type component.
	 The Machines request for the process plan of the part type to be processed from DBA1,
	develop bids and submit the bids to the PPA.
	 PPA computes submitted bids using the embedded AHP algorithm to select the best
	machine (refer to Nyanga et al [23] and Nyanga et al [22]).
5.	 PPA notifies the MA about the winner/successful bidder.
	 The MA notifies the USER about the selected Machine to process part type.

Table 1 Agents' roles and conversations

3.2.2 Constructing Conversations

At this stage of the design phase, conversations that take place between agents, the initiator and the responder are now described in detail. The initiator always begins the conversation by sending the first message. When an agent receives a message, it compares it to its active conversations. If it finds a match, the agent transitions the appropriate conversation to a new state and performs any required actions or activities from either the transition or the new state. Otherwise, the agent assumes the message is a request to start a new conversation and compares it to all the possible conversations the agent can participate in with the agent that sent the message.

The detailed designs of conversations were derived from the concurrent tasks associated with roles. Agents' communication diagrams showing stages of communication from the point a Part type Agent (PA) is created up until the overall goal of allocating that part type component to the selected machine were developed. Figure 7 Communication class diagram (PPA - MA)shows the communication class diagram between the Production Planning



Agent (PPA) and the Machine Agent (MA). The same communication diagrams were developed for all other agents in the system.



Figure 7 Communication class diagram (PPA - MA)

PPA - **MA**: The Production Planning Agent (PPA) invites the Machine Agent(MA) for submitting of bids to process part type with the requested PTID, waits to receive the bids and compute the bids after receiving the submitted bids. PPA computes bids using a predetermined and embedded AHP algorithmin the system to select the best machine with highest priority bid value and send the message of the winning bidder to the Machine Agent (MA).

MA - **PPA:** The Machine Agent (MA) first request for the processing part type details using the PTID from DBA1 and develop bids based on the provided process plan. It then submits the bids (available machines) and waits to receive response of acceptance/rejection.

3.2.3 Winner Determination criteria, Analytic Hierarchy Process (AHP)

Analytic Hierarchy Process (AHP) algorithm was formulated for multi-criteria decision-making in selection of the right machine to process a part type component in the manufacturing of a transformer tank. It uses a multi-level hierarchical structure of objectives, criteria, sub-criteria and alternatives. The methodology of selectingthe best machine to perform the intended task is based on the priority ranking of each and every machine in the plant developed by Nyanga et al[23]. The machine priority ranking values will Production Planning Agent (PPA) requests the DataBase Agent (DBA) for retrieval of the best suitable machine according to the AHP machines; the Production Planning Agent (PPA) interacts with Machine Agent (MA) to submit available machines as bids and in order to determine the winner by computing the submitted bids, the winner by computing the submitted bids, the Production Planning Agent (PPA) requests the DataBase Agent (DBA) for retrieval of the bast suitable machines as bids and in order to determine the winner by computing the submitted bids, the Production Planning Agent (PPA) requests the DataBase Agent (DBA) for retrieval of the bast suitable machines as bids.

3.2.4 The AHP hierarchy structure of machine selection

The machine selection when a job is introduced in the manufacturing system, it is based on time, cost and quality as shown in Figure 8 Machine selection top hierarchy. The best machine selection being the overall objective of the MAS system is decomposed from top level hierarchy to second level (sub criteria). Time, cost and quality are the three selection criteria factors considered.



Figure 8 Machine selection top hierarchy



3.2.5 Sub criteria (Time, Cost and Quality)

The factors (set up time, processing time and machine availability) are considered in determining the time priority. In order to increase throughput of the production process, the machine with the minimum set up time, loading time and processing time are given high priority values. If the machine with the highest priority value is busy, the next available machine is considered according to its priority value. The same is being done in terms of cost and quality/capability.

For the cost analysis, operational/processing cost of machine, machine maintenance cost and material handling cost were considered to come up with an annual cost of a machine. The objective was to select the machine that process part type component at a minimum cost meanwhile greatly benefiting from that operation. The annual worth of machine was calculated using the formula developed by Cimren and Budak[24].

$$AW_{K} = \frac{1}{P_{K}} \left(\frac{(1+r)^{n}-1}{i(1+r)^{n}} \right)$$
(1)

$$P_K = A_{K0}(1+r)^0 + A_{K1}(1+r)^{-1} + A_{K2}(1+r)^{-2} + \dots + A_{Kn}(1+r)^{-n}$$
(2)

$$A_{kt} = PC_{kt} + MMC_{kt} \tag{3}$$

Where Ak_t is the annual cost of the machine k in year t, r is the annual interest rate, P_k is the net present value of the machine k, AW_k being the annual worth of machine k, PC_{kt} is the processing cost of the machine k in year t, MMC_{kt} is the machine maintenance cost of machine k in year t, and nk is the economic life of machine k in year t. (t = 0, ..., nk).

In coming up with quality (or capability) priority, productivity and flexibility sub criterion were considered as illustrated in Nyanga et al[23], Nyanga et al [25] and Nyanga et al [26]. Productivity sub-criterion is being determined by the following machine attributes; machine power, machine speed, machine precision and dependability. Flexibility sub-criterion considers the ability of the machine to perform different operations; number of tool slots, operation flexibility and mobility (fixed/mobile) are the machine attribute being considered in prioritising the selection of the best machine to process a part type component

3.2.6 Assembling Agents

The internals of the agents were created at this stage. Architectures of each of the agents were developed from the logic based aspect of how the agent should behave. The behaviour of the each agent is explained in detail and the conversations that take place between agents. When developing agents, each task from each role played by an agent defines a component in that agent class. The components for each of the agent are represented in the table showing the conversation ID of each as shown in Table 2.

Agent No.	Agents	Components	Conversation ID
1	PA	Announce arrival	1.1
2	PPA	Request for materials	2.1
		Inform PA about materials availability	2.2
		Request PT operations from DBA1	2.3
		cfp from MA	2.4
		Compute bids	2.5
		Inform winner	2.6
3	MA	Request for PP and compatible machines	3.1
		Submit bids	3.2
		Inform USER	3.3
4	SPA	Check for materials availability	4.1
		Inform PPA	4.2
		Inform USER	4.3
5	DBA1	Retrieves PT Operations	5.1
		Retrieves compatible machines	5.2
6	DBA2	Retrieves materials available	6.1
7	USER	Input PTID	7.1
		Acknowledge availability of materials	7.2
		Acknowledge unavailability of materials	7.3

Table 2 Components of agents and the associated conv ID

3.2.7 Production Planning Agent Architecture

Out of all the agents, Production Planning Agent (PPA) is the agent which plays many roles because it acts as the intermediator between the Machine Agent (MA) and Part Agent (PA). It frequently communicates with Database Agent (DBA) as well. It is made of different components as in Figure 9. The request resource information will be connected to DBA1, request for materials from SPA and request for bids from MA. How the components interact is represented by the thin arrows connecting them together. The architecture of other agents were developed showing their different roles and how they interact with each other.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 9 Production Planning Agent Architecture

3.2.8 Message Templates

The message template shown in Table 3 shows the interactions that takes place between agents, the interaction protocol (IP) that is used in the conversation, the *role* of initiating and responding *with* other agents *when* triggered by a certain event occurs. The template column describes the conversation IDs for each interaction that takes place on how the conversations IDs were determined.

Interaction	IP	Role	With	When	Template
Announce arrival (PA)	Query if	I	PPA	Order is placed	Conv 1.1
Request for materials (PPA)	Request protocol	I	SPA	PTID is announced into the system	Conv 2.1
Access stores DataBase (SPA)	Request	Ι	DBA	PPA makes a request	Conv 4.1
Retrieves materials info (DBA2)	Query if	R	SPA	SPA makes a request	Conv 6.1
Informs USER about materials status (SPA)	Query if	R	USER	DBA retrieves info	Conv 4.3
Informs PPA about materials status (SPA)	Query if	R	PPA	DBA retrieves info	Conv 4.2
Informs PA about materials status (PPA)	Query if	R	PA	SPA informs PPA about materials	Conv 2.2
Call for proposal (PPA)	Contract.net protocol	I	MA	SPA informs availability of	Conv 2.4
Submit bids (machine availability), (MA)	Contract.net protocol	R	PPA	It receives a call for proposal	Conv 3.2
Compute bids (PPA)	Request protocol	I	DBA	It receives bids	Conv 2.5
Retrieves highest priority machines among the submitted machines and inform MA. (PPA)	Request protocol	R	PPA &USER	Request for priority by PPA	Conv 2.6
Inform User (MA)	Query	I	USER	Informed about the winner	Conv 3.3

Table 3 Message Template

3.2.9 Agent Behaviours

For each agent to perform a particular role, it should have behaviour in order to execute the intended role. An agent can have different behaviours depending with the roles that it is supposes to play. The behaviours of the agent are executed concurrently depending with flow of interactions between agents. The behaviours of the two main agents; Production Planning Agent (PPA) (Figure 10) and Machine Agent (MA)(Figure 11)were considered in this paper. However other agents' behaviours were developed as well for easy implementation of the whole MAS system.





Figure 10 Dynamic behavior pattern for Production Planning Agent (PPA)

When the PA announces arrival to the PPA, the Production Planning Agent (PPA) starts to execute its behaviours which are *one shot* and *cyclic behaviour* concurrently. The ticker behaviour is also nested in the cyclic behaviour to periodically execute the set operation. The one shot behaviour is being applied when it is requesting for materials from the Stores Person Agent, this is done only once. The moment the one shot behaviour is terminated, the cyclic behaviour is initialised as the PPA interacts with the Machine Agent (MA) when calling for proposals (cfp). This is done repeatedly since there should always be something in the PPA catalogue (PT) ready for processing and this is when the ticker behaviour is being applied as well. The cyclic behaviour is executed at two scenarios including when the PPA is computing bids to determine the winner using the Analytical Hierarchy Process (AHP) priority algorithm. Every moment, the Machine Agent (MA) submits available machines as bids, the Production Planning Agent periodically computes them and informs winner.



Figure 11 Dynamic behaviour pattern for Machine Agent (MA)

How the Machine Agent (MA) responds to the request of the jobs that would have been placed in the catalogues is determined by its behaviour. The Machine Agent (MA) should always check for jobs placed in catalogue repeatedly so that it can submit bids. This behaviour is executed by the ticker behaviour of the machine agent as it always checks for availability of machines. The other behaviour being executed by the Machines Agent (MA) is one shot behaviour as it informs the USER about which machine is suppose do perform the auctioned job, Figure 11shows the all the behaviours of MA.

4. IMPLEMENTATION OF MAS (DEPLOYMENT)

The Production Planner user interface was developed using Active Server Pages (ASP.NET). The (ASP.NET) is an objected oriented language that is not limited to script languages; it allows the developer to make use of .NET languages like C#, J#, VB. However, ASP does not support agent based system in the JADE environment; therefore JADE LEAP was used to interconnect the ASP.NET developed interface with JADE as shown Figure 12.





Figure 12 ASP.NET connected to JADE using JADE LEAP

The interface has a provision where the user first request for availability of the materials from the Stores Department by entering the Part ID and quantity required as shown in Figure 13. At the background the Production Planning agent request for materials from the Stores Person agent and informs the user about the materials status.

MULTI-AGENT BASED SYSTEM				
File About				
REQUEST FOR MA	TERIALS			
Enter Part ID	Description Quantity Required			
PT1 -	Transformer base 2			
	Materials			
Materials Requested Availabl	MS Sheet Metal			
Request Materials App	hy Cancel			

Figure 13Production planning user interface

The Multi Agent System (MAS) is implemented using Java Agent Development Environment (JADE). JADE is a software platform that provides basic middleware-layer functionalities which are independent of the specific application and which simplify the realization of distributed applications that exploit the software agent abstraction. The developed Multi Agent System (MAS) model supports integration of departments (Stores department, Production planning department and the Mechanical department). It also allows running of distributed application across a network in different working environments. The developed model allows the production planner to schedule jobs in real time. Figure 14 shows how the production planner plans for a placed order obtained in running of the MAS developed model during testing.

Agent container Main-Container@localhost is ready.
Production Planning Agent PPA@localhost:1099/JADE is ready. Request for processing for Part: PT1 PT1 exists.
Proceeding to request for machines to process
Database Agent DBA@localhost:1099/JADE is ready.
Machine Agent MA@localhost:1099/JADE is ready.
Waiting for machines to process Parttype PT1
Found the following Machine Agents:
MA@localhost:1099/JADE
PartID process request received by Machine-Agent from PPA: PT1
The following processes will be applied on:PT1
PT1 Cutting
PT1 Marking
PT1 Punching
PT1 Bending
Machine-Agent responds to PPA with Machine Catalogue.
Attempt failed: PT1 cannot be processed/ Machines busy.
The following machine Gillotine Machine and priority 980 is available for:PT1
The following machine Marking 1 (scribers/m.rules) and priority 6825 is available for:PT1
The following machine Marking 2 (scribers/m.rules) and priority 6811 is available for:PT1
The following machine Punching Machine 1 and priority 1374 is available for:PTI
The following machine Punching Maching 2 and priority 1179 is available for:PTI
The following machine Cropping Machine 1 and priority 124/ is available for:Pl1
The following machine Cropping Machine 2 and priority 1193 is available for:Pli
The following machine Cropping Machine 5 and priority 1197 is available for:Pli
The following machine Cropping Machine 4 and priority 102 is available for Pri
Maiting for machines to process Party and T1
Found the following Machine Agents:
Mallocalbest:1099/100F
PartID process request received by Machine-Agent from PPA: PT1
Machine-Agent responds to PPA with Machine Catalogue.
Bids received by PPA from MA: ((ConversationId: Parttype-trade) AND (InReplyTo: cfp1433279925128))
PPA computing Bids from MA
Confirmation of machine allocation to PPA from DBA

Figure 14 Results of the developed Multi Agent System



5. RESULTS

Order processing of 100kva transformers shown in Table 4 is used to assess the effect of the MAS by comparing the decisions made by a human expert and the MAS.

Time (Days)	Number of orders/day
1	1
2	0
3	3
4	4
5	2
6	3
7	4
8	0
9	1
10	6

Table 4 Number of orders per for 100kva

The current existing scenario of developing a work order for a raised job at the company takes an average of 2.5 minutes as time is being spent in interacting with relevant departments to enquire manufacturing and resource information. Implementation of a real time Multi Agent System (MAS) will reduce the lead time in processing of work order by 83% as shown in Figure 15, thus increased overall throughput is achieved.



Figure 15 Lead times of processing work order

A sample of frequency orders per for a 100kva that was used in creating arrival rates in the simulation of the existing production system at the company as shown in Figure 15. The average machine utilization at the workstations when a human expert is used to plan for production and when a MAS is used are shown in Figure 16. The results show that machine utilisation is increased from 73% to 84%. Prioritization of machines is increased from 0.193 using a human expert to 0.197 using the MAS.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 16 Machine utilization

6. CONCLUSION

The application of Analytical Hierarchy Process (AHP) in conjunction with a Multi Agent System (MAS) ensures high machine utilization and increased throughput. The process of machine selection is automated by developing a multi-agent system that handles the interactions and negotiation in order to achieve an optimised solution. Different departments of a single factory are used to access the performance of the system. The departments can be extended to different companies in an Industrial cluster. This enables the creation of anE-manufacturing system in which resources are shared using a network. Multi Agent System (MAS) enhances efficiencies in management of system files through integration of departmental functions. The proposed Multi Agent System will outperform the current existing system in execution of operations during production planning and control.

REFERENCEE-MANUS

- [1] Ayton and Lawley, 2005. Executing production schedules in the face of uncertainties: a review and some future direction. *European Journal of Operatio Research*, 161(1), pp 86-110.
- [2] Shen, W.; Norrie, D.H.1999. Agent-Based Systems for Intelligent Manufacturing: A State-of-the-Art Survey. Knowl. Inf. Syst. 1, 129-156.
- [3] Colombo,A. W., Schoop, R. and Neubert, R. 2006. An agent-based intelligent control platform for industrialholonic manufacturing systems, *IEEE Transactions on Industrial Electronics*, 53(1), pp. 322-337.
- [4] Zhang, D. Z., Anosike, A. and Lim, M. K. 2007. Dynamically integrated manufacturing systems (DIMS) a multi-agent approach, *IEEE Transactions on Systems, Man, and Cybernetics, Part A*, 37(5), pp. 824-850.
- [5] Nyanga, L. and Van Der Merwe A.F., (2012), E-Manufacturing: A Framework For Increasing Manufacturing Resource Utilisation, *CIE42 Proceedings*, 16-18 July 2012, Cape Town, South Africa
- [6] Russel, S. & Norvig, P., 1995. Artificial Intelligence: A Modern Approach. Eaglewood Cliffs, USA: Prentice-Hall
- [7] Wooldridge, M. and Jennings, N. R. 1995. Intelligent agents: theory and practice, *The Knowledge Engineering Review*, 10(2), pp. 115-152.
- [8] Wooldridge, M. 1999. "Intelligent agents," in *Multi-Agent Systems, M. Wooldridge and G. Weiss, Eds.*, pp. 3-51, MIT Press, Cambridge, Mass, USA.
- [9] Gordillo, A. and Giret, A., 2014, Performance Evaluation of Bidding-Based Multi-Agent Scheduling Algorithms for Manufacturing Systems, *Machines* 2014, 2, pp233-254; doi:10.3390/machines2040233
- [10]Quynh-Nhu N. T. and Low, G, 2004, A Preliminary Comparative Feature Analysis of Multi-agent Systems Development Methodologies.
- [11]Omicini, AandMolesini, A, 2008, The SODA Methodology Multiagent Systems LS SistemiMultiagente LS, Universita di Bologna a Cesena, Academic Year 2007/2008.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

- [12]DeLoach,S.A., 2004, "The MaSE methodology", in Methodologies and Software Engineering for Agent Systems, F. Bergenti, M.P Gleizes, F. Zambonelli, Eds. *The Agent-oriented Software Engineering Handbook*. Kluwer Academic Publishers, 2004, pp. 107-125
- [13]Deloach, S.A and Mark M, 2001., Multiagent Systems Engineering, International Journal of Software Engineering and Knowledge Engineering, 11(3), pp231-258
- [14]Florea, A. M., Kayser, D, Pentiuc, S., 2002, A. El FallahSegrounichi, Intelligents agents, Agents Intelligents, Politechnica University of Bucharest, 2002.
- [15]Isern, D., Sanchez, D. and Moreno, A., 2011, Organizational structures supported by agent-oriented methodologies, *The Journal of Systems and Software*, 84(2), pp. 169-184.
- [16]Moraitis, P. and Petraki, E., 2001. Engineering JADE Agents with the Gaia Methodology, Dept. of Computer Science University of Cyprus.
- [17]Omicini, A and Molesini, A, 2008, The SODA Methodology Multiagent Systems LS SistemiMultiagente LS, Universita di Bologna a Cesena, Academic Year 2007/2008.
- [18]Caire, G, Leal. 2001. Agent oriented analysis using MESSAGE/UML. Agent-Oriented Software Engineering II: 2nd International Workshop, LNCS 2222, pages 119-135. Springer Verlag.
- [19] Wang, Y-M. and Chin, K-S. 2009. A new approach for the selection of advanced manufacturing technologies: DEA with double frontiers. *International Journal of Production Research*, 47 (23), pp 6663-6679.
- [20]Ababutain, A., 2002. A Multi-Criteria Decision-Making model for selection of bot toll road proposals within the public sector, pp 39
- [21]Saaty, T.L.1977. A Scaling Method for Priorities in Hierarchical Structures, Journal of Mathematical Psychology, 15, pp 57-68.
- [22]Nyanga, L., Van Der Merwe, A.F., Tsikira, C., Matope, S. and Dewa, M.T., 2016, Design of a Multi Agent System for machine selection., *International Conference on Competitive Manufacturing* 16, 27 -29 January, Stellenbosch, South Africa
- [23]Nyanga. L, Van der Merwe, A.F. Matope S. and Dewa, M.T. 2014. A decision support system framework for machine selection, *SAIIE26 Proceedings*, 14th 16th of July 2014, Muldersdrift, South Africa © 2014 SAIIE
- [24]Cimren, E. and Budak, E. 2000, Development of a machine tool selection system using Analytical Hierarchy Process, Intelligent computation in manufacturing Engineering-4.
- [25]Nyanga, L., Van der Merwe, A.F., Matope, S. and Dewa, M.T., 2014, A web based manufacturability agent framework for an E- manufacturing system, *3rd CIRP Global Web Conference*, June 3-5
- [26]Nyanga, L., Van Der Merwe, A.F., Burawa, M., Matope, S. and Dewa, M.T., 2016, Agent based job scheduling for a vehicle engine reconditioning machine shop, International Conference on Competitive Manufacturing 16, 27 - 29 January, Stellenbosch, South Africa



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



DESIGN OPTIMIZATION OF A SCHEDULING SYSTEM FOR A MULTI-PRODUCT FLOW LINE ENTERPRISE IN ZIMBABWE

T. Munetsiwa^{1*}, L. Masiyazi², M.T Dewa³, D. Museka⁴

^{1,2,4} Department of Industrial and Manufacturing Engineering Harare Institute of Technology, Zimbabwe <u>tmunetsiwa@gmail.com</u> <u>masiyazil@gmail.com</u> <u>musekahit@gmail.com</u>

> ³Department of Industrial Engineering University of Stellenbosch, South Africa <u>mnce2009@gmail.com</u>

ABSTRACT

This paper presents a simulation-based scheduling tool which assists production planners in Zimbabwe to generate optimal production schedules, quickly react to unexpected events and give customers more accurate delivery promises. It was noticed that a cable manufacturing company suffered a lot by failing to meet customer due dates as a result of using manual static scheduling techniques which are inadequate to use in such a dynamic environment. The proposed system employed discrete-event simulation modeling as a means to evaluate the sequencing rules and then use tardiness and flow times as performance measurement parameters in order to select one optimum rule. The objectives pursued include developing the description of processes document, designing a simulation model for the plant operations, designing an information system and performing experimental designs of the simulation model. A user friendly excel interface was designed so as to retrieve data from Arena and present it in a more comprehensive manner. 2 sets of experiments were done under different conditions and the Simulation based Scheduling system was developed. The system is recommended for use by small to medium cable manufacturers in Zimbabwe.

^{1*}The author was enrolled for a B.Tech (Industrial and Manufacturing) degree in the Department of Industrial and Manufacturing Engineering, Harare Institute of Technology

^{2,4} The authors are lecturers in the Department of Industrial and Manufacturing Engineering, Harare Institute of Technology.

³The author is enrolled for a PhD (Industrial) degree in the Department of Industrial Engineering, University of Stellenbosch.



1. INTRODUCTION:

1.1 Background to the Study

A cable manufacturing company that manufactures and supplies an extensive range of cables and allied products for the transmission and distribution of information and energy for the Southern and Central African markets, has a wide range of products. The policy used is a make to order policy. For some stock the cost of cable manufacturing is high therefore each product has a minimum order quantity and when this quantity has been satisfied, the order is accepted for manufacture.

Currently the scheduling system being used is a static manual scheduling system whereby the schedule is generated manually by the production planner and is unaltered until the next scheduling period. However, besides being complex and time consuming, this production scheduling technique is ineffective in a dynamic environment since it does not manipulate occurrences of unexpected events such as cancellation of jobs, changes in job priority and much more. If an unexpected event should occur, the available schedule might then be far from optimal, or will be invalidated. This has resulted in many instances where there would be bottlenecks, more idle times, and low machine utilization. Consequently this has seen the company constantly failing to meet targeted customer due dates.

The focus of the paper was to address the problem of customer due dates not being met due to inadequate scheduling techniques therefore a simulation based production scheduling system was developed. Objectives pursued included design of an information system and the determination of optimal and effective methods for schedule reconfiguration to accommodate the dynamic changes.

2. RELATED LITERATURE

2.1 Dynamic Scheduling and Real Time Events

De Jong [1] pointed out that a scheduling problem is dynamic if changes in the problem properties or in the problem environment cause a previously produced schedule to be invalidated. Cowling and Johnson [2] highlighted that a dynamic scheduling system is one that uses real time information as it arrive. One frequent assumption of scheduling theory, which rarely holds in practise, is that the scheduling environment is static. Petrovic et al. [3], stated that manufacturing environments are dynamic in nature and are subject to various disruptions, referred to as real-time events, which can change system status and affect its performance. Vieira et al. [4], highlighted that these real time events can be categorized into resource-related and job-related. Resource-related include machine breakdown, unavailability or tool failure. Job related include rush jobs, job cancellation, due date changes, change in job priority etc.

2.2 Dispatching Rules and Simulation

Vieira et al. [4], noted that no single dispatching rule excels well in all criteria. They highlighted that in order to assess the performance of various dispatching rules dynamically under different dynamic and stochastic conditions of the shop floor, simulation can be used. Simulation has a long and strong track record in the analysis of manufacturing systems whose complexity and interaction of components defy closed-form methods according to Clark [5]. Tamaki et.al [6] also stated that one of the most promising manufacturing activities to employ simulation techniques and tools is production scheduling. Further, recent advances in techniques of analysis and computer software enable simulation to provide excellent support to real-time, hence dynamic and adaptable scheduling, and this was purported by Harmonosky [7]. Simulation allows the execution of several dispatching rules, and the rule that yields the best performance is selected.

2.3 Heuristics

According to Vieira et al. [4] heuristics are problem specific schedule repair methods, which do not guarantee to find an optimal schedule, but have the ability to find reasonably good solutions in a short time. Heuristics may therefore be used to increase the chances of obtaining solutions of acceptable quality. A heuristic can be defined as an inexact algorithm that is based on non-rational and credible arguments which may lead to reasonable solutions n a short time for a given problem, but are not guaranteed to do so according to De Jong [1]. Michelle [8] suggests that the use of heuristics will help decrease the number of variables and unknowns, resulting in a more efficient scheduling process. As a result, the use of heuristics may decrease the time it requires to solve a problem, but the trade-off is optimality in the solution. Heuristic functions and algorithms De Jong [1]. Heuristic algorithm applied may or may not accept the new schedule as a replacement of the current schedule. This only goes to show that heuristic optimization involves much fine tuning on the side of the algorithm, the neighbourhood function, and objective function Coy et. al [9].



2.4 Approaches to Dynamic Scheduling

Leus et al. [10], classified dynamic scheduling into four classes which are: complete reactive scheduling, predictive-reactive scheduling and robust pro-active scheduling.

2.4.1 Complete Reactive Scheduling

In a completely reactive scheduling no firm schedule is generated in advance and decisions are made locally in real-time. Priority dispatching rules are frequently used. A dispatching rule is used to select the next job with highest priority to be processed from a set of jobs awaiting service at a machine that becomes free.

2.4.2 Predictive Reactive Scheduling

Predictive-reactive scheduling is a scheduling/rescheduling process in which schedules are revised in response to real-time events. Predictive-reactive scheduling is a two-step process. First, a predictive schedule is generated in advance with the objective of optimizing shop performance without considering possible disruptions on the shop floor. This schedule is then modified during execution in response to real-time events.

2.4.3 Robust Predictive-reactive scheduling

Most of the predictive-reactive scheduling strategies are based on simple schedule adjustments which consider only shop efficiency. The new schedule may deviate significantly from the original schedule, which can seriously affect other planning activities that are based on the original schedule and may lead to poor performance of the schedule. It is therefore desirable to generate predictive-reactive schedules that are robust. Robust predictivereactive scheduling focuses on building predictive-reactive schedules to minimize the effects of disruption on the performance measure value of the realized schedule.

2.4.4 Robust Pro-active Scheduling.

Vieira et al. [4], highlighted that a robust pro-active scheduling approach focuses on building predictive schedules which satisfy performance requirements predictably in a dynamic environment.

2.5 When to Reschedule in Real time Events

There are three policies that have been proposed in the literature, which are: Periodic, Event driven, and hybrid.

2.5.1 Periodic Policy

Under this policy, schedules are generated at regular intervals, which gather all available information from the shop floor. The dynamic scheduling problem is decomposed into a series of static problems that can be solved by using classical scheduling algorithms. The schedule is then executed and not revised until the next period begins, where the planning horizon is renewed by taking into account new information gathered from the current shop floor status. The periodic policy yields more schedule stability and less schedule nervousness. Unfortunately, following an established schedule in the face of significant changes in the shop floor status may compromise performance since unwanted products or intermediates may be produced. Determining the rescheduling period is also a difficult task.

2.5.2 Event Driven Policy

In this policy, rescheduling is triggered in response to an unexpected event that alters the current system status.

2.5.3 Hybrid

A hybrid policy reschedules the system periodically and also when an exception occurs. Events usually considered are machine breakdowns, arrival of urgent jobs, cancellation of jobs, or job priority changes.

It has been reviewed that although the use of heuristics in dynamic scheduling is simple, the drawback with them is their possibility of becoming stuck in poor local optima and it requires much fine tuning on the side of the algorithm, the neighbourhood function, and objective function. The use of simulation provides a very promising method to solve dynamic scheduling problems. Furthermore, recent advances in techniques of analysis and computer software enable simulation to provide excellent support to real-time, hence dynamic and adaptable, scheduling.



3. RESEARCH METHODS

The procedures adopted by the researchers was an experimental design research, and used a combination of quantitative and qualitative techniques. Focus was on the following product lines:

- PVC 35×4C RS (UA) (Poly Vinyl Chloride insulated, 35mm diameter, 4 Cored, Red Striped, Unarmoured cable).
- PVC 150×4C RS (SWA) (Poly Vinyl Chloride insulated, 150mm diameter, 4 Cored, Red Striped, Steel Wire Armoured cable).
- LHC 70×4C BS (UA) (Low Halogen Compound insulated, 70mm diameter, 4 Cored, Blue Striped, Unarmoured cable).

3.1 Interviews

Informal interviews with two production personnel at the cable manufacturing company were carried out. From the interview it was revealed that the company suffers a lot by not meeting customer due dates. It was also highlighted that there are a lot of bottlenecks in the system which usually results in low machine utilization. A root cause analysis and several brainstorming sessions were carried out and it was then noted that the scheduling system being used is not only manual and time consuming, but it is inadequate to use in a dynamic environment.

3.2 System Design

After obtaining the processing times from a time study, Microsoft Visio was used to generate the conceptual model of the system through flow process diagrams and workflow diagrams prior to the development of the simulation model. The Simulation model was then designed using Rockwell ARENA. Microsoft Excel was used to design the user friendly interface of the scheduling system. The user interface allows the user to enter data such as the orders for the day, rush orders and system capacity for a particular scheduling period, and allows the retrieval of simplified information of the statistics of scheduling rules that may be adopted.

The Arena Simulation Model was designed using the steps summarized in Fig. 1.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Fig 1: Steps Taken in developing a Simulation Model



4. SYSTEM DEVELOPMENT

Operations at the cable manufacturing company are summarized in the workflow and process flow diagrams given Fig. 2 and Fig. 3.



Fig 2: Workflow diagram




Key:

- Product A represent the PVC 35×4C RS (UA)
- Product B represent the LHC 75×4C BS (UA)
- Product C represent the PVC 150×4C RS (SWA)

Fig. 2 and 3 show that all the products go through almost the same processing routes. This makes scheduling of these products difficult since there are many products to be processed with limited resources.

4.1 Information System

The information system provided a user-friendly interface to write the orders to be scheduled. It automatically retrieved the simulated results from Arena and presented them in a more comprehensive manner where it is analysed. The database was designed using Microsoft Excel. The interfaces are as shown in Fig. 4 and Fig. 5:



	DAT	A EN	TRY	USE	R INTEF	RFACE		
			QUANTITY		Est. ORDER PROCESSING TIMES	EXPECTED		
PRODUCT DESCRIPTON	PRODUCT TYPE	Order Number	(km)	ARRIVAL DATE	(days)	FINISHING DATE	DUE DATES	COVERT
PVC 35X4C RS (UA)	А	5	2.5	10/08/2016	4.0	16/08/2016	18/08/2016	0.5
LHC 70X4C BS (UA)	В	3	3	09/08/2016	5.0	17/08/2016	18/08/2016	0.2
LHC 70X4C BS (UA)	BS (UA) B 6		0.8	08/08/2016	1.0	13/08/2016	14/08/2016	1
PVC I50X4C RS (SWA)	OX4C RS (SWA) C		1.2	11/08/2016	2.0	14/08/2016	14/08/2016	0
					0	0		0
					0	0		0
FISFS	EDD							
		SEE RES	<u>ULTS</u>					
COVERT	SPT							<u></u>

Fig 4: Order data entry interface

	RESULTS ANALYSIS								
				FLOW TIME		DUEDATE	TARDINESS		
PRODUCT DESCRIPTION	ENTITYTYPE	Order Number	STARTDATE	(Days)	ACTUAL FINISH DATE	DUEDATE	(Days)		
PVC 35X4C RS (UA)	A	5	12/08/2016	4.38	16/08/2016	18/08/2016	-1.62		
LHC 70X4C BS (UA)	В	3	12/08/2016	6.57	18/08/2016	18/08/2016	0.57		
LHC 70X4C BS (UA)	В	6	12/08/2016	0.97	12/08/2016	14/08/2016	-1.03		
PVC I50X4C RS (SWA)	С	1	12/08/2016	2.63	14/08/2016	14/08/2016	0.63		
				0.00			0		
				0.00			0		
	4								
		AVERAGE FL	OW TIME	3.64					
		MAXIMUM F	LOW TIME	6.57					

Fig 5: Simulated results interface

4.2 Simulation Model Design

The simulation model was designed using a discrete event simulation software called ARENA. The role of simulation is to mimic the real plant operations under some given conditions and evaluate the system performance, then automatically feed the results to the Microsoft Excel Database. The Arena model logic is shown in Fig. 6.



Fig 6: The Model Logic



5. RESULTS AND DISCUSSIONS

5.1 Experimentation

Typical scenarios were replicated to represent a manufacturing system. Table 1 shows four orders used to test the model.

ORDER	QUANTITY (Km)	ARRIVAL DATE	DUE DATE
C1	7	24/11/2015	25/12/2015
A5	7	25/11/2015	20/12/2015
B3	12	05/12/2015	25/12/2015
A6	15	20/11/2015	22/12/2015

Table 1: Orders Table

Table 2 shows simulation results obtained using the Earliest Due Date dispatching rule.

EARLIEST DUE DATE (EDD)								
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)						
C1	6.42 21.42							
A5	-0.78 9.21							
B3	1.73 16.73							
A6	0.88	12.88						
Average (hrs.)	2.26	15.06						
Maximum (hrs.)	6.42	21.42						
Total Tardiness (hrs.)	9.03							
Number Of Tardy	3							
SH	ORTEST PROCESSING TIME (SF	РТ)						
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)						
C1	0.47	15.47						
A5	-0.34	9.66						
B3	2.39	17.39						
A6	7.55	19.55						
Average (hrs.)	2.60	15.51						
Maximum (hrs.)	7.55 19.55							
Total Tardiness (hrs.)	10.41							
Number Of Tardy	3							
FIRS	T IN SYSTEM FIRST SERVED (F	ISFS)						
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)						
C1	-0.39	14.61						
A5	6.93	16.93						
B3	4.09	19.09						
A6	-3.80	8.20						
Average (hrs.)	2.75	14.71						
Maximum (hrs.)	6.93	19.09						
Total Tardiness (hrs.)	11.02							
Number Of Tardy	2							
	COST OVER TIME (COVERT)							
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)						
C1	4.55	19.55						
A5	7.39	17.39						
B3	10.23	25.23						
A6	15.95 27.95							
Average (hrs.)	9.53 22.53							
Maximum (hrs.)	15.95 27.95							
Total Tardiness (hrs.)	38.12							
Number Of Tardy	4							

Table 2: Experimental Results Table



Comment

From the simulation results it can be deduced that in order to reduce the number of tardy jobs and to reduce flow time, the FISFS dispatching rule can be used.

Continuing with the experiments by including two more orders that have just arrived into the current schedule. Table 3 and Table 4 give the results obtained when using the different dispatching rules.

ORDER	QUANTITY	ARRIVAL DATE	DUE DATE
A6	15	20/11/2015	22/12/2015
C1	7	24/11/2015	25/12/2015
A5	7	25/11/2015	20/12/2015
B3	12	05/12/2015	25/12/2015
C4	9	10/12/2015	29/12/2015
B2	8	10/12/2015	30/12/2015

Table 3:	Orders Table

FARLIEST DUE DATE (EDD)									
CT	1 92	11.02							
AJ	1.03	21 E4							
DJ	0.00	21.34							
A0	3.71	23.00							
C4	7.10	20.10							
	8.40	26.40							
Average (nrs.)	6.0	21.23							
	8.80	28.46							
Iotal lardiness (hrs.)	36.50								
Number Of Tardy									
SH	ORTEST PROCESSING TIME (SP								
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)							
C1	5.14	20.14							
A5	-1.74	8.26							
В3	6.92	21.92							
A6	12.13	24.14							
C4	7.74	26.74							
B2	-3.72	16.28							
Average (hrs.)	5.32	19.58							
Maximum (hrs.)	12.14	26.74							
Total Tardiness (hrs.)	31.95								
Number Of Tardy	4								
FIRS	T IN SYSTEM FIRST SERVED (F	ISFS)							
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)							
C1	2.87	17.88							
A5	12.17	22.17							
B3	9.29	24.29							
A6	2.26	14.29							
C4	7.53	26.53							
B2	8.88	28.88							
Average (hrs.)	i.) 7.17 22.34								
Maximum (hrs.)	12.17	28.88							
Total Tardiness (hrs.)	43.04								
Number Of Tardy	6								
	COST OVER TIME (COVERT)								
ORDER	TARDINESS (hrs.)	FLOW TIME (hrs.)							
C1 6.20 21.20									
A5	5.58	15.58							

Table 4: Experimental results Table



B3	10.23	25.23
A6	15.95	27.95
C4	23.37	4.37
B2	10.33	-9.67
Average (hrs.)	7.06	20.61
Maximum (hrs.)	15.95	27.95
Total Tardiness (hrs.)	42.34	
Number Of Tardy	5	

Comment

It was observed that continued use of FISFS resulted in having all jobs tardy. Using EDD also resulted in all jobs being tardy although it reduced maximum tardiness. Adopting COVERT rule resulted in one of the jobs being delivered in time; however it further increased the maximum tardiness. Using SPT in this scenario gave the best option which resulted in both jobs being delivered in time and also reduced the average and maximum tardiness of the jobs. SPT in this scenario again resulted in a significant reduction in flow time.

It can be concluded from these results that no one dispatching rule excels in all criterion, hence the need for a Simulation based Scheduling System to quickly evaluate the system performance under the different criterion and help decision makers to take action in the shortest possible time.

5.2 Conclusion

In this paper, we have developed a simulation based scheduling tool which is aimed to assist production planners to quickly generate optimal production schedules and to quickly react to real time events. Several experiments were carried out and it was concluded that no one scheduling rule excels in all real-time events, hence the need for the simulation-based scheduling tool to help production planners to quickly react to unexpected changes and generate optimal schedules.

REFERENCES

- [1] Jeroen de Jong. (2012), Heuristics in Dynamic Scheduling, A practical framework with a case study in elevator dispatching.
- [2] Cowling, P. I. and Johansson, M., (2002) Using real-time information for effective dynamic scheduling, *European Journal of Operational Research*, 139 (2), 230-244.
- [3] Ouelhadj D., Petrovic S. (2007), Survey of dynamic scheduling in manufacturing systems. Automated Scheduling, Optimization and Planning Research Group, School of Computer Science and IT, University of Nottingham, Nottingham NG8 1BB, UK.
- [4] Vieira, G.E., Herrmann, J.W., and Lin, E., (2003), Rescheduling manufacturing systems: a framework of strategies, policies, and methods, *Journal of Scheduling*, 6(1):35-58; 36-92.
- [5] Clark, Gordon M. 1996. Intoduction to manufacturing applications. In Proceedings of the 1996 Winter Simulation Conference.
- [6] Tamaki, H., Kryssanov, V.V., and Kitamura, S. (1998) A Simulation engine to support production scheduling using genetics-based machine learning.
- [7] Harmonosky, Catherine M. (1995) Simulation-based real-time scheduling: Review of recent developments. In Proceedings of the 1995 Winter Simulation Conference, eds. Christos Alexopoulos, Keebom Kang, William R. Lilegdon, and David Goldsman, 220-225.
- [8] Michelle Leung. (2009), Production Scheduling Optimization of a Plastics Compounding Plant with Quality Constraints, A thesis presented to the University of Waterloo in fulfilment of the thesis requirement for the degree of Master of Applied Science In Chemical Engineering Waterloo, Ontario, Canada
- [9] Coy, S. P., Golden, B. L., Runger, G. C., and Wasil, E. A. (2001), Using experimental design to find effective parameter settings for heuristics. *Journal of Heuristics*, V7 (1):77-97.
- [10] Leus R, Herroelen W. (2005) Project scheduling under uncertainty: Survey and research potentials, European Journal of Operational Research, 165 (2), 289-306.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

THE IMPACT OF OBESITY ON MUSCULOSKELETAL DISORDER IN SOUTH AFRICAN AUTOMOTIVE INDUSTRY

T.M. Mallane¹ and T.B. Tengen^{2,*}

Department of Industrial Engineering and Operations Management, Faculty of Engineering and Technology, Vaal University of Technology, Vanderbijlpark 1900, Private Bag X021, South Africa,

tabitham@vut.ac.za

Department of Industrial Engineering and Operations Management, Faculty of Engineering and Technology, Vaal University of Technology, Vanderbijlpark 1900, Private Bag X021, South Africa,

thomas@vut.ac.za

ABSTRACT

In Automotive industries a number of repetitive tasks (RT) are performed frequently leading to musculoskeletal disorder (MSD). Due to prevalence of obesity and MSD in South Africa, this paper deals with the impact of obesity on MSD in SA automotive industry. Workers were classified into ten subgroups, each of which obesity status (measured in Kg/m²) fell within same Body Mass Index . All workers were within active working age group of 25-28 years. Five workers from each subgroup were chosen and subjected to RT. The average RT was considered as the subgroup value. Regression analysis was used to test the relationship between workers' weights and RT.

It was observed that as workers' weights increased, RT increased reaching a maximum, which then starts decreasing with continuous increase in weights of workers. It was concluded that obesity and underweight have negative impacts on RT or productivity.

Keyword: Automotive industry, Repetitive tasks, Musculoskeletal disorders



Introduction

In today's fast pace and highly competitive world, production processes are rapid (i.e. aimed at reducing cycle time) and Musculoskeletal Disorders (MSDs) are common as a result of repetitive works that are prominent in automotive industry [1].

Vehicle manufacturing is a major industry worldwide, accounting for at least 7 per cent of the manufacturing workforce in countries as diverse as the U.S.A., Canada, Mexico, Brazil, Sweden, Spain, Japan, India, and Australia [2]. In general, this sector is characterized by high capital investment in manufacturing facilities and an organized workforce [2]. As in many other industries, musculoskeletal disorders are the largest type of "injury" in the automotive industry. A number of studies have found that MSDs are associated with the physical demands of jobs in automobile assembly and the manufacturing stages (e.g. for engines, upholstery, and so forth) that precede final vehicle assembly [3].

Lacks of attentions to workplace design and work-method design have unfavourable consequences to workers' health and safety, including musculoskeletal disorders, psychological distress, just to name a few [4]. In addition to the resulting human pain and suffering, which cannot be quantified monetarily, there are also increased costs to employers through workers' compensation claims, lost productivity, scrap and decreased production quality, medical insurance premiums, and ultimately the sustainability of a company or even of an entire sector. It has been estimated that the direct and indirect economic burden of musculoskeletal disease in the United States, including arthritis and repetitive trauma disorders, totalled \$149 billion in 1992 [4]. However, within individual firms, typically large proportions of these costs often are not specifically identified by traditional accounting methods as they are due to ergonomic problems in the work process. There is evidence that the frequency of work-related musculoskeletal disorders (MSDs) is severely underestimated when relying on traditional administrative data sources such as workers' compensation records and OSHA logs of recordable injury and illness [5].

The current paper investigates on the Impact of obesity on MSDs in automotive industry in S.A. The work was carried out at a local car maintenance or repair workshop. The study involved observing operators while they lift and fit the gearbox. The back and shoulder disorders were recorded by counting the number of lift-and-fit processes that were performed before the operators complained of tiredness or could be seen stressing themselves. The observation lasted for a period of three months.

Method

Body Mass Index (BMI)

Body mass (kg) (measured with an electronic scale [5]) and the body height (m) (measured with metric rule [5]) were used to calculate BMI (BMI (kg/m²) given by body mass (kg) divided by the squared of body height (m^2) .

Participants

Workers from automotive manufacturing based in Pretoria participated in this study and were divided into ten groups. The participants were workers whose jobs required manual force and awkward posture as repetitive operations. All the workers were within an active working age group of 25-28 years. They were divided into the subgroup according to their BMI values and were subjected to repetitive tasks (i.e. assembly of gear boxes). The BMI Values ranged from 17-to-33 Kg/m² with an interval of 1.5Kg/m².

Data collection and analysis

MSD is understood as a sign of abnormal condition (abnormality) experienced by employees, typically expressed in the form of body pain or other related illnesses. MSD, which is a form of physiological disorders typically causes change in efficiency of the workers. Since time is used by Industrial Engineers as one of the parameters of measuring efficiency, cycle time was used to quantify efficiency in this report. As such the measure of MSD was cycle time which was quantified using stop watches. The onset of MSD was identified by the fact that a task cycle time started becoming longer than (i.e. changing from) those of the previous ones that were constant.

While onsite data collection process was performed, others sources of information were consulted such as worker's medical records. Data was collected from a database on the medical records of the company's clinic for acute and chronic musculoskeletal disorders. Information about symptoms and respective working environments were obtained by employing work-study techniques. The two work-study techniques (method study and work measurement) were used to systematically record and examine the manner in which the work was done necessary improvements [6]. Before any recording was made, according to these techniques, a series of observations were conducted to satisfy the choice of job selection for study and application of techniques. The workers (or patient) were then referred to the medical practitioner for medical examination.

Regression analysis technique

Regression analysis was used to establish the relationship between the two variables, which are the workers' weights and the number of repetitive tasks that a worker could bear before the MSD was noticed per shift (to avoid influence of resting) for workers from each BMI subgroup. Two sets of data were obtained, which can be likened to the use of normal working shift and over time. The first set of data (that can be used to establish a normal shift workload for workers of the specific weight group) was when the workers could easily recover



from MSD through simple resting after shift. The second set (see Fig.1), which could be used to study the impact of over-times, was when the workers were asked to do further jobs after they started experiencing the MSD, and the maximum number of repetitive tasks that they could perform with no further ability to perform more tasks were recorded. *Since most of the workers were around the same average height, the weights of the workers were used* during analysis of results as depicted in Figure 1. Subsequently, the cost implication of MSD to the company was investigated and is presented in Figure 2. The cost implication was obtained by getting unit product done cost multiplied by the number of repetitive tasks and the cost incurred due to MSD or over-time.

Results and Discussion

It can be observed from Figure 1 that as the weights or the workers were increasing the number of repetitive tasks that they could perform before MSD were increasing. The maximum number of repetitive tasks was measured for workers whose weights were around 70Kg. The number of repetitive tasks started decreasing with increasing weights (obese) workers (see Fig.1).



Figure 1: The relationship between tasks and weight

The cost implication, as depicted in Fig.2, indicates that as the number of repetitive tasks increases the cost of job also increases linearly until the onset of MSD were the cost start increasing exponential with additional repetitive tasks. It can then be seen from this Fig .2 that underweight and overweight (obese) workers will cost the company more as the plots of their costs are above those of normal weight workers who have weights of 70Kg.



Figure 2: The cost implications of MSD and repetitive tasks

The prevalence of repetitive tasks in automotive industries escalates the risk of musculoskeletal disorders on employees and would increase in the long run due to cumulative exposure [7]. The results of this study (Fig. 1) indicates that those employees with a normal weight can perform tasks assigned to them for a standard shift until excess overtime tasks with minimal amount of fatigue, and those with low or high BMI level experienced deteriorating level of operations leading to more MSDs, ultimately increases the costs to employers.

In performing tasks, generally there is a standard time set for an operation, and repetitive tasks are no exclusion to this. Workers must follow a predetermined pace [8]. With a higher work pace, the levels of muscle activities



are also higher, which contribute to muscle fatigue and MSD risks [6]. Muscle fatigue is a stage, which the muscle is not able to sustain the required force or work output level [3]. Accordingly, figure 1 indicates that a general increase in the number of repetitive tasks was associated with the decrease or reduction of employee performance for those with weights between 30kg - 45kg and 95kg-120kg; and sustaining the active working weight of between 50kg and 90kg should be the better option.

Most industries prefer over-time as opposed to hiring more staff member because the cost-to-company when a company employs more employees is higher due to financial reasons [9]. The impact of overtime has been observed to be problematic to all employees particularly to those underweight and overweight persons/employees as classified by BMI [10]. A person whose weight is between 30kg - 45kg and 95kg - 120kg, when subjected to more repetitive tasks, would require longer resting break to return to their initial state of muscle function. This would cost the company as the productivity rate would reduce.

Conclusion

The results of this study showed that both overweight (i.e. obesity) and underweight have negative impacts on number of repetitive tasks or productivity [11]. It was observed that as workers' weights increased, RT increased reaching a maximum, which then starts decreasing with continuous increase in weights of workers. It was concluded that obesity and underweight have negative impacts on RT or productivity.

Therefore, the employee's health and safety were affected and that reduced their performances and subsequently affected the company productivity. This research ignites the desire to know how lack of performance resulting from MSD affects the employer with regards to employee's productivity

References

1. Kumar, S. (2001). Theories of musculoskeletal injury causation. *Ergonomics*, 44(1), 17-47.

2. Devereux, J., Rydstedt, L., Kelly, V., Veston, P., Buckle, P., 2004. The role of work stress and

psychological factors in the development of musculoskeletal disorders. 273 pg 1-139. 3. Ma, L., Chablat, D., & Zhang, W. (2009). Dynamic Muscle Fatigue Evaluation in Virtual Working

Environment. Ergonomics, 1(January), 211-220.

4. Waters, T. R. (2004). National efforts to identify research issues related to prevention of work-related musculoskeletal disorders. Journal of electromyography and kinesiology: official journal of the International Society of Electrophysiological Kinesiology, *14*(1), 7-12.

International Society of Electrophysiological Kinesiology, 14(1), 7-12.
Global database on Body Masa Index. World Health Organization 2006. Retrieved July, 27, 2012.
Okorodudu, D.O. M. F. Jumean, V. M. Montori et al., "Diagnostic performance of body mass index to identify obesity as deŐned by body adiposity: a systematic review and meta-analysis," International Journal of Obesity, vol. 34, no. 5, pp. 791-799, 2010.

7. Nordander, C., Ohlsson, K., Akesson, I., Arvidsson, I., Balogh, I., Hansson, G.-A. ... Skerfving, S. (2009). Risk of musculoskeletal disorders among females and males in repetitive/constrained work. *Ergonomics*, 52(10), 1226-39.

8. Chiasson, M., Imbeau, D., Aubrey k., Delisle., 2012. Industrial Ergonomics: Comparing the results of eight methods used to evaluate risk factors associated with Musculoskeletal Disorder. International Journal of Industrial Ergonomics, 42.pg 478-488.

9. Guanyan, L.I., 2010. Current techniques for assessing physical exposure to work - related musculoskeletal risks, with emphasis on posture - based methods.

10. Poston, W. S. C., Haddock, C. K. Jahnke, S. A Jitnarin, N. Tuley, B. C and. Kales, S.N. " prevalence of overweight, obesity, and substandard Tones in a population-based cohort," Journal of Occupational and Environmental Medicine, vol. 53, no. 3, pp. 266-273, 2011.

11. Xu, Z., Ko, J., Cochran, D. J., & Jung, M. (2012). Design of assembly lines with the concurrent consideration of productivity and upper extremity musculoskeletal disorders using linear models. Computers & Industrial Engineering, 62(2), 431-441.



PHYSICAL ASSET MANAGEMENT MATURITY IN MINING: A CASE STUDY

B.J. Mona¹, & B.P. Sunjka¹

¹School of Mechanical, Industrial and Aeronautical Engineering University of the Witwatersrand, Johannesburg, South Africa. <u>bj.mona@mabhelengwane.co.za; bernadette.sunjka@wits.ac.za</u>

ABSTRACT

Successful implementation and execution of asset management strategy is found to be a critical element in driving value, which depends on physical assets' performance. This paper shows that strategic asset management targets measured as AM Maturity are often not attained. This study identifies key issues that prevent mining organisations from attaining the desired level of Asset Management Maturity. A qualitative methodology was utilised on a single case study design in order to investigate a particular phenomenon which is Asset Management Maturity at "A-Coal" site. The study revealed the following key issues: lack of strong Asset Management leadership; lack of a favourable organisational structure which will provide Asset Managers with credibility and the authority to make decisions in support of organisational objectives; no leadership support; and a lack of training limits the benefits that can be realised from improvement initiatives which motivated, committed and enthusiastic employees will be delivering. In addition to these challenges are organisational culture and a certain level of employees' competitiveness in the field of asset management.

Key words: Asset Management (AM), AM Maturity



1 INTRODUCTION

Asset Management is defined as the coordinated activity of an organisation to realise value from assets, where an asset is defined as an item, thing or entity that has potential or actual value to an organisation [14]. Societies are faced with massive challenges of managing assets in their endeavour to identify the lowest cost or high return on investments; understanding the lifecycle costs, as well as extending the life of these assets. If these assets are not managed effectively it will be a wasteful exercise. AM is important especially for asset intensive organisations, like mining, because it can assist to improve financial performance; to make informed asset investment decisions; to improve services and outputs; where some benefits can be directly quantified such as reducing capital and maintenance costs [14]. Additional benefits [6] are the potential reduction of the health impacts and the safety risks as well as minimising the environmental impact of operating assets. Managing assets assists mining organisations to reduce legal risks; improving the regulatory performance; thus maintaining and improving the reputation of the organisation.

Asset Management Maturity is a measure of how well an organisation is capable of optimising and sustaining the performance, risk, investment and costs related to their assets [11]. The global forum on maintenance and asset management GFMAM [8] defines AM maturity as the extent to which the capabilities, performance and ongoing assurance of an organisation are fit for purpose to meet the current and future needs of its stakeholders, including the ability of an organisation to foresee and respond to its operating context.

The purpose of this study is to identify key challenges experienced in a mining organisation in South Africa inhibiting it from achieving the desired level of Asset Management Maturity. A qualitative case study method was chosen. Three data sources of evidence were used to obtain information, these included; archived documents, semi-structured interviews and participant observation. Data was analysed using content analysis.

2 LITERATURE REVIEW

The purpose of this chapter is to, outline the fundamental concepts in asset management including issues associated with implementation and execution; and to, uncover the key challenges experienced in mining in relation to physical asset management maturity. A typical AM framework to measure Asset Management maturity will be presented and compared with other AM Maturity frameworks. This chapter is correspondingly used as a foundation for the development of the questionnaire in the next chapter.

2.1 Asset

An asset is an item, thing or entity that has potential or actual value to an organization. The value will vary between different organizations and their stakeholders, and can be tangible or intangible, financial or non-financial [14]. Any item of economic value owned by an individual or corporation is defined as an asset [6]. PAS55 [22] defines an asset as a plant, machinery, property, buildings, vehicles, and other items that have a distinct value to the organisation.

2.2 Asset Management

The activities of asset management start when there is an existing tangible asset and it is being operated [12]. There are four key areas to be developed to improve AM practise namely [5]:

Effective decision making - Improving decision making across the organisation, through better use of longer term financial, and non-financial, metrics to deliver value for all involved in managing assets;

Organisational changes - The organisation must evolve to enable better decision making and share knowledge and skills, breaking down silos and boundaries resulting from functional specialism and multiple cost centres; Data capture, sharing and standards - Improving the quality and availability of the information available for

decision making and Predictive Analytics - New information technologies are available to improve AM, but several barriers prevent

Predictive Analytics - New information technologies are available to improve AM, but several barriers prevent their effective use.

AM is best seen as an integration framework that uses any combination of tools and techniques to achieve these aims, provided that they are appropriately applied and deliver added value [11]. AM embodies the important principle of proportionality or fitness-for-purpose, so any objective definitions of capability and Maturity must recognise context and are appropriate, possible and worthwhile in such an environment [29]. The purpose of the Physical Asset Management function is to provide resources and expertise to support the acquisition, inservice support and disposal of the physical assets required by the organisation. More precisely, the asset management function, when present at Company level, should provide inputs to asset planning, take a role in major acquisitions and developments and provide the systems and facilities needed to support assets throughout their life [9]. An asset management system provides a structured approach for the development, coordination and control of activities undertaken on assets by the organisation over the different life cycle stages, and for aligning these activities with its organisational objectives [14].



2.3 Asset Management and Maintenance Management

Maintenance management focuses more on the short-term activities, where the primary purpose is to determine which activities are performed on which asset, including the cost of maintenance. Asset management, however, is intended as a proactive approach to managing organisational investments over the longer term. Maintenance management should be imagined as a subset of asset management [25] Asset Management achieves the realisation of AM which has evolved from maintenance management to provide a holistic approach to managing the life of physical assets [20]. The terms "maintenance management" and "asset management" are frequently used interchangeably in mining industry when, in fact, they should be viewed separately [25].

2.4 Asset Management (AM) Maturity

AM maturity can also be defined as an extent to which the capabilities, performance and ongoing assurance of an organisation are fit for purpose to meet the current and future needs of its stakeholders, including the ability of an organisation to foresee and respond to its operating context [8]. It refers to the organisation level of Asset Management practice [25]. AM Maturity is also defined as a measure of how well an organisation is capable of optimising and sustaining the performance, risk, investment and costs related to their assets [11]. The importance of measuring, knowing and understanding Asset Management Maturity practices aids organisations to optimise their business performance, through reduction of risks, higher returns on assets (ROA) invested in and reduced costs [11].

Asset Management Maturity knowledge helps asset owners to understand if they are really getting value from the asset management systems put in place in ensuring assets perform as expected, reliably and safely [12]. Increasing asset management maturity may deliver the following benefits [8]: Aligning individual activities and behaviours that achieve the organisation's objective; improve both financial and non-financial results; provide clarity about the goals and direction of the organisation; optimise lifecycle cost; aligning risk with stakeholders' risk appetite; and enables benchmarking, even between organisations managing different assets in different operating environments. There are three important questions that a company should be asking with regards to AM Maturity [21]:

- 1. Where are they? Mining organisations should establish a baseline level that indicates their current assets, how well they are being taken care of and how these assets contribute to supporting service delivery.
- 2. Where do they want to be? What are the mine's plans for the future and what assets are required in order to support current and future service needs? Having this in mind the organisation will set targets for improving the condition and performance of assets so that they match future requirements or needs.
- 3. How do they get there? How will the organisation move from the current to desired level of AM Maturity? The prioritisation of future investments in assets will be important to match their future service needs.

2.5 AM Maturity Models

An Asset Management Maturity model can be viewed as a set of structured levels that describe how well different processes of an organisation are able to produce the required outcomes in a reliable and sustainable way [25]. There exist similarities between AM maturity frameworks because most of them were developed based on PAS 55, the first Institute of Asset Management (IAM) maturity scale to be published. The AM framework used for this research has level 1 to level 5, starting from Fire-Fighting, Stabilising, Preventing, Optimising and Excellence [3]. An overview of AM Maturity levels definition is described below [25]:

Level 1 - an organisation has a clear Asset Management vision. This includes a policy statement that provides top-down direction regarding Asset Management expectations, a strategy that summarises the methodology for completing the policy, and a plan that details the people, activities, and resources needed for addressing the policy and strategy.

Level 2 - an organisation has one or more asset inventories with condition data that support multiple business processes. There is a well-defined owner and processes for all this data to maintain its reliability.

Level 3 - The organisation can conduct a risk analysis or a performance assessment to evaluate the assets' current performance and to assess how well the organisation is doing against the policy and strategy objectives.

Level 4 - At this level, an organisation can set priorities amongst and throughout all asset classes based on risk and performance data available. This is important in the development of the capital program, operations and maintenance budget.

Level 5 - Herein an organisation can apply performance modelling and other analytical tools to optimise how funding is allocated to all the organisation's asset classes.



2.6 **Key AM Implementation Steps**

Implementing the changes suggested to become a mature asset management organisation requires a well thought out planning and execution strategy. Organisations that are highly productive, they typically place more emphasis on executing strategy than formulating it [16]. These are the principles for implementing AM [25]:

- Understand the organisations asset management drivers Organisations undertake an AM improvement program for various reasons. The organisation should develop an implementation approach that maintains a focus; however, the approach should be flexible enough that it can shift focus as priorities change.
- Build upon existing strengths and practices The organisation should leverage its departments' existing asset management activities, identifying best practices and lessons learned with one asset class and applying these practices and lessons to others.
- Provide value immediately Through incremental implementation activities, an organisation can quickly achieve results that demonstrate the value of implementing improvements to asset management practice and provide momentum for future activities.
- Recognise that asset management is a process Identify the core processes that provide a starting point for developing an asset management process that will pay dividends in improved service delivery and asset sustainability.
- Prioritise people, tools, and information Asset management is, at its core, about data-driven management, so your managers should identify the people who can understand and lead this change initiative and establish the data and develop tools that best support the organisation's decisionmaking processes.
- Invest smartly Managers should identify the investments that will provide the best "bang for the buck" and only if these investments support the organisational strategy.
- Develop your human resources Managers should identify the appropriate skillsets needed to implement the asset management strategy and invest in those people with recognition, incentives, and training.
- Provide top-down leadership and assign clear ownership for asset management activities Strong leadership will set expectations and accountability for implementation, while your asset owners should "own" and drive implementation by developing and implementing lifecycle management plans.

2.7 AM Challenges and Associated Risk

Some of the institutional and technical challenges faced within organisations are: to secure senior management support and leadership throughout the period of asset management implementation, which may extend over several years and to improve the life cycle analysis methods and incorporate them fully into planning and programme development [19]. Repenning & Sterman [18] suggests that the inability of most organisations to gain the full benefits of the implementation of an innovation lie in the manner in which this improvement programme has been introduced to the Company and how the process reflects the whole system. One of the main practical challenges to the improvement of AM practices in organisations is the disconnect between tactical and implementation groups [33]. Management Involvement and commitment are the most essential fundamentals in adding any of the desired productivity improvement initiatives [1], [4], [10]. Since strong leadership permeates a vision and strategy for generating, while permitting a flexible organisational structure. Good leadership ultimately encourages effective skills and knowledge enhancement amongst its workforce [27]. Maintenance is technically considered as a "necessary evil" and not pertinent at the strategic level as a business issue [7].

High performing organisations are those with a culture of sustainable and proactive improvement [27]. The Asset Management Council of Australia (AMC) defines Asset Management Culture as: "the extent to which all levels of the organization have the knowledge, skills and commitment to achieve the documented asset management goals of the organization". Organisational capabilities are critical for addressing the cultural challenges within the organisation [26]. Most people think that organisational culture is the biggest single obstacle to improving the asset performance [21]. This requires the training of people to utilise the programme, financial resources that organisations are often not willing to invest. Risk management is the identification, assessment, and prioritisation of risks followed by coordinated and economical application of resources to minimise, monitor, and control the probability and/or impact of unfortunate events or to maximise the realisation of opportunities [13]. There are other risks involved which if not addressed properly will inhibit AM growth, these risks contribute to an organisation's failure to manage its assets optimally. These risks are overly exposed for example [17]:

- 1. If asset owners don't know what assets they have;
- If they are under or over maintaining their assets;
 If operators are not properly operating their assets;
- 4. The asset risk at hand with regards to assets is not managed properly
- 5. And Sub-optimised asset management systems are in place.



3 RESEARCH METHODOLOGY

The primary approach utilises a Case Study design as the research investigates a particular phenomenon, Asset Management Maturity, in a particular context, the case company, "A-Coal". The essence of a case study is the tendency among all types of case study as it tries to illuminate a decision or set of decisions: why they were taken, how they were implemented and with what result [30]. "A-Coal" is a coal washing plant situated around Ogies in Mpumalanga a small town outside eMalahleni. It is a 50/50 percentile joint venture of "Client A" and "Client B". The plant is designed to beneficiate 16 million tons of ROM coal per annum at 2400 tons/hour. This study sought to establish the key challenges inhibiting the achievement of the desired levels of AM Maturity in A-Coal by:

- establishing the importance of stakeholder's involvement when it comes to executing asset management activities and the improvement of AM Maturity.
- assessing employees' attitudes, culture, perceptions and understanding of AM and its associated benefits for properly executing Asset Management Processes.
- analysing A-Coal employees' understanding of Change Management, Asset Management and Business Risk Management.

A qualitative approach was chosen because of direct and personal contact with the employees at the A-Coal site in the natural setting of the phenomenon (AM implementation and execution). A summary of data source used is detailed below:

- Documentary evidence in the form of information from A-Coal when the initial Maturity assessment was conducted, implementation took place, and all the other annual assessments were captured with findings. The initial strategy, policy, and master plan were also captured.
- Semi-structured interviews with thirteen interviewees were conducted at the case site (excluding the pre-test interviews). The interview sessions duration ranged from 15 to 45 minutes. A set of interview guides with a list of questions were crafted in such a way that they are mostly the same for all interviewees; however, some were restructured differently as follow-up questions emerged during the sessions. The profile of the interview participants was from senior, middle, and lower management, as well as engineers and planners. A set of interview guides that summarised a list of questions following Wengraf's Pyramid Model [28].
- Participant Observations: The researcher worked for over three years at case site as an Asset Care Centre Manager offering asset management services, and has been involved in meetings, when certain initiatives were implemented, but only observed and documented some of the things that he was exposed to but also of interest to the study. The data collected through this method also contributed as a trigger to this study.

Content analysis was used to analyse the data. This method relies on the subjective interpretation of the content of text data through the systematic classification process of coding and identifying themes or patterns" [23]. The process is designed to reduce collected data into categories or themes based on valid inference and interpretation [24]. Analysing data from three exclusive sources ensured data quality, integrity, reliability and validity of the research.

4 RESULTS AND DISCUSSION

This chapter presents the processed data from the documents, interviews and participant observations. The current state of AM and AM maturity is first outlined leading to the initiation of this study. This largely relies on documentary data sources. Subsequently, the data from the interviews and participant observations is analysed with the objective of answering the research question for this study. The strategy data received is qualitative by its very nature, and was examined by the researcher and any change in the organisational strategy, mission and goals, which can be triggered by internal or external influences mainly from the shareholders (Client A and B as referred to in the interviews transcribed works). Data extracted from the transcribed works contained both but mostly qualitative than quantitative data. As for Asset Management Maturity, this is represented as a figure or can be expressed by describing the level of Maturity (such "fire-fighting, or excellence").

4.1 Current State-Analysis

The following describes the state of AM in A-Coal prior to the start of this study. In this phase of analysis, the focus was on analysing data received from archived documentation of the implemented AM systems at A-Coal site. This data provided a holistic understanding of the implemented system detailing the financials, methodology, framework used, assessment results conducted, developed AM Strategies, AM Policy, setting up AM Objectives and the AM Master-plan, with AM activities that were set to be actioned during the initial year of implementation, based on the AM gaps identified. The establishment of technical and steering committees to drive AM activities occurred during this period. The objective of this analysis was to provide a brief background of A-Coal site's context, AM capabilities and a description of the AM implementation leading to this research study.



Company B, the AM consulting company which was employed to assist Company A (asset service providers) who operate and maintain the plant and the implementation and execution of AM. Company B has developed AM Maturity framework which is aligned with the newly adopted ISO55000 and with GFMAM 39 Subjects. The model used is called Asset Management Improvement Planning (AMIP) and has grouped the Key Performance Areas into 17KPAs. Company B assessed A-Coal AM maturity in relation to best practices and the associated Key Performance Indicators (KPIs). Table 1 below illustrates the initial AMIP assessment results that were obtained against the target set for the organization to be achieved within a 12 months' period.

	BP So	ores	KPI S	cores		
	BP Target	BP Maturity	KPI Target	KPI Maturity		
Strategy Management	4	1,6	4	4		
Information Management	4	2,5	4	2,8		
Technical Information	3	2	3	2		
Organisation and Development	3	2	3	3,8		
Contractor Management	3	2,4	3	2		
Financial Management	3	3,1	3	3		
Risk Management	3	2,3	3	3		
Environment, Health and Safety	3	3,3	3	2,5		
Asset Care Plans	3	1,5	3	2,7		
Work Planning and Control	3	1,9	3	2		
Operator Asset Care	3	2,4	3	2,5		
Materials Management	3	2,2	3	2,4		
Support Facilities and Tools	3	2,9	3	3,3		
Life Cycle Management	4	2,7	4	4		
Project and Shutdown Management	3	2	3	3		
Performance Measurement	3	2	3	1		
Focused Improvement	3	1,9	3	3		
Average	3,18	2,28	3,18	2,76		
	BP Target Average	BP Maturity Average	KPI Target Average	KPI Maturity Average		

Table 1: A-Coal initial AMIP	assessment results and set i	target [3]	Ľ
	assessment results and see		

The average Best Practice (BP) obtained was at 2.28 of a possible 5 but the worrying thing was that the key maintenance areas of asset management were scoring the lowest (such as Strategy Management, Asset Care Plans). The desired average AM Maturity BP level was set at 3.17, with only three KPA targets set at level 4 and the rest of the BP KPAs targeted for level 3 for the next 12 months. Some of the supporting maintenance processes (finance, safety,) were well established and working well. However, a lot of asset history was lost in the initial phase, although a basic Enterprise Asset Management System (EAMS) was in place with standards and business processes documented. There were basic asset care plans "asset maintenance strategies" which also needed to be improved. There was also a basic material manager function under control [3]. On the other hand, Company A did not have a consolidated AM Policy or Strategy and a platform to discuss AM issues and goals as part of the shareholders' operations review. The EAMS systems implemented was not configured as per documented standards resulting in under performance or sub-utilising the system. With the shortage of staff complement it was a challenge for the operation to escape the fire-fighting mode of operation. The planning department needed to be enhanced from a one-man operation in capacity and capability [3].

There were limited asset management KPIs available compared to production related KPIs which were readily available. These KPIs were not broadly communicated amongst all the employees on site. Some optimisation that needed to occur, was at the stores department where stock levels and critical spares needed to be identified. Only a very limited structured improvements process was in place. This included the use of routine failure analysis on all breakdowns experienced. There was very little control in the process of updating asset manuals and drawings [3].

At the start of the implementation phase the AM Steering Committee and AM Technical Committee were established to drive and coordinate all asset management activities. The stakeholders of the meetings also took responsibility for change management during the implementation of the ACC [3]. The members of both the technical and steering committees were trained in the overview of asset management to create a knowledgeable forum where all major decisions regarding asset management can be made. Company A employees were trained in generic ACC principles and practices in preparation for the execution phase [3].

4.2 Semi-Structured Interviews

The semi-structured interviews were used to explore any problems and successes related to the implementation and execution of asset management. They were also used to gain understanding of some of the data obtained from the documentation with regards to the current execution of AM. The most predominant themes which emerged during the analysis of the interviews are:



- 1. Reactive mode of operation;
- 2. Asset Management Understanding;
- 3. Stakeholders' involvement;
- 4. People's training and development;
- 5. Organisational Culture;
- 6. Skills Retention and Skills Transfer.

4.2.1 Reactive mode of operation

The Asset Management programme was not implemented during the commissioning phase of the A-Coal [3]. It was discovered that a call for AM implementation was made because investors realised that the plant was operating in a fire-fighting mode. They commissioned A-Coal without a developed AM strategy, without adequate documented proper maintenance strategies. This, in turn, indicates a reactive mode of operation because such documentation should have been implemented before the plant was operated considering that it is a new plant. It is considered a reactive mode of operation for a plant to be commissioned without an appointed Engineering manager. The asset management department consisted of one person who was not capable to execute the AM activities as set by Brown and Humphrey [3].

4.2.2 People's training and development

Leadership and management play a vital role in driving change in an organisation. For successful execution of AM activities Company B required strong support from Company A and both the shareholders Client A & Client B. Senior management must play their role in ensuring that all affected by the implementation of AM, are sensitised and well trained for the success of the implementation, execution and sustainability of AM. Company A's employees were trained in generic ACC principles and practices in preparation for the execution phase [3]. However, this will not bring about a total understanding of what AM is to these employees, simply because employees were going through the change process and having to deal with the pressure of being in a fire-fighting mode and the new changes brought in. In addition to that was a mental shift on the way they used to execute their AM activities.

A few noteworthy points can be highlighted as follows; Client A and Client B made a call to have AM implemented two years later after commissioning and this could have been done right at the start of the project. The operations and maintenance staff focused on equipment breakdown rather than on preventative maintenance. The responsible engineering manager was appointed in 2011 which should have been right at the beginning or commissioning phase of A-Coal. A feasibility study should have been conducted at the beginning of the project to understand the skillset required and capacity for driving asset management for an organisation of A-Coal size and nature than to appoint only one resource to drive AM.

AM management should not be driven by a one-man-band show; an organisation should not find itself in a state whereby AM is driven by an individual. Generally, interest is created through exposure. It was found that the majority of the participants interviewed had never received formal AM training. Training must be used as a platform to create interest in the subject in question and as such employees will understand better why they have to do what they opt to do. Training also stimulates people's minds to be more proactive than reactive as it creates an inquisitive mind. Another subliminal finding is the loss of skill, including tacit knowledge. It is a fact that people change position for various reasons. And tacit knowledge is not easy to transfer. These findings can be considered to augment and agree with other challenges mentioned in the literature review [7], [1], [34], [25].

4.2.3 Asset Management Understanding

An important phenomenon occurred when different views and understanding of what asset management from the participants was observed. Majority of the interviewees' understanding of asset management is that an asset needs to be fully utilised, or maximising value from that asset, whereas some of the participants explained asset management as maintenance, tracking and measuring performance. One out of the thirteen participants explained asset management as coordinating activities to optimise asset life cycle cost. Having different views on what asset management is, is not a good start where AM Maturity is concerned and where team work is essential to improve asset management capability. Senior, Middle and lower management must align in working towards improving AM capabilities. This can be seen as a concern where management needs to have a clear vision of why they are driving asset management.

4.2.4 Stakeholder's Involvement

About 75 per cent of the respondents indicated that they are aware of the Steering Committee Meetings (SCM), and 16 per cent of the 13 are attending the SCM. This can be seen as a concern, which hampers the success and improvement of asset management's capabilities when there is no leadership or a lack of leadership's involvement in steering committee meetings. Between 60 per cent and 80 per cent of all companies fall short of the strategic targets that they have set [15]. It is highly desirable to have a certain degree of leadership engagement throughout the Company, long-term focus of management and the strategic team while executing AM activities. In many organisations, focusing the business practices around improved asset management can



require a change in the way the Company does business. This means leadership needs to provide a clear direction regarding people's responsibilities. They must also provide suitable accountability mechanisms, develop and follow a structured communications strategy. The steering committee (SCM) and technical committee meetings (TCM) are the platforms that enable such activities to take place

If lower and Middle Management believes that Senior Management is not leading them in the right direction of AM excellence, they will lose trust and commitment to where matters of Asset Management are concerned. And most importantly lose confidence of Senior Management to leading them in AM. A very strong AM leadership is required to drive AM capabilities or Maturity to excellence. An organisation with low Maturity increases the organisation's business risk, because when a Company is matured it will have a strong competitive advantage over its rivals. If Asset Management is not executed properly, the three functions; cost, risk and performance which AM aims to balance will be compromised as such exposing the business risk to be higher.

Changing AM leadership exposes an organisation to high risks for not accomplishing AM activities set to improve AM Maturity. A-Coal experienced this challenge as key asset management position were affected due to staff turnover from the three parties involved. The four key positions that were affected are the following:

- The Plant manager (Company A)
- The Engineering manager (Client A)
- The Plant Engineer (Company A)
- Asset Care Engineer (Company B)

Looking at the above staff turnover picture of asset management the leaders within A-Coal fail to present a favourable picture for AM capability improvement. It helps to have a minimal staff turnover within an organisation in order to maintain the level of commitment and execution to AM activities. Woodhouse [29], mentioned that managers often rotate or move every 2-3 years and it can be as worse as 6 months. This leads to employees getting changing messages as AM leaders are changing. It was found that at A-Coal most of the senior managers who were initially responsible to drive asset management and were involved during the implementation of AM were no longer working for A-Coal for various reasons during the period when this research was conducted. It is without doubt that such experience will result in plummeting the enthusiasm of driving AM activities which will hinder AM maturity growth.

None of the senior management and middle management had received formal AM training, such as an overview of asset management, ISO 55000, GFMAM 39 Subjects. It is rather surprising that an organisation can strive to attain a certain level of AM maturity with a leadership that is not fully abreast on current matters of AM. To understand Asset Management one needs to have the love for it and must understand the benefits that can be reaped if AM is executed properly. But having the passion only, will not bring about changes and improvement in the AM Maturity of an organisation. Passion combined with AM understanding and knowing the benefits associated if AM is executed properly are the ingredients to AM Maturity improvements and success and of course implementing the required actions.

Prior to AM implementation Company A had standards developed for the use of EAMS, without an AM Strategy, AM Policy or AM Objectives. During implementation these documents were developed and then became static documents rather than dynamic. These documents had never been updated from the time they were developed. The documents are supposed to be dynamic to align with organisational objectives, which supposedly gets updated at least on a yearly basis [3]. Strong AM leadership which understands how AM contributes to an organizations competitive strategy and how to translate an organization's objectives to AM activities is a must for the success of AM activities' execution and the improvement of AM Maturity thereof [8].

4.2.5 Organisational Culture

When establishing an organisation's asset management system, consideration should be taken on internal and external context. As ISO 55000 states that the internal context includes the organisational cultures and the environment as well as the mission, vision and values of the organisation. Stakeholders' inputs, concerns and expectations are also part of the context of the organisation [14]. Stakeholders' influence is key to setting rules for consistent decision making and also to contribute in the formulation of organisational objectives.

Organisational structure plays an important role in the execution of AM activities and success thereof. Structures shape the way in which assets are managed across the business and for the whole life cycle of the assets. A silo mentality obstructs the efficiency and effectiveness of AM as compared to team work. Key to asset management success is gaining the commitment from top management to drive change in organisational



culture.

Leadership needs to improve their understanding of how good asset management contributes to achieving organisational objectives. Once leadership understand this, they need to encourage it throughout the life of the plant and across the organisation by encouraging system thinking and working towards eliminating organisational limitations to improve AM practices. An improvement in AM Maturity requires an organisational culture shift from reactive to a proactive mode with a culture of proactive improvements efforts. Employees must also have pride in what they do, especially if they understand the benefits of AM. It is important that all employees understand the importance of AM and their relation with business risk. There is no distinct link between AM maturity growth and business risk. However, an improvement of how well organisation is able to carry out AM activities indirectly influences the risk that the business is exposed

The results obtained from the study indicate that there is some alignment with what was found in the literature as Markow & Rackosy [19] stated that some of the technical and institutional challenges faced within organisation is securing senior management support and leadership throughout AM implementation. Management needs to supports change and execution of AM to realise improvement in AM maturity. A culture where all the levels in the organisation have the knowledge, skills and commitment will help achieve asset management goals. If the organisational culture is not supportive and personnel don't have pride in taking care of the asset they are looking after AM execution will fail. Without senior management support and commitment attempts of implementing AM with not succeed. Lack of senior management buy-in can also mean that AM training will not be allocated any budgets or minimal at best. It was found that at A-Coal most of the senior managers who were initially responsible for driving asset management and were involved during the implementation of AM were no longer working for A-Coal. In essence, these inconsistencies and lack of AM continuity with new management will hinder AM maturity growth.

4.3 Summary of Findings

The study aimed to identify challenges faced by mining organisations for not attaining their desired level of Maturity, herein is a list of findings compiled after an analysis was made from the various sources used.

- 1. The current AM Organisational structure has limitations to the improvement of AM maturity growth. Asset managers are currently reporting to Asset service providers instead of asset service providers to repot to asset managers [2].
- 2. Lack of alignment between organisational objectives and operational activities. (AM strategy has never been updated since it was developed to align with business objectives. Therefore, poor communication of the Company objectives to translate into AM activities) [3].
- 3. Company A management supports Technical Committee meetings. Through lack of management backing they don't fully support Steering Committee meetings.
- 4. There is a general lack of knowledge and understanding of Asset Management principles from management.
- 5. Lack of systems which resulted in key man/woman dependencies syndrome. (AM initiatives are driven by individuals and skills retention).
- 6. Poor management support to drive AM capabilities of employees. Employees have training matrices which focus on technical skills development.
- 7. Resistance from people to do the same things in a different way, as AM was introduced.

5 CONCLUSION

The identified key challenges have been noted not only to have surfaced from the execution stage, but also what happens during the implementation of Asset Management plays a massive role. It has been cited in this study that there are many challenges and issues faced with the implementation of Asset Management, little has been written with regards to challenges faced during the execution phase of AM. This study maintains that the key issues identified prevent mining organisations from attaining the desired level of Asset Management Maturity which is; lack of a strong Asset Management leadership; A favourable organisational structure which will provide Asset Managers with credibility and the authority to make decisions in support of organisational objectives; No leadership support; and a lack of training limits the benefits that can be realised from improvement initiatives which motivated, committed and enthusiastic employees will be delivering.

In addition to these challenges are organisational culture and a certain level of employees' competitiveness in the field of asset management. It is noted that there is an indirect correlation between Asset Management Maturity and Business Risk. An organisation with low level of Maturity it would mean they are operating in a reactive mode. Organisations that execute reactive activities are doing a minimum of planning or planned work (pro-active). Organisations that operate in a proactive mode are most likely to survive rather than those in reactive mode, thus, exposing the organisation to a high risk of failure.

Continuous improvement is the core of Asset Management. Challenges identified during the study at A-Coal are not unique to the case site, and are surely similar to challenges experienced in some other organisations.



Future work in some other organisations or industries on different sectors could lead to a wider spectrum of challenges identified at A-Coal in order to develop a more solid base of the key challenges that hinder organisations from attaining their desired level of asset management Maturity.

5.1 Limitations of Research Findings

One of the limitations of this study is that it was based on a single case study. There is always a possibility of the researcher making errors when dealing with qualitative data due to the imprecision of the data, interpretation, presentation or analysis. Given the methods used to collect data, especially from interviews, errors such as the possibility of poor recording, cannot completely be eradicated. It could happen that one single data could be interpreted twice or misinterpreted completely. Apart from the fact that this is a single case study and the sample was not well represented, especially where only one position participated, such as Finance Manager, there can only be one Finance Manager in an organisation. With these limitations in mind, the findings can still be compared to similar studies. The analysis depended on a single point of view of the researcher, which could be a weak point or distort the results or misinterpret participants' opinions and expressions.

The findings or results of this study could be improved and validated by future studies, which could take place in larger samples or multiple case studies. The aim is to generalise whatever could be generalized in mining, as key challenges that impede the AM Maturity improvement as desired. And this study could still be expanded not only within mining, but to other sectors, such as manufacturing or public sectors.

5.2 Industry Significance

To mature an AM process costs money and time, and AM benefits are not realised immediately, thus Senior Management might not give it priority of implementation or spend the monies required in relation to the AM initiatives. This is probably one of the reasons why senior management does not really commit, as they do not clearly see how their investment in improving the AM system/process will benefit their organization financially. Therefore, it is extremely important to pick a few KPAs which are in line with organisational objectives, ensure availability of resources and focus on those, once the desired Maturity has been attained, then one can pick the next set of KPAs or aiming for higher target on current KPAs driven or of focus. It is also important to ensure that a proper balance is attained since the KPAs are not in isolation and some could be linked. Following a structured and focused approach to improving AM Maturity could move the organisation to a more competitive position in the mining industry, and will provide results in the organisation which will have the following advantages:

- 1. A more focused and effective approach in executing AM activities;
- 2. An easy to follow translation of organisational objectives to AM activities;
- 3. Employees at 'coal-face' level will have a clear understanding of why they do what they are doing and how it contributes to the bigger picture for the organisation and what that means to them;
- 4. A proactive organisational culture can be developed.

6 REFERENCES

- [1] Antony, J., & Banuels, R. (2001). A Strategy for Survival. *Manufacturing Engineer*, 80 (3), 119-121.
- [2] Brown, R., & Humphrey, B. (2005). Asset Management for Transamission and Distribition. *Power and Energy Magazine*, *IEEE*, 3 (3), 39-45.
- [3] Company B. (2011). AM Assessment Report for A-Coal Plant.
- [4] Coronado, R., & Antony, J. (2002). Critical Success Factors for the Succesful Implementation of Six Sigma Projects in organisations. *The TQM Magazine*, 14 (2), 92-99.
- [5] CSA. (2012). Engineering Asset Management Issues and Challenges. Cambridge Service Alliance.
- [6] Davis, R. (2016, February 6). An Introduction to Asset Management. Retrieved from https://theiam.org/Whatis-Asset-Management
- [7] Dwight, R., & El-Akruti, K. O. (2009). The role of asset management in enterprise strategy success. 13th Annual ICOMS, Asset Management Conference, (pp. 66-76).
- [8] GFMAM. (2015). Asset Management Maturity: A Position Statement. Retrieved February 23, 2016, from Global Forum on Maintenance and Asset Management: www.gfmam.org/files/ISBN_978-0-98706-4-2
- [9] Hastings, N. A. (2010). Physical Asset Management.
- [10] Henderson, K., & Evans, J. (2000). Succesful Implementation of Six Sigma: Benchmarking General Electric Company. Benchmarking and International Journal, 7 (4), 260-281.
- [11] IAM. (2012). Asset Management an anatomy. The Institute of Asset Management.
- [12] IBM. (2007). The Evolution of Asset Management, Finding the Best Parctise, Not All Theory. USA.
- [13] ISO31000. (2009). Patent No. ISO 31000.
- [14] ISO55000. (2014). Patent No. ISO55000.



- [15] Kaplan, R., & Norton, D. (2001). The strategy focused organization: How balanced score-card companies thrive in the new business environment,. Harvard Business School Press.
- [16] Kaplan, R., & Norton, D. (2008). The execution premium: Linking strategy to operations for competitive advantage. Harvard Business School Press.
- [17] March, C. (2010). Life Cycle Engineering. Retrieved January 13, 2016, from <u>https://www.lce.com/The-Five-Biggest-Risks-to-Effective-Asset-Management-1224.html</u>
- [18] Repenning, N. P., & Sterman, J. D. (2001). Nobody Ever Gets Credit for Fixing Problems that Never Happened : Creating and Sustaining Process Improvement. *Carlifonia Management Review*, 43 (4).
- [19] Markow, M., & Rackosy, J. (2001). Asset management implementations plan and tiered system process. Transport. Cambridge, MA 02140.
- [20] Mollentze, F. J. (2005). Asset Management Auditing The Roadmap to Asset Management Excellence. MSc, Faculty of Engineering, Built Environment and Information Technology.
- [21] O'Hanlon, T. (2014). Asset Management Practices, Investments and Challenges 2014-2019. Reliabilityweb.com.
- [22] PAS55, B. (2008). Patent No. PAS55.
- [23] Hsieh, H.-F., & Shannon, S.E. (2005). Three approaches to qualitative content analysis. Qualitative Health Research, 15(9), 1277-1288
- ^[24] Patton, M.Q. (2002). Qualitative Research and Evaluation Methods. Thousand Oaks, CA: Sage.
- [25] Rose, D., Isaac, L., Shah, K., & Blake, T. (2012). Asset Management Guide: Focusing on the Management of Our Transit Investments. Retrieved January 13, 2016, from http://www.fta.dot.gov/research
- [26] Shah, M., & Lattlefield, M. (2008, November). Asset Performance Management.
- [27] Achanga, P., Shehab, E., Roy, R., & Nelder, G. (2006). Critical Success Factors for Lean Implementation within SMEs. The International Journal of Manufacturing Technology Management, Vol.17 (Issue.04), (pp. 460-471).
- [28] Wengraf, T. (2001). Qualitative Research Interviewing: Biographic Narrative and Semi-structured methods (4th Edition ed.). London: Thousand Oaks, California: Sage Publications Inc.
- [29] Woodhouse, J. (2014). Joining up the jigsaw puzzle coordinating the implementation of new practices, methodologies, education and culture change, all in a sustainable manner.
- [30] Yin. (2009). Case Study Research: Design and Methods (4th Edition ed.). Carlifonia: Thousands Oaks: Sage Publications.
- [31] Zikmund, W. (2003). Business Reasrch Methods (7th Edition ed.). Ohio, USA: Thomson South-Western.
- [32] Zook, C., & Allen, J. (2001). Profit from the core: Growth strategy in a era of turbulence. Harvard Business School Press.
- [33] Hodkiewicz, M., & Pascual, R. (2006). Education in Engineering Asset Management Current Trends and Challenges. International Physical Asset Management Conference. Tehran.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



THE EFFECTIVENESS OF THE INTERNAL QUALITY AUDITING AT A CABLE MANUFACTURING COMPANY IN SOUTH AFRICA

F. Chiromo1^{1*}and F. H. Ngobeni2²

¹Department of mechanical and Industrial Engineering Technology University of Johannesburg, South Africa <u>fchiromo@uj.ac.za</u>

²Department of Mechanical and Industrial Engineering Technology University of Johannesburg, South Africa hariethngobeni@gmail.com

ABSTRACT

This is a case study that investigates the effectiveness of internal quality auditing at a South African cable manufacturing company, hereafter referred as Cable Manufacturer. Cable Manufacturer is a supplier of a wide range of electrical cables for household and industrial applications. It receives raw material in the form of copper rod, 7.9mm diameter, and draws it to various conductors ranging from 0.205mm to 2.59mm in diameter. The conductors are then bunched into strands varying in cross-sectional area from 0.5mm² to 70mm². Cable Manufacturer is ISO 9001 certified. Data for this study was collected by a University of Johannesburg Industrial Engineering Student through interviews with employees, observation of processes and review of company documents. The research revealed that Cable Manufacturer's internal audit team does not adhere to set audit schedules. Moreover it does not have control on the system thereby resulting in line managers not adhering to plans. It focuses on compliance as well as continuous improvement audits. These findings have an impact on the company's growth and its ability to eliminate waste. Lastly, the study contributes by suggesting the internal quality auditing approach that Cable Manufacturer should adopt.

Key Words: Cable Manufacturer, ISO 9001 quality system, Internal quality Auditing, competitiveness, Customer specifications



1 INTRODUCTION

Cable Manufacturer is one of the leading cable manufacturers in South Africa. It manufactures an extensive range of electrical cables for application in power transmission, power distribution, rail, petrochemical, mining, ports, airports, wholesale, construction and domestic building environments. The company receives raw material in the form of copper rod, 7.9mm diameter, and takes it through a cable making pipeline that involves drawing, annealing, bunching, insulation, layup, bedding, armouring and sheathing. Figure 1 shows the schematic diagram of the cable making process flow in the company.

The company is one of the first organisations in South Africa to be awarded the South African bureau of standards (SABS) quality assurance certification, and has been ISO 9001 certified since 1991. Cable Manufacturer takes pride in its ability to identify and satisfy customer needs. The company serves on a number of South African bureau of standards (SABS) working groups. It produces an unarmoured cable designed to reduce the propagation of fire. Cable Manufacturer is listed with underwriters laboratories of United Kingdom.



Figure 1: Cable Manufacturer process flow diagram

This case study investigates the effectiveness of an internal quality auditing at Cable Manufacturer. The focus is on:

- interactions that the internal auditing team does with suppliers, factory management, workers and customers;
- internal quality auditing schedules;
- the role of internal quality auditing as a management tool in the company; and
- the contribution of the internal quality auditing in improving quality across the plant.

Internal audit is one of the techniques required by ISO 9000 standard [1]. It is conducted by a company on itself to check for compliance with requirements of the ISO 9000 standard and to report any identified non-conformities as a basis for corrective actions [2]. By eliminating non-conformities the company formally meets requirements for attaining or maintaining ISO 9000 certificate [2]. As the quality management system becomes more mature, the management's expectation of the internal quality audit changes. The purpose and orientation of the internal audit is changed from a compliance assessment to an assessment of continuous improvement and management system [3]. It becomes imperative that organisations concentrate on continuous improvement - in fact, it becomes a survival issue [4]. Continuous improvement is required in product or service quality, timeliness and in costs [4]. Figure 2 illustrates the focus of first party, second party and third party auditing. Second party and third party audits are beyond the scope of this study.

Kapapetrovic and Willborn [5], argue that first party (internal) auditing is a set of interdependent processes or activities using human, material, infrastructural, financial, information and technical resources to achieve objectives related to the continuous improvement of performance. It is an approach used by top management to



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

follow up on the status of a management system and to provide input to improve processes and change attitudes. When the quality system conforms to ISO 9000, it becomes necessary that:

- all areas including the management review and the audit activities are audited;
- internal audits are documented;
- a documented audit plan is prepared for a particular period;
- people carrying out the audits are sufficiently competent;
- results of the audit are documented; and
- audit findings are used as a basis for improvement.

Measuring internal quality audit effectiveness involves the evaluation of the whole audit system, including its objectives, processes and resources. According to Beckmerhagen et al [6], an effective audit is defined as a joint probability that the audit will be:

- suitable to achieve set objectives under a defined scope;
- reliable;
- valuable; and
- maintainable.

In order for an audit to achieve its objectives, Beckmerhagen et al [6] suggest that the following principles should be adhered to:

- any audit should be conceived according to the guidelines specified in the applicable audit standard;
- the auditor should plan the audits; identify the goal, purpose, available resources and possible problems;
- general criteria or standards for proper audit performance should be clarified, such as auditor qualification and competence;
- the evaluation of audit effectiveness must be based on facts;
- the auditor must assess his/her own competence before accepting an auditing assignment;
- the auditor should assess and measure his/her own performance in order to identify improvements;



Figure 2: Types of Quality Audits (Orr, [7] Lecture Material- The Queen's University of Belfast)

- reviews of audit effectiveness should be conducted periodically as a matter of procedure or when requested by the auditor. The best time is shortly after a regular audit has been conducted and completed;
- linkages to preceding and subsequent audits should be considered. Individual audits are interconnected in such a way that the output of a preceding audit must be considered as one of the inputs into the subsequent audit;
- the auditor must be able to discontinue an audit when risks become unacceptable;



- an effective audit is aimed at continuous improvement;
- all parties interested in the successful audit performance, including the client, auditee, and the audit management, should be involved in the process of determining audit effectiveness.

Moreover Beckmerhagen et al [6] suggest that the criteria for effective audits are:

- defining adequate and feasible audit objectives, which are approved by all interested parties;
- preparing a suitable audit plan, which is accepted by all interested parties;
- providing adequate resources and time to complete the audit;
- planning and executing the audit by a properly appointed and competent auditor;
- conducting the audit in accordance with recognised audit standards and procedures;
- finding valid non-conformances;
- audit results that lead to corrections and improvements;
- satisfying the client completely in terms of achievement of stated objectives; and
- providing objective evidence to all interested parties that the audit resulted in an improvement of the quality management system.

The purpose and outcome of internal quality auditing depend on the purpose or motivation for introducing the quality management system [2]. A company that has an external motive, only implements a compliance audit, which just checks formal conformity with the quality standard. This enables the company to maintain the certificate [2]. Since the business environment demands not the status quo, but continuous improvement, then an effective audit must be adaptable to the changes in both the quality management system and the overall business [6]. By virtue of being a goal-oriented activity, the audit process and its achieved outcomes (audit performance) are always compared with the planned objectives [6].

The success of internal auditing depends on the co-operation of both auditors and auditees within the audit process. They need to have an open conversation without hiding the facts [2]. Managers should show some understanding in the case of detected non-conformities and regard them as a chance for improvement [2]. Moreover internal quality Auditors can detect cases of good practices and spread these good practices and ideas to other environments within the company.

According to Alic and Rusjan [2], other benefits derived from internal audits are:

- they give an in-depth understanding of the company's quality system and its requirements;
- they help employees become familiar with the other organisational units. This gives better knowledge of the internal "customers" and "suppliers";
- they encourage cross-organisational learning and experience sharing;
- they influence workers' attitudes to the quality of their work;
- when preparing for an audit both auditors and workers review work processes, documentation and records. So they renew their knowledge and possibly find something in the documentation that should be corrected;
- they are a means to stimulate business improvements; and
- they help managers attain the business objectives of the company.

Once an internal audit and review process has been done, a challenge that one faces is prioritisation of the findings. As the number of deficiencies or opportunities for improvement increases, the probability of implementing effective solutions decreases [4]. In this regard one would find refuge in focusing on the vital few areas than to take on too many changes and implement none [4].

2 METHODOLOGY

Data for this study was collected by a University of Johannesburg industrial engineering student. The student had interviews with the quality assurance manager, quality auditors, production manager, procurement manager, and calibration officers. She had attained her Bachelor of Technology in industrial engineering and had just started her studies for an M-tech in industrial engineering. The student undertook the following:

- arranged appointments with departmental heads;
- delivered and explained the contents of the questionnaire;
- collected the completed questionnaire and verified the answers by taking tours of the production floors and
 offices of the facility.

Moreover, the student collected secondary data from the company working documents and reports that satisfied the research objectives. Lastly a reviewed of journal articles was done to gain knowledge on the current direction of internal quality auditing.



3 FINDINGS

3.1 Principles upon which audits are based

Cable Manufacturer believes that ethical conduct is the foundation of professionalism. All internal quality auditors at Cable Manufacturer are obliged to observe this requirement.

The auditors are bound to report truthfully and accurately the audit findings. The company requires them to report correctly, significant obstacles encountered during the audit as well as unresolved or diverging opinions between the audit team and the auditee.

They are bound to be independent of the activity or process being audited and be free from bias and conflict of interest. This ensures that the findings and conclusions are based purely of evidence found.

There is a drive that encourages that audit evidence generated at Cable Manufacturer must be verifiable.

3.2 Internal audit

Cable Manufacturers does the audits to:

- to verify the effectiveness of the quality management system; and
- meet the objectives of Cable Manufacturer ISO 9000:2008.

To facilitate effective execution of the audits, the company develops audit plans every calendar year. The plan is then uploaded onto the SAP system for easy access by the respective line manager. The plan identifies:

- what is to be audited;
- when audits will be done; and
- who does the audits. Figure 3 does not show the auditor concerned. This is to protect the identity of Cable Manufacturer.

3.3 Audit scope

Internal audits at Cable Manufacturer focus largely on ensuring compliance of products and processes with; the ISO 9000 standard, legal requirements, regulations as well as specific customer requirements. The audits cover marketing, sales, technical department, operation, warehouse, and engineering. The task of improving products and processes is delegated to the continuous improvement department. This is a shortcoming in the effectiveness of the internal quality auditing department. There is however a strong working relationship between the auditing department and the continuous improvement department. Improvement opportunities and updates are communicated across the departments.

Figure 3 shows a representative sample of the audit plan at Cable Manufacturer. The plans are for the 2014 and 2015 calendar years. The figure shows that 2014 was a smooth year and did not have disruptions to the audit plan. However in 2015 things did not go well. There were eighteen planned internal audit activities that were cancelled. Two were not executed as initially planned.



		2014 2015																							
REF:	Description	L	F	м	A	м	L	L	A	s	0	N	D	L	F	м	A	м	L	L	A	s	o	N	D
150 0001	Document Control													1											
130 9001	Control of Records																								
	Management Commitment																								
	Group Sherq Policy	1				1				1				1		1									
ISO 9001	Quality Objectives				1		1		1		1			1											
	Responsibility, Authourity & Communication	1				1				1		1		1		1		1							
	Management Review										1			1											
	Management Resources													1											
	Competence, Training & Awareness				1		1		1					1			1			[
	Infrastructure Maintenance	1								1				1			[
	Planning of Product																								
	Customer Process	1				1				1		1		1				1							
	Design Control													1											
	Purchasing					<u> </u>								1											
	Verification of Purchased Product validation of processes for production and service		9)				6																		
	Rod Break, WD, MWD &Bunching				\square	7			†		÷			1	·····	11111	.			·					
ISO 9001	Insulation and Bedding	1							1	1		1		1		1		1					[
	Lay Up & Armouring	1			1	1	1		1	1	1	1		1			100			[,,,,,,,			
	Sheathing	1		1	1	1	1		1	1		1					í uu								
	Final Test	1			1	1	1		1	1	1	1		1	1			1							
	Inspection and Testing	1		1	1		Í		1	1	1	1		1			1	1							
	Identification & Traceability	1			1	·			1	1		Í		1	1		1	1							
	Preservation of product (and Warehouse)	1			1	1	1		1	1				1			1	IIIII	(
	Control of monitoring & Measuring Equip.	1			1	1		[1	1	1	1		1		1	1	<u> </u>							
	Audits													1											
	Customer Satisfaction				1		1		1		1						1								
	Monitoring and Measure of Processes	1			1	1				Ĺ	1	1		1				1							
	Monitoring and Measure of Prod	1			1	1					1	1		1			1	1				IIIII			
ISO 9001	Controll of Non-Conforming Prod.	1			1		((1		[1				1							
	Analysis of Data (to be audited during audits)	1			1	(1	1	1	(1											
	Continual Improvement (to be audited during audits)	1	[1	T	1	T		T	1	T	1	[1	[
	Corrective Action (to be audited during audits)	T		[1	T	1			I	1	T		1	1										
	Preventative Action (to be audited during audits)																								

Figure 3: Disruptions to 2014 and 2015 internal quality audits at Cable Manufacturer



Key:

Dark black - means the planned internal audits activity was successfully completed. Note this does not mean it was done on time. Yellow with black stripes - means that the planned audit activity was cancelled. Medium black - planned audit activity Light black - the initially cancelled activity was rescheduled

The disruption in the 2015 plan (Figure 3) were partly due to:

- the resignation of one of the internal auditors in December 2014. A suitable replacement was only recruited in August of 2015;
- lack of cooperation from some line managers. They did not avail themselves on the planned dates. This resulted in the non-fulfilment of the audits.

These disruptions had a negative impact on customer complaints. The period January 2015 to July 2015 saw a major increase in customer complaints. These complaints were associated short lengths, wrong labelling, damage, rough sheathing, tight stripping, wrong cable supplied and oversize outside diameter. These complaints point to the worsening of processes and management system in Cable manufacturer. During that time the performance of the company was getting worse. Table 1 shows the extent to which 2015 customer complaints compare with those of 2014.

The situation started improving from August 2015, when a new auditor arrived. This drop in customer complaint, starting August 2015, shows significance of internal quality auditors in the company.

	Number of C	Complaints
Month	2014	2015
January	7	15
February	4	19
March	7	31
April	9	5
May	14	18
June	18	20
July	12	25
August	43	6
September	17	15
October	20	1
November	18	12
December	11	2

Table 1: 2014-2015 Cable Manufacturer customer complaints

3.4 Corrective Action

Audit findings are presented to respective line managers as feedback for continual improvement and verification of conformance to the quality. Table 2 shows a typical internal quality audit report for the sheathing department. The company classifies the findings into three categories; comment, minor and major.



SAIIE27 Proceedings, 25^{th} - 27^{th} of October 2016, Stonehenge, South Africa $\ensuremath{\mathbb C}$ 2016 SAIIE

Ref No.	Standard and clause number	Finding (classification	Description of Findings
FH230315-001	ISO9001:2008 8.2.3	Minor	Incorrect product loaded onto the Machine Execution System (MES) - $2.5mm^2 \times 4$ cores instead of $2.5mm^2 \times 2$ cores as per job shop order (work instruction) number 7621378
FH230315-002	ISO9001:2008 7.5.3	Minor	The label used for identification and traceability at machine PR0352 could not be traced to the previous shop order number and the product size was not recorded on the ticket.
FH230315-003	ISO9001:2008 8.2.4	Minor	The label used for identification and traceability found at machine PR0352 did not have an inspection and test criteria as required on the identification ticket. (QC approval)
FH230315-004	ISO9001:2008 4.2.3	Minor	Extrusion pointer and die selection chart found on machine PR0995, Doc No. MWD 7510 issued by Hamilton Ndlovu, was not controlled in the Aberdare Integrated System, i.e. it was not a controlled document.
FH230315-005	ISO9001:2008 6.4 and OHSAS 1800:4.4.6	Minor	Two electrical switch boxes were open (cover broken) at the machine payoff at PR0995.
FH230315-006	ISO9001:2008 8.2.4 and SANS 1507 part 2	Minor	When running 2.5mm ² x 2 suffix wiring cable black, the Clinton spark tester model Hf-15AC was set incorrectly at 6 KV instead of 8KV.

Table 2: Typical Internal Quality Audit Report (Sheathing department)

A comment is a token of appreciation of the adherence to quality standards. A minor finding is a moderate nonconformance. The standard practice in the company is that all minor non-conformances should have been rectified by the time the next audit is done. Lastly, a major non-conformance is a serious finding and it calls for an immediate corrective action. The affected department is given a maximum of seven working days, from the day an audit report is presented, to rectify the non-conformance.

The categorisation of findings has an added advantage of helping in prioritization of findings. As a rule of the thumb, major findings are prioritized ahead of minor findings. This emanates from the benefits associated with attending to major findings.

Irrespective of the category of the finding, Cable Manufacturer follows a standard approach in effecting a corrective action. Figure 4 shows the steps that were taken in attending to a minor non-conformance number FH230315-002 in Table 2.



Description of finding:

The label on machine PR0352 used for identification and traceability could not be traced to the previous phase. The previous shop order number and the product size were not recorded on the ticket.

Root cause of finding:

Operator at previous phase had not filled in the ticket correctly.

Immediate action taken: (Correction)

The ticket was filled in correctly and then the quality assurance department verified the test status.

Corrective action taken: (Man/Machine/Method)

The significance of the identification and traceability ticket was explained to the machine operator. Moreover the operator was shown how to adequately and correctly fill in the ticket.

Figure 4: Corrective action taken to finding FH230315-002 in Table 2.

To encourage respective departments to take corrective action on findings reported, the internal auditors make follow up at set dates. They close all non-conformances for which corrective action has been taken. This way the internal quality auditing team influences continuous improvement.

3.5 Analysis of performance of Stock Items

Cable manufacturer has specifications for all its stock items. These specifications are only reviewed and changed when there is a change in product design. What is lacking is a trend analysis in stock item performance. If done, this could be one source of information for continuous improvement initiatives.

3.6 Benefits derived from internal quality audits.

Benefits that comes with internal audits at Cable Manufacturer are:

- that processes that are not performing in accordance to the set standards are highlighted and are resolved and/or controlled as soon as they are identified;
- the internal quality audits have helped decrease quality costs as well as the number of customer complaints received;
- there is sustainable production of quality products;
- the internal quality audits have shortened the gap between the workforce and management. They have
 induced a unifying effect on driving workers and management in working towards attaining the company
 objectives.

3.7 Audit of an audit

In addition to internal audits conducted at Cable Manufacturer, the company has engaged South African Bureau of Standards (SABS) to conduct third party audits for ISO 9001 quality standard certification. Eskom being a major customer also conducts second party audits. To help market its products in United Kingdom Cable Manufacturer has engaged British Approval Service for Cables (BASEC) to conduct third party audits. This assures European customers of quality products. All these external audits help the internal audit team in Cable Manufacturer improve.



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE

4 CONCLUSIONS

From the study done on Cable Manufacturer the following conclusions were arrived at:

- Cable Manufacturer has both internal and external motives. The audits on one hand check formal conformity with ISO 9000 quality standards. On the other hand they are used as a basis for quality improvements.
- As shown in Figure 3, there are no adequate controls put in place to avoid disruptions to internal audit plans. These disruptions compromise the effectiveness of the audits.
- Cable Manufacturer has a system to capture non-conformities (Table 1). This is the basis for effecting corrective actions and improvements.
- The company also has a standard procedure of effective corrective action.
- Lastly Cable Manufacturer internal audit team work with three external auditors, SABS, Eskom and BASEC. These collaborations help improve the operations of the auditing team.

5 RECOMMENDATIONS

- The internal auditing system at Cable Manufacturer has not done much to involve workers. Quality circles are one way the company could involve workers. When workers are involved they feel motivated and this way they make significant contribution towards improving the quality system.
- Cable Manufacturer has a rigorous quality control system for all stock items. On purchase, each item is subjected to compliance tests. However, the company does not have a supplier selection system. Having a supplier selection system in place helps decision makers monitor the performance of existing supplier portfolio [8]. This, in turn can be used to negotiate further contracts in different segments with the existing vendors, given their current capabilities [8].
- Cable Manufacturer developed skills matrix in all production departments. The matrix determines the adequacy of workers in addressing the challenges of the company. There is also a need to develop such a matrix in the internal auditing department.

6 **REFERENCES**

- [1] **Terziovski, M. and Guerrero, J.L. 2014.** ISO 9000 quality system certification and its impact on product and process innovation performance, *International Journal , Production Economics 158, 198-207*).
- [2] Alic, M. and Rusjan, B. 2012. Managerial relevance of internal audit- Business benefits of using ISO 9000 internal audit as a managerial tool, *The TQM Journal, Volume 23, Number3, 284-300*.
- [3] Alic, M. and Rusjan, B. 2010. Contribution of the ISO 9001 internal audit to business performance, International Journal of Quality and Reliability Management, Volume 27, Number 8, 916-937.
- [4] **Beecroft, G.D. 1996.** Internal quality audits obstacles or opportunities?, *Training for Quality, Volume 4, Number 3, 32-34.*
- [5] **Karapetrovic, S., and Willborn, W. 2000.** Quality assurance and effectiveness of audit systems, International Journal of Quality and Reliability Management, Volume 17, number 6, 697-703.
- [6] Beckmerhagen, I.A., Berg, H.P., Karapetrovic, S.V., and Willborn, W.O. 2004. On the effectiveness of quality management system audits, *The TQM Magazine, Volume 16, number 1, 14-25*.
- [7] Orr, J. 1995. Unpublished lecture material, The Queen's University of Belfast, Northern Ireland
- [8] Koprulu, A., and Albayrakoglu, M.M. 2007 Supply Chain management in the Textile Industry: A supplier Selection Model with the analytical Hierarchy Process, *ISAHP 2007, Vina Del Mar, Chile*.



PRACTICAL IMPLEMENTATION OF THE ISO 50 001 STANDARD FOR LARGE INDUSTRIES

M. van Heerden^{1*}, M.J. Mathews² & J.H. Marais³

This work was sponsored by HVAC International. ^{1,2,3}Center for Research and Continued Engineering Development (CRCED) North-West University, Pretoria, South Africa ¹mvanheerden@researchtoolbox.com ²mjmathews@researchtoolbox.com ³jhmarais@researchtoolbox.com

ABSTRACT

Energy management is becoming a necessity when competing in large industries in South Africa. An effective energy management system will improve a company's competitiveness by decreasing energy inputs while increasing production outputs. Therefore, one of the main challenges in energy management is to implement an effective energy management system to best benefit the company's needs.

The stipulated requirements and practical guidelines for a recognised energy management system are available in the ISO 50 001:2011 (Energy Management System) International Standard. However, due to the complexity of the standard, most companies defer their energy management strategies to outside contractors or cease implementation altogether.

There is therefore a need to simplify the process of implementing the ISO 50 001 standard for large industry. This study describes a simplified implementation model for the industry. The aim of this model is to keep the implementation time to a minimum. To ensure minimum effort for the implementing company, an industrial template was also developed.

Sectional checklists and practical examples form the basis of the model and the template can be adjusted as specified in the different industrial requirements. A practical execution of the model also showed promising results.

^{*}Corresponding author



1. ENERGY MANAGEMENT IN THE INDUSTRY

South Africa is in a stressed situation involving electricity usage. Constraints in Eskom's electricity supply places users in a difficult situation in terms of increases in electricity tarrifs and this has a negative economic impact on users.

The annual electricity tariff increase has been higher than the consumer price index (CPI) for a significant number of years. This results in an increase on production costs within local industries. With these price increases, the need to obtain cost savings in various ways has become crucial [1].

The consumption of energy within large industries is manageable to a certain extent. Therefore, the need to develop and implement an Energy Management System (EnMS) arises. Companies with the focus on continuous improvements for their energy management processes and waste reduction, have increased their profitability and operating quality. This has resulted in improved productivity, global respect and significant profit increases [2].

Organisations strive to implement effective energy management strategies in order to reduce costs and energy consumption. They also strive to improve their corporate image in the relevant market by reducing their environmental impact. In order to realise these advantages, organisations should aim to reduce their total energy waste by following specific programs. [3]

The most effective programs are being developed according to the set requirements and guidelines from the ISO 50 001 standard [4]. However, before a successful energy management strategy can be developed, the international standard should be understood completely. This is one of the most challenging tasks a company faces in terms of the ISO 50 001 standard.

It is not necessary for an organisation to be certified in order to be successful in managing their energy. Thus, some organisations comply with the standard but never complete the step of registering for certification [5]. The key to implementing the ISO 50 001 standard in an organisation is to develop a sustainable system within the shortest time possible. This study will thus detail the ISO 50 001 standard and provide a framework for its effective implementation.

2. OVERVIEW OF ISO 50 001

ISO 50 001 is a published standard that describes how to develop an EnMS. The ISO 9 001 (Quality Management System) and ISO 14 001 (Environmental Management System) standards assisted in the development of this standard. Thereafter the ISO 50 001 standard was published and released by ISO in 2011 [6].

The ISO 50 001 standard provides specific requirements and guidelines to develop, implement, manage and improve an EnMS [7]. The main intention of the standard is to control and minimise the total energy consumption for companies. Employee behaviour and process changes, within feasible boundaries, can outline the activities that will ensure the achievement of this purpose [8].

Unlike the ISO 9 000 series, ISO 50 001 does not have a series of standards. The ISO 50 001: 2011 is a single standard that describes energy management using an EnMS. Compliance to the standard can also be certified by external certification bodies. An integrated certification from some of the other ISO standards can also be achieved and accepted by ISO 50 001 [7].

The main focus of the standard is for an organisation to develop a strategic model to successfully manage energy consumptions and performance. It is expected that the ISO 50 001 standard could influence about 60% of the energy used internationally [2].

The ISO 50 001 standard follows the continual Plan-Do-Check-Act (PDCA) cycle. Organisations need to implement this cycle into their EnMS and daily energy management strategies [9]. Figure 1 indicates the corresponding PDCA steps outlined in the ISO 50 001 standard. Every aspect in the development and implementation of an EnMS using the ISO 50 001 standard is based on a continuous improvement cycle.

The energy policy and corresponding energy plan should be investigated and developed during the planning phase. The developed energy plan should then be implemented during the second phase and the performance thereof verified during the checking phase. During the act phase, an organisation's energy performance and effectiveness of the EnMS should be analysed and the necessary amendments should be modified on the EnMS. The management's decisions will be based on the performance results. These managerial alterations should be integrated in the energy policy and energy plan to continually improve the EnMS.





Figure 1: The corresponding PDCA steps in ISO 50 001

The ISO 50 001 standard guides an organisation in order to achieve the following [9]:

- Develop a system to manage energy more efficiently,
- Establish unique requirements and goals to achieve within the system,
- Process data to explain and control energy usage more effective,
- Compare and provide intelligent feedback on the processed data,
- Monitor the system, and
- Improve the EnMS continually.

General recommendations and frameworks can be established within each uniquely developed EnMS. This will serve as the benchmark by which energy performances can be measured. By implementing a strategy with benchmarking targets unique to a specific organisation, more realistic financial benefits could be seen.

3. IMPLEMENTATION STRATEGY

3.1 Introduction

An EnMS serves as a centralised strategy for companies to manage their energy performance. This paper will identify several shortcomings in the implementation of previously developed energy management systems.

The model developed will also be discussed according to three levels of implementation. These levels are defined to assist companies in choosing the level of detail required in which they want to manage their energy needs.

3.2 Shortcomings of implementation

Every organisation develops or uses their own specialised EnMS in order to best benefit their operations and objectives. As there is no set strategy for implementing the ISO 50 001 standard, shortcomings could exist in every system.

It should thus be a critical step for any organisation to identify and investigate the possible shortcomings from similar systems before developing their own. The implementation time for an EnMS should also be kept to a minimum. This can be done by eliminating all identified shortcomings before implementation.

Some inefficient EnMS could be due to organisations not planning properly. An EnMS should be developed according to the specific industry or service provided. Problems may occur when organisations do not address certain challenges correctly. These challenges may include reductions in resources for the same production targets. These resources may involve energy as well as materials [10].



The ISO 50 001 standard does not provide information about the most productive procedures when implementing the standard according to specific industry. Companies might have insufficient infrastructure available to implement an EnMS. A lack of measuring devices or knowledgeable resources can be a big concern [10].

Issues such as the availability of data should be addressed as soon as possible. These issues should be identified in the plan phase of the PDCA cycle. The collection of historical energy data should also be considered. This process of data collection could be inaccurate and may delay the process, which will influence the implementation time [11].

A successful energy management model should thus be developed by using the PDCA cycle approach. This will ensure proper planning as well as continual improvements [12]. The basic ISO 50 001 standard represents good practices. However, this does not always ensure that the model will be sufficient to achieve ideal energy performances [13].

3.3 Levels of implementation

3.3.1 Introduction

Developing an EnMS for an organization can take months or even years to complete. There is however several action steps that will form the main part of the system. Every fundamental step helps develop a more detailed model and ultimately an effective EnMS for an organization.

Three standard levels of implementation were identified in this article. They include the following:

- Basic model,
- Intermediate checklist, and
- Advanced EnMS.

3.3.2 Basic model

After investigating the requirements stipulated in the ISO 50 001 standard, a high-level framework was developed. This framework indicates a simplified division of the various criteria involved in developing an EnMS. These criteria were grouped according to the PDCA cycle steps for a more understandable approach. Figure 2 indicates this simplified framework.




SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Figure 2: ISO 50 001 requirements simplified according to PDCA cycle steps (Basic model)

The basic model is indicated in Figure 2 above. This represents the high level investigation of an ISO 50 001 system for any organization.

As soon as an organization considers developing an EnMS, they could use this model to identify the steps that need to be followed according to the PDCA cycle. They can then identify all the basic requirements that should be included in the final EnMS.

However, organisations will not be able to register for an ISO 50 001 EnMS certification audit after completing this model. This can only be seen as a blueprint of an organisation's proposed EnMS and does not provide any detailed information regarding the energy performances of the company.

3.3.3 Intermediate checklist

After a basic proposal for an EnMS is created, the detailed investigation should proceed. At this level, data will be captured and collected from various sources. Data will then be compared to baselines and thereafter compared to targets. An EnMS checklist should be created in order to identify the detailed areas to be investigated next. The checklist allows the user to identify missing criteria in the EnMS. This should be used to create an auditable system for accreditation in the future.

The developed checklist consists of the same five steps as mentioned in the basic model above. Under every step extended criteria are mentioned to ensure that those elements are included in the organisation's EnMS. The level of detail for each element in the checklist is dependent on the type of organisation, the size of the company and the need for certification. An example of this checklist is illustrated in Figure 3 below.



Company name:	
Company type:	
Date checked:	
Step 1: Energy Policy	Yes/No
Management commitment	
Defining scope of system	
State resources and information to achieve targets	
Define significant energy users	
Legal requirements	
12 month comparison for EnPI	
Step 2: Energy Plan	Yes/No
Analyse and compare baselines	
Prioritise energy savings opportunities	
Realistic objectives and targets	
Predictions on energy savings	
Verification procedures	
Energy performance tracking	
Step 3: Implementation	Yes/No
Design specifications for SEUs	
Hardware and software	
Training	
Internal and external communication	
Document templates and latest versions	
Step 4: Checking	Yes/No
Monitor the energy performance	
Calibration of measuring devices	
Identify and investigate outliers in the performances	
Corrective actions for non-conformaties	
Preventative actions in place	
a . a a	Yes/No
Step 5: Management review	
Step 5: Management review Internal audit results	
Step 5: Management review Internal audit results Continual improvement strategies	
Step 5: Management review Internal audit results Continual improvement strategies External audit	

Figure 3: Developed EnMS checklist for organisations (Intermediate checklist)

A descriptive energy policy should be compiled during the first step. This policy should primarily include the commitment from management to comply with the set targets. The energy performance targets should be stipulated in accordance with the identified Significant Energy Users (SEU) of the company. The legal requirements identified for the type of company should also be stated in the policy.

The energy plan is created in the second step in accordance with the energy policy defined previously. The main part of the energy plan is the continuous improvement of the identified SEUs. Performance data should be compared with the developed baselines. This could determine whether the set targets are realistic. Energy performance tracking should be conducted in order to revise the set targets and objectives.

Step 3 forms the body of the EnMS. The specific energy related projects, as identified through the SEUs, should be designed to ensure the most effective energy management on the site. The required instrumentation and applications should be obtained to best fit the organization and its projects. Continuous training should be provided on new initiatives and projects. This will ensure awareness and help minimise the implementation time of the EnMS before results are available.



Overall system monitoring should be done during the fourth step. This is also a continuous process and should be analysed and improved as regularly as possible. The energy performances should be monitored in order to verify the accuracy of the targets at all times. Calibrations should be done where required to ensure energy savings are measured accurately. This allows action to be taken on the actual performance of energy saving initiatives.

The last step of the process is the management review which must ensure that the EnMS is performing as required and decide on how it can be improved. In this step audits should be done internally and externally in order to determine the success of the EnMS. External audit results will influence the certification process. Therefore the intermediate checklist should be completed as far as possible before completing the advanced EnMS as mentioned in the next section.

The responsibilities of all the internal and external stakeholders which may include the energy management team members should be clearly stated. Document control should be followed for all the relevant instructions, procedures, specifications and reports. All documents should be reviewed and approved before being filed.

3.3.4 Advanced EnMS

The ISO develops standards, but does not provide certification. External certification bodies are responsible for the certification process [14]. This means that a company can be ISO (standard requirements) compliant, but not certified as such.

For companies to be ISO 50 001 certified, they need to adapt to the standard requirements and prove their improved energy performance. The management strategy and results should then be certified by an external party [15].

During this level of implementation, organisations should focus on completing the missing elements in detail after completing the intermediate checklist. This will ensure an ISO 50 001 compliant EnMS to be audited.

4. PRACTICAL IMPLEMENTATION ON A GOLD MINING COMPANY

The developed method was implemented at a selected mining group in order to develop their EnMS. Various system phases were investigated to evaluate the efficiency of the EnMS.

All the steps involved in the PDCA cycle were conducted individually to achieve sustainable results. The plan and do phases focused mainly on developing the EnMS and its functionalities. The defined energy objectives and targets were set forth in the energy policies of the mining group as well as for the individual mines. The SEUs were defined for every mine and can be seen in Table 1.

Mine	Projects	SEU	Savings target (%)	
Mine A	Initiative A	Pumping	19%	
Mine A	Initiative B	Air network	11%	
Mine B	Initiative C	Cooling Air	20%	
Mine B	Initiative D	Water network	9%	
Mine C	Initiative E	Air network	12%	
Mine C	Initiative F	Pumping	15%	
Mine D	Initiative G	Air network	26%	
Mine E	Initiative H	Air network	28%	

From the energy policy, the energy action plan was developed. The actual implementation of the activities for data collection, system setups and report generations were delivered according to the energy action plan. Energy savings initiatives were also implemented as per this plan.

The energy performance results of the initiatives were analysed during the check and act phases. The performance impacts were compared to the targets set during the planning stage. These targets were confirmed after the initiatives were implemented and enhanced where necessary.



Through the process of developing their own EnMS the mining group achieved significant savings on their electricity cost. All eight identified initiatives achieved their specified target savings during the course of assessment. A summary of the project performances after 8 months since implementation is given in Table 2.

Mine	Project	Savings target (%)	Actual savings (%)	To	tal cost savings
Mine A	Initiative A	19%	21%	R	1 513 482.00
	Initiative B	11%	26%	R	11 297 061.00
Mine B	Initiative C	20%	32%	R	5 272 794.00
	Initiative D	9%	21%	R	6 661 934.00
Mine C	Initiative E	12%	13%	R	5 170 715.00
	Initiative F	15%	17%	R	2 444 439.00
Mine D	Initiative G	26%	45%	R	12 261 942.00
Mine E	Initiative H	28%	33%	R	7 472 724.00
TOTAL				R	52 095 091.00

Table 2: Energy management projects performance summary

5. CONCLUSION

Many different models have been developed in order to manage the energy usage in large industries. Organisations also often develop their own EnMS strategies to improve their energy performance. This can however result in overly long and complicated implementations. This study therefore describes a model which divides the implementation strategy of an EnMS into three main levels.

The first level (basic model) is defined to ensure that the company has the basic elements required according to the ISO 50 001 standard. These elements include a basic description of an energy policy, energy teams and management targets. The next level (intermediate level) is used with the basic model to ensure compliance with the standard. Relevant data, processes and actions are compared to a developed checklist to determine the status of the company's EnMS.

The third and final level of implementation (advanced EnMS) would be where the implementing organisation gathers and develops all the missing detailed elements from the second level. This is the level of implementation where companies ensure continual improvements and is the point at which the EnMS can be certified as compliant to the ISO 50 001 standard.

After implementing the first two levels of this model, a selected gold mining company already saved over R 52 million in electricity costs during 2014. This combined saving was recorded after eight months of implementation of eight separate energy savings initiatives.

REFERENCES

- [1] **Dr. J van Rensburg, HJ Groenewald, Dr JH Marais**, 2014 "Business case for industrial DSM maintenance," presented at the Industrial and Commercial Use of Energy Conference (ICUE), Cape Town, South Africa.
- [2] P. Ranky, 2012 "Sustainable energy management and quality process models based on ISO 50001: 2011 the International Energy Management Standard," in *Sustainable Systems and Technology (ISSST)*, 2012 *IEEE International Symposium on*, pp. 1-6.
- [3] V. Introna, V. Cesarotti, M. Benedetti, S. Biagiotti, and R. Rotunno, 2014 "Energy Management Maturity Model: an organizational tool to foster the continuous reduction of energy consumption in companies," *Journal of Cleaner Production*, vol. 83, pp. 108-117.
- [4] **A. McKane,** 2010 "Thinking Globally: How ISO 50001-Energy Management can make industrial energy efficiency standard practice," *Lawrence Berkeley National Laboratory*.
- [5] National Cleaner Production Centre of South Africa, 2013 Introduction and Implementation of an Energy Management System and Energy Systems Optimization: Case study of Arcelormittal Saldanha Works. Available: <u>http://ncpc.co.za/files/CaseStudies/EnMS_</u>2013_ArcelorMittal_Suldanah_Works_CS.pdf



- [6] ISO. (4 August). *ISO 50001 Energy management*. Available: <u>http://www.iso.org/iso/home/standards/management-standards/iso50001.htm</u>
- [7] ISO helpline. (4 August 2014). EnMS ISO 50000 | ISO 50001:2011 PPT | Energy Manual. Available: http://www.isohelpline.com/enms-iso-500012011-ppt-energy-manual.htm
- [8] **T. Fiedler and P. Mircea,** 2012 "Energy management systems according to the ISO 50001 standard— Challenges and benefits," in *Applied and Theoretical Electricity (ICATE)*, 2012 International Conference on, pp. 1-4.
- [9] SABS, 2011 "SANS 50001:2011 Energy management systems Requirements with guidance for use," ed. Pretoria: SABS Standards Division.
- [10] M. Dörr, S. Wahren, and T. Bauernhansl, 2013"Methodology for Energy Efficiency on Process Level," *Procedia CIRP*, vol. 7, pp. 652-657.
- [11] I. Dzene, I. Polikarpova, L. Zogla, and M. Rosa, 2015 "Application of ISO 50001 for Implementation of Sustainable Energy Action Plans," *Energy Procedia*, vol. 72, pp. 111-118.
- [12] P. Antunes, P. Carreira, and M. Mira da Silva, 2014 "Towards an energy management maturity model," *Energy Policy*, vol. 73, pp. 803-814.
- [13] B. Jovanović and J. Filipović, 2016 "ISO 50001 standard-based energy management maturity model proposal and validation in industry," *Journal of Cleaner Production*, vol. 112, Part 4, pp. 2744-2755.
 [14] ISO. Certification... Available: <u>http://www.iso.org/iso/home/standards/certification.htm</u>
- [15] M. Wheeland, 2011, 6 Things You Need to Know to Get Certified Under ISO 50001. Available: http://www.greenbiz.com/news/2011/07/29/6-things-you-need-know-get-certified-under-iso-50001?page=full



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



THE IMPACT OF MAJOR ENVIROMENTAL, SOCIAL AND ECONOMIC FORCES ON THE FIELD OF INDUSTRIAL ENIGINEERING

H. Darwish^{1*}

¹Department of Industrial Engineering North West University, South Africa <u>Hasan.Darwish@nwu.ac.za</u>

ABSTRACT

Every profession is bound to adapt in response to changes in the macro environments. Industrial Engineering is especially affected due to its broad application across various fields, i.e. manufacturing, customer service design, consulting and more. This article argues that the need for change is envisaged to be radical due the deeprootedness of Industrial Engineering in specific areas where environmental, social and economic (ESE) forces are prompting fundamental transformation. This transformation often manifests itself in an evolving technology. The argument of this article is that 1) Means of production are changing due to 3D printing and additive manufacturing 2) Employees work arrangements and corporate structures are changing 3) Customer behaviour is changing due to sharing and knowledge economy 4) Supply chains are changing (supply/demand) due to drones and radically different supply/demand relationships 5) Human concerns/values (such as education, health and other concerns) are changing. Therefore, Industrial Engineering must adapt to these changes due to its connection with these fields. This article further addresses the major forces impacting Industrial Engineering from 3 starting points: top down (macro-landscape challenge view), bottom-up (technology impact view) and middle ground (IE sub-discipline point of view).

Keywords: Industrial Engineering, Economic Forces, Social Forces, Environmental Forces, Impact Analysis

¹ The author is enrolled for an PhD (Development & Management) degree in the Department of Industrial Engineering and is a lecturer at North West University



1. INTRODUCTION

The world is changing and so is Industrial Engineering. A Coldwell Banker Richard Ellis (CBRE) report finds that "50% of occupations in corporations today will no longer exist by 2025" [1, pp.6]. Yet, this does not mean the other 50% are safe. Although they will not become obsolete, their occupations are most likely to change dramatically. The simple fact of the matter is, in the knowledge era, the inability to pick up on relevant forces shaping the context or work of a field of study can result in misalignment, outmodedness and deterioration of that field or its tools and techniques. "The next 15 years will see a revolution in how we work, and a corresponding revolution will necessarily take place on how we plan and think about workplaces" - Peter Andrew [2]. Technology is perhaps the greatest force that requires consideration. A failure to adjust and integrate technology into practice results in ineffective, minimally applicable and obsolete practices. It is important to remember, though, that technology development is often instigated by economic, social or environmental factors.

The topic of interest for this article is the occupation of Industrial Engineering. After rising to a high standard and developing capabilities that are seen as immensely valuable by industry, it would be disappointing to see that some of methodologies, tools and skills are no longer seen as useful (especially since it is possible to adapt them). The concern here is that creating something of value without the perception that it is valuable often results in similar effects as something not being valuable at all. Much like the companies that Industrial Engineers work in have contingency, human capability and risk plans; the Industrial academic and work departments have to be structured in a way that integrates and caters for all potential changes to macro environments. After all, the tools for change management and integration are being taught in those same departments; it would be unwise not to use them to do a self-analysis. In summary, in the words of Sperotto [3, pp. 8]:

Rapid changes in the social and industrial environment continue unabated, driven by accelerating technological developments, availability of information, networking, and globalization. Developments in polymers and new materials from nanotechnology, for example; the variety of energy sources; new manufacturing processes and paradigms, such as additive manufacturing and distributed manufacturing; aerial and robotic networking; neuromorphic technology and predictive intelligence; the Internet of Things (IoT); the spread of integrating techniques, technologies and systems throughout the whole supply chain and its parallel, waste flow management; the increasing value-adding applications in the service industries; and the blossoming world of virtual reality; will extend the complexity of integrated systems and change the focus of industrial engineering, the needs of industry, the nature of work, the human-machine relationship, and the culture and lifestyle of people. Inevitably, the industrial engineering profession and its institutions will need to embrace these changes in order to remain relevant and to contribute meaningfully to industry and society.

1.1 Background and Summary of Argument

The Industrial Engineering profession faces many challenges in the 21st century. Yet, one can still argue that these forces are unlikely to completely change Industrial Engineering and the need for Industrial Engineers. Especially since, in many countries, it is considered a scarce skill (8th in South Africa) [4]. However, this does not change the fact that IE can remain ahead by conducting self-assessment and improvement studies. Additionally, doing so would be keeping true to the vision and core values of Industrial Engineering. There are, however, challenges arising from specific economic, social and environmental forces. Perhaps the most pressing issue is that the 21st century is experiencing the evolution of a new macro-economic system that is better suited for the needs of society and the environment. Many have labelled this new economic system as the "knowledge economy" and the "sharing economy" [5-6].

Although one cannot neglect the immense impact of the latter 2 macro-systems on the economy, it would seem that the economy is the one where humans have the most direct control. Therefore, it makes them the preferred option to influence change and actually 'do' something. This is mainly because participating in the economy is almost fully mandatory and virtually everything seems to be linked to it in some way or another [6]. Simple day to day reference to global warming, income inequality and employment wellbeing all have deep social and environmental instigators [6]. However, they seem to attract much more attention when one finds a way to display them on a balance sheet and show their link to the economy (carbon taxes, knowledge accounting, intangible accounting, talent capital...etc).

The problem of dealing with massive technological, social and economic changes is not unique to Industrial departments. "Rapid changes inevitably destabilise established institutions, and cause misunderstandings and stresses between conservative and progressive groups and individuals" [3, pp. 8]. The beauty of the field of Industrial Engineering is that it understands "that adapting to the new economic regime would require philosophies and methods tailored to the forms of work that defined the new mode of production" [7, pp. 737]. However, technological forces of the 21st century (reinforced by social and environmental forces of awareness) have completely changed the meaning of key words like 'supply chain', 'work', 'production', 'system' and even



'value'. Arguably, though, it is a key responsibility of Industrial departments to provide the frameworks, connections and strategies to deal with such rapid changes (since there is an implicit claim to understand this complexity best and since Industrial Engineers are often in management positions and are well known for designing the systems that deal with these changes). This is also partially due the broadness of IE sub-disciplines. These sub-disciplines can be defined as: "Management Information Systems; Systems Engineering; Modelling and Simulation; Mechatronics and Automation; Robotics and Artificial Intelligence; Operations Research and Operations Management; Facilities Management and Maintenance Systems; Quality Management Systems; Project Management; Logistics and Supply Chain Management" [3, pp. 6].

The solution, however, must not simply focus on future graduates (most likely from generation Y and Z). The Industrial Engineering department has to embrace a transformative role which develops existing graduates already in industry, similar minded professions in other disciplines and interested individuals willing to progresses their holistic abilities (both privately and at universities). Yet, the most valuable network to an Industrial Engineer might be fellow Industrial Engineers (and their networks within other industries) who have mastery over some skills and can further one's own connections. The common understanding between Industrial Engineers allows for ease of communication, while the clear distinction in specialty allows for a wide influence. This also requires understanding the perception of Industrial Engineers, both in academia and in industry, on the forces changing the environmental, economic and social landscape. The overall aim of this article is to suggest developmental areas for Industrial Engineering knowledge. This will help adapt, reform, integrate and promote sustainable and healthy changes, whilst opposing, eliminating and transforming obsolete theories and technology. In short the focus of the argument in this article is as follows:

- 1) Means of production are changing due to 3D printing & additive manufacturing
- 2) Employees and corporate structures are changing, and customer behaviour is changing due to sharing and knowledge economy
- 3) Supply chains are changing (supply/demand) due to drones and radically different supply/demand Relationships
- 4) Human concerns/values (such as education, health and other concerns) are changing
- 5) Industrial Engineering is deep-rooted in above systems

: Industrial Engineering must develop to adapt to these changes

Section 2 review the impact of major forces in the economic, social and environmental spheres. Section 3 further explores a specific technology (additive manufacturing) as a case study to show how specific research projects can contribute to adapting to these changes. Lastly, section 4 aims to classify the challenges into relevant IE disciplines.

2. MACRO CHALLENGES ON THE INDUSTRIAL ENGINEERING DISCIPLINE

2.1 Economic Specific Challenges

The rise of drone delivery, 3D printing, EBay, Airbnb, Fiverr and even Uber will challenge every conceivable industry. This is not only because they add complexity to the network, but also because they change the rules of the game with regards to supply and demand. This has led to the rise of supply networks (example shown in Fig. 1) which "should visually depict all the linkages between buyers and suppliers throughout the world" [8]. For example, large organizations offering services and products have very high overheads with shareholder expectations as well as big boss bonuses built into their pricing might be replaced by decentralized producers. The local 3D printer/Uber driver/Airbnb room simply have their capital repayment of their hardware, living costs and some savings built in. Additionally, there might be an interpersonal relationship between the producer and supplier of goods and services. Yet, they are producing the same product or offering virtually the same core service. In fact, these applications are going a step further in truly answering the customers need because they are offering only what the customer is willing to pay for. This in itself is an excellent display of value offering design, minimum viable product thinking and system thinking. The sharing economy is a result of complex socioeconomic forces with some very positive environmental effects (due to less resources being used and more being shared). Part of the forces shaping this new idea are a disillusionment with consumption, need to connect, requirement for more specialized/customized service and a lack of opportunity in the mainstream economy [1]. Additionally, a CBRE report notes that "a growing proportion of jobs in the future will require creative intelligence, social and emotional intelligence and the ability to leverage artificial intelligence. And for most people that will be a route to happiness and fulfilment." [1, pp.33].





Figure 1: Supply Chain to Supply Network [8-10]

Additionally, the sharing economy seems determined to eliminate traditional hierarchical structures. It is also commonly referred to as 'collaborative consumption', "which is a trending business concept that highlights the ability (and perhaps the preference) for individuals to rent or borrow goods rather than buy and own them" [11]. Part of the reasons this shift is occurring are that traditional capitalistic and consumption models (idea that 'more' consumption holds the solution) did not create value for the overwhelming majority of agents participating in them [12]. For some, it showed a shift towards an inhumane way of living, especially since a large part of the assumptions of the previous model depend on less knowledgeable consumers who do not question as much. Additionally, the previous model resulted in more than half of the world's wealth being owned by 1% of the population [13]. Even though one might debate whether humanity has reached the sharing economy stage, it is commonly agreed that the knowledge, experience and ideas. More often than not, this content has been freely available on the internet. The knowledge economy recognizes 'intellectual capital' and with that change "a significant part of a company's value may consist of intangible assets, such as the value of its workers' knowledge (intellectual capital)" [14]. The fact that more knowledge is freely available and attainable will naturally lead to more wealth distribution since wealth is no longer viewed merely as capital assets.

2.2 Social Specific Challenges

The concern for human well-being, health, ethics and culture is not just a passing concern in Industrial Engineering. It is a fundamental pillar of the engineering profession [15-16]. In the words of Nelson Mandela "like the gardener, a leader must take responsibility for what he cultivates" [17]. Engineers, in general, face a similar responsibility especially since the products, services, technologies and processes they design affect virtually every human being on earth [16]. Industrial Engineering are especially responsible since most of the time they are specifically responsible for the designs of the interfaces with society at large of engineering products (or at the very least, they are responsible for coordinating efforts with those who do). This is what makes humanity/society the second anchor of IE focus [15-16].

Industrial Engineering is not only capable of providing relevant approaches to managing the facilities, infrastructure and systems governing urban life; it can setup the collaborative approaches, adoption strategies and decentralization necessary to avert unhealthy effects. One of the key specialties in Industrial Engineering is the ability to adapt to scaling up and scaling down. This is why Industrials should naturally work with city planners to "modify the design and use of our cities [and] develop approaches to allow us to live in high-density ''urban villages'' separated by parklands, recreation facilities and garden plots, and connected by light-rail transport" [18, pp.1123]. In addition to traditional IE tasks, several vigilant IE postgraduates and professionals have noticed the immense benefits that IE can have in developing the systems necessary for governance and providing services (especially health and education). Uzosoy in Christensen discusses that "like industrial supply chains, the health care 'supply chain' consists of multiple independent agents, such as insurance companies, hospitals, doctors, employers and regulatory agencies" [19, pp.143] and "the task is complicated because demand for service is determined by both available technology and financial considerations (such as whether or not certain treatments are covered)" [19, pp.143].

Government services (education, health and voting) face similar challenges and have had some benefit from Industrial Engineering intervention [20]. Although IE is still in very early stages of intervention in the public sector, one can only wonder how much benefit can arise from IE using systems principles to restructure government services to be more transparent, effective and consistent [20]. The most relevant point here is that humans naturally interact with systems. There is no doubt these systems are complex, but IE tools can extract



relevant factors. Additionally, they can participate in designing systems which best reflect the virtues of its agents. Whether this system is education, healthcare or something else is not the issue. Industrial Engineers will naturally work with the available professionals and specialists in that field to develop a system which answers a principle need [15]. Part of answering a need requires Industrial Engineers to review some theory from industrial psychology and organizational behaviour. The increasing connectedness of the world has resulted in people's voice being better heard [8]. Social media is a great example of how each person can create an online presence which represents them as an agent [8]. The communication between these agents, their exchange of culture and mail has created an effect that converted this once disperse world into a small village. Another effect of this is rapid urbanization as more people try to find more opportunity, interact with similar minded people and experience new things as they setup their lives [1].

The most pressing issue, however, remains that modern human psyche is different and the impact of perception greater. Part of this lies in the fact that "evolutionary emergence of advanced symbolizing capacity enabled humans to transcend the dictates of their immediate environment and made them unique in their power to shape their life circumstances and the courses their lives take. In this conception, people are contributors to their life circumstances, not just products of them" [21, pp.164]. The prime role of an Industrial Engineer, with regards to the above, is aligning perception with reality. An additional focus can be on finding a way to "balance between the individual and collective aspirations" [15, pp.6]. Lastly, perhaps the Industrial Engineer should adopt a clear focus on application. There is no shortage of people creating knowledge/technologies, so Industrials would be perceived better if they focused on applying it for the benefit of society.

A good place to start benefiting society is radically changing the idea of work. When asked about work, the perception of an 8-5 workday comes to mind (with 1-2 hours each side for traffic) [1-2]. This is unfortunate and perhaps the reason flexible hours and deliverable based pay work systems came to be. The idea of work was largely developed to suit the needs of a previous century where it was believed that the 8-5 model provided the best productivity and health. When this pre-21st century idea is applied in the 21st century, family relations are strained (since working parents have less time to interact), time is actually wasted at work (due to an imbalance between pay and expectation) and so on [1]. Thus, overall productivity is lower and the perception is self-defeating. However, work, in the broadest sense, is a physical or mental effort directed towards the production or accomplishment of something. Due to the fact that this creates value, one can naturally expect a share of the value in the form of a salary, benefits...etc. Ironically, even in the definition, physics arguably reaches better results than the perception of fixed hours. The definition in physics is the transfer of energy from one system to another. If IE works on converting this definition to accurately account for work/energy coming from individuals and collectives, then Industrial Engineers will have something better to base a "work" theory on (which will determine remuneration).

2.3 Environmental Specific Challenges

The 21st century has seen a rise in environmental regulation. Interestingly, this is only partially due to government efforts. The devastating cycle of consumption has lead scientists, sociologists, engineers and others to seek alternative technologies. Customers all around the world are becoming more aware of the effects of their actions and consumer choices [15]. Additionally, a larger focus is being placed on sustainability, preservation and conversation of the resources that allow us to live our lifestyles. Although some of these technologies have existed for a long time, the reason they fall under the environmental section is because the rising interest in them as a true alternative only grew in response to environmental problems [15]. The technologies that will be discussed share a common idea in that they aim to be wasteless (in doing so, they are also sustainable and environmentally friendly). For Industrials, a wasteless process is an ideal process. Theoretically it is impossible to produce no waste at all; yet, when one studies nature, one finds that the idea is more about creating waste which is usable by another system as a resource.

Environmentally-driven technologies are striving to provide the closest thing to wastelessness and symbiotic relationships humanity has known. They might hold the key to a healthier environment with less polluting chemicals and better sustainability [12]. If an Industrial Engineer is asked what an ideal manufacturing process would look like, there is no doubt that the definition will include eliminating the waste. Additive manufacturing and 3D printing not only do this, but they also simplify a production process which used to take 12 machines in a 100m line into a 3 by 3m build area. These developments are constantly brought to light in the media world. However, there is a need to understand how additive manufacturing affects the entire world around us. Beyond reviewing what techniques are available and what design processes are used today, a deeper understanding must be developed on how these developments affect our supply chains, facilities, customizability, quality of and even the way companies will design our products. The rapid destruction of our natural resources prompted a revision of our energy production [12]. Some countries were fast to convert fully to alternative energy while others remain sceptic. Whether there is direct links to global warming and natural catastrophes, it is no secret that the chemicals involved and created by non-renewables are harmful to the environment and everything living in it, both directly and indirectly. It also simplifies the energy conversion process immensely (from raw potential energy into usable energy). Alternative energy technologies include geothermal, solar, wind and various others. Yet, the problem does not seem to be with the technology, but rather it seems to be an inability to see the true financial benefit of alternative energy. Industrial Engineers naturally need to take a proactive part in resolving



this problem by utilizing skills learnt in management accounting, engineering economics and other courses to accurately reflect the financial and non-financial benefits of alternative energy to companies and decision makers [15]. Accounting for the right things might lead to very different decisions and will facilitate a swifter transition to more sustainable energies.

Energy and products are definitely important but they are not nearly as important and vital as food to human survival. Food security is a growing concern across the globe. "Eating is the most important act of human existence" - Bob Cat [22]. The ability of the planet to produce food sustainably for 7 billion people has decreased due to environmental decay [12]. What adds more complexity to the issue of the food supply chain is the nature of its variability in the supply demand relationship. Different cultures have different diets and different people demand different types of food daily (which is perhaps one of the most unpredictable sources of demand). The question here is how does one provide demand locally (to ensure high quality), whilst ensuring rapid response times to demand (keeping low stock which also increases quality), and ensuring that minimal resources are used in the process?

Hydroponic and aeroponic faming might provide the answer since they utilize a water or air based medium to give plants the exact nutrients the need [22-23]. This is often done in a controlled environment, requires virtually no soil and produces high quality good consistently due to the lack of severe environmental changes typically encountered in normal farming [23]. The idea itself was largely conceived by recognizing that soil is merely a transfer agent. The actual nutrients are what is really required by the plants (differentiating perceived need from true need). Additionally, business cases can be made for both small scale decentralized production and large scale centralized production.

Sustainable food cycles recognise and utilize the waste of one organism to feed another until a complete cycle is achieved (forming symbiotic relationships) [24]. This naturally occurs in nature, but restaurants and businesses are starting to notice its positive impacts. A great example of this is a symbiotic relationship between beer, fish and plants which several restaurants have taken advantage of on [24]. The fish produce waste used to grow the beer, the fish eat some of the vegetation to grow and both of them are served at a restaurant which provides the funds to run the entire operation [24]. This shows that these methods can be applied to the business environment as well to create more sustainable and healthy competition less focused on the 'dog eat dog' metaphor. Drawing parallels between the natural world and business world can create for some very interesting answers.

3. TECHNOLOGY IMPACT REVIEW: ADDITIVE MANUFACTURING AS A CASE STUDY

During the past decade humanity saw the rise of consumer electronics due to the rise of the micro-processor. Moore's law predicted that processing power will double every 1-2 years with components becoming more affordable, reliable, usable and widespread; and he was correct. However, a new buzzword has hit the technology world: Additive Manufacturing. Additive manufacturing is a process by which components are built, altered or repaired using a variety of processes [25-27]. This concept mainly focusses on the addition of material rather than subtraction; thus creating less waste whilst saving time, money and ensuring top quality. Processes vary from laser manufacturing to cold spraying. Some processes (mainly known as 3D printing) only require heating and placing certain polymers into the preferred form. However, one thing is clear from the scientific literature and technology markets: Additive Manufacturing is here to stay [25, 26]. Additive manufacturing by nature is a superior process capable of very advanced designs. In the past 3-4 years there has been a trend developing within this technology and it is expected that the next decade will see the same effect as the processor (shown in Fig. 2).

Printing a pizza at your local shop or having an old piston rebuilt like new with metal powder were topics mentioned solely in science fiction only a couple of years ago. However, developments in the additive manufacturing world have led to a new production method which is not restricted by tolerances, geometries or complexities [25, 26]. Although somewhat expensive at this point in time, it is argued that additive manufacturing will be the way we produce any product in the future [26]. Whether it be our food, metallic service parts or even our organs; scientists have found a way to use additive manufacturing techniques to use raw powders to create an endless possibility of products [26]. These developments are constantly brought to light in the media world, however, there is a need to understand how additive manufacturing affects the entire world around us. Beyond reviewing what techniques are available and what design processes are used today, a deeper understanding must be developed on how these developments affect our supply chains, quality of our products and even the way companies will design our products [26].

With any growth of this kind it is natural to see problem areas such as: inefficiency, quality, process and other multidimensional problems. Since this technology deals with aspects of man, money and machine; it is vital that an Industrial Engineering perspective is developed to help save millions in future and assist in advancing this fantastic technology. Additionally, the simplified supply chain and facilities can be a research area on their own within IE. Perhaps due to the smaller facility sizes and improved power efficiency, alternative energy can be used.



Additive Manufacturing Adoption Timeline



Figure 2: Growth and Maturity of Additive Manufacturing [25]

3.1 Potential IE Research Topics in Response to Additive Manufacturing Technology

3.1.1 Quality Standards & Decision Support Data for Additive Manufacturing

A potential idea is to create a universal test jig/board, using quality management and design of experiment tools, to be able compare different additive manufacturing technologies to each other and thus be able to understand to what industry they can be used in. The aim is to test one powder material at a time on the base given below with different scenarios such as different base materials and conditions to create a variety of shapes and figures designed to test the full capabilities of the technology.

Each technology will be compared on the basis of 3 aspects: Cost, Quality and Time. Each of these aspects has sub qualities indicated in the table below to show the exact area where technologies surpass each other. Such a deliverable will provide a very good platform for analysing the different technologies. Different "boards" can be developed by experts for different test fields and material however the concept must test a variety of operating conditions, materials and builds while ensuring that accurate data is recorded about the time, cost and quality. This can help create a way to compare apples with apples. Overall this would provide an excellent platform for companies to test machines, technologies, materials or suppliers. This will give them an excellent test before commencing production with that specific supplier and will ensure that their criteria are met. A visual representation is shown in Fig. 3.



Figure 3: Universal Additive Technology Test Jig/Boards [Own Work]



3.1.2 Production and Manufacturing Requirements & Specifications

From this above it can easily be seen how specific requirements can arise for different industries. Industries will most likely be grouped by product size and complexity; changing the way in which production and manufacturing requirements are classified and communicated (shown in Fig. 4 below).



Figure 4: Production and Manufacturing Industry Classification [Own Work]

3.1.3 Wasteless Design Methodology

The development of this new hardware will naturally require a new industrial design methodology. The design processes available today are somewhat lacking if we try to apply them to additive manufacturing. The main problem is that they do not take the full spectrum of possibility when it comes to additive manufacturing. There are modern techniques on how to define and design a product which allow rapid prototyping and testing. In addition to this most design processes do not consider the ability to refine designs in real time. Additive manufacturing allows easy additions to designs as well as reconstructions. There is much less waste involved and designs are modified relatively quickly. Even manufacturing errors can be corrected easily without creating unnecessary waste. If the product has already reached the customer, a simply process can allow them to get their product fixed on site without scrapping or returning the product. Products can be upgraded in this manner as well. This is why a define-design-refine and manufacture technique should be considered when designing products for the 21st century.



3.1.4 Supply Chains, Facilities Planning and other Considerations

In addition to having immense impact on our products, the additive manufacturing technique has major impacts on the way we source our products, produce and store raw material (powder in this case) and even on the facilities that will produce products. Some argue this might be decentralized but some products will have to be proceed in centralized facilities (for intellectual property, legal and health reasons). Designing a future manufacturing facility, as well as the considerations that need to be taken, can form the basis of many studies. These facilities can be decentralized or centralized, multi-product group focused or single product group focused, produced on order or customized from stock. More importantly, traditional location models, production scheduling, inventory management (which has to adapt to powder based inventory) would have to adapt massively to these changes.

4. IMPACT ON SPECIFIC INDUSTRIAL ENGINEERING SUB DISCIPLINES

Without context on where the literature and case study (from sections 2 and 3) fit within the broader structure of the Industrial Engineering body of knowledge; several vital questions and connections can be missed or overlooked. This section explores the impact of some of the developments and arguments discussed above on specific sub disciplines of Industrial Engineering in dialogue format.

System Engineering/ Project Management: It would seem the forces above will definitely result in more and more projects becoming available to Industrial Engineers. Although some of them will be large scale, it is likely that a large proportion will be small scale development and management projects. However, traditional project management and system engineering tools and methodologies simply do not cater for these types of projects. Industrial Engineering must be proactive in finding ways to distil the core of the theories and tools and develop appropriate module based models to assist in implementing successful project management and system



engineering regardless of scale. The focus, then, will be on adapting theory to the specific scopes and requirements of each respective project.

Logistics and Supply Chain: The fact that most additive technologies are powder/particle /nutrient based will most likely radically change the technology, vehicles and methods of warehousing and delivery of products. The biggest question with regards to the above is: will supply chains be reduced to a simple 5 step process that: 1) converts ores/raw materials into powers 2) transfers them to appropriate warehousing facilities for storage 3) delivers them to relevant AM technologies for conversion to parts 4) assembles them into final products and 5) ships them to clients? Will returns management be a corrective process where a 3D printer is taken to client site to correct product? Is the carbon footprint for this activity significantly less than existing supply chains? The interesting thing here is hydroponic farming is very similar to additive technologies in that it simply adds what is required by the plant (natural machine in this case) in order to produce the final product. Does this make the future of everything we eat and use (accounting for most products) additive based?

Information System Design: Advances in payment, booking and tracking technologies is forcing traditional information systems out of their comfort zones due to a significant difference in user interaction available (arising due to new supply-demand relationships). Additionally, the sheer amount of data required to understand this complexity is engulfing. Yet, the widespread use of smartphones as a data collecting tool is becoming a reality. The concerning questions are: how will the data be collected? How can it be stored effectively? How does one best make use of this data for competitiveness? And, most importantly, how does this data transition to information and then knowledge and perhaps artificial intelligence in the near future? The customer is most likely going to want answers to these questions especially because of rising concerns on privacy.

Simulation/Automation: The ability to simulate this uncertainty and adapt automation technologies to react appropriately is most definitely a challenge. With so many factors to test for sensitivity, is it even practical to conduct simulations? Or is there a way to simplify the chaos into specific agent based characteristics? Can this be done proactively or is it destined to be reactive? Additionally, simulations models will most likely have to develop a way of simulating generational human culture and ethics, especially if this sub-field hopes to create convincing models/arguments for government services.

Business Engineering: With the rise of the sharing economy comes the reality that many business engineering tools will have to be restructured to cater for the individual instead of the corporate structure. The biggest impact of the forces above is that money that is used to flow to traditional business hierarchies will now go directly to individuals. Thus, traditional strategy development, process modelling and business models need to adapt. The question is: does each individual then become his own business or do they follow the business model of the app which brings them customers? If they are unique and are not app reliant, how will these individual 'businesses' differentiate their processes/value offering? Will the sharing economy find symbiotic relationships with the traditional economy or remain competitive?

Operations Research: The algorithms in the code of every application are constantly determining the most effective 'route', 'producer' and more broadly 'connection'. These optimization algorithms are essentially the decision makers. Yet, with this continuously changing environment, a static decision maker will result in tremendous waste. Dynamic decision making and optimization is the most likely the future of operations research. Whether it is a que, production or infrastructure problem: optimality is based on dynamic factors and thus will have a dynamic answer. The question remains, however: is it possible to truly get the optimal answer in these scenarios or will optimization radically change to accept certain confined area as 'optimal'? What if this confined area radically changes in the solution space? Does that mean the introduction of sensitivity as a core element in operations research problems?

Operations/ Facilities Management: The sharing economy will perhaps more accurately reveal the meaning of the word 'facility' and 'operation'. In a nutshell, the facility will become the encompassing structure that facilitates business operations to take place. People are most likely to convert their garages, gardens and bedrooms to facilities (production, servicing and other). However, with this comes massive facilities designs and safety regulation problems. Are there realistic methods to properly oversee such operations?

Quality Management: Perhaps the biggest question for quality management going forward into the knowledge economy is how one assesses the quality of intangible assets (mainly knowledge, human capability and talent). As for the sharing economy, will new standards for decentralized producers come to existence? Will perception play a bigger role in quality? Or will technology facilitate a way to create more accurate quantitative measures for perception?

General IE: The sharing economy and knowledge economy are most definitely challenging legal and business regulations both nationally and internationally. The simple fact of the matter is they create interactions and assets that governments and lawmakers are not traditionally accustom to deal with or account for. More worryingly though, some sharing economy transactions are not even trackable by most governments and some knowledge assets require a whole set of new depreciation and taxation laws.



5. CONCLUSION

It would seem, at some level, that adopting and adapting to even a few macro ESE trends will naturally lead to the others. Maybe this is because once a new ideology of wastelessness, equality and humanity comes into operation they indicate and make us realize (individuals and collectives) how bad the previous ideology was; which can naturally result in a continuous cycle of reconsideration and improvement. There is no doubt in the Industrial Engineers ability to deal with economic changes. However, environmental and social changes are unfortunately more difficult to adapt to and require an authentic focus. This paper recommends, for example, that Industrial Engineers naturally need to take a proactive stance in resolving environmental problem by utilizing skills learnt in management accounting, engineering economics and other sub-disciplines to accurately reflect the financial and non-financial benefits of alternative energy to companies and decision makers. Accounting for the right things might lead to very different decisions and will facilitate a swifter transition to more sustainable energies. Additionally, although there is no doubt that humanitarian engineering projects exist, yet, they represent a minority of projects. This is not to say that projects don't take human concerns into consideration, but, it does show that most projects have a different central focus: money, enormity or competition. Over time this can lead to human factors being disregarded and left to satisfy the minimum. Industrial Engineering should be the champion of the humanitarian engineering movement. After all, the roots of IE are so intertwined with workers, their rights, their health and balancing or superseding it over financial considerations.

6. **REFERENCES**

- [1] CBRE. 2016. Fast Forward 2030. [cited 2016 24 April]; Available from: http://www.cbre.com/~/media/images/research%20reports/apac/2015/cbre_genesis_fast_forward_workpl ace_2030_full_report_e.pdf [Acessed 24 April 2016].
- [2] CIPD. 2016. Half of current occupations won't exist by 2025, finds report. People Management Magazine Online. Available from: http://www.cipd.co.uk/pm/peoplemanagement/b/weblog/archive/2014/11/11/half-current-occupationswon-t-exist-by-2025-finds-report.aspx [Acessed 24 April 2016].
- [3] **Sperotto, Sperotto, F.** 2015. The Development of the Industrial Engineering Profession in South Africa, South African Journal of Industrial Engineering. The South African Journal of Industrial Engineering, 26(2), p.1-9.
- [4] INSETA. 2014. National Scarce Skills List: Top 100 Occupations In Demand, H.E. Training, Editor. South African Government: Pretoria.
- [5] Powell, W.W. and Snellman, K. 2004. The knowledge economy. Annual review of sociology, p.199-220.
- [6] Hamari, J., Sjöklint, M. and Ukkonen, A. 2015. The sharing economy: Why people participate in collaborative consumption. Journal of the Association for Information Science and Technology.
- [7] Bailey, D.E. and Barley, S.R. 2005. *Return to work: Toward post-industrial engineering*. IIE Transactions, 37(8), p.737-752.
- [8] Bellini, J. 2015. Why Supply Chains Should Be More Socially Engaged Supply Chain. Available from: http://www.supplychain247.com/article/why _supply_chains_should_be_more_socially_engaged [Acessed 1 May 2016].
- [9] Ferrari-Group. 2014. Supply Chain 3.0, Sherman-SSN3_0.png, Editor. Available from: http://www.theferrarigroup.com/supply-chain-matters/2014/04/07/supply-chain-matters-guest-contribution-web-3-0-enables-the-smart-supply-chain-network/ [Acessed 1 May 2016].
- [10] **Cmuscm**. 2014. *Coca Cola and its Supply Chain System*. Available from: http://cmuscm.blogspot.co.za/2014/01/as-personwho-is-new-to-field-of-supply.html [Acessed 1 May 2016].
- [11] SearchCIO. 2016. What is the sharing economy?. Available from: http://searchcio.techtarget.com/definition/sharing-economy. [Acessed 24 April 2016].
- [12] **Project S.o.S.** 2007. Annie Leonard in Story of Stuff, Free Range Studios.
- [13] Walker, J. 2015. Richest 1 Percent To Own More Than Half Of The World's Wealth By 2016, Oxfam Finds, in The Huffington Post.
- [14] Investopedia. 2015. Knowledge Economy. Available from: http://www.investopedia.com/terms/k/knowledge-economy.asp [Acessed 24 April 2016].
- [15] Sperotto, F. 1994. In the footsteps of homo industrialis, Johannesburg: Picsie Books.
- [16] Mandela, N. 2013. Long Walk To Freedom, Little, Brown Book Group.
- [17] **EWB.** 2015. Engineering for Humanity: What Does It Mean to Be a Humanitarian Engineer?, Engineers Without Borders Australia. Available from: http://www.ewb.org.au/announcements/2/11414 [Acessed 7 April 2016].
- [18] McMichael, A.J. 2000. The urban environment and health in a world of increasing globalization: issues for developing countries. Bulletin of the World Health Organization, p. 1117-1126.
- [19] Christensen, C.M. 2005. Building a Better Delivery System: A New Engineering/Health Care Partnership, National Academies Press.



- [20] Schutte, C., Kennon, D., & Bam, W. 2016. The Status and Challenges of Industrial Engineering in South Africa. The South African Journal of Industrial Engineering, 27(1), p.1-19.
- [21] Bandura, A. 2006. Toward a psychology of human agency, Perspectives on psychological science.
- [22] Cat, B. 2015. *Eating is the most important act of human existence*, in Ted X. Available from: https://www.youtube.com/watch?v=310--GIH2mw [Acessed 1 September 2016].
- [23] **Resh, H.M.** 2002. Hydroponic Food Production: A Definitive Guidebook for the Advanced Home Gardener and the Commercial Hydroponic Grower, Sixth Edition, Taylor & Francis.
- [24] Parks, J. 2012. The Plant: My Beer Feeds Your Fish, in Green Minute. Available from: https://www.youtube.com/watch?v=zMBxJTQqnRI [Acessed 1 September 2016].
- [25] Forbes. 2016. Forbes.com. Available at: http://www.forbes.com/sites/louiscolumbus/2015/03/31/2015roundup-of-3d-printing-market-forecasts-and-estimates/#35dbc4bf1dc6 [Acessed 7 May 2016].
- [26] Economist. 2012. A third industrial revolution, The Economist, no. 403, pp. 1-14.
- [27] Sherman, E. 2016. *Materializing the Third Industrial Revolution Oerlikon Blog Without limits*, Oerlikon Blog Without limits. Available at: https://www.oerlikon.com/stories/2016/07/21/materializing-the-third-industrial-revolution [Acessed 1 September 2016].



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



BENCHMARK INDICATORS FOR COMPETITIVENESS ANALYSIS OF TIER 2 AUTOMOTIVE INDUSTRY

Opeyeolu Timothy Laseinde¹ & Mukondeleli Grace Kanakana²

¹Department for Industrial Engineering, Tshwane University of Technology, Pretoria <u>laseindeo@tut.ac.za</u>

² Department for Industrial Engineering, Tshwane University of Technology, Pretoria <u>kanakanamg@tut.ac.za</u>

ABSTRACT

The importance of micro economic drivers is well established globally, and leading economies are largely driven by small and medium scale enterprises. The paper seeks to determine major internal benchmarks for determining competitiveness of tier 2 manufacturers who in turn supply automotive components to tier 1 enterprises, who are the only approved suppliers recognized by Original Equipment Manufacturers (OEM). This is against the backdrop that the automotive industry contributes largely to South Africa's GDP and as such, its lower tier need to be strengthened. The study sampled tier 2 automotive component suppliers and initial competitiveness assessments were conducted. The relevance and suitability of different competitiveness benchmarking parameters were revealed based on tested Key Performance Indicators (KPI). To achieve higher competitiveness, lower tier automotive component manufacturers are advised to promote workplace cooperation, improve employee working conditions and apply productivity improvement systems that will boost efficiency and improve product quality.

KEYWORDS:

Benchmark, Indicators, Competitiveness, Automotive, South Africa

¹ The author is enrolled for a PhD (Mechanical) degree in the Department of Engineering Management, North West University, South Africa.



1. INTRODUCTION

From a retrospective standpoint, some large international firms currently notable as industrial giants in the context of competitiveness, all started out as Small, Medium and Micro Enterprises (SMMEs), and these were able to survive massive technological changes during the industrial revolution. The technological changes pruned volumes of enterprises off the radar, thereby creating opportunities for new entrants. However, the manufacturing space is fast changing as the world is not only contending with disruptive manufacturing innovation according to Campbell et al. [1], but also, it is faced with the unfavourable outcomes of globalization, resulting in product divergence and market convergence [2]. Imports are major bane to growth of local enterprises in some countries with unfavourable policies affecting SMMEs [3] [4] as witnessed in most developing nations. Nevertheless, quality of products and service delivery by local manufacturers has been found wanting, within the lower tier automotive component value chain. Accordingly, larger enterprises that depend on local suppliers alternately resolve in sourcing components from suppliers that guarantee guality, less rejects and better lead times, even when products cost more to acquire than local substitutes. Furthermore, some large manufacturers prefer sourcing components from affiliate enterprises abroad. The effect this has on a nation's economic growth cannot be overlooked, which calls for proactive measures that can effectively boost competitiveness of home grown industries. Enterprise Improvement (EI) is essential at this point in time because for the SMMEs in the automotive industry to be in existence and be entities to reckon with in future, they must become competitive locally and progressively work towards global relevance.

The study focuses on internal benchmarking indicators which tend to rank the competitiveness level of enterprises using multiple indicative matrix. It aims at identifying areas which need attention, in order to build a competitive industry. Institutions and agencies charged with the responsible of improving local production will therefore have empirical data to base decisions upon, when focusing on productivity improvement initiatives within this industry segment.

Numerous studies have been carried out at higher levels involving tier 1 enterprises and Original Equipment Manufacturers (OEMs); however, little information exists at the tier 2 enterprise level which are more often smaller in terms of product volumes, structures and workforce, and the industry conformance standard (according to ISO demands), are less demanding in comparison to higher level enterprises. Indicators adopted in other countries or other industries could have easily been copied and adopted directly for competitiveness evaluation; however, the results might then be skewed because of a lack of relevance in the context of the industry where it has to be tested based on employment act differentials and economic variances.

This is the pilot phase of a larger Enterprise Improvement (EI) project and as such, this paper focuses specifically on the identification of relevant benchmarking indicators which can be adopted on a larger scale for Enterprise Improvement (EI) projects within this industry segment.

2. REVIEW OF LITERATURE

Competitiveness of an industry can be viewed from various perspectives depending on the aims and objectives of the evaluation. Industry competitiveness is also attributed to economical competitiveness of nations based on further classification matrix. According to Saranga [5], the indicative elements worth considering for classification of competitiveness grouped by Durand and Giorno [6] are: (1) financial indicators such as price, exchange rate, Return on Investment (ROI) and Return on Assets (ROA) (2) Non-financial indicators such as customer satisfaction, market share, inventory, market share growth, sales volume, sales growth, quality measures, productivity, lead time, performance to schedule and preventive maintenance measures.

Enterprise improvement is believed to yield higher productivity and in turn, competitiveness. Competitiveness Key Performance Indicators (KPIs) as explained by the World Economic Forum (WEF), are the set of institutions, policies, and factors which determine productivity level of a country. Equally, success and economic prosperity of a nation is attributed to the level of productivity of enterprises, generating wealth from internal production within, and not from its externally generated revenue from branch outlets abroad [7].

Productivity is a major factor considered when measuring competitiveness as identified from literature. Major components identified in literature are: (1) Current state of manufacturing and complementing infrastructures (2) ability to innovate (3) macro-economic environment (4) training structure especially at the higher education level (5) institutional performance (6) efficiency of Labour market (7) market efficiency (8) financial market inclusion (9) size of market (10) technological readiness (11) level of sophistication of businesses (12) accessibility to health care (13) accessibility to primary education [8][9][10][11]. These factors all complement productivity but they are not always used in a single evaluation due to factors peculiar to various industries. Furthermore, some of these factors are influenced by other external measures. For instance, automotive component manufacturers compulsorily require to operate in global value chains to remain competitive in Korea and Malaysia [12], which is not the case in South Africa's tier 2 automotive industry. According to bargaining agreements unique to each industry in every country, labour cost varies across countries which can be linked to the level of prevailing economic dynamics. Just as oil prices are regulated at international levels, so also pricing is habitually determined at international levels for homogenous standardized components, produced across continents. Lack of comprehensive bilateral price data however, affects the pricing policy which leads to uncertainties and



subjective claims, according to [13]. As such, it is important to consider export prices, logistics cost and labour cost [13] when determining competitiveness. The level of Research and Development (R&D) also determines growth of an industry. Based on a two-way fixed effect model for estimation of a firm's growth [14], industry growth level was linked to percentage of sales revenue channeled into Research and Development (R&D) and existing technology utilized, as against foreign equity participation and import of capital goods. Another important factor worth considering is service delivery time. On time service delivery is generally seen as higher lead times, which is attributed to time based competitiveness [15] [16]. The ability of a manufacturing system to respond to customer demand promptly shows the effectiveness of internal systems and processes so long quality of products is maintained. The lead time is critical to business success as amplified in international levels [17].

3. METHODOLOGY

The use of appropriate feedback mechanisms greatly contributes to information harnessing in industry competitiveness assessment. To adequately identify the views of various levels of the workforce within the enterprises, instruments for employees were different from those of employers/management. The study was approached with this understanding and as such, primary data was harnessed using observation sheets, employee evaluation forms and standard questionnaires with KPIs for a benchmarking enterprise baseline. The questions were designed to sufficiently provide information that will guide in taking informed decisions regarding strategy adoption, for improving the automotive industries competitiveness level. The questionnaire identified gaps relating to enterprise productivity, workstation environment, production systems, Health Safety & Environment (HSE) practices, resource allocation, resource planning, quality management system, supply chain, visual management, suggestion schemes, problem solving techniques, research & development, workplace cooperation amongst others. It was also designed to identify existing practices, which make some of the enterprises competitive in a bid to learn from them and promote same among others. The tools adopted were developed for consistency of result analysis and this guaranteed effective communication of KPIs during the project. The project was in 3 phases as presented in figure 1.



Figure 1: Project implementation plan

For the pilot phase of the study, the 3 tools were tested in 15 enterprises while they were applied in 5 enterprises during the baseline study. Feedback was received from shop floor employees, shop stewards, supervisors, middle managers and some management staff of the evaluated enterprises, according to accessibility of stakeholders. The evaluation instruments (observation sheet) and employee feedback form were designed using the Likert scale of measurement. Relative weight was important due to the large population of respondents. Data collected at the baseline assessment phase was mainly to correlate accuracy of initial data during the needs assessment. Due to multiple instruments utilized, the collected data were carefully matched for balanced result. The observation sheet results obtained from the interviewers were normalized and analyzed for accuracy and consistency, and were further averaged. The initial assessment had 73 structured questions which covered all internal competitiveness areas articulated from literature. Some critical competitiveness indicators that are externally influenced were not included in the study, such as Research & Development (R&D), which is hypothetically measured through the level of research outputs and R&D infrastructures existing within a country and around the region.

Effort was made to link literature findings pertaining competitiveness benchmarking indicators at this level with determining factors considered globally. The focus was on developing internal productivity enhancers, through improvement of soft skills and to some extent, hard skills needed for day to day functions. This article is structured not only for academics and high level professionals, but in a format that can be easily understood by enterprise improvement teams responsible for competitiveness improvement within the tier 2 automotive component manufacturing segment. The information harnessed from the survey, informed the proposed KPIs herein referred to as benchmark indicators. During the EI project phase, a table of KPIs was issued to the



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Enterprise Improvement Teams (EIT) within the enterprises and each were at liberty to select 5 indicators from the list of 20 generic KPIs provided. The selections were weighed relative to data captured in the observation sheets and the corresponding outcomes supported the shortlisted benchmarked indicators presented in the results. Furthermore, Enterprise Improvement Plans (EIPs) were developed by each of the enterprises at the commencement of EI projects, and the resonating indicators across all enterprises validated choice of benchmark indicators.

Due to very large data obtained from the various phases of the project, a multi level criteria approach for indicator selection was utilized. Based on time constraint, averages of the various feedback averages were taken for employee assessment while average of the modal evaluation scores obtained from all interviewers using observation sheets were used and further normalized for standardization. Figure 2 shows the implementation model adopted for the study.



Figure 2: Implementation model

4. DATA ANALYSIS

The data collected was normalized using Range Equalization Method (REM) to improve its interpretation. Normalization = $\left[\frac{\text{Actual Value-Minimum Value}}{\text{Maximum Value-Minimum Value}}\right] \times 100\%$ (i)

To evaluate the collective importance of response received from each indicator, Relative Importance Value (RIV) was used for analyzing collected data. RIV describes level of importance of respondent's views, thus an index value was adopted based on the 5-point rating scale for questions and answers within the questionnaire [18]. The measure RIV for each individual competitiveness indicator was obtained from the weighted average using the survey data through the following formula: $RIV = 100 \times \frac{\sum ax}{EN}$ (ii)

X: frequency of the responses for a specific indicator; N: total number of responses; a: the weighting value corresponding to a specific question (ranging from 1 to 5 as suggested in the questionnaire) where 1 is least acceptable (most unacceptable) or insignificant and 5 is extremely important (most acceptable) or critical. The aggregate index for each of the indicators was derived first using relevant variables (sub-indicators).

$$V_i = \frac{1}{2} \sum_{i=1}^n x_i$$

(iii)

(iv)

where, V_i is i_{th} indicator, xi is the i_{th} sub-indicator; n: number of sub indicators within the indicators. The aggregating of all the indicators into competitiveness index for a firm in the automobile industry.

$$C_{j} = \frac{\sum_{i=1}^{n} W_{i} V_{i}}{\sum_{i=1}^{n} W_{i}}$$

where, C_j is the competitiveness index of j_{th} firm, W_i is weight of the i_{th} indicator, V_i is i_{th} indicator and n is the number of indicators.

The Likert scale of measurement is usually utilized in an attempt to weigh respondent's opinions in a survey [19]. It was applied by coding the results on a cumulative basis according to scale of 1 to 5 with the following representations: strongly agree-5, agree-4, Indifferent (Neither agree nor disagree)-3, partially disagree-2, Strongly disagree-1. To effectively analyze the Likert data, the results were arranged in an ordinal measurement scale using the descriptive statistical approach for ordinal measurement scale.

For the Centre of Tendency measurements, the median and mode was utilized while the frequencies were observed for the variability measurements and this was achieved by analyzing the variables at intervals. The



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE

Likert scale items were tested for reliability at the onset and also tested for outliers and missing data. Since most of the harnessed data were normally distributed, one tailed Grubb's test was used to analyze the data to eliminate outliers as recommended by Marino and Gladyshev [20]. This was achieved using a software add-on to Microsoft Excel called "real statistics".

	Summarized evaluated parameters (measured elements)
K1	Team work
K2	Suggestion Scheme & Grievance Mechanism (policies, grievance procedures, suggestion box)
K3	Adherence to Standard Operating Procedures (SOP) and effectiveness of Process Failure Mode Effects Analysis (PFMEA)
K4	Ergonomics (working position, temperature, weight of objects hand lifted, condition of air)
K5	Error proofing utilization
K6	Calibration / validation of measurement devices as required
k7	HSE (Health Safety and Environment), HIV/AIDs awareness)
k8	Legal Compliance and workforce wellness (workers safety, welfare, minimum wage, sexual harassment prohibition, role understanding, demography mix, performance linked payment)
k9	Workforce Management (Policy related to gender equality,
k10	HSE (handling of sharp objects, exposed electrical cabling, hazardous wastes; complete use of PPE, fire prevention, building inspection; availability of potent first aid items)
k11	Planning (SWOT, order driven production planning, contingency plan, continuous improvement plan, skill development plan)
k12	Cleaner Production (TIMWOODS status, monitoring of consumption, reference to consumption norms, leakages and waste elimination)
k13	Visual Management (use of display boards during production & green area meetings, clearly defined and understood goals & targets)
k14	Communication (Daily meetings, news letters, display boards)
k15	5S (toilet hygiene, organized inventory, environmental hygiene, availability of drinking water)
k16	Customer Need & Expectation Assessment (list of customer internal & external needs and expectations)
k17	Data availability (display of trends & analysis & understanding of same by workforce)
k18	Statistical Process Control (display of analysed components on X-bar, R chart etc.; Understanding of Stability, variation, capability, over adjustment etc.)
k19	Use of Quality Tools (Pareto chart, why why analysis, fishbone diagram, histogram, analytics tools)
k20	Product development status, OEE, Inventory Turnover Ratio (ITR), Downtime/Uptime, On Time Delivery (OTD), Turn Around Time (TAT), scrap rate, rework, defects, costs savings, absenteeism, injury, near misses
k21	Level of implementation of productivity improvement tools utilized by employees (VSM, line balancing , KANBAN, KAIZEN, continuous production flow, QC tools, time-motion study)



SAIIE27 Proceedings, 25th - 27th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 3: Analyzed results ranked according to KPI Scores, showing need for intervention

4.1 Discussion of results and recommendations

The study explored preselected automotive component SMMEs and exposed internal competitiveness gaps existing within the firms during the pilot phase of the study. 60% level of conformance was set as minimum ranking upon which the indicators were considered sufficient, and any results below this mark are clear pointer for comprehensive intervention. Above 80% is the desired level according to the study matrix employed. From the results, only K20 was sufficiently above the minimum bar which shows that most elements within the selected indicators are not adequately addressed across the analyzed enterprises. From closer analysis of K20, despite being above the minimum bar, it did not meet the 80% mark which is the desired conformance level for which the indicator will not be considered. K3, K10, K11 and K12 were close to the 60% mark, and from analysis of these indicators, they were all critical conditions which must be addressed for the enterprises to meet regulatory standards (k10) and others were essential for productivity improvement. From the results, team work (k1) was rather weak and the formation of Enterprise Improvement Teams (EIT) as shown in the implementation model in Figure 2 was a right step in addressing this issue, common to most enterprises surveyed. From the study, there is need for improved communication policies through effective visual management boards and displays. Walkways also need to be clearly demarcated using approved safety colour codes. Employees should be trained and must adopt problem solving techniques especially in the shop floor. Regular green area meetings using defined meeting agenda is equally required because effective communication guarantees higher productivity from employees. Taking a closer look at the results, enterprise 5 contributed largely to the high scores, neutralizing some of the scores from lower performing firms. An in depth study of the companies showed that company 5 had to some extent, effective communication procedures which encouraged workplace cooperation even though communication policies were not documented. Team work is equally emphasized in company 5 which is lacking in the other companies studied. Furthermore, productivity is given utmost priority in this company with emphasis on TIMWOODS (Transport, Inventory, Motion, Waiting time, Overproduction, Over processing, Defects and Skills). Identified waste is constantly being investigated and soft measures are in place for reduction of such. However, emphasis on total waste elimination was not seen in some instances which was linked to cost-benefit considerations. Company 5 is found lagging in terms of legal compliance and work force wellness. Equally, the results reflect very little use of quality tools for the enterprise day to day operations. Formation of Enterprise Improvement Teams (EITs) is highly recommended because the team will serve as "change agents" for promoting required changes.

Since the project is in its pilot phase, to take the work further, a Decision Support Systems (DSS) may be considered for further analysis of the data, which is an alternative selection criteria mechanism. Additionally, the use of quality tools should be emphasized and statistical process control for line area meetings should be considered because of the potential they have in effectively communicating critical production elements. The results obtained should be compared in detail with the method adopted in this phase of the project.

5. CONCLUSION

Competitiveness of any industry requires holistic improvement of processes, management systems and all elements that promote productivity. Internal factors have to be addressed at the enterprise level while external factors have fiscal economic bearings which are controlled by government policies and global market forces. Tier 2 automotive component producers within South Africa need to address competitiveness gaps which make their products less competitive, by improving internal systems in order to meet international industry standards, thereby entering global markets since there is continual convergence of market as it is currently. The industry should equally take advantage of favorable policies already in place, which will guarantee sustainability, so long their product offerings meet internationally acceptable quality standards with zero defects. Based on the results



obtained, there are evidences to show that all the hypothetical Key Performance Indicators (KPIs) assumed at the onset of the study have to be considered.

REFERENCES

- [1] Campbell, T., Williams, C., Ivanova, O. and Garrett, B. 2011. Could 3D printing change the world. Technologies, Potential, and Implications of Additive Manufacturing, Atlantic Council, Washington, DC.
- [2] Ford, D., 2011. IMP and service-dominant logic: Divergence, convergence and development. Industrial Marketing Management, 40(2), pp.231-239.
- [3] Holmlund, M., Kock, S. and Vanyushyn, V. 2007. Small and medium-sized enterprises' internationalization and the influence of importing on exporting. International Small Business Journal, 25(5), pp.459-477.
- [4] Von Blottnitz, M. 2009. An Economic Overview of SMMEs in 2008. The Small Business Monitor, 5(1), pp.4-11.
- [5] Saranga, H. 2010. Competitiveness of the Indian auto component industry, An empirical study, chapter 5, JRP. http://www.ide.go.jp/English/Publish/Download/Jrp/pdf/154_ch5.pdf
- [6] Durand, M. and Giorno, C. 1987. Indicators of international competitiveness: Conceptual Aspects and Evaluation, OECD, Economic Studies, pp 147-197.
- [7] Organization for Economic Cooperation and Development. 2010. Economic policy reforms 2010: Going for Growth, OECD publishing
- [8] Srivastava, S. C., & Teo, T. S., 2010). E-government, e-business, and national economic performance. Communications of the association for information systems, 26(1), pp 14
- [9] Huselid, M.A., 1995. The impact of human resource management practices on turnover, productivity, and corporate financial performance. Academy of management journal, 38(3), pp.635-672.
- [10] Branstetter, L. and Chen, J.R., 2006. The impact of technology transfer and R & D on productivity growth in Taiwanese industry: Micro econometric analysis using plant and firm-level data. Journal of the Japanese and international Economies, 20(2), pp.177-192
- [11] Baily, M.N. and Solow, R.M., 2001. International Productivity Comparisons Built from the Firm Level (Digest Summary). Journal of Economic Perspectives, 15(3), pp.151-172.
- [12] Narayanan, K. 2004. Technology Acquisition and Growth of Firms: Indian automobile sector under changing policy regimes, Economic and Political Weekly, 39(6), pp 461-470
- [13] Tan, Y.T., Shen, L.Y., Yam, M.C.H. and Lo, A.A.C. 2007. Contractor Key Competitiveness Indicators (KCIs), A Hong Kong Study Surveying and Built Environment, 18(2), pp 33-46
- [14] Sahu, S.K. and Narayanan, k. 2015. Technology Import, R & D Spillover and Export: A Study of automobile sector in India, Working paper 98/2015
- [15] Buckley, P.J., Pass, C.L. and Prescott, K. 1988. Measure of International Competitiveness: A Critical Survey, Journal of Marketing Management, 4(2), pp 175-200.
- [16] Buckley, P.J., Pass, C.L. and Prescott, K. 1990. Measure of international competitiveness: Empirical findings from British manufacturing companies, Journal of Marketing Management, 6(1), pp 1-13.
- [17] Saranga, H., Mukherji, A. and Shah, J. 2009. Determinants of inventory trends in the Indian automotive industry: An empirical study, IIM Bangalore Research Paper, (286).
- [18] **Onwuegbuzie, A.J. and Leech, L.N.** 2012. Qualitative analysis techniques for the review of the literature, The Qualitative Report, 17(56), pp 1-28R
- [19] Allen, I.E. and Seaman, C.A. 2007. Likert scales and data analyses. Quality Progress, 40(7), pp 64-65
- [20] Marino, S.M. and Gladyshev, V.N.. 2010. Cysteine function governs its conservation and degeneration and restricts its utilization on protein surfaces. Journal of molecular biology, 404(5), pp.902-916.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



A DYNAMIC OPTIMAL CONTROL SYSTEM FOR COMPLEX COMPRESSED AIR NETWORKS

S.W. van Heerden¹, J.N. du Plessis² & J.H. Marais³

¹North-West University, CRCED-Pretoria, Pretoria, South Africa swvanheerden@researchtoolbox.com
²North-West University, CRCED-Pretoria, Pretoria, South Africa juplessis@researchtoolbox.com
³North-West University, CRCED-Pretoria, Pretoria, South Africa jhmarais@researchtoolbox.com

ABSTRACT

Mines use large compressed air networks to supply shafts and processing plants with compressed air. These networks can however be very complex as multiple compressors are located at different locations. Each end user of compressed air is also often a separate business entity - each following its own schedule and usage requirements. Despite this complexity, most mines still use static compressor control, which is based on typical usage patterns, on these networks. However, the static control strategy is only effective when all the end users use compressed air matching the typical usage pattern, since if one end user deviates from this pattern, the strategy becomes outdated. This typically happens on a daily basis. The article describes a dynamic compressor controller that offers an optimised solution for controlling the network. This optimised solution will specify the running compressors as well as the minimum pressure set point for these compressors. This ensures that the compressed air network is controlled optimally. The system was tested repeatedly and has shown that the system has the potential to reduce the running costs of a typical mine by up to R17 million per annum.

This work was sponsored by HVAC International

¹ Dr. S.W. van Heerden is a registered professional engineer and holds a PhD and M.Eng. degree in Computer and Electronic Engineering from the North-West University as well as a B.Eng. Computer Engineering degree from the University of Pretoria. He is currently a post-doctoral student at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.

² Dr. Johan N. du Plessis is a registered professional engineer and holds a PhD in computer/electronic engineering from the North-West University. He is currently a post-doctoral student at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.

³ Dr. Johan H. Marais is a registered professional engineer and holds a PhD in electrical engineering from the North-West University. He is currently a lecturer at the North-West University's Centre for Research and Continued Engineering Development (CRCED) in Pretoria.



1. INTRODUCTION

South Africa has a large mining sector which forms an essential part of the economy [1]. However, mining is an energy intensive industry, especially from an electricity consumption point of view [2]. In 2011, mining consumed 16% of South Africa's total electricity [3]. Gold and platinum mines further accounted for 80% of the total energy use [4],[5].

According to Fraser, compressed air generation on mines consumes approximately 25% of the total electrical energy use for gold mines and 40% for platinum mines [6]. This is approximately 9% of the total electricity consumption of South Africa's industrial sector [7]. Further, when examining the total cost of a compressor over 10 years, the electricity usage represents approximately 75% of the total cost [8]. With this in mind, it is clear that the energy usage of compressed air generation is an obvious target for electricity consumption reduction.

Most South African gold- and platinum mines supply compressed air to the shafts and processing plants through large compressed air networks called compressed air rings. Compressed air is supplied to these rings from either a single or multiple locations. However, each shaft or processing plant does not usually have its own compressed air source. These compressed air rings are also usually massive, consisting of up to 75 km of piping [9]. By using these compressed air rings, the reliability of compressed air is improved by providing backup compressors.

In order to ensure the optimal operation of the compressed air ring, a supply pressure profile can be generated depending on the combination of equipment, the number of the compressed air requiring equipment and the usage times of said equipment. These pressure profiles show the necessary supply pressure throughout the day. Each mine has a unique pressure profile with a noted maximum and a minimum pressure. If the pressure is too high, it could damage equipment; if the pressure is too low, the equipment ceases to work.

Each end user on a compressed air ring has its own compressed air requirements, thus if all end users on the compressed air ring are combined, a total pressure profile can be generated. However, if one end user changes its requirements, the pressure profile of the ring will change. Mines routinely deviate from their normal schedules, causing the schedules to become obsolete [10]. This makes the pressure profile generation of a compressed air ring a complicated problem.

As a solution to this, Van Heerden developed the Dynamic Compressor Selector (DCS) system [11] based on the work done by Venter [12]. The DCS system can observe the real-time requirement of compressed air networks and dynamically control the compressors to satisfy that need. This system was successfully implemented on a platinum mine [13].

2. SHORTCOMINGS OF PREVIOUS WORK

The DCS was not without its problems. Van Heerden mentioned that the DCS suffered from poor stability and non-optimal compressor selection [11]. Van Heerden thus further developed the dynamic compressor controller system and created the Dynamic Compressor Controller (DCC) [14]. However, during the development of the DCC system Van Heerden identified even more shortcomings in the DCS system.

The DCS system calculated a pressure set-point dynamically for each compressor house. However, ideally each compressor requires its own individual set-point. This is required because the compressors on a compressed air ring are located at different locations. The distance between the supply and demand can therefore be so extensive that the pressure drop becomes significant.

Further, when controlling compressors it is often assumed that compressors are most efficient when running at full power or that the largest compressor available is the most efficient. The DCS system also operated on this assumption. However, when looking at compressor efficiency curves it can be noted that the most efficient control point for a compressor is not always at full power, neither is the largest compressor always the most power efficient. When selecting the trimming compressor this assumption is also made, and in turn will cause the smallest running compressor to always be selected as the trimming compressor. However, advance larger compressors will sometimes be able to control their flow delivery more effectively and energy efficiently.

From a maintenance perspective reducing cycling, which is the continuous on and off switching of equipment, is a key objective. Therefore, in order to control the compressor optimally and prevent cycling, the future state of the network needs to be estimated. In order to be able to estimate the future state of the network the future flow requirement of the network needs to be estimated. This enables the control system to start the optimal sized compressor for the required present flow rate as well as the future flow rate.

Figure 1 One-dimensional network [14]



The DCS system worked on a one-dimensional level by only having a total supply and total demand. An example of such a one-dimensional network can be viewed in Figure 1. The controller then attempts to do flow matching. However, to ensure that the dynamic compressor controller is able to work on most networks the controller needs to be location attentive and work on a two-dimensional level [14]. This means that more than one supply point as well as demand point is present. An example of this can be viewed in Figure 2. Tying in with this is the fact that the controller needs to be able to identify when a compressed air pipe is closed in real-time and adapt its control philosophy accordingly.



3. THE DYNAMIC COMPRESSOR CONTROLLER

3.1 Simulation

The method of simulation used by the DCC system [14] was based on how the DCS system completed simulations [10]. This simulation model was developed by Venter [12]. He made the assumption that using the compressor exit temperature is satisfactory for isothermal flow calculations [12]. This assumption was used made Sun [15] as well as Ke and Ti [16].

The DCS requires the user to determine the pressure drop over a pipe empirically and incorporates this into the surface friction of the pipe. This ensures that any pressure drop attributed to temperature changes is automatically accounted for. For this reason empirical simulations utilise an average temperature throughout the system.

Simulating transient temperature changes on a pipe network that spans hundreds of kilometres is not practical, as there are simply too many variables influencing the temperature. This would require a magnitude of temperature measurements on a system that changes continuously.

Van Heerden proved that the simulation method could successfully calculate large compressed air networks [11] and Van Niekerk successfully used the system in a project [13]. The system of nodes and pipes are used to build up a complete network and by using Bernoulli's theorem [17], equation (1), the network can be simulated. However, one of the reasons the DCS system [11] was having stability issues was in its inherent design which made recovering from erroneous output almost impossible. This can be viewed in Figure 3.



$$\frac{\rho v^2}{2} + pgz + P = constant$$

(1)

- ρ : Fluid density [kg/m³].
- v : Fluid velocity [m/s].
- g : Gravitational acceleration $[m/s^2]$. z : Measured height [m].
- P : Pressure [Pa].



Figure 3 The DCS simulation process

All the calculations of the DCS system were also based on the idea of having a supply pressure to calculate the demand pressure. This is then used to simulate the current state of the network. To calculate the supply setpoint pressure, the supply pressure is increased or decreased until all the demand pressures are equal to or greater than their respective set-point pressures. However, by following this approach only one supply pressure set-point per compressed air ring can be calculated.

Furthermore the DCC system's calculations are based on the idea of calculating the ideal state of the network and not the current state [14]. It will then attempt to assign inputs in order to create this ideal network state. The DCC simulation process can be viewed in Figure 4. The "viable inputs" is data captured from the compressors in order to estimate their ability to deliver compressed air.



Figure 4 The DCC simulation process

The simulation process that Van Heerden [11] used is based on nodes, an example of which can be viewed in Figure 5. By using this structure and having nodes represent end users, supply points and junctions, a complete network can be built. An example of this can be viewed in Figure 6.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



For calculating the pressure, it is assumed that the end point pressure is known. The center node pressure is thus iterated and with each iteration the flow rate through the pipes are calculated. The center pressure is then iterated until equation (2) is valid. This whole process is repeated for all nodes until the rate of change of the flow through the pipes converge to 0.

$$\sum m_{flow into node} = \sum m_{flow out of node}$$
(2)

m : Mass flow [kg/s].

For the End User nodes the pressure must be calculated based on the flow as only the flow is measured upstream of the control valve. For this calculation equation (3) is used. The DCC also makes use of a "driven node" when simulating. This node is the demand node which drives the supply set-point up. Practically it means that the driven node will have its pressure be equal to the set-point pressure whereas all other nodes will have their pressure exceed the set-point pressure. This is done in order for the input pressure to be calculated at the lowest possible value while still supply enough pressure.



$$P = P_n - \left[\left(\frac{v^2}{2} \right) \bullet \rho \left(f \frac{L}{D} + k \right) \right]$$
(3)

P : Pressure [Pa].
Pn: Pressure of other node [Pa].
v : Fluid velocity [m/s].
ρ : Fluid density [kg/m³].
f : Darcy friction factor.
L : Measured length of pipe [m].
D : Pipe diameter [m].
k : K-loss factor.

Like the calculations of the DCS, the DCC systems read the flow values from the demand in order to use them in the calculations. However, because compressed air systems usually have leaks, this read value must be adjusted. The formula used to adjust the flow can be viewed below in equation (4). Finally, if all of the measured demand flow is adjusted, flow matching of the supply and demand can be done.

$$m_{adjusted} = m_{measured} \bullet \left(1 + \frac{\sum m_{total \, supply}}{\sum m_{total \, demand}} \right)$$
(4)
m : Mass flow [kg/s]

3.2 Future flow

The future flow estimation for the DCC [14] is based on historical averages with a resolution of 30 minutes which can then be used to generate a profile for each day. The current flow rate is logged every 2 minutes and in order to keep the values current the amount of loggings are limited to 650. The data is logged as a count and a total, this is then adjusted with each log. When the maximum logs of 650 is reached the formula in equation (5) is used to calculate a new average.

$$Total flow_{new} = \left(\frac{Total flow_{old}}{650}\right) \bullet 200$$
⁽⁵⁾

This new total flow is updated with a fake logged count of 200. This will allow the future flow estimation to keep estimating the flow historically while adjusting the historical value with time. For the simulations, the highest flow value between the current flow, 30 minute predicted flow and 60 minute predicted flow is used. This is to ensure that the compressed air supply will always be able to supply the required flow of the network within the hour period.

3.3 Compressor performance estimation

Compressor performance will change over time, since there are many factors influencing compressor performance such as inlet filters and seasonal temperature changes. The DCC system will thus actively log the pressure and the flow that the compressor can deliver. The power usage is also logged with this data, which is routinely updated to insure that the logged data is as relevant as possible.

This data is stored in a lookup table, which allows the DCC to seek the historic energy usage as well as the historic supply pressure and flow ranges. By combining these, the DCC can estimate how much power a compressor will use to deliver the specified flow. This estimated power usage is used as a cost metric.



3.4 Integrated solution

3.4.1 Overview

The DCC system works within a structure of solutions, worker, network, pipe and node structures. The relationship of these can be viewed in Figure 7. Each solution contains a number of individual workers. each worker contains one network, which in turn contains many pipes and nodes. The pipes and nodes are used to create the network.

3.4.2 Solution object

The solution represents a running combination of compressors. When the DCC simulates a network it will create multiple solutions in order to test different compressor combinations. By taking inspiration from genetic algorithms the DCC creates different combinations of running compressors for simulation. In order to reduce cycling and improve simulation speed a solution tree is built, this can be viewed in Figure 8. The starting solution is always taken as being the current running combination of compressors.



Figure 8 The DCC solution tree [14]

Solutions are created by representing compressor states as either 0 or 1. These combinations then form a binary string. A degree of evolution is made as one solution is evolved to all possible states by only changing one bit. A second degree of evolution is when all the children of the first solution are evolved. Figure 9 is an example of one degree of evolution. These solutions are evolved using modifications on genetic algorithms as they require a cost to find the best solutions to modify. As the cost metric is only available after simulation, the solution must be evolved to multiple possible solutions. To reduce cycling the amount of evolutions are limited as each evolution is a change of a compressor state.



Figure 9 The solution evolution [14]



Before simulation the solutions are filtered for unfeasible solutions. These are considered as solutions which don't supply enough compressed air, supply too much compressed air or are a duplicate child solution. This reduces the amount of simulations that require computing.

3.4.3 Worker

The purpose of the worker is to simulate its own network and calculate the energy usage of its network. Each worker runs on a different CPU thread (multithreading) in order to improve simulation efficiency. The solution assigns different running set-points to all the compressors supplying compressed air in the simulation. These flow set-points are calculated using a shotgun hill climbing algorithm. This can be viewed in Figure 10. The same problem as with the solutions exists here in that the cost metric is energy usage and this is only available after simulation. However, by having multiple workers, different possible solutions are made which allows the DCC to find the optimal running point of the current compressors.

3.4.4 Energy estimation

After all the workers of all the solutions have finished with the simulations, the DCC can calculate an estimation as to the energy requirement of the simulations done. The DCC will then choose the most energy efficient completed simulation as the ideal state of the compressed air network. Further, in order to promote energy savings by stopping the DCC system cycling compressors, a scaling value can be selected to artificially reduce the energy consumption of the current running solution. This will ensure that a new solution is only chosen if the new solutions reduces the energy consumption by the scaling value or the current solution is rendered invalid by either supplying too much compressed air or not enough compressed air.



Figure 10 Hill climbing search flow diagram [14]

4. RESULTS

4.1 Node balance simulation testing

In order to test the DCC's network simulation, the network in Figure 11 was simulated in the DCC. The x represents the supply node and its pressure must be calculated. All the pipes in the network are 1 000 m in length, have an internal friction of 45 μ m and have a diameter of 0.6 m. The results can be seen in Figure 12.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE





Figure 12 Sample small network result

Figure 13 and Figure 14 shows how the pressure and flow differences change during the simulation process of AirNode 2. In Figure 14 it can clearly be seen how the pressure of AirNode 2 is iterated twice. This is because the pressure of AirNode 6 has changed and the flow in AirPipe 3 has changed as well requiring the pressure of AirNode 2 to be recalculated.



Figure 14 flow difference convergence of AirNode 2

2752-9



From the above mentioned figures it can be seen how the DCC system successfully simulates the future state of a compressed air network. The DCC also assigned a supply set-point pressure that will ensure that every end user on the compressed air ring will receive their respective minimum air pressure.

4.2 Network testing

In order to verify if the DCC system can simulate a network accurately, it was compared to the DCS and KYPIPE by simulating the network in Figure 15. KYPIPE is a commercial compressed air simulation program. The simulated network can be seen in Figure 16.



Figure 16 Simulated test network

The results of these comparative simulations can be viewed in Table 1 and Table 2. It should however be noted that the DCS and the DCC works on gauge pressure while KYPIPE works on absolute pressure. Gauge pressure is defined in equation (6). For these tests the atmospheric pressure was 89 kPa.

$$P_{gauge} = P_{absolote} - P_{Atmospheric}$$

(6)

P : Pressure [Pa].

From the below mentioned tables, it can be seen how the DCC favourably compares to the DCS and KYPipe in terms of simulation accuracy. The standard deviation of the results was less than 1%. From this it can be said that the DCC simulations are accurate.

In order to show the effect of the CPU threads on the calculations, a screenshot was taken of Windows® Task Manager after a simulation was completed. This can be viewed in Figure 17. As stated, the load is spread out on the different threads of the CPU, taking full advantage of modern CPU design.

KYPipe value (kPa)	DCS value (kPa)	DCC value (kPa)	Difference: K-pipe vs DCC (kPa)	Similarity: K-pipe vs DCC	Difference: DCS vs DCC (kPa)	Similarity: DCS vs DCC
500.0	508.8	500.0	0.0	100.00%	8.8	98.27%
500.0	508.4	500.0	0.0	100.00%	8.4	98.35%
546.2	553.4	545.8	0.4	99.93%	7.6	98.63%
701.6	701.8	690.9	10.7	98.47%	10.9	98.45%

Table 1 Pressure comparison [14]


735.0	735.0	724.0	11.0	98.50%	11.0	98.50%
735.0	735.0	722.6	12.4	98.31%	12.4	98.31%
Average difference		5.75	99.20%	5.75	99.20%	
Standard deviation		5.64	0.78%	5.64	0.78%	

KYPipe value (kg/s)	DCS value (kg/s)	DCC value (kg/s)	Difference: K-pipe vs DCC (kg/s)	Similarity: K-pipe vs DCC	Difference: DCS vs DCC (kg/s)	Similarity: DCS vs DCC
47.5	47.4	47.5	0.0	100.00%	0.1	99.79%
47.5	47.6	47.5	0.5	98.95%	0.6	98.74%
95.0	94.8	94.8	0.2	99.79%	0.0	100.00%
47.5	47.5	48.0	0.5	98.95%	0.0	98.95%
47.5	47.5	47.0	0.5	98.95%	0.5	98.95%
Average difference			0.34	99.33%	0.34	99.33%
Standard deviation			0.2	0.47%	0.2	0.47%

Table 2 Flow comparison [14]



Figure 17 Simulation CPU usage

4.3 Simulated energy reduction

According to Van Heerden the DCC can reduce the energy consumption of a compressed air system by up to 86MWh which, by using 2014-2015 Eskom tarrifs, extrapolates to R17 800 000 per annum [14]. By following a worst case scenario of compressed air energy use and using the energy usage data metioned in the introduction, the DCC could potentially reduce the energy usage of South Africa by 1%.

5. CONCLUSION

The DCS system designed by Van Heerden was the first dynamic compressor controller for compressed air rings on mines [11]. Van Niekerk *et al* proved that the concept worked and that it could succesfully reduce the energy usuage of a compressed air ring [18]. Van Heerden's results also showed a reduction in the flow fluctuations of the delivered compressed air, which is a byproduct of proper demand-supply matching [11].



However, the DCS system cannot function on all compressed air networks, the DCC systems addresses this and improves on the energy usage according to Van Heerden [14]. The DCC system was however only tested in parralel to current working systems and in simulations.

By continsously trying to optimise the compressed air network and doing supply-demand flow matching, the DCC system provides a way to efficiently control compressors in such a way as to reduce the energy usage. As an additional benifit the DCC will also reduce flow fluctiations during control as the supply flow will be better matched to the demand flow. The DCC system also successfully addressed all the shortcomings of the DCS system.

Bibliography

van Heerden, S. W. 2016, "A dynamic optimal control system for complex compressed air networks," PhD thesis, Electrical Engineering, North-West University, Potchefstroom.

REFERENCES

- [1] Bronkhorst, Q. 2014, "The biggest companies in South Africa," BusinessTech.
- [2] Fisher-Vanden, K., Jefferson, G. H., Liu, H., and Tao, Q. 2004, "What is driving China's decline in energy intensity?," *Resour. Energy Econ.*, vol. 26, no. 1, pp. 77-97.
- [3] City of Cape Town 2011, "State of energy and energy futures report," Cape Town.
- [4] Departmnent of Energy 2012, "Draft 2012 integrated energy planning report," South Africa.
- [5] Tshisekedi, J. R. N. 2008, "Energy consumption standards and costs in South African gold and platinum mines," University of Witwatersrand.
- [6] **Fraser**, **P.** 2008, "Saving energy by replacing compressed air with localized hydropower systems: A 'half level' model approach," *South. African Inst. Min. Metall.*
- [7] Marais, J. H., Kleingeld, M., and Pelzer, R. 2009 "Increased energy savings through a compressed air leakage documentation system," *Ind. Commer. Use Energy, Cape Penins. Univ. Technol.*
- [8] **Cameron compression Systems** 2007 "Turbo-air 3000 centrifugal compressor." Houston, Texus.
- [9] van Tonder, A. J. M. 2014, "Automation of compressor networks through a dynamic control system," PhD thesis, Electrical Engineering, North-West University, Potchefstroom.
- [10] Heyns, G. P. 2014, "Challenges faced during implementation of a compressed air energy savings project on a gold mine," North West University.
- [11] van Heerden, S. W. 2014, "Developing a dynamic control system for mine compressed air networks," M. Eng. Dissertation, Computer and Electronic Engineering, North-West University, Potchefstroom.
- [12] Venter, J. 2012, "Development of a dynamic centrifugal compressor selector for large compressed air networks in the mining industry," M. Eng. Dissertation, Mechanical Engineering, North-West University, Potchefstroom.
- [13] van Niekerk, M. H. P. 2014, "The implementation of a dynamic air compressor selector system in mines," M. Eng. Dissertation, Mechanical Engineering, North-West University, Potchefstroom.
- [14] van Heerden, S. W. 2016, "A dynamic optimal control system for complex compressed air networks," PhD thesis, Electrical Engineering, North-West University, Potchefstroom.
- [15] Sun, L. 2012, "Mathematical modeling of the flow in a pipeline with a leak," *Math. Comput. Simul.*, vol. 82, no. 11, pp. 2253-2267.
- [16] Ke, L. S. and Ti, H. C. 2000, "Transient analysis of isothermal gas flow in pipeline network," *Chem. Eng. J.*, vol. 76, no. 2, pp. 169-177.
- [17] Durgaiah, D. R. 2002, Fluid Mechanics and Machinery. New Delhi, India: New Age International.
- [18] van Niekerk, M. H. P., van Heerden, S. W. and van Rensburg, J. F. 2015, "The implementation of a Dynamic air Compressor Selector system in mines," in 2015 International Conference on the Industrial and Commercial Use of Energy (ICUE), pp. 129-132.



AN INVESTIGATION INTO THE CONTRIBUTION OF VARIOUS HUMAN FACTORS TOWARDS THE SUCCESSFUL OUTCOME OF A COMPLEX WEAPON SYSTEM ACQUISITION PROCESS

G.D.P. Pretorius ¹, Prof. N.D. du Preez, Dr. L Louw

Department of Industrial Engineering University of Stellenbosch, South Africa gert.dp.pretorius@gmail.com

Department of Industrial Engineering University of Stellenbosch, South Africa niek@indutech.co.za

Department of Industrial Engineering University of Stellenbosch, South Africa louisl@sun.ac.za

ABSTRACT

This research investigates the effect of a specific set of human factors on the successful outcome of an acquisition project in a complex design to order environment. This study addresses the following overall research question:

• What are the significant human related factors contributing towards the success of a project or enterprise during the acquisition phase of a complex system when interacting with a customer?

The study is limited to the supply of a complex (engineered to order) system through a structured acquisition process. The example used is the supply of a military vehicle system as produced by one specific company.

¹ The author was enrolled for a PhD (Industrial Engineering) degree in the Department of Industrial Engineering, Stellenbosch University.



1 INTRODUCTION

This paper presents the study of a selection of projects in an enterprise in the military vehicle supply business domain. The acquisition process spans a number of years and follows a predetermined (prescribed) process. The acquisition process involves the successful passing of specific technical milestones in order to lower the technical risk profile as the design matures. In order to pass these milestones successfully, a significant interaction between the supplier and the customer is required.

The study therefore addresses the following problem statement:

• What are the significant human related factors contributing towards the success of a project or enterprise during the acquisition phase of a complex system when interacting with a customer?

In figure 1 below the success events, factors influencing the outcome, as well as their inter-relationships are presented. As depicted, the study is about the central focus in the diagram, which is to investigate the process and influences on the interactive and consequential events that occur between the supplier and the customer throughout the life cycle of the product's development, production and support.



Figure 1: The Success Event

The evaluation and categorisation of these interactions, defined as success events, are done utilising a specific a case study (Yin) [1]. The case study under review comprises of six sub cases involving three different vehicle projects spanning over three phase of the enterprise lifecycle (Ward) [2] as well as four phases of the product life cycle (DOD) [3].

A framework was developed in order to define and contextualise the effect of specific human interaction factors on the successful outcome of a project. A comprehensive structured questionnaire was also developed to investigate the contribution of these human factors in relation to the process defined above. This questionnaire was distributed to 144 respondents of which 120 replied. One response was incomplete and was discarded resulting in a response rate of 83%. The results were analysed and are presented in this paper.

The author was actively involved in these projects in the company at executive level, for at least ten years. Triangulation of the results is obtained through the summary of the questionnaire results, comparing it to the current executive opinion of the outcome and also adding the author's interpretation of the outcome.

2 THE ACQUISITION PROCESS

2.1 The Defence Acquisition Processes

The acquisition of a complex military vehicle system is well documented in the western world with the USA's Department of Defence (DOD) [4] and the UK's Ministry of Defence (MOD)'s [5] acquisition system guidance explaining in detail the process required to ensure a successful acquisition of these military systems. The generic phases and decision points in this process are depicted in figure 2 below.





Figure 2: DOD Acquisition Policy 5000.02: Generic Acquisition Phases and Decision Points [5]

The UK, European, North American, Australian and South African acquisition supply processes were used to derive BAE System's Engineering Life Cycle Handbook [6] to be applied in all their global businesses.

2.2 BAE Systems Life Cycle Handbook

The supply of a weapon system has to be aligned in co-ordination with the customer's acquisition processes. To this end BAE Systems [6] defined the generic process in their Engineering Life Cycle Handbook System as having five major phases as follows:

- Concept Creation Phase
- Development and Qualification Phase
- Manufacture Phase
- Support during Deployment Phase
- Disposal Phase



Figure 3: The Engineering Life Cycle of a Product or System, BAE Systems.[6]

Thirteen design review gates (0 to 8 in figure 3 above) have been identified to manage the success of the knowledge migration throughout the life cycle.

2.3 The Engineer to Order Product Life Cycle

The acquisition phase for a military vehicle system typically covers between two and five years in time. We investigate this process in more detail by considering the specific engineer to order product process defined by Land Systems South Africa [7] as depicted in figure 4.

The engineer to order product life cycle consist out of three distinct phases' namely:

- The development of a generic or base vehicle. This can also be the previous model produced. The knowledge attributes of the base product must reach a minimum level of maturity to obtain an **acceptable risk level** to continue with potential contracting.
- The customisation phase where the design is completed against the stated commitments. This phase's main aim is to mature the product's **knowledge attributes**. If the customisation is extensive, it will increase the technical risk level and lower the current design maturity level as measured against the required performance.
- The industrialisation as well as the successful production of the customised product this phase is primarily orientated to mature the knowledge attributes of the **processes** used during the production phase.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 4: The Engineer to Order Product Life Cycle, Land Systems South Africa. [7]

The sequential phases of the engineering life cycle define an increasing maturity level of knowledge of the item under development. This is defined as a set of technical states or baselines indicating the growth in Design Maturity Levels of the product. It is important to note that as the design maturity levels increase the technical risk levels decrease.

3 SUCCESS REDEFINED

According to Porter [8], a successful acquisition project can be described with the following two results:

- A successful value outcome for the business owner.
- A successful product value realised as experienced by the customer.

3.1 Business Owner Value

The value for the business owner is a measure of the increase in the monetary value of the business for which there are many published financial models. Although the business owner value is of paramount importance to the investor, it will only be sustainable with continued delivery of true customer value.

3.2 Customer Value

The customer normally defines mandatory attributes as well as a weighted scale for the required functional performance for a specific project. In an engineer-to-order project [8], the customer also contracts a future commitment of the product performance. This is normally defined in a performance matrix, against stated requirements and timeline. Alignment of the offered product's attributes is based on subjective and objective elements combined in the three-dimensional space against the customer requirement set. Competitive product analysis is also done within this framework.

The three domains of presenting a knowledge item are defined as a structural or hierarchical view, a functional view as well as a logic or timeline view one can superimpose the required vs. the achieved attributes of the knowledge item during its application, we can define the following values and expectation gaps in the various views as defined in figure 5:





Figure 5: The Customer Value Realised - Pretorius et al. [9]

As can be seen from the above it is always critical to meet, but not exceed, the customer expectations in all the areas. We need to involve the customer through continuous engagement in order to understand his real needs and value system as this may differ from the explicit knowledge available during the process. The continuous interaction is defined as events where the supplier can demonstrate success and, in themselves, can also contribute towards the satisfaction of the customer as this is the only way in a complex product environment to obtain and react to the real (and perceived) value system of the customer.

3.3 Critical Success Factors

Kerzner [10] defines "The goal of any project is to achieve technical objectives within the allotted time period and within the allocated budget." Davis [11] stipulates that the project is only as successful as the added business value to the organization. The US Government Accountability Office (GAO) [12] stipulates that military acquisition management systems can only be successful at the macro level (DOD) if the correct strategies and tactics are implemented at the micro (program) level. Rockart [13] defined Critical Success Factors as "the areas of operation where results must be positive for the organization to achieve goals and attain satisfactory performance". Bjorn [14] in his study of the USA Department of Defence acquisition process, employed multiple methods to assess the impact of system engineering reviews in order to define the correct approach to knowledge and risk in line with the funding of the program.

Dobbins and Donnelly [15] developed a standardised process for the formulation of Critical Success Factors specifically for defence program execution. This process is based on risk management when it considers the probability to achieve each Critical Success Factor. The DoD Risk Management Guide [16], "risk is a measure of future uncertainties in achieving program performance goals and objectives within defined cost, schedule and performance constraints." Risks are often divided into three categories based on the program performance measurement that is impacted upon realization of the risk. These are cost, schedule and performance risks. The significant amount of risk prevalent to the acquisition of advanced systems must be managed by a knowledge-based approach to system acquisition and development (GAO, 2008) [17]. "Knowledge-based approaches are characterized by early management and technical reviews coupled with technology development activities to replace risk with knowledge and reduce first-time events and program uncertainty. According to the Defence Acquisition Guidebook [18] the knowledge-based approaches also require systematic re-assessments as knowledge becomes available thereby reducing uncertainty as the program progresses. New Knowledge provides decision authorities with greater degrees of certainty (wisdom), increasing the propensity for success that the system will provide the required capabilities and be delivered on cost and schedule. We can now focus on the events surrounding the technical reviews where, as indicated above, the potential for the influence on the outcome resides.

3.4 Success Events

A success event is defined as a series of premeditated, tacit and explicit knowledge exchanges that contributes towards the ultimate success of a technical baseline, the project and thus also that of the business. Examples of success events are:

• A communication interaction like a meeting, a letter, an e-mail, during work or social sessions with the customer.



- A formal presentation of a bid, a product description or contract negotiations as part of the acquisition life cycle.
- A formal design, program and contract reviews with the customer, i.e. milestones in the acquisition life cycle.
- A sincere, honest and speedy reaction towards mitigation of a problem or a perceived problem.
- An informal and formal presentation of in-process (stage buy-off) and the final product presentation.
- The interactions during the delivery of a service.
- The actual performance of the deployed product against the expected performance.

In summary success (or failure) events occur before, during and after each technical review gate in the acquisition process [5], is cumulative in nature [12], is mostly based on the perception of the customer [14] and requires consistent human interaction between the supplier and the customer.

3.5 Factors contributing to Success

Bob Seelert identifies leadership as a critical factor contributing towards successful client interaction [19]: "Leadership is intangible. The first step is earning the trust of people you work with. Everything else follows from that."

Soupios et al. [20] defines the need to investigate relationships with their statement "We are convinced that enduring success is ultimately traceable to those rare men and women who, by virtue of their personal insights and integrity, are able to command the loyalties and commitments of their subordinates. To our way of thinking, failure to see matters in these terms constitutes the real naïveté as well as the greatest impediment to building a meaningful enterprise."

Courtney et al. [21] studied the social learning at the Australian Defense Organization (ADO) in order to identify factors that will enable the generation and transfer of knowledge to contribute to creation of organizational culture in support of continuous learning. These factors are described as "common identity, morale, problem solving, team building, performance management, workplace design, organizational culture, records keeping, information exchange, IT infrastructure, professional training, and induction and enculturation." It will be important to understand the contribution of corporate culture on the success of the project.

3.6 Human Intervention

According to Briskin et al. [22] "change happens on a macro systems level but also on a micro level—one conversation at a time, one group at a time, one new idea spawned among a group of committed people, setting off a chain reaction of new possibilities". They conclude that collective wisdom is the human ability in finding alternatives ways in pursuing "wise action". Wise action is defined as the capability to exercise good judgment, and to reflect real understanding of the issue at hand. Collective wisdom result in unexpected and positive results that holds more value than any single individual can contribute.

The human's interaction within an organisation is very complex and cannot be segregated into specific functions. For example, a leader needs relationships, communication, teamwork, knowledge, sound judgement etc. to affect his leadership within an organisation. Similarly, communication cannot be isolated as it requires at least two parties to be successful which implies some sort of relationship at play.

During the preceding years of working in a complex organisation the writer identified that the effect of leadership, relationships, collective wisdom and company culture did have some role to play in the success of a project. We now investigate these specific human related factors as follows:

3.6.1 Leadership

Ward [3], through the study of companies at different stages of the organization's lifecycle, concluded that there are unique challenges facing organizations and their leadership at each transition between the different stages, as the demand for a specific leadership style between phases differ significantly. Consequently, it requires significant changes in managerial and leadership skills to lead an organisation at these different transitional periods. Hence, the Leadership Lifecycle with the leadership styles of the Creator, Accelerator, Sustainer, Transformer and Terminator are required for success in each phase of the life of the organisation.

3.6.2 Relationships

The human interaction is depicted by a relationship model as defined by Stiglingh [23] where he combined the ABC model (Activity, Belief and Consequence) of both the customer and the supplier as depicted in figure 6 below:





Figure 6: The Influencing Framework adapted from Stiglingh [23]

To obtain the best results from a relationship, it is advantageous if the supplier and the customer's belief systems are closely aligned. The belief system is the actual set of precepts from which you live your daily life, those which govern your thoughts, words, and actions [24]. As no complete alignment between the belief systems of the customer, as well as the supplier, at all levels, are realistically possible, one needs to focus on areas of miss-alignment that have the potential to turn a success event into a failure. According to the literature [25], the business belief system comprises of the following four elements:

- Behavioural Performance, Action, Inaction How people respond to what is happening.
- Cognitions, Mind-sets, Clarity, Perceptions How people think of what is happening.
- Philosophy, Core Beliefs, Personal Values The meaning people attach to what is happening.
- Emotional Climate, Relationship Management, Emotional Information How people feel about what is happening.

From this model, it seems that a positive consequence must be sought through careful planning of the sequential success event actions as well as a careful choice of the teams interacting with the customer.

3.6.3 Wisdom

Management is defined as a series of decisions combined with supportive actions. Knowledge, on the other hand, exists as tacit knowledge within the people as well as explicit knowledge within the enterprise [26]. Knowledge Management is one of the key factors that drive competitiveness within an enterprise [27]. The making of enlightened decisions by combining good insight based on experience and one's own belief system, augmented by appropriate knowledge levels, gives rise to the concept of wisdom [28].

3.6.4 Organizational culture

Organizational culture is the behaviour of humans within an organisation and the meaning that people attach to those behaviours. Culture includes the organisation's vision, values, norms, systems, symbols, language, assumptions, beliefs, and habits [29].

This industrial belief system manifests itself in different corporate culture forms like:

- An Innovation Culture stemming from the need for urgent renewal at the forefront of technologically based competition [30].
- A learning-based culture with the underlying premise of natural growth for an enterprise [31].

Although this breakdown compartmentalises specific cultures, reality on the ground actually consists of a combination of the above with one or more taking the dominant role.

As is the nature of the complex human, we do not expect these four factors to be all inclusive nor do we believe that they will be completely independent or mutually exclusive.

4 THE FRAMEWORK FOR SUCCESS

In order to allow a more detail view on the problem, we must investigate the various elements that potentially make up the framework for success.

- The first element that is potentially part of the framework is the success event. The success event is any opportunity that can be used to obtain and increase the customer's actual and perceived value expectation through direct and honest interaction.
- The next part of the framework is the definition of the processes required to execute the normal business. These processes spans within and across three business domains defined as the product, the enterprise as well as the customer domains. This part of the framework is simply called the executable factors.
- In order to study the human potential effect on the required processes of the executables, four specific study domains of leadership, relationships, wisdom and culture were considered based on the literature reviewed as well as own experience. These factors are not intended to be all inclusive as defined by their inter-relations, however, it allow us to study their specific effect on the success of the outcome when viewed separately. These factors grouped together, are termed the enabler factors.



The executables, the enablers and the success events is combined to form the framework in depicted in figure 7 as follows:

The process starts with an idea and this knowledge baseline is developed by applying all the normal business processes in the strategic tactical and operational areas through the product, enterprise and customer domains to end with value realised by the customer when he deploys the product successfully. This part of the framework is called the executables. The second part of the framework considers the human factors that can enhance (or hinder) the success of the outcome of applying the executable processes. These factors are defined as the enablers.



Figure 7: The Framework for Success

The effect of the human interaction (enables) on the executable factors can now be investigated by means of the case studies of three different products spanning five enterprise life cycle phases and four engineering phases in a military vehicle system supplier environment.

5 CASE STUDIES

5.1 Introduction to the Case Studies

The guidelines of Yin [1] were used in choosing and construction the specific case studies defined below. The cases considered for this research spans the life cycle of the enterprise from growth through to maturity and finally, decline. It is also set against the backdrop of the USA and NATO forces activities in Iraq and Afghanistan where both the RG31 and RG32 were successfully deployed by various armed forces.

Case no:	Project Name	Description	Customer	Product Life Cycle	Enterprise Life Cycle	Specific Atributes	Financial Outcome	Customer Satisfaction	Success Rating
1	RG 32 Development	Develop of new vehicle against a specific user requirement	Europe	Development	Growth	New Cusomer and New Product. User very satisfied with outcome	Medium	Very High	4
2	RG 32 Production Batch 1&2	Production of first two batched of vehicles	Europe	Production	Maturity	Production runs resulted in new level of quality demand for enterprise	Low	Medium	3
3	RG 32 Production Batch 3&4	Production of final two batched of vehicles	Europe	Production	Decline	Quality expectation gap grew out of control resulting in "Pain Plan" by customer	Very Low	Very Low	1
4	B Vehicle Support	Support of aging B vehicle fleet for the South African Army	South Africa	Support	Maturity	Need to support local customer with low budget	Low	Low	2
5	RG 31 MRAP Production	Supply of Urgent Operational Requirement	USA	Production	Growth	Urgent Operational Need in Iraq to save lives	High	Very High	5
6	RG 31 MRAP Upgrade	Midlife Upgrade of the MRAP fleet returning fom Theatre	USA	Support	Maturity	Re-use of vehicles as part of USA army inventory	Very High	Very High	6

Table 2:	Summary	of Case	Studies
----------	---------	---------	---------

The different cases that make up the case study are further described below:



5.1.1 Cases 1, 2 and 3: RG 32 Development and Production (2004 - 13)

The development of the RG 32 started as an evolutionary development process from a previous design. This was a complete design to order project with the Swedish government procurement agency against the user specific specification. Within a two-year period, the team concluded the development process to the point where the senior project officer wrote a poem to thank the development team in Swedish, tone set it and promptly sang it in the design review meeting. This can be seen as the highlight of the entire program. This project was conducted in the growth phase of the company.

The next project in the RG 32 program was the production of two batches of 100 vehicles each. During this time the project lead was moved to a program manager who brought his own value set to bear on proceedings. This changed the frequency of the success events between the customer and the company teams. This project was conducted mainly in the mature phase of the company.

After a production gap of about two years, two new batches of vehicle orders were concluded with the customer. These vehicles were needed for urgent use in Afghanistan. In this process the customer program team changed as well, which resulted in a new dynamic interaction and relationship between the customer and the company teams. This result was that very few success events were noted and an overall dissatisfied customer.

5.1.2 Case 4: B Vehicle Fleet Support (2005 - 14)

The South African customer purchased a fleet of logistic trucks in the previous century and needed to keep up the maintenance on them as the replacement programs were not funded.

The company set up a wide range of support centres throughout the country close to the user bases in order to support these aging vehicles and conducted the maintenance business at a very low cost base.

5.1.3 Cases 5 and 6: RG 31 MRAP Production and Upgrade Projects (2007 - 12)

During the Iraq war the USA realised that their Humvee unarmoured fleet of vehicles were not capable to withstand the Incendiary Explosive Devices (IED's) and that they needed an urgent solution. One of the solutions they decided on was the RG 31 Mine Protected Vehicle design and owned by the company being researched. The company was put under a challenge to produce circa 150 vehicles within a nine-month period as this was an urgent purchase program from the USA. It took on this challenge and proceeded to produce the vehicle successfully on time.

As indicated above more than a thousand RG 31 MRAP vehicles were built for the USA army and deployed in Iraq and Afghanistan where they performed very well. This resulted in the USA decision to keep this fleet of RG 31 MRAP vehicles and to perform a midlife upgrade program on them before being shipped back to the USA from Afghanistan. This project was conducted in the mature and early decline phases of the company. By this time the USA customer was very satisfied with the RG 31 vehicle's performance and this remains rated as very high.

5.2 Questionnaire

The relative contribution towards the success of the enterprise (project) of the enabling factor's interaction with the executable factors was interrogated by means of a structured questionnaire. This questionnaire was constructed by comparing and defining a question on any potential interaction of the enabler factors on each of the detail executable factors.

5.3 Statistical Analysis

In the case where the data is presented in the Likert scale format, a statistical regression model was used to study the effect of the enabling factors relationship, wisdom and culture on the successful outcome of the projects, thereby defining the most influential factor.

5.3.1 The Initial Regression Model

The first step in the analysis involved the compilation of the relevant questions' influence on the possible outcome of the cases defined in the "generic" regression model below:

 $Y_{\text{ProjectSuccess}} = B_0 + B_1 Q_1 + B_2 Q_2 + B_3 Q_3 \dots + B_n Q_n$

where:

Y_{ProjectSuccess} = Project Success Rating from 1 to 6

 B_0 = Correction factor

 B_1 = Contribution factor related to question 1

 B_n = Contribution factor related to question n



Using the R software [32], we can test for the validity of the Gauss Markov Assumptions to define that the OLS regression is reliable:

5.3.2 Multi-collinearity

The next step is to test the data for Multi-collinearity.

Gauss Markov assumptions state that there must be low to zero multi-collinearity between the independent variables of the model.

5.3.3 Heteroscedasticity

The test for Homo or Heteroscedasticity is performed next.

The revised model with no multi-collinearity is now test for Heteroscedasticity. Heteroscedasticity is tested using Breusch-Pagan test, with the below hypothesis:

H₀: Homoscedasticity

*H*₁: *Heteroscedasticity*

Using R software again [30], we run the test of the P-value of the Breusch-Pagan test is 0.5685 which is significantly higher than 0.05. We can therefore not reject the H0 and assume the model is homoscedastic.

5.3.4 Stochastic Regressors

Gauss-Markov theorem requires any of the explanatory variables to be uncorrelated with the error terms. Correlation of regressors to the residual should be close to if not equal to zero.

5.3.5 Hypothesis Test

The Hypothesis Test is then carried out which results in the ranked significance of the factors actually having an effect on the outcome. These are presented by a final regression model and also represents the outcome of the research questions posed as:

What are the significant human related factors contributing towards the success of a project or enterprise during the acquisition phase of a complex system, when interacting with a customer?

5.4 Results

5.4.1 Leadership

Some of the questionnaire answers where only a specific indication (e.g. yes or no answers) were required, were analysed by means of graphic results. This is done to observe the effect of the leadership styles on the successful outcome of the projects in the various phases of the enterprise. The confirmation of leadership styles prevalent during the different enterprise phases as defined by Ward are the creator (creation), the accelerator (growth), the sustainer (maturity), the transformer (turn around) and and the terminator (decline) resulted in a close correlation overall. A remarkably close correlation between the theoretical and observed leadership style was observed in Cases 1, 2 and 3 (see to Figure 8 below) where a lot of leadership focus was expended in order achieve success.



Figure 8: Mapping Leaderships Style on the Enterprise Phase

The example above shows the RG 32 cases (1, 2 and 3) which were challenged in their execution by the customer and required specific leadership focus in both the development and the production phases. We therefore see a very good alignment between the theoretical and the observed leadership styles during each enterprise phase.



5.4.2 Relationship

Question	Number	β Value	Question Topic
The company's senior management got on very well with the customer's senior management	2.2b	0.36	Relationship: Senior Management
The project team got on very well with the customer's project team	2.2c	0.33	Relationship: Project Team
The operational / quality team got along very well with the customer's quality team	2.2d	-0.34	Relationship: Quality Team
It is very difficulty to do business with the customer's quality team	2.3d	-0.43	Relationship: Quality Team
Which of the following areas did the company's operational and quality teams not see eye to eye with the customer?	2.4c	-0.29	Relationship: Quality Team

Table 3: The results of the Relationship Analysis

As seen from the results presented in table 3 the relationships of the senior management had a significant positive contribution to success, while the quality team's interaction resulted in a significant contribution to the failure of the specific projects.

5.4.3 Wisdom

As depicted in table 4 below the respondents agree that informed decision making from a tacit and explicit knowledge base in all cases were a major contributor towards their success.

Question	Number	β Value	Question Topic
The company's documented product	3.1b	0.44	Wisdom: Company's
knowledge i.e. functional design,			Project Team Explicit
data pack and specifications were			Knowledge
well defined			
The customer's tacit?? product	3.4a	-0.54	Wisdom: Customer
knowledge contributed significantly			Project Team Tacit
towards the success of the project			Knowledge
The customer's documented	3.4b	0.42	Wisdom: Customer
product knowledge i.e. functional			Project Team Explicit
design, data pack and specifications			Knowledge
were well defined			
The customer's project	3.5b	0.60	Wisdom: Customer
management made very good			Project Team Decisions
decisions			

Table 4: The results of the Wisdom Analysis

The company and the customer's project teams, through their expression of wisdom (good decision making supported by structured information) contributed significantly to the success of the company. However in case 2 and 3 the customer had more tacit knowledge about the product as the company lost its knowledge base of the company's chief design engineer that left the company and this contributed negatively to the outcome of these projects.

5.4.4 Culture

The statistical analysis indicates that there is no correlation between the project success and culture. We can therefore conclude that the culture of the organisation did not significantly influence the success of the organisation.

5.5 Triangulation of the results

The following table summarises the outcomes of the statistical analysis of the case studies, and compares it to the executive management opinion of the company studied, as well as the primary author's own experiences about the cases that were studied.

	Table 5;	I mangulation of Results	
Enabling Domain	Case Study Results	Company Executive	Researcher
Leadership	Close correlation	Agree with the outcome.	Leadership style were
	between the theory	Leadership style needs to	constantly adapted for
	of the specific	be adapted depending	specific application. I
	leadership style	circumstances. This can	concur with the
	against the	create confusion, since	observation of a close
	leadership style	employees sometimes find	correlation between the

Table 5: Triangulation of Result



	observed	the change in "character" of the leader difficult to	theoretical and actual leadership styles
Relationship	The positive relationship with the customer's senior and project management were observed. There were also a very negative result regarding the customer quality team relationship with the operational team of the company contributing towards failure.	Agree with the outcome: Excellent relationships between Management and the Customer remains the single most important factor during difficult times on any contract. The value of a "good" contract is secondary. The opposite is true in terms of the respective QA teams, where pre-agreed quality standards usually ensures a successful relationship between the QA teams.	observed In the case of the RG 31 and early RG 32 development the company had excellent relationships with both customers. This observation agrees with the view that good relationships in both the senior management and the program management level significantly contributed to success of these cases. The negative contribution of the customer quality teams' relationships can be identified with the RG 32 production cases as well as the B vehicle support program where both the Swedish and local embedded quality teams did not contribute towards success.
Wisdom	The contribution of explicit knowledge with both the customer and the company's teams were defined as having a positive influence on the outcome. However when the customer's tacit knowledge becomes more dominant than the company's tacit knowledge it resulted in a negative influence. It also helps if the customer makes robust decisions to promote the project as a whole	The retention of tacit knowledge on both the technical and commercial aspects of a project is a key factor, amongst others, for successful project execution.	The company and both the RG 31 and RG 32 customers have an extensive process of defining the explicit knowledge of the product. The loss of the senior design engineer on the RG 32 after development left an imbalance of tacit knowhow between the customer and the company and this was exploited by the customer's team resulting in a negative influence towards success. The RG 31 customer on the other hand demonstrated very sound judgement and decision making and this contributed towards the success of these two cases.
Culture	No correlation between culture and company success were evident in the case studies under review	Partially agree with the outcome: Company culture is normally enforced and cultivated from "above". Strong governance and policy enhances a strong culture which is needed when a project experience major challenges, especially if corporate interference is at stake. Cultural differences between the company and the customer can however lead to friction and/or conflict and sometimes need careful management.	The underlying culture of learning as well as the "forced " BAE corporate culture did contribute to inherent capability and order within the company however there are not one significant event that I can recall that had a direct bearing on the successful outcome of these cases. I therefore support the case study observation.



The various results and opinions defined in the table 5 above indicate a close correlation between the case study results, the company executive as well as the experience of the researcher who worked at the same company for thirty-three years.

6 CONCLUSION

6.1 Interpretation of the Summary Results

It can be concluded that all four Enabler Domains contribute towards the corporate success. However, the perceived contribution varies between the cases. The following detail observations are made:

- The more leadership focus is required; the more defined correlation is demonstrated with the theoretical leadership style.
- In all cases relationships are a prerequisite for success with a marked increase of the relationship contribution in the most successful project (case 3).
- The prevalence of bad decision-making results in a less successful project.
- Enforcing a corporate culture from the top does not necessarily contribute to the success of the company.

The Success Event Roadmap can be used to establish the potential contribution of any of the defined factors within the various processes. It also facilitates understanding of the contribution of the most appropriate factors in a specific scenario.

The analysis defined above considered all the detail answers to the all the questions in the questionnaire but through the statistical model application only isolated only those answers which were clearly tested to have a significant contribution to the successful outcome.

The results reflected in this article is therefore only as summary of the ongoing analysis of the case studies. More detail conclusions regarding the sub-factors interaction of both the enablers and the executables in different phases of both the product development and the enterprise phases will be reflected in the final study.

6.2 Recommendations and Supplementary Research

This study contributed the following knowledge items:

- A framework for success that defines all the conceptualisation of a success event.
- A specific relationship model.
- Some insight as to the importance of relationships and collective wisdom within the business framework.

The following future research topics will broaden our knowledge about conducting successful projects and enterprises:

- The study of the interrelationships between the different parts of the Executable Factors
- The study of the application of the framework for success in the non military and non complex business environments.
- The further development of the relationship model in all human interface applications.
- The study of knowledge growth in relation to risk of a specific product under development.

REFERENCES

- [1] Yin, Robert K. 2009. Case Study Research, design and Methods, 4th edition, Sage Publications, pp. 40-45, 86 and 98.
- [2] DOD 5000.01 Directive. 2007. The Defense Acquisition System, Certified Current as of November 20, 2007.
- [3] DOD 5000.02 Instruction. 2015. Operation of the Defense Acquisition System, January 7, 2015.
- [4] Ward, A. 2003. The Leadership Lifecycle, Matching Leaders to Evolving Organizations, Palgrave Macmillan.
- [5] <u>https://www.gov.uk/guidance/acquisition-operating-framework</u>, 13 May 2015.
- [6] BAE Systems. 2011. Lifecycle Management. The Handbook to the Engineering Lifecycle, HB05/01, this is an unpublished work created in 2005 2011.
- [7] BAE, Land Systems South Africa. 2013. Design to Order, Internal Best Practice.
- [8] Magretta, J. 2012. Understanding Michael Porter The Essential Guide to Competition and Strategy, Boston, MA: Harvard Business Review Press.
- [9] Pretorius, G.D.P., Du Preez, N. 2011. The Knowledge Cube, A Universal Framework to describe all Knowledge Items. COMA 13.



- [10] Kerzner, H. 2006. Project Management: A Systems Approach to Planning, Scheduling, and Controlling. Hoboken, NJ: John Wiley and Sons
- [11] Davis, B. 2009. 97 Things Every Project Manager Should Know, Collective Wisdom from the Experts, Download at Boykma.Com. p36
- [12] GAO. 2006. "Improved Business Case Is Needed for Future Combat System's Successful Outcome." GAO-06-367.
- [13] Rockart, J.F. 1979. "Chief Executives Define Their Own Data Needs." Harvard Business Review. (March-April).
- [14] Bjorn, Brig. J. 2012. Critical Success Factors for Evolutionary Acquisition Implementation, Electronic theses and Dissertations, Doctoral Dissertation (Open Access), University of Central Florida.
- [15] Dobbins, H.H. and Donnelly, R.G. 1998. "Critical Success Factor Analysis for DoD Program Management." Acquisition Review Quarterly. No. 5, pp 61-81.
- [16] Department of Defense. 2006. "Risk Management Guide for DoD Acquisition". Sixth Edition, Version 1.0. (August).
- [17] GAO. 2008. "A Knowledge-based Funding Approach Could Improve Major Weapon System Program Outcomes." GAO-08-619. (July).
- [18] Defense Acquisition Guidebook. 2011. "Defense Acquisition Guidebook". Defense Acquisition University, Dated: June 2011.
- [19] Seelert, B. Chairman, Saatchi & Saatchi. 2009. Start with the Answer, and other Wisdom for Aspiring Leaders, John Wiley & Sons, Inc. p147.
- [20] Soupios, M.A. and Mourdoukoutas, P. 2015. *The Ten Golden Rules of Leadership*, Classical Wisdom for Modern Leaders, M. A. American Management Association. pp128 and 9.
- [21] Courtney, J.F. University of Central Florida, USA, Haynes, J.D. University of Central Florida, USA, David B. Paradice, D.B. Florida State University, USA. 2005. Inquiring Organizations: Moving from Knowledge Management to Wisdom, by Idea Group Publishing. pp316, 317.
- [22] Briskin, A., Erickson, S., Ott, J., Callanan, T. 2009. The Power of Collective Wisdom and the Trap of Collective Folly, Berrett-Koehler Publishers. pp5,8,20,24.
- [23] Stiglingh, W. 2013. Unpublished works. (One-to-one disclosure, October 2013, Pretoria).
- [24] http://www.fringewisdom.com/your_belief_system.php. Accessed 11 December 2013.
- [25] Simon, S. D., & Pajares, A. F., 2001. From Neo-Behaviorism to Social Constructivism? The Paradigmatic Non-Evolution of Albert Bandura. A thesis submitted to the Faculty of Emory College of Emory University, p6.
- [26] Mohammad, A. H. 2010. Developing a Theoretical Framework for Knowledge Acquisition. European Journal of Scientific Research, 42(3), pp439-449.
- [27] Greiner, M. E., Böhmann, T., & Krcmar, H. 2007. A strategy for knowledge management. Journal of Knowledge Management, 11(6), pp3-15.
- [28] Jardim-Goncalves, R., Sarraipa, J., Agostinho, C., & Panetto, H. 2009. Knowledge framework for intelligent manufacturing systems. Journal of Intelligent Manufacturing, 6(3), 725-735.
- [29] http://en.wikipedia.org/wiki/Organizational_culture. Accessed January 2015.
- [30] Mortara, L., & Minshall, T. 2011. Technovation How do large multinational companies implement open innovation. Technovation, 31(10-11), 586-597. /j.technovation.
- [31] Gottschalk, P. 2006. Case Studies in Knowledge Management. Knowledge Management Research, 4(1), 75-76. /palgrave.kmrp.
- [32] https://www.r-project.org



7 BIOGRAPHY



Gert Pretorius obtained his Master's degree in Engineering Management from the University of Pretoria. In 2009 he was appointed Vice President for New Product Development at BAE Systems, Global Tactical Systems in the USA. He is currently the Automotive Advisor for NIMR LLC in the UAE.



Niek du Preez is Emeritus Professor in Enterprise Engineering at Stellenbosch University, South Africa. He is founder of the Global Competitiveness Centre in Engineering and is currently the CEO of Indutech, an Enterprise-wide Innovation Management Company, based in Stellenbosch.



Louis Louw is a senior lecturer in Industrial Engineering at Stellenbosch University. His research field include Enterprise Engineering and Innovation Management.



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



APPLICATION OF BERNOULLI PRINCIPLE IN ADDRESSING BOTTLENECKS AT A LOCAL PRODUCTION COMPANY IN THE VAAL REGION

T.B. Tengen

Department of Industrial Engineering and Operations Management, Faculty of Engineering and Technology, Vaal University of Technology, Vanderbijlpark 1900, Private Bag X021, South Africa, thomas@vut.ac.za, tel: +27835631768, fax2email: +27(0)866515179.

ABSTRACT

Production flow is similar to most physical flows, especially liquids that frequently encounter bottlenecks, an issue being addressed by the current paper. This paper investigates the applicability of the concept of the Elementary Law of Energy Conservation in fluid flow given by the Bernoulli Principle as one of the approaches of addressing production bottlenecks. Empirical data on managerial behaviour, managers' attitudes towards workers, production rate and the extent of bottlenecks were analysed following the concept of Bernoulli Principle.

It was found that the total effort required in executing a job is a function of production rate, the managers' closeness to workstations and "pressure" imposed by the bottleneck. Where pressure imposed by the bottleneck was high, the production rate was reduced. Increased pressure on operators by managers was equivalent to a reduction of distance from managers to workstations that led to increased speed of job-flow and/or reduction of bottleneck. It was concluded that the Bernoulli Principle is useful when applied correctly to address bottlenecks.

Keywork: bottleneck, conservation of energy, height of management structure, productivity



Introduction

"Energy can neither be created nor destroyed: it can only be converted from one form to another" (i.e. energy cannot be lost from the "entire universe", it can only be transferred from one system to another) is a fundamental law of nature that forms the foundation of most dynamic systems [1]. The field of energy science and energy engineering have been derived from this law, which subsequently have given birth to most popular subfields such as thermodynamic [2], steam plants [3], locomotive [4], aerodynamics [5] and so on. Energy is elementary defined as the "ability' to do work, which can be stated in a simple term as the "effort" required to do work [1]. In the same light, the effort required to do work can neither be created no destroyed: it can only be converted from one form to another or the effort can only be transferred from one system to another (i.e. in a typical physical system, energy is converted from potential energy to kinetic energy and the energy due to frictional losses or pressure against motion). This simply means that the "total effort" required to do work in a production system should be "constant" i.e. for a system that is well monitored and managed, if an additional effort or resource, which is unnecessary (e.g. unnecessary raw material, unnecessary management, unnecessary workstation, and so on), is added in performing a piece of job then the additional effort or resource would be available by the completion of that job, which would lead to a situation of under utilization of resources [6]. Most managers are aware of the impacts of the under utilization of resources and, hence, are engaged in capacity planning [7] and line balancing [6] so as to avoid or minimize this lost of or wasted effort.

In most production systems, series of processing operations (known as work or tasks or jobs) are performed on a product before the good that is finally acceptable by the external customer is produced [7]. Since energy (or work) is represented by the effort employed to produce a product, it follows that the effort required to produce a product from a production system should neither be created nor destroyed i.e. the effort should be constant. Production effort can only be converted from one form to another: the resources deployed by management are production efforts that can be seen from the variations in production speed (Known as production rate) and level of pressure at the workstation or degree of bottleneck.

The above premise opens up an interesting point of discussion and investigation. This is because, for a simple physical system, work is defined as the product of "force and distance", which are quantities that can be easily measured with measuring instruments; while in a production system the flow of work-piece is determined by the physical processing operations and other service activities. Because of the integration of both product processing and service activities, the interpretations of the variables involved in defining work have to be handled with care. For example, the word "pressure" has a simple meaning or interpretation in a physical system, but its interpretation can be confusing in the service sector. Thus, this article is aimed at attempting to apply a typical physical theory to an integrated product processing and service sector.

Method

Most products require several processes to be performed on them before they are acceptable as finished goods by external customers [7]. In order to improve on productivity or to achieve mass production or mass customization, job specialization is required where each process being performed on the product is assigned to an "operator" or a "workstation" [7]. If not well controlled, there will arise occasions where some operators will have to share the same space or stage so as to perform their operations, which will lead to congestion and traffic related issues on the stage. Thus, flow line production is recommended where each process is assigned to its own workstation [7] and each workstation has its designated space. This flow line production also allows for mass production or mass customization [6,7]. The flow line production is adopted in this report as it allows for easy interpretation of terminologies and collection of relevant data.

Bernoulli Principle [8,9,10] is a concept that has been widely proven to be very useful for fluids flowing in pipes of "different" sizes. It states that the energy or effort required for fluid or liquid to flow across different cross sections of a pipe is constant, or the total energy required to ensure fluid flow in a pipe can neither be created nor destroyed: it can only be converted from one form to another. That is to say, the total work done on a fluid at different cross sections of a pipe is constant. Bernoulli Principle is given by the following equation for a unit mass (i.e. unit mass is equivalent to "per unit product" produced in the current research) and for an incompressible flow as, [8,9,10]

$$E = \frac{v^2}{2} + gx + \frac{p}{r} \tag{1}$$

where E is energy of (or effort required by) a fluid flowing in a section of a pipe, v is the velocity of the flow, g is a (gravitational) constant, x is a distance from the reference point, p is pressure on the fluid at that



section and rho (ρ) is the density of the fluid. It should be noted that on the right-hand-side of equation (1), the first term stands for the kinetic energy (k.e.), the second term represents the potential energy (p.e.) and the third (last) term represents the frictional energy. It is common knowledge, in physical system, that when a body that had been raised to a height falls then it loses its potential energy, which is being converted to a gain in the kinetic energy and frictional (or resistance) energy. One has to return to the theme of the current project: a manufacturing system and not a mere physical system where every quantity in expression (1) can be measured with a measuring instrument.

As stated earlier, the key issue is how to interpret the parameters found in equation (1) for an integrated product and service industry (i.e. for a production system). The scenarios given below are used as examples. Before such an attempt, it is imperative to translate those parameters found in expression (1) to terminologies used within production settings. In this regards, E can be interpreted as the total investment made available to produce a product, v should be proportional to the production rate or feed rate or the "speed" at which a product is produced, g is also a constant that is related to the "speed" at which managers move closer to operators or workstations i.e. a function of management respond time, x is the distance or hierarchical position of managers (who take decisions) away from the operators/workstation with (+x) representing when managers consider themselves to be above operators (overhead managers or white collar managers) and (-x) being managers who listen to operators or consider themselves to be below operators, with (zero x) being blue collar managers or those who also perform the operators' tasks, p is the pressure imposed by the operator or workstation and rho (ρ) is the production density or production quantity per unit space. In this work, as also applicable to the initial Bernoulli Principle, it is assumed that one is dealing with a constant density flow, which is tantamount to incompressible flow or a production system with zero scrap rate or rework. In this case, a section of the case-study company that deals with final packaging of soft drinks was chosen for the case study.

Data collection

Data were collected at an isolated (one) workstation and also at a group of workstations as given in the scenarios below. It should be noted that within a production setting, the speed of production or what is normally termed the production rate is a depended variable that depends on the managerial attitude (or managerial pressure) and/or the pressure from the workstations (or bottlenecks). The management/overhead is responsible for providing production resources whenever and wherever they are needed. The pressures from the workstations (i.e. bottlenecks) are indicated by its cycle/standard times, which can vary with machine deterioration or changing machining speeds leading to bottlenecks [11].

Scenario 1: changing managerial closeness (x) towards a single or particular workstation

The management/overhead contribution in the effort required to produce a product is represented by the second term on the Right Hand Side (RHS) of expression (1). So, when the speed of respond of the manager is constant or the speed with which a manager moves closer to workstation is constant, where g is constant as to be assumed in this project, the distance between the workstation(s) and the manager x can be represented by the time it takes the manager to respond to a call i.e. tall managerial (reporting) structure increases the x or *time*. In a simplistic form, an increment of +1 in x can be used to represent the next (management) level at which decision is made with the assumption that it takes the same time for decision to move from one level to the next e.g. x=+2 represents a situation where an operator has an immediate Line Manager who in turn has his/her own Line Manager i.e. a second Line of command. In this case, when a call is made, the First Line Manager spends a given time T in an attempt to address the problem, and if the problem cannot be addressed, then the matter is escalated to the Second Level Line Manager who takes the same time T (in an attempt) to address the problem. It must be noted that the effective time spent before the Second Line Manager addresses the problem from when call was made is 2T.

Scenario 2: changing managerial closeness (x) and attitude towards different workstations (p)

When managers have different attitudes towards different workstations, then it is natural that bottlenecks will (or can) arise: this typically represents a situation where the line is not balanced (i.e. poor Line Balancing) and management does not take serious the non-uniform workloads at the different workstations. Obviously, different workstations will experience different pressures (p). The pressure induced by a workstation or bottleneck (p) was studied by varying the cycle time, more specifically by reducing the amount of resources at



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

the bottleneck workstation so that it created a situation whereby unfinished products started accumulating in front of the bottleneck stations. This led to more increase pressures on the bottleneck stations. The managerial closeness (x) to a workstation was changed as stated in scenario 1 above.

Scenario 3: Increase pressure or effort from manager

Under this scenario care has to be taken when interpreting (uncontrolled) pressures from managers and those pressures from workstations because these should be the common sources of misinterpretations and wrong results. The increased pressure from workstation is when the workstation is under serious strain because it is overloaded (or under resourced) thus leading to a state where the workstation slows down the entire production plant; and the increased pressure from the manager is when more resources are deployed onto the production line in the form of additional investments (e.g. manager spending more time at the workstation, incentives, overtimes, paying more attention to workflow, etc.). Obviously, under this second situation of increase pressure or effort from management, the operators perform their tasks harder so as to derive the benefits thus increasing the output. This situation of increase pressure or effort from manager is equivalent to reducing the distance (x) from the workstation. It is simply called increased pressure on the workstation because, more often, the operators are said to work under pressure because of the sight of their managers

The above three scenarios were used during investigation and data collection process, and the results are presented and discussed below.

Results and discussion

The results are discussed as per the three scenarios above, with scenario 1 being split into three sub-scenarios: 1.1, 1.2 and 1.3.

<u>Under scenario 1.1</u>, it was practically observed that the closer the manager (or decision-making process) was to the operators (+ $x \sqrt{0}$ *i.e.* for positive x with x approaching zero from above or manager getting closer to the shop floor)

- 1. the <u>fastest speed</u> of flow (i.e. the highest the production rate) occurred when management focused on maintaining or reducing the pressures on the workstations or operators. In this case, the operators appreciated the fact that management was committed to appraise their performance that should end with proper recognition
- in other cases, the operators/workstations experienced <u>higher pressures</u>, which in some instances led to <u>reduction</u> in production rates or slight <u>increase</u> in the production rates when compared to case 1 above. Some operators even complained that they were not reporting to Top Managers who were at their shop floors.
- 3. In some cases, minor incidences of <u>reduced production speeds</u> with very high working pressures from the operators/workstations were also recorded and were explained to be due the fact that the operators were panicking because their managers were around them leading to the operators not being able to focus on producing the products. Basically, those involved were of the view that could be expressed in their own words as "keep Top Management away from stalking our daily activities".

These above three sub-scenarios were all as predicted by the Bernoulli Principle and can be used for job appraisals i.e. reducing x may lead to increase speed with constant pressure from workstation (case 1.1.1); or higher pressure with either reduced or increased speed as in case (1.1.2); or a complete reduction in speed due to very high (uncontrollable) pressure on the operators (as in case 1.1.3).

<u>Under scenario 1.2</u>, where (x=0) or decisions were made at the level of the operators or the operators were given powers to make decision as in Total Productive Maintenance (TPM) where operators are empowered to take decisions, it was observed that

4. There were cases of both increase in production rates and increase of pressures from the workstations/operators since the workloads of the operators were indirectly increased



- 5. In the case where operators were empowered to reduce the pressures from workstations such as in Total Productive Maintenance (TPM), the highest speed of production or the highest production rates were recorded.
- 6. In the case where the pressures on the workstations were unchecked and they became so high, less increase in production speeds or complete reduction in production rates were observed.

<u>Under scenario 1.3 of $(-x\uparrow 0)$ </u>, where the management was advised to approach the operators in their decision making processes with the perception that the managers were there just to endorse the views of the operators and to recognise them for those views, it was observed that

- 7. Cases of very high production rates were achieved where the pressures on the workstations were monitored so as not to have them increased.
- 8. Cases of not so high production rates or reduced production rates were also recorded when the pressures on the workstations were not monitored to stay the same or were allowed to increase due to abuse of procedures by operators since they were the ultimate decision makers with no checks-and-balances.

<u>Under scenario 2 or bottleneck situation</u>, it should be recalled that this is the situation where there is an increase in pressure on/from the workstation.

- 9. Changing management structure as outlined in scenarios 1.1 -to 1.3 above, then the production speeds or production rates were equally getting higher, as the management gets closer to the operators.
- 10. For a constant management structure, it was observed that increased bottleneck brought about a reduction in production rate.

Hence the best approach to manage bottleneck was to bring management closer to the bottleneck workstations/operators.

Under scenario 3 or additional investment or increase effort or pressure from managers, it was observed that

- 11. the production rate increased
- 12. there were also increase fatigue and Musculoskeletal Disorders (MSD)

It can be seen from all these subcases (1)-(12) above that the correct application of the Bernoulli Principle can give useful information about the sources of success and threats in a SWOT analysis [12]. It was generally observed that increased pressures on workstations or increased bottlenecks brought about reduced production rates, and an approach of alleviating it was by suggesting that management should get closer to the sources of the bottlenecks.

Quantifying the parameters of the Bernoulli Equation is a bit involved. It requires quantifying bottlenecks or pressures from bottlenecks, the closeness of management to the shop floor and the production rate. Quantifying the first two parameters are not trivial issues since it involves quantifying human inward attributes and also one mostly encounters additional constraints such as scrap production with variable respond times or speeds with which management gets closer to the workstations.

Conclusion

As a concluding remark, with a manager being defined as someone who is allowed to take critical decisions represented by x, it has been demonstrated by employing Bernoulli Principle that bottleneck can be addressed by reducing the tall management structure or empowering the operators to make decisions (although with checks-and-balances); and investing additional resources or pressure from management may also be exploited for solution although it has its own cost implications. It is advisable to bear in mind that there is a need for a proper interpretation of the word "pressure", most especially where it is coming from. It would also be interesting to quantity those parameters of Bernoulli Equation so that the outcome can be fully quantified, which is a bit involved and is a subject matter of a different study.



Acknowledgement

This material is based upon work supported financially by the National Research Foundation. Any opinion, findings and conclusions or recommendations expressed in this material are those of the authors and therefore the NRF does not accept any liability in regard thereto.

Reference

- [1] Resnick, R. and Halliday, D. (1960), section 18-4, Physics, John Wiley & Sons, Inc.
- [2] T.D. Eastop and A. McConkey, Applied Theromodynamics for Engineering Technologists, 5th Ed., pearson Prentice Hall, (1993), England
- [3] R.L. Mott, Applied Fluid Mechanics, 6th Ed., pearson Prentice Hall, (2006), Singapore
- [4] P.F. Ostwaald &J. Munoz, Manufactruing Processes and Systems, 9th Ed., john Wiley & Sons, (1997), Canada
- [5] J.F. Douglass & R.D. Mathews, Solving Problems: Fluid Mechanics, Vol.1, pearson Prentice Hall, (1996), England
- [6] J.A. Tompkins, J.A. White, Y.A. Bozer and J.M.A. Tanchoco, *Facilities Planning*, 4th Ed. John Wiley & Sons, Inc. Printed in the United States, 2010.
- [7] J. M.P. Groover, Automation, Production Systems, and Computer-Integrated Manufacturing, 3rd Ed., Pearson education Inc, printed in the United States, 2008.
- [8] "Hydrodynamica". Britannica Online Encyclopedia. Retrieved 2008-10-30.
- [9] Streeter, V.L., Fluid Mechanics, Example 3.5, McGraw-Hill Inc. (1966), New York.
- [10] Mulley, Raymond (2004). Flow of Industrial Fluids: Theory and Equations. CRC Press. ISBN 0-8493-2767-9., 410 pages. See pp. 43-44.
- [11] TB Tengen (2013), Impacts of Random Demand and Cycle Times on Workstation Percentage Load of flow-Line Production, in Proc. of the 43nd International Conference on Computer & Industrial Engineering, 16th - 18th October, Hong Kong, pages [191]-1-7, ISSN (USB Media): 2164-8670, ISSN (online): 2164-8689
- [12] Heizer and B. Render, Operations Management, 10th Ed. Pearson Education Limited, Printed in the United States, 2011.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

CRITICAL SUCCESS FACTORS FOR IMPLEMENTING A LABOUR PRODUCTIVITY IMPROVEMENT INITIATIVE IN A COMPETITIVE SOUTH AFRICAN MANUFACTURING PLANT FOR GREATER INTERNATIONAL COMPETITIVENESS

R.N. Govender¹ and B. Emwanu²

¹School of Mechanical Industrial and Aeronautical Engineering, University of the Witwatersrand, South Africa riselgov@gmail.com

²School of Mechanical, Industrial and Aeronautical Engineering, University of the Witwatersrand, South Africa Bruno.Emwanu@wits.ac.za

ABSTRACT

High manufacturing competitiveness in South Africa has the potential to contribute towards mitigating the poor economic performance and high unemployment rate prevailing in the country. However, manufacturing competitiveness is lacking, majorly due to labour productivity issues that are idiosyncratic to South Africa. One of the dilemmas is how to increase manufacturing productivity without major capitalisation, as doing so usually leads to redundancy and retrenchment of employees. A possible solution is to implement manufacturing performance improvement programmes without major capital investment (or capital-labour substitution). Although various programmes exist and are successful elsewhere in the world, these have been found to fail in South African firms due to implementation challenges.

This study assesses the results from implementing a manufacturing performance improvement programme called Integrated Work Systems (IWS) in a South African factory of a multinational company, which has already proven successful in the multinational's factories in other parts of the world, to determine the critical success factors and develop a framework for the successful implementation of such a programme in South Africa.

It is expected that the critical success factors for implementation in a South African firm will differ from other parts of the world, providing valuable information for practitioners.

¹ Corresponding author



1 INTRODUCTION

1.1 Background and motivation

The research is based in a multinational fast-moving consumer goods (FMCG) company ABC that has recently embarked on implementing a manufacturing performance improvement programme in a number of its factories around the world. The method called 'Integrated work systems' (IWS) was developed by P&G (another multinational FMCG company). P&G has factories globally running at 85-90 percent equipment efficiency, which is extremely high compared to ABC. They have achieved this through combining various best practices and partnering with Toyota to develop a customised manufacturing system after over 30 years of development (now called IWS) [1].

Broadly summarised, IWS aims at improving equipment efficiency through integrating a set of activities, systems and tools to eliminate manufacturing losses and defects. This is achieved by engaging the entire organisation, and through autonomous maintenance practices, builds the culture and capability of machine operators and the line leadership structure [1]. The system combines elements of Lean in waste elimination, six-sigma in process control, Total Quality Management (TQM) and Total Productive Maintenance (TPM) to drive equipment efficiency [1]. IWS pillars focus on the production line and involves rigorous monitoring to eliminate line stoppages through daily management systems such as clean-inspect-lubricate (CIL), Centre-line and Defect Handling that serve as preventive countermeasures to any loss or defect experienced [2].

A 12-week proof of concept (PoC) was conducted in three ABC factories in 2014 around the world and after being deemed a success (after demonstrating significant improvements on pilot production lines in key performance indicators such as increased throughput by 20 percent in Russia and decrease in line stoppages by 54 percent in Switzerland) was rolled out to strategic ABC factories including South Africa.

The purpose of the research is to assess the results of implementing a manufacturing performance improvement programme without major capital investment in a South African factory that has already proven to be successful in similar factories in other parts of the world. Critical success-factors are identified that are used for the development of a framework for successfully implementing manufacturing performance improvement initiatives in South Africa without major capital investment.

1.2 Research question and hypotheses

The question is: 'What are the critical success-factors for implementing a manufacturing performance improvement initiative that is not technology or automation-related, in a South African manufacturing environment with idiosyncratic labour characteristics such as low productivity, inadequate skills, poor education, and industrial relations dynamism and related unrest?

The hypotheses are:

- H1: A manufacturing improvement programme that is not driven by capital investment can achieve significant change in manufacturing performance in a South African plant not withstanding labour constraints characterised by low productivity, inadequate skills, poor education and strong trade union influence
- H2: Critical success-factors for implementing a manufacturing improvement programme that is not driven by capital investment in a South African plant differ from other manufacturing improvement programmes that have been successful elsewhere
- H3: Operators in the selected South African plant will generally demonstrate a different view from senior staff regarding how the improvement programme should be implemented, given their background of poor education and inadequate skills.

1.3 Assumptions

The following assumptions were made:

- The ABC SA factory is a microcosm of the South African manufacturing environment as the social characteristics and challenges are similar to those faced by other factories in the country, that includes the environmental influences such as shortage of skilled labour, more cost-competitive international factories and inflexible labour regulations that are similarly faced by manufacturers in the country.
- Implementation of the IWS project influences the workforce and organisational culture, and implementation of the initiative requires employing change management.
- The limiting factors influencing the manufacturing sector including education levels, skills availability, government support and exchange rates will continue to exist in the foreseeable future.



2 LITERATURE REVIEW

The South African manufacturing sector is judged to be in decline according to measures such as the Kagiso Manufacturing Purchasing Managers Index (PMI) [3], the World Economic Forum's [4] Competitiveness index and Global Manufacturing Indices [6]. A Deloitte [6] report pointed out that one of the key issues in South Africa is that of labour costs increasing at a faster rate than productivity. According to the report, the issue of labour costs and availability is driven by shortcomings in the country's education, skills development, spatial development and community safety. With the bulk of South African manufacturers producing basic or low-technology products, competition from countries with lower unskilled labour costs is stronger [6]. Additional challenges to productivity during the tough economic conditions that South Africa is in [7] includes the availability of capital to invest in technology and automation as firms come under increasing financial strain, as well as the weakening rand that hinders the ability to import equipment and raw materials [8]. This provides a challenge for local manufacturers to improve cost-competitiveness to survive in the global market place.

Human capital in South Africa is often regarded as a constraint on the economy and particularly the manufacturing sector's productivity [9] [10] [11]. According to Kleynhans and Labuschagne [9], the nature of labour supply quality and availability in South Africa is restrictive. These views suggest that the workforce is generally a constraint on South African manufacturers' international competitiveness. The literature points out that the negative effect of labour supply on productivity is caused by the mismatch of skills supply and demand where there is high availability of low to unskilled labour with unmatched demand for greater skills that is partly attributed to the poor education system [10], inflexible labour laws and regulations that favour employees such that additional difficulties and costs are incurred by employers [9], as well as increasing events of industrial action and social unrest that directly and indirectly impacts economic and plant productivity [11][9].

Manufacturing performance improvement initiatives (MPIPs) in this paper refer to implementation of Lean, continuous improvement and improvement of organisational culture or change management, all with the purpose of increasing productivity, quality or reliability of the production process. Various frameworks are suggested by studies such as Ahrens [12] and Jozaffe [13] on the basis of surveys and investigations into why manufacturing improvement initiatives tend to fail or lack sustainability in South Africa compared to other parts of the world such as Japan and Germany. The literature on manufacturing improvement specific to South Africa is focused on lean implementation. Such literature generally overlaps with change management, changing organisational culture and continuous improvement. Nevertheless, only a few studies could be found regarding similar manufacturing improvement initiatives that were implemented in a South African plant and assessed for impact (success or failure). In particular, a study on positive cultural impact that can be directly related to a quantified labour productivity measure (continuous improvement philosophy e.g. lean thinking) for a South African manufacturing plant could not be found and is therefore scarce.

Based on the findings of the studies, there is some indication that continuous or manufacturing improvement initiatives that affect the ways of work or culture of a plant's workforce have a very low success rate both locally and internationally. The study in this paper, based in a large South African manufacturing plant, sets out to demonstrate a measured productivity improvement, without capital-labour substitution, while incorporating the critical success-factors identified from the literature and confirmed through empirical methods hence making a contribution.

To undertake measurements for the study, performance indicators in the literature were considered and the most relevant key performance indicators (KPIs) were found to be overall equipment effectiveness (OEE) and throughput [14]. Average number of production line stoppages per day or 'stops' was another KPI chosen for analysis in the study because it is fundamental to the IWS methodology, as a leading indicator of production efficiency.

2.1 CSF framework from the literature reviewed

A thematic method of content analysis was used to develop a CFS framework from the literature [15]. Critical success factors (CSFs) from studies on manufacturing improvement initiatives and their success or failure [12][16][13][17][18]were grouped according to themes and consolidated where CSFs were closely related. The CSFs and the corresponding themes are indicated in <u>table 1</u> below. This framework formed the basis for the survey used in the study at ABC SA.



SAIIE27 Proceedings, 27^{th} - $29^{th}\,\text{of}$ October 2016, Stonehenge, South Africa $\ensuremath{\mathbb C}$ 2016 SAIIE

Table 1: CSFs from the literature review

#	Critical success-factor	Theme
1	Leaders and supervisors motivate, coach, train and facilitate the work of those adding value rather than to tell them what to do	Bottom-up approach
2	Employee pull (employees fully understand the benefits of the programme for themselves)	Bottom-up approach
3	Employees are free to allocate time to improvement (empowerment)	Bottom-up approach
4	Resources freed up by productivity gains are reinvested into the search for still greater improvements	Bottom-up approach
5	Converting from top-down leadership to bottom-up initiatives	Bottom-up approach
6	Getting shop floor commitment and employee trust	Employee engagement
7	Involving operators through empowered Kaizen teams	Employee engagement
8	Enrolment of stakeholders for commitment, i.e. workers' council	Employee engagement
9	Avoidance of any linkage between the project and headcount reduction	Employee engagement
10	Employees understand company objectives	Employee engagement
11	Employees are engaged proactively on the project throughout the company	Employee engagement
12	Employees learn from personal and colleagues' positive and negative experiences	Employee engagement
13	Creation and communication of a vision attainment plan	Employee engagement
14	Linking successes experienced to the project	Employee engagement
15	Employee input for measuring performance to stimulate understanding and accountability	Employee engagement
16	Management demonstration of trust in employees' abilities	Employee engagement
17	Finding a good change agent or champions to remove blocks in the organisation	Key resource
18	Experienced and full time programme facilitator	Key resource
19	Organisational structure should enable the programme to succeed	Key resource
20	Executive or senior manager appointed to oversee the project	Key resource
21	Creation of a programme office or lean promotion office to support implementation	Key resource
22	Board and top management actively driving and supporting change	Leadership commitment
23	Top management presence and availability on the shop floor, engaging with employees	Leadership commitment
24	Managerial push (mandatory participation in workshops and training)	Leadership commitment
25	Executives join kaizen events on a regular basis	Leadership commitment
26	Managers actively demonstrate commitment to the CI programme at all levels	Leadership commitment
27	Building internal customer-supplier relationships (by integrating support functions) such that employees work effectively across boundaries at all levels of the organisation	Organisational integration
28	Integrating suppliers and customers into the transformation	Organisational integration
29	Employees have a shared set of cultural values that influence the way CI is incorporated in everyday work	Organisational integration
30	HR Partnership with production management with involvement in issues and concerns of people on the shop floor	Organisational integration
31	Employee recognition or reward system linked to the programme	Recognition and reward
32	Performance monitoring	Regular review of performance
33	Define measures for critical processes	Regular review of performance
34	Availability of a crises or strategic need that motivates the organisation to change	Strategic importance
35	Top management support should be established before communicating or	Strategic importance



	beginning implementation	
36	Company structure, systems and procedures are regularly assessed to ensure the CI system is supported	Support system
37	Beginning as soon as possible with an important and visible activity	Timing and planning
38	Setting a Kaizen agenda for the organisation	Timing and planning
39	Management identification of which tools will work best in their organisation	Timing and planning
40	Target areas of rapid success first and communication of successes	Timing and planning
41	Organisational stability at each stage before advancing implementation	Timing and planning
42	Comprehensive training such that everyone has necessary skills and understands the philosophy	Training
43	Standardised portion of a manager's work day	Work standards
44	Standardisation of work for operators and lower-level employees	Work standards
45	Standardised management system	Work standards
46	Discipline in adhering to standardised work	Work standards
47	Individual and group learning is captured and deployed systematically	Work standards

3 METHODOLOGY

A mixed method approach, both qualitative and quantitative, is used. For qualitative, a survey is carried out to determine participant views and characteristics in the population that have a relationship to performance trends[19], while for quantitative a document review is done to study the recorded data.

3.1 Document review

This involved reviewing trends of key performance indicators by analysing weekly OEE and brand changes, average weekly number of stops, and throughput since implementation of IWS on Line E, past OEE performance of ABC SA's factory, planned versus actual capital expenditure in ABC SA's factory.

3.2 The survey method and participants

Pre-testing of the questionnaire was done through a set of interviews with a sample of the population of ABC SA employees that was to eventually complete the survey. They were asked for approximately ten CSFs from their perspective, to be checked against the questionnaire's list of CSFs to confirm completeness and applicability of the success-factors in the survey.

Everyone directly involved in the IWS programme participated in the final survey, making a total of 37 respondents. This includes operators of two production lines, technicians responsible for both lines, Shift Leads (supervisors), the line structure responsible for this section of the factory including Line Lead, Process Lead and Maintenance Leads, Manufacturing Manager, IWS Programme Manager, and programme coaches.

A Likert scale was used for respondents to rate each success-factor on the level of impact it had on transforming culture or improving performance. To improve the quality of answers given, respondents were additionally asked the extent to which each success-factor was applied in ABC SA. This was to mitigate the risk of respondents rating a success-factor highly because they like the idea of the factor itself even though it wasn't a prevalent feature in ABC's implementation. This becomes useful during analysis as a check for consistency whereby factors that were deemed by respondents to be partially applied or not applied are given a lower impact on the results, regardless of the Likert rating given. The outcomes should therefore be success-factors considered to be prominent in ABC SA's implementation, and thought to have a highly positive impact on the results and/or culture.

3.3 Survey data analysis

Survey data were prepared by checking if the data for all variables fall within the rules of the survey. There are two categories of variables in this study, namely the respondents and the success-factors (SFs), which were treated separately. The first check was conducted on the 94 SF variables with 37 cases (responses) and no violations were found. The second check was conducted on the 37 respondents as variables with 94 cases (two answers for each of 47 SFs). In this instance two violations were found against the rule of maximum percentage of cases in a single category. These two respondents were therefore removed from the data, resulting in 94 items and 35 cases for further analyses. A test for unusual cases in the data was then conducted using SPSS to identify any cases that may warrant further examination. This procedure creates a model to identify clusters within the data and then isolates cases outside the clusters as anomalous cases [20]. The benefit of this



procedure with regards to the current study is that it also assigns cases to peer groups through its analysis of relationships between the variables that are the basis of the clustering process [20]. No anomalies were found in the 35 cases, which were statistically split into two peer identification groups.

Two rankings were required by respondents for each SF to reduce bias and intended to prompt the respondent into thinking more objectively about each SF and the role it played at ABC SA. SFs ranked according to mean criticality rating were identified and deemed to be the CSFs forming the basis of the framework recommended by this study.

An aggregated analysis of the survey results according to the themes behind each SF in various ways provided further insights into where industry practitioners and managers should focus their efforts, as well as greater direction when planning improvement initiatives. Two types of graphs were developed to provide these insights. A pie chart of CSF themes for each respondent group takes the top 20 ranked CSFs according to the themes assigned to each and shows which themes appear most frequently as a percentage, where a frequency of 20 would be 100 percent. This shows which themes the respondent group was most drawn to in answering the survey (even though they did not know which CSF belonged to which theme), indicating the areas that industry practitioners and managers should focus on when implementing initiatives like IWS. Because themes that were allocated to a greater number of CSFs would have a higher probability of showing up in the pie chart with a higher percentage, a bar graph was developed to indicate out of the frequency of a certain theme in the 47 CSFs, how many times the theme appeared in the 20 most critical CSFs for a given respondent group (figure 5).

4 RESULTS

4.1 Impact of the MPIP in ABC SA

Figure 1 below indicates key performance results for the duration of the study on the pilot production line in ABC SA including OEE and average number of stops per day and throughput. Note that baseline (BL) results at the start of each graph indicate the average weekly performance of the total month's data before implementation of the programme.



Figure 1: Trends of OEE and stops for pilot line in ABC SA

<u>Table 2</u> below summarises the KPIs before, during and after the implementation period. Implementation of IWS began in April (quarter two) of 2015, with roll out onto three lines out of 27 by quarter three and a number of IWS principles were partly applied throughout the factory in this period. OEE for the given period and recent history is provided in <u>figure 2</u> below.



Line E	Baseline	Weeks 1-12 average	Weeks 13-16 average	Improvement
OEE (%)	42	45	54	29 %
Average stops per day (#)	376	235	150	60%
Throughput (volume)	4.61	4.73	5.42	18%

SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Table 2: Analysis of KPIs since IWS implementation in ABC SA



Comparing trends of OEE and capital expenditure (CapEx) shows a particular relationship in the graph of Figure 3 below.



Figure 3: Comparing OEE and Captial Expenditure (CapEx) at ABC SA

4.2 CSFs of the MPIP implementation in ABC SA

4.2.1 Survey results - CSF rankings

The top 20 CSFs according to the survey results for the total respondent group is shown below in Table 3. SFs ranked according to mean criticality rating were identified and deemed to be the CSFs forming the basis of the framework recommended by this study. The mean score (out of 5) and percentage score are shown adjacently. Shading used in darker colour (e.g. green) highlights the higher values obtained from respondents while lighter shade (e.g. shades of yellow) indicates lower values.For the ranking, the darker shade indicates the first ten SFs considered most critical by respondents while the lighter shade indicates the next 10 most critical SFs.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa $\ensuremath{\mathbb C}$ 2016 SAIIE

Table 3: CSFs for total respondents

Ranking	Success-factor description	Application		Criticality	
1	Top management joins improvement events/meetings on a regular basis	4.17	83%	4.40	88%
2	Addition of leading KPIs that operators understand (stops, MTBF)	4.43	89%	4.37	87%
3	Clearly defined measures for key performance areas, with regular monitoring	4.37	87%	4.34	87%
4	Targeting areas of rapid success first	4.37	87 %	4.34	87%
5	Clearly defined measures for critical processes, with regular monitoring	4.17	83%	4.34	87 %
6	Management identification of which tools will work best in their organisation	4.11	82%	4.29	86%
7	Financial investment to support programme success	4.31	86%	4.26	85%
8	Setting and communicating an improvement agenda and roll-out plan	4.29	86%	4.26	85%
9	Employee 'pull' the new ways of working as they understand the benefits of the programme for themselves	4.11	82%	4.26	85%
10	Standard work with visual management	4.11	82%	4.26	85%
11	Standardised management system	4.23	85%	4.23	85%
12	Managers actively demonstrate commitment to the programme at all levels	4.20	84%	4.23	85%
13	Executive or senior manager appointed to oversee the project	4.34	87 %	4.17	83%
14	Discipline in adhering to standardised work	4.03	81%	4.17	83%
15	Organisational structure should enable the programme to succeed	4.00	80%	4.17	83%
16	Reliable data enabling regular detailed analyses	4.11	82%	4.14	83%
17	Conducting an important and visible activity of the programme as soon as possible (e.g. leadership deep clean)	4.11	82%	4.14	83%
18	Comprehensive training such that everyone has necessary skills and understands the philosophy	4.09	82 %	4.14	83%
19	Standardisation of part of a manager's work day	3.97	79 %	4.14	83%
20	Employees learn from personal and colleagues' positive and negative experiences	4.09	82 %	4.11	82 %

When the results of <u>table 3</u> above are separated into the two category of participants in the study, namely operators and senior staff, table 4 below is obtained. The shades used here (green colour) indicate SFs that are common to both operators and senior staff.



Table 4: Comparing	CSFs for op	perators and	senior staff
--------------------	-------------	--------------	--------------

Operators		Senior staff		
Ranking	Success-factor description	Ranking	Success-factor description	
1	Addition of leading KPIs that operators understand (stops, MTBF)	1	Clearly defined measures for key performance areas, with regular monitoring	
2	Management identification of which tools will work best in their organisation	2	Reliable data enabling regular detailed analyses	
3	Top management joins improvement events/meetings on a regular basis	3	Standardisation of work for operators and non-management employees	
4	Setting and communicating an improvement agenda and roll-out plan	4	Clearly defined measures for critical processes, with regular monitoring	
5	Employee 'pull' the new ways of working as they understand the benefits of the programme for themselves	5	Targeting areas of rapid success first	
6	Targeting areas of rapid success first	6	Top management joins improvement events/meetings on a regular basis	
7	Managers actively demonstrate commitment to the programme at all levels	7	Standardised management system	
8	Clearly defined measures for critical processes, with regular monitoring	8	Financial investment to support programme success	
9	Clearly defined measures for key performance areas, with regular monitoring	9	Conducting an important and visible activity of the programme as soon as possible (e.g. leadership deep clean)	
10	Standard work with visual management	10	Addition of leading KPIs that operators understand (stops, MTBF)	

4.2.2 Survey results - CSF Themes

The frequency of theme representation in the theoretical framework of 47 CSFs is shown below, followed by theme representation according to the ranking of top 20 CSFs by the total respondent group.



Figure 4: CSF Themes from theoretical framework



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE





40%

60%

80%

100%

20%

0%

5 DISCUSSION

5.1 Impact of the MPIP in ABC SA

The trend lines in Figure 1 of increasing OEE and decreasing number of daily stops show that the programme had a significant and almost immediate positive effect on manufacturing performance of the lead line in ABC SA. The gradients of the trend lines for 'OEE' and 'stops' indicate that 'stops' (the downward trend) improved at a faster rate than OEE (the upward trend) for observed period of 28 weeks after implementation commenced. This is understandable because the IWS methodology is focused on reducing the most frequent type of stops on a daily basis that should in turn increase OEE [1]. 'Number of stops' is therefore a leading indicator of efficiency or performance (less stoppages equals higher running or uptime), while OEE is a lagging indicator in relation. The survey results for the total respondent group found "the addition of a leading indicator that operators understand" to be second most highly ranked CSF (Table 3: CSFs for total respondents). This could indicate that 'stops' is a measure or KPI that is more effective on the production floor than OEE possibly because operators find it more tangible and easier to understand and influence instead of OEE. Significant throughput improvement of 18 percent after 16 weeks on the pilot line, shown in Table 2: Analysis of KPIs since IWS implementation , is the main business benefit in a factory as it generally enables the reduction of overheads such as energy usage, overtime and crewing. This table also shows a reduction from 376 to 150 stops on average per day. This less tangible benefit could hypothetically have a positive impact on employee morale as operators would exert less energy in repairing line stoppages throughout their shift.

Knowing that ABC factories are constantly pursuing OEE improvements, it is evident from the trends in Figure 3 (of OEE and CapEx) that OEE improvements came largely as a result of capital expenditure in ABC SA and efforts in more recent years to improve OEE were largely unsuccessful and resulted in stagnating performance from 2010 to 2014. With the OEE results in Figure 2 showing an increase from 55.5 in 2014 to 58.8 in Q3 2015, evidence of performance data indicate that the MPIP of IWS is delivering breakthrough performance improvements in light of ABC SA's recent history, and without capital-labour substitution (indicated by the lack



of CapEx). This is heavily supported by the KPIs tracked for the lead line in OEE, number of stops and throughput.

In light of these results, the research value seems quite significant in understanding what the CSFs were that drove these KPI improvements in ABC SA for managers and practitioners in the South African manufacturing community.

5.2 CSFs of the MPIP implementation in ABC SA

Table 3 presents the ranking of CSFs for the total group of respondents. Top management is seen to have the biggest influence on results and driving the sought after culture by regularly joining meetings or programme events. Even though this factor was not applied to its fullest extent in relation to other factors according to the respondents (83 percent), the criticality of 88 percent indicates that if more of this were to happen then results in ABC SA could be even better. 'Clarity of what performance areas to focus on' and 'which processes are most critical', both with clearly defined measures, is another point deemed vital for successful implementation by ABC SA, seen in CSF's three and five. 'Targeting areas of rapid success first' as a CSF ranked in the top five is not surprising when aiming to drive a certain culture and implementing a programme that requires operator buy-in to work. This is because it gives the opportunity to everyone associated with the project to see positive results and experience the improvements, which in turn drives motivation and belief in the programme. 'Standardisation of work' and 'standardisation of the management systems' as CSFs 10 and 11 illustrate how important it is for everyone to know what to do, and for everyone to do it in the same way by following the same routine.

From Figure 4 (the pie chart of theoretical framework themes) it is worth noting the prominence of certain themes, with employee engagement seen as the biggest area to consider when implementing a MPIP with 17 percent of the SFs, followed equally by organisational integration, bottom-up approach, leadership commitment, key resources and work standards all with 11 percent. However, according to the total group of survey respondents in ABC SA with regards to IWS implementation, the results of Figure 5 depict a contrasting picture. Only seven out of the 11 themes derived from the literature chosen by respondents feature in the 20 CSFs, with a notable absence being 'employee engagement' that is clearly seen by ABC SA as less critical. 'Standardisation of work', 'timing and planning the way the programme is executed' and 'focusing regularly on key processes and performance dimensions' are deemed the most critical areas to consider, each represented by 20 percent.

Figure 6 indicates that all four out of 47 SFs under the theme 'regular review of performance' were deemed critical by appearing in the top 20 CSFs by the total group of respondents. This factor has a possible correlation with South Africa's labour supply idiosyncrasies where it was discussed in the literature review that skill and education levels are low, indicating that close controlling, reviewing and focusing on key processes and performance dimensions on a regular basis is necessary to influence results and culture positively. Work standardisation, particularly for non-management employees is logically a vital input for the achievement of discipline and productivity improvement in a workforce with low skills and education levels - this is reflected in Figure 6 with 80 percent of items of the theme considered critical.

The top 10 SFs chosen to be most critical for both the operator and senior staff group are compared in Table 4. 50 Percent of the CSFs are common to both operators' and senior staff's top 10, as indicated by matching colours (numbers 1, 3, 6, 8, 9 on the left, and numbers 1, 4, 5, 6, 10 on the right). It could be argued that the two CSFs marked in a separate colour (number 10 on the left and 3 on the right) are also common as they are similar in construct in that both speak to standardisation of work. As seen in <u>table 4</u>, the top 10 SFs chosen to be most critical for both the operator and senior staff group are compared and it is seen that the CSFs from the top 10 SFs have fifty percent that are common to both operators' and senior staff (this is shown by shading numbers 1, 3, 6, 8, 9 on the left, and numbers 1, 4, 5, 6, 10 on the right). It can also be argued that two other CSFs marked (number 10 on the left and 3 on the right) are also common as they are similar in construct in that both speak to standardiset in the percent that are common to both operators' and senior staff (this is shown by shading numbers 1, 3, 6, 8, 9 on the left, and numbers 1, 4, 5, 6, 10 on the right). It can also be argued that two other CSFs marked (number 10 on the left and 3 on the right) are also common as they are similar in construct in that both speak to standardisation of work; the only difference between these two being that one includes visual management and the other emphasises non-management.

From the findings, it is noteworthy that both operators and senior staff hold similar views regarding CSFs, which indicates alignment between managers and the shop floor operators in understanding the programme and what drove its success thus far, despite generally different perspectives. None the less, differences are to be expected in the above comparison, specifically where operators and senior staff are exposed to different elements of the programme. This can be seen in CSFs 2 and 8 of the senior staff's list, where operators do not have as much interaction with aspects like data analysis beyond a single shift, as well as implementation costs.

6 CONCLUSIONS AND RECOMMENDATION

In line with the hypotheses proposed:

1. Implementation period showed 29 percent OEE improvement, 60 percent reduction of stops per day and 18 percent increase in throughput. A negligible OEE improvement was seen in preceding years. H1 was therefore supported.



- 2. Of the top three CSF themes in the ABC SA survey results, only 'work standards' features in the top five themes of the theoretical framework. H2 was therefore supported.
- 3. Although many of the CSFs ranked in the top 10 were common for both employee groups, comparing the themes indicates a significant difference as to what factors each group would emphasise in implementation of the improvement programme. H3 was therefore supported to an extent.

A framework is proposed similar to Table 3: CSFs for total respondents that should be used by managers and practitioners in South African manufacturing plants when implementing MPIPs (manufacturing performance improvement programmes) that require a change in mind-set or shift in culture on the factory floor for the new processes, ways of working or systems to succeed.

7 REFERENCES

- [1] Proctor & Gamble Co. (2014, June 4). Integrated Work Systems.
- [2] EFESO Consulting. (1999, December). Procter & Gamble: objective zero defects and 100% involvement. *Results*, pp. 5-7.
- [3] Lings, K. (2015). SA manufacturing had a shocking start to 2015. Retrieved September 3, 2015, from Stanlib Economic Focus: http://www.stanlib.com/EconomicFocus/Pages/SAmanufacturing2015.aspx
- [4] World Economic Forum. (2015). South Africa Competitiveness Index. Retrieved from Trading Economics: http://www.tradingeconomics.com/south-africa/competitiveness-index
- [5] Deloitte. (2013). Enhancing Manufacturing competitiveness in South Africa. Deloitte & Touche.
- [6] Alves, P., and Edwards, L. (2005). South Africa's Export Performance, Determinants of Export Supply. Cape Town: School of Economics: University of Cape Town.
- [7] Lipton, D. (2015). *Reflections on South Africa's Challenges and Opportunities for Reform*. Cape Town: International Monetary Fund at the University of Cape Town.
- [8] England, A. (2015, April 9). Weak rand fails to ease South African manufacturers' woes. Retrieved November 8, 2015, from Financial Times: http://www.ft.com/cms/s/0/23266d4e-de95-11e4-b9ec-00144feab7de.html#axzz421WsmHCN
- [9] Kleynhans, E., and Labuschagne, J. (2012). The quality of human capital in South Africa: Evidence from a firm survey. *African Journal of Business Management Vol.* 6(23), 6914-6923.
- [10] Bhorat, H., Hirsch, A., Kanbur, R., and Ncube, M. (2013). *Economic Policy in South Africa, Past, Present, and Future*. Oxford Companion to the Economics of South Africa.
- [11] Go, D., Kearney, M., Korman, V., Robinson, S., and Thierfelder, K. (2009). Wage Subsidy and Labour Market Flexibility in South Africa. University of Sussex.
- [12] Ahrens, T. (2006). Lean production: Successful implementation of organisational change in operations instead of short term cost reduction efforts. Germany: Lean Alliance.
- [13] Jozaffe, L. B. (2006). Implementing lean manufacturing to improve production efficiency in the manufacturing operations at the Aspen General Facility. Faculty of business and economic sciences at the Nelson Mandela Metropolitan University.
- [14] Jonsson, P., and Lesshammar, M. (1999). Evaluation and improvement of manufacturing performance measurement systems - the role of OEE. International Journal of Operations & Production Management, 55-78.
- [15] Wilkinson, D., and Birmingham, P. (2003). Using research instruments: a guide for researchers. London and New York: Routledge Falmer: Taylor & Francis Group.
- [16] Bhuiyan, N., and Baghel, A. (2005). An overview of continuous improvement: from the past to the present. *Management decision*, 761 771.
- [17] van der Merwe, K., Pieterse, J., and Lourens, A. (2014). The development of a theoretical lean culture causal frameowrk to support the effective implementation of lean in automotive component manufacturers. The NMMU Business School, Departmeth of Industrial Engineering.
- [18] Vermaak, T. (2008). Critical success factors for the implementation of lean thinking in South African manufacturing organisations. Johannesburg: Faculty of Management at the University of Johannesburg.
- [19] Leedy, P., and Ormrod, J. (2013). Practical Research. New Jersey: Pearson.
- [20] IBM. (2011). SPSS Help.


FACTORS AFFECTING THE QUALITY OF SERVICE DELIVERY IN A GOVERNMENT DEPARTMENT OVERSEEING EFFICIENT MAINTENANCE OF PHYSICAL ASSETS IN HEALTHCARE SERVICES IN A SELECTED PROVINCE IN SOUTH AFRICA

T. Taruvinga¹ and B. Emwanu²

¹School of Mechanical Industrial and Aeronautical Engineering, University of the Witwatersrand, South Africa gaiyos07@gmail.com

²School of Mechanical, Industrial and Aeronautical Engineering, University of the Witwatersrand, South Africa Bruno.Emwanu@wits.ac.za

ABSTRACT

An efficient delivery of healthcare service depends on reliable, safe and readily available infrastructure and equipment for use. This necessitates a well-planned and managed maintenance programme that is able to guide and support maintenance of the infrastructure and equipment. Inefficiencies on the maintenance programme have detrimental effects on the healthcare services. This study investigates the factors affecting quality of service delivery by a selected provincial government department of infrastructure development (PGDID), particularly factors that affect management and its maintenance arm in ensuring maintenance of the physical assets at public health institutions. The study was based on the perceptions of the PGDID personnel that were directly involved in the service delivery. Data were collected using a structured questionnaire from a sample of 91 PGDID personnel consisting of engineers, inspectors, foremen, artisans, Supply Chain Management (SCM) staff and regional directors. The results of the study provide insight on the importance of soft issues such as bureaucracy, motivation of staff, and communication as being comparable to the execution of the maintenance task itself. Jointly, the soft issues play a significant role and should be taken seriously as it has implications when planning maintenance of physical assets in healthcare services at provincial level.

¹ Corresponding Author.



1. INTRODUCTION

Increased allocations of the national budget towards the health sector in the recent years has seen the construction of more health and social development physical infrastructure and capital assets [1][2]. However, inadequate attention has been given to the maintenance of already existing infrastructure and equipment, and this has led to a deteriorating state of infrastructure and equipment, particularly at the oldest institutions [1]. On the other hand, 80% of South African population rely on the government for healthcare, as influenced by socio-political and economic issues [3], hence a need for government institutions to be continuously and efficiently operating all the time. It is imperative to prolong the life of public infrastructure, improve accessibility and keep equipment in a functional state for the proper and safe functioning of the health and social development institutions in order to maximise the benefit to the communities they serve [4][5]. Management of operations and maintenance of infrastructure and equipment are critical to the service delivery of the departments of Health and Social Development.

The selected provincial government department of infrastructure development (PGDID) is mandated to maintain or procure all maintenance services for the maintenance of all the infrastructure and equipment at the Department of Health and Social Development institutions in the selected province of South Africa. Ever since PGDID was formed, there has been a negative public/government/stakeholder perception on the service delivered by PGDID [6]. The challenges affecting PGDID are not limited to but include the under-expenditure of the allocated budget when there is increased deterioration of public infrastructure and equipment at institutions and the frequent equipment breakdowns at institutions resulting in unavailability of critical equipment such as boiler, chiller plant or autoclave are adversely affecting service delivery of the client department. Also, there is a high number of cases of poor or non-performance by the contractors PGDID appoints or sub-contracts to provide service to its client department and the slow turn around in processing payments to consultants on services rendered to GDID.

The objectives of the investigation are to identify the factors that are affecting the effectiveness of maintenance management by GDID and to provide recommendations for improvements on the factors identified to be affecting service delivery.

2. LITERATURE REVIEW

Equipment and buildings deteriorate depending on age, usage, method of design, materials and environmental conditions [10]. These conditions make it essential to perform maintenance work, so as to preserve the buildings and equipment and allow them to continue operating efficiently [11]. Maintenance work is also done to ensure that facilities are safe to work in, and that they meet specified working standards. According to [12], maintenance management ensures the protection of capital investments and guarantees acceptable working environment for users. However, maintenance work increases operational costs, and some organizations resort to inadequately maintaining their facilities in order to lower costs, but by so doing, they compromise safety. Neglecting maintenance has long-term negative financial consequences due to reduced asset life and premature replacements [10]. Also, inadequate or lack of attention given to maintenance of facilities often results in large and unbudgeted expenditure occurring [13].

The requirements for good practice in maintenance management have been established over a considerable period, but the achievement of good practice is by no means universal [14]. While some organizations consider planned preventive maintenance more economical as compared to ad hoc replacement of parts when they fail, some consider otherwise. In the case of hospital equipment and buildings, their performance depends largely on continuous and planned maintenance [15].

2.1 Maintenance management

Maintenance management is the planning and controlling of all aspects associated with maintenance of a plant or facility. It involves the administrative, financial, and technical framework for assessing and planning maintenance operations on a scheduled basis. An effective maintenance management system involves three separate entities of work, *viz*, maintenance, inspection and verification, which can be done by different groups in the same company or different companies specifically subcontracted for the purpose [16].

Good maintenance management keeps equipment and infrastructure in a state that helps protect operators' health and safety, as well as from environmental damage. It also yields longer asset life with fewer breakdowns, which in turn result in lower operating costs in the long run [17]. A good maintenance management system needs to be backed up with well-informed and competent maintenance staff.

Maintenance management is classified into planned and unplanned, where unplanned maintenance refers to work necessitated by unpredicted breakdowns or damage due to external forces [10]. Planned maintenance management is further subdivided into three categories namely reactive or corrective maintenance (CM),



preventive maintenance (PM) and predictive maintenance (PdM) [18]. Each of the strategies is useful under different circumstances and has associated advantages and disadvantages, such that organizations are faced with the option of choosing one strategy or more to follow [12].

Optimal maintenance of equipment cannot be achieved by utilising only one maintenance strategy, but a wellorchestrated maintenance program will include a practical and reasonable mix of the different maintenance strategies. The right maintenance mix should be adopted, where the three types of maintenance strategies are distributed across different equipment. The right maintenance mix should be decided upon after considering a number of factors, for example, the type of equipment or assets and costs of replacement *versus* costs of maintaining [18]. Figure 1 provides a correlation between the maintenance costs and repair costs associated with the three different maintenance strategies.



Figure 1: Association between repair cost, maintenance cost and maintenance strategies [18]

Figure 1 shows that Predictive Maintenance (PdM) generally has the highest maintenance cost, but it will result in the lowest repair costs. On the other hand, Reactive Maintenance (CM) has the lowest maintenance cost but the highest corresponding costs associated with asset repairs. The costs of repairing equipment are generally higher as compared to maintenance costs, implying that in order to lower costs, CM should be allocated a smaller ratio of the maintenance mix [19]. As much as PdM result in the least repair cost, the installation of indicators that help predict failures brings additional costs. With the given information, Preventive Maintenance (PM) should be allocated the highest percentage of about 70% in the maintenance mix [20].

2.2 Factors affecting maintenance management at health institutions

Public hospitals find it difficult to compete in a market-driven economy due to their nature, characteristics and structure. In general, the structure in the hospitals promotes inefficiencies and inflexibility, and allows for the imposition of bureaucratic impediments to operational effectiveness [21]. These factors instil researchers to investigate public hospitals with regard to their performance and maintenance, in order to suggest ways of improving services provided by these institutions.

[22] investigated the strategies that are employed in rolling-out a maintenance plan and identified the challenges in managing hospital maintenance in the Eastern Cape Province. The factors identified in the study included realistic budgets, adherence to occupational health and safety regulations, engagement of qualified and skilled maintenance staff, good record management in relation to the equipment under maintenance, availability of tools and materials and keeping abreast with the latest technologies and trends of machinery



advancement. [22] recommended that for maintenance to be effective, strategic planning that takes into account carefully thought-out maintenance management systems should be the first step in the direction of setting out definite tangible objectives and goals.

To explain the ineffective and sub-sequent poor performance of medical equipment in hospitals and health centres in developing countries, [23] incorporated a management perspective onto the way in which health care equipment is acquired and used. The management related factors identified include poor equipment acquisition procedures, poor maintenance and repair procedures, poor equipment replacement procedures, poor management of human resources development and poor equipment performance evaluation procedures. Other factors identified in the study which contribute to poor maintenance management at health and social development institutions include lack of trained maintenance staff, lack of trained equipment operators, weak technical support system, inadequate finances for equipment management, poor logistics support (transport, information), poor equipment conservation culture and weak infrastructure and organisation [23].

[24] identified the main reasons for the poor service delivery as attributed mainly to the non-existence of good governance framework with proper checks and balances particularly in the purchase of sophisticated equipment which were underutilized or never used due to lack of technical expertise, bad usage and maintenance which reduce life of equipment. Other important maintenance management related factors identified were the over-purchase of accessories and extra spare parts and modifications to facilities initially unforeseen due to lack of expertise in choosing appropriate equipment, over-buying due to corruptive deals, lack of standardization resulting in increased cost of spare parts or additional purchases and extra workload on limited competent staff, excessive down time of equipment due to lack of preventive maintenance, inexperience in repair and lack of spare parts and unfavourable purchasing contracts. The high percentage failure of equipment led to resource wastage and poor quality of health care delivery in public sector hospitals.

[25] identified maintenance management system adopted by an organisation as an important factor for the service delivery of maintenance of equipment. A maintenance management system that prioritizes medical equipment for maintenance decisions to ensure safety and reliability of medical equipment was proposed. It is highlighted that in order to mitigate functional failures, significant and critical devices should be identified and prioritized. Reliability Centred Maintenance (RCM) was proposed as a solution to achieve this.

2.3 Best practices in maintenance management

Best practices are good practices that have been used and worked well elsewhere. They are proven and have produced successful results. In maintenance management, [26] provided a set of best principles as given below:

- The maintenance budget must be drawn from the maintenance needs on the ground and adequate funding must be provided
- Occupational health and safety regulations must be followed
- Maintenance plans must be in place to guide maintenance and are adhered to
- Maintenance records of service history must be available
- Maintenance is carried out by skilled personnel
- Personnel operating machinery are competent
- The right tools and adequate spares must be available for maintenance
- Maintenance and equipment replacements are carried out as per manufacturer's specifications and scheduling
- Competent contractors are sub-contracted for specialised services
- Procurement processes enable the appointment of the best competent contractor
- Availability of adequate support resources such as transport, computers, fax and telephones.

3. METHODOLOGY

The purpose of the research was to investigate the factors affecting service delivery by PGDID so as to give recommendations on how to improve PGDID maintenance management. A quantitative research methodology was used in the study. The method makes use of numerical data to explain a particular phenomenon and can be used where there is a large amount of data that can be standardised [27]. A quantitative descriptive research approach was used in the study where some of the data collected was descriptive whilst the other data was numerical. The type of research was chosen because substantial amount of data could be collected at low costs, and the method gives a description of the state of affairs as it exists, which suits the purpose for this research to explain why PGDID was failing on its mandate.

Preliminary interviews in the form of a focus group which consisted of eight PGDID engineers were conducted to discuss the problems PGDID is facing. The factors affecting PGDID service delivery were identified and hypotheses were drawn based on the factors identified by the focus group interviews. Self-administered



questionnaires (which were semi-structured) were then designed based on the hypotheses to be tested and administered to 120 respondents that are directly involved in PGDID service delivery for their participation in the research. The research participants included PGDID engineers, inspectors, foremen, artisans, supply chain management (SCM) staff and regional directors.

A set of research hypothesis tested are as follows:

- i. The maintenance management system (MMS) in place is not effective
- ii. Infrastructure and equipment are not effectively maintained
- iii. PGDID is not adequately resourced to provide an efficient service
- iv. The PGDID Supply Chain Management (SCM) department is not efficient
- v. The staff are not motivated and morale is low
- vi. There is no effective communication between top management and the staff
- vii. There are no significant effects of bureaucracy on the service delivery

The Cronbach's alpha was calculated to test the reliability and internal consistency of the factors measured and a reliability of 0.70 or higher is acceptable [28].

4. RESULTS AND DISCUSSION

The mean score for each hypothesis tested obtained from a semi-structured questionnaire was calculated as shown on Table 1. The mean score determined whether the hypothesis was accepted or rejected.

ltem	Hypothesised factors	Mean	Hypothesis
		score	
1	The maintenance management system (MMS) in place is not effective	2.624	Accepted
2	Infrastructure and equipment are not effectively maintained	2.308	Accepted
3	PGDID is not adequately resourced to provide an efficient service	3.399	Rejected
4	The Supply Chain Management (SCM) department is not efficient	2.849	Accepted
5	The staff are not motivated and morale is low	2.719	Accepted
6	There is no effective communication between top management and	2.505	Accepted
	the staff		
7	There are no significant effects of bureaucracy on the service delivery	3.684	Rejected

Table 1 : Summary of results of hypothesis testing

4.1 Maintenance Management System (MMS)

The research findings indicated that the maintenance management system (MMS) currently in place at PGDID is inefficient. This explains why the department is experiencing budget under-expenditures and failing to maintain infrastructure and equipment effectively as evidenced by the research findings. Similar findings were obtained by [22] on the investigation of effective management of machinery in government operated hospitals in the Eastern Cape province of South Africa. Based on these two research findings, it raises concerns on the MMS that are in implementation at the government maintained institutions in South Africa. A sound MMS will inform of the maintenance needs of equipment and infrastructure, maintenance resources, spares and tools necessary and when they should be available. The budget that is required to implement maintenance is also extracted from the MMS. MMS will also assist maintenance record keeping which is essential for an effective maintenance management by PGDID.

4.2 Supply Chain Management department efficiency

The perceptions shown by the research findings indicate that the SCM is ineffective which explains some of the problems the department is facing which include the sub-standard quality of service provided by the contractors, the poor or non-performance by contractors which has led to a significant number of contracts



terminated. Since PGDID externally procures the bulk of services it provides to the client department, thus the quality of service the PGDID provides to its client department is dependent on the quality of contractors it appoints, therefore the SCM department within PGDID plays a critical role and it is imperative that it operates efficiently. The research findings indicate the SCM contractor appointment process is not yielding the appointment of the right competent contractors which is significantly contributing to the poor service delivery by PGDID.

4.3 Availability of resources

It is essential that adequate resources be available for effective maintenance of infrastructure and equipment. The research findings show that PGDID is generally resourced as shown by the average mean rating of 3.399 given to the availability of resources. This is contrasting to the research findings obtained by [21] where insufficiency of funds for maintenance programs was identified as one of the main factors affecting maintenance management of public hospital buildings in Lagos state, Nigeria. This differentiates the situation of PGDID maintained institutions to those of Lagos government run institutions. However, since PGDID is perceived not delivering a quality service to its client department despite being resourced, it is an indication that the resources available to PGDID are under-utilised. However, the availability of resources in PGDID also provides a foundation on where improvements can be built upon.

4.4 Motivation and morale of staff

The employees play a significant role in the quality of service delivery by PGDID. It is important that the staff be motivated and the morale to be high in order to obtain maximum productivity from the employees. From the below average rating of 2.719 that the research results showed on the perception of staff motivation and morale, the poor service delivery by PGDID can be attributed to lack of motivation and low morale of the staff.

4.5 Communication

The research results have showed a negative perception on the effectiveness of communication within PGDID. This has contributed to the staff not understanding the vision and mission as well as a perception that management is not addressing the concerns of the staff as shown by the research findings. Communication plays an important role in an organisation. It is important for management and the staff to have aligned goals in their approach to maintenance. The staff needs to be informed of the vision and mission of the organisation and understand the strategies that are in implementation to achieve the set targets as well as their role in contributing to the attainment of the vision and mission. The staff also need to be able to inform the management of the challenges and concerns on the ground which need to be addressed. This can only be achieved through effective communication. Lack of effective communication within PGDID also contributes to the poor service delivery by the department.

4.6 Bureaucracy in PGDID

The survey results indicated that there is bureaucracy in the organisation and that it is impacting on the service delivery by PGDID. The bureaucracy is causing delays in contractor appointments even when unplanned critical breakdowns occur which is resulting in negative perception on effective maintenance of infrastructure and equipment by the client department. Implementation of changes even where they are critical is also slow due to bureaucracy within the system. The bureaucracy is attributed to top management centralising decision making which is also affecting the morale of the PGDID staff who indicated that they are not consulted in decision making. For PGDID to give efficient service to its client department, it is imperative that the barrier of bureaucracy be eliminated.

5. CONCLUSION

The study findings indicated that the perceptions on the current maintenance practices show inadequacy for an efficient service delivery by PGDID. It is concluded that there is poor service delivery by PGDID to its client department which is attributed to: an ineffective maintenance management system in place; infrastructure and equipment not being effectively maintained; inefficiency of the supply chain management within PGDID; low morale and motivation amongst the staff; the ineffective communication in PGDID and bureaucracy within the organisation systems.

6. **RECOMMENDATIONS**



- It is recommended that PGDID make use of term contracts with contractors for the maintenance of its electro-mechanical equipment to improve responsiveness and to minimise the effects of bureaucracy that is inherent in PGDID procurement processes.
- Delegation of authority to regional management will assist in minimising the bureaucracy that is in the PGDID systems.
- PGDID needs to increase the percentage of services that are done by its in-house staff so as to increase its responsiveness in case of breakdowns. Recruitment of technical staff is essential to achieve this.
- Renovations on all old infrastructure to give a facelift to the institutions and the replacement of all obsolete equipment is recommended to reduce age related breakdowns on machinery and improve the plant availability and productivity.
- It is recommended that PGDID introduce a computerised maintenance management system (CMMS) to guide maintenance. CMMS will enable a better record management system and informs the maintenance plans, material required and budget needs.
- Rating of performance of every contractor appointed after execution of services so that all poor performing contractors can be removed from the PGDID contractor database.
- Top management is recommended to engage in regular meetings with the staff and that issues raised by the staff be taken into consideration in decision making so that there will be buy-in from the staff. Internal workshops will assist in informing the staff of the vision and mission of PGDID and also that they understand their role in contributing to the department's vision and mission.
- Best practice benchmarking can be used by PGDID to improve its service delivery. Private hospitals, clinics and forensic mortuaries which are perceived to be efficiently maintained can be used as benchmarks for PGDID institutions.

REFERENCES

[1] Nkomfe, M., 2012. 'Selected province' Finance MEC Budget Speech on 06 March 2012, Johannesburg.

[2] Bathembu, C., 2011. SA to pump billions into health care. [Online]. Available: <u>http://www.southafrica.info/business/economy/policies/budget2011e.htm</u>. [Accessed: 28-02-12].

[3] South Africa info, 2012. Health care in South Africa. [online]. Available <u>http://www.safrica.info/about/health/health.htm</u>. [Accessed: 29-02-2012].

[4] Strock, C., Songer, A. and Borrego, M., 2008. A social-technical systems model for public health infrastructure in resource-poor communities. CIB W107 Construction in Developing Countries International Symposium. Trinidad and Tobago.

[5] Knotek, R. and Stenerson, J., 2006. Mechanical Principles and Systems. New Jersey: Pearson education, Inc.

[6] Nkosi, B. S., 2012. Meeting briefing by the MEC at 'PGDID' on 28 January 2012, Tulisa Park, Alberton.

[7] Chauke, K., 2012. Hospital CEO's are now responsible for minor maintenance of hospitals. [Online]. Available: <u>http://www.did.gpg.gov.za/Pages/HospitalCEO%E2%</u> 80%99sarenowresp onsibleforminormaintenanceofhospitals.aspx. [Accessed: 28-02-2012].

[8] Modipa, M., 2012. Meeting briefing by the HOD at 'PGDID' on 28 January 2012, Tulisa Park, Alberton.

[9] PGDID, 2011. 'Selected province' Annual report 2010/2011.

[10] Chanter, B. and Swallow, P., 2007. Building Maintenance Management. Oxford: Blackwell Scientific.

[11] Moubray, J., 1997. Reliability Centred Maintenance. (2e). Butterworth Heinemann.

[12] Cloete, C. E., 2002. Introduction to Facilities Management. South African Property Education Trust. Business Print Centre, Pretoria.

[13] Martin, D., 2006. The A-Z of Facilities and Property Management. Thorogood.

[14] Turrell, P., 1997. Small is Different: A Strategy of Effective Management of Maintenance in Nonprofitmaking Organization. The Royal Institution of Chartered Surveyors: 1-3.

[15] Shohet, I. M. and Perelstien, E., 2004. Decision Support Model for the Allocation of Resources in Rehabilitation Projects. Journal of Construction Engineering and Management 130 (2): 1-5.

[16] Misra, K. B., 2008. Handbook of Performability Engineering. London: Springer Publishing Company.

[17] Marquez, A. C., 2007. The Maintenance Management Framework: Models and methods for complex systems maintenance. London: Springer_Publishing Company.



[18] CHOA, RDH Building Engineering Ltd and the Real Estate Foundation of BC, 2011. What is Maintenance? Information Bulletin No. 5. [Online]. Available: <u>http://www.choa.bc.ca/resour</u> ces/bulletin05_what_is_maintenance.pdf. [Accessed: 15-06-12].

[19] Lazarus, S. J. and Hauptfleisch, A. C., 2010. The application of Facilities Maintenance within the public sector: An exploratory study. ASOCSA 2010-18. University of the Free State, Bloemfontein.

[20] Suttel, R., 2006. Preventive HVAC Maintenance is a Good Investment. Buildings. [Online]. Available: http://www.buildings.com/ArticleDetails/tabid/3321/ArticleID/3183/Default.aspx [Accessed: 26-06-12].

[21] Adenuga O. A., Odusami, K. T. and Faremi, J. O., 2007. Management of public hospital buildings in Lagos state, Nigeria. London: RICS, Georgia Institute of Technology

[22] Gatang'i, P. G., 2010. Effective management of machinery in government operated hospitals. MSc Thesis, Nelson Mandela Metropolitan University, Port Elizabeth.

[23] Rammelzwaal, B. L., 1997. The effective management of medical equipment in developing countries. FAKT Project number 390, Stuttgart.

[24] Dasanayaka, S. W. S. B., 2008. Performance of Health Care Equipments in the Public Sector Hospitals in the Eye of Good Governance, A Case Study Based on the Sri Lankan Public Sector Hospitals, Department paper, University of Moratuwa, Moratuwa.

[25] Taghipour, S., Banjevic, D. and Jardine, A. K. S., 2010. Prioritization of medical equipment for maintenance decisions. Journal of the Operational Research Society 1: 1-22.

[26] Wireman, T., 2005. Developing Performance Indicators for managing maintenance (2e). London: Industrial Press Inc

[27] Babbie, E, 2008. The Practice of Social Research (11e). Belmont: Thomson Wadsworth.

[28] Nunnally, J C, 1978. Psychometric theory (2 e). New York: McGraw-Hill.



ENGINEERING WORK INTEGRATED LEARNING: A CASE STUDY IN PROBLEM-BASED RESEARCH AND DEVELOPMENT PROJECTS

M. Della Tamin¹, J. Meyer² and H. Nel³ ¹ School of Electrical and Electronic Engineering Sciences University of Johannesburg, P.O. Box 524, Auckland Park 2006, South Africa Tel: +27 11 559 2462, Fax: +27 11 559 2462 dellatamin@gmail.com

² School of Electrical and Electronic Engineering Sciences
University of Johannesburg, P.O. Box 524, Auckland Park 2006, South Africa
Tel: +27 11 559 2462, Fax: +27 11 559 2462

johanm@uj.ac.za,

³ School of Electrical and Electronic Engineering Sciences
University of Johannesburg, P.O. Box 524, Auckland Park 2006, South Africa
Tel: +27 11 559 2462, Fax: +27 11 559 2462

hannelien@uj.ac.za

ABSTRACT

The case study presented in this paper is based on an internship program at a company located in Johannesburg South Africa. It aims to show an innovative problem-based research and development approach through an engineering work-integrated learning program with technologist engineering interns. Through the case study, theoretical, practical and proper decision making have emphasized the understanding of problem solving strategy in research and development with interns. A thorough follow-up of the processes described in this paper could potentially enable decision makers to develop the skills of more engineers. A working model is presented to show how the system can be replicated.

Keywords: Work-Integrated Learning, Research and Development, Internship, Interns, systems engineering.



1. INTRODUCTION:

Over the past two decades, a considerable number of people have moved to South Africa for diverse reasons such as political and economic stability and advancement [1]. Similarly, the population growth has increased over the same period [2, 3]. The increasing number of people has created the need for infrastructure development, better life styles and commercial opportunities which stem from engineering activities. Engineering activities require specific competencies [4]. Many of these competencies, including infrastructural and industrial design as examples, must be addressed by engineers. Engineers are considered crucial to economic development, sustainability and national growth. The question that arises is whether adequate engineering skills are available to address the growing South African economy.

According to the Engineering Council of South Africa (ECSA), one engineer serves 3000 people in South Africa in comparison with 227 in Brazil and 543 in Malaysia. This statistic validates the shortage of skilled engineers in South Africa [5].

In order to address the shortage, the development of engineering skills can be achieved through normal tuition supported by work-integrated learning (WIL). Practically, most graduates lack industrial problem solving exposure and effective mentoring which may be attributed to the absence of industry placement. However, for certain companies, employing engineering interns shift them from their economic targets since the time of productive staff is absorbed to mentor future graduates. Due to the recursive nature of this situation, it becomes a problem for students to graduate with their National Diploma (ND) of which the Work Integrated Learning (WIL) component is an exit-level outcome as required by ECSA. Furthermore, the graduation certificate confirming a degree or an ND in engineering technology does not automatically translate to experience, expertise and resultant employment. Consequently, recent graduates seeking their first job opportunity may lack realistic expectations and experience relevant to the kinds of challenges and responsibilities encountered in the work place [6]. According to Leinhardt *et al.* (1995) and Entwistle and Entwistle (1997), the theoretical learning from universities has been of huge criticism for focusing much effort on 'declarative knowledge' and unsatisfactory efforts on relevant professional understanding [7, 8]. In response, the combination of theoretical knowledge and industrial exposure could be seen as the bridge to link graduates with industry for the development of skills and value-adding engineers.

In this paper, the case study presented indicates how a well-managed ND engineering technicians' intern can be integrated into industrial projects to add value; maintaining budget and time constraints in a scenario where a new group of up to 300 interns is enrolled every 6 months.

2. BACKGROUND

The introduction of the practice for linking theory and practical work to enhance education is unclear. Historically, a practical program was introduced for engineering and architecture students in 1903 at Sunderland Technical Collage in Northern England [9]; and in1906, at the University of Cincinnati, Herman Schneider, an engineering lecturer believed that professional concepts and skills acquisition required more than just classroom lecturing. In order to master professional concepts and skills, he understood that students must undertake intense practical experience development. He suggested the key solution would be to alternate employment experiences on a work site with theoretical work on campus [10]. The notion of "learning by doing" has been in formal operation for over a century and from the late 1950s to mid-1980, led by the United States of America (USA), this notion has expanded globally [10].

Many papers have been published by a large spectrum of authors on the WACE International Conference portal to examine the effects of WIL on the students [11]. In 2009, Matsutaka *et al.* studied students' consciousness and its related effects on their academic and employment outcome by means of a sample of over 1300 students at a University in Japan [12]. Carlson and Kwan (2010) explored the effects of WIL on learning outcomes by means of 1040 students from a Hong-Kong University [13], and in 2009 and 2010, Green and Mendez focused smaller groups of engineering students to study the effects of work employment on academic performance in some Universities in Great Britain [14, 15].

In Australia, research has shown the importance and effectiveness of WIL in preparing university students for their work readiness on completion of their studies. From the employers' point of view, as revealed by the survey conducted for the Office of the Chief Scientist (OCS) in 2014, the key difficulties encountered when recruiting graduates were deficiencies in interpersonal skills, understanding of business and workplace experience [16]. As such, practical learning and critical thinking can be visualized as the two most important attributes a graduate should be equipped with on entry to the workplace [17]. According to the Australian Survey of Student Engagement (AUSSE), students who have experienced WIL are less likely to drop out of the University as compared to those who have no WIL experience [16]. The data retrieved from the AUSSE state that students who have experienced WIL have been learning things in their academic program that 'improved knowledge and skills that contribute to employability' [18, 19].



Most of underdeveloped countries such as Uganda, Cameroon and Ivory Coast are agricultural-based economy countries with young and fast growing populations. Uganda, as an example, has a population estimated at 34 million people with a growth rate of 3.2% per annum, endowed with low industrialization and engineering value addition [20]. In spite of the economic structural transformation over the past two decades, it still faces huge challenges, which include high unemployment and inadequate skills development [21]. Due to the lack of organisational forecast, some efforts in addressing WIL are often not visible. Providing an answer to the skills development would definitely warrant economic growth through entrepreneurship and job creation. This applies to most of the underdeveloped countries worldwide.

In South Africa (RSA), the unemployment rate is in the range of 26.7% for the first guarter of 2016 [22]. Globally, the economies of first-world countries are recovering from recession whereas South Africa is entering one which may be attributed, to some extent, to the lack of a proper employment policies. In RSA, some opportunities aimed at contributing to the economic growth also lack specific engineering skills such as electrical, electronic, civil and chemical engineering as listed on the national scarce skills list [23]. Therefore, it is imperative and expected from the Universities to produce graduates who are job ready through WIL. But in reality, a large amount of students are not exposed to WIL due to a deficiency of opportunities offered in the local industrial environment. Additionally, the labor market fluctuates between skills scarceness and graduates without work. According to Coll and Zegward (2006), skills scarceness and graduates without work might arise because most students do not have behavioral or soft skills such as analytical and team work skills, and are therefore unable to organize and manage themselves [24]. The Resolution Circle (RC), a company based in Johannesburg, is a dedicated research and development environment with multi-disciplinary student teams. These teams are supported by internal technical experts. RC is an initiative of the University of Johannesburg (UJ) that aims to address specific technological challenges brought forward by industry partners. The RC internship WIL program was designed to address the lack of opportunity for engineering skills development in industry and to provide a platform for graduate employability skills development.

Previous studies conducted in first-world countries determined the technical and individual abilities required of engineers by today's industry [25, 26]. These studies have influenced countries like the United States of America [27], the United Kingdom [28] and Australia [29] to revise their national accreditation criteria for engineering programs. In these countries, the accreditation requirement has moved from what is being taught to what is being learned [30]. Therefore, current engineering programs must demonstrate that graduates achieve sets of learning outcomes but they have autonomy in implementation methodology [31]. Two concepts arise from the studies, including problem-based learning and project-based learning and in South Africa, these concepts are integrated in the WIL outcomes. Problem-based learning assists students to develop problem-solving, team and interpersonal skills; whilst project-based learning introduces methods for effective design and completion of projects.

Several universities that have introduced problem-based learning courses, including Monarch University, Curtin University and Griffith University in Australia. Project-based learning courses have been introduced at Hogskole Telemark in Norway, the Colorado School of Mines, Rose-Hulman University of Technology in the USA, Aalborg and Roskilde in Denmark, and TU Berlin in Germany [31]. In South Africa, the combination of problem-based and project-based learning is provided at RC under the umbrella of WIL.

3. A RESEARCH AND DEVELOPMENT WIL CASE STUDY

3.1 The Resolution Circle

In South Africa, universities receive subsidy from the government when students graduate. However, many students are unable to graduate since they have not received placement in companies to fulfil their WIL requirements. In order to overcome income loss and address this challenge, the University of Johannesburg (UJ), has invested in the Resolution Circle to recruit students for their WIL internship program. An on-site range of services is orchestrated in the company (RC) to support marketable projects throughout the commercialization life cycle, including research, development, prototyping, and incubation. Through commercialization, interns who have shown entrepreneurial skills and have developed product based ideas are selectively granted few sits to our incubation program.

RC attempts to close the gap between industry and academia by providing training opportunities and industrial support. The services offered by RC are presented in Figure 1.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa $\ensuremath{\mathbb C}$ 2016 SAIIE



Provide training opportunities

Provide industry support



Figure 1 indicates that the services offered by RC lie between providing of training opportunities for students and industrial support. The interns, led by technicians and engineers, are exposed to experimental training and commercialization projects through the research and development (R&D) department. Another important focus of RC is research-based projects. The layout of the RC R&D WIL program with its functionality is presented in Figure 2. Figure 2 displays the interaction between the main components of the company with respect to the WIL program. As required by ECSA, the ND WIL students must undertake practical training for a period of one year of which they do Practical 1 (P1) and Practical 2 (P2) for the first and second semester respectively in their final year of study.



Figure 2. Resolution Circle R&D WIL program layout

One hundred and fifty engineering technician interns from the electrical, mechanical, electronics and instrumentations disciplines are enrolled each semester to complete P1 training for a period of six months. The interns are exposed to six trades which include boiler making, carpentry, turn and fitting, automation, electronic and electrical skills. Exposure to the six trades trains the interns with the necessary hand skills required by the respective disciplines. When the R&D department is executing a project with a manufacturing component, the P1 interns may be involved as well. After completing the six trades, the P1 graduates receive certificates as proof of completion and then move to the P2 phase for another 6 month period.

During P2 the interns are mostly exposed to soft skills (technical report writing, formal presentation, interview readiness and engineering ethics), problem solving, team work and design. The WIL program includes students from other Universities as well.

3.2 Resolution Circle P2 WIL Layout

In the RC P2 WIL program layout, the interns are grouped per engineering discipline for a period of two months. They are exposed to individual stations for two weeks and then rotate to the next station. In mechanical engineering as example, the stations are: G-coding [32], Solidworks [33] and 3-D printing, small-scale



manufacturing, and Revit mechanical-electrical-piping [34]. During the two week period, the interns are guided and monitored by trained instructors (senior and junior technicians). Relevant computer-assisted design (CAD) or other software packages are introduced to all the interns at the respective stations. The station arrangement for electrical engineering is presented in Figure 3.



Figure 3. Station rotation layout for electrical engineering interns

Figure 3 indicates that the interns enrolled for electrical engineering are exposed to four stations where they learn relevant software packages such as ETAP [35] for power system, photo voltaic (PV) SOL [36] for solar PV systems and national instruments [37] programming packages. Professional report writing, presentation and entrepreneurial skills are introduced to each team. The interns enrolled for instrumentation and electronics are few and kept in small groups where they are introduced to programming using micro controllers such as Arduino [38], and conveyor belt control technologies using Allen Bradley Programmable Logic Controller (PLC) [39]. This last group is also introduced to electronic software for simulation and Printed Circuit Board (PCB) design.

Report writing is an important aspect of the WIL training program required by ECSA, and monitored weekly. At the end of each station period, the interns write group and individual reports aimed at providing personal and team work reflection and development. After the two month period rotation, all the interns are expected to use their skills to solve relevant problems from industry. The program is structured to satisfy the practical requirements set by the various universities, and to provide the interns with exposure to real-world problem solving. A minimum of 24 weeks must be completed by the interns per semester.

3.3 Resolution Circle P2 WIL R&D Problem Solving

After the initial two-month period, the interns are introduced to higher activity. During this time the P2 interns are placed in smaller groups per speciality. For diversity purposes, interns from different Universities are represented in each group.

The projects are multi-disciplinary including electrical, mechanical, instrumentation and electronic engineering. Some projects are engineering-based whilst others may not be. Clients range from government, the banking sector, industry and entrepreneurs. As projects are received, they are placed in a protocol pool and ranked by time and cost. The projects are then distributed to teams with the relevant discipline requirements; and managed by the Agile project management scheme [40]. Agile is employed since most customers are not fully aware of their requirements at the start of the project and the needs of the project cannot be clearly defined during inception. The project execution flow chart is presented in Figure 4.



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



Figure 4. Project management WIL layout.

Projects are selected from the available projects pool and a meeting is organized between the clients and the engineering teams. The engineering teams consists of senior engineers as the project owner, the R&D P2 project manager, a team of four interns per group per speciality and a senior technician per project per speciality. A statement of work is drafted and a list of deliverables are drafted from the meeting including the project due dates. In preparation of the draft proposal, the teams meet to discuss the time needed for the project, the materials or components required, the risk involved, the remedies for risk, the schedule for the deliverables and the project plan. The project backlog is often used in the Agile project management scheme [40]. The project backlog provides a clear vision as an important factor. The backlog is converted into tasks and then assigned to each team respectively during execution. In order to execute the project, a specific time frame of not more than two weeks must be defined respectively. This time frame is known as the Run [40]. A run period is the time interval when teams come together to review the project's progress. When the teams are satisfied with their draft proposal, it is forwarded to the project manager who reviews, edits and forwards it to the senior engineer responsible of the specific project. The senior engineer is the project owner and the scrum master is someone in the execution team [40]. When the senior engineer is satisfied with the draft proposal, s/he sends it to the scrum master for pricing. If modifications are required, the senior engineer sends the files back to the teams and the process reiterates until consensus is reached. When this occurs, the scrum master discusses the proposal with the client and issues the project term sheets and the quotations. The term sheets clarify stakeholder expectations and the respective deliverable dates. The project commences officially once the term sheets are signed by all stakeholders.

A senior technician can work with up to four teams in his speciality on different projects. At commencement, each team selects its leader who holds a local scrum meeting every day on the progress of the project. The technicians hold informal meetings with the project manager twice a week. At the completion of the project, the scrum master and senior engineers hand over to the customer. The project and the client satisfaction is fed back to the teams after which the project is closed.

As an example, one of the projects that involved more than one team was the electric car which comprised of the electrical, mechanical and electronic engineering teams. At the commencement of the project, all the teams meet to discuss the project backlog and the assigned tasks. The project team elected one-week runs run for the project. Team leaders were chosen and procurement and communication issues were addressed. Project management was allocated to the interns as a learning experience. The interns held local scrum meetings every morning which were monitored by a senior technician. The mechanical team was assigned to design the car, the suspension, its road worthiness and the aerodynamics and suspension system.

The tasks assigned to the electrical team were to design the electric drive system, the battery size and the electrical wiring of the car. Through team consultation, cost effective and high performance parts were designed. The task allocated to the electronic team was to design a charger for the battery bank and the lighting of the car. The project was executed according to plan and is a showcase project for WIL development.

3.4 Challenges of WIL projects

A challenge in performing projects with a new groups of interns every six months makes management of the R&D P2 project a complicated task. A new group of interns introduces new personalities and new cultural dynamics. Placing the new group at the level where the previous P2 left becomes more difficult with unwilling interns. Human behavioural and cultural implications therefore make the program both exciting and challenging. This transition and human factor challenge necessitates the introduction of a systems engineering approach.



According to the International Council of Systems Engineering (INCOSE), Systems Engineering (SE) is an interdisciplinary approach that provides the means to enable the realization of successful and complex systems [6]. SE focuses on:

- Defining customer needs and required functionality early in the development cycle;
- Documentation requirements;
- Proceeding with design synthesis; and
- Structure validation and verification whilst considering the complete problem.

SE integrates all the disciplines and specialty groups into a team effort forming a structured development process that proceeds from concept to production to operation. SE considers both the business and the technical needs of all customers with the goal of providing a quality product that meets the user needs.

Challenges experienced during the execution of WIL projects at RC include: interns with no industry experience; limited budgets; inefficient procurement processes; and cultural differences in the team composition (age, gender, religion, ethnic group and race). On the other hand, the computer labs, workshops and experienced senior engineers and technicians are available to support the intern teams. The systems engineering approach facilitates the execution of projects as follows:

- Clients' needs are defined as soon as possible in the project process as per project backlog;
- Drafting the statement of work clarifies team expectation and the project deliverables and dates,
- The teams involved in one project discuss, design, harmonize costs and optimize the human resources; and
- The system validation and verification are addressed using the systems engineering V-model.

The V-model works well with small projects where requirements are very well understood and where tasks are completed individually. The V-model is presented in Figure 5.



Figure 5. R&D Project Agile project management model.

From Figure 5, the life cycle of the project is implemented with the interaction with the client until completion.

4. RESULTS

A number of WIL projects at RC have been success stories. These include the circuit board project, the intelligent solar inverter, the electric car and the thin film solar modules testing project. The success of all the projects is ensured by a clear definition of the project scope, as well as the execution of project with the Agile project management techniques. Another important element of project success is communication amongst team members which is facilitated by the systems engineering approach.

5. CONCLUSION

The involvement of the Resolution Circle with the internship program of the ND interns has not been an easy task. The lessons learned have channeled the P2 program to important results in terms of a project's effectiveness, consistency, management and repeatability. The authors provide solutions to engineering skills development though WIL problem solving R&D at the Resolution Circle as a specific case study. The value added through the program was both shared by the interns and RC. The interns have specifically gained and earned



SAIIE27 Proceedings, 27th - 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE

their skills through the program whilst RC has contributed by providing a future strategy for engineering skills development in South Africa. Therefore, after their graduation at respective Universities, these former interns would possibly address the needs from clients and effectively contribute to the economy. As recommendation, decision makers and business owners may replicate the model presented in this paper for enhanced skills development to reduce technical skills shortage in South Africa and Africa.

6. **REFERENCES**

- [1] V. Kalitany and K. Visser, 2010, 'African immigration in South Africa': Job takers or job creators, Sajems, NS 13 (2010), ISBN: 1015-881
- [2] N. Tait, A. Whiteford, J. Joubert and J. Van Zyl, 1996, 'Socio-economic Atlas for South Africa', HSRC Publishers, Pretoria, ISBN: 0-7969-1786-8
- [3] C. Wright, 2010, 'City of Cape Town Discussion Paper, Demographics Scenario', August 2010, Strategy and Planning Strategic Information and GIS, https://www.capetown.gov.za/en/sdf/Documents/Nov2010/DemographicsDiscussionPaperAugust2010.p df
- [4] H. E. Hanrahan, N. Beute, D. M. Fraser, J. Gosling, A. Lawless, I. R. Jandrell, 2006, 'Engineering education in South Africa': In fragile good health, denial or crisis?, Third African Regional Conference on Engineering Education, Pretoria
- [5] Margret Bauer, Kevin S. Brooks, Carl Sandrock, 2014, 'Industry Expectations and Academic Practice in Control Engineering Education - A South African Survey', The International Federation of Automatic Control, Cape Town, South Africa
- [6] M. A. Malloy, 2007, 'The MITRE Corporation', 903 Gateway Boulevard, Suite 200 Hampton, Virginia 23666, INCOSE, 2007.
- [7] G. Leinhardt, K. McCarthy Young & J. Merriman, 1995, 'Integrating professional knowledge': the theory of practice and the practice of theory. Learning and Instruction, 5, 401-408.
- [8] N. Entwistle and A. Entwistle, 1997, 'Revision and the experience of understanding in N.F. Marton', D. Hounsell and N. Entwistle (eds) The Experience of Learning, Edinburgh, UK: Scottish Universities Press.
- [9] P. Franks and O. Blomqvist, 2004, 'The World Association for Cooperative Education': The global network that fosters work-integrated learning. In R.K. Coll & C. Eames, C. (Eds.). International handbook for cooperative education: An international perspective of the theory, research and practice of work-integrated learning (pp. 283-289). Boston, MA: World Association for Cooperative Education.
- [10] E. S. Sovilla and J. W. Varty, Cooperative education in the USA, past, present: some lessons learned. In: R.K. Coll & C. Eames (Eds.). International handbook for cooperative education: an international perspective of the theory, research and practice of work-integrated learning. Boston: World Association for Cooperative Education: 3-16.
- [11] Y. Tanaka and K. Carlson, 2012, 'An international comparison of the effect of work-integrated learning on academic performance': A statistical evaluation of WIL in Japan and Hong Kong, Asia-Pacific Journal of Cooperative Education, 2012, 13(2), 77-88
- [12] M. Matsutaka, Y. Tanaka & P. Churton, 2009, 'Assessing the effectiveness of co-op education in Japan': A panel data analysis at KSU, Paper presented at the 16th WACE World Conference, Vancouver 2009.
- [13] K. S. Carlson and K. P. Kwan, 2010, 'You either get it or you don't': The bifurcation of student learning outcomes in WIL context, Paper presented at WACE International Conference, Hong Kong, 2010.
- [14] J. P. Green, 2009, 'The impact of a work placement year on student final year performance': An empirical study, Paper presented at the 16th WACE World Conference, Vancouver, 2009.
- [15] R. Mendez, 2010, 'The role of work-integrated learning in academic performance': Is there a correlation between industrial placements and degree performance, Paper presented at WACE International Conference, Hong Kong, 2010.
- [16] D. Perkins, K. Pearce and J. Pearce and J. Hong, 2015, 'Work Universities Australian in STEM Report -Final report', ACER, June 2015
- [17] Deloitte Access Economics, 2014, 'STEM Employer Survey, Canberra': Office of the Chief Scientist, 2014.
- [18] H. Coates, 2009, 'Engaging Students for Success Australasian Student Engagement Report. Melbourne': Australian Council for Educational Research, 2009.
- [19] A. Radloff and H. Coates, 2010, 'Doing more for learning': Enhancing engagement and outcomes -Australasian Student Engagement Report. Melbourne: Australian Council for Educational Research, 2010.
- [20] J. Mutambi, 2014, 'Stimulating Industrial Development in Uganda Through open Innovation Business Incubators', Doctoral Dissertation Series, Division of Technoscience Studies, Department of Technology and Aesthetics, School of Planning and Media Design, Blekinge Institute of Technology, Sweden, 2014
- [21] MFPED, 2010/2011, 'Republic of Uganda-Background to the Budget': 2010/11 Fiscal Year, Kampala, http://www.finance.go.ug. 2010.
- [22] Quarterly Labor Force Survey, 2016, 'Quarter 1:2016, South Africa Statistics P0211', www.statssa.gov.za/P02111stQuater2016.pdf



- Department of Higher Education and Training, 2014, 'Top 100 occupations in Call for comments on the [23] national scarce skills list demand', Notice 380 of 2014, Staatskoerant, 23 May 2014, No. 376783. www.gpwonline.co.za
- R. K. Coll and K. E. Zegward, 2006, 'Perceptions of desirable graduate competencies for science and [24] technology new graduates', Research in Science and Technology Education, 24(1):29-58, 2006.
- R. Henshaw, 1991, 'Desirable attributes for professional engineers. In Agnew, J.B. & Creswell, C. (Eds.) [25] Broadening Horizons of Engineering Education', 3rd Annual conference of Australasian Association for Engineering Education. 15-18 December, 1991.
- J. D. Lang, S. Cruise, F. D. McVey, & J. McMasters, 1999, 'Industry expectations of new engineers: A [26] survey to assist curriculum designers', Journal of Engineering Education, 88, 1, 43-51, (1999).
- [27] ABET, 2001, 'Criteria for Accrediting Engineering Programs. Engineering Accreditation Commission of the Accreditation Board of Engineering and Technology', Baltimore, Maryland, (2001). Available on-line at http://www.abet.org/criteria.html
- [28] SARTOR, 2000, 'Standards and Routes to Registration', SARTOR (3 Rd ed), (2000). On-line from http://www.engc.org.uk/registration/sartor.asp
- Institution of Engineers, 1999, 'Australia, Manual for the accreditation of professional engineering [29] programs', Revised: 7 October 1999. Canberra: Institution of Engineers, (1999).
- [30] E. Koehn, 1999, 'ABET program criteria for educating engineering students'. International Conference on Engineering Education, ICEE'99, Paper 413, (1999). Available on-line at http://www.fs.vsb.cz/akce/1999/ICEE99/Proceedings/papers/413/413.htm
- [31] J. E. Mills and D. F. Treagust, 2003, 'Engineering Education: Is Problem based or Project based Answer', The Australasian Association for Engineering Education Inc, AAEE, 2003 ISSN 1324-5821
- S. Krar and A. Gill, Computer Numerical Control Programming Basics, Industrial Press Inc. 200 Madison [32] Avenue, New York, NY 10016
- [33] http://files.solidworks.com/pdf/
- [34]
- Autodesk Revit systems, 2008, 'BIM for MEP engineering', www.autodesk.com/revitsystems Operation Technology, 2008, 'Inc. ETAP 7.0 Demo', Registered to ISO 9001:2008, copy ISO 9001:2008, copyright 2009, [35] http://etap.com/downloads/brochures/etap-70-demo-guide.pdf
- [36] PV*SOL® Expert Version 6.0, Design and Simulation of Photovoltaic Systems Manual, Valentin Software, Inc. 31915 Rancho California Rd, #200-285 Temecula, CA 92591 USA, http://www.valentinsoftware.com/sites/default/files/downloads/handbuecher/en/manual-pvsolen.pdf
- [37] National Instruments Corporation. All rights reserved, User Manual, C 2006-2009, http://www.ni.com/pdf/manuals/374483d.pdf
- [38] https://www.arduino.cc/
- MicroLogix[™], 1000 [39] Allen-Bradley, User Manual, Programmable Controllers, http://literature.rockwellautomation.com/idc/groups/literature/documents/um/1761-um003_-enp.pdf
- R. Pichler, 2010, 'Agile Product Management with Scrum': Creating Products that Customers Love, [40] Addison-Wesley, 2010, ISBN-13: 978-0-321-60578-8



SAIIE27 Proceedings, 27th – 29th of October 2016, Stonehenge, South Africa © 2016 SAIIE



ASSESSMENT OF CONTINUOUS IMPROVEMENT INITIATIVES' IMPACT WITHIN THE MANUFACTURING SECTOR IN SOUTH AFRICA

S. Theko¹, M.G. Kanakana² & K. Mpofu³

¹Supplier and Enterprise Development Department, Automotive Industry Developmet Centre (AIDC), South Africa e-mail address: <u>stheko@aidc.co.za</u>

> ²Department of Industrial Engineering, Faculty of Engineering & Built Environment Tshwane University of Technology, South Africa e-mail address: KanakanaMG@tut.ac.za

> ³Department of Industrial Engineering, Faculty of Engineering & Built Environment Tshwane University of Technology, South Africa e-mail address: <u>Mpofuk@tut.ac.za</u>

ABSTRACT

The aspect of sustainability has been widely debated in the Industrial Engineering and Operations management fields, with researchers arguing that many companies are doing at least one initiative of continuous improvement to make their businesses profitable, by minimizing operations costs and the world-wide involvement of companies are taking part in process improvement activities. However, these arguments were based on work done in the UK and USA and not covering the South African (SA) perspective on the sustainability issue. This paper addresses the issue of sustainability with special attention to the work done by the SA government through agencies, such as the Automotive Industry Development Centre (AIDC) in order to ensure that SA manufacturing companies remain competitive. The project specifically focused on process improvements done by the AIDC. The argument is that SA companies are not sustaining these process improvements and therefore not realizing the benefits of these government funded projects. The research findings indicate that generally South African companies in manufacturing sector feel that they are sustaining continuous improvement initiatives to a minimum level, therefore new approach is needed to structure government funded initiatives differently in order to ensure sustainability at a benchmark level of 75%.

¹ The author was enrolled for an MTech (Industrial - Structured) degree in the Department of Industrial Engineering, Tshwane University of Technology

² The author was main supervisor for author one in the Department of Industrial Engineering, Tshwane University of Technology. *Corresponding author

³ The author was co-supervisor for author one in the Department of Industrial Engineering, Tshwane University of Technology. *Corresponding author



1. INTRODUCTION

Sustainability is a very important aspect in continuous improvement. It is through this aspect that productivity and efficiency benefits are realized. Statistics SA reports that: "the manufacturing sector continues to occupy a significant share of the South Africa economy, despite its relative importance declining from 19 percent in 1993 to about 17 percent in 2012 in real terms" (STATISTICS SA. Economic Growth, 2015).

The South African government in view of reports like these, is encouraging South African businesses to embark on process improvements in order to be more efficient and productive. This is evident because the government, through its agencies such as the Automotive Industry Development Centre (AIDC), The Innovation Hub (TIH), Enterprise Propeller and Productivity SA, etc. is making funding available for such developments. The AIDC is a government agency that is primarily mandated to increase efficiency and productivity more especially in the automotive sector. Funded by the Gauteng government, the AIDC is running several programmes in order to make process improvements through their Supplier and Enterprise Development Department (SEDD). All of these programmes are designed to assist companies to reduce wastes and be more efficient in their operations. They are also designed to improve the way people think in terms of continuous improvement and to help companies improve their efficiencies and maintain them. These structured programmes range from 3-month to one year long during which the AIDC proves to the participating companies that with dedications, continuous improvements are possible. Figure 1 shows some of the process improvement tools that are used by the AIDC (AIDC SEDD Templates, 2012). These tools are also widely used throughout the entire world.



Figure 1: AIDC Process Improvement tools (AIDC templates, 2012)

Although not limited to, the AIDC makes use of these tools in their specialized programmes such as Total Productive Maintenance (TPM), Lean Manufacturing and Rapid Process Improvement (RPI).

This study is done to establish if the companies, that took part in one of the AIDC's programmes, have maintained the process improvements, and have also continued implementing the methodologies as learnt during the AIDC's intervention and what impact has this made to their company. The study also aims to develop a framework which will enable the organisations in the South African manufacturing sector to sustain their continuous improvement initiatives should it be found that the sector does not maintain the improvement or if there is room for improvement.

2. BACKGROUND

2.1 Report Analysis

In 2011/12 financial year, the SED department in AIDC have interacted with 89 companies by doing at least one of the development programmes in each company. The BlueIQ annual report for the same year indicated that all companies achieved a range between 9% and 34% productivity and efficiency improvements. (GAUTENG GROWTH AND DEVELOPMENT. Automotive Industry Development Centre. 2012. Annual report. GGDA).

In 2013/14 financial year, the AIDC continued their programmes with 15 different companies and it is reported that 15% average efficiency was achieved in that financial year. The percentage here ranges from 10% to 30%. (GAUTENG GROWTH AND DEVELOPMENT. Automotive Industry Development Centre. 2014. Annual report. GGDA).



The AIDC has produced the same reports for 2014/15 financial year, with 25 companies, and showed an efficiency improvement of 45% on average. (AUTOMOTIVE INDUSTRY DEVELOPMENT CENTRE. Supplier and Enterprise Development Department. 2015. Annual report, Pretoria: AIDC).

Given the information above it is obvious that whenever the AIDC is involved in a continuous improvement initiative at a company, positive results are yielded. Since the AIDC is only at these companies for a limited period of time, it is necessary to establish whether these companies are sustaining the improvements made during anyone of the SEDD's programmes and whether they are actually using the methodologies learnt during the intervention with AIDC to improve their businesses.

2.2 Literature Review

Sustainability is viewed by the AIDC as the important measure for performance because it is through this measure that the benefits are realised. For this reason, it is very imperative for companies to have a clear strategic focus in achieving sustainability of improvements made. Ali, A.J. 2012 came to a conclusion that Strategic Focus, management of Continuous Improvement (CI) and learning/knowledge sharing were the most influential factors of sustainability of CI. Since the element of sustainability is growing at a very fast pace, sustainability thinking is challenging businesses to change how they operate (Lubin and Asty, 2010). As a developing country, South Africa cannot afford to ignore the direction that the developed countries have taken in terms of the element of sustainability of continuous improvement to gain competitive advantage and to respond to the major trends that include growing of human population, rapid response to developing economies, both of which they put a major train to the resources and create unpredictable pricing in commodities (McPhee 2014).

According to literature, several models that aim to measure sustainability in business were identified. In the main scientific databases like Scopus, Ebsco, Web of Science, and Wiley, for example, approximately 64 models were identified(Satolo and Simon, 2014).

Meanwhile, in Mexico, Suárez and et al (2011) conducted an empirical study in 49 different Mexican industries. The study focused on the application of Kaizen and reported the main causes of abandonment of Kaizen philosophy, among others, is resistance to change from employees, lack of proper implementation and monitoring of Kaizen projects. AIDC in one way or the other do meet the above challenges especially at the beginning of the project but the willingness to participate is overcome by introducing the issue of change management to participating companies. The study will also establish whether the companies continue managing this change or not.

According to Garcia and Rivera (2013), the continuous Improvement (Kaizen) philosophy, "was applied for a large number of entrepreneurs, restless and highly competitive managers that were eager to employ methodologies that strengthen the highly competitive environment where they were involved". Despite this argument, Rink J, 2005 reported that in the 3000 industrial organisations that took part in the survey conducted with the US Manufacturers, 90% had implemented continuous improvement projects; however, only 10% felt they were achieving the results they desired. Japanese companies on the other hand are seeing more value in implementing the Kaizen projects than anywhere in the world. Cheser, 1998, has conducted a research in Japanese manufacturing industries, and concluded that the Kaizen generated an increase in motivation and positive change in attitude in employees.

Bateman, (2001) argued that globally not all companies that embark on process improvements are able to sustain those improvements and continue to make further improvements. She in fact states that some of the companies will go back to where they were before the improvement was done.

The United Kingdom (UK) has adopted what they termed Industry Forum MasterClass where a formal process structure to continuous improvement was established.

In a study that was done in UK by Bateman (2001), where the various phases of an improvement activity was followed and studied over time considering the outcomes of each phase in terms of sustainability. Bateman, 2001, came up with a model for sustainability that explains the findings of her study in 2001. She concluded her study by classifying the companies into Class A to Class E (Class A being good, Class E being worse). The summary of the findings as described by Bateman (2001) is as follows:

"During the post follow up phase, Class A activities continue to improve and Class B activities only continued for few weeks and maintain the current situation. Class C activities only maintained the improvement done during the intervention. All other classes literally dropped from the achieved improvement with class E going back to the original state. Continuous Improvement takes place when team uses the tools and techniques that were learnt during the workshop to solve new and to improve further the performance of the model area. The activities that achieve this are defined as class A activities".

Figure 2 below shows a Sustainability Model as defined by Bateman, 2001.





Figure 2: Sustainability Model (Bateman, 2001)

Bateman, 2003, further looked at the differences of these classes in terms of what A and B are doing versus C, D and E, and she found out that companies in class A and B have the following in place:

- i. Daily time dedicated to 5S activities,
- ii. Quality, cost and delivery measures are displayed at cell level,
- iii. Problems are communicated properly from situational level,
- iv. Managers stay focused on continuous improvement activities,
- v. Operators are involved in decision making,
- vi. The companies are supporting "evidence based management".

3. THE SOUTH AFRICAN CONTEXT

3.1 South African operating environment

Majority of manufacturing companies in South Africa are multi-national automotive component suppliers who understand the global status core. These companies are the biggest role players in the enforcement of lean manufacturing in South Africa, hence the government is encouraging them to have a supplier and enterprise development programmes with their suppliers.

Through the AIDC programmes, it has been realized that companies are keen to pursue continuous improvement programmes. While the above is true, not all companies are realizing the full potential of such programmes. Sustaining process improvements is the key to a competitive manufacturing sector. In view of the above literature, the study is very important and relevant for South Africa to start understanding our position in terms of sustainability of continuous improvement. In realizing that the government is investing capital in industry development to enhance the competitiveness of the industry, it is also imperative to find out if the investment is worthy or to be improved.

3.2 Research Questions in South African context

The questions, therefore, are: Are South African companies in the automotive and manufacturing sectors sustaining their Kaizen projects? What interventions do these companies need in order to see value in their Kaizen projects? And what is the role of the government and the support agencies like AIDC in developing the manufacturing companies in the automotive sector?

The above literature does not make mention of the state of affairs with regards to general manufacturing in South Africa never mind the automotive industries in South Africa. This paper will therefore, focus on South African manufacturing industries in comparison with the above literature.



3.3 Current Processes and Frameworks

Currently, the AIDC has no defined model but are carrying out work in a manner that is shown below:



Figure 3: Current SEDD process (AIDC templates, 2014)

The current approach used to carry out the above programmes, is based on the target set by the AIDC for the SED department. The department then allocates the targets for each project manager or each programme. The project manager is then responsible for ensuring that the targets are met, so from marketing to project completion, project manager for each programme focuses on what the programme targets are and strives to achieve those targets.

Since the approach is target driven, marketing of the programmes depends largely on the individual project managers. Once the project manager has reached their number of companies to be sign on to the programme, then implementation with the individual companies starts immediately. The project manager will then report progress on a bi-monthly basis and the implementation results are summarized at the end of the financial year which is end of March each year. Figure 11 in section 5.3 shows the draft of the current process within the AIDC. Although the above is the way the process is conducted currently, the process is not formally described anywhere within the AIDC

The disadvantage of the current operations (figure 3) is that programmes are conducted by companies that avail themselves and but are not necessarily ready for some of these interventions. This results in companies not sustaining their improvement and therefore not realizing the benefits.



4. RESEARCH METHODOLOGY

The study was characterized by primary research which involved quantitative research methods. The research was primary in that it consisted of questionaire that was sent directly to the respondents by means of SurveyMonkey. In this way, the research was as interactive as possible as it was conducted electronically with an allowance to ask clarity to questions by e-mail. This research is based on some underlying philosophical assumptions about what constitutes a 'valid' research and the research methods that are appropriate for the development of knowledge in a given study. An example of the assuptions is that the respondents will give answers to the best of their knowledge. The research method is a strategy of enquiry, which moves from the underlying assumptions to research design, and data collection (Myers, 2009).

Thethree common classifications of research methods were considered during this research, namely: qualitative - research that relies primarily on the collection of qualitative data, quantitative - research that relies primarily on the collection of quantitative data and mixed - research that involves the mixing of quantitative and qualitative methods or paradigm characteristics. These self-explanatory terms are really dependent on the assumption made before the data collections, the way data is collected and the way that data is analysed.

4.1 Research Strategy

There is certain esearch strategies that could be considered for a particular project depending on their ability to answer the research questions and meet the objectives of the study. This could be in the form of survey, case study, experiment, action research, ethnography, and so on. (Saunders et al., 2012). All research strategies carry the same value when compared to each other hence the researcher is urged to select the one strategy that best suits their research.

The research design included literature reviews and studies where existing information was analyzed and equated to the South African environment. A descriptive investigation was conducted using a qualitative survey. This method was chosen because it allows the study of things in their natural setting, attempting to make sense of certain type of behaviours.

4.2 Target Population and sampling strategy

A questionnaire was sent in a form of monkey survey targeting a sample of 40 companies that formed part of the AIDC' Supplier and Enterprise Development programmes in the past 4 financial years. Although this study accommodated companies that took part in other programmes within the AIDC, more emphasis was placed on companies that attended the Rapid Process Improvement programme as they appear in the AIDC data-base. This means that all such companies were invited to take part in the study. In order to achieve the goal, the cluster sampling approach under probability sampling was used. The targeted population for this purpose is approximately 150 companies that interacted with the AIDC in the past 4 years.

A sample is the representative of its own population if it has to produce valid and reliable estimates of the population from which it was drawn. The Longman South African school dictionary defines a sample as a noun that means the small amount of something that shows what the rest is like, (Maskew Miller Longman, 2011). So in other words a sample is a subset of a population.

Sampling methods can either be non-probability, where sample members are not selected randomly, or probability, where sample members are selected randomly.

During this study, a probability sampling method was employed, and the following steps were followed:

- Based on the research questions and objectives, a suitable sampling frame was identified,
- A decision taken on what sample size would best represent the population,
- The most suitable sampling technique was then selected
- The sample was rechecked to ensure it is the representative of the population (Saunders et al, 2009)

For this research study, the cluster sampling approach which is classified under the probability sampling was used. The sample size was 40 companies, representing a response of hundred percent.

4.3 Data collection strategy

Each participant was requested to complete the self-administered online survey and return it to the researcher electronically via Survey Monkey so that it could be collected at a central point. It was made clear to all participants via the information leaflets that the information will be kept confidential, and that companies can choose to remain anonymous.



On receipt of the data the researcher used statistical methods such as Bar Charts, Percentage, Weighted average, etc to analyze, quantify and interpret it.

All questions in the questionnaire were structured in such a way that they aggressively determine the attitudinal instrument of the respondents.

4.4 Questionnaire construction and format used

The questionnaire in this research study was structured in such a way that it starts with all the general questions about the respondents and their personal experiences, then grouped into the following measures of lean to ensure that all the learnings from the AIDC interventions were assessed:

- Continuous Improvement Initiatives,
- Quality Policy Deployment,
- Operator Involvement in Decision Making,
- Communication Strategies,
- Visual Management,
- Performance Measuring and
- Standard Work Procedures.

These measures are more company specific and assess how the respondent see or view their company. All questions pertaining to the above measures were structured. This means that they were either closed-ended or fixed questions. The researcher did not use any open-ended questions to avoid ambiguity and any non-specific responses.

This attitudinal study, as explained under aims above, was set out to understand the attitudes of companies towards continuous improvement initiatives. The weighted average score of 63.4 % was achieved. This figure was calculated as follows:

The study was divided into these seven elements of sustainability, which each contained a set of questions related to it. Each question within the elements was then weighted against the number of responses received. The weight was then averaged per sustainability element. By dividing the average weight of each element by 5 (the maximum possible score per question), we could get the percentages as shown in table 2.

5. FINDINGS

5.1 How was data analysed

The data collected was analyzed per questionnaire sections as outlined in section 4.4. The following indicators were used to determine the attitudes of the respondents towards each one of the topics discussed:

Table 1:	Defined	decision	indicators
----------	---------	----------	------------

0%-50%	Poor Sustainability
51%-75%	Sustainability can be Improved
76%-100%	Good Sustainability

These are the main indicators that were used to draw conclusion based on the collected data.

5.2 Findings from the Study

The study has found that South African companies in manufacturing sector are sustaining their improvement although there is room for improvement. Table 2 shows the findings of the study in terms of the 7 elements as discussed in section 4.4 above.



SUMMARY OF RESULTS			
01. CONTINUOUS IMPROVEMENT	61,4%		
02. QUALITY MANAGEMENT SYSTEM	60,8%		
03. OPERATOR FLEXIBILITY	69,8%		
04. COMMUNICATION	56,6%		
05. VISUAL MANAGEMENT	66,6%		
06. PERFORMANCE MEASUREMENT (INTERNAL)	58,8%		
07. STANDARD WORK	69,8%		
AVERAGE	63,4%		

Table 2: Results per sustainability element

This means that the majority of the respondents felt that although there is room for improvement, there is some level of sustainability of improvements. So according to this assessment, the current continuous improvement initiatives by government agencies have the minimum required impact, as discussed in table 2 above, but can largely be improved by restructuring and refining the current programmes further. According to the set indicators, these companies are really not rewduced in sustaining improvements but are also neither very good. This means that by redefining the current programmes, the sustainability of improvements within these companies will be achieved much better than the current findings.

Figure 4 shows a radar graph outlining the results of the study and also comparing these results to world class companies. As can be seen the red line shows the lower limit for the companies to be deemed as being able to maintain their continuous improvement while the grey line is showing the sustainability target for companies to be deemed as reaping the benefits of continuous improvements. The orange line is the maximum target that is available for companies to strive for. There are very few companies, if any at all, that are achieving the maximum benefits of continuous improvements. This means that with the gathered information as discussed in section 2.2, shows that generally companies around the world are not sustaining their improvements.

The blue line indicates the results of this survey. It denotes that South African manufacturing sector is meeting the minimum level of sustainability but is not reaping the benefits of improvement sustainability as yet.



Figure 4: Overall results of the study - All measures fall within the 50% and 75% range



According to literature, 90% had implemented continuous improvement projects; however, only 10% felt they were achieving the results they desired (Rink, 2005). The fact that Japanese companies are seeing more value in implementing and maintaining Kaizen projects (Cheser, 1998), is an indication that continuous improvement initiatives can be sustained and make the required impact. Also the fact that Bateman (2001), could classify the companies into the five (5) categories as described in section 2.2, means that there are companies that are sustaining their improvements by practicing a standardised organisational culture.

Since the element of sustainability is growing at a very fast pace, sustainability thinking is challenging businesses to change how they operate (Lubin and Asty, 2010). As a developing country, South Africa cannot afford to ignore the direction that the developed countries have taken in terms of the element of sustainability of continuous improvement. This is necessary to gain competitive advantage and to respond to the major trends that include growing of human population, rapid response to developing economies, both of which they put a major train to the resources and create unpredictable pricing in commodities.

In order for manufacturing companies in South Africa to grow the national GDP contribution, the current approach needs to change for the following reasons:

- Companies need their top management to commit fully in the continuous improvement programme, but current no assessment of such commitment at the beginning of the programme and therefore government's investments to these programmes end up being null and void.
- Supplier and Enterprise development programmes, especially when funded by the government, should be given to the companies that need them and to the companies that are realizing the need. Currently companies are being begged to come on board in order to meet the targets.
- Companies need to continue the process with little or no assistance at all from the AIDC. This means that after achieving at least 10% efficiency and productivity improvement at the end of such intervention, the company should continue instead of stopping just on 10% improvement.
- To ensure that companies have the required commitment, in order to grow their revenues and in turn growing the most important pillars of sustainability, such as Social, Economic development, and environmental protection.

5.3 Researcher's proposed way forward from the study

The proposed frame-work (figure 5 below) requires the AIDC SEDD to first do a thorough assessment before allowing companies into the longer term projects. These include working with these companies on a shortened version of the Rapid process improvement workshop and analyze the willingness of the company to change.



Proposed Frame work for SEDD Programmes					
DESCRIPTION	TIMING (Weeks)	PROPOSED FLOW	FULL SEDD PROGRAMMES		
Project Channing Project Chaner High level Project Plans Budget Planning Setting of Project KPI's Project Marketing Adverts on stake holder websites Brochure distribution to seminars, etc - Approaching potential Companies	Last two 8 Months of Q4 Last & First 6 Months of FY	START (Darle grevious PT)			
Company Assessments - Company need analysis - Company management's attitude to CI initiatives - Ability of company to allocate Human resource to project	Last two weeks of 2 First Month of FY	ASSESSMENTS	Total Productive		
Run Workshops: Give companies a case study.		ND VIS FOR VURLPROGRAMME VIS FOR WORKSHOPS	Quality		
to solve Visual Mangement workshops Shortened version of RPIW Autonomous Maintenance workshops Standardisation workshops Introduction to SS, 7Wastes, etc VSM / MIFD Workshops	First two weeks of 2 Second Month of FY	Run Mini Workshops	Rapid Process		
First progress Review - Assess project status vs plan - Advise companies where necessary	One week after the mini workshop	FILET FOLLOW-UP	Lean Mundschring Profess		
Second progress Beview - Assess project status vs plan - Advise companies where necessary - Assess whether company is ready for bigger intervention - Assess 1 company is showing interest - Measure percentage achievement from mini workshops	One week after the first follow up	SECOND TOLLOW UP	Cleaner Production		
Decision I Company can prove that they are interested (achieve 10% from mini workshop), subsidise them to take part in the IBPT and / or sign - If company is not interested, close the project by preparing a close out report	Two weeks 2 after Second follow up	Prepare Close out report Achieve 10%			
Recommend encouragement initiatives, sq. IBP - If company can prove that they are interested (achives 105 from mini workshop), chance to subsidise them to take part in the IBPT and then sign them in for a bigger project	Run Project as per Requirement till end of financial year	Industry Best Protective Tours (IBPT) Subaday			

Figure 5: Proposed frame work for SEDD programmes

This frame work also stipulates that if the company does not qualify in the assessment stage, recommendations will be made to that company explaining exactly what the requirements are. The company will then be left alone to address those challenges and will be considered in the future. This way only companies that are ready get to benefit from the government subsidized programmes.

It also ensures that companies continue the process with little or no assistance at all from the AIDC. This means that after achieving at least 10% efficiency and productivity improvement at the end of such intervention, the company will be re-evaluated to ensure that they continue instead of stopping just on 10% improvement.

6. CONCLUSION

Literature present that continues improvement sustainability has been studies in UK, Spain, US, etc however nothing has been done in South African environment, literature also indicated that factors that contribute to sustainability is the approached followed during the interventions.

The research findings indicate that generally South African companies in manufacturing sector, feel that they are sustaining continuous improvement initiatives to a minimum level, therefore new approach is needed to structure government funded initiatives differently in order to ensure sustainability is at a benchmark level of 75%.

Using the study results and the literature review finding, the research developed a new approach which can be followed to ensure high level of sustainability, this new approached will be tested in 2017 during new cluster programs and if find not suitable revision will be effected therefore.

7. AREAS FOR FURTHER RESEARCH

This study has only determined where South African manufacturing companies are in terms of sustaining their improvements. More research and factors need to be considered in order to fully contribute to the sustainability and growth of the industry through improvement initiatives. Although the impression and practical findings in this study are not far-reaching, it does provide a useful groundwork for further investigation of other relational elements essential in improving sustainability of improvement in manufacturing industry.



Future studies may explore the following:

- 1. This study measured the attitude of the respondents towards sustainability of continuous improvement initiatives, but did not physically measure to confirm the finding. It is the researcher's view that further comprehensive time series studies should be conducted with sampled individuals being observed over a period of time in order to provide more concrete evidence.
- 2. Since the sustainability concept is growing in manufacturing industry, a study should be conducted to compare the local industry with other developing countries world-wide.
- 3. Future studies could also look at frame work to educate South African individuals in management positions to drive lean manufacturing concepts in order to enhance sustainability of improvements by overcoming cultural challenges through continuous improvement sustainability.

ACKNOWLEDGMENT

A special word of thanks all the respondents for taking their valuable time to complete the questionnaire and to the AIDC for allowing me to take time to work on this research and also for making information available for me to conduct this research. I would also like to extend my word of thanks to Tshwane University of Technology (TUT) for assistance provided during this challenging times.

REFERENCES

- [1] Ali A. J., Md. Aminul Islam, Lim Poon Howe, 2012. A study of sustainability of continuous improvement in the manufacturing industries in Malaysia (Organisational self-assessment as a mediator.
- [2] AUTOMOTIVE INDUSTRY DEVELOP CENTRE. SEDD Templates, 2012. Internal document.
- [3] AUTOMOTIVE INDUSTRY DEVELOPMENT CENTRE. Supplier and Enterprise Development Department. 2014. Annual report, Pretoria: AIDC
- [4] AUTOMOTIVE INDUSTRY DEVELOPMENT CENTRE. Supplier and Enterprise Development Department. 2015. Annual report, Pretoria: AIDC.
- [5] Bateman, N, (in association with the SMMT Industry Forum), 2001, Sustainability ... a guide to ... Process Improvement... Cardiff University Business School, UK
- [6] Bateman, N, 2005, Sustainability: the elusive element of process improvement, International Journal of Operations & Production Management, Vol. 25 Iss: 3, pp.261 - 276, Cardiff University Business School, UK
- [7] Carey A. and Parsons S. (2009), Improving sustainability through the 21st century workplace, and IBM's vision of the office of the future, IBM Global Business Services
- [8] Cheser R., 1998. The effect of Japanese Kaizen on employee motivation in US manufacturing. Int J Organ Anal 6(3):197-217
- [9] Clough, G.W., J. Chameau, and C. Carmichael (2006), "Sustainability and the University." The Presidency. Winter 2006, pp. 30-40
- [10] Curtis, T. (2011). Share slides Presentation: www.shareslides.net/onimproving
- [11] Cusumano, M.A., 1985, The Japanese Automobile Industry: Technology and Management at Nissan and Toyota, Harvard University Press, Cambridge, MA.
- [12] Durward K. Sobek, II, A3 Reports: Tool for Process Improvement, Dept. of Mechanical and Industrial Engineering: Montana State University Bozeman, MT 59717-3800
- [13] Garcia, J. L., Rivera, D. G. (2013). Critical success factors for Kaizen implementation in Manufacturing Industries in Mexico.
- [14] GAUTENG GROWTH AND DEVELOPMENT. Automotive Industry Development Centre. 2012. Annual report. GGDA
- [15] GAUTENG GROWTH AND DEVELOPMENT. Automotive Industry Development Centre. 2014. Annual report. GGDA
- [16] Goodman, I., Kuniavsky, M., & Moed, A., (2012). Observing the user experience. Burlington, Massachusetts: Morgan Kaufmann.
- [17] Hicks, P.E., 1977, Introduction to Industrial Engineering and Management Science, McGraw-Hill, New York, p. 24.
- [18] Hignite, K., 2006. "Will Sustainability Take Root?" Business Officer. April 2006, pp. 12-22.
- [19] http://www.autofieldguide.com/columns/article.cfm, 2001."*Getting Lean Everywhere and Every Day*," [20] http://www.Lean.org/Lean/Community/Resources/thinkers-start.cfm, 2001."*Creating* a Future
- State," 2001. "History of Loop The
- [21] http://www.Optiprise.com/Company/Overview/History/Lean.htm, 2001. "History of Lean: The Evolution of a New Manufacturing Paradigm,"
- [22] Kumar, R. (2005). Research Methodology. 2nd Edition. London: SAGE Publications Ltd.
- [23] Lubin, D. A. and Esty, D. C. (2010), "The Sustainability Imperative", Harvard Business Review, May.
- [24] Malhotra, NK. (2004) Marketing research: an applied orientation (fourth edition). New Jersey: Pearson Education.



[25] Maskew Miller Longman, (2011), Longman South African School Dictionary with CD-ROM

- [26] McPhee, W. (2014), "A new Sustainability model: engaging the entire firm"
- [27] Ohno, T., 1988, Toyota Production System Beyond Large-Scale Production, Productivity Press, Portland, OR
- [28] Rajesh Kumar Singh, H.R. Murty, S.K. Gupta, A.K. Dikshit (2011), An overview of sustainability assessment methodologies, PE Sustainability Solutions, Bhilai Steel Plant, SAIL, and Centre for Environmental Science and Engineering, Indian Institute of Technology
- [29] Rink J. 2005 Lean can save American manufacturing. Reliable plant. [Online] http://www.rmdonovan/articles/pdf_2005/Lean_. Accessed 15 Sept 2011.
- [30] Satolo, E. G. and Simon, A. T. (2014), "Critical analysis of assessment methodologies for intraorganizational sustainability"
- [31] Saunders, Mark N. K, (2012). Research Methodology for Business Students, 6/E Financial Times Press -(See more at: http://catalogue.pearsoned.co.uk/educator/product/Research -Methods-for-Business-Students/9780273750758.page#sthash.8SBMe0Vq.dpuf, Accessed August 2015)
- [32] STATISTICS SOUTH AFRICA, Economic growth, 2015, prepared to compare economic growth of current year versus previous years: Done for South African manufacturing sector (29 June 2015) http://beta2.statssa.gov.za/?page_id=735&id=1
- [33] Suárez B, Miguel J, Castillo I (2011). The application of Kaizen in Mexican organizations: an empirical study. GCG 5(4):60-74 (in Spanish as quoted by Garcia, J. L., Rivera, D. G. (2013))
- [34] Tustin, D.H., Ligthelm, A.A., Martins, J.H. & Van Wyk, H.D. 2005. *Marketing Research in Practice*. Pretoria: UNISA Press.
- [35] Womack, J. And Jones, D. (1996): Lean Thinking. Simon & Schuster New York.
- [36] Womack, J., D.T. Jones, and D. Roos, 1990, *The Machine that Changed the World*: The Story of Lean Production, Harper Perennial, New York.
- [37] www.lo.dk/dba, The Sustainable Workplace An introduction to the concept and some good advice for those who get started